# BELLSOUTH® / CLEC Agreement

# Customer Name: Business Telecom, Inc.

Business Telecom, Inc.	2
Amendment 03/13/00	3
Amendment A 05/02/00	15
Amendment B 05/02/00	23
Amendment 10/20/00	42
BTIVolumeandTermResaleAmend	211
CoinPortAmendment	368
TN Permanent Rate Amendment	381
USOC Change Amendment	417

Note: This page is not part of the actual signed contract/amendment, but is present for record keeping purposes only.

The Customer Care Contract System is unable to display this rendered document at present. Please report this to the **TSC** (Technical Support Center) at: **404-230-2586** and provide to them the exact name of this document. We will make every effort to correct this situation as soon as possible.

We apologize for any inconvenience.

Thanks!

# Amendment to Interconnection Agreement by and between BellSouth Telecommunications, Inc. and Business Telecom, Inc. dated February 21, 2000

This Agreement refers to the Interconnection Agreement ("the Agreement") entered into by Business Telecom, Inc. ("BTI") and BellSouth Telecommunications, Inc. ("BellSouth") on February 21, 2000. This Amendment ("Amendment") is made by and between BTI and BellSouth and shall be deemed effective on the date executed by BTI and BellSouth.

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, BTI and BellSouth (individually, a "Party" and collectively, the "Parties") hereby covenant and agree as follows:

- 1. The Agreement is hereby amended to include a new Attachment 11 Disaster Recovery incorporated herein as Exhibit 1.
- 2. The Parties agree that all other provisions of the Agreement, dated February 21, 2000, shall remain in full force and effect.
- 3. The Parties further agree that either or both of the Parties is authorized to submit this Amendment to the Public Service Commission or other regulatory body having jurisdiction over the subject matter of this Amendment, for approval subject to Section 252(e) of the federal Telecommunications Act of 1996.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

Business Telecom, Inc.	BellSouth Telecommunications, Inc.
Signature on File	Signature on File
Signature	Signature
Anthony M. Copeland Name	Jerry D. Hendrix Name
EVP/General Counsel Title	Senior Director Title
10 March 00 Date	3/13/00 Date

# **EXHIBIT 1**

# Attachment 11 BellSouth Disaster Recovery Plan

# 2000 BELLSOUTH

# DISASTER RECOVERY PLANNING

For

**CLECS** 

# **CONTENTS**

	PAGE
1.0 Purpose	4
2.0 Single Point of Contact	4
3.0 Identifying the Problem	4
3.1 Site Control	5
3.2 Environmental Concerns	6
4.0 The Emergency Control Center (ECC)	6
5.0 Recovery Procedures	7
5.1 CLEC Outage	7
5.2 BellSouth Outage	7
5.2.1 Loss of Central Office	8
5.2.2 Loss of a Central Office with Serving Wire Center Functions	8
5.2.3 Loss of a Central Office with Tandem Functions	8
5.2.4 Loss of a Facility Hub	9
5.3 Combined Outage (CLEC and BellSouth Equipment	9
6.0 T1 Identification Procedures	9
7.0 Acronyms	10

### 1.0 PURPOSE

In the unlikely event of a disaster occurring that affects BellSouth's long-term ability to deliver traffic to a Competitive Local Exchange Carrier (CLEC), general procedures have been developed to hasten the recovery process. Since each location is different and could be affected by an assortment of potential problems, a detailed recovery plan is impractical. However, in the process of reviewing recovery activities for specific locations, some basic procedures emerge that appear to be common in most cases.

These general procedures should apply to any disaster that affects the delivery of traffic for an extended time period. Each CLEC will be given the same consideration during an outage and service will be restored as quickly as possible.

This document will cover the basic recovery procedures that would apply to every CLEC.

## 2.0 SINGLE POINT OF CONTACT

When a problem is experienced, regardless of the severity, the BellSouth Network Management Center (NMC) will observe traffic anomalies and begin monitoring the situation. Controls will be appropriately applied to insure the sanity of BellSouth's network; and, in the event that a switch or facility node is lost, the NMC will attempt to circumvent the failure using available reroutes.

BellSouth's NMC will remain in control of the restoration efforts until the problem has been identified as being a long-term outage. At that time, the NMC will contact BellSouth's Emergency Control Center (ECC) and relinquish control of the recovery efforts. Even though the ECC may take charge of the situation, the NMC will continue to monitor the circumstances and restore traffic as soon as damaged network elements are revitalized.

The telephone number for the BellSouth Network Management Center in Atlanta, as published in Telcordia's National Network Management Directory, is 404-321-2516.

### 3.0 IDENTIFYING THE PROBLEM

During the early stages of problem detection, the NMC will be able to tell which CLECs are affected by the catastrophe. Further analysis and/or first hand observation will determine if the disaster has affected CLEC equipment only; BellSouth equipment only or a combination. The initial restoration activity will be largely determined by the equipment that is affected.

Once the nature of the disaster is determined and after verifying the cause of the problem, the NMC will initiate reroutes and/or transfers that are jointly agreed upon by the affected CLECs' Network Management Center and the BellSouth NMC. The type and percentage of controls used will depend upon available network capacity. Controls necessary to stabilize the situation will be invoked and the NMC will attempt to re-establish as much traffic as possible.

For long term outages, recovery efforts will be coordinated by the Emergency Control Center (ECC). Traffic controls will continue to be applied by the NMC until facilities are re-established.

Version 1Q00:3/6/00

Subject to the Confidentiality and Proprietary agreement of the General Terms and Conditions of the Interconnection Agreement.

As equipment is made available for service, the ECC will instruct the NMC to begin removing the controls and allow traffic to resume.

### 3.1 SITE CONTROL

In the total loss of building use scenario, what likely exists will be a smoking pile of rubble. This rubble will contain many components that could be dangerous. It could also contain any personnel on the premises at the time of the disaster. For these reasons, the local fire marshal with the assistance of the police will control the site until the building is no longer a threat to surrounding properties and the companies have secured the site from the general public.

During this time, the majority owner of the building should be arranging for a demolition contractor to mobilize to the site with the primary objective of reaching the cable entrance facility for a damage assessment. The results of this assessment would then dictate immediate plans for restoration, both short term and permanent.

In a less catastrophic event, i.e., the building is still standing and the cable entrance facility is usable, the situation is more complex. The site will initially be controlled by local authorities until the threat to adjacent property has diminished. Once the site is returned to the control of the companies, the following events should occur.

An initial assessment of the main building infrastructure systems (mechanical, electrical, fire & life safety, elevators, and others) will establish building needs. Once these needs are determined, the majority owner should lead the building restoration efforts. There may be situations where the site will not be totally restored within the confines of the building. The companies must individually determine their needs and jointly assess the cost of permanent restoration to determine the overall plan of action.

Multiple restoration trailers from each company will result in the need for designated space and installation order. This layout and control is required to maximize the amount of restoration equipment that can be placed at the site, and the priority of placements.

Care must be taken in this planning to insure other restoration efforts have logistical access to the building. Major components of telephone and building equipment will need to be removed and replaced. A priority for this equipment must also be jointly established to facilitate overall site restoration. (Example: If the AC switchgear has sustained damage, this would be of the highest priority in order to regain power, lighting, and HVAC throughout the building.)

If the site will not accommodate the required restoration equipment, the companies would then need to quickly arrange with local authorities for street closures, rights of way or other possible options available.

## 3.2 ENVIRONMENTAL CONCERNS

In the worse case scenario, many environmental concerns must be addressed. Along with the police and fire marshal, the state environmental protection department will be on site to monitor the situation.

Items to be concerned with in a large central office building could include:

- 1. Emergency engine fuel supply. Damage to the standby equipment and the fuel handling equipment could have created "spill" conditions that have to be handled within state and federal regulations.
- 2. Asbestos containing materials that may be spread throughout the wreckage. Asbestos could be in many components of building, electrical, mechanical, outside plant distribution, and telephone systems.
- 3. Lead and acid. These materials could be present in potentially large quantities depending upon the extent of damage to the power room.
- 4. Mercury and other regulated compounds resident in telephone equipment.
- 5. Other compounds produced by the fire or heat.

Once a total loss event occurs at a large site, local authorities will control immediate clean up (water placed on the wreckage by the fire department) and site access.

At some point, the companies will become involved with local authorities in the overall planning associated with site clean up and restoration. Depending on the clean up approach taken, delays in the restoration of several hours to several days may occur.

In a less severe disaster, items listed above are more defined and can be addressed individually depending on the damage.

In each case, the majority owner should coordinate building and environmental restoration as well as maintain proper planning and site control.

## 4.0 THE EMERGENCY CONTROL CENTER (ECC)

The ECC is located in the Colonnade Building in Birmingham, Alabama. During an emergency, the ECC staff will convene a group of pre-selected experts to inventory the damage and initiate corrective actions. These experts have regional access to BellSouth's personnel and equipment and will assume control of the restoration activity anywhere in the nine-state area.

In the past, the ECC has been involve with restoration activities resulting from hurricanes, ice storms and floods. They have demonstrated their capabilities during these calamities as well as

Version 1Q00:3/6/00

Subject to the Confidentiality and Proprietary agreement of the General Terms and Conditions of the Interconnection Agreement.

during outages caused by human error or equipment failures. This group has an excellent record of restoring service as quickly as possible.

During a major disaster, the ECC may move emergency equipment to the affected location, direct recovery efforts of local personnel and coordinate service restoration activities with the CLECs. The ECC will attempt to restore service as quickly as possible using whatever means is available; leaving permanent solutions, such as the replacement of damaged buildings or equipment, for local personnel to administer.

Part of the ECC's responsibility, after temporary equipment is in place, is to support the NMC efforts to return service to the CLECs. Once service has been restored, the ECC will return control of the network to normal operational organizations. Any long-term changes required after service is restored will be made in an orderly fashion and will be conducted as normal activity.

### 5.0 RECOVERY PROCEDURES

The nature and severity of any disaster will influence the recovery procedures. One crucial factor in determining how BellSouth will proceed with restoration is whether or not BellSouth's equipment is incapacitated. Regardless of who's equipment is out of service, BellSouth will move as quickly as possible to aid with service recovery; however, the approach that will be taken may differ depending upon the location of the problem.

## 5.1 CLEC OUTAGE

For a problem limited to one CLEC (or a building with multiple CLECs), BellSouth has several options available for restoring service quickly. For those CLECs that have agreements with other CLECs, BellSouth can immediately start directing traffic to a provisional CLEC for completion. This alternative is dependent upon BellSouth having concurrence from the affected CLECs.

Whether or not the affected CLECs have requested a traffic transfer to another CLEC will not impact BellSouth's resolve to re-establish traffic to the original destination as quickly as possible.

## 5.2 BELLSOUTH OUTAGE

Because BellSouth's equipment has varying degrees of impact on the service provided to the CLECs, restoring service from damaged BellSouth equipment is different. The outage will probably impact a number of Carriers simultaneously. However, the ECC will be able to initiate immediate actions to correct the problem.

A disaster involving any of BellSouth's equipment locations could impact the CLECs, some more than others. A disaster at a Central Office (CO) would only impact the delivery of traffic to and from that one location, but the incident could affect many Carriers. If the Central Office is a Serving Wire Center (SWC), then traffic from the entire area to those Carriers served from that switch would also be impacted. If the switch functions as an Access Tandem, or there is a tandem in the building, traffic from every CO to every CLEC could be interrupted. A disaster that destroys a facility hub could disrupt various traffic flows, even though the switching equipment may be unaffected.

Version 1Q00:3/6/00

Subject to the Confidentiality and Proprietary agreement of the General Terms and Conditions of the Interconnection Agreement.

The NMC would be the first group to observe a problem involving BellSouth's equipment. Shortly after a disaster, the NMC will begin applying controls and finding re-routes for the completion of as much traffic as possible. These reroutes may involve delivering traffic to alternate Carriers upon receiving approval from the CLECs involved. In some cases, changes in translations will be required. If the outage is caused by the destruction of equipment, then the ECC will assume control of the restoration.

## 5.2.1 Loss of a Central Office

When BellSouth loses a Central Office, the ECC will

- a) Place specialists and emergency equipment on notice;
- b) Inventory the damage to determine what equipment and/or functions are lost;
- c) Move containerized emergency equipment and facility equipment to the stricken area, if necessary;
- d) Begin reconnecting service for Hospitals, Police and other emergency agencies; and
- e) Begin restoring service to CLECs and other customers.

# **5.2.2** Loss of a Central Office with Serving Wire Center Functions

The loss of a Central Office that also serves as a Serving Wire Center (SWC) will be restored as described in section 5.2.1.

### **5.2.3** Loss of a Central Office with Tandem Functions

When BellSouth loses a Central Office building that serves as an Access Tandem and as a SWC, the ECC will

- a) Place specialists and emergency equipment on notice;
- b) Inventory the damage to determine what equipment and/or functions are lost;
- c) Move containerized emergency equipment and facility equipment to the stricken area, if necessary;
- d) Begin reconnecting service for Hospitals, Police and other emergency agencies;
- e) Re-direct as much traffic as possible to the alternate access tandem (if available) for delivery to those CLECs utilizing a different location as a SWC;
- f) Begin aggregating traffic to a location near the damaged building. From this location, begin re-establishing trunk groups to the CLECs for the delivery of traffic normally found on the direct trunk groups. (This aggregation point may be the alternate access tandem location or another CO on a primary facility route.)

Version 1000:3/6/00

Subject to the Confidentiality and Proprietary agreement of the General Terms and Conditions of the Interconnection Agreement.

g) Begin restoring service to CLECs and other customers.

## 5.2.4 Loss of a Facility Hub

In the event that BellSouth loses a facility hub, the recovery process is much the same as above. Once the NMC has observed the problem and administered the appropriate controls, the ECC will assume authority for the repairs. The recovery effort will include

- a) Placing specialists and emergency equipment on notice;
- b) Inventorying the damage to determine what equipment and/or functions are lost;
- c) Moving containerized emergency equipment to the stricken area, if necessary;
- d) Reconnecting service for Hospitals, Police and other emergency agencies; and
- e) Restoring service to CLECs and other customers. If necessary, BellSouth will aggregate the traffic at another location and build temporary facilities. This alternative would be viable for a location that is destroyed and building repairs are required.

## 5.3 COMBINED OUTAGE (CLEC AND BELLSOUTH EQUIPMENT)

In some instances, a disaster may impact BellSouth's equipment as well as the CLECs'. This situation will be handled in much the same way as described in section 5.2.3. Since BellSouth and the CLECs will be utilizing temporary equipment, close coordination will be required.

## 6.0 T1 IDENTIFICATION PROCEDURES

During the restoration of service after a disaster, BellSouth may be forced to aggregate traffic for delivery to a CLEC. During this process, T1 traffic may be consolidated onto DS3s and may become unidentifiable to the Carrier. Because resources will be limited, BellSouth may be forced to "package" this traffic entirely differently then normally received by the CLECs. Therefore, a method for identifying the T1 traffic on the DS3s and providing the information to the Carriers is required.

## 7.0 ACRONYMS

CO - Central Office (BellSouth)

DS3 - Facility that carries 28 T1s (672 circuits)

ECC - Emergency Control Center (BellSouth)

CLEC - Competitive Local Exchange Carrier

NMC - Network Management Center

SWC - Serving Wire Center (BellSouth switch)

T1 - Facility that carries 24 circuits

## **Hurricane Information**

During a hurricane, BellSouth will make every effort to keep CLECs updated on the status of our network. Information centers will be set up throughout BellSouth Telecommunications. These centers are not intended to be used for escalations, but rather to keep the CLEC informed of network related issues, area damages and dispatch conditions, etc.

Hurricane-related information can also be found on line at <a href="http://www.interconnection.bellsouth.com/network/disaster/dis\_resp.htm">http://www.interconnection.bellsouth.com/network/disaster/dis\_resp.htm</a>. Information concerning Mechanized Disaster Reports can also be found at this website by clicking on CURRENT MDR REPORTS or by going directly to <a href="http://www.interconnection.bellsouth.com/network/disaster/mdrs.htm">http://www.interconnection.bellsouth.com/network/disaster/mdrs.htm</a>.

## **BST Disaster Management Plan**

BellSouth maintenance centers have geographical and redundant communication capabilities. In the event of a disaster removing any maintenance center from service another geographical center would assume maintenance responsibilities. The contact numbers will not change and the transfer will be transparent to the CLEC.

## **AMENDMENT**

TO

# THE INTERCONNECTION AGREEMENT BETWEEN BUSINESS TELECOM, INC. AND BELLSOUTH TELECOMMUNICATIONS, INC. DATED FEBRUARY 21, 2000

Pursuant to this Agreement (the "Amendment"), Business Telecom, Inc. ("BTI") and BellSouth Telecommunications, Inc. ("BellSouth") hereinafter referred to as the "Parties", hereby agree to amend the Interconnection Agreement between the Parties dated February 21, 2000 ("Interconnection Agreement").

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby covenant and agree as follows:

- 1. Attachment 2 of the Agreement is hereby amended to include the following terms and conditions for the provision of Port/Loop Combinations in the state of North Carolina.
  - 1. Port/Loop Combinations
  - 1.1 At BTI's request, BellSouth shall provide access to combinations of port and loop network elements, as set forth in the Rate section of the agreement, for those combinations that are currently combined in BellSouth's network, except as specified in Sections 1.1.1 and 1.1.2 below.
  - 1.1.1 BellSouth is not required to provide access to combinations of port and loop network elements in locations where BellSouth is not required to provide circuit switching.
  - 1.1.2 BellSouth is not required to provide circuit switching in Density Zone 1, as defined in 47 C.F.R. 69.123 as of January 1, 1999, of the Charlotte MSA to BTI if BTI's customer has 4 or more DS0 equivalent lines.

# 1.2. <u>Definition</u>

- 1.2.1 For purposes of this Amendment, references to "Currently Combined" network elements shall mean that such network elements are in fact already combined by BellSouth in the BellSouth network to provide service to a particular end user at a particular location.
- 1.2.2 Combinations of port and loop network elements provide local exchange service for the origination or termination of calls. Section 1.4 following provides the combinations of port and loop network elements that may be ordered by BTI when Currently Combined except in those locations where BellSouth is not

required to provide circuit switching, as set forth in Section 1.1.2 above.

- 1.3 Rates for Combinations of Loop and Port Network Elements
- 1.3.1 Rates for combinations of loop and port network elements are provided in Exhibit A of this Amendment.
- 2. Exhibit C of Attachment 2 of the Interconnection Agreement is hereby amended to include the North Carolina Public Utilities Commission's permanent UNE rates adopted in Docket No. P-100, Sub 133d on March 13, 2000 attached hereto as Exhibit A.
- 3. Exhibit A of Attachment 3 of the Interconnection Agreement is hereby amended to include the North Carolina Public Utilities Commission's permanent UNE rates adopted in Docket No. P-100, Sub 133d on March 13, 2000 attached hereto as Exhibit B.
- 4. Exhibit A of Attachment 5 of the Interconnection Agreement is hereby amended to include the North Carolina Public Utilities Commission's permanent UNE rates adopted in Docket No. P-100, Sub 133d on March 13, 2000 attached hereto as Exhibit C.
- 5. Exhibit A North Carolina of Attachment 4 of the Interconnection Agreement is hereby amended to add the following terms and conditions for the provision of Space Preparation in the state of North Carolina and to include the North Carolina Public Utilities Commission's permanent Collocation rates adopted in Docket No. P-100, Sub 133d on March 13, 2000 attached hereto as Exhibit D.

Space Preparation Fee in North Carolina. In North Carolina, the Space Preparation Fee is a monthly recurring charge, assessed per arrangement, per location, which is due beginning with the date on which BellSouth releases the Collocation Space for occupancy or on the date BTI first occupies the Collocation Space, which include survey, engineering, design and modification costs for network, building and support systems. In the event BTI opts for cageless space, the space preparation charge will be assessed based on the total floor space dedicated to BTI as described in Section 7.5. The Space Preparation Fee always consists of charges for Firm Order Processing, Central Office Modifications, Power, and Common Systems Modifications. The charge for Common Systems Modification will be on a per square foot basis for cageless and on a per cage basis for caged collocation. The charge for Power will be assessed per the nominal –48V DC ampere requirements specified by BTI on the Bona Fide Application.

5. Section 5 of the General Terms and Conditions of the Interconnection Agreement is hereby amended to include the following language:

- 5.3.1 Notwithstanding any provision(s) to the contrary, BTI agrees to provide to BellSouth, and BellSouth agrees to accept, BTI's Subscriber Listing Information (SLI) relating to BTI's customers in the geographic area(s) covered by this Interconnection Agreement. BTI authorizes BellSouth to release all such BTI SLI provided to BellSouth by BTI to qualifying third parties via either license agreement or BellSouth's Directory Publishers Database Service (DPDS), General Subscriber Services Tariff, Section A38.2, as the same may be amended from time to time. Such CLEC SLI shall be intermingled with BellSouth's own customer listings of any other CLEC that has authorized a similar release of SLI. Where necessary, BellSouth will use good faith efforts to obtain state commission approval of any necessary modifications to Section A38.2 of its tariff to provide for release of third party directory listings, including modifications regarding listings to be released pursuant to such tariff and BellSouth's liability therunder. BellSouth's obligation pursuant to this Section shall not arise in any particular state until the commission of such state has approved modifications to such tariff.
- 5.3.2 No compensation shall be paid to BTI for BellSouth's receipt of BTI SLI, or for the subsequent release to third parties of such SLI. In addition, to the extent BellSouth incurs costs to modify its systems to enable the release of BTI's SLI, or costs on an ongoing basis to administer the release of BTI SLI, BTI shall pay to BellSouth its proportionate share of the reasonable costs associated therewith.
- 5.3.3 BellSouth shall not be liable for the content or accuracy of any SLI provided by BTI under this Agreement. BTI shall indemnify, hold harmless and defend BellSouth from and against any damages, losses, liabilities, demands claims, suits, judgments, costs and expenses (including but not limited to reasonable attorneys' fees and expenses) arising from BellSouth's tariff obligations or otherwise and resulting from or arising out of any third party's claim of inaccurate BTI listings or use of the SLI provided pursuant to this Agreement. BellSouth shall forward to BTI any complaints received by BellSouth relating to the accuracy or quality of BTI listings.
- 5.3.4 Listings and subsequent updates will be released consistent with BellSouth system changes and/or update scheduling requirements.
- 6. All of the other provisions of the Interconnection Agreement dated February 21, 2000 shall remain unchanged and in full force and effect until the expiration date.

7. Either or both of the Parties is authorized to submit this Amendment to the appropriate regulatory agencies for approval subject to Section 252 (e) of the Federal Telecommunications Act of 1996.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

Busines	ss Telecom, Inc.	BellSo	uth Telecommunications, Inc.
Sig	nature on File		Signature on File
Ву:		By:	
\$	Signature		Signature
Name: <u> </u>	Anthony M. Copeland	Name:	Jerry D. Hendrix
Title: <u>l</u>	Executive VP/General Coun	selTitle:	Senior Director
Date:	April 27 2000	Date:	May 2, 2000

# EXHIBIT A: BELLSOUTH/BTI RATES – NORTH CAROLINA PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate	Non-Recurring
			(RC)	Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3,850.00
DE1G1			27.4	Φ1 COO OO
PE1CA	Subsequent Application Fee (Note	Per request	NA	\$1,600.00
	1)			Minimum
	Space Preparation Fee	Daniel G	¢1.57	
	Central Office Modification	Per sq. ft.	\$1.57	
	Common Systems Modification –	Per sq. ft.	\$3.26	
	Cageless Common Systems Modification	Damasas	\$110.70	
	Common Systems Modification –	Per cage	\$110.79	
	Caged Power	Per nominal –48v	\$5.76	
	Power	DC Amp	\$3.70	
		DC Allip		
	Space Enclosure (Note 2)			
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$102.76	NA
PE1CW	Welded Wire-mesh	Per add'l 50 sq. ft.	\$10.44	NA
TLICV	Weided Wife-mesh	Ter add 130 sq. it.	Ψ10.44	1171
PE1PJ	Floor Space	Per sq. ft.	\$3.45	NA
PE1BD	Cable Installation	Per cable	NA	\$2,305.00
PE1PM	Cable Support Structure	Per entrance cable	\$21.33	NA
	2			
DE1DI	Power	D	00.05	ICD
PE1PL	-48V DC Power	Per amp	\$6.65	ICB
PE1FB PE1FD	120V AC Power single phase*	Per breaker amp	\$5.50	ICB ICB
PE1FD PE1FE	240V AC Power single phase* 120V AC Power three phase*	Per breaker amp Per breaker amp	\$11.00 \$16.50	ICB ICB
PE1FG	277 AC Power three phase*	Per breaker amp	\$38.20	ICB ICB
FEIFU	211 AC Fower timee phase.	rei bleaket allip	\$36.20	ICB
	Cross Connects (Note 3)	Per cross connect		First/Add'1
PE1P2	2-wire	1 of offost conflict	\$0.32	\$41.78/\$39.23
PE1P4	4-wire		\$0.64	\$41.91/\$39.25
PE1P1	DS-1		\$2.34	\$71.02/\$51.08
PE1P3	DS-3		\$42.84	\$69.84/\$49.43
PE1F2	2-fiber		\$15.99	\$67.34/\$48.55
PE1F4	4-fiber		\$28.74	\$82.35/\$63.56

NORTH CAROLINA (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
	Co-Carrier Cross-Connect (Note			
	4)			
PE1ES	Fiber Cable Support Structure,	Per linear ft.	\$0.06	NA
Fiber	existing		40.00	
PE1DS	Copper or Coaxial Cable Support	Per linear ft.	\$0.03	NA
Copper	Structure, existing	D	NIA	ICD
(TBD)	Cable Support Structure	Per new	NA	ICB
	Construction, new	construction		
PE1AX	Security Access System Security	Per premises	\$52.00	
	System*	P	40-100	
	New Access Card Activation*	Per card		\$55.00
PE1AA	Administrative change, existing	Per card		\$35.00
	card*			
PE1AR	Replace lost or stolen card	Per card		\$250.00
PE1SR	Space Availability Report*	Don mamisos		\$550.00
PEISK	Space Availability Report*	Per premises		\$330.00
		requested		
	POT Bay Arrangements	Per cross-connect		
	Prior to 6/1/99			
PE1PE	2-Wire Cross-Connect		\$0.10	NA
PE1PF	4-Wire Cross-Connect		\$0.19	NA
PE1PG	DS1 Cross-Connect		\$0.79	NA
PE1PH	DS3 Cross-Connect		\$4.85	NA
PE1B2	2 Fiber Cross-Connect		\$39.67	NA
PE1B4	4 Fiber Cross-Connect		\$53.49	NA
	Security Escort	Per half hr./Add'l		
	Security Escort	half hr.		
PE1BT	Basic Time	11411 111.	NA	\$42.92/\$25.56
PE1OT	Overtime		NA NA	\$54.51/\$32.44
PE1PT	Premium Time		NA NA	\$66.10/\$39.32
12.11			1111	Ψου.10/Ψον.32
AEH	Additional Engineering Fee (Note	Per request, first		First/Add'1
	5)	half hr/add'l half hr.		Basic Time
				\$31.00/\$22.00
				Overtime
				\$37.00/\$26.00

# EXHIBIT A: BELLSOUTH/CLEC-1 RATES – NORTH CAROLINA PHYSICAL COLLOCATION (continued)

### Note(s):

N/A refers to rate elements which do not have a negotiated rate.

- (1) Subsequent Application Fee: BellSouth requires the submission of an Application Fee for modifications to an existing arrangement. However, when the modifications do not require BellSouth to expend capital, BellSouth will assess the Subsequent Application Fee in lieu of the Application Fee. Proposed modifications that could result in assessment of a Subsequent Application Fee would cause BellSouth to analyze the following but are not limited to: floor loading changes, changes to HVAC requirements, power requirement changes which may result in a power plant upgrade, environmental or safety requirements, or equipment relocation. Should the Subsequent Application Fee not be included as part of this Attachment, CLEC-1 will be assessed the full Application Fee for all subsequent activity for completed arrangements.
- (2) **Space Enclosure Fee**: The Space Enclosure Construction Fee is a monthly recurring fee, assessed per enclosure, per location with a one-hundred (100) square foot minimum enclosure. It recovers costs associated with providing an optional equipment arrangement enclosure, which include architectural and engineering fees, materials, and installation costs. The cost for additional square feet is applicable only when ordered with the first 100 square feet and must be requested in fifty (50) square foot increments. CLEC-1 may, at its option, arrange with a BellSouth Certified Contractor to construct the space enclosure in accordance with BellSouth's guidelines and specifications. In this event, the BellSouth Certified Contractor shall directly bill CLEC-1 for the space enclosure, and this fee shall not be applicable.
- (3) **Cross Connect:** The charges for cross connects are for orders placed electronically. Cross connect elements may also be ordered manually for which there is an additional charge per element.

	First/Additional
2-wire	\$46.53/\$43.98
4-wire	\$46.64/\$43.98
DS-1	\$75.72/\$55.78
DS-3	\$74.54/\$54.13

(4) **Co-Carrier Cross-Connect**. As stated in Section 5 of the Collocation Attachment, CLEC-1 may connect to other CLECs within the designated Premises in addition to, and not in lieu of, interconnection to BellSouth services and facilities. Where BellSouth must construct a cable rack structure to house the co-Carrier cross-connection, construction charges will be applied on an individual case basis as described in Section 5.6.1 of the Collocation Attachment. BellSouth shall provide an estimate of these charges in the Application Response. Where an existing cable rack structure is in place and has sufficient capacity to accommodate the co-Carrier cross-connection requested, the recurring charges as stated in this Exhibit A shall apply.

# EXHIBIT A: BELLSOUTH/CLEC-1 RATES – NORTH CAROLINA PHYSICAL COLLOCATION (continued)

(5) Additional Engineering Fee: BellSouth's additional engineering, and other labor costs associated with handling CLEC-1-requested modifications to requests in progress or augmentations for existing arrangements shall be recovered as Additional Engineering charges, under provisions in BellSouth's F.C.C. Number 1 Tariff, Sections 13.1 and 13.2. Should Additional Engineering rates not be included, CLEC-1 agrees not to make changes to collocation arrangement after a Bona Fide Firm Order is submitted.

# AMENDMENT

TO

# THE INTERCONNECTION AGREEMENT BETWEEN BUSINESS TELECOM, INC. AND BELLSOUTH TELECOMMUNICATIONS, INC. DATED FEBRUARY 21, 2000

Pursuant to this Agreement (the "Amendment"), Business Telecom, Inc. ("BTI") and BellSouth Telecommunications, Inc. ("BellSouth") hereinafter referred to as the "Parties", hereby agree to amend the Interconnection Agreement between the Parties dated February 21, 2000 ("Interconnection Agreement").

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby covenant and agree as follows:

- 1. Attachment 2 of the Agreement is hereby amended to include the following terms and conditions for the provision of Port/Loop Combinations in the state of North Carolina.
  - 1. Port/Loop Combinations
  - 1.1 At BTI's request, BellSouth shall provide access to combinations of port and loop network elements, as set forth in the Rate section of the agreement, for those combinations that are currently combined in BellSouth's network, except as specified in Sections 1.1.1 and 1.1.2 below.
  - 1.1.1 BellSouth is not required to provide access to combinations of port and loop network elements in locations where BellSouth is not required to provide circuit switching.
  - 1.1.2 BellSouth is not required to provide circuit switching in Density Zone 1, as defined in 47 C.F.R. 69.123 as of January 1, 1999, of the Charlotte MSA to BTI if BTI's customer has 4 or more DS0 equivalent lines.

## 1.2. Definition

- 1.2.1 For purposes of this Amendment, references to "Currently Combined" network elements shall mean that such network elements are in fact already combined by BellSouth in the BellSouth network to provide service to a particular end user at a particular location.
- 1.2.2 Combinations of port and loop network elements provide local exchange service for the origination or termination of calls.

  Section 1.4 following provides the combinations of port and loop network elements that may be ordered by BTI when Currently



Combined except in those locations where BellSouth is not required to provide circuit switching, as set forth in Section 1.1.2 above.

- 1.3 Rates for Combinations of Loop and Port Network Elements
- 1.3.1 Rates for combinations of loop and port network elements are provided in Exhibit A of this Amendment.
- 2. Exhibit C of Attachment 2 of the Interconnection Agreement is hereby amended to include the North Carolina Public Utilities Commission's permanent UNE rates adopted in Docket No. P-100, Sub 133d on March 13, 2000 attached hereto as Exhibit A.
- 3. Exhibit A of Attachment 3 of the Interconnection Agreement is hereby amended to include the North Carolina Public Utilities Commission's permanent UNE rates adopted in Docket No. P-100, Sub 133d on March 13, 2000 attached hereto as Exhibit B.
- 4. Exhibit A of Attachment 5 of the Interconnection Agreement is hereby amended to include the North Carolina Public Utilities Commission's permanent UNE rates adopted in Docket No. P-100, Sub 133d on March 13, 2000 attached hereto as Exhibit C.
- 5. Exhibit A North Carolina of Attachment 4 of the Interconnection Agreement is hereby amended to add the following terms and conditions for the provision of Space Preparation in the state of North Carolina and to include the North Carolina Public Utilities Commission's permanent Collocation rates adopted in Docket No. P-100, Sub 133d on March 13, 2000 attached hereto as Exhibit D.

Space Preparation Fee in North Carolina. In North Carolina, the Space Preparation Fee is a monthly recurring charge, assessed per arrangement, per location, which is due beginning with the date on which BellSouth releases the Collocation Space for occupancy or on the date BTI first occupies the Collocation Space, which include survey, engineering, design and modification costs for network, building and support systems. In the event BTI opts for cageless space, the space preparation charge will be assessed based on the total floor space dedicated to BTI as described in Section 7.5. The Space Preparation Fee always consists of charges for Firm Order Processing, Central Office Modifications, Power, and Common Systems Modifications. The charge for Common Systems Modification will be on a per square foot basis for cageless and on a per cage basis for caged collocation. The charge for Power will be assessed per the nominal –48V DC ampere requirements specified by BTI on the Bona Fide Application.

- 5. Section 5 of the General Terms and Conditions of the Interconnection Agreement is hereby amended to include the following language:
  - 5.3.1 Notwithstanding any provision(s) to the contrary, BTI agrees to provide to BellSouth, and BellSouth agrees to accept, BTI's Subscriber Listing Information (SLI) relating to BTI's customers in the geographic area(s) covered by this Interconnection Agreement. BTI authorizes BellSouth to release all such BTI SLI provided to BellSouth by BTI to qualifying third parties via either license agreement or BellSouth's Directory Publishers Database Service (DPDS), General Subscriber Services Tariff, Section A38.2, as the same may be amended from time to time. Such CLEC SLI shall be intermingled with BellSouth's own customer listings of any other CLEC that has authorized a similar release of SLI. Where necessary, BellSouth will use good faith efforts to obtain state commission approval of any necessary modifications to Section A38.2 of its tariff to provide for release of third party directory listings, including modifications regarding listings to be released pursuant to such tariff and BellSouth's liability therunder. BellSouth's obligation pursuant to this Section shall not arise in any particular state until the commission of such state has approved modifications to such tariff.
  - 5.3.2 No compensation shall be paid to BTI for BellSouth's receipt of BTI SLI, or for the subsequent release to third parties of such SLI. In addition, to the extent BellSouth incurs costs to modify its systems to enable the release of BTI's SLI, or costs on an ongoing basis to administer the release of BTI SLI, BTI shall pay to BellSouth its proportionate share of the reasonable costs associated therewith.
  - 5.3.3 BellSouth shall not be liable for the content or accuracy of any SLI provided by BTI under this Agreement. BTI shall indemnify, hold harmless and defend BellSouth from and against any damages, losses, liabilities, demands claims, suits, judgments, costs and expenses (including but not limited to reasonable attorneys' fees and expenses) arising from BellSouth's tariff obligations or otherwise and resulting from or arising out of any third party's claim of inaccurate BTI listings or use of the SLI provided pursuant to this Agreement. BellSouth shall forward to BTI any complaints received by BellSouth relating to the accuracy or quality of BTI listings.
  - 5.3.4 Listings and subsequent updates will be released consistent with BellSouth system changes and/or update scheduling requirements.

- 6. All of the other provisions of the Interconnection Agreement dated February 21, 2000 shall remain unchanged and in full force and effect until the expiration date.
- 7. Either or both of the Parties is authorized to submit this Amendment to the appropriate regulatory agencies for approval subject to Section 252 (e) of the Federal Telecommunications Act of 1996.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

Business Telecom, Inc.	BellSouth Telecommunications, Inc
By: Allow W. C. D.	Ву:
Signature	Signature
Name: Anthony M. Copeland	Name: /Jerry D. Hendrix
Title: Executive VP/General Counsel	Title: Senior Director
Date: April 27, 2000	Date: 5/2 (19)

# Exhibit A BELLSOUTH/BTI RATES NETWORK ELEMENTS AND OTHER SERVICES OSS/SWA BXX/DATABASES

OSS/SWA 8XX/DATABASES		
DESCRIPTION	USOC	NC
	0000	
8XX Access Ten Digit Screening (all types), per call (Note 2)	N/A	\$0,00050
Reservation Charge per 8XX number reserved		40.000
NRC - 1st	N8R1X	\$7.05
NRC - Addi'i	N8R1X	\$0.96
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'i	SOMAN	NA
Per 8XX # Established w/o POTS (w/8XX No.) Translations		
NRC - 1st	N/A	\$23.82
NRC - Addi'l	N/A	\$2.73
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$41.35
NRC - Incremental Charge - Manual Service Order - Add'i	SOMAN	NA
Per 8XX # Established with POTS Translations		
NRC - 1st	N8FTX	\$23.82
NRC - Addi'l	N8FTX	\$2.73
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$41.35
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA
Customized Area of Service per 8XX Number		
NRC - 1st	N8FCX	\$5.63
NRC - Addi'i	N8FCX	\$2.82
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA
Multiple Inter LATA Carrier Routing per Carrier Requested per 8XX#		
NRC - 1st	N8FMX	\$6.59
NRC - Addi'l	N8FMX	\$3.77
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA
NRC - Incremental Charge - Manual Service Order - Add't	SOMAN	NA
Change Charge per request NRC - 1st	NOTAV	4- 4-
NRC - Addi'l	N8FAX	\$8.01
NRC - Addit NRC - Incremental Charge - Manual Service Order - 1st	N8FAX	\$0.96
NRC - Incremental Charge - Manual Service Order - 1st  NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$26.94
Call Handling and Destination Features	SOMAN	NA NA
NRC - 1st	N8FDX	\$5.63
NRC - Add1	N8FDX	\$5.63 NA
TWO YAG	1401-07	INA
	grosses w	• • • • • • • • • • • • • • • • • • • •
LIDB Common Transport per query	OQT	\$0.0003
IDB Validation per query	OQU	\$0.013400
IDB Originating Point Code Establishment or Change		
NRC	TBD	\$62.26
NRC - Incremental Charge - Manual Service Order	SOMAN	\$26.94
\$6000000000000000000000000000000000000		
	4.1 A.	
CS7 Signaling Connection, per 56Kbps		\$18.22
NRC		\$278.02
NRC - Incremental Charge - Manual Service Order	SOMAN	\$41.35
CS7 Signaling Termination, per STP port per month		\$132.83
CS7 Signaling Usage, per Call Setup Message [(applicable when measurement and billing capability exists.)		\$0.00004
		#A COOC-
CCS7 Signaling Usage, per TCAP message [(applicable when measurement and billing capability exists.)		\$0.00009
CS7 Signaling Usage Surrogate, per 56Kbps, per LATA per month		\$338.98
COT Orginality Osage Surrogate, per Serups, per CATA per month	or the stage of the	\$30.86
perator Provided Call Handling per min - Using BST LIDB	N/A	\$1.20
Call Completion Access Termination Charge per call attempt	N/A N/A	\$1.20 NA
perator Provided Call Handling per min - Using Foreign LIDB	N/A	\$1.24
Call Completion Access Termination Charge per call attempt	N/A	NA
perator Provided Call Handling, per call	N/A	NA NA
ully Automated Call Handling per call - Using BST LIDB	N/A	\$0.11
ully Automated Call Handling per call - Using Foreign LiDB	N/A	\$0.12
		-
erification, per minute	N/A	\$1.15
erification and Emergency Interrupt, per minute	N/A	\$1.15
IRECTORY ASSISTANCE SERVICES		
Pirectory Assist Call Completion Access Svc (DACC), per call attempt	N/A	\$0.062
lumber Services Intercept per query	N/A	\$0.0110
Directory Assistance Access Service Calls, per call		\$0.260000
		60-0-
Pirectory Transport - Local Channel DS1, per month	N/A	\$35.68
NRC - 1st	N/A	\$534.48
NRC - Add'l	N/A	\$462.69
NRC - Incremental Charge-Manual Svc Order - NRC - 1st	SOMAN	\$86.15
NRC - Incremental Charge-Manual Svc Order - NRC -addl	TBD	\$1.77

Version 1Q00: 3/27/00

# EXHIBIT A BELLSOUTH/BIT RATES NETWORK ELEMENTS AND OTHER SERVICES OSS/SWA 8XX/DATABASES

DESCRIPTION	USOC	NC
Directory Transport - Dedicated DS1 Level Interoffice per mile per mo	N/A	\$0.5753
Directory Transport - Dedicated DS1 Level Interoffice per facility termination per mo	N/A	\$71.29
NRC - 1st	N/A	\$217.17
NRC - Add'l	N/A	\$163.75
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$38.07
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$38.07
Switched Common Transport per DA Access Service per call	N/A	\$0.00020
Switched Common Transport per DA Access Service per call per mile	N/A	\$0.00003
Access Tandem Switching per DA Access Service per call	N/A	\$0.0021
DA Interconnection, per DA Access Service Call	N/A	\$0.00
Directory Transport-Installation NRC, per trunk or signaling connection	N/A	
NRC - Manual Service Order - 1st	SOMAN	\$407.53
NRC - Manual Service Order - Add'l	SOMAN	\$10.98
REMANN PERSONAL OF THE FOREST STREET, AND THE STREET STREET, AND S		27.01122
Directory Assistance Database Service charge per listing	N/A	\$0.04460
Directory Assistance Database Service, per month	DBSOF	\$126.26
	55050	
Direct Access to Directory Assistance Service, per month	DBSDS	\$6,930.00
Direct Access to Directory Assistance Service, per query	DBSDA	\$0.0456
Direct Access to Directory Assistance Service, svc estab charge	DBSDE	
NRC	DBSDE	\$1,164.00
		ļ
AIN, per message	<u> </u>	<del> </del>
AIN - BellSouth AIN SMS Access Service	ļ	ļ
Service Establishment Charge, per state, initial set-up		
NRC	CAMSE	\$294.77
Port Connection - Dial/Shared Access		
NRC	CAMDP	\$86.94
Port Connection - ISDN Access		
NRC	CAM1P	\$86.94
User ID Codes - per User ID Code		
NRC	CAMAU	\$200.83
Security Card per User ID Code, initial or replacement		1
NRC	CAMRC	\$172.05
Storage, per unit (100Kb)	N/A	\$0.0023
Session per minute	N/A_	\$0.0791
C0. Performed Session, per minute		\$2.08
AIN - BellSouth AIN Toolkit Service		
AIN, Service Creation Tools		
Service Establishment Charge, per state, initial set-up		
NRC	BAPSC	\$290.05
Training Session, per customer	5450	00.000.00
NRC SN T AMERICA	BAPVX	\$8,363.00
Trigger Access Charge, per trigger, per DN, Term. Attempt	DADTT	670.70
NRC	BAPTT	\$72.76
Trigger Access Charge, per trigger per DN, Off-Hook Delay	54575	270.70
NRC	BAPTD	\$72.76
Trigger Access Charge, per trigger, per DN, Off-Hook Immediate	DADTM	\$70.70
NRC	BAPTM	\$72.76
Trigger Access Charge, per trigger, per DN, 10-Digit PODP	BAPTO	\$149.95
NRC	BAPIU	\$148.85
Trigger Access Charge, per trigger, per DN, CDP	BAPTC	\$149.95
NRC	BAFIC	\$148.80
Trigger Access Charge, per trigger, per DN, Feature Code	BAPTF	\$149.95
NRC	DAPIF	\$0.02
Query Charge, per query	ļ	\$0.02
Type 1 Node Charge, per AIN Toolkit Subscription, per node, per query	<del>                                     </del>	\$0.000
	N/A	\$1.45
Monthly Report - per AIN Toolkit Service Subscription	BAPMS	\$15.98
	BAPMS	\$71.80
I INPC	BAPLS	\$0.08
NRC Special Study - per AIN Toolkit Service Subscription		
Special Study - per AIN Toolkit Service Subscription		34/20
Special Study - per AIN Toolkit Service Subscription	BAPLS	\$47.20 \$15.90
Special Study - per AIN Toolkit Service Subscription NRC Call Event Report - per AIN Toolkit Service Subscription	BAPLS BAPDS	\$15.90
Special Study - per AIN Toolkit Service Subscription   NRC   Call Event Report - per AIN Toolkit Service Subscription   NRC	BAPLS BAPDS BAPDS	\$15.90 \$71.80
Special Study - per AIN Toolkit Service Subscription   NRC   Call Event Report - per AIN Toolkit Service Subscription   NRC   Call Event special Study - per AIN Toolkit Service Subscription	BAPLS BAPDS BAPDS BAPES	\$15.90 \$71.80 \$0.003
Special Study - per AIN Toolkit Service Subscription  NRC  Call Event Report - per AIN Toolkit Service Subscription  NRC  Call Event special Study - per AIN Toolkit Service Subscription  NRC	BAPLS BAPDS BAPDS	\$15.90 \$71.80
Special Study - per AIN Toolkit Service Subscription    NRC  Call Event Report - per AIN Toolkit Service Subscription    NRC  Call Event special Study - per AIN Toolkit Service Subscription    NRC  Operations Support Systems (OSS)	BAPLS BAPDS BAPDS BAPES BAPES	\$15.90 \$71.80 \$0.003 \$47.20
Special Study - per AIN Toolkit Service Subscription   NRC  Call Event Report - per AIN Toolkit Service Subscription   NRC  Call Event special Study - per AIN Toolkit Service Subscription   NRC  Operations Support Systems (OSS)   Recovery of incremental OSS costs, per CLP, per month	BAPLS BAPDS BAPDS BAPES BAPES TBD	\$15.90 \$71.80 \$0.003 \$47.20 \$305.00
Special Study - per AIN Toolkit Service Subscription  NRC  Call Event Report - per AIN Toolkit Service Subscription  NRC  Call Event special Study - per AIN Toolkit Service Subscription  NRC  Operations Support Systems (OSS)  Recovery of incremental OSS costs, per CLP, per month  OSS CLEC daily usage file (billing): recording, per message	BAPLS BAPDS BAPDS BAPES BAPES TBD TBD	\$15.90 \$71.80 \$0.003 \$47.20 \$305.00 \$0.0003
Special Study - per AiN Toolkit Service Subscription    NRC   Call Event Report - per AiN Toolkit Service Subscription   NRC   Call Event special Study - per AiN Toolkit Service Subscription   NRC   Operations Support Systems (OSS)   Recovery of incremental OSS costs, per CLP, per month   OSS CLEC daily usage file (billing): recording, per message   OSS CLEC daily usage file (billing): message distribution, per message	BAPLS BAPDS BAPDS BAPES BAPES BAPES TBD TBD TBD	\$15.90 \$71.80 \$0.003 \$47.20 \$305.00 \$0.0003
Special Study - per AIN Toolkit Service Subscription   NRC   Call Event Report - per AIN Toolkit Service Subscription   NRC   Call Event special Study - per AIN Toolkit Service Subscription   NRC   Operations Support Systems (OSS)   Recovery of incremental OSS costs, per CLP, per month   OSS CLEC daily usage file (billing): recording, per message	BAPLS BAPDS BAPDS BAPES BAPES TBD TBD	\$15.90 \$71.80 \$0.003 \$47.20 \$305.00 \$0.0003

11		
DESCRIPTION TO THE PROPERTY OF	USOC	NC
STEELS OF THE SECRET SECRET AND SECRET SECRE	UNDAX	\$1.01
ID per 2-Wire Analog VG Loop, Per Month NRC - 1st	UNDAX	\$1.42
The state of the s	UNDAX	\$1.42
NRC - Add'! NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76
ID per 4-Wire Analog VG Loop, Per Month	UNDAX	\$1.14
NRC - 1st	UNDAX	\$1.42
NRC - Add'l	UNDAX	\$1.42
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76
ID per 2-Wire ISDN Digital VG Loop, Per Month	UNDAX	\$1.01
NRC - 1st	UNDAX	\$1.42
NRC - Add'l	UNDAX	\$1.42
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76
ID per 2-Wire Asymmetrical Dig Subscriber Line (ADSL) Loop, Per Mo.	UNDAX	\$1.01
NRC - 1st	UNDAX	\$1.42
NRC - Add'i	UNDAX	\$1.42
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add't	SOMAN	\$12.76
ID per 2-Wire High Bit Rate Dig Subscriber Line (HDSL) Loop	UNDAX	\$1.01
NRC - 1st	UNDAX	\$1.42
NRC - Add'1	UNDAX	\$1.42
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76
ID per 4-Wire High Bit Rate Dig Subscriber Line (HDSL) Loop	UNDAX	\$1.14
NRC - 1st	UNDAX	\$1.42
NRC - Add'i	UNDAX	\$1.42
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76
ID per 4-Wire 58 Kbps Dig Grade Loop	UNDAX	\$1.14
NRC - 1st	UNDAX	\$1.42
NRC - Add'I	UNDAX	\$1.42
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76
ID per 4-Wire 64 Kbps Dig Grade Loop	UNDAX	\$1.14
NRC - 1st	UNDAX	\$1.42
NRC - Add'l	UNDAX	\$1.42
NRC - Incremental Charge - Manual Svc Ord - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Svc Ord - Add'l	SOMAN	\$12.76
onrecurring Charge - customer transfer, feature additions, changes (1)		NA
-Wire Analog VG Loop-SL1	LIEALO	\$15.88
RC - Statewide, per month	UEAL2 TBD	\$15.88 TBD
RC - Zone 1, per month (Note 2)	TBD	TBD
RC - Zone 2, per month (Note 2)	TBD	TBD
RC - Zone 3, per month (Note 2)	TBD	NA NA
RC - Zone 4, per month (Note 2)	UEAL2	\$57.99
		\$42.37
NRC - 1st		\$26.94
NRC - 1st NRC - Add'1	UEAL2	32n MA
NRC - 1st NRC - Add1 NRC - Incremental Charge - Manual Service Order - 1st	UEAL2 SOMAN	
NRC - 1st NRC - Add'1 NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'1	UEAL2 SOMAN SOMAN	\$12.76
NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Manual Order Coordination - 1st	UEAL2 SOMAN SOMAN TBD	\$12.76 \$61.38
NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Incremental Charge - Manual Service Order - Add'I NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - addI	UEAL2 SOMAN SOMAN TBD TBD	\$12.76 \$61.38 \$61.38
NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - addi NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	UEAL2 SOMAN SOMAN TBD TBD TBD	\$12.76 \$61.38
NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Incremental Charge - Manual Service Order - Add'I NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - addI NRC - Incremental Charge - Order Coordination - Time Specific (per LSR) NRC - Loop Make-Up	UEAL2 SOMAN SOMAN TBD TBD	\$12.76 \$61.38 \$61.38 \$45.34
NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Incremental Charge - Manual Service Order - Add'I NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - addI NRC - Incremental Charge - Order Coordination - Time Specific (per LSR) NRC - Loop Make-Up -Wire Analog VG Loop-SL2	UEAL2 SOMAN SOMAN TBD TBD TBD	\$12.76 \$61.38 \$61.38 \$45.34
NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - addI NRC - Incremental Charge - Order Coordination - Time Specific (per LSR) NRC - Loop Make-Up -Wire Analog VG Loop-SL2 RC - Statewide, per month	UEAL2 SOMAN SOMAN TBD TBD TBD UEANM	\$12.76 \$61.38 \$61.38 \$45.34 TBD
NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - addI NRC - Incremental Charge - Order Coordination - Time Specific (per LSR) NRC - Loop Make-Up Wire Analog VG Loop-SL2 RC - Statewide, per month RC - Zone 1, per month (Note 2)	UEAL2 SOMAN SOMAN TBD TBD TBD UEANM	\$12.76 \$61.38 \$61.38 \$45.34 TBD
NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Incremental Charge - Manual Service Order - Add'I NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - addI NRC - Incremental Charge - Order Coordination - Time Specific (per LSR) NRC - Loop Make-Up -Wire Analog VG Loop-SL2 RC - Statewide, per month RC - Zone 1, per month (Note 2) RC - Zone 2, per month (Note 2)	UEAL2 SOMAN SOMAN TBD TBD TBD UEANM UEAR2 TBD	\$12.76 \$61.38 \$61.38 \$61.38 \$45.34 TBD \$19.50 TBD
NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Incremental Charge - Manual Service Order - Add'I NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - addI NRC - Incremental Charge - Order Coordination - Time Specific (per LSR) NRC - Loop Make-Up Wire Analog VG Loop-SL2 RC - Statewide, per month RC - Zone 1, per month (Note 2) RC - Zone 2, per month (Note 2) RC - Zone 3, per month (Note 2)	UEAL2 SOMAN SOMAN TBD TBD TBD UEANM UEAR2 TBD TBD	\$12.76 \$61.38 \$61.38 \$45.34 TBD \$19.50 TBD
NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Incremental Charge - Manual Service Order - Add'I NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - addI NRC - Incremental Charge - Order Coordination - Time Specific (per LSR) NRC - Loop Make-Up	UEAL2 SOMAN SOMAN TBD TBD TBD UEANM  UEAR2 TBD TBD TBD TBD	\$12.76 \$61.38 \$61.38 \$45.34 TBD \$19.50 TBD TBD TBD
NRC - 1st  NRC - Add'I  NRC - Incremental Charge - Manual Service Order - 1st  NRC - Incremental Charge - Manual Service Order - Add'I  NRC - Incremental Charge - Manual Service Order - Add'I  NRC - Manual Order Coordination - 1st  NRC - Manual Order Coordination - addI  NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)  NRC - Loop Make-Up  -Wire Analog VG Loop-SL2  RC - Statewide, per month  RC - Zone 1, per month (Note 2)  RC - Zone 2, per month (Note 2)  RC - Zone 3, per month (Note 2)  RC - Zone 4, per month (Note 2)  NRC - Incremental Charge - Manual Service Order - 1st	UEAL2 SOMAN SOMAN TBD TBD TBD UEANM  UEAR2 TBD TBD TBD TBD	\$12.76 \$61.38 \$61.38 \$45.34 TBD \$19.50 TBD TBD TBD
NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Incremental Charge - Manual Service Order - Add'I NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - addI NRC - Incremental Charge - Order Coordination - Time Specific (per LSR) NRC - Loop Make-Up NRC - Loop Make-Up -Wire Analog VG Loop-SL2 RC - Statewide, per month RC - Zone 1, per month (Note 2) RC - Zone 2, per month (Note 2) RC - Zone 3, per month (Note 2) RC - Zone 4, per month (Note 2) NRC - 1st NRC - Add'I	UEAL2 SOMAN SOMAN TBD TBD TBD UEANM  UEAR2 TBD	\$12.76 \$61.38 \$61.38 \$45.34 TBD \$19.50 TBD TBD TBD NA \$142.97
NRC - 1st NRC - Add'! NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'! NRC - Incremental Charge - Manual Service Order - Add'! NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - add! NRC - Incremental Charge - Order Coordination - Time Specific (per LSR) NRC - Loop Make-Up -Wire Analog VG Loop-SL2 RC - Statewide, per month RC - Zone 1, per month (Note 2) RC - Zone 2, per month (Note 2) RC - Zone 3, per month (Note 2) RC - Zone 4, per month (Note 2) NRC - 1st NRC - Add'! NRC - Incremental Charge - Manual Service Order - 1st	UEAL2 SOMAN SOMAN TBD TBD TBD UEANM  UEAR2 TBD	\$12.76 \$61.38 \$61.38 \$45.34 TBD \$19.50 TBD TBD TBD NA \$142.97 \$106.56
NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Incremental Charge - Manual Service Order - Add'I NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - addI NRC - Incremental Charge - Order Coordination - Time Specific (per LSR) NRC - Loop Make-Up Wire Analog VG Loop-SL2 RC - Statewide, per month RC - Zone 1, per month (Note 2) RC - Zone 1, per month (Note 2) RC - Zone 3, per month (Note 2) RC - Zone 4, per month (Note 2) NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I	UEAL2 SOMAN SOMAN TBD TBD TBD UEANM  UEAR2 TBD TBD TBD TBD TBD UEAR2 TBD	\$12.76 \$61.38 \$61.38 \$45.34 TBD \$19.50 TBD TBD TBD NA \$142.97 \$106.56 \$26.94
NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Incremental Charge - Manual Service Order - Add'I NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - addI NRC - Incremental Charge - Order Coordination - Time Specific (per LSR) NRC - Loop Make-Up Wire Analog VG Loop-SL2 RC - Statewide, per month RC - Zone 1, per month (Note 2) RC - Zone 2, per month (Note 2) RC - Zone 3, per month (Note 2) RC - Zone 4, per month (Note 2) NRC - 1st NRC - 1st NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	UEAL2 SOMAN SOMAN TBD TBD TBD TBD UEANM  UEAR2 TBD TBD TBD TBD TBD UEAR2 TBD TBD TBD TBD TBD TBD UEAR2 UEAR2 SOMAN SOMAN	\$12.76 \$61.38 \$61.38 \$45.34 TBD \$19.50 TBD TBD TBD NA \$142.97 \$106.56 \$28.94
NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Incremental Charge - Manual Service Order - Add'I NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - addI NRC - Incremental Charge - Order Coordination - Time Specific (per LSR) NRC - Loop Make-Up Wire Analog VG Loop-SL2 RC - Statewide, per month RC - Zone 1, per month (Note 2) RC - Zone 2, per month (Note 2) RC - Zone 3, per month (Note 2) RC - Zone 4, per month (Note 2) NRC - 1st NRC - 1st NRC - Add'I NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'I NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	UEAL2 SOMAN SOMAN TBD TBD TBD UEANM  UEAR2 TBD	\$12.76 \$61.38 \$61.38 \$45.34 TBD \$19.50 TBD TBD NA \$142.97 \$106.56 \$26.94 \$12.76
NRC - 1st NRC - Add'l NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'l NRC - Manual Order Coordination - 1st NRC - Manual Order Coordination - addl NRC - Incremental Charge - Order Coordination - Time Specific (per LSR) NRC - Loop Make-Up Wire Analog VG Loop-St.2 RC - Statewide, per month RC - Zone 1, per month (Note 2) RC - Zone 2, per month (Note 2) RC - Zone 3, per month (Note 2) RC - Zone 4, per month (Note 2) NRC - 1st NRC - 1st NRC - Add'l NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'l	UEAL2 SOMAN SOMAN TBD TBD TBD TBD UEANM  UEAR2 TBD TBD TBD TBD TBD UEAR2 TBD TBD TBD TBD TBD TBD UEAR2 UEAR2 SOMAN SOMAN	\$12.76 \$61.38 \$61.38 \$45.34 TBD \$19.50 TBD TBD TBD NA \$142.97 \$106.56 \$28.94

Version 1Q00:3/27/00

DESCRIPTION	USOC	NC
RC - Zone 3, per month (Note 2)	TBD	TBD
RC - Zone 4, per month (Note 2)	TBD	NA
NRC - 1st	UEAL4	\$288.47
NRC - Add'l	UEAL4	\$237.45
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76
NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.34
2-Wire ISDN Digital Grade Loop		
RC - Statewide, per month	U1L2X	\$24.98
RC - Zone 1, per month (Note 2)	TBD	TBD
RC - Zone 2, per month (Note 2)	TBD	TBD
RC - Zone 3, per month (Note 2)	TBD	TBD
RC - Zone 4, per month (Note 2)	TBD	NA
NRC - 1st	U1L2X	\$325.91
NRC - Add'l	U1L2X	\$251.31
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'I	SOMAN	\$12.76
NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.34
2-Wire Asymmetrical Dig Subscriber Line (ADSL) Compatible Loop		
RC - Statewide, per month	UAL2X	\$14.60
RC - Zone 1, per month (Note 2)	TBD	TBD
RC - Zone 2, per month (Note 2)	TBD	TBD
RC - Zone 3, per month (Note 2)	TBD	TBD
RC - Zone 4, per month (Note 2)	TBD	NA.
NRC - 1st	UAL2X	\$504.90
NRC - Add'l	UAL2X	\$456.17
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76
NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.34
2-Wire High Bit Rate Dig Subscriber Line (HDSL) Compatible Loop		<b>V</b>
RC - Statewide, per month	UHL2X	\$11.98
RC - Zone 1, per month (Note 2)	TBD	TBD
RC - Zone 2, per month (Note 2)	TBD	TBD
RC - Zone 3, per month (Note 2)	TBD	TBD
IRC - Zone 4, per month (Note 2)	TBD	NA.
NRC - 1st	UHL2X	\$504.90
NRC - Add'I	UHL2X	\$456.17
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76
NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.34
4-Wire High Bit Rate Dig Subscriber Line (HDSL) Compatible Loop		• • • • • • • • • • • • • • • • • • • •
RC - Statewide, per month	UHL4X	\$13,97
RC - Zone 1, per month (Note 2)	TBD	TBD
RC - Zone 2, per month (Note 2)	TBD	TBD
RC - Zone 3, per month (Note 2)	TBD	TBD
RC - Zone 4, per month (Note 2)	TBD	NA.
NRC - 1st	UHL4X	\$531.35
NRC - Add'l	UHL4X	\$482.62
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76
NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.34
4-Wire DS1 Digital Loop		
RC - Statewide, per month	USLXX	\$62.78
RC - Zone 1, per month (Note 2)	TBD	TBD
RC - Zone 2, per month (Note 2)	TBD	TBD
RC - Zone 3, per month (Note 2)	TBD	TBD
RC - Zone 4, per month (Note 2)	TBD	NA
NRC - 1st	USLXX	\$714.84
NRC - Add'I	USLXX	\$421.47
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$42.19
NRC - Incremental Charge - Manual Service Order - Add'I	SOMAN	\$12.76
NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$48.31
4-Wire 56 Kbps Dig Grade Loop		
RC - Statewide, per month	UDL56	\$32.67
RC - Zone 1, per month (Note 2)	TBD	TBD
RC - Zone 2, per month (Note 2)	TBD	TBD
RC - Zone 3, per month (Note 2)	TBD	TBD
RC - Zone 4, per month (Note 2)	TBD	NA NA
NRC - 1st	UDL56	\$489.04
NRC - Add'1	UDL56	\$337.51
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76

DESCRIPTION	USOC	NC
INRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.34
Wire 64 Kbps Dig Grade Loop	00002	<b>\$</b> 40.04
RC - Statewide, per month	UDL64	\$32.67
RC - Zone 1, per month (Note 2)	TBD	TBD
RC - Zone 2, per month (Note 2)	TBD	TBD
RC - Zone 3, per month (Note 2)	TBD	TBD
RC - Zone 4, per month (Note 2)	TBD	NA
NRC - 1st	UDL64	\$489.04
NRC - Add'l	UDL64	\$337.51
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76
NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL.	\$45.34
op Channelization System - Digital Loop Carrier	TBD	NA COLE CA
RC - Loop Channelization System - Digital Loop Carrier	TBD	\$315.61
NRC-1st	TBD	\$426.48
NRC- Addl	TBD	\$103.42
NRC- Incremental Cost - Manaul Service Order- 1st	TBD TBD	\$42.19 \$12.76
NRC- Incremental Cost - Manaul Service Order- Addi	TBD	\$12.76 \$0.89
annel Interface 2-wire voice grade  NRC 1st	TBD	\$35.73
NRC 1st	TBD	\$35.73 \$35.49
INTO AUUT	180	430.48
2-Wire Analog Line Port (Res., Bus.), per month	UEPRL	\$2,19
NRC - 1st	UEPRL	\$21.60
NRC - Add'l	UEPRL	\$21.60
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76
4-Wire Analog VG Port, per month	UEP4A	\$8.69
NRC - 1st	UEP4A	\$21.69
NRC - Add'l	UEP4A	\$21.69
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.85
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.67
2-Wire DID Port, per month	UEPP2	\$12.36
NRC - 1st	UEPP2	\$81.84
NRC - Add'1	UEPP2	\$81.84
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76
4-Wire DS1 Port w/DID capability, per month	UEPDD	\$123.65
NRC - 1st	UEPDD UEPDD	\$116.59 \$69.92
NRC - Add'l NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$12.76
2-Wire ISDN Port(2) (3), per month	U1PMA	\$24.50
NRC - 1st	U1PMA	\$62.29
NRC - Add'l	U1PMA	\$62.29
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$55.30
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$55.30
4-Wire ISDN DS1 Port, per month	UEPEX	\$179.75
NRC - 1st	UEPEX	\$241.63
NRC - Add'l	UEPEX	\$241.63
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$53.89
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$53.89
2-Wire Analog Line Port (PBX), per month	UEPPC	\$2.18
NRC - 1st	UEPPC	\$21.60
NRC - Add'l	UEPPC	\$21.60
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76
Coin Port, per month		\$2.59
NRC - 1st		\$21.60
NRC - Add'l		\$21.60
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$26.94
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.76

	SCRIPTION	USOC	NC
	VERTICAL FEATURES		
Ļ	Local Switching Features offered with Port, Per month		\$0.89
ť	Three-Way Calling, per month NRC		\$1.51
╁	INC		<b>V1.01</b>
1	Customer Changeable Speed Calling, per month		\$0.17
Ť	INRC		\$1.51
t			· · · · · · · · · · · · · · · · · · ·
7	Call Waiting		\$0.09
Τ	NRC		\$1.51
I			
1	Remote Activation of Call Fordwarding, per month		\$0.85
Ļ	NRC		\$1.51
ļ,	A		\$0.01
4	Cancel Call Waiting, per month		\$1.51
╀	INC		<b>V1.51</b>
1	Automatic Caliback, per month		\$0.66
ť	INRC		\$1.51
t			
1	Automatic Recall, per month		\$0.29
Ι	NRC		\$1.51
Ι			
9	Calling Number Delivery, per month		\$0.33
1	NRC		\$1.51
Ļ	O III - N - b - D II - D III - D II		\$0.02
4	Calling Number Delivery Blocking, per month NRC		\$0.02
+	NRC .		\$1.01
+	Customer Originated Trace, per month		\$0.14
ť	NRC		\$1.51
t	1870		
1	Selective Call Rejection, per month		\$0.13
T	NRC		\$1.51
T			
1	Selective Call Forwarding, per month		\$0.28
1	NRC		\$1.51
4			<b>#0.00</b>
1	Selective Call Acceptance, per month		\$0.33 \$1.51
+	NRC		\$1.51
+	Multiline Hunt Service (Rotary)		
١	Service per line, (in addition to port) , per month		\$0.14
ť	NRC		\$1.51
t			
1	Call Forwarding Variable, per month		\$0.10
Ι	NRC		\$1.51
I			
10	Call Forwarding Busy Line, per month		\$0.08
+	NRC		\$1.51
Ŧ,	Call Converting Don't Anguar All Calle you month		\$0.09
4	Call Forwarding Don't Answer All Calls, per month NRC		\$1.51
+			<b>\$1.01</b>
١,	Remote Call Forwarding, per month		\$0.95
Ť	NRC		\$1.51
1			
Ī	Call Transfer, per month		\$0.14
Ţ	NRC		\$1.51
1			#n 45
#	Call Hold, per month		\$0.15 \$1.51
+	NRC		\$1.51
+	Toll Restricted Service, per month		\$0.10
+	NRC		\$1.51
+			<del></del>
†	Message Waiting Indicator – Stutter Dial Tone, per month	1	\$0.03
ť	NRC		\$1.51
1			2
1	Anonymous Call Rejection, per month		\$1.29 \$1.51

LESCRIPTION	USOC	NC		
DESCRIPTION USOC Shared Call Appearances of a DN, per month				
INRC		\$0.29 \$1.47		
INC	<del> </del>			
Multiple Call Appearances, per month	<del>                                     </del>	\$0.07		
INRC		\$1.47		
ISDN Bridged Call Exclusion, per month		\$0.0011		
INRC		\$1.47		
Call by Call Access, per month		\$19.83		
INRC		\$33.33		
Privacy Release, per month		\$0.0041		
NRC		\$1.51		
Muiti Appearance Directory Number Calls, per month		\$0.13		
NRC		\$1.51		
	<del> </del>	L		
Make Set Busy, per month		\$0.0020		
NRC	<del> </del>	\$1.51		
		00.00		
Teen Service (Res. Dist. Alerting Service), per month		\$0.26		
NRC		\$1.51		
NRC - Disconnect		NA CO. CO.		
Code Restriction and Diversion, per month		\$0.09 \$1.51		
NRC		\$1.51		
		\$0.09		
Call Park, per month		\$1.51		
NRC	<del></del>	\$1.51		
		\$0.14		
Automatic Line, per month		\$1.51		
ISDN Message Waiting Indication-Lamp, per month	<del> </del>	\$0,0107		
INRC	<del> </del>	\$1.47		
INC		<b>V</b>		
ISDN Feature Function Buttons		NA		
NRC		\$1.51		
Subsequent Ordering Charge – (per order, per line)		NA		
NRC - 1st		\$5.42		
NRC - Add'l		\$0.95		
NRC - Incremental Charge - Manual Service Order - 1st		\$1.89		
NRC - Incremental Charge - Manual Service Order - Add'l		NA		
OTES:	<u> </u>	<u> </u>		
In states where a specific NRC for customer transfer, feature additions and changes is not stated, the applicable NRC from the appropriate tariff applies.				
2 Deaveraged Loop Rates by Zone will be available effective May 1, 2000. The status of rates shown by state is as follows:				
North Carolina - To be decided.				
3 All rates are interim and subject to true-up.				

ESCRIPTION	USOC	NC
	REPORT OF THE PROPERTY OF THE	
ommon (Shared) Transport		
Common (Shared) Transport per mile per mou	N/A	\$0.00001
Common (Shared) Transport Facilities Termination per mou	N/A	\$0.00034
	1 A 1 A 1 A 1 A 1 A 1 A 1 A 1 A 1 A 1 A	
Interoffice Channel - Dedicated Transport - 2-Wire VG - per mile	1L5XX	\$0.0282
Interoffice Channel - Dedicated Transport - 2-Wire VG - facility termination per month	U1TV2	\$18.00
NRC - 1st	U1TV2	\$137.48
NRC - Add'l	U1TV2	\$52.58
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$38.07
NRC - Incremental Charge - Manual Service Order - Add'I	SOMAC	\$38.07
teroffice Channel - Dedicated Transport - DS0 - 58/64 KBPS		
I Unteroffice Channel - Dedicated Transport - DS0 - per mile per month	1L5XX	\$0.0282
Interoffice Channel - Dedicated Transport - DS0 - facility termination per month	U1TD6	\$17.40
NRC - 1st	U1TD6	\$137.48
NRC - Add'l	U1TD6	\$52.58
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$38.07
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$38.07
teroffice Channel - Dedicated Transport - DS1		
Interoffice Channel - Dedicated Transport - DS1- per mile per month	1L5XX	\$0.5753
Interoffice Channel - Dedicated Transport - DS1 facility termination per month	U1TF1	\$71.29
NRC - 1st	U1TF1	\$217.17
NRC - Add'l	U1TF1	\$163.75
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$38.07
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$38.07
teroffice Channel - Dedicated Transport - DS3		
Interoffice Channel - Dedicated Transport - DS3 - per mile per month	1L5XX	\$12.98
Interoffice Channel - Dedicated Transport - DS3 - facility termination per month	U1TF3	\$720.38
NRC - 1st	U1TF3	\$794.94
NRC - Add'I	U1TF3	\$579.55
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$91.26
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$91.26
ocal Channel - Dedicated Transport		,
ocal Channel - Dedicated Transport - 2-Wire VG		
Monthly Recurring	ULDV2	\$14.82
NRC - 1st	ULDV2	\$553.80
NRC - Add'I	ULDV2	\$86.69
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$42.17
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$12.76
ocal Channel - Dedicated Transport - 4-Wire VG		<u> </u>
Monthly Recurring	ULDD6	\$15.87
NRC - 1st	ULDD6	\$562.23
NRC - Add'l	ULDD6	\$92.67
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$42.17
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$12.76
ocal Channel - Dedicated Transport - DS1		1
Monthly Recurring	TMECS	\$35.68
NRC - 1st	TMECS	\$534.48
NRC - Add'l	TMECS	\$462.69
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$86.15
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$1.77
ocal Channel - Dedicated Transport - DS3		<u> </u>
Monthly Recurring	ULDF3	\$498.87
NRC - 1st	ULDF3	\$562.25
NRC - Add'l	ULDF3	\$527.88
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$56.25
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$56.25

# Exhibit A BELLSOUTH/BTI RATES NETWORK ELEMENTS AND OTHER SERVICES LOOP/PORT COMBINATIONS

LOOP/PORT COMBINATIONS		
CRIPTION	USOC	NC NC
UNBUNDLED LOOP BILLING USOC (REQUIRES ONE PER PORT)	UEPLX	
		<u> </u>
LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	LNPCX	
rently Combined		
customers with less than 4 DS0 Equivalent		
2-Wire Voice Grade Loop with 2-Wire Line Port		
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Statewide	TBD	\$16.46
RC - 2- Wire Voice Grade Loop	UEPLX	Note 1
RC - Exchange Port - 2-Wire Line Port	TBD	Note 1
NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st	USACC	\$2.77
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'i	USACC	\$0.40
	TDD	\$1.42
NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update  NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Incremental Manual Service	TBD	\$1.42
	TBD	\$10.27
Order Charles and	180	\$10.27
2- Wire Voice Grade Loop with 2 -Wire DID Trunk Port	TOD	£22.70
RC- 2 Wire Voice Grade Loop with 2 - Wire Line Port	TBD	\$23.79
NRC- 2- Wire Voice Grade Loop with 2- Wire Line Port - 1st		\$13.26
NRC- 2- Wire Voice Grade Loop with 2- Wire Line Port - Addl	TBD	\$8.39
NRC- 2- Wire Voice Grade Loop with 2- Wire Line Port - Incremental Cost- Manual Service Order - 1st	TBD	\$53.89
NRC- 2- Wire Voice Grade Loop with 2- Wire Line Port - Incremental Cost- Manual Service Order - Addl	TBD	\$11.34
2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Port		4:0.15
RC- 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Port	TBD	\$43.45
NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Port - 1st conversion	USACB	\$174.35
NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Port - Add'l conversion	USACB	\$174.35
NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Port - Non Feature Subsequent Activity	USASB	\$286.15
		ļ
4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port		2011 70
RC - 4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port	TBD	\$241.72
NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port - 1st	TBD	\$481.51
NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port - Addl	TBD	\$481.51
NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port - Subsequent Channel Activation - Per		
Channel	TBD	\$36.92
NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port - Subsequent Inward/2way Telephone		
Numbers	TBD	\$1.17
NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port - Subsequent Outward Telephone Numbe	rs TBD	\$28.17
NEW AND ROLD IN THE STREET CONTROL OF THE ST	700	<b>\$50.00</b>
NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port - Subsequent Inward Telephone Numbers	TBD	\$56.33
NIDO A MAN DOA DISHALL WAS A MAN IODN DOA DISHALT IN DAT ON THE COMMENT OF	TOD	\$255.25
NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port - Subsequent Service Order Per Order	TBD	\$233.25
4 Miles DC4 Digital I can with 4 Miles DID Trunk Bort		
4 - Wire DS1 Digital Loop with 4 - Wire DID Trunk Port	TBD	\$186.23
RC - 4 - Wire DS1 Digital Loop with 4 - Wire DID Trunk Port		\$490.38
NRC -4 - Wire DS1 Digital Loop with 4 - Wire DID Trunk Port - 1st	TBD	
NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port - Addl	TBD	\$490.38
NRC -4 - Wire DS1 Digital Loop with 4 - Wire DIDTrunk Port - Subsequent Channel Activation - Per Channel	TBD	\$146.91
NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port - Subsequent Telephone Numbers	TBD	\$120.96
NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port - Subsequent Signaling Changes	TBD	\$29.65
	1	
	700	
NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port - Subsequent Service Order Per Order	ТВО	\$127.63
Note 1 - In the absence of ordered rates by a State Commission, the rates for Currently Combined combinations of	ТВО	\$127.63
	TBD	\$127.63

#### Exhibit B

#### BELLSOUTH/BTI RATES LOCAL INTERCONNECTION

	1			
	usoc	NC		
DESCRIPTION CONTRACTOR AND TERMINATION	0300	110		
OCAL INTERCONNECTION (CALL TRANSPORT AND TERMINATION)				
End Office Switching,		00.0047		
per mou (includes E.O. trunk port shared per MOU)	NA NA	\$0.0017		
Tandem Switching				
per mou (includes tandem trunk port - shared per MOU)	N/A	\$0.0009		
INTEROFFICE TRANSPORT				
Common (Shared) Transport				
Common (Shared) Transport per mile per mou	N/A	\$0.00001		
Common (Shared) Transport Facilities Termination per mou	N/A	\$0,00034		
	147	40.00001		
Interoffice Channel Transport - Dedicated - VG	41.575	\$0.0282		
Interoffice Transport - Dedicated - 2-Wire VG - per mile	1L5XF	*****		
Interoffice Transport - Dedicated - 2-Wire VG - facilities termination per month	1L5XF	\$18.00		
NRC - 1st	1L5XF	\$137.48		
NRC - Add'l	1L5XF	\$52.58		
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$38.07		
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$38.07	1	
Interoffice Channel Transport - Dedicated - DS0 - 56/64 KBPS				
Interoffice Cramer Transport - Dedicated - DS0 - per mile per month	1L5XK	\$0.0282		
	1L5XK	\$17,40		
Interoffice Transport - Dedicated - DS0 - facility termination per month	1L5XK	\$137.48		
NRC - 1st				
NRC - Add'I	1L5XK	\$52.58		
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$38.07		
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$38.07		
Interoffice Channel Transport - Dedicated - DS1				
Interoffice Transport - Dedicated - DS1 - per mile per month	1L5XL	\$0.5753		
Interoffice Transport - Dedicated - DS1 - facility termination per month	1L5XL	\$71.29		
NRC - 1st	1L5XL	\$217.17		
	1L5XL	\$163.75		
NRC - Add'l	SOMAC	\$38.07		
NRC - Incremental Charge - Manual Service Order - 1st				
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$38.07		
Interoffice Channel Transport - Dedicated - DS3				
Interoffice Transport - Dedicated - DS3 - per mile per month	1L5XM	\$12.98		
Interoffice Transport - Dedicated - DS3 - facility termination per month	1L5XM	\$720.38		
NRC - 1st	1L5XM	\$794.94		
NRC - Add'l	1L5XM	\$579.55		
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$91.26		
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$91.26		
	COMPAG	401.20		
Local Channel - Dedicated				
Local Channel - Dedicated - 2-Wire VG		211.00		
Monthly Recurring	TEFV2	\$14.82		
NRC - 1st	TEFV2	<b>\$</b> 553.80		
NRC - Add'l	TEFV2	\$86.69		
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$42.17		
NRC - Incremental Charge - Manual Service Order - Add'I	SOMAC	\$12.76		
Local Channel - Dedicated - 4-Wire VG				
Monthly Recurring	TEFV4	\$15.87		
NRC - 1st	TEFV4	\$562.23		
NRC - 1st	TEFV4	\$92.67	<del></del>	
	SOMAC	\$42.17		
NRC - Incremental Charge - Manual Service Order - 1st				
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$12.76	<del></del>	
Local Channel - Dedicated - DS1			<b></b>	
Monthly Recurring	TEFHG	\$35.68		
NRC - 1st	TEFHG	\$534.48		
NRC - Add'l	TEFHG	\$462.69		
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$42.17		
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$12.76		
Local Channel - Dedicated - DS3				
Monthly Recurring	TEFHJ	\$498.87		
	TEFHJ	\$562.25	<del> </del>	
NRC - 1st	TEFHJ	\$527.88	<del></del>	
NRC - Add'I			<del>                                     </del>	
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$56.25		
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$56.25		
NOTES:				
If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable Bell	South tariff or as negotiated	d by the parties upor	n request by eith	er party.
In the table to the surface of the contract, the fact for the appeals do not need on the application of the surface of the sur	I			
		<u> </u>	<del> </del>	

## Exhibit C BELLSOUTH/BTI RATES SERVICE PROVIDER NUMBER PORTABILITY

ESCRIPTION	USOC	NC
	<u> </u>	
CF, per number ported (Business Line)	TNPBL	\$1.66
NRC -	TNPBL	\$0.71
RCF, per number ported (Residence Line)	TNPRL	\$1.66
INRC	TNPRL	\$0.71
RCF, per additional path	N/A	\$0.32
	(++) Bus = TNPBD	
RCF, per service order, per location	Res = TNPRD	
NRC - 1st	TNP++	\$2.73
NRC - Add'i	TNP++	\$2.73
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$43.07
NPC - incremental Chame - Manual Service Order - Add'l	SOMAN	\$43.07
Des 200 by 12.18 1 12.48 2.48 2.11 12.11 12.11 12.11 12.11	3.	
DID per number ported, Residence - NRC	TNPDR	\$2.25
DID per number ported, Business - NRC	TNPDB	\$2.25
DID per service order, per location		
INRC - 1st	TNPRD	\$2.73
NRC - Add'I	TNPRO	\$2.73
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$43.07
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$43.07
DID, per trunk termination, initial	TNPT2	\$11.43
INRC	TNPT2	\$217.88
DID, per trunk termination, subsequent	TNPT2	\$11.43
NRC	TNPT2	\$73.56
NOTES:		
If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable	te BellSouth tariff or as negotiated by the Parties upon request by e	ither Party.

## EXHIBIT A: BELLSOUTH/BTI RATES – NORTH CAROLINA PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate	Non-Recurring
DE15		Day we want	(RC)	Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3,850.00
DELCA	G.1 A Liestian Fee (Note	Per request	NA	\$1,600.00
PE1CA	Subsequent Application Fee (Note	rerrequest	INA	Minimum
	1)			Willimmum
	Space Preparation Fee			
	Central Office Modification	Per sq. ft.	\$1.57	
	Common Systems Modification –	Per sq. ft.	\$3.26	
	Cageless	1 01 54. 16.	1	
	Common Systems Modification –	Per cage	\$110.79	
	Caged	1010050		
	Power	Per nominal –48v	\$5.76	
		DC Amp		
	Space Enclosure (Note 2)			
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$102.76	NA
PE1CW	Welded Wire-mesh	Per add'1 50 sq. ft.	\$10.44	NA
PE1PJ	Floor Space	Per sq. ft.	\$3.45	NA
22122		D1.1.	NTA	\$2,305.00
PE1BD	Cable Installation	Per cable	NA	\$2,303.00
PE1PM	Cable Support Structure	Per entrance cable	\$21.33	NA
IEIIWI	Cable Support Structure	T CT CHITAINCE CASIC	Ψ21.33	
	Power			
PE1PL	-48V DC Power	Per amp	\$6.65	ICB
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.50	ICB
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.00	ICB
PE1FE	120V AC Power three phase*	Per breaker amp	\$16.50	ICB
PE1FG	277 AC Power three phase*	Per breaker amp	\$38.20	ICB
				7m* / A 4 4A4
	Cross Connects (Note 3)	Per cross connect		First/Add'l
PE1P2	2-wire		\$0.32	\$41.78/\$39.23
PE1P4	4-wire		\$0.64	\$41.91/\$39.25
PE1P1	DS-1		\$2.34	\$71.02/\$51.08
PE1P3	DS-3		\$42.84	\$69.84/\$49.43
PE1F2	2-fiber		\$15.99	\$67.34/\$48.55
PE1F4	4-fiber		\$28.74	\$82.35/\$63.56

NORTH CAROLINA (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
	Co-Carrier Cross-Connect (Note			
	4)			
PE1ES	Fiber Cable Support Structure,	Per linear ft.	\$0.06	NA
Fiber	existing			27.
PE1DS	Copper or Coaxial Cable Support	Per linear ft.	\$0.03	NA
Copper	Structure, existing			.co
(TBD)	Cable Support Structure	Per new	NA	ICB
	Construction, new	construction		
PE1AX	Security Access System Security	Per premises	\$52.00	
IDIAA	System*	Tor promises	<b>Q</b> 00.00	
	New Access Card Activation*	Per card		\$55.00
PE1AA	Administrative change, existing	Per card		\$35.00
ILIAA	card*	1010010		
PE1AR	Replace lost or stolen card	Per card		\$250.00
TEITAC	Replace less of sterem sure			
PE1SR	Space Availability Report*	Per premises		\$550.00
		requested		
	POT Bay Arrangements	Per cross-connect		
	Prior to 6/1/99		60.10	NA
PE1PE	2-Wire Cross-Connect		\$0.10	NA NA
PE1PF	4-Wire Cross-Connect		\$0.19	NA NA
PE1PG	DS1 Cross-Connect		\$0.79	NA NA
PE1PH	DS3 Cross-Connect		\$4.85	NA NA
PE1B2	2 Fiber Cross-Connect		\$39.67	
PE1B4	4 Fiber Cross-Connect		\$53.49	NA
	Security Escort	Per half hr./Add'l		
	Security Escort	half hr.		
PE1BT	Basic Time	110417 111.	NA	\$42,92/\$25.56
PEIOT	Overtime		NA	\$54.51/\$32.44
PEIPT	Premium Time		NA	\$66.10/\$39.32
1211	Tromum Timo			
AEH	Additional Engineering Fee (Note	Per request, first		First/Add'l
	5)	half hr/add'l half hr.		Basic Time
				\$31.00/\$22.00
				Overtime
				\$37.00/\$26.00

## EXHIBIT A: BELLSOUTH/CLEC-1 RATES – NORTH CAROLINA PHYSICAL COLLOCATION (continued)

#### Note(s):

N/A refers to rate elements which do not have a negotiated rate.

- (1) Subsequent Application Fee: BellSouth requires the submission of an Application Fee for modifications to an existing arrangement. However, when the modifications do not require BellSouth to expend capital, BellSouth will assess the Subsequent Application Fee in lieu of the Application Fee. Proposed modifications that could result in assessment of a Subsequent Application Fee would cause BellSouth to analyze the following but are not limited to: floor loading changes, changes to HVAC requirements, power requirement changes which may result in a power plant upgrade, environmental or safety requirements, or equipment relocation. Should the Subsequent Application Fee not be included as part of this Attachment, CLEC-1 will be assessed the full Application Fee for all subsequent activity for completed arrangements.
- (2) Space Enclosure Fee: The Space Enclosure Construction Fee is a monthly recurring fee, assessed per enclosure, per location with a one-hundred (100) square foot minimum enclosure. It recovers costs associated with providing an optional equipment arrangement enclosure, which include architectural and engineering fees, materials, and installation costs. The cost for additional square feet is applicable only when ordered with the first 100 square feet and must be requested in fifty (50) square foot increments. CLEC-1 may, at its option, arrange with a BellSouth Certified Contractor to construct the space enclosure in accordance with BellSouth's guidelines and specifications. In this event, the BellSouth Certified Contractor shall directly bill CLEC-1 for the space enclosure, and this fee shall not be applicable.
- (3) Cross Connect: The charges for cross connects are for orders placed electronically. Cross connect elements may also be ordered manually for which there is an additional charge per element.

	First/Addition:		
2-wire	\$46.53/\$43.98		
4-wire	\$46.64/\$43.98		
DS-1	\$75.72/\$55.78		
DS-3	\$74.54/\$54.13		

(4) Co-Carrier Cross-Connect. As stated in Section 5 of the Collocation Attachment, CLEC-1 may connect to other CLECs within the designated Premises in addition to, and not in lieu of, interconnection to BellSouth services and facilities. Where BellSouth must construct a cable rack structure to house the co-Carrier cross-connection, construction charges will be applied on an individual case basis as described in Section 5.6.1 of the Collocation Attachment. BellSouth shall provide an estimate of these charges in the Application Response. Where an existing cable rack structure is in place and has sufficient capacity to accommodate the co-Carrier cross-connection requested, the recurring charges as stated in this Exhibit A shall apply.

## EXHIBIT A: BELLSOUTH/CLEC-1 RATES – NORTH CAROLINA PHYSICAL COLLOCATION (continued)

(5) Additional Engineering Fee: BellSouth's additional engineering, and other labor costs associated with handling CLEC-1-requested modifications to requests in progress or augmentations for existing arrangements shall be recovered as Additional Engineering charges, under provisions in BellSouth's F.C.C. Number 1 Tariff, Sections 13.1 and 13.2. Should Additional Engineering rates not be included, CLEC-1 agrees not to make changes to collocation arrangement after a Bona Fide Firm Order is submitted.

#### AMENDMENT

TO

# THE INTERCONNECTION AGREEMENT BETWEEN BUSINESS TELECOM, INC. AND BELLSOUTH TELECOMMUNICATIONS, INC. DATED FEBRUARY 21, 2000

Pursuant to this Agreement (the "Amendment"), Business Telecom, Inc. ("BTI") and BellSouth Telecommunications, Inc. ("BellSouth") hereinafter referred to as the "Parties", hereby agree to amend the Interconnection Agreement between the Parties dated February 21, 2000 ("Interconnection Agreement").

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby covenant and agree as follows:

- 1. Attachment 2 of the Agreement, including Exhibit A of Attachment 2, is hereby deleted in its entirety and replaced with a new Attachment 2, including a new Exhibit A, attached hereto as Exhibit 1.
- 2 All of the other provisions of the Interconnection Agreement dated February 21, 2000 shall remain unchanged and in full force and effect until the expiration date.
- 3. Either or both of the Parties is authorized to submit this Amendment to the appropriate regulatory agencies for approval subject to Section 252 (e) of the Federal Telecommunications Act of 1996.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

Business '	Felecom, Inc.	BellSouth Telecommunications, Inc		
Ву:	Signature on File Signature	By: Signature on File Signature		
Name: An	thony M. Copeland	Name: Jerry D. Hendrix		
Title: <u>Ex</u>	ecutive VP/General Counsel	Title: Senior Director		
Date:	17 Oct. 00	Date: 10/20/00		

#### **Attachment 2**

**Network Elements and Other Services** 

#### **TABLE OF CONTENTS**

1.	INTRODUCTION	3
2.	UNBUNDLED LOOPS, INTEGRATED DIGITAL LOOP CARRIERS, NETWORINTERFACES DEVICE, UNBUNDLED LOOP CONCENTRATION (ULC) SYSTEM, SUB LOOPS AND DARK FIBER	
3.	HIGH FREQUENCY SPECTRUM NETWORK ELEMENT	
4.	SWITCHING	28
5.	UNBUNDLED NETWORK ELEMENT COMBINATIONS	36
6.	TRANSPORT, CHANNELIZATION AND DARK FIBER	44
7.	BELLSOUTH SWA 8XX TOLL FREE DIALING TEN DIGIT SCREENING SERVICE	50
8	LINE INFORMATION DATABASE (LIDB)	52
9	SIGNALING	54
10.	OPERATOR CALL PROCESSING, INWARD OPERATOR SERVICES AND DIRECTORY ASSISTANCE SERVICES	63
11.	CALLING NAME (CNAM) DATABASE SERVICE	69
12.	BASIC 911 AND E911	70
13.	TRUE-UP	72
LII	OB Storage Agreement Exhi	bit A
CN	AM Database Service Exhi	bit B
Rat		

#### ACCESS TO NETWORK ELEMENTS AND OTHER SERVICES

#### 1. Introduction

- 1.1 This Attachment sets forth the unbundled network elements and combinations of unbundled network elements that BellSouth agrees to offer to BTI in accordance with its obligations under Section 251(c)(3) of the Act. The specific terms and conditions that apply to the unbundled network elements are described below in this Attachment 2. The price for each unbundled network element and combination of unbundled Network Elements is set forth in Exhibit D of this Agreement.
- 1.2 For purposes of this Agreement, "Network Element" is defined to mean a facility or equipment provided by BellSouth on an unbundled basis as is used by the CLEC in the provision of a telecommunications service. These unbundled network elements are consistent with the requirements of the FCC 51.319 rule. For purposes of this Agreement, combinations of Network Elements shall be referred to as "Combinations."
- 1.2.1 Except as otherwise required by law, BellSouth shall not impose limitation restrictions or requirements or request for the use of the network elements or combinations that would impair the ability of BTI to offer telecommunications service in the manner BTI intends.
- 1.2.2 Except upon request by BTI, BellSouth shall not separate requested network elements that BellSouth currently combines.
- 1.2.2.1 Unless otherwise ordered by an appropriate state or federal regulatory agency, currently combined Network Elements are defined as elements that are already combined within BellSouth's network to a given location.
- 1.3 BellSouth shall, upon request of BTI, and to the extent technically feasible, provide to BTI access to its network elements for the provision of BTI's telecommunications service. If no rate is identified in the contract for a specific service or function, the rate for the specific service or function will be as negotiated by the Parties upon request by either Party.
- 1.4 BTI may purchase network elements and other services from BellSouth for the purpose of combining such network elements in any manner BTI chooses to provide telecommunication services to its intended users, including recreating existing BellSouth services. With the exception of the sub-loop elements which are located outside of the central office, BellSouth shall deliver the network elements purchased by BTI for combining to the designated BTI collocation space. The network elements shall be provided as set forth in this Attachment.

Version 3Q00: 10/10/00

.

- In the event that any effective legislative, regulatory, judicial or other legal action modifies or redefines the "Network Elements" in a manner which materially affects the terms of this Attachment or the Network Elements and/or prices set forth herein, either Party may, on thirty (30) days written notice, require renegotiation of such terms, and the Parties shall renegotiate in good faith such new terms in accordance with such legislative, regulatory, judicial or other legal action. In the event such new terms are not renegotiated within ninety (90) days after the notice for renegotiation, either Party may petition the Commission for resolution of the dispute between the Parties. Each Party reserves the right to seek judicial review of any Commission ruling concerning this Attachment.
- 1.6 BTI will adopt and adhere to the standards contained in the applicable BellSouth Operational Understanding regarding maintenance of service.
- 1.7 Standards for Network Elements
- 1.7.1 BellSouth shall comply with the requirements set forth in the technical references, as well as any performance or other requirements identified in this Agreement, to the extent that they are consistent with the greater of BellSouth's actual performance or applicable industry standards.
- 1.7.2 If one or more of the requirements set forth in this Agreement are in conflict, the Parties shall mutually agree on which requirement shall apply. If the Parties cannot reach agreement, the dispute resolution process set forth in Section 12 of the General Terms and Conditions of this Agreement, incorporated herein by this reference, shall apply.
- 1.8 Rates

The prices that BTI shall pay to BellSouth for Network Elements and Other Services are set forth in Exhibit D to this Attachment. If BTI purchases a service(s) from a tariff, all terms and conditions and rates as set forth in such tariff shall apply.

1.9 **Operational Support Systems (OSS)** 

The terms, conditions and rates for OSS are as set forth in Section 2.13 of this Attachment.

2. Unbundled Loops, Network Interfaces Device, Unbundled Loop Concentration (ULC) System, Sub loops and Dark Fiber

All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of unbundled loops.

Version 3000: 10/10/00

#### 2.1 **Unbundled Loops**

#### 2.1.1 Definition

- 2.1.2 The local loop network element ("Loop(s)") is defined as a transmission facility between a distribution frame (or its equivalent) in BellSouth's central office and the loop demarcation point at an end-user customer premises, including inside wire owned by BellSouth. The local loop network element includes all features, functions, and capabilities of the transmission facilities, including dark fiber and attached electronics (except those used for the provision of advanced services, such as Digital Subscriber Line Access Multiplexers) and line conditioning.
- 2.1.3 The provisioning of service to a CLEC's collocation space will require cross-office cabling and cross-connections within the central office to connect the loop to a local switch or to other transmission equipment. These cross-connects are a separate component, that are not considered a part of the loop, and thus, have a separate charge.
- 2.1.4 BellSouth Order Coordination referenced in Attachment 2 includes two types: "Order Coordination" (OC) and "Order Coordination Time Specific" (OC-TS).
- 2.1.5 "Order Coordination" refers to standard BellSouth service order coordination involving the reuse of facilities for SL2 voice loops and all digital loops, where BTI is requesting that their loop order be provisioned over an existing circuit that is currently providing service to the end user. Order coordination for physical conversions will be scheduled at BellSouth's discretion during normal working hours on the committed due date and BTI will be advised.
- 2.1.6 "Order Coordination Time Specific" refers to service order coordination in which BTI requests a specific time for a service order conversion to take place. Loops on a single service order of 14 or more loops will be provisioned on a project basis. This is a chargeable option for any coordinated order and is billed in addition to the OC charge. BTI may specify a time between 9:00 a.m. and 4:00 p.m. (location time) Monday through Friday (excluding holidays). If BTI specifies a time outside this window, or selects a time or quantity of loops that requires BellSouth technicians to work outside normal work hours, overtime charges will apply in addition to the OC and OC-TS charges. Overtime charges will be applied according to actual costs based on type of force group required to perform the work, overtime hours worked and any special circumstances.

Version 3000: 10/10/00

SL-1	Not available	Not available	Not available	Chargeable Option	Charged for Dispatch inside & outside Central Office
SL-2	Included	Chargeable Option*	Included	Included	Charged for Dispatch outside Central Office
Unbundled Digital Loop	Included	Chargeable Option* (except on Universal Digital Channel)	Included (where appropriate)	Included	Charged for Dispatch outside Central Office
Unbundled Copper Loop	Chargeable Option	Not available	Included	Included	Charged for Dispatch outside Central Office

- Where facilities are available, BellSouth will install loops in compliance with BellSouth's Interval Guide available at the website at http://www.interconnection.bellsouth.com. For orders of 14 or more loops, the installation will be handled on a project basis and the intervals will be set by the BellSouth project manager for that order. Some loops require a Service Inquiry (SI) to determine if facilities are available prior to issuing the order. The interval for the SI process is separate from the installation interval. For expedite requests by BTI, expedite charges will apply for intervals less than 5 days. The charges outlined in BellSouth's FCC No. 1 Tariff, Section 5, will apply. If BTI cancels an order for network elements and other services, any costs incurred by BellSouth in conjunction with the provisioning of that order will be recovered in accordance with FCC No. 1 Tariff, Section 5.
- 2.1.8 If BTI modifies an order after being sent a Firm Order Confirmation (FOC) from BellSouth, any reasonable and demonstrable costs incurred by BellSouth to accommodate the modification will be reimbursed by BTI.
- 2.1.9 BellSouth will offer Unbundled Voice Loops (UVL) in two different service levels Service Level One (SL1) and Service Level Two (SL2).
- 2.1.10 SL1 loops will be non-designed, will not have test points, and will not come with any OC or engineering information/circuit make-up data. Upon issuance of an order in the service order system, SL1 loops will be activated on the due date in the same manner and time frames that BellSouth normally activates POTS-type loops for its customers. If BTI requests work to be done for SL1s that requires BellSouth technicians to work outside normal work hours, overtime charges will

<sup>\*</sup>Order Coordination-Time Specific charge for orders due on same day at same location will be applied on a per LSR basis.

be applied according to actual costs based on type of force group required to perform the work, overtime hours worked and any special circumstances.

- 2.1.11 SL2 loops shall have test points, will be designed with a design layout record provided to BTI, and will be provided with OC. The OC feature will allow BTI to coordinate the installation of the loop with the disconnect of an existing customer's service and/or number portability service. In these cases, BellSouth will perform the order conversion with standard order coordination at its discretion during normal work hours.
- 2.1.12 BellSouth will also offer Unbundled Digital Loops (UDL) such as DS1s and DS3s. UDL will be designed, will be provisioned with test points (where appropriate), and will come standard with OC and a Design Layout Record (DLR).

Due to technical limitations associated with certain Digital Loop Carrier (DLC) systems, some ISDN-capable loops that are provisioned using DLC systems may not support IDSL (Integrated Digital Subscriber Line) service. BellSouth will not reconfigure its ISDN-capable loop to support IDSL service.

Instead, BellSouth agrees to offer the Universal Digital Channel (UDC), which may also be referred to as an IDSL-capable loop as a part of its Unbundled Digital Loop offerings. The UDC loop is intended to be compatible with IDSL service and has the same physical characteristics and transmission specifications as BellSouth's ISDN-capable loop. These specifications are listed in BellSouth's TR73600.

Like the ISDN-capable loop, the UDC loop may be provisioned on copper or through a DLC system. However, when UDC loops are provisioned using a DLC system, BellSouth will ensure that they are only provisioned on time slots that are compatible with data-only services such as IDSL.

- 2.1.13 As a chargeable option on all loops except UVL-SL1, Universal Digital Channel (UDC) and Unbundled Copper Loop (UCL), BellSouth will offer OC-TS. This will allow BTI the ability to specify the time that the coordinated conversion takes place. The OC-TS charge for orders due on the same day at the same location will be applied on a per Local Service Request (LSR) basis.
- 2.1.14 In addition to the UVLs and UDLs, BellSouth shall make available Unbundled Copper Loops (UCLs). The UCL will be a copper twisted pair loop that is unencumbered by any intervening equipment (e.g., filters, load coils, range extenders, digital loop carrier, or repeaters). The UCL will be offered in two versions Short and Long. A short UCL (18 kft or less) will be provisioned

according to Resistance Design parameters, may have up to 6kft of bridged tap and will have up to 1300 ohms of resistance. The long UCL (beyond 18kft) will be any dry copper pair longer than 18kft and may have up to 12kft of bridged tap and up to 2800 ohms of resistance. Unbundled Loop Modifications (ULM) may be used when a CLEC wants to condition copper loops by removing load coils and other intervening equipment. In almost every case, the UCL long will require ULM to remove load coils. BellSouth will only ensure electrical continuity and balance relative to tip and ring on UCLs.

- 2.1.15 The UCL is a designed circuit, is provisioned with a test point and comes standard with a DLR. OC will be offered as a chargeable option on all UCL loops. OC is required on UCLs where a reuse of existing facilities has been requested by BTI. Order Coordination Time Specific (OC-TS) will not be offered on UCLs.
- 2.1.16 The UCL is a dry copper loop and is not intended to support any particular telecommunications service. BTI may use the UCL loop for a variety of services, including xDSL (e.g., ADSL and HDSL) services, by attaching appropriate terminal equipment of BTI's choosing. BTI will determine the type of service that will be provided over the loop.
- 2.1.17 Because the UCL loop shall be an unbundled loop offering that is separate and distinct from BellSouth's ADSL and HDSL capable loop offerings, BTI agrees that BellSouth's UCL loop will not be held to the service level and performance expectations that apply to its ADSL and HDSL unbundled loop offerings. BellSouth shall only be obligated to maintain copper continuity and provide balance relative to tip and ring on UCL loops.
- 2.1.18 The UCL loop shall be provided to BTI in accordance with BellSouth's Technical Reference 73600.
- 2.1.19 BTI will be responsible for testing and isolating troubles on the loops. Once BTI has isolated a trouble to the BellSouth provided loop, BTI will issue a trouble to BellSouth on the loop. BellSouth will take the actions necessary to repair the loop if a trouble actually exists. BellSouth will repair these loops in the same time frames that BellSouth repairs similarly situated loops to its customers.
- 2.1.20 If BTI reports a trouble on SL1 loops and no trouble actually exists, BellSouth will charge BTI for any dispatching and testing (both inside and outside the CO) required by BellSouth in order to confirm the loop's working status.
- 2.1.21 If BTI reports a trouble on SL2 loops and no trouble actually exists, BellSouth will charge BTI for any dispatching and testing, (outside the CO) required by BellSouth in order to confirm the loop's working status.
- 2.1.22 Technical Requirements

- 2.1.22.1 To the extent available within BellSouth's Network at a particular location, BellSouth will offer loops capable of supporting telecommunications services such as: POTS, Centrex, basic rate ISDN, analog PBX, voice grade private line, ADSL, HDSL, and digital data (up to 64 kb/s). If a requested loop type is not available, then the CLEC can use the Special Construction process to request that BellSouth place facilities or otherwise modify facilities in order to meet BTI's request.
- 2.1.22.2 BTI will be responsible for providing BellSouth with a Service Profile Identifier (SPID) associated with a particular ISDN-capable loop and end user. With the SPID, BellSouth will be able to adequately test the circuit and ensure that it properly supports ISDN service.
- 2.1.22.3 The loop will support the transmission, signaling, performance and interface requirements of the services described in 2.1.2 above. It is recognized that the requirements of different services are different, and that a number of types or grades of loops are required to support these services. Services provided over the loop by BTI will be consistent with industry standards and BellSouth's TR73600.
- 2.1.22.4 BTI may utilize the unbundled loops to provide any telecommunication service it wishes. However, BellSouth will only provision, maintain and repair the loops to the standards that are consistent with the type of loop ordered. For example, if BTI orders an ISDN-capable loop but wants to use the loop for a service other than ISDN, BellSouth will only support that the loop is capable of providing ISDN service. For non-service specific loops (e.g. UCL, loops modified by BTI using the Special Construction process), BellSouth will only support that the loop has copper continuity and balanced tip-and-ring.
- 2.1.22.5 In some instances, BTI will require access to a copper twisted pair loop unfettered by any intervening equipment (e.g., filters, load coils, range extenders, etc.), so that BTI can use the loop for a variety of services by attaching appropriate terminal equipment at the ends. BTI will determine the type of service that will be provided over the loop. In some cases, BTI may be required to pay additional charges for the removal of certain types of equipment. BellSouth's Unbundled Loop Modifications (ULM) process will be used to determine the costs and feasibility of these activities.
- 2.1.22.6 In those cases where BTI has requested that BellSouth modify a loop so that it no longer meets the technical parameters of the original loop type (e.g., voice grade, ISDN, ADSL, etc.) the resulting modified loop will be ordered and maintained as a UCL.
- 2.1.22.7 The loop shall be provided to BTI in accordance with BellSouth's TR73600 Unbundled Local Loop Technical Specification and applicable industry standard technical references.

#### 2.2 **Unbundled Loop Modifications (Line Conditioning)**

- 2.2.1 Subject to applicable and effective FCC rules and orders, BellSouth shall condition loops, as requested by BTI, whether or not BellSouth offers advanced services to the End User on that loop.
- 2.2.2 Loop conditioning is defined as the removal from the loop of any devices that may diminish the capability of the loop to deliver high-speed switched wireline telecommunications capability, including xDSL service. Such devices include, but are not limited to, load coils, bridge taps, low pass filters, and range extenders.
- 2.2.3 The Unbundled Loop Modifications (ULM) offering provides the following elements: 1) removal of equipment on loops equal to or less than 18kft; 2) removal of equipment of loops longer than 18kft; and 3) removal of bridged-taps on loops of any length.
- 2.2.4 BellSouth shall recover the cost of line conditioning requested by BTI through a recurring charge and/or nonrecurring charge(s) in accordance with the FCC's forward-looking pricing principles promulgated pursuant to Section 252 (d) (1) of the Act and in compliance with FCC Rule 52.507 (e).

#### 2.3 **Integrated Digital Loop Carriers**

Where BellSouth uses Integrated Digital Loop Carrier (IDLC) systems to provide the local loop and BellSouth has a suitable alternate facility available, BellSouth will make arrangements to permit BTI to order a contiguous local loop. To the extent it is technically feasible, these arrangements will provide BTI with the capability to serve end users at a level that is at parity with the level of service BellSouth provides its customers. If no alternate facility is available, BellSouth will utilize its Special Construction (SC) process to determine the additional costs required to provision the loop facilities. BTI will then have the option of paying the SC rates to place the loop facilities or BTI may chose some other method of providing service to the end-user (e.g., Resale, private facilities, etc.).

#### 2.4 Network Interface Device

#### 2.4.1 Definition

The NID is defined as any means of interconnection of end-user customer inside wire to BellSouth's distribution plant, such as a cross-connect device used for that purpose. The NID is a single-line termination device or that portion of a multiple-line termination device required to terminate a single line or circuit at the premises. The NID features two independent chambers or divisions that separate the service provider's network from the End User's on-premises wiring. Each chamber or division contains the appropriate connection points or posts to which the service provider and the end user each make their connections. The NID

Version 3Q00: 10/10/00

provides a protective ground connection and is capable of terminating cables such as twisted pair cable.

- 2.4.2 BellSouth shall permit BTI to connect BTI's loop facilities the end-user's inside wire through the BellSouth NID or at any other technically feasible point.
- 2.4.3 <u>Access to Network Interface Device (NID)</u>
- 2.4.3.1 Due to the wide variety of NIDs utilized by BellSouth (based on subscriber size and environmental considerations), BTI may access the end user's wire by any of the following means: BellSouth shall allow BTI to connect its loops directly to BellSouth's multi-line residential NID enclosures that have additional space and are not used by BellSouth or any other telecommunications carriers to provide service to the premise. It is the responsibility of BTI to leave undisturbed the existing form of electrical protection and to maintain the physical integrity of the NID.
- 2.4.3.2 Where an adequate length of the end user's inside wire is present and environmental conditions permit, either Party may remove the inside wire from the other Party's NID and connect that wire to that Party's own NID; or
- 2.4.3.3 Enter the subscriber access chamber or "side" of "dual chamber" NID enclosures for the purpose of extending a connecterized or spliced jumper wire from the inside wiring through a suitable "punch-out" hole of such NID enclosures; or
- 2.4.3.4 Request BellSouth to make other rearrangements to the inside wiring terminations or terminal enclosure on a time and materials cost basis to be charged to the requesting Party (i.e., BTI, its agent, the building owner or the subscriber). Such charges will be billed to the requesting Party.
- 2.4.3.5 In no case shall either Party remove or disconnect the other Party's loop facilities from either Party's NIDs, enclosures, or protectors unless the applicable Commission has expressly permitted the same and the disconnecting Party provides prior notice to the other Party. In such cases, it shall be the responsibility of the Party disconnecting loop facilities to leave undisturbed the existing form of electrical protection and to maintain the physical integrity of the NID. It will be the CLEC's responsibility to ensure there is no safety hazard and will hold BellSouth harmless for any liability associated with the removal of the BellSouth loop from the BellSouth NID. Furthermore, it shall be the responsibility of the disconnecting Party, once the other Party's loop has been disconnected from the NID, to reconnect the disconnected loop to a nationally-recognized-testinglaboratory-listed station protector, which has been grounded as per Article 800 of the National Electrical Code. If no spare station protector exists in the NID, the disconnected loop must be appropriately cleared, capped and stored. If CLEC does not wish to accept these responsibilities, other options exist in which BellSouth installs a NID for the CLEC as a chargeable option.

- 2.4.3.6 In no case shall either Party remove or disconnect ground wires from BellSouth's NIDs, enclosures, or protectors.
- 2.4.3.7 In no case shall either Party remove or disconnect NID modules, protectors, or terminals from BellSouth's NID enclosures.
- 2.4.3.8 Due to the wide variety of NID enclosures and outside plant environments BellSouth will work with BTI to develop specific procedures to establish the most effective means of implementing this Section, 2.4.3.

#### 2.4.4 Technical Requirements

- 2.4.4.1 The NID shall provide an accessible point of interconnection and shall maintain a connection to ground.
- 2.4.4.2 The NID shall be capable of transferring electrical analog or digital signals between the subscriber's inside wiring and the Distribution Media and/or cross connect to BTI's NID, consistent with the NID's function at the Effective Date of this Agreement.
- 2.4.4.3 Where a BellSouth NID exists, it is provided in its "as is" condition. BTI may request BellSouth do additional work to the NID in accordance with Section 2.4.3.8. When BTI deploys its own local loops with respect to multiple-line termination devices, BTI shall specify the quantity of NIDs connections that it requires within such device.

#### 2.4.5 Interface Requirements

2.4.5.1 The NID shall be equal to or better than all of the requirements for NIDs set forth in the applicable industry standard technical references.

#### 2.5 Unbundled Loop Concentration (ULC) System

- 2.5.1 BellSouth will provide to BTI Unbundled Loop Concentration (ULC). Loop concentration systems in the central office concentrate the signals transmitted over local loops onto a digital loop carrier system. The concentration device is placed inside a BellSouth central office. BellSouth will offer ULC with a TR008 interface or a TR303 interface.
- 2.5.2 ULC will be offered in two sizes. System A will allow up to 96 BellSouth loops to be concentrated onto multiple DS1s. The high-speed connection from the concentrator will be at the electrical DS1 level and may connect to BTI at BTI's collocation site. System B will allow up to 192 BellSouth loops to be concentrated onto multiple DS1s. System A may be upgraded to a System B. A minimum of two DS1s is required for each system (i.e., System A requires two DS1s and System B would require an additional two DS1s or four in total). All DS1 interfaces will terminate to the CLEC's collocation space. ULC service is

Version 3Q00: 10/10/00

offered with or without concentration and with or without protection. A Line Interface element will be required for each loop that is terminated onto the ULC system. Rates for ULC are as set forth in this Attachment.

#### 2.6 **Sub-loop Elements**

- 2.6.1 Where facilities permit and subject to applicable and effective FCC rules and orders, BellSouth shall offer access to its Unbundled Sub Loop (USL) and Unbundled Sub-loop Concentration (USLC) System. BellSouth shall provide non-discriminatory access, in accordance with FCC Rule 51.311 and Section 251(c) (3) of the Act, to the sub-loop on an unbundled basis and pursuant to the following terms and conditions and the rates approved by the Commission and set forth in this Attachment.
- 2.6.2 Sub-loop components include but are not limited to the following:
- 2.6.2.1 Unbundled Sub-Loop Distribution;
- 2.6.2.2 Unbundled Sub-Loop Concentration/Multiplexing Functionality; and
- 2.6.2.3 Unbundled Sub-Loop Feeder.
- 2.6.2.4 Unbundled Sub-Loop Intrabuilding Network Cable
- 2.7 Unbundled Sub-Loop (distribution facilities)
- 2.7.1 Definition
- 2.7.1.1 Subject to applicable and effective FCC rules and orders, the unbundled sub-loop distribution facility is dedicated transmission facility that BellSouth provides from a customer's point of demarcation to a BellSouth cross-connect device. The BellSouth cross-connect device may be located within a remote terminal (RT) or a stand-alone cross-box in the field or in the equipment room of a building. The unbundled sub-loop distribution media is a copper twisted pair that can be provisioned as a 2 Wire or 4 Wire facility. Following are the current sub-loop distribution offerings:
- 2.7.1.1.1 Voice grade Unbundled Sub-Loop Distribution (USL-D) is a sub-loop facility from the cross-box in the field up to and including the point of demarcation, at the end user's premises.
- 2.7.1.1.2 Unbundled Sub-Loop distribution facilities were originally built as part of the entire voice grade loop from the BellSouth central office to the customer network interface. Therefore, the voice grade Unbundled Sub-Loop may have load coils, which are necessary for transmission of voice grade services.

- 2.7.1.1.3 Unbundled Copper Sub-Loop (UCSL) is a non-loaded copper facility of any length provided from the cross-box in the field up to and including the end-user's point of demarcation.
- 2.7.1.1.3.1 If available, this facility will not have any intervening equipment such as load coils between the end-user and the cross-box.
- 2.7.2 If BTI requests a UCSL and a non-loaded pair is not available, BTI may order Unbundled Sub-Loop Modification to remove load coils and/or bridge tap from an existing sub-loop facility. If load coils are removed from an existing sub-loop, that sub-loop will be classified as a UCSL. BTI may order Loop Make-up to determine what loop modifications will be required.
- 2.7.3 Unbundled Sub-Loop distribution facilities shall support functions associated with provisioning, maintenance and testing of the Unbundled Sub-Loop. For access to Voice Grade USL-D and UCSL, BTI would be required to deliver a cable to the BellSouth remote terminal or cross-box in the field to provide continuity to BTI's feeder facilities. This cable would be connected, by a BellSouth technician, within the BellSouth RT/cross-box during the set-up process. BTI's cable pairs can then be connected to BellSouth's USL within the BellSouth cross-box by the BellSouth technician.
- 2.7.4 Unbundled Sub-Loop Intrabuilding Network Cable (USL-INC) (a.k.a. riser cable) is the distribution facility inside a subscriber's building or between buildings on one customer's same premises (continuous property not separated by a public street or road). USL-INC includes the facility from the cross-connect device in the building equipment room up to and including the point of demarcation, at the end user's premises.
- 2.7.4.1 In a scenario that requires connection in a building equipment room, BellSouth will install a cross connect panel for the purpose of accessing USL-INC pairs. The cross-connect panel will function as a single point of interconnection (SPOI) for USL-INC and will be accessible by multiple carriers as space permits. BellSouth will place cross-connect blocks in 25-pair increments for BTI's use on this cross-connect panel. BTI will be responsible for connecting its facilities to the 25-pair cross-connect block(s).
- 2.7.5 BellSouth will provide Unbundled Sub-Loops where available. Through the firm order Service Inquiry (SI) process, BellSouth will determine if it is technically feasible to place the required facilities where BTI has requested access to Unbundled Sub-Loops. If existing capacity is sufficient to meet the CLEC demand, then BellSouth will perform the set-up work as described in Section 2.7.6. If any work must be done to modify existing BellSouth facilities or add new facilities (other than adding the cross-connect panel in a building equipment room as noted in Section 2.7) to accommodate BTI's request for Unbundled Sub-Loops, BTI may request BellSouth's Special Construction (SC) process to

determine additional costs required to provision the Unbundled Sub-Loops; provided that the one-time SC rate shall not exceed the costs of a similarly situated Bell South retail end user customer would pay when requesting a new sub-loop to that location. BTI will have the option of paying the SC charges to modify the BellSouth facilities.

- 2.7.6 Set-up work must be completed before BTI can order sub-loop pairs. During the set-up in a BellSouth cross-connect box in the field, the BellSouth technician will perform the necessary work to splice the CLEC's cable into the cross-connect box. For the set-up inside a building equipment room, BellSouth will perform the necessary work to install the cross-connect panel and the connecting block(s) that will be used to provide access to the requested USLs.
- 2.7.6.1 Once the set-up is complete, the CLEC will request sub-loop pairs through submission of a Local Service Request (LSR) form to the Local Carrier Service Center (LCSC). Order Coordination is required with USL pair provisioning when BTI requests reuse of an existing facility and is in addition to the USL pair rate. For expedite requests by BTI for sub-loop pairs, expedite charges will apply for intervals less than 5 days.
- 2.7.6.2 Unbundled Sub-Loop shall be equal to or better than each of the applicable requirements set forth in the applicable industry standard technical references.
- 2.7.6.3 Unbundled Sub-Loops will be provided in accordance with technical reference TR73600.
- 2.8 Unbundled Network Terminating Wire (UNTW)
- 2.8.1 BellSouth agrees to offer its Unbundled Network Terminating Wire (UNTW) to BTI pursuant to the following terms and conditions at rates as set forth in this Attachment.
- 2.8.2 Definition
- 2.8.2.1 Subject to applicable and effective FCC rules and orders, UNTW is a dedicated transmission facility that BellSouth provides from the Wiring Closet /Garden Terminal (or other type of cross-connect point) at the point of termination of BellSouth's loop distribution facilities to the end user's point of demarcation. UNTW is the final portion of the loop owned by BellSouth.
- 2.8.3 <u>Requirements</u>
- 2.8.3.1 On a multi-unit premises where Provisioning Party owns the network terminating wire, and by request of Requesting Party, Provisioning Party will provide access to UNTW pairs on an Access Terminal that is suitable for use by multiple carriers at each Garden Terminal or Wiring Closet.

- In new construction where possible, both Parties may at their option and with the property owner's agreement install their own Network Terminating Wire (NTW). In existing construction, the Provisioning Party shall not be required to install new or additional NTW beyond existing NTW to provision the services of the Requesting Party.
- 2.8.3.3 Upon notice from the Requesting Party to the Provisioning Party that the Requesting Party desires access to the Provisioning Party's UNTW pairs in a multi-unit premises, representatives of both Parties will participate in a meeting at the site of the requested access. The purpose of the site visit will include discussion of the procedures for Access Terminal installation, location and addresses of the Access Terminals and to discuss an estimated completion date. Upon completion of site visit, the Requesting Party will submit a Service Inquiry (SI) to the person or organization designated by the Provisioning Party to receive the SI. The SI will initiate the work for the Provisioning Party to begin the Access Terminal installation. In multi-tenant unit (MTU) scenarios, Provisioning Party will provide access to UNTW pairs on an Access Terminal(s). By request of the Requesting Party, an Access Terminal will be installed either adjacent to each Provisioning Party's Garden Terminal or inside each Wiring Closet on the requested MTU. All the UNTW pairs served by a Garden Terminal/Wiring Closet will be made available on the Access Terminals. Requesting Party will deliver and connect its central office facilities to the UNTW pairs within the Access Terminal. Requesting Party may access any available pair on an Access Terminal unless the Provisioning Party or another service provider is using the pair to concurrently provide service. Prior to connecting Requesting Party's service on a pair previously used by Provisioning Party, Requesting Party is responsible for ensuring the end-user is no longer using Provisioning Party's service or another CLEC's service before accessing UNTW pairs.
- 2.8.3.4 Provisioning Party will use best efforts to complete installation of the Access Terminals within 30 business days of the receipt by the Provisioning Party of the Service Inquiry from the Requesting Party.
- 2.8.3.5 Requesting Party is responsible for obtaining the property owner's permission for Provisioning Party to install an Access Terminal(s) on behalf of the Requesting Party. The submission of the SI by the Requesting Party will serve as certification by the Requesting Party that such permission has been obtained.
- 2.8.3.6 Requesting Party will be billed for non-recurring and recurring charges for accessing UNTW pairs at the time the Requesting Party activates the pair(s). BTI will report use of the UNTW pairs on a Local Service Request (LSR) form submitted to BellSouth's Local Carrier Service Center (LCSC).
- 2.8.3.7 Requesting Party will isolate and report repair problems to the UNE center.

  Requesting Party must tag the UNTW pair that requires repair. If Provisioning Party dispatches a technician on a reported trouble call and no UNTW trouble is

found, Provisioning Party will charge Requesting Party for time spent on the dispatch and testing the UNTW pair(s).

- 2.8.3.8 If Requesting Party initiates the Access Terminal installation and the Requesting Party has not activated at least one pair on the Access Terminal installed pursuant to Requesting Party's request for an Access Terminal within 6 months of installation of the Access Terminal, Provisioning Party will bill Requesting Party a non-recurring charge equal to the actual cost of provisioning the Access Terminal.
- 2.8.3.9 If Provisioning Party determines that Requesting Party is using the UNTW pairs without reporting such usage to BellSouth, the following charges shall apply in addition to any fines which may be established by state commissions and any other remedies at law or in equity available to the Provisioning Party:
- 2.8.3.10 If Requesting Party issued a LSR to disconnect an end-user from BellSouth in order to use a UNTW pair, Requesting Party will be billed for the use of the pair back to the disconnect order date.
- 2.8.3.11 If Requesting Party activated a UNTW pair on which Provisioning Party was not previously providing service, Requesting Party will be billed for the use of that pair back to the date the end-user began receiving service using that pair. Upon request, Requesting Party will provide copies of its billing record to substantiate such date. If Requesting Party fails to provide such records, then Provisioning Party will bill the Requesting Party back to the date of the Access Terminal installation.

#### 2.9 Unbundled Sub-Loop Concentration System (USLC)

- 2.9.1 Where facilities permit and where necessary to comply with an effective Commission order, BellSouth will provide BTI with the ability to concentrate its sub-loops onto multiple DS1s back to the BellSouth Central Office. The DS1s will then be terminated into BTI's collocation space. TR-008 and TR303 interface standards are available.
- USLC, using the Lucent Series 5 equipment, will be offered in two different systems. System A will allow up to 96 of BTI's sub-loops to be concentrated onto multiple DS1s. System B will allow an additional 96 of BTI's sub-loops to be concentrated onto multiple DS1s. One System A may be supplemented with one System B and they both must be physically located in a single Series 5 dual channel bank. A minimum of two DS1s is required for each system (i.e., System A requires two DS1s and System B would require an additional two DS1s or four in total). The DS1 level facility that connects the RT site with the serving wire center is known as a Feeder Interface. All DS1 Feeder Interfaces will terminate to the CLEC's collocation space within the SWC that serves the RT where the

CLEC's sub-loops are connected. USLC service is offered with or without concentration and with or without a protection DS1.

2.9.3 In these scenarios BTI would be required to place a cross-box, remote terminal (RT), or other similar device and deliver a cable to the BellSouth RT. This cable would be connected, by a BellSouth technician, to a cross-connect panel within the BellSouth RT/cross-box and would allow BTI's sub-loops to then be placed on the ULSC and transported to their collocation space at a DS1 level.

#### 2.10 **Unbundled Sub-Loop Feeder**

- 2.10.1 Definition
- 2.10.1.1 Unbundled Sub-Loop Feeder (USLF) provides connectivity between BellSouth's central office and its cross-box (or other access point) that serves an end user location.
- 2.10.2 USLF is intended to be utilized for voice traffic and can be configured as 2-wire voice (USLF-2W/V) or 4-wire voice (USLF-4W/V).
- 2.10.3 USLF can also be utilized for digital traffic and can be configured as 2-wire ISDN (USLF-2W/I); 2-wire Copper (USLF-2W/C); 4-wire Copper (USLF-4W/C) facilities: 4-wire DS0 level loop (USLF-4W/D0); or 4-wire DS1 and ISDN (USLF-4W/DI).
- 2.10.4 USLF will provide the facilities needed to provision a 2W or 4W communications pathway from the BellSouth central office to the BellSouth cross-box. This element will allow for the connection of BTI's loop distribution elements onto BellSouth's feeder system.
- 2.10.5 Requirements
- 2.10.5.1 BTI will extend its compatible cable to BellSouth's cross-box. The cable will then be connected to a panel inside the BellSouth cross-box to the requested level of feeder element. In those cases when there is no room in the BellSouth cross-box to accommodate the additional cross-connect panels mentioned above, BellSouth will utilize its Special Construction process to determine the costs to provide the sub-loop feeder element to BTI. BTI will then have the option of paying the special construction charges or canceling the order.
- 2.10.5.2 USLF will be a designed circuit and BellSouth will provide a Design Layout Record (DLR) for this element.
- 2.10.5.3 BellSouth will provide USLF elements in accordance with applicable industry standards for these types of facilities. Where industry standards do not exist, BellSouth's TR73600 will be used to determine performance parameters.

#### 2.11 **Dark Fiber**

#### 2.11.1 Definition

2.11.1.1 Dark Fiber is optical transmission facilities without attached signal regeneration, multiplexing, aggregation or other electronics that connects two points within BellSouth's network. Dark Fiber is unused strands of optical fiber. It may be strands of optical fiber existing in aerial or underground structure. No line terminating elements terminated to such strands to operationalize its transmission capabilities will be available.

#### 2.11.2 Requirements

- 2.11.2.1 BellSouth shall make available Dark Fiber where it exists in BellSouth's network and where, as a result of future building or deployment, it becomes available. If BellSouth has pre-existing plans to make specific use of the fiber within a two year planning period, there is no requirement to provide said fiber to BTI, provided that upon BTI's request, BellSouth shall provide written certification to BTI that BellSouth has plans to use such fiber and shall describe generally its plans to make such use subject to BTI signing a separate confidentiality agreement with respect to such plans.
- 2.11.2.2 If the requested dark fiber has any lightwave repeater equipment interspliced to it, BellSouth will remove such equipment at BTI's request subject to time and materials charges.
- 2.11.2.3 BTI may test the quality of the Dark Fiber to confirm its usability and performance specifications.
- 2.11.2.4 BellSouth shall use its best efforts to provide to BTI information regarding the location, availability and performance of Dark Fiber within ten (10) business days for a records-based answer and twenty (20) business days for a field-based answer, after receiving a request from BTI ("Request"). Within such time period, BellSouth shall send written confirmation of availability of the Dark Fiber ("Confirmation"). From the time of the Request to one hundred and twenty (120) days after Confirmation, BellSouth shall hold such requested Dark Fiber for BTI's use and may not allow any other Party to use such media, including BellSouth. If a Dark Fiber firm order is not received within the one hundred and twenty day period, the fiber will revert to BellSouth's Dark Fiber inventory.
- 2.11.2.5 BellSouth shall use its best efforts to make Dark Fiber available to BTI within thirty (30) business days after it receives written confirmation from BTI that the Dark Fiber previously deemed available by BellSouth is wanted for use by BTI. This includes identification of appropriate connection points (e.g., Light Guide Interconnection (LGX) or splice points) to enable BTI to connect or splice BTI provided transmission media (e.g., optical fiber) or equipment to the Dark Fiber.

- 2.11.2.6 Dark Fiber shall meet BellSouth Technical Reference TR73603, Unbundled Dark Fiber Technical Specifications and/or industry standards, where available.
- 2.11.2.7 BTI may splice and test Dark Fiber obtained from BellSouth using BTI or BTI designated personnel. BellSouth shall provide appropriate interfaces to allow splicing and testing of Dark Fiber. BellSouth shall provide an excess cable length of 25 feet minimum (for fiber in underground conduit) to allow the uncoiled fiber to reach from the manhole to a splicing van.

#### 2.12 **Rates**

The prices that BTI shall pay to BellSouth for Network Elements and Other Services are set forth in Exhibit D to this Attachment.

#### 2.13 Operational Support Systems (OSS)

2.13.1 BellSouth has developed and made available the following electronic interfaces by which BTI may submit LSRs electronically.

LENS Local Exchange Navigation System
EDI Electronic Data Interchange
TAG Telecommunications Access Gateway

2.13.2 LSRs submitted by means of one of these electronic interfaces will incur an OSS electronic ordering charge as specified in the table below. An individual LSR will be identified for billing purposes by its Purchase Order Number (PON). LSRs submitted by means other than one of these interactive interfaces (mail, fax, courier, etc.) will incur a manual order charge as specified in the table below:

OPERATIONAL SUPPORT SYSTEMS	AL, GA, LA, MS, NC, SC	FL, KY, TN
OSS LSR charge, per LSR received from the CLEC by one of the OSS interactive interfaces	\$3.50	\$3.50
CLEC by one of the OSS interactive interfaces	SOMEC	SOMEC
Incremental charge per LSR received from the	See applicable rate	\$19.99
CLEC by means other than one of the OSS interactive interfaces	element	SOMAN

#### 2.13.3 Denial/Restoral OSS Charge

Version 3Q00: 10/10/00

- 2.13.3.1 In the event BTI provides a list of customers to be denied and restored, rather than an LSR, each location on the list will require a separate PON and, therefore will be billed as one LSR per location.
- 2.13.4 <u>Cancellation OSS Charge</u>
- 2.13.4.1 BTI will incur an OSS charge for an accepted LSR that is later canceled by BTI.

Note: Supplements or clarifications to a previously billed LSR will not incur another OSS charge.

- 2.13.5 <u>Network Elements and Other Services Manual Additive</u>
- 2.13.5.1 The Commissions in some states have ordered per-element manual additive non-recurring charges (NRC) for Network Elements and Other Services ordered by means other than one of the interactive interfaces. These ordered Network Elements and Other Services manual additive NRCs will apply in these states, rather than the charge per LSR. The per-element charges are listed on the Rate Tables in Exhibit D.
- 2.14 **Preordering Loop Makeup (LMU)**
- 2.14.1 <u>Description of Service</u>
- 2.14.1.1 BellSouth shall make available to BTI loop makeup (LMU) data for BellSouth's network facilities. This section addresses LMU as a *preordering* transaction, distinct from BTI ordering any other service(s). Loop Makeup *Service Inquiries* (LMUSI) for preordering loop makeup are likewise unique from other preordering functions with associated service inquiries (SI) as described in this Agreement.
- 2.14.1.2 BellSouth will provide BTI with loop makeup information consisting of the composition of the loop material (copper/fiber); the existence, location and type of equipment on the loop, including but not limited to digital loop carrier or other remote concentration devises, feeder/distribution interfaces, bridge taps, load coils, pair-gain devices; the loop length; and the wire gauge. The LMUSI may be utilized by BTI for the purpose of determining whether the loop requested is capable of supporting DSL service or other advanced data services. The determination shall be made solely by BTI and BellSouth shall not be liable in any way for the performance of the advanced data services provisioned over said loop.
- 2.14.1.3 BellSouth's LMU information is provided to BTI as it exists either in BellSouth's databases or in its hard copy facility records. BellSouth does not guarantee accuracy or reliability of the LMU information provided.
- 2.14.1.4 BellSouth offers LMU information for the sole purpose of allowing BTI to determine whether, in BTI's judgment, BellSouth's loops will support the specific services that BTI wishes to provide over those loops. BTI may choose to use

equipment that it deems will enable it to provide a certain type and level of service over a particular BellSouth loop; however, such configurations may not match BellSouth's or the industry's standards and specifications for the intended type and level of service. Accordingly, BTI shall be responsible for insuring that the specific loop type (ADSL, HDSL, or otherwise) ordered on the LSR matches the LMU of the facility requested. BTI bears full responsibility for being knowledgeable of BellSouth's technical standards and the specifications of BellSouth's loops. BTI bears full responsibility for making the appropriate ordering decisions of matching BellSouth loops with BTI's equipment for accomplishing BTI's end goal for the intended service it wishes to provide its enduser(s). BTI is fully responsible for any of its service configurations that may differ from BellSouth's technical standard for the loop type ordered.

#### 2.14.2 Submitting Loop Makeup Service Inquiries

- 2.14.2.1 BTI will be able to obtain LMU information by submitting a LMUSI mechanically or manually. Mechanized LMUSIs should be submitted through BellSouth's Operational Support Systems interfaces. After obtaining the resulting loop data from the mechanized LMUSI process, if BTI determines that it needs further loop data information in order to make a determination of loop service capability, BTI may initiate a separate manual SI for a separate nonrecurring charge as set forth in Section 2.14.3. Mechanized LMU has been made available for limited deployment to those CLECs that have effective X-Digital Subscriber Line (xDSL) Beta Test Agreements in place with BellSouth. CLECs will be notified once a successful Beta Test has been completed, and mechanized LMU shall then be available to BTI.
- 2.14.2.2 **Manual** LMUSIs shall be submitted on the preordering manual LMUSI form by means of fax or electronic-mail to BellSouth's Complex Resale Support Group (CRSG)/Account Team utilizing the Preordering Loop Makeup Service Inquiry form. The standard service interval for the return of a Loop Makeup Manual Service Inquiry is seven business days. This service interval is distinct from the interval applied to the subsequent service order. Manual LMUSIs are not subject to expedite requests.

#### 2.14.3 LMUSI Types and Associated Charges

BTI may request LMU information by submitting LMUSIs in accordance with the rate elements in Exhibit D.

2.14.3.1 BTI will be assessed a nonrecurring charge for each facility queried as specified in the table above. Rates for all states are interim and subject to true-up pending approval of final rates by the respective State Commissions. True-ups will be retroactive to the effective date of this Agreement.

- 2.14.3.2 BTI may reserve facilities for up to four (4) days in connection with a LMUSI. Reserved facilities for which BTI does not plan to place a UNE local service request (LSR) should be cancelled by BTI. Should BTI wish to cancel a reservation on a spare facility, the cancellation will require a facility reservation number (RESID/FRN).
- 2.14.3.3 The reservation holding timeframe is a maximum of four days from the time that BellSouth's LMU data is returned to BTI for the facility queried. During this holding time and prior to BTI's placing an LSR, the reserved facilities are rendered unavailable to other customers, whether for CLEC(s) or for BellSouth. Notwithstanding the foregoing, BellSouth does not guarantee that a reservation will assure BTI's ability to order the exact facility reserved.
- 2.14.3.4 If BTI does not submit an LSR for a UNE service on a reserved facility within the four-day reservation timeframe, the reservation of that spare facility will become invalid and the facility will be released.
- 2.14.3.5 Charges for preordering LMUSI are separate from any charges associated with ordering other services from BellSouth.
- 2.14.4 Ordering of Other UNE Services
- 2.14.4.1 Whenever BTI has reserved a facility through BellSouth's preordering LMU service, should BTI seek to place a subsequent UNE LSR on a reserved facility, BTI shall provide BellSouth the RESID/FRN of the single spare facility on the appropriate UNE LSR., BTI will be billed the appropriate rate element for the specific type UNE loop ordered by BTI as set forth in this Attachment. BTI will not be billed any additional Loop Makeup charges for the loop so ordered. Should BTI choose to place a UNE LSR having previously submitted a request for preordering LMU without a reservation, BTI will be billed the appropriate rate element for the specific UNE loop ordered as well as additional Loop Makeup charges as set forth in this Attachment. Rates are provided in Exhibit D in this Attachment.
- 2.14.4.2 Where BTI submits an LSR to order facilities reserved during the LMUSI process, BellSouth will use its best efforts to assign to BTI the facility reserved as indicated on the return of the LMU. Multi-facility reservations per single RESID/FRN as provided with the mechanized LMUSI process are less likely to result in the specific assignment requested by BTI. For those occasions when BellSouth cannot assign the specific facility reserved by BTI during the LMU preordering transaction, due to incomplete or incorrect information provided by BTI during the ordering process, BellSouth will assign to BTI, subject to availability, a facility that meets the BellSouth technical standards of the BellSouth type loop as ordered by BTI. If the ordered loop type is not available, BTI may utilize the Unbundled Loop Modification process or the Special Construction process, as applicable, to obtain the loop type ordered.

#### 2.15 **Rates**

The prices that BTI shall pay to BellSouth for Network Elements and Other Services are set forth in Exhibit D to this Attachment. If BTI purchases a service(s) from a tariff, all terms and conditions and rates as set forth in such tariff shall apply.

#### 2.16 **Operational Support Systems (OSS)**

The terms, conditions and rates for OSS are as set forth in Section 2.13 of this Attachment.

#### 3. High Frequency Spectrum Network Element

- 3.1 General
- 3.1.1 BellSouth shall provide BTI access to the high frequency portion of the local loop as an unbundled network element only where BellSouth is the voice service provider to the end user ("High Frequency Spectrum") at the rates set forth in Exhibit D. BellSouth shall provide BTI with the High Frequency Spectrum irrespective of whether BellSouth chooses to offer xDSL services on the loop.
- 3.1.2 The High Frequency Spectrum is defined as the frequency range above the voiceband on a copper loop facility carrying analog circuit-switched voiceband transmissions. Access to the High Frequency Spectrum is intended to allow BTI the ability to provide Digital Subscriber Line ("xDSL") data services to the end user for which BellSouth provides voice services. The High Frequency Spectrum shall be available for any version of xDSL presumed acceptable for deployment pursuant to 47 CFR Section 51.230, including, but not limited to, ADSL, HDSL, and any other xDSL technology that is presumed to be acceptable for deployment pursuant to FCC rules. BellSouth will continue to have access to the low frequency portion of the loop spectrum (from 300 Hertz to at least 3000 Hertz, and potentially up to 3400 Hertz, depending on equipment and facilities) for the purposes of providing voice service. BTI shall only use xDSL technology that is within the PSD mask parameters set forth in T1.413 or other applicable industry standards. BTI shall provision xDSL service on the High Frequency Spectrum in accordance with the applicable Technical Specifications and Standards.
- 3.1.3 The following loop requirements are necessary for BTI to be able to access the High Frequency Spectrum: an unconditioned, 2-wire copper loop. An unconditioned loop is a copper loop with no load coils, low-pass filters, range extenders, DAMLs, or similar devices and minimal bridged taps consistent with ANSI T1.413 and T1.601. BellSouth will provide BTI access to the Unbundled Loop Modification (Line Conditioning), in accordance with Section 2.2 of this Agreement. BellSouth is not required to condition a loop for access to the high frequency spectrum if conditioning of that loop significantly degrades BellSouth's

voice service. If BTI requests that BellSouth condition a loop longer than 18,000 ft. and such conditioning significantly degrades the voice services on the loop, BTI shall pay for the loop to be restored to its original state.

- 3.1.4 BTI's termination point is the point of termination for BTI on the toll main distributing frame in the central office ("Termination Point"). BellSouth will use jumpers to connect BTI's connecting block to the splitter. The splitter will route the High Frequency Spectrum on the circuit to BTI's xDSL equipment in BTI's collocation space.
- 3.1.5 BTI shall have access to the splitter for test purposes, irrespective of where the splitter is placed in the BellSouth premises. Should BTI desire to collocate its own splitter, the Parties will negotiate the terms and conditions associated with BTI's provision of its own splitter.
- 3.2 Provisioning of High Frequency Spectrum and Splitter Space
- 3.2.1 BellSouth will provide BTI with access to the High Frequency Spectrum as follows:
- 3.2.1.1 BellSouth will install splitters within forty-two (42) calendar days of BTI's submission of such order to the BellSouth Complex Resale Support Group; provided, however, that in the event BellSouth did not have reasonable notice that a particular central office was to have a splitter installed therein, the forty-two (42) day interval shall not apply. Collocation itself or an application for collocation will serve as reasonable notice.
- 3.2.1.2 Once a splitter is installed on behalf of BTI in a central office, BTI shall be entitled to order the High Frequency Spectrum on lines served out of that central office.
- 3.2.1.2.1 BellSouth will bill and BTI shall pay the SOMAN and SOMEC charges as described in Section 2.13 of this Agreement when BTI orders High Frequency Spectrum for end-user service.
- 3.2.1.3 BellSouth will select, purchase, install, and maintain a central office POTS splitter and provide BTI access to data ports on the splitter. At least 30 days before making a change in splitter suppliers, BellSouth will provide BTI with a carrier notification letter, informing BTI of change. BTI shall purchase ports on the splitter as set forth more fully below.
- 3.2.1.4 BellSouth will install the splitter in (i) a common area close to the BTI collocation area, if possible; or (ii) in a BellSouth relay rack as close to the BTI DS0 termination point as possible. For purposes of this section, a common area is defined as an area in the central office in which both Parties have access to a common test access point. BellSouth will cross-connect the splitter data ports to a specified BTI DS0 at such time that a BTI end user's service is established.

- 3.2.1.5 The High Frequency Spectrum shall only be available on loops on which BellSouth is also providing, and continues to provide, analog voice service directly to the end user. In the event the end-user terminates its BellSouth provided voice service for any reason, and BTI desires to continue providing xDSL service on such loop, BTI shall be required to purchase a full stand-alone loop unbundled network element. In the event BellSouth disconnects the enduser's voice service pursuant to its tariffs or applicable law, and BTI desires to continue providing xDSL service on such loop, BTI shall be permitted to continue using the line by purchasing the full stand-alone loop unbundled network element. To the extent commercially practicable, BellSouth shall give BTI notice in a reasonable time prior to disconnect, which notice shall give BTI an adequate opportunity to notify BellSouth of its intent to purchase such loop. In those cases in which BellSouth no longer provides voice service to the end user and BTI purchases the full stand-alone loop, BTI may elect the type of loop it will purchase. BTI will pay the appropriate recurring and non-recurring rates for such loop as set forth in Exhibit D to this Attachment. In the event BTI purchases a voice grade loop, BTI acknowledges that such loop may not remain xDSL compatible.
- 3.2.1.6 Only one competitive local exchange carrier shall be permitted access to the High Frequency Spectrum of any particular loop.
- 3.3.1 To order High Frequency Spectrum on a particular loop, BTI must have a DSLAM collocated in the central office that serves the end-user of such loop. BTI may order splitters in a central office once it has installed its Digital Subscriber Line Access Multiplexer ("DSLAM") in that central office. BellSouth will install these splitters within the interval provided in paragraph 3.2.1.1.
- 3.3.2 BellSouth will devise a splitter order form that allows BTI to order splitter ports in increments of 24 ports.
- 3.3.2.1 BellSouth will provide BTI the Local Service Request ("LSR") format to be used when ordering the High Frequency Spectrum.
- 3.3.3 BellSouth will provide access to the High Frequency Spectrum within the following target intervals: BellSouth will return a manual Firm Order Confirmation ("FOC") in no more than two (2) business days after receipt of a valid, error free manual LSR. When BTI submits an electronic LSR for High Frequency Spectrum, BellSouth will return a FOC in four (4) hours ninety-five percent (95%) of the time, or, for orders that do not flow-through, in two (2) business days. BellSouth will provide BTI with access to the High Frequency Spectrum at the following target intervals:
- 3.3.3.1 For 1-5 lines at the same address within three (3) business days from date BellSouth issues the FOC; 6-10 lines at same address within 5 business days from

date BellSouth issues the FOC and more than 10 lines at the same address is to be negotiated.

- 3.3.4 BellSouth will provide to BTI BellSouth's Loop Qualification System that BellSouth uses to qualify loops for its own ADSL offering as described below.
- 3.3.5 BellSouth will provide BTI access to the Preordering Loop Makeup (LMU), in accordance with Section 2.14 of this Agreement. BellSouth shall bill and BTI shall pay the rates for such services, as described in Exhibit D.
- 3.4 Maintenance and Repair
- 3.4.1 BTI shall have access, for test, repair, and maintenance purposes, to any loop as to which it has access to the High Frequency Spectrum. BTI may access the loop at the point where the voice and data signal exits the central office splitter.
- 3.4.2 BellSouth will be responsible for repairing voice services and the physical line between the network interface device at the customer's premises and the Termination Point of demarcation in the central office. BTI will be responsible for repairing data services. Each Party will be responsible for maintaining its own equipment.
- 3.4.3 BTI shall inform its end users to direct data problems to BTI, unless both voice and data services are impaired, in which event the end users should call BellSouth.
- 3.4.4 Neither Party will initiate any marketing efforts with respect to the other Party's service nor make any remarks as to the services provided by the other Party.
- 3.4.5 Once a Party has isolated a trouble to the other Party's portion of the loop, the Party isolating the trouble shall notify the end user that the trouble is on the other Party's portion of the loop.
- In the event BTI's deployment of xDSL on the High Frequency Spectrum significantly degrades, from the perspective of the end user, the performance of other advanced services or of BellSouth's voice service on the same loop, BellSouth shall notify BTI and allow twenty-four (24) hours to cure the trouble. If BTI fails to resolve the trouble, BellSouth may discontinue BTI's access to the High Frequency Spectrum on such loop.

#### 3.5 Rates

The prices that BTI shall pay to BellSouth for Network Elements and Other Services are set forth in Exhibit D to this Attachment. If BTI purchases a service(s) from a tariff, all terms and conditions and rates as set forth in such tariff shall apply.

#### 3.6 **Operational Support Systems (OSS)**

The terms, conditions and rates for OSS are as set forth in Section 2.13 of this Attachment.

#### 4. Switching

All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of local and tandem switching.

#### 4.1 **Local Switching**

- 4.1.1 BellSouth shall provide non-discriminatory access to local circuit switching capability, and local tandem switching capability, on an unbundled basis, except as set forth below in Section 4.1.3.3 to BTI for the provision of a telecommunications service. BellSouth shall provide non-discriminatory access to packet switching capability on an unbundled basis to BTI for the provision of a telecommunications service only in the limited circumstance described below in Section 4.4.6.
- 4.1.2 Except as otherwise provided herein, BellSouth shall not impose any restrictions on BTI regarding the use of Switching Capabilities purchased from BellSouth provided such use does not result in demonstrable harm to either the BellSouth network or personnel or the use of the BellSouth network by BellSouth or any other telecommunication carrier.

#### 4.1.3 Local Circuit Switching Capability, including Tandem Switching Capability

#### 4.1.3.1 Definition

Local Circuit Switching Capability is defined as: (A) line-side facilities, which include, but are not limited to, the connection between a loop termination at a main distribution frame and a switch line card; (B) trunk-side facilities, which include, but are not limited to, the connection between trunk termination at a trunk-side cross-connect panel and a switch trunk card; and (C) all features, functions, and capabilities of the switch, which include, but are not limited to: (1) the basic switching function of connecting lines to lines, line to trunks, trunks to lines, and trunks to trunks, as well as the same basic capabilities made available to BellSouth's customers, such as a telephone number, white page listings, and dial tone; and (2) all other features that the switch is capable of providing, including but not limited to customer calling, customer local area signaling service features, and Centrex, as well as any technically feasible customized routing functions provided by the switch; (D) switching provided by remote switching modules.

4.1.3.2 Notwithstanding BellSouth's general duty to unbundle local circuit switching, BellSouth shall not be required to unbundle local circuit switching for BTI when BTI serves end-users with four (4) or more voice-grade (DS-0) equivalents or

lines in locations served by BellSouth's local circuit switches, which are in the following MSAs: Atlanta, GA; Miami, FL; Orlando, FL; Ft. Lauderdale, FL; Charlotte-Gastonia-Rock Hill, NC; Greensboro-Winston Salem-High Point, NC; Nashville, TN; and New Orleans, LA, and BellSouth has provided non-discriminatory cost based access to the Enhanced Extended Link (EEL) throughout Density Zone 1 as determined by NECA Tariff No. 4 as in effect on January 1, 1999.

- 4.1.3.3 In the event that BTI orders local circuit switching for a single end user account name at a single physical end user location with four (4) or more 2-wire voice-grade loops from a BellSouth central office in an MSA listed above, BellSouth shall charge BTI the market based rate in Exhibit D for use of the local circuit switching functionality for the affected facilities.
- 4.1.3.4 A featureless port is one that has a line port, switching facilities, and an interoffice port. A featured port is a port that includes all features then capable or a number of then capable features specifically requested by BTI. Any features that are not currently then capable but are technically feasible through the switch can be requested through the NBR/BFR process.
- 4.1.3.5 BellSouth will provide to BTI customized routing of calls: (i) to a requested directory assistance services platform; (ii) to an operator services platform pursuant to Section 10 of Attachment 2; (iii) for BTI's PIC'ed toll traffic in a two (2) PIC environment to an alternative OS/DA platform designated by BTI. BTI customers may use the same dialing arrangements as BellSouth customers.
- 4.1.3.6 Remote Switching Module functionality is included in Switching Capability. The switching capabilities used will be based on the line side features they support.
- 4.1.3.7 Switching Capability will also be capable of routing local, intraLATA, interLATA, and calls to international customer's preferred carrier; call features (e.g. call forwarding) and Centrex capabilities.
- 4.1.3.8 Where required to do so in order to comply with an effective Commission order, BellSouth will provide to BTI purchasing local BellSouth switching or reselling BellSouth local exchange service under Attachment 1, selective routing of calls to a requested directory assistance services platform or operator services platform. BTI customers may use the same dialing arrangements as BellSouth customers, but obtain a BTI branded service.
- 4.1.4 <u>Technical Requirements</u>
- 4.1.4.1 The requirements set forth in this Section apply to Local Switching, but not to the Data Switching function of Local Switching.
- 4.1.4.2 Local Switching shall be equal to or better than the requirements for Local Switching set forth in the applicable industry standard technical references.

- 4.1.4.3 When applicable, BellSouth shall route calls to the appropriate trunk or lines for call origination or termination.
- 4.1.4.4 Subject to this section, BellSouth shall route calls on a per line or per screening class basis to (1) BellSouth platforms providing Network Elements or additional requirements (2) Operator Services platforms, (3) Directory Assistance platforms, and (4) Repair Centers. Any other routing requests by BTI will be made pursuant to the BFR/NBR Process as set forth in General Terms and Conditions.
- 4.1.4.5 BellSouth shall provide unbranded recorded announcements and call progress tones to alert callers of call progress and disposition.
- 4.1.4.6 BellSouth shall activate service for BTI customer or network interconnection on any of the Local Switching interfaces. This includes provisioning changes to change a customer from BellSouth's services to BTI's services without loss of switch feature functionality as defined in this Agreement.
- 4.1.4.7 BellSouth shall perform routine testing (e.g., Mechanized Loop Tests (MLT) and test calls such as 105, 107 and 108 type calls) and fault isolation on a mutually agreed upon schedule.
- 4.1.4.8 BellSouth shall repair and restore any equipment or any other maintainable component that may adversely impact Local Switching.
- 4.1.4.9 BellSouth shall control congestion points such as those caused by radio station call-ins, and network routing abnormalities. All traffic shall be restricted in a non-discriminatory manner.
- 4.1.4.10 BellSouth shall perform manual call trace and permit customer originated call trace.
- 4.1.4.11 Special Services provided by BellSouth will include the following:
- 4.1.4.11.1 Telephone Service Prioritization;
- 4.1.4.11.2 Related services for handicapped;
- 4.1.4.11.3 Soft dial tone where required by law; and
- 4.1.4.11.4 Any other service required by law.
- 4.1.4.12 BellSouth shall provide Switching Service Point (SSP) capabilities and signaling software to interconnect the signaling links destined to the Signaling Transfer Point Switch (STPS). These capabilities shall adhere to the technical specifications set forth in the applicable industry standard technical references.

Version 3Q00: 10/10/00

- 4.1.4.13 BellSouth shall provide interfaces to adjuncts through Telcordia (formerly BellCore) standard interfaces. These adjuncts can include, but are not limited to, the Service Circuit Node and Automatic Call Distributors.
- 4.1.4.14 BellSouth shall provide performance data regarding a customer line, traffic characteristics or other measurable elements to BTI, upon a reasonable request from BTI. BTI will pay BellSouth for all costs incurred to provide such performance data through the Business Opportunity Request process.
- 4.1.4.15 BellSouth shall offer Local Switching that provides feature offerings at parity to those provided by BellSouth to itself or any other Party.
- 4.1.4.16 BellSouth shall offer to BTI all AIN triggers in connection with its SMS/SCE offering which are supported by BellSouth for offering AIN-based services
- 4.1.4.17 Where capacity exists, BellSouth shall assign each BTI customer line the class of service designated by BTI (e.g., using line class codes or other switch specific provisioning methods), and shall route directory assistance calls from BTI customers to BTI directory assistance operators at BTI's option.
- Where capacity exists, BellSouth shall assign each BTI customer line the class of services designated by BTI (e.g., using line class codes or other switch specific provisioning methods) and shall route operator calls from BTI customers to BTI operators at BTI's option. For example, BellSouth may translate 0- and 0+ intraLATA traffic, and route the call through appropriate trunks to BTI Operator Services Position System (OSPS). Calls from Local Switching must pass the ANI-II digits unchanged.
- 4.1.4.19 Local Switching shall be offered in accordance with the technical specifications set forth in the applicable industry standard references.
- 4.1.5 <u>Interface Requirements.</u> BellSouth shall provide all technically feasible and available interfaces to loops including the following interfaces to loops:
- 4.1.5.1 Standard Tip/Ring interface including loopstart or groundstart, on-hook signaling (e.g., for calling number, calling name and message waiting lamp);
- 4.1.5.2 Coin phone signaling;
- 4.1.5.3 Basic Rate Interface ISDN adhering to appropriate Telcordia (formerly BellCore) Technical Requirements;
- 4.1.5.4 Two-wire analog interface to PBX;
- 4.1.5.5 Four-wire analog interface to PBX;

4.1.5.6 Four-wire DS1 interface to PBX or customer provided equipment (e.g. computers and voice response systems); 4.1.5.7 Primary Rate ISDN to PBX adhering to ANSI standards Q.931, Q.932 and appropriate Telcordia (formerly BellCore) Technical Requirements; 4.1.5.8 Switched Fractional DS1 with capabilities to configure Nx64 channels (where N =1 to 24); and Loops adhering to Telcordia (formerly BellCore) TR-NWT-08 and TR-NWT-303 4.1.5.9 specifications to interconnect Digital Loop Carriers. 4.1.6 BellSouth shall provide access to the following but not limited to: 4.1.6.1 SS7 Signaling Network or Multi-Frequency trunking if requested by BTI; 4.1.6.2 Interface to BTI operator services systems or Operator Services through appropriate trunk interconnections for the system; and 4.1.6.3 Interface to BTI Directory Assistance Services through the BTI switched network or to Directory Assistance Services through the appropriate trunk interconnections for the system; and 950 access or other BTI required access to interexchange carriers as requested through appropriate trunk interfaces. 4.2 **Tandem Switching** 4.2.1 Definition Tandem Switching is the function that establishes a communications path between two switching offices through a third switching office (the Tandem switch). 4.2.2 **Technical Requirements** Tandem Switching shall have the same capabilities or equivalent capabilities as those described in Bell Communications Research TR-TSY-000540 Issue 2R2, Tandem Supplement, 6/1/90. The requirements for Tandem Switching include, but are not limited to the following: 4.2.2.1 Tandem Switching shall provide signaling to establish a tandem connection; 4.2.2.2 Tandem Switching will provide screening as jointly agreed to by BTI and BellSouth; 4.2.2.3 Tandem Switching shall provide Advanced Intelligent Network triggers

supporting AIN features where such routing is not available from the originating

end office switch, to the extent such Tandem switch has such capability;

4.2.2.4 Tandem Switching shall provide access to Toll Free number portability database as designated by BTI; 4.2.2.5 Tandem Switching shall provide all trunk interconnections discussed under the "Network Interconnection" section (e.g., SS7, MF, DTMF, DialPulse, PRI-ISDN, DID, and CAMA-ANI (if appropriate for 911)); 4.2.2.6 Tandem Switching shall provide connectivity to PSAPs where 911 solutions are deployed and the tandem is used for 911; and 4.2.2.7 Where appropriate, Tandem Switching shall provide connectivity to transit traffic to and from other carriers. 4.2.3 Tandem Switching shall accept connections (including the necessary signaling and trunking interconnections) between end offices, other tandems, IXCs, ICOs, CAPs and CLEC switches. 4.2.4 Tandem Switching shall provide local tandeming functionality between two end offices including two offices belonging to different CLECs (e.g., between a CLEC end office and the end office of another CLEC). 4.2.5 Tandem Switching shall preserve CLASS/LASS features and Caller ID as traffic is processed. 4.2.6 Tandem Switching shall record billable events and send them to the area billing centers designated by BTI. Tandem Switching will provide recording of all billable events as jointly agreed to by BTI and BellSouth. 4.2.7 Upon a reasonable request from BTI, BellSouth shall perform routine testing and fault isolation on the underlying switch that is providing Tandem Switching and all its interconnections. The results and reports of the testing shall be made immediately available to BTI. 4.2.8 BellSouth shall maintain BTI's trunks and interconnections associated with Tandem Switching at least at parity to its own trunks and interconnections. 4.2.9 BellSouth shall control congestion points and network abnormalities. All traffic will be restricted in a non-discriminatory manner. 4.2.10 Selective Call Routing through the use of line class codes is not available through the use of tandem switching. Selective Call Routing through the use of line class codes is an end office capability only. Detailed primary and overflow routing plans for all interfaces available within BellSouth's switching network shall be mutually agreed to by BTI and BellSouth.

local switch.

4.2.11

Tandem Switching shall process originating toll-free traffic received from BTI's

- 4.2.12 In support of AIN triggers and features, Tandem Switching shall provide SSP capabilities when these capabilities are not available from the Local Switching Network Element, to the extent such Tandem Switch has such capability.
- 4.2.13 **Interface Requirements**
- 4.2.13.1 Tandem Switching shall provide interconnection to the E911 PSAP where the underlying Tandem is acting as the E911 Tandem.
- 4.2.13.2 Tandem Switching shall interconnect, with direct trunks, to all carriers with which BellSouth interconnects.
- 4.2.13.3 BellSouth shall provide all signaling necessary to provide Tandem Switching with no loss of feature functionality.
- 4.2.13.4 Tandem Switching shall interconnect with BTI's switch, using two-way trunks, for traffic that is transiting via BellSouth's network to interLATA or intraLATA carriers. At BTI's request, Tandem Switching shall record and keep records of traffic for billing.
- Tandem Switching shall provide an alternate final routing pattern for BTI's traffic 4.2.13.5 overflowing from direct end office high usage trunk groups.
- 4.2.13.6 Tandem Switching shall be equal to or better than the requirements for Tandem Switching set forth in the applicable technical references.
- 4.3 AIN Selective Carrier Routing for Operator Services, Directory Assistance and Repair Centers
- 4.3.1 BellSouth will provide AIN Selective Carrier Routing at the request of BTI. AIN Selective Carrier Routing will provide BTI with the capability of routing operator calls, 0+ and 0- and 0+ NPA (LNPA) 555-1212 directory assistance, 1+411 directory assistance and 611 repair center calls to pre-selected destinations.
- 4.3.2 BTI shall order AIN Selective Carrier Routing through its Account Team. AIN Selective Carrier Routing must first be established regionally and then on a per central office, per state basis.
- 4.3.3 AIN Selective Carrier Routing is not available in DMS 10 switches.
- 4.3.4 Where AIN Selective Carrier Routing is utilized by BTI, the routing of BTI's end user calls shall be pursuant to information provided by BTI and stored in BellSouth's AIN Selective Carrier Routing Service Control Point database. AIN Selective Carrier Routing shall utilize a set of Line Class Codes (LCCs) unique to a basic class of service assigned on an 'as needed' basis. The same LCCs will be assigned in each central office where AIN Selective Carrier Routing is established.

Version 3000: 10/10/00

- 4.3.5 Upon ordering of AIN Selective Carrier Routing Regional Service, BTI shall remit to BellSouth the Regional Service Order non-recurring charges set forth in Exhibit D of this Attachment. There shall be a non-recurring End Office Establishment Charge per office due at the addition of each central office where AIN Selective Carrier Routing will be utilized. Said non-recurring charge shall be as set forth in Exhibit D of this Attachment. For each BTI end user activated, there shall be a non-recurring End User Establishment charge as set forth in Exhibit D of this Attachment, payable to BellSouth pursuant to the terms of the General Terms and Conditions, incorporated herein by this reference. BTI shall pay the AIN Selective Carrier Routing Per Query Charge set forth in Exhibit D of this Attachment.
- 4.3.6 This Regional Service Order non-recurring charge will be non-refundable and will be paid with 1/2 due up-front with the submission of all fully completed required forms, including: Regional Selective Carrier Routing (SCR) Order Request-Form A, Central Office AIN Selective Carrier Routing (SCR) Order Request Form B, AIN\_SCR Central Office Identification Form Form C, AIN\_SCR Routing Options Selection Form Form D, and Routing Combinations Table Form E. BellSouth has 30 days to respond to the client's fully completed firm order as a Regional Service Order. With the delivery of this firm order response to the client, BellSouth considers that the delivery schedule of this service commences. The remaining 1/2 of the Regional Service Order payment must be paid when at least 100% of the Central Offices listed on the original order have been turned up for the service.
- 4.3.7 The non-recurring End Office Establishment Charge will be billed to the client following BellSouth's normal monthly billing cycle for this type of order.
- 4.3.8 End-User Establishment Orders will not be turned-up until the second payment is received for the Regional Service Order. The non-recurring End-User Establishment Charges will be billed to the client following BellSouth's normal monthly billing cycle for this type of order.
- 4.3.9 Additionally, the AIN Selective Carrier Routing Per Query Charge will be billed to the client following the normal billing cycle for per query charges.
- 4.3.10 All other network components needed, for example, unbundled switching and unbundled local transport, etc, will be billed per contracted rates.

#### 4.4 Packet Switching Capability

#### 4.4.1 Definition

The packet switching capability network element is defined as the function of routing or forwarding packets, frames, cells or other data units based on address or

other routing information contained in the packets, frames, cells or other data units.

- 4.4.2 BellSouth shall be required to provide non-discriminatory access to unbundled packet switching capability only where each of the following conditions are satisfied:
- 4.4.2.1 BellSouth has deployed digital loop carrier systems, including but not limited to, integrated digital loop carrier or universal digital loop carrier systems; or has deployed any other system in which fiber optic facilities replace copper facilities in the distribution section (e.g., end office to remote terminal, pedestal or environmentally controlled vault);
- 4.4.2.2 There are no spare copper loops capable of supporting the xDSL services BTI seeks to offer:
- 4.4.2.3 BellSouth has not permitted BTI to deploy a DSLAM at the remote terminal, pedestal or environmentally controlled vault or other interconnection point, nor has BTI obtained a virtual collocation arrangement at these sub-loop interconnection points as defined by 47 CFR § 51.319 (b); and
- 4.4.2.4 BellSouth has deployed packet switching capability for its own use.
- 4.4.3 If there is a dispute as to whether BellSouth must provide Packet Switching, such dispute will be resolved according to the dispute resolution process set forth in Section 12 of the General Terms and Conditions of this Agreement, incorporated herein by this reference.

#### 4.5 Rates

The prices that BTI shall pay to BellSouth for Network Elements and Other Services are set forth in Exhibit D to this Attachment. If BTI purchases a service(s) from a tariff, all terms and conditions and rates as set forth in such tariff shall apply.

# 4.6 **Operational Support Systems (OSS)**

The terms, conditions and rates for OSS are as set forth in Section 2.13 of this Attachment.

#### 5. Unbundled Network Element Combinations

5.1. Unbundled Network Element Combinations shall include: 1) Enhanced Extended Links (EELs); 2) UNE Loops/Special Access Combinations; 3) Loop/Port Combinations; and 4) Transport Combinations.

5.2. For purposes of this Section, references to "Currently Combined" network elements shall mean that such network elements are in fact already combined by BellSouth in the BellSouth network to provide service to a particular end user at a particular location.

# **5.3.** Enhanced Extended Links (EELs)

- Where facilities permit and where necessary to comply with an effective FCC and/or State Commission order, or as otherwise mutually agreed by the Parties, BellSouth shall offer access to loop and transport combinations, also known as the Enhanced Extended Link ("EEL") as defined in Section 5.3.2 below.
- 5.3.2 Subject to Section 5.3.3 below, BellSouth will provide access to the EEL in the combinations set forth in Section 5.3.4 following. This offering is intended to provide connectivity from an end user's location through that end user's SWC to BTI's POP serving wire center. The circuit must be connected to BTI's switch for the purpose of provisioning telephone exchange service to BTI's end-user customers. Except as set forth in Section 5.3.5.1.3 below, the EEL will be connected to BTI's facilities in BTI's collocation space at the POP SWC, or BTI may purchase BellSouth's access facilities between BTI's POP and BTI's collocation space at the POP SWC.
- 5.3.3 BellSouth shall provide EEL combinations to BTI in Georgia regardless of whether or not such EELs are Currently Combined. In all other states, BellSouth shall make available to BTI those EEL combinations described in Section 5.3.4 below only to the extent such combinations are Currently Combined. Furthermore, BellSouth will make available EEL combinations to BTI in density Zone 1, as defined in 47 CFR 69.123 as of January 1, 1999, in the Atlanta, GA; Miami, FL; Orlando, FL; Ft. Lauderdale, FL; Charlotte-Gastonia-Rock Hill, NC; Greensboro-Winston Salem-High Point, NC; Nashville, TN; and New Orleans, LA, MSAs regardless of whether or not such EELs are Currently Combined. Except as stated above, EELs will be provided to BTI only to the extent such network elements are Currently Combined.
- 5.3.4 EEL Combinations
- 5.3.4.1 DS1 Interoffice Channel + DS1 Channelization + 2-wire VG Local Loop
- 5.3.4.2 DS1 Interoffice Channel + DS1 Channelization + 4-wire VG Local Loop
- 5.3.4.3 DS1 Interoffice Channel + DS1 Channelization + 2-wire ISDN Local Loop
- 5.3.4.4 DS1 Interoffice Channel + DS1 Channelization + 4-wire 56 kbps Local Loop
- 5.3.4.5 DS1 Interoffice Channel + DS1 Channelization + 4-wire 64 kbps Local Loop
- 5.3.4.6 DS1 Interoffice Channel + DS1 Local Loop

- 5.3.4.7 DS3 Interoffice Channel + DS3 Local Loop
   5.3.4.8 STS-1 Interoffice Channel + STS-1 Local Loop
- 5.3.4.9 DS3 Interoffice Channel + DS3 Channelization + DS1 Local Loop
- 5.3.4.10 STS-1 Interoffice Channel + DS3 Channelization + DS1 Local Loop
- 5.3.4.11 2-wire VG Interoffice Channel + 2-wire VG Local Loop
- 5.3.4.12 4wire VG Interoffice Channel + 4-wire VG Local Loop
- 5.3.4.13 4-wire 56 kbps Interoffice Channel + 4-wire 56 kbps Local Loop
- 5.3.4.14 4-wire 64 kbps Interoffice Channel + 4-wire 64 kbps Local Loop
- 5.3.5. Special Access Service Conversions
- BTI may not convert special access services to combinations of loop and transport network elements, whether or not BTI self-provides its entrance facilities (or obtains entrance facilities from a third party), unless BTI uses the combination to provide a significant amount of local exchange service, in addition to exchange access service, to a particular customer. To the extent BTI requests to convert any special access services to combinations of loop and transport network elements at UNE prices, BTI shall provide to BellSouth a letter certifying that BTI is providing a significant amount of local exchange service (as described in this Section) over such combinations. The certification letter shall also indicate under what local usage option BTI seeks to qualify for conversion of special access circuits. BTI shall be deemed to be providing a significant amount of local exchange service over such combinations if one of the following options is met:
- 5.3.5.1.1 BTI certifies that it is the exclusive provider of an end user's local exchange service. The loop-transport combinations must terminate at BTI's collocation arrangement in at least one BellSouth central office. This option does not allow loop-transport combinations to be connected to BellSouth's tariffed services. Under this option, BTI is the end user's only local service provider, and thus, is providing more than a significant amount of local exchange service. BTI can then use the loop-transport combinations that serve the end user to carry any type of traffic, including using them to carry 100 percent interstate access traffic; or
- 5.3.5.1.2 BTI certifies that it provides local exchange and exchange access service to the end user customer's premises and handles at least one third of the end user customer's local traffic measured as a percent of total end user customer local dialtone lines; and for DS1 circuits and above, at least 50 percent of the activated channels on the loop portion of the loop-transport combination have at least 5 percent local voice traffic individually, and the entire loop facility has at least 10 percent local voice traffic. When a loop-transport combination includes multiplexing, each of the

individual DS1 circuits must meet this criteria. The loop-transport combination must terminate at BTI's collocation arrangement in at least one BellSouth central office. This option does not allow loop-transport combinations to be connected to BellSouth tariffed services; or

- 5.3.5.1.3 BTI certifies that at least 50 percent of the activated channels on a circuit are used to provide originating and terminating local dialtone service and at least 50 percent of the traffic on each of these local dialtone channels is local voice traffic, and that the entire loop facility has at least 33 percent local voice traffic. When a loop-transport combination includes multiplexing, each of the individual DS1 circuits must meet this criteria. This option does not allow loop-transport combinations to be connected to BellSouth's tariffed services. Under this option, collocation is not required. BTI does not need to provide a defined portion of the end user's local service, but the active channels on any loop-transport combination, and the entire facility, must carry the amount of local exchange traffic specified in this option.
- 5.3.5.2 In addition, there may be extraordinary circumstances where BTI is providing a significant amount of local exchange service, but does not qualify under any of the three options set forth in Section 5.3.5.1.3. In such case, BTI may petition the FCC for a waiver of the local usage options set forth in the June 2, 2000 Order. If a waiver is granted, then upon BTI's request the Parties shall amend this Agreement to the extent necessary to incorporate the terms of such waiver for such extraordinary circumstance.
- 5.3.5.3 After BellSouth has provisioned the loop-transport combination to CLEC, or converted a special access arrangement to a loop-transport combination, BellSouth may at its sole discretion audit BTI records in order to verify the type of traffic being transmitted over combinations of loop and transport network elements. The audit shall be conducted by a third party independent auditor, and BTI shall be given thirty days written notice of scheduled audit. Such audit shall occur no more than one time in a calendar year, unless results of an audit find noncompliance with the significant amount of local exchange service requirement, as defined in Section 5.3.5.1.3 above. In the event of noncompliance, BTI shall reimburse BellSouth for the cost of the audit. If, based on its audits, BellSouth concludes that BTI is not providing a significant amount of local exchange traffic over the combinations of loop and transport network elements, BellSouth may file a complaint with the appropriate Commission, pursuant to the dispute resolution process as set forth in the Interconnection Agreement seeking a determination that BellSouth may convert such combinations of loop and transport network elements to special access services and obtain appropriate retroactive reimbursement from BTI.

- 5.3.5.4 BTI may convert special access circuits to combinations of loop and transport UNEs pursuant to the terms of this Section and subject to the termination provisions in the applicable special access tariffs, if any.
- 5.3.5.5 For the purposes of this section 5.3.5, traffic is local if it is defined as such in BTI's state approved local exchange tariff and/or subject to reciprocal compensation arrangement between BellSouth and BTI.
- 5.3.6 Rates
- 5.3.6.1 Georgia
- 5.3.6.2 The non-recurring and recurring rates for the EEL Combinations of network elements set forth in 5.3.4, whether Currently Combined or new, are as set forth in Exhibit D of this Attachment.
- 5.3.6.3 On an interim basis, for combinations of loop and transport network elements not set forth in Section 5.3.4, where the elements are not Currently Combined but are ordinarily combined in BellSouth's network, the non-recurring and recurring charges for such UNE combinations shall be the sum of the stand-alone non-recurring and recurring charges of the network elements which make up the combination. These interim rates shall be subject to true-up based on the Commission's review of BellSouth's cost studies.
- 5.3.6.4 To the extent that BTI seeks to obtain other combinations of network elements that BellSouth ordinarily combines in its network which have not been specifically priced by the Commission when purchased in combined form, BTI, at its option, can request that such rates be determined pursuant to the BFR/NBR process set forth in this Agreement.
- 5.3.6.5 All Other States
- 5.3.6.5.1 Subject to Section 5.3.2 and 5.3.3 preceding, for all other states, the non-recurring and recurring rates for the Currently Combined EEL combinations set forth in Section 5.3.4 and other Currently Combined network elements will be the sum of the recurring rates for the individual network elements plus a non recurring charge set forth in Exhibit D of this Attachment.
- 5.3.6.6 Multiplexing
- 5.3.6.6.1 Where multiplexing functionality is required in connection with loop and transport combinations, such multiplexing will be provided at the rates and on the terms set forth in this Agreement.
- 5.4 Other Network Element Combinations

- 5.4.1.1 In the state of Georgia, BellSouth shall make available to BTI, in accordance with Section 5.4.2.1 below: (1) combinations of network elements other than EELs that are Currently Combined; and (2) combinations of network elements other than EELs that are not Currently Combined but that BellSouth ordinarily combines in its network. In all other states, BellSouth shall make available to BTI, in accordance with Section 5.4.2.2 below, combinations of network elements other than EELs only to the extent such combinations are Currently Combined.
- 5.4.2 Rates
- 5.4.2.1 Georgia
- 5.4.2.1.1 The non-recurring and recurring rates for Other Network Element combinations, whether Currently Combined or new, are as set forth in Exhibit D of this Attachment.
- On an interim basis, for Other Network Element combinations where the elements are not Currently Combined but are ordinarily combined in BellSouth's network, the non-recurring and recurring charges for such UNE combinations shall be the sum of the stand-alone non-recurring and recurring charges of the network elements which make up the combination. These interim rates shall be subject to true-up based on the Commission's review of BellSouth's cost studies.
- 5.4.2.1.3 To the extent that BTI seeks to obtain other combinations of network elements that BellSouth ordinarily combines in its network which have not been specifically priced by the Commission when purchased in combined form, BTI, at its option, can request that such rates be determined pursuant to the BFR/NBR process set forth in this Agreement.
- 5.4.2.2 All Other States
- 5.4.2.2.1 For all other states, the non-recurring and recurring rates for the Other Network Element Combinations that are Currently Combined will be the sum of the recurring rates for the individual network elements plus a non recurring charge set forth in Exhibit D of this Attachment.

# 5.5 **UNE/Special Access Combinations**

5.5.1 Additionally, BellSouth shall make available to BTI a combination of an unbundled loop and tariffed special access interoffice facilities. To the extent BTI will require multiplexing functionality in connection with such combination, BellSouth will provide access to multiplexing within the central office pursuant to the terms, conditions and rates set forth in its Access Services Tariffs. The tariffed special access interoffice facilities and any associated tariffed services, including but not limited to multiplexing, shall not be eligible for conversion to UNEs as described in Section 5.3.7.

- 5.5.2 Rates
- 5.5.2.1 The non-recurring and recurring rates for UNE/Special Access Combinations will be the sum of the unbundled loop rates as set forth in Exhibit D and the interoffice transport rates and multiplexing rates as set forth in the Access Services Tariff.

# 5.6 **Port/Loop Combinations**

- 5.6.1 At BTI's request, BellSouth shall provide access to combinations of port and loop network elements, as set forth in Section 5.6.3 below, that are Currently Combined in BellSouth's network except as specified in Sections 5.6.1.1 and 5.6.1.2 below.
- 5.6.1.1 BellSouth shall not provide combinations of port and loop network elements on an unbundled basis in locations where, pursuant to FCC rules, BellSouth is not required to provide circuit switching as an unbundled network element.
- In accordance with effective and applicable FCC rules, BellSouth shall not be required to provide circuit switching as an unbundled network element in density Zone 1, as defined in 47 CFR 69.123 as of January 1, 1999 of the Atlanta, GA; Miami, FL; Orlando, FL; Ft. Lauderdale, FL; Charlotte-Gastonia-Rock Hill, NC; Greensboro-Winston Salem-High Point, NC; Nashville, TN; and New Orleans, LA, MSAs to BTI if BTI's customer has 4 or more DS0 equivalent lines.
- 5.6.2 Combinations of port and loop network elements provide local exchange service for the origination or termination of calls. BellSouth shall make available the following loop and port combinations at the terms and at the rates set forth below:
- In Georgia, BellSouth shall provide to BTI combinations of port and loop network elements to BTI on an unbundled basis regardless of whether or not such combinations are Currently Combined except in those locations where BellSouth is not required to provide circuit switching, as set forth in Section 5.6.1.2 above. The rates for such combinations shall be the cost based rates set forth in Exhibit D of this Attachment.
- In all other states, BellSouth shall provide to BTI combinations of port and loop network elements on an unbundled basis if such combinations are Currently Combined, except in those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 5.6.1.1 and 5.6.1.2 above. The rates for such combinations shall be the cost based rates set forth in Exhibit D of this Attachment.
- 5.6.2.3 In all states other than Georgia, except in those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 5.6.1.1 and 5.6.1.2, BellSouth shall provide to BTI combinations of port and loop network elements that are not Currently Combined. The rates for such combinations shall be negotiated by the Parties.

- 5.6.2.4 In those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 5.6.1.1 and 5.6.1.2, BellSouth shall provide to BTI combinations of port and loop network elements whether or not such combinations are Currently Combined. The rates for Currently Combined combinations are the market based rates as set forth in Exhibit D. The rates for not Currently Combined combinations shall be negotiated by the Parties.
- 5.6.3 Combination Offerings
- 5.6.3.1 2-wire voice grade port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.6.3.2 2-wire voice grade DID port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.6.3.3 2-wire CENTREX port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.6.3.4 2-wire ISDN Basic Rate Interface, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.6.3.5 2-wire ISDN Primary Rate Interface, DS1 loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.6.3.6 4-wire DS1 Trunk port, DS1 Loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.

#### 5.7 Rates

The prices that BTI shall pay to BellSouth for Network Elements and Other Services are set forth in Exhibit D to this Attachment. If BTI purchases a service(s) from a tariff, all terms and conditions and rates as set forth in such tariff shall apply.

## 5.8 **Operational Support Systems (OSS)**

The terms, conditions and rates for OSS are as set forth in Section 2.13 of this Attachment.

Version 3Q00: 10/10/00

## 6. Transport, Channelization and Dark Fiber

All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of unbundled transport and dark fiber.

## 6.1 **Transport**

- 6.1.1 BellSouth shall provide nondiscriminatory access, in accordance with FCC Rule 51.311 and Section 251(c)(3) of the Act, to interoffice transmission facilities on an unbundled basis to BTI for the provision of a telecommunications service .Interoffice transmission facility network elements include:
- 6.1.1.1 Dedicated transport, defined as BellSouth's transmission facilities, including but not limited to DS1, DS3, and OCN levels, is dedicated to a particular customer or carrier that provides telecommunications between wire centers or switches owned by BellSouth, or between wire centers and switches owned by BellSouth and BTI.
- Dark Fiber transport, defined as BellSouth's optical transmission facilities without attached signal regeneration, multiplexing, aggregation or other electronics;
- 6.1.1.3 Common (Shared) transport, defined as transmission facilities shared by more than one carrier, including BellSouth, between end office switches, between end office switches and tandem switches, and between tandem switches, in BellSouth's network.
- 6.2 BellSouth shall:
- 6.2.1 Provide BTI exclusive use of interoffice transmission facilities dedicated to a particular customer or carrier, or shared use of the features, functions, and capabilities of interoffice transmission facilities shared by more than one customer or carrier;
- Provide all technically feasible transmission facilities, features, functions, and capabilities that BTI could use to provide telecommunications services;
- 6.2.3 Permit, to the extent technically feasible, BTI to connect such interoffice facilities to equipment designated by BTI, including but not limited to, BTI's collocated facilities; and
- 6.2.4 Permit, to the extent technically feasible, BTI to obtain the functionality provided by BellSouth's digital cross-connect systems in the same manner that BellSouth provides such functionality to interexchange carriers.
- 6.3 Common (Shared) Transport

Version 3Q00: 10/10/00

6.3.1	Definition	of Common	(Shared)	) Trans	port

- 6.3.1.1 Common (Shared) Transport is an interoffice transmission path between two BellSouth end-offices, BellSouth end-office and a local tandem, or between two local tandems. Where BellSouth Network Elements are connected by intra-office wiring, such wiring is provided as a part of the Network Elements and is not Common (Shared) Transport.
- 6.3.2 <u>Technical Requirements of Common (Shared) Transport</u>
- 6.3.2.1 Common (Shared) Transport provided on DS1 or VT1.5 circuits, shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for Central Office to Central Office ("CO to CO") connections in the appropriate industry standards.
- 6.3.2.2 Common (Shared) Transport provided on DS3 circuits, STS-1 circuits, and higher transmission bit rate circuits, shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for CO to CO connections in the appropriate industry standards.
- 6.3.2.3 BellSouth shall be responsible for the engineering, provisioning, and maintenance of the underlying equipment and facilities that are used to provide Common (Shared) Transport.
- 6.3.2.4 At a minimum, Common (Shared) Transport shall meet all of the requirements set forth in the applicable industry standard technical references.

# 6.4 **Dedicated Transport**

- 6.4.1 Definitions
- Dedicated Transport is defined as BellSouth transmission facilities dedicated to a particular customer or carrier that provide telecommunications between wire centers owned by BellSouth or requesting telecommunications carriers, or between switches owned by BellSouth or requesting telecommunications carriers.
- 6.4.3 <u>Unbundled Local Channel</u>
- Unbundled Local Channel is the dedicated transmission path between BTI's Point of Presence and the BellSouth Serving Wire Center's collocation.
- 6.4.5 Unbundled Interoffice Channel.
- 6.4.6 Unbundled Interoffice Channel is the dedicated transmission path that provides telecommunication between BellSouth's Serving Wire Centers' collocations.

6.4.7 BellSouth shall offer Dedicated Transport on a nondiscriminatory basis in each of the following ways: 6.4.7.1 As capacity on a shared UNE facility. 6.4.7.2 As a circuit (e.g., DS0, DS1, DS3, OCN) dedicated to BTI. This circuit shall consist of an Unbundled Local Channel or an Unbundled Interoffice Channel or both. 6.4.8 When Dedicated Transport is provided it shall include: 6.4.8.1 Transmission equipment such as, line terminating equipment, amplifiers, and regenerators; 6.4.8.2 Inter-office transmission facilities such as optical fiber, copper twisted pair, and coaxial cable. 6.4.9 Rates for Dedicated Transport are listed in this Attachment. For those states that do not contain rates in this Attachment the rates in the applicable State Access Tariff will apply as interim rates. When final rates are developed, these interim rates will be subject to true up, and the Parties will amend the Agreement to reflect the new rates. 6.4.10 **Technical Requirements** 6.4.10.1 This Section sets forth technical requirements for all Dedicated Transport. 6.4.10.2 When BellSouth provides Dedicated Transport, the entire designated transmission service (e.g., DS0, DS1, DS3, OCN) shall be dedicated to BTI designated traffic. 6.4.10.3 BellSouth shall offer Dedicated Transport in all technologies that become available including, but not limited to, (1) DS0, DS1 and DS3 transport services, and if and when available, OC3, OC12, OC48, and OC192 and (2) SONET at available transmission bit rates. 6.4.10.4 For DS1 or VT1.5 circuits, Dedicated Transport shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for Customer Interface to Central Office ("CI to CO") connections in the appropriate industry standards. 6.4.10.5 Where applicable, for DS3, Dedicated Transport shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for CI to CO connections in the appropriate industry standards. 6.4.10.6 BellSouth shall offer the following interface transmission rates for Dedicated Transport:

6.4.10.6.1	DS0 Equivalent;
6.4.10.6.2	DS1;
6.4.10.6.3	DS3;
6.4.10.6.4	OCN;
6.4.10.6.5	SDH (Synchronous Digital Hierarchy) Standard interface rates in accordance with International Telecommunications Union (ITU) Recommendation G.707 and Plesiochronous Digital Hierarchy (PDH) rates per ITU Recommendation G.704.
6.4.10.6.6	When Dedicated Transport is provided, BellSouth shall design it according to BellSouth's network infrastructure to allow for the termination points specified by BTI.
6.4.11	At a minimum, Dedicated Transport shall meet each of the requirements set forth in the applicable industry technical references.
6.4.11.1	BellSouth Technical References:
6.4.11.2	TR-TSY-000191 Alarm Indication Signals Requirements and Objectives, Issue 1, May 1986.
6.4.11.3	TR 73501 LightGate <sup>®</sup> Service Interface and Performance Specifications, Issue D, June 1995.
6.4.11.4	TR 73525 MegaLink®Service, MegaLink Channel Service and MegaLink Plus Service Interface and Performance Specifications, Issue C, May 1996.
6.4.12	Provided that the facility is used to transport a significant amount of local exchange services as defined in Section 5.3.5.1.3, BTI shall be entitled to convert existing interoffice transmission facilities (i.e., special access) to the corresponding interoffice transport network element option.
6.5	<b>Unbundled Channelization</b>
6.5.1	BellSouth agrees to offer access to Unbundled Channelization when available pursuant to following terms and conditions and at the rates set forth in the Attachment. Channelization will be offered with both the high and the low speed sides to be connected to collocation.
6.5.2	Definition
6.5.2.1	Unbundled Channelization (UC) provides the multiplexing capability that will allow a DS1 (1.544 Mbps) or DS3 (44.736 Mbps) or STS-1 Unbundled Network Element (UNE) or collocation cross-connect to be multiplexed or channelized at a

BellSouth central office. This can be accomplished through the use of a standalone multiplexer or a digital cross-connect system at the discretion of BellSouth. Once UC has been installed, BTI can have channels activated on an as-needed basis by having BellSouth connect lower level UNEs via Central Office Channel Interfaces (COCIs).

- 6.5.3 Channelization capabilities will be as follows:
- 6.5.3.1 DS3 Channelization System: An element that channelizes a DS3 signal into 28 DS1s/STS-1s.
- DS1 Channelization System: An element that channelizes a DS1 signal into 24 DS0s.
- 6.5.3.3 Central Office Channel Interfaces (COCI): Elements that can be activated on a channelization system.
- DS1 Central Office Channel Interface elements can be activated on a DS3 Channelization System.
- Voice Grade and Digital Data Central Office Channel Interfaces can be activated on a DS1 Channelization System.
- 6.5.6 AMI and B8ZS line coding with either Super Frame (SF) and Extended Super Frame (ESF) framing formats will be supported as options.
- 6.5.7 COCI will be billed on the lower level UNE order that is interfacing with the UC arrangement and will have to be compatible with those UNEs.
- 6.5.8 Technical Requirements
- 6.5.8.1 In order to assure proper operation with BellSouth provided central office multiplexing functionality, the customer's channelization equipment must adhere strictly to form and protocol standards. Separate standards exist for the multiplex channel bank, for voice frequency encoding, for various signaling schemes, and for subrate digital access.
- 6.5.8.2 DS0 to DS1 Channelization
- The DS1 signal must be framed utilizing the framing structure defined in ANSI T1.107, Digital Hierarchy Formats Specifications and ANSI T1.403.02, DS1 Robbed-bit Signaling State Definitions. DS0 to DS1 Channelization requirements are essential the same as defined in BellSouth Technical Reference 73525, MegaLink<sup>®</sup> Service, MegaLink<sup>®</sup> Channel Service, MegaLink<sup>®</sup> Plus Service, and MegaLink<sup>®</sup> Light Service Interface and Performance Specification.
- 6.5.8.3 DS1 to DS3 Channelization

- 6.5.8.3.1 The DS3 signal must be framed utilizing the framing structure define in ANSI T1.107, Digital Hierarchy Formats Specifications. DS1 to DS3 Channelization requirements are essentially the same as defined in BellSouth Technical Reference 73501, LightGate<sup>®</sup> Service Interface and Performance Specifications. The asynchronous M13 multiplex format (combination of M12 and M23 formats) is specified for terminal equipment that multiplexes 28 DS1s into a DS3.
- 6.5.8.4 DS1 to STS Channelization
- 6.5.8.4.1 The STS-1 signal must be framed utilizing the framing structure define in ANSI T1.105, Synchronous Optical Network (SONET) Basic Description Including Multiplex Structure, Rates and Formats and T1.105.02, Synchronous Optical Network (SONET) Payload Mappings. DS1 to STS Channelization requirements are essentially the same as defined in BellSouth Technical Reference TR 73501, LightGate® Service Interface and Performance Specifications.

# 6.6 **Dark Fiber Transport**

- 6.6.1 Definition
- Dark Fiber is optical transmission facilities without attached multiplexing, aggregation or other electronics that connects two points within BellSouth's network. Dark Fiber is unused strands of optical fiber. It may be strands of optical fiber existing in aerial or underground structure. No line terminating elements terminated to such strands to operationalize its transmission capabilities will be available.
- 6.6.3 Requirements
- 6.6.3.1 BellSouth shall make available Dark Fiber where it exists in BellSouth's network and where, as a result of future building or deployment, it becomes available. If BellSouth has pre-existing plans to make specific use of the fiber within a two-year period, there is no requirement to provide said fiber to BTI, provided that upon BTI's request, BellSouth shall provide written certification to BTI that BellSouth has plans to use such fiber and shall describe generally its plans to make such use subject to BTI signing a separate confidentiality agreement with respect to such plans.
- 6.6.3.2 If the requested dark fiber has any lightwave repeater equipment interspliced to it, BellSouth will remove such equipment at BTI's request subject to time and materials charges.
- 6.6.3.3 BTI may test the quality of the Dark Fiber to confirm its usability and performance specifications.
- BellSouth shall use its best efforts to provide to BTI information regarding the location, availability and performance of Dark Fiber within ten (10) business days

for a records based answer and twenty (20) business days for a field based answer, after receiving a request from BTI ("Request"). Within such time period, BellSouth shall send written confirmation of availability of the Dark Fiber ("Confirmation"). From the time of the Request to one hundred and twenty (120) days after Confirmation, BellSouth shall hold such requested Dark Fiber for BTI's use and may not allow any other Party to use such media, including BellSouth. If a Dark Fiber firm order is not received within the one hundred and twenty day period, the Dark Fiber will revert to BellSouth's Dark Fiber inventory.

- 6.6.3.5 BellSouth shall use its best efforts to make Dark Fiber available to BTI within thirty (30) business days after it receives written confirmation from BTI that the Dark Fiber previously deemed available by BellSouth is wanted for use by BTI. This includes identification of appropriate connection points (e.g., Light Guide Interconnection (LGX) or splice points) to enable BTI to connect or splice BTI provided transmission media (e.g., optical fiber) or equipment to the Dark Fiber.
- Dark Fiber shall meet BellSouth Technical Reference TR73603, Unbundled Dark Fiber Technical Specifications and/or industry standards, where available.
- 6.6.3.7 BTI may splice and test Dark Fiber obtained from BellSouth using BTI or BTI designated personnel. BellSouth shall provide appropriate interfaces to allow splicing and testing of Dark Fiber. BellSouth shall provide an excess cable length of 25 feet minimum (for fiber in underground conduit) to allow the uncoiled fiber to reach from the manhole to a splicing van.
- 6.7 Rates
- 6.7.1 The prices that BTI shall pay to BellSouth for Network Elements and Other Services are set forth in Exhibit D to this Attachment. If BTI purchases a service(s) from a tariff, all terms and conditions and rates as set forth in such tariff shall apply.
- 6.8 **Operational Support Systems (OSS)**

The terms, conditions and rates for OSS are as set forth in Section 2.13 of this Attachment.

7. BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service

All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of 8XX Access Ten Digit Screening Services.

- 7.1 BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service database
- 7.1.1 The BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service database (herein known as 8XX SCP) is a SCP that contains customer record information

and functionality to provide call-handling instructions for 8XX calls. The 8XX SCP IN software stores data downloaded from the national SMS and provides the routing instructions in response to queries from the SSP or tandem. The BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service (herein know as 8XX TFD), utilizes the 8XX SCP to provide identification and routing of the 8XX calls, based on the ten digits dialed. 8XX TFD is provided with or without POTS number delivery, dialing number delivery, and other optional complex features as selected by BTI. BellSouth shall provide 8XX TFD in accordance with the following:

## 7.1.2 Technical Requirements

- 7.1.2.1 BellSouth shall provide BTI with access to the 8XX record information located in the 8XX SCP. The 8XX SCP contains current records as received from the national SMS and will provide for routing 8XX originating calls based on the dialed ten digit 8XX number.
- 7.1.2.2 The 8XX SCP is designated to receive and respond to queries using the American National Standard Specification of Signaling System Seven (SS7) protocol. The 8XX SCP shall determine the carrier identification based on all ten digits of the dialed number and route calls to the carrier, POTS number, dialing number and/or other optional feature selected by BTI.
- 7.1.2.3 The SCP shall also provide, at BTI's option, such additional feature as described in SR-TSV-002275 (BOC Notes on BellSouth Networks, SR-TSV-002275, Issue 2, (Telcordia (formerly BellCore), April 1994)) as are available to BellSouth. These may include but are not limited to:
- 7.1.2.3.1 Network Management;
- 7.1.2.3.2 Customer Sample Collection; and
- 7.1.2.3.3 Service Maintenance.

#### 7.2 Rates

The prices that BTI shall pay to BellSouth for Network Elements and Other Services are set forth in Exhibit D to this Attachment. If BTI purchases a service(s) from a tariff, all terms and conditions and rates as set forth in such tariff shall apply.

## 7.3 Operational Support Systems (OSS)

The terms, conditions and rates for OSS are as set forth in Section 2.13 of this Attachment.

## 8 Line Information Database (LIDB)

- 8.1 All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of LIDB.
- 8.2 BellSouth will store in its LIDB only records relating to service in the BellSouth region. The LIDB Storage Agreement is included in this Attachment.

## 8.2.1 Definition

8.2.2 The Line Information Database (LIDB) is a transaction-oriented database accessible through Common Channel Signaling (CCS) networks. It contains records associated with end user Line Numbers and Special Billing Numbers. LIDB accepts queries from other Network Elements and provides appropriate responses. The query originator need not be the owner of LIDB data. LIDB queries include functions such as screening billed numbers that provides the ability to accept Collect or Third Number Billing calls and validation of Telephone Line Number based non-proprietary calling cards. The interface for the LIDB functionality is the interface between BellSouth's CCS network and other CCS networks. LIDB also interfaces to administrative systems.

# 8.2.3 <u>Technical Requirements</u>

- 8.2.4 BellSouth will offer to BTI any additional capabilities that are developed for LIDB during the life of this Agreement.
- 8.2.4.1 BellSouth shall process BTI's Customer records in LIDB at least at parity with BellSouth customer records, with respect to other LIDB functions. BellSouth shall indicate to BTI what additional functions (if any) are performed by LIDB in the BellSouth network.
- 8.2.4.2 Within two (2) weeks after a request by BTI, BellSouth shall provide BTI with a list of the customer data items, which BTI would have to provide in order to support each required LIDB function. The list shall indicate which data items are essential to LIDB function, and which are required only to support certain services. For each data item, the list shall show the data formats, the acceptable values of the data item and the meaning of those values.
- 8.2.4.3 BellSouth shall provide LIDB systems for which operating deficiencies that would result in calls being blocked shall not exceed 30 minutes per year.
- 8.2.4.4 BellSouth shall provide LIDB systems for which operating deficiencies that would not result in calls being blocked shall not exceed 12 hours per year.
- 8.2.4.5 BellSouth shall provide LIDB systems for which the LIDB function shall be in overload no more than 12 hours per year.

Version 3Q00: 10/10/00

- 8.2.4.6 All additions, updates and deletions of BTI data to the LIDB shall be solely at the direction of BTI. Such direction from BTI will not be required where the addition, update or deletion is necessary to perform standard fraud control measures (e.g., calling card auto-deactivation).
- 8.2.4.7 BellSouth shall provide priority updates to LIDB for BTI data upon BTI's request (e.g., to support fraud detection), via password-protected telephone card, facsimile, or electronic mail within one hour of notice from the established BellSouth contact.
- 8.2.4.8 BellSouth shall provide LIDB systems such that no more than 0.01% of BTI customer records will be missing from LIDB, as measured by BTI audits. BellSouth will audit BTI records in LIDB against DBAS to identify record mismatches and provide this data to a designated BTI contact person to resolve the status of the records and BellSouth will update system appropriately. BellSouth will refer record of mis-matches to BTI within one business day of audit. Once reconciled records are received back from BTI, BellSouth will update LIDB the same business day if less than 500 records are received before 1:00PM Central Time. If more than 500 records are received, BellSouth will contact BTI to negotiate a time frame for the updates, not to exceed three business days.
- 8.2.4.9 BellSouth shall perform backup and recovery of all of BTI's data in LIDB including sending to LIDB all changes made since the date of the most recent backup copy, in at least the same time frame BellSouth performs backup and recovery of BellSouth data in LIDB for itself. Currently, BellSouth performs backups of the LIDB for itself on a weekly basis and when a new software release is scheduled, a backup is performed prior to loading the new release.
- 8.2.4.10 BellSouth shall provide BTI with LIDB reports of data, which are missing or contain errors, as well as any misrouted errors, within a reasonable time period as negotiated between BTI and BellSouth.
- 8.2.4.11 BellSouth shall prevent any access to or use of BTI data in LIDB by BellSouth personnel that are outside of established administrative and fraud control personnel, or by any other Party that is not authorized by BTI in writing.
- 8.2.4.12 BellSouth shall provide BTI performance of the LIDB Data Screening function, which allows a LIDB to completely or partially deny specific query originators access to LIDB data owned by specific data owners, for Customer Data that is part of an NPA-NXX or RAO-0/1XX wholly or partially owned by BTI at least at parity with BellSouth Customer Data. BellSouth shall obtain from BTI the screening information associated with LIDB Data Screening of BTI data in accordance with this requirement. BellSouth currently does not have LIDB Data Screening capabilities. When such capability is available, BellSouth shall offer it to BTI under the BFR/NBR as set forth in General Terms and Conditions.

- 8.2.4.13 BellSouth shall accept queries to LIDB associated with BTI customer records, and shall return responses in accordance with industry standards.
- 8.2.4.14 BellSouth shall provide mean processing time at the LIDB within 0.50 seconds under normal conditions as defined in industry standards.
- 8.2.4.15 BellSouth shall provide processing time at the LIDB within 1 second for 99% of all messages under normal conditions as defined in industry standards.
- 8.2.5 <u>Interface Requirements</u>
- 8.2.6 BellSouth shall offer LIDB in accordance with the requirements of this subsection.
- 8.2.6.1 The interface to LIDB shall be in accordance with the technical references contained within.
- 8.2.6.2 The CCS interface to LIDB shall be the standard interface described herein.
- 8.2.6.3 The LIDB Data Base interpretation of the ANSI-TCAP messages shall comply with the technical reference herein. Global Title Translation shall be maintained in the signaling network in order to support signaling network routing to the LIDB.
- 8.3 Rates

The prices that BTI shall pay to BellSouth for Network Elements and Other Services are set forth in Exhibit D to this Attachment. If BTI purchases a service(s) from a tariff, all terms and conditions and rates as set forth in such tariff shall apply.

## 8.4 **Operational Support Systems (OSS)**

The terms, conditions and rates for OSS are as set forth in Section 2.13 of this Attachment.

## 9. Signaling

- 9.1 All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of Signaling Transport Services.
- 9.2 BellSouth shall offer access to signaling and access to BellSouth's signaling systems and databases subject to compatibility testing and at the rates set forth in this Attachment. BellSouth may provide mediated access to BellSouth signaling systems and databases. Available signaling elements include signaling links, signal transfer points and service control points. Signaling functionality will be available with both A-link and B-link connectivity.

## 9.3 **Signaling Link Transport**

Version 3Q00: 10/10/00

9.3.1	Definition Signaling Link Transport is a set of two or four dedicated 56 Kbps. transmission paths between CLEC-designated Signaling Points of Interconnection (SPOI) that provides appropriate physical diversity.
9.3.2	Technical Requirements
9.3.2.1	Signaling Link Transport shall consist of full duplex mode 56 kbps transmission paths.
9.3.3	Of the various options available, Signaling Link Transport shall perform in the following two ways:
9.3.3.1	As an "A-link" which is a connection between a switch or SCP and a home Signaling Transfer Point Switch (STP) pair; and
9.3.3.2	As a "B-link" which is a connection between two STP pairs in different company networks (e.g., between two STP pairs for two Competitive Local Exchange Carriers (CLECs)).
9.3.4	Signaling Link Transport shall consist of two or more signaling link layers as follows:
9.3.4.1	An A-link layer shall consist of two links.
9.3.4.2	A B-link layer shall consist of four links.
9.3.5	A signaling link layer shall satisfy a performance objective such that:
9.3.5.1	There shall be no more than two minutes down time per year for an A-link layer; and
9.3.5.2	There shall be negligible (less than 2 seconds) down time per year for a B-link layer.
9.3.5.3	A signaling link layer shall satisfy interoffice and intraoffice diversity of facilities and equipment, such that:
9.3.5.3.1	No single failure of facilities or equipment causes the failure of both links in an A-link layer (i.e., the links should be provided on a minimum of two separate physical paths end-to-end); and
9.3.5.3.2	No two concurrent failures of facilities or equipment shall cause the failure of all four links in a B-link layer (i.e., the links should be provided on a minimum of three separate physical paths end-to-end).
9.3.5.4	Interface Requirements

- 9.3.5.4.1 There shall be a DS1 (1.544 Mbps) interface at the BTI designated SPOIs. Each 56 kbps transmission path shall appear as a DS0 channel within the DS1 interface.
- 9.4 **Signaling Transfer Points (STPs)**
- 9.4.1 <u>Definition</u> Signaling Transfer Points is a signaling network function that includes all of the capabilities provided by the signaling transfer point switches (STPs) and their associated signaling links which enable the exchange of SS7 messages among and between switching elements, database elements and signaling transfer point switches.
- 9.4.2 <u>Technical Requirements</u>
- 9.4.2.1 STPs shall provide access to Network Elements connected to BellSouth SS7 network. These include:
- 9.4.2.1.1 BellSouth Local Switching or Tandem Switching;
- 9.4.2.1.2 BellSouth Service Control Points/DataBases:
- 9.4.2.1.3 Third-party local or tandem switching;
- 9.4.2.1.4 Third-party-provided STPs.
- 9.4.2.2 The connectivity provided by STPs shall fully support the functions of all other Network Elements connected to the BellSouth SS7 network. This explicitly includes the use of the BellSouth SS7 network to convey messages which neither originate nor terminate at a signaling end point directly connected to the BellSouth SS7 network (i.e., transient messages). When the BellSouth SS7 network is used to convey transient messages, there shall be no alteration of the Integrated Services Digital Network User Part (ISDNUP) or Transaction Capabilities Application Part (TCAP) user data that constitutes the content of the message.
- 9.4.2.3 If a BellSouth tandem switch routes calling traffic, based on dialed or translated digits, on SS7 trunks between a BTI local switch and third party local switch, the BellSouth SS7 network shall convey the TCAP messages that are necessary to provide Call Management features (Automatic Callback, Automatic Recall, and Screening List Editing) between BTI local STPs and the STPs that provide connectivity with the third party local switch, even if the third party local switch is not directly connected to BellSouth STPs.
- 9.4.2.4 STPs shall provide all functions of the MTP as defined in the applicable industry standard technical references.
- 9.4.2.5 STPs shall provide all functions of the SCCP necessary for Class 0 (basic connectionless) service, as defined in Telcordia (formerly BellCore) ANSI

Interconnection Requirements. In particular, this includes Global Title Translation (GTT) and SCCP Management procedures, as specified in T1.112.4. In cases where the destination signaling point is a BTI or third party local or tandem switching system directly connected to BellSouth SS7 network, BellSouth shall perform final GTT of messages to the destination and SCCP Subsystem Management of the destination. In all other cases, BellSouth shall perform intermediate GTT of messages to a gateway pair of STPs in an SS7 network connected with BellSouth SS7 network, and shall not perform SCCP Subsystem Management of the destination. If BellSouth performs final GTT to a BTI database, then BTI agrees to provide BellSouth with the Destination Point Code for the BTI database.

- 9.4.2.6 STPs shall provide on a non-discriminatory basis all functions of the OMAP commonly provided by STPs, as specified in the reference in Section 12 of this Attachment. All OMAP functions will be on a "where available" basis and can include:
- 9.4.2.6.1 MTP Routing Verification Test (MRVT); and
- 9.4.2.6.2 SCCP Routing Verification Test (SRVT).
- 9.4.2.7 In cases where the destination signaling point is a BellSouth local or tandem switching system or database, or is a BTI or third party local or tandem switching system directly connected to the BellSouth SS7 network, STPs shall perform MRVT and SRVT to the destination signaling point. In all other cases, STPs shall perform MRVT and SRVT to a gateway pair of STPs in an SS7 network connected with the BellSouth SS7 network. This requirement shall be superseded by the specifications for Internetwork MRVT and SRVT if and when these become approved ANSI standards and available capabilities of BellSouth STPs, and if mutually agreed upon by BTI and BellSouth.
- 9.4.2.8 STPs shall be on parity with BellSouth.
- 9.4.2.9 SS7 Advanced Intelligent Network (AIN) Access
- 9.4.2.9.1 When technically feasible and upon request by BTI, SS7 Access shall be made available in association with switching. SS7 AIN Access is the provisioning of AIN 0.1 triggers in an equipped BellSouth local switch and interconnection of the BellSouth SS7 network with the BTI SS7 network to exchange TCAP queries and responses with a BTI SCP.
- 9.4.2.9.2 SS7 AIN Access shall provide BTI SCP access to BellSouth local switch in association with switching via interconnection of BellSouth SS7 and BTI SS7 Networks. BellSouth shall offer SS7 access through its STPs. If BellSouth requires a mediation device on any part of its network specific to this form of access, BellSouth must route its messages in the same manner. The

interconnection arrangement shall result in the BellSouth local switch recognizing the BTI SCP as at least at parity with BellSouth's SCP's in terms of interfaces, performance and capabilities.

- 9.4.3 Interface Requirements
- 9.4.3.1 BellSouth shall provide the following STPs options to connect BTI or BTI-designated local switching systems or STPs to the BellSouth SS7 network:
- 9.4.3.1.1 An A-link interface from BTI local switching systems; and,
- 9.4.3.1.2 A B-link interface from BTI local STPs.
- 9.4.3.2 Each type of interface shall be provided by one or more sets (layers) of signaling links.
- 9.4.3.3 The Signaling Point of Interconnection (SPOI) for each link shall be located at a cross-connect element, such as a DSX-1, in the Central Office (CO) where BellSouth STP is located. There shall be a DS1 or higher rate transport interface at each of the SPOIs. Each signaling link shall appear as a DS0 channel within the DS1 or higher rate interface. BellSouth shall offer higher rate DS1 signaling for interconnecting BTI local switching systems or STPs with BellSouth STPs as soon as these become approved ANSI standards and available capabilities of BellSouth STPs. BellSouth and BTI will work jointly to establish mutually acceptable SPOIs.
- 9.4.3.4 BellSouth CO shall provide intraoffice diversity between the SPOIs and BellSouth STPs, so that no single failure of intraoffice facilities or equipment shall cause the failure of both B-links in a layer connecting to a BellSouth STP. BellSouth and BTI will work jointly to establish mutually acceptable SPOIs.
- 9.4.3.5 STPs shall provide all functions of the MTP as defined in the applicable industry standard technical references.
- 9.4.3.6 Message Screening
- 9.4.3.6.1 BellSouth shall set message screening parameters so as to accept valid messages from BTI local or tandem switching systems destined to any signaling point within BellSouth's SS7 network where the BTI switching system has a legitimate signaling relation.
- 9.4.3.6.2 BellSouth shall set message screening parameters so as to pass valid messages from BTI local or tandem switching systems destined to any signaling point or network accessed through BellSouth's SS7 network where the BTI switching system has a legitimate signaling relation.

- 9.4.3.6.3 BellSouth shall set message screening parameters so as to accept and pass/send valid messages destined to and from BTI from any signaling point or network interconnected through BellSouth's SS7 network where the BTI SCP has a legitimate signaling relation.
- 9.4.4 STPs shall be equal to or better than all of the requirements for STPs set forth in the applicable industry standard technical references.

#### 9.5 Service Control Points/Databases

# 9.5.1 <u>Definition</u>

- 9.5.1.1 Databases are the Network Elements that provide the functionality for storage of, access to, and manipulation of information required to offer a particular service and/or capability. Databases include, but are not limited to: Local Number Portability, LIDB, Toll Free Number Database, Automatic Location Identification/Data Management System, Calling Name Database, access to Service Creation Environment and Service Management System (SCE/SMS) application databases and Directory Assistance.
- 9.5.2 A Service Control Point (SCP) is a specific type of Database functionality deployed in a Signaling System 7 (SS7) network that executes service application logic in response to SS7 queries sent to it by a switching system also connected to the SS7 network. Service Management Systems provide operational interfaces to allow for provisioning, administration and maintenance of subscriber data and service application data stored in SCPs.

## 9.5.3 Technical Requirements for SCPs/Databases

- 9.5.3.1 Requirements for SCPs/Databases within this section address storage of information, access to information (e.g. signaling protocols, response times), and administration of information (e.g., provisioning, administration, and maintenance). All SCPs/Databases shall be provided to BTI in accordance with the following requirements.
- 9.5.3.2 BellSouth shall provide physical access to SCPs through the SS7 network and protocols with TCAP as the application layer protocol.
- 9.5.3.3 BellSouth shall provide physical interconnection to databases via industry standard interfaces and protocols (e.g. SS7, ISDN and X.25).
- 9.5.3.4 The reliability of interconnection options shall be consistent with requirements for diversity and survivability.

# 9.5.4 <u>Database Availability</u>

- 9.5.4.1 Call processing databases shall have a maximum unscheduled unavailability of 30 minutes per year. Unavailability due to software and hardware upgrades shall be scheduled during minimal usage periods and only be undertaken upon proper notification to providers, which might be impacted. Any downtime associated with the provision of call processing related databases will impact all service providers, including BellSouth, equally.
- 9.5.4.2 The operational interface provided by BellSouth shall complete Database transactions (i.e., add, modify, delete) for BTI customer records stored in BellSouth databases within 3 days, or sooner where BellSouth provisions its own customer records within a shorter interval.

# 9.6 **Local Number Portability Database**

# 9.6.1 <u>Definition</u>

9.6.2 The Permanent Number Portability (PNP) database supplies routing numbers for calls involving numbers that have been ported from one local service provider to another. PNP is currently being worked in industry forums. The results of these forums will dictate the industry direction of PNP. BellSouth agrees to provide access to the PNP database at rates, terms and conditions as set forth by BellSouth and in accordance with an effective FCC or Commission directive.

#### 9.7 **SS7 Network Interconnection**

- 9.7.1 Definition.
- 9.7.2 SS7 Network Interconnection is the interconnection of BTI local Signaling Transfer Point Switches (STP) and BTI local or tandem switching systems with BellSouth STPs. This interconnection provides connectivity that enables the exchange of SS7 messages among BellSouth switching systems and databases (DBs), BTI local or tandem switching systems, and other third-party switching systems directly connected to the BellSouth SS7 network.
- 9.7.3 Technical Requirements
- 9.7.3.1 SS7 Network Interconnection shall provide connectivity to all components of the BellSouth SS7 network. These include:
- 9.7.3.1.1 BellSouth local or tandem switching systems;
- 9.7.3.1.2 BellSouth DBs; and
- 9.7.3.1.3 Other third-party local or tandem switching systems.

- 9.7.4 The connectivity provided by SS7 Network Interconnection shall fully support the functions of BellSouth switching systems and DBs and BTI or other third-party switching systems with A-link access to the BellSouth SS7 network.
- 9.7.5 If traffic is routed based on dialed or translated digits between a BTI local switching system and a BellSouth or other third-party local switching system, either directly or via a BellSouth tandem switching system, then it is a requirement that the BellSouth SS7 network convey via SS7 Network Interconnection the TCAP messages that are necessary to provide Call Management services (Automatic Callback, Automatic Recall, and Screening List Editing) between the BTI local STPs and BellSouth or other third-party local switch.
- 9.7.6 When the capability to route messages based on Intermediate Signaling Network Identifier (ISNI) is generally available on BellSouth STPs, the BellSouth SS7 Network shall also convey TCAP messages using SS7 Network Interconnection in similar circumstances where the BellSouth switch routes traffic based on a Carrier Identification Code (CIC).
- 9.7.7 SS7 Network Interconnection shall provide all functions of the MTP as specified in ANSI T1.111. This includes:
- 9.7.7.1 Signaling Data Link functions, as specified in ANSI T1.111.2;
- 9.7.7.2 Signaling Link functions, as specified in ANSI T1.111.3; and
- 9.7.7.3 Signaling Network Management functions, as specified in ANSI T1.111.4.
- 9.7.8 SS7 Network Interconnection shall provide all functions of the SCCP necessary for Class 0 (basic connectionless) service, as specified in ANSI T1.112. In particular, this includes Global Title Translation (GTT) and SCCP Management procedures, as specified in T1.112.4. Where the destination signaling point is a BellSouth switching system or DB, or is another third-party local or tandem switching system directly connected to the BellSouth SS7 network, SS7 Network Interconnection shall include final GTT of messages to the destination and SCCP Subsystem Management of the destination. Where the destination signaling point is a BTI local or tandem switching system, SS7 Network Interconnection shall include intermediate GTT of messages to a gateway pair of BTI local STPs, and shall not include SCCP Subsystem Management of the destination.
- 9.7.9 SS7 Network Interconnection shall provide all functions of the Integrated Services Digital Network User Part (ISDNUP), as specified in ANSI T1.113.
- 9.7.10 SS7 Network Interconnection shall provide all functions of the TCAP, as specified in ANSI T1.114.

9.7.11 If and when Internetwork MTP Routing Verification Test (MRVT) and SCCP Routing Verification Test (SRVT) become approved ANSI standards and available capabilities of BellSouth STPs, SS7 Network Interconnection shall provide these functions of the OMAP. 9.7.12 SS7 Network Interconnection shall be equal to or better than the following performance requirements: 9.7.12.1 MTP Performance, as specified in ANSI T1.111.6; 9.7.12.2 SCCP Performance, as specified in ANSI T1.112.5; and 9.7.12.3 ISDNUP Performance, as specified in ANSI T1.113.5. 9.7.13 **Interface Requirements** 9.7.13.1 BellSouth shall offer the following SS7 Network Interconnection options to connect BTI or BTI-designated local or tandem switching systems or STPs to the BellSouth SS7 network: 9.7.13.1.1 A-link interface from BTI local or tandem switching systems; and 9.7.13.1.2 B-link interface from BTI STPs. 9.7.13.2 The Signaling Point of Interconnection (SPOI) for each link shall be located at a cross-connect element, such as a DSX-1, in the Central Office (CO) where the BellSouth STP is located. There shall be a DS1 or higher rate transport interface at each of the SPOIs. Each signaling link shall appear as a DS0 channel within the DS1 or higher rate interface. BellSouth shall offer higher rate DS1 signaling links for interconnecting BTI local switching systems or STPs with BellSouth STPs as soon as these become approved ANSI standards and available capabilities of BellSouth STPs. BellSouth and BTI will work jointly to establish mutually acceptable SPOI. 9.7.13.3 BellSouth CO shall provide intraoffice diversity between the SPOIs and the BellSouth STP, so that no single failure of intraoffice facilities or equipment shall cause the failure of both B-links in a layer connecting to a BellSouth STP. BellSouth and BTI will work jointly to establish mutually acceptable SPOI. 9.7.13.4 The protocol interface requirements for SS7 Network Interconnection include the MTP, ISDNUP, SCCP, and TCAP. These protocol interfaces shall conform to the applicable industry standard technical references. 9.7.13.5 BellSouth shall set message screening parameters to accept messages from BTI local or tandem switching systems destined to any signaling point in the BellSouth

relation.

SS7 network with which the BTI switching system has a legitimate signaling

9.7.13.6 SS7 Network Interconnection shall be equal to or better than all of the requirements for SS7 Network Interconnection set forth in the applicable industry standard technical references.

#### 9.8 Rates

The prices that BTI shall pay to BellSouth for Network Elements and Other Services are set forth in Exhibit D to this Attachment. If BTI purchases a service(s) from a tariff, all terms and conditions and rates as set forth in such tariff shall apply.

# 9.9 Operational Support Systems (OSS)

The terms, conditions and rates for OSS are as set forth in Section 2.13 of this Attachment.

# 10. Operator Call Processing, Inward Operator Services and Directory Assistance Services

10.1 All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of Operator Call Processing, Inward Operator Services and Directory Assistance Services.

## 10.2 **Operator Systems**

10.2.1 <u>Definition.</u> Operator Systems is the Network Element that provides operator and automated call handling and billing, special services, end user telephone listings and optional call completion services. The Operator Systems, Network Element provides two types of functions: Operator Service functions and Directory Assistance Service functions, each of which are described in detail below.

## 10.3 **Operator Service**

10.3.1 <u>Definition</u>. Operator Service provides: (1) operator handling for call completion (for example, collect, third number billing, and manual credit card calls), (2) operator or automated assistance for billing after the end user has dialed the called number (for example, credit card calls); and (3) special services including but not limited to Busy Line Verification and Emergency Line Interrupt (ELI), Emergency Agency Call, Operator-assisted Directory Assistance, and Rate Quotes.

#### 10.3.2 Requirements

- 10.3.2.1 When BTI requests BellSouth to provide Operator Services, the following requirements apply:
- 10.3.2.1.1 BellSouth shall complete 0+ and 0- dialed local calls.

10.3.2.1.2	BellSouth shall complete 0+ intraLATA toll calls.
10.3.2.1.3	BellSouth shall process calls that are billed to BTI end user's calling card that can be validated by BellSouth.
10.3.2.1.4	BellSouth shall complete person-to-person calls.
10.3.2.1.5	BellSouth shall complete collect calls.
10.3.2.1.6	BellSouth shall provide the capability for callers to bill to a third party and complete such calls.
10.3.2.1.7	BellSouth shall complete station-to-station calls.
10.3.2.1.8	BellSouth shall process emergency calls.
10.3.2.1.9	BellSouth shall process Busy Line Verify and Emergency Line Interrupt requests.
10.3.2.1.10	BellSouth shall process emergency call trace, as it does for its own end users prior to the Effective Date. Call must originate from a 911 provider.
10.3.2.1.11	BellSouth shall process operator-assisted directory assistance calls.
10.3.2.1.12	BellSouth shall adhere to equal access requirements, providing BTI local end users the same IXC access as provided to BellSouth end users.
10.3.2.1.13	BellSouth shall exercise at least the same level of fraud control in providing Operator Service to BTI that BellSouth provides for its own operator service.
10.3.2.1.14	BellSouth shall perform Billed Number Screening when handling Collect, Personto-Person, and Billed-to-Third-Party calls.
10.3.2.1.15	BellSouth shall direct customer account and other similar inquiries to the customer service center designated by BTI.
10.3.2.1.16	BellSouth shall provide a feed of customer call records in "EMI" format to BTI in accordance with CLEC ODUF standards specified in Attachment 7.
10.3.3	Interface Requirements
10.3.3.1	With respect to Operator Services for calls that originate on local switching capability provided by or on behalf of BTI, the interface requirements shall conform to the then current established system interface specifications for the platform used to provide Operator Service and the interface shall conform to industry standards.
10.4	Directory Assistance Service

- 10.4.1 <u>Definition.</u> Directory Assistance Service provides local end user telephone number listings with the option to complete the call at the caller's direction separate and distinct from local switching.
- 10.4.2 <u>Requirements</u>
- Directory Assistance Service shall provide up to two listing requests per call. If available and if requested by BTI's end user, BellSouth shall provide caller-optional directory assistance call completion service at rates contained in this Attachment to one of the provided listings, equal to that which BellSouth provides its end users. If not available, BTI may request such requirement pursuant to the BFR/NBR Process as set forth in General Terms and Conditions.
- 10.4.4 <u>Directory Assistance Service Updates</u>
- 10.4.4.1 BellSouth shall update end user listings changes daily. These changes include:
- 10.4.4.1.1 New end user connections: BellSouth will provide service to BTI that is equal to the service it provides to itself and its end users;
- 10.4.4.1.2 End user disconnections: BellSouth will provide service to BTI that is equal to the service it provides to itself and its end users; and
- 10.4.4.1.3 End user address changes: BellSouth will provide service to BTI that is equal to the service it provides to itself and its end users;
- 10.4.4.1.4 These updates shall also be provided for non-listed and non-published numbers for use in emergencies.
- 10.4.5 Branding for Operator Call Processing and Directory Assistance
- 10.4.5.1 The BellSouth Operator Systems Branding Feature provides a definable announcement to BTI end users using Directory Assistance (DA)/Operator Call Processing (OCP) prior to placing them in queue or connecting them to an available operator or automated operator system. This feature allows BTI to have its calls custom branded with BTI's name on whose behalf BellSouth is providing Directory Assistance and/or Operator Call Processing. Rates for Custom Branding, Operator Call Process and Directory Assistance are set forth in this Attachment.
- 10.4.5.2 BellSouth offers four service levels of branding to BTI when ordering Directory Assistance and/or Operator Call Processing.
- 10.4.5.2.1 Service Level 1 BellSouth Branding
- 10.4.5.2.2 Service Level 2 Unbranded

10.4.5.2.3	Service Level 3 - Custom Branding
10.4.5.2.4	Service Level 4 - Self Branding (applicable only to BTI for Resale or use with an Unbundled Port when routing to an operator service provider other than BellSouth).
10.4.6	For Resellers and Use with an Unbundled Port
10.4.6.1	BellSouth Branding is the Default Service Level.
10.4.6.2	Unbranding, Custom Branding, and Self Branding require BTI to order selective routing for each originating BellSouth end office identified by BTI. Rates for Selective Routing are set forth in this Attachment.
10.4.6.3	Custom Branding and Self Branding require BTI to order dedicated trunking from each BellSouth end office identified by BTI, to either the BellSouth Traffic Operator Position System (TOPS) or BTI Operator Service Provider. Rates for trunks are set forth in applicable BellSouth tariffs.
10.4.6.4	Unbranding - Unbranded Directory Assistance and/or Operator Call Processing calls ride common trunk groups provisioned by BellSouth from those end offices identified by BTI to the BellSouth TOPS. These calls are routed to "No Announcement."
10.4.7	For Facilities Based Carriers
10.4.7.1	All Service Levels require BTI to order dedicated trunking from their end office(s) point of interface to the BellSouth TOPS Switches. Rates for trunks are set forth in applicable BellSouth tariffs.
10.4.7.2	Customized Branding includes charges for the recording of the branding announcement and the loading of the audio units in each TOPS Switch, Interactive Voice Subsystem (IVS) and Network Applications Vehicle (NAV) equipment for which BTI requires service.
10.4.8	Directory Assistance customized branding uses:
10.4.8.1	the recording of the name;
10.4.8.2	the front-end loading of the Digital Recorded Announcement Machine (DRAM) in each TOPS switch.
10.4.9	Operator Call Processing customized branding uses:
10.4.9.1	the recording of the name;
10.4.9.2	the front-end loading of the DRAM in the TOPS Switch;

- the back-end loading in the audio units in the Automated Alternate Billing System (AABS) in the Interactive Voice Subsystem (IVS);
- the 0- automation loading for the audio units in the Enhanced Billing and Access Service (EBAS) in the Network Applications Vehicle (NAV).
- BellSouth will provide to BTI purchasing local BellSouth switching and reselling BellSouth local exchange service, selective routing of calls to a requested directory assistance services platform or operator services platform. BTI end users may use the same dialing arrangements as BellSouth end users, but obtain a BTI branded service.

## 10.5 Directory Assistance Database Service (DADS)

- 10.5.1 BellSouth shall make its Directory Assistance Database Service (DADS) available solely for the expressed purpose of providing Directory Assistance type services to BTI end users. The term "end user" denotes any entity which obtains Directory Assistance type services for its own use from a DADS customer. Directory Assistance type service is defined as Voice Directory Assistance (DA Operator assisted and Electronic Directory Assistance (Data System assisted)). BTI agrees that DADS will not be used for any purpose which violates federal or state laws, statutes, regulatory orders or tariffs. Except for the permitted users, BTI agrees not to disclose DADS to others and shall provide due care in providing for the security and confidentiality of DADS. Further, BTI authorizes the inclusion of BTI Directory Assistance listings in the BellSouth Directory Assistance products.
- BellSouth shall provide BTI initially with a base file of subscriber listings which reflect all listing change activity occurring since BTI's most recent update via magnetic tape, and subsequently using electronic connectivity such as Network Data Mover to be developed mutually by BTI and BellSouth. BTI agrees to assume the costs associated with CONNECT: Direct TM connectivity, which will vary depending upon volume and mileage.
- 10.5.3 BellSouth will require approximately one month after receiving an order to prepare the Base File. BellSouth will provide daily updates which will reflect all listing change activity occurring since CLEC's most recent update. BellSouth shall provide updates to BTI on a Business, Residence, or combined Business and Residence basis. BTI agrees that the updates shall be used solely to keep the information current. Delivery of Daily Updates will commence the day after BTI receives the Base File.
- BellSouth is authorized to include BTI Directory Assistance Listing Information in its DADS. Any other use by BellSouth of BTI Directory Assistance Listing Information is not authorized and with the exception of a request for DADS, BellSouth shall refer any request for such information to BTI.

- 10.5.5 Rates for DADS are as set forth in this Attachment.
- 10.6 **Direct Access to Directory Assistance Service**
- 10.6.1 Direct Access to Directory Assistance Service (DADAS) will provide BTI's directory assistance operators with the ability to search all available BellSouth subscriber listings using the Directory Assistance search format. Subscription to DADAS will allow BTI to utilize its own switch, operator workstations and optional audio subsystems.
- 10.6.2 Rates, terms and conditions for provisioning DADAS are as set forth in the FCC tariff No. 1.
- 10.7 Automatic Location Identification/Data Management System (ALI/DMS)
- The ALI/DMS Database contains end user information (including name, address, telephone information, and sometimes special information from the local service provider or end user) used to determine to which Public Safety Answering Point (PSAP) to route the call. The ALI/DMS database is used to provide more routing flexibility for E911 calls than Basic 911. BellSouth shall provide the Emergency Services Database in accordance with the following:
- 10.7.2 <u>Technical Requirements</u>
- 10.7.2.1 BellSouth shall offer BTI a data link to the ALI/DMS database or permit BTI to provide its own data link to the ALI/DMS database. BellSouth shall provide error reports from the ALI/DMS database to BTI immediately after BTI inputs information into the ALI/DMS database. Alternately, BTI may utilize BellSouth, to enter end user information into the database on a demand basis, and validate end user information on a demand basis.
- 10.7.2.2 The ALI/DMS database shall contain the following end user information:
- 10.7.2.2.1 Name;
- 10.7.2.2.2 Address:
- 10.7.2.2.3 Telephone number; and
- 10.7.2.2.4 Other information as appropriate (e.g., whether an end user is blind or deaf or has another disability).
- When BellSouth is responsible for administering the ALI/DMS database in its entirety, ported number NXXs entries for the ported numbers should be maintained unless BTI requests otherwise and shall be updated if BTI requests, provided BTI supplies BellSouth with the updates.

Page 69

- 10.7.2.4 When Remote Call Forwarding (RCF) is used to provide number portability to the local end user and a remark or other appropriate field information is available in the database, the shadow or "forwarded-to" number and an indication that the number is ported shall be added to the customer record.
- 10.7.2.5 If BellSouth is responsible for configuring PSAP features (for cases when the PSAP or BellSouth supports an ISDN interface) it shall ensure that CLASS Automatic Recall (Call Return) is not used to call back to the ported number. Although BellSouth currently does not have ISDN interface, BellSouth agrees to comply with this requirement once ISDN interfaces are in place.
- 10.7.3 **Interface Requirements**

The interface between the E911 Switch or Tandem and the ALI/DMS database for BTI end users shall meet industry standards.

10.8 **Rates** 

> The prices that BTI shall pay to BellSouth for Network Elements and Other Services are set forth in Exhibit D to this Attachment. If BTI purchases a service(s) from a tariff, all terms and conditions and rates as set forth in such tariff shall apply.

10.9 **Operational Support Systems (OSS)** 

> The terms, conditions and rates for OSS are as set forth in Section 2.13 of this Attachment.

#### 11. Calling Name (CNAM) Database Service

- 11.1 All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of CNAM.
- 11.2 The Agreement for CNAM with standard pricing is included as Exhibit B to this Attachment. BTI must provide to its account manager a written request with a requested activation date to activate this service. If BTI is interested in requesting CNAM with volume and term pricing, BTI must contact its account manager to request a separate CNAM volume and term Agreement.
- 11.3 SCPs/Databases shall be equal to or better than all of the requirements for SCPs/Databases set forth in the applicable industry standard technical references.
- 11.4 Service Creation Environment and Service Management System (SCE/SMS) **Advanced Intelligent Network (AIN) Access**
- 11.4.1 BellSouth's Service Creation Environment and Service Management System (SCE/SMS) Advanced Intelligent Network (AIN) Access shall provide BTI the

capability that will allow BTI and other third parties to create service applications in a BellSouth SCE and deploy those applications in a BellSouth SMS to a BellSouth SCP. The third party service applications interact with AIN triggers provisioned on a BellSouth SSP.

- BellSouth's SCE/SMS AIN Access shall provide access to SCE hardware, software, testing and technical support (e.g., help desk, system administrator) resources available to BTI. Scheduling procedures shall provide BTI equivalent priority to these resources.
- BellSouth SCP shall partition and protect BTI service logic and data from unauthorized access, execution or other types of compromise.
- When BTI selects SCE/SMS AIN Access, BellSouth shall provide training, documentation, and technical support to enable BTI to use BellSouth's SCE/SMS AIN Access to create and administer applications. Training, documentation, and technical support will address use of SCE and SMS access and administrative functions, but will not include support for the creation of a specific service application.
- When BTI selects SCE/SMS AIN Access, BellSouth shall provide for a secure, controlled access environment in association with its internal use of AIN components. BTI access will be provided via remote data connection (e.g., dialin, ISDN).
- When BTI selects SCE/SMS AIN Access, BellSouth shall allow BTI to download data forms and/or tables to BellSouth SCP via BellSouth SMS without intervention from BellSouth (e.g., service customization and end user subscription).

### 11.5 **Rates**

The prices that BTI shall pay to BellSouth for Network Elements and Other Services are set forth in Exhibit D to this Attachment. If BTI purchases a service(s) from a tariff, all terms and conditions and rates as set forth in such tariff shall apply.

# 11.6 Operational Support Systems (OSS)

The terms, conditions and rates for OSS are as set forth in Section 2.13 of this Attachment.

## 12. Basic 911 and E911

All of the negotiated terms and conditions set forth in this Section pertain to the provision of Basic 911 and E911.

- If BTI orders network elements and other services, then BTI is also responsible for providing E911 to its end users. BellSouth agrees to offer access to the 911/E911 network pursuant to the following terms and conditions set forth in this Attachment.
- 12.3 Definition
- Basic 911 and E911 is an additional requirement that provides a caller access to the applicable emergency service bureau by dialing a 3-digit universal telephone number (911).
- 12.5 Requirements
- Basic 911 Service Provisioning. For Basic 911 service, BellSouth will provide to BTI a list consisting of each municipality that subscribes to Basic 911 service. The list will also provide, if known, the E911 conversion date for each municipality and, for network routing purposes, a ten-digit directory number representing the appropriate emergency answering position for each municipality subscribing to 911. BTI will be required to arrange to accept 911 calls from its end users in municipalities that subscribe to Basic 911 service and translate the 911 call to the appropriate 10-digit directory number as stated on the list provided by BellSouth. BTI will be required to route that call to BellSouth at the appropriate tandem or end office. When a municipality converts to E911 service, BTI will be required to discontinue the Basic 911 procedures and being using E911 procedures.
- E911 Service Provisioning. For E911 service, BTI will be required to install a 12.5.2 minimum of two dedicated trunks originating from the BTI serving wire center and terminating to the appropriate E911 tandem. The dedicated trunks shall be, at a minimum, DS-0 level trunks configured either as a 2-wire analog interface or as part of a digital (1.544 Mb/s) interface. Either configuration shall use CAMAtype signaling with multifrequency ("MF") pulsing that will deliver automatic number identification ("ANI") with the voice portion of the call. If the user interface is digital, MF pulses, as well as other AC signals, shall be encoded per the u-255 Law convention. BTI will be required to provide BellSouth daily updates to the E911 database. BTI will be required to forward 911 calls to the appropriate E911 tandem, along with ANI, based upon the current E911 end office to tandem homing arrangement as provided by BellSouth. If the E911 tandem trunks are not available, BTI will be required to route the call to a designated 7-digit local number residing in the appropriate Public Service Answering Point ("PSAP"). This call will be transported over BellSouth's interoffice network and will not carry the ANI of the calling party. BTI shall be responsible for providing BellSouth with complete and accurate data for submission to the 911/E911 database for the purpose of providing 911/E911 to its end users.

- 12.5.3 <u>Rates.</u> Charges for 911/E911 service are borne by the municipality purchasing the service. BellSouth will impose no charge on BTI beyond applicable charges for BellSouth trunking arrangements.
- Basic 911 and E911 functions provided to BTI shall be at least at parity with the support and services that BellSouth provides to its end users for such similar functionality.
- 12.5.5 Detailed Practices and Procedures. The detailed practices and procedures contained in the E911 Local Exchange Carrier Guide For Facility-Based Providers as amended from time to time during the term of this Agreement will determine the appropriate practices and procedures for BellSouth and BTI to follow in providing 911/E911 services.

## 13. True-Up

This section applies only to Tennessee and other rates that are interim or expressly subject to true-up under this attachment.

- The interim prices for Network Elements and Other Services and Local Interconnection shall be subject to true-up according to the following procedures:
- The interim prices shall be trued-up, either up or down, based on final prices determined either by further agreement between the Parties, or by a final order (including any appeals) of the Commission which final order meets the criteria of (3) below. The Parties shall implement the true-up by comparing the actual volumes and demand for each item, together with interim prices for each item, with the final prices determined for each item. Each Party shall keep its own records upon which the true-up can be based, and any final payment from one Party to the other shall be in an amount agreed upon by the Parties based on such records. In the event of any disagreement as between the records or the Parties regarding the amount of such true-up, the Parties agree that the body having jurisdiction over the matter shall be called upon to resolve such differences, or the Parties may mutually agree to submit the matter to the Dispute Resolution process in accordance with the provisions of Section 12 of the General Terms and Conditions and Attachment 1 of the Agreement.
- 13.3 The Parties may continue to negotiate toward final prices, but in the event that no such Agreement is reached within nine (9) months, either Party may petition the Commission to resolve such disputes and to determine final prices for each item. Alternatively, upon mutual agreement, the Parties may submit the matter to the Dispute Resolution Process set forth in Section 12 of the General Terms and Conditions and Attachment 1 of the Agreement, so long as they file the resulting Agreement with the Commission as a "negotiated Agreement" under Section 252(e) of the Act.

(a) 13.4An effective order of the Commission that forms the basis of a true-up shall be based upon cost studies submitted by either or both Parties to the Commission and shall be binding upon BellSouth and BTI specifically or upon all carriers generally, such as a generic cost proceeding.

## **EXHIBIT A**

## LINE INFORMATION DATA BASE (LIDB)

### STORAGE AGREEMENT

## I. SCOPE

- A. This Agreement sets forth the terms and conditions pursuant to which BellSouth agrees to store in its LIDB certain information at the request of BTI and pursuant to which BellSouth, its LIDB customers and BTI shall have access to such information. BTI understands that BellSouth provides access to information in its LIDB to various telecommunications service providers pursuant to applicable tariffs and agrees that information stored at the request of BTI, pursuant to this Agreement, shall be available to those telecommunications service providers. The terms and conditions contained herein shall hereby be made a part of this Interconnection Agreement upon notice to BTI's account team to activate this LIDB Storage Agreement. The General Terms and Conditions of the Interconnection/Resale Agreement shall govern this LIDB Storage Agreement. The terms and conditions contained in the attached Addendum is hereby made a part of this LIDB Storage Agreement as if fully incorporated herein.
- B. LIDB is accessed for the following purposes:
  - 1.Billed Number Screening
  - 2. Calling Card Validation
  - 3.Fraud Control
- C. BellSouth will provide seven days per week, 24-hours per day, fraud monitoring on Calling Cards, bill-to-third and collect calls made to numbers in BellSouth's LIDB, provided that such information is included in the LIDB query. BellSouth will establish fraud alert thresholds and will notify BTI of fraud alerts so that BTI may take action it deems appropriate. BTI understands and agrees BellSouth will administer all data stored in the LIDB, including the data provided by BTI pursuant to this Agreement, in the same manner as BellSouth's data for BellSouth's end user customers. BellSouth shall not be responsible to BTI for any lost revenue which may result from BellSouth's administration of the LIDB pursuant to its established practices and procedures as they exist and as they may be changed by BellSouth in its sole discretion from time to time.

BTI understands that BellSouth currently has in effect numerous billing and collection agreements with various interexchange carriers and billing clearinghouses. BTI further understands that these billing and collection customers of BellSouth query BellSouth's LIDB to determine whether to accept various billing options from end users. Additionally, BTI understands that presently BellSouth has no method to differentiate between BellSouth's own billing and line data in the LIDB and such data which it includes in the LIDB on BTI's behalf pursuant to this Agreement. Therefore, until such time as BellSouth can and does implement in its LIDB and its supporting systems the means to differentiate BTI's data from BellSouth's data and the Parties to this Agreement execute appropriate amendments hereto, the following terms and conditions shall apply:

- (a) BTI agrees that it will accept responsibility for telecommunications services billed by BellSouth for its billing and collection customers for BTI's end user accounts which are resident in LIDB pursuant to this Agreement. BTI authorizes BellSouth to place such charges on BTI's bill from BellSouth and agrees that it shall pay all such charges. Charges for which BTI hereby takes responsibility include, but are not limited to, collect and third number calls.
- (b) Charges for such services shall appear on a separate BellSouth bill page identified with the name of the entity for which BellSouth is billing the charge.
- (c) BTI shall have the responsibility to render a billing statement to its end users for these charges, but BTI's obligation to pay BellSouth for the charges billed shall be independent of whether BTI is able or not to collect from BTI's end users.
- (d) BellSouth shall not become involved in any disputes between BTI and the entities for which BellSouth performs billing and collection. BellSouth will not issue adjustments for charges billed on behalf of an entity to BTI. It shall be the responsibility of BTI and the other entity to negotiate and arrange for any appropriate adjustments.

## II. FEES FOR SERVICE AND TAXES

- A. BTI will not be charged a fee for storage services provided by BellSouth to BTI, as described in Section I of this Agreement.
- B. Sales, use and all other taxes (excluding taxes on BellSouth's income) determined by BellSouth or any taxing authority to be due to any federal, state or local taxing jurisdiction with respect to the provision of the service set forth herein will be paid by

BTI in accordance with the tax provisions set forth in the General Terms and Conditions of this Agreement.

## III. MISCELLANEOUS

- A. It is understood and agreed to by the Parties that BellSouth may provide similar services to other companies.
- B. All terms, conditions and operations under this Agreement shall be performed in accordance with, and subject to, all applicable local, state or federal legal and regulatory tariffs, rulings, and other requirements of the federal courts, the U. S. Department of Justice and state and federal regulatory agencies. Nothing in this Agreement shall be construed to cause either Party to violate any such legal or regulatory requirement and either Party's obligation to perform shall be subject to all such requirements.
- C. This LIDB Storage Agreement constitutes the entire Agreement between BTI and BellSouth with respect to the subject matter hereof and supersedes all prior Agreements or contracts, oral or written representations, statements, negotiations, understandings, proposals and undertakings with respect to LIDB Storage.

# FACILITIES BASED ADDENDUM

# TO LINE INFORMATION DATA BASE (LIDB)

# STORAGE AGREEMENT

	This is a Facilities Based Addendum to the Line Information Data Base Storage
Agreen	nent dated, between BellSouth mmunications, Inc. ("BellSouth"), and("BTI"),
effectiv	ve the,
I.	GENERAL
	This Addendum sets forth the terms and conditions for BTI's provision of billing number information to BellSouth for inclusion in BellSouth's LIDB. BellSouth will store in its LIDB the billing number information provided by BTI, and BellSouth will provide responses to on-line, call-by-call queries to this information for purposes specified in Section I.B. of the Agreement.
II.	DEFINITIONS
A.	Billing number - a number that BTI creates for the purpose of identifying an account liable for charges. This number may be a line or a special billing number.B.Line number - a ten digit number that identifies a telephone line administered by BTI.
C.	Special billing number - a ten-digit number that identifies a billing account established by BTI.
D.	Calling Card number - a billing number plus PIN number.
E.	PIN number - a four-digit security code assigned by BTI which is added to a billing number to compose a fourteen-digit calling card number.
F.	Toll billing exception indicator - associated with a billing number to indicate that it is considered invalid for billing of collect calls or third number calls or both, by BTI.
G.	Billed Number Screening - refers to the activity of determining whether a toll billing exception indicator is present for a particular billing number.

- H. Calling Card Validation refers to the activity of determining whether a particular calling card number exists as stated or otherwise provided by a caller.
- I. Billing number information information about billing number, Calling Card number and toll billing exception indicator provided to BellSouth by BTI.

## III. RESPONSIBILITIES OF PARTIES

- A. BTI will provide its billing number information to BellSouth's LIDB each business day by a method that has been mutually agreed upon by both Parties.
- B. BellSouth will store in its LIDB the billing number information provided by BTI. Under normal operating conditions, BellSouth shall include BTI's billing number information in its LIDB no later than two business days following BellSouth's receipt of such billing number information, provided that BellSouth shall not be held responsible for any delay or failure in performance to the extent such delay or failure is caused by circumstances or conditions beyond BellSouth's reasonable control. BellSouth will store in its LIDB an unlimited volume of BTI's working telephone numbers.
- C. BellSouth will provide responses to on-line, call-by-call queries to the stored information for the specific purposes listed in the next paragraph.
- D. BellSouth is authorized to use the billing number information provided by BTI to perform the following functions for authorized users on an on-line basis:
  - 1. Validate a 14 digit Calling Card number where the first 10 digits are a line number or special billing number assigned by BTI, and where the last four digits (PIN) are a security code assigned by BTI.
  - 2. Determine whether BTI or the subscriber has identified the billing number as one which should not be billed for collect or third number calls, or both.
- E. BTI will provide its own billing number information to BellSouth for storage and to be used for Billed Number Screening and Calling Card Validation. BTI will arrange and pay for transport of updates to BellSouth.

## IV. COMPLIANCE

Unless expressly authorized in writing by BTI, all billing number information provided pursuant to this Addendum shall be used for no purposes other than those set forth in this Addendum.

### **EXHIBIT B**

# CALLING NAME DELIVERY (CNAM) DATABASE SERVICES

## 1. Definitions

For the purpose of this Attachment, the following terms shall be defined as:

**CALLING NAME DELIVERY DATABASE SERVICE (CNAM) -** The ability to associate a name with the calling party number, allowing the end user subscriber (to which a call is being terminated) to view the calling party's name before the call is answered. This service also provides BTI the opportunity to load and store its subscriber names in the BellSouth CNAM SCPs.

**CALLING PARTY NUMBER (CPN)** - The number of the calling party that is delivered to the terminating switch using common channel signaling system 7 (CCS7) technology, and that is contained in the Initial Address Message (IAM) portion of the CCS7 call setup.

**COMMON CHANNEL SIGNALING SYSTEM 7 (CCS7)** - A network signaling technology in which all signaling information between two or more nodes is transmitted over high-speed data links, rather than over voice circuits.

**SERVICE CONTROL POINTs (SCPs)** - The real-time data base systems that contain the names to be provided in response to queries received from CNAM SSPs.

**SERVICE MANAGEMENT SYSTEM (SMS)** - The main operations support system of CNAM DATABASE SERVICE. CNAM records are loaded into the SMS, which in turn downloads into the CNAM SCP.

**SERVICE SWITCHING POINTs (SSPs)** - Features of computerized switches in the telephone network that determine that a terminating line has subscribed to CNAM service, and then communicate with CNAM SCPs in order to provide the name associated with the calling party number.

**SUBSYSTEM NUMBER (SSN)** - The address used in the Signaling Connection Control Part (SCCP) layer of the SS7 protocol to designate an application at an end signaling point. A SSN for CNAM at the end office designates the CNAM application within the end office. BellSouth uses the CNAM SSN of 232.

### 2. Attachment

- 2.1 This Attachment contains the terms and conditions where BellSouth will provide to BTI access to the BellSouth CNAM SCP for query or record storage purposes.
- 2.2 BTI shall submit to BellSouth a notice of its intent to access and utilize BellSouth CNAM Database Services pursuant to the terms and conditions of this Attachment. Said notice shall be in writing, no less than 60 days prior to BTI's access to

BellSouth's CNAM Database Services and shall be addressed to BTI's Account Manager.

# 3. Physical Connection and Compensation

- 3.1 BellSouth's provision of CNAM Database Services to BTI requires interconnection from BTI to BellSouth CNAM Service Control Points (SCPs). Such interconnections shall be established pursuant to Attachment 3 of this Agreement. The appropriate charge for access to and use of the BellSouth CNAM Database service shall be as set forth in this Attachment.
- 3.2 In order to formulate a CNAM query to be sent to the BellSouth CNAM SCP, BTI shall provide its own CNAM SSP. BTI's CNAM SSPs must be compliant with TR-NWT-001188, "CLASS Calling Name Delivery Generic Requirements".
- 3.3 If BTI elects to access the BellSouth CNAM SCP via a third party CCS7 transport provider, the third party CCS7 provider shall interconnect with the BellSouth CCS7 network according to BellSouth's Common Channel Signaling Interconnection Guidelines and Telcordia (formerly BellCore)'s CCS Network Interface Specification document, TR-TSV-000905. In addition, the third party provider shall establish CCS7 interconnection at the BellSouth Local Signal Transfer Points (LSTPs) serving the BellSouth CNAM SCPs that BTI desires to query.

# 3.4 Out-Of-Region Customers

If the customer queries the BellSouth CNAM SCP via a third party national SS7 transport provider, the third party SS7 provider shall interconnect with the BellSouth CCS7 network according to BellSouth's Common Channel Signaling Interconnection Guidelines and Telcordia's (formerly BellCore's) CCS Network Interface Specification document, TR-TSV-000905. In addition, the third party provider shall establish SS7 interconnection at one or more of the BellSouth Gateway Signal Transfer Points (STPs). The payment of all costs associated with the transport of SS7 signals via a third party will be established by mutual agreement of the Parties in writing and shall, by this reference become an integral part of this Agreement.

# 4. CNAM Record Initial Load and Updates

- 4.1 The mechanism to be used by BTI for initial CNAM record load and/or updates shall be determined by mutual agreement. The initial load and all updates shall be provided by BTI in the BellSouth specified format and shall contain records for every working telephone number that can originate phone calls. It is the responsibility of BTI to provide accurate information to BellSouth on a current basis.
- 4.2 Updates to the SMS shall occur no less than once a week, reflect service order activity affecting either name or telephone number, and involve only record additions, deletions or changes.

4.3 BTI CNAM records provided for storage in the BellSouth CNAM SCP shall be available, on a SCP query basis only, to all Parties querying the BellSouth CNAM SCP. Further, CNAM service shall be provided by each Party consistent with state and/or federal regulation.

ПП		AND OTHER SE	KVICES							
						TES BY STA				
DESCRIPTION NIDs	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
NID to NID Cross Connect, 2-Wire or 4-Wire, NRC	UNDC2	\$11.63	\$6.15	NA	\$11.79	\$11.72	NA	\$11.68	NA	NA NA
NID to NID Cross Connect, 2-Wire of 4-Wire, NRC	UNDC4	\$11.63	\$6.15	NA NA	\$11.79	\$11.72	NA NA	\$11.68	NA NA	NA NA
NID, 1-2 lines, per month	UND12	\$11.63 NA	NA	NA NA	Φ11.79 NA	Ψ11.72 NA	NA NA	NA	NA NA	NA NA
NRC - 1st	UND12	TBD	\$94.50	TBD	\$94.56	\$93.90	TBD	TBD	TBD	TBD
NRC - Add'l	UND12	TBD	\$94.50 \$57.22	TBD	\$94.56 \$57.28	\$93.90 \$56.67	TBD	TBD	TBD	TBD
NRC - Disconnect Charge - 1st	UND12	TBD	NA	TBD	\$37.26 NA	νA	TBD	TBD	TBD	TBD
NRC - Disconnect Charge - 1st	UND12	TBD	NA NA	TBD	NA NA	NA NA	TBD	TBD	TBD	TBD
NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	NA	\$3.50	\$3.50	NA NA	\$3.50	NA	TBD
NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA NA	NA	NA	NA NA	NA	NA	TBD
NRC - Service Order submitted Manually, per LSR	SOMAN	NA NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	TBD
NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA NA	\$3.84	NA NA	\$3.94	NA NA	NA NA	NA NA	NA NA	TBD
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	TBD	NA	NA	TBD	TBD	TBD	TBD
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA NA	TBD	NA	NA	TBD	TBD	TBD	TBD
NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA NA	TBD	NA NA	NA	TBD	TBD	TBD	TBD
NID, 1-6 lines, per month	UND16	NA	NA NA	NA NA	NA	NA	NA	NA	NA	NA NA
NRC - 1st	UND16	TBD	\$136.75	TBD	\$136.91	\$135.29	TBD	TBD	TBD	TBD
NRC - Add'l	UND16	TBD	\$99.47	TBD	\$99.63	\$98.07	TBD	TBD	TBD	TBD
NRC - Disconnect Charge - 1st	UND16	TBD	NA	TBD	NA	NA	TBD	TBD	TBD	TBD
NRC - Disconnect Charge - Add'l	UND16	TBD	NA	TBD	NA	NA	TBD	TBD	TBD	TBD
NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	NA	\$3.50	\$3.50	NA	\$3.50	NA	TBD
NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	TBD
NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	TBD
NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	TBD
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	\$21.56	TBD	NA	NA	TBD	TBD	TBD	TBD
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	TBD	NA	NA	TBD	TBD	TBD	TBD
NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	\$3.84	TBD	NA	NA	TBD	TBD	TBD	TBD
Nonrecurring Charge - customer transfer, feature additions, changes (1)		\$5.00	NA	NA	NA	NA	\$5.00	NA	NA	NA
LOOP, EXCLUDING NID										
2-Wire Analog VG Loop (Standard), per month		NA	NA	NA	\$18.20	NA	NA	NA	NA	NA
NRC - 1st		NA	NA	NA	\$86.08	NA	NA	NA	NA	NA
NRC - Add'l		NA	NA	NA	\$58.57	NA	NA	NA	NA	NA
2-Wire Analog VG Loop (Customized), per month		NA	NA	NA	\$21.41	NA	NA	NA	NA	NA
NRC - 1st		NA	NA	NA	\$236.75	NA	NA	NA	NA	NA
NRC - Add'l		NA	NA	NA	\$177.10	NA	NA	NA	NA	NA NA
4-Wire Analog VG Loop (Standard), per month		NA NA	NA NA	NA NA	\$26.38	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - 1st NRC - Add'I		NA NA	NA NA	NA NA	\$457.14 \$348.83	NA NA	NA NA	NA NA	NA NA	NA NA
2-Wire ISDN Digital Grade Loop (Standard), per month		NA NA	NA NA	NA NA	\$348.83	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - 1st		NA NA	NA NA	NA NA	\$29.65	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - Add'l		NA NA	NA NA	NA NA	\$431.61	NA NA	NA NA	NA NA	NA NA	NA NA
2-Wire ADSL Loop (Standard), per month		NA NA	NA NA	NA NA	\$10.63	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - 1st		NA NA	NA NA	NA NA	\$713.50	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - Add'I		NA NA	NA NA	NA NA	\$609.44	NA NA	NA NA	NA NA	NA	NA NA
2-Wire HDSL Loop (Standard), per month		NA NA	NA NA	NA NA	\$7.40	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - 1st		NA NA	NA	NA	\$713.50	NA	NA	NA	NA	NA NA
NRC - Add'l		NA NA	NA NA	NA NA	\$609.44	NA NA	NA NA	NA NA	NA	NA NA
4-Wire HDSL Loop (Standard), per month		NA NA	NA NA	NA NA	\$9.70	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - 1st		NA NA	NA NA	NA.	\$748.93	NA	NA	NA NA	NA.	NA NA
NRC - Add'l		NA	NA NA	NA	\$646.17	NA	NA	NA	NA	NA NA
LOOP, INCLUDING NID		1			Ţ					1
2-Wire Analog VG Loop-SL1		1								
LII TOWN TO THE TO	·	1	·							

			AND OTHER SE									
	RC - Statewide, per month	UEAL2	NA	NA	NA	NA	NA	NA	\$15.88	NA	NA	
	RC - Zone 1, per month (Note 2)	UEAL2	\$15.24	\$13.75	\$14.21	\$14.79	\$14.96	\$15.58	TBD	\$18.48	\$15.92	
	RC - Zone 2, per month (Note 2)	UEAL2	\$24.75	\$20.13	\$16.41	\$27.68	\$25.69	\$20.65	TBD	\$27.87	\$20.79	
	RC - Zone 3, per month (Note 2)	UEAL2	\$44.85	\$44.40	\$26.08	\$47.78	\$52.47	\$29.51	TBD	\$36.91	\$27.18	
	RC - Zone 4, per month (Note 2)	UEAL2	NA	NA	NA	NA	NA	\$38.94	NA	NA	NA	
	NRC - 1st	UEAL2	\$59.03	\$83.20	\$42.54	\$86.08	\$40.69	\$59.25	\$57.99	\$70.44	\$78.93	
	NRC - Add'l	UEAL2	\$43.14	\$35.12	\$31.33	\$58.57	\$29.96	\$43.67	\$42.37	\$44.05	\$50.98	
	NRC - Disconnect Charge - 1st	UEAL2	\$15.21	\$55.97	NA	NA	\$11.48	\$16.35	NA	NA	NA	
	NRC - Disconnect Charge - Add'l	UEAL2	\$3.22	\$10.35	NA	NA	\$3.36	\$4.06	NA	NA	NA	
	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	TBD	
	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.22	NA	
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA	
	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA	
	NRC - Loop Make-Up	UEANM	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	
2-Wi	re Analog VG Loop-SL2 w/loop or ground start signaling											
ШΤ	RC - Statewide, per month	UEAL2	NA	NA	NA	NA	NA	NA	\$19.50	NA	NA	
	RC - Zone 1, per month (Note 2)	UEAL2	\$17.95	\$18.48	\$16.84	\$17.27	\$17.65	\$18.35	TBD	\$21.57	\$15.92	
$\Box \Box \Box$	RC - Zone 2, per month (Note 2)	UEAL2	\$29.16	\$22.43	\$19.45	\$32.32	\$30.32	\$24.33	TBD	\$32.53	\$20.79	
	RC - Zone 3, per month (Note 2)	UEAL2	\$52.84	\$27.87	\$30.92	\$55.78	\$61.93	\$34.77	TBD	\$43.08	\$27.18	
	RC - Zone 4, per month (Note 2)	UEAL2	NA	NA	NA	NA	NA	\$45.88	NA	NA	NA	
	NRC - 1st	UEAL2	\$145.46	\$218.96	\$104.17	\$236.75	\$99.69	\$144.01	\$142.97	\$178.12	\$192.97	
	NRC - Add'l	UEAL2	\$108.40	\$136.44	\$78.10	\$177.10	\$74.73	\$107.70	\$106.56	\$128.80	\$140.72	
	NRC - Disconnect Charge - 1st	UEAL2	\$40.31	\$113.41	NA	NA	\$28.73	\$40.98	NA	NA	NA	
	NRC - Disconnect Charge - Add'l	UEAL2	\$26.01	\$20.58	NA	NA	\$18.87	\$26.95	NA	NA	NA	
	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	TBD	
	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.42	NA	
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA	
	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$26.95	NA	NA	NA	
	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$36.18	\$34.22	\$36.18	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00	
2-Wi	re Analog VG Loop-SL2 w/ reverse battery signaling											
	RC - Statewide, per month	UEAR2	NA	NA	NA	NA	NA	NA	\$19.50	NA	NA	
	RC - Zone 1, per month (Note 2)	UEAR2	\$17.95	\$18.48	\$16.84	\$17.27	\$17.65	\$18.35	TBD	\$21.57	\$15.92	
	RC - Zone 2, per month (Note 2)	UEAR2	\$29.16	\$22.43	\$19.45	\$32.32	\$30.32	\$24.33	TBD	\$32.53	\$20.79	
	RC - Zone 3, per month (Note 2)	UEAR2	\$52.84	\$27.87	\$30.92	\$55.78	\$61.93	\$34.77	TBD	\$43.08	\$27.18	
	RC - Zone 4, per month (Note 2)	UEAR2	NA	NA	NA	NA	NA	\$45.88	NA	NA	NA	
	NRC - 1st	UEAR2	\$145.46	\$218.96	\$104.17	\$236.75	\$99.69	\$144.01	\$142.97	\$178.12	\$192.97	
	NRC - Add'l	UEAR2	\$108.40	\$136.44	\$78.10	\$177.10	\$74.73	\$107.70	\$106.56	\$128.80	\$140.72	
	NRC - Disconnect Charge - 1st	UEAR2	\$40.31	\$113.41	NA	NA	\$28.73	\$40.98	NA	NA	NA	
	NRC - Disconnect Charge - Add'l	UEAR2	\$26.01	\$20.58	NA	NA	\$18.87	\$26.95	NA	NA	NA	
	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	TBD	
	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.42	NA	
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA	
	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$26.95	NA	NA	NA	
	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOCL	\$45.99	\$36.18	\$34.22	\$36.18	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00	
4-Wi	re Analog VG Loop			1			1					
	RC - Statewide, per month	UEAL4	NA	NA	NA	NA	NA	NA	\$27.49	NA	NA	

				AND OTHER SE	RVICES								
		RC - Zone 1, per month (Note 2)	UEAL4	\$24.01	\$24.26	\$22.26	\$20.92	\$24.36	\$22.38	TBD	\$29.47	\$15.92	
		RC - Zone 2, per month (Note 2)	UEAL4	\$39.00	\$35.51	\$25.70	\$39.14	\$41.85	\$29.67	TBD	\$44.44	\$20.79	
		RC - Zone 3, per month (Note 2)	UEAL4	\$70.67	\$78.35	\$40.85	\$67.56	\$85.47	\$42.40	TBD	\$58.85	\$27.18	
		RC - Zone 4, per month (Note 2)	UEAL4	NA	NA	NA	NA	NA	\$55.96	NA	NA	NA	
		NRC - 1st	UEAL4	\$293.70	\$141.00	\$206.95	\$457.14	\$198.10	\$289.06	\$288.47	\$383.39	\$58.50	
H		NRC - Add'l	UEAL4	\$241.76	\$43.00	\$170.57	\$348.83	\$163.26	\$238.19	\$237.45	\$286.77	\$31.00	
		NRC - Disconnect Charge - 1st	UEAL4	\$108.96	\$122.15	NA	NA	\$74.27	\$108.14	NA	NA	NA	
H		NRC - Disconnect Charge - Add'l	UEAL4	\$57.01	\$27.42	NA	NA	\$39.44	\$57.28	NA	NA	NA	
		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
H		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA NA	NA	NA	NA NA	NA NA	TBD	
+++		NRC - Service Order submitted Manually, per LSR	SOMAN	NA NA	\$21.56	NA NA	\$29.24	NA NA	NA NA	NA NA	NA NA	\$19.99	
HH		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA NA	\$3.84	NA NA	\$3.94	NA	NA NA	NA NA	NA NA	TBD	
HH	_	NRC - Service Order submitted Manually, per LSR, Disconnect  NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	Ψ3.04 NA	\$18.94	Ψ3.94 NA	\$18.14	\$25.52	\$26.94	\$44.06	NA NA	
HH								\$8.06				NA NA	
++-		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN SOMAN	\$12.97	NA NA	\$8.42	NA NA		\$11.34	\$12.76	\$13.55	NA NA	
++-		NRC - Incremental Charge - Manual Service Order - Disconnect		\$17.77	\$36.18	NA \$34.22		\$11.41	\$16.06	NA \$45.34	NA \$45.43	\$55.00	
Щ.		NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$36.18	\$34.22	\$36.18	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00	
2-1		ISDN Digital Grade Loop											
HH		RC - Statewide, per month	U1L2X	NA	NA	NA	NA	NA	NA	\$24.98	NA	NA	
H		RC - Zone 1, per month (Note 2)	U1L2X	\$23.23	\$32.34	\$21.89	\$23.66	\$21.15	\$21.86	TBD	\$26.68	\$15.92	
		RC - Zone 2, per month (Note 2)	U1L2X	\$37.74	\$47.35	\$25.27	\$44.28	\$36.22	\$28.97	TBD	\$40.24	\$20.79	
		RC - Zone 3, per month (Note 2)	U1L2X	\$68.38	\$104.47	\$40.17	\$76.42	\$74.19	\$41.40	TBD	\$53.29	\$27.18	
		RC - Zone 4, per month (Note 2)	U1L2X	NA	NA	NA	NA	NA	\$54.64	NA	NA	NA	
		NRC - 1st	U1L2X	\$331.85	\$306.00	\$233.38	\$541.28	\$223.27	\$326.38	\$325.91	\$423.04	\$58.50	
		NRC - Add'l	U1L2X	\$255.87	\$283.00	\$180.35	\$431.61	\$172.63	\$252.00	\$251.31	\$301.75	\$31.00	
		NRC - Disconnect Charge - 1st	U1L2X	\$108.95	\$111.10	NA	NA	\$74.27	\$108.14	NA	NA	NA	
		NRC - Disconnect Charge - Add'l	U1L2X	\$57.01	\$18.28	NA	NA	\$39.44	\$57.27	NA	NA	NA	
		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	TBD	
		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
H		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	TBD	
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA NA	\$18.94	NA NA	\$18.14	\$25.52	\$26.94	\$44.42	NA NA	-
H		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA NA	
+++		NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA NA	NA	NA NA	\$11.41	\$16.06	NA	NA	NA NA	
+++		NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$36.18	\$34.22	\$36.18	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00	
1		,		945.99 NA	930.16 NA	νA.22	930.16 NA	NA	NA		NA	NA	
		Universal Digital Carrier (UDC), statewide, per month  Zone 1, per month	UDC2X UDC2X	\$23.23	\$32.34	\$21.89	\$23.66	\$21.15	\$21.86	\$24.98 TBD	\$26.68	\$15.92	
++-													
++-		Zone 2, per month	UDC2X	\$37.74	\$47.35	\$25.27	\$44.28	\$36.22	\$28.97	TBD	\$40.24	\$20.79	$\longrightarrow$
H	_	Zone 3, per month	UDC2X	\$68.38	\$104.47	\$40.17	\$76.42	\$74.19	\$41.40	TBD	\$53.29	\$27.18	
		Zone 4, per month	UDC2X	NA	NA Table 22	NA Table 22	NA 2511.00	NA	\$54.64	NA 1005.04	NA 0100.01	NA 050.50	
++		NRC - 1st	UDC2X	\$331.85	\$306.00	\$233.38	\$541.28	\$223.27	\$326.38	\$325.91	\$423.04	\$58.50	
H		NRC - Add'l	UDC2X	\$255.87	\$283.00	\$180.35	\$431.61	\$172.63	\$252.00	\$251.31	\$301.75	\$31.00	
HL		NRC - Disconnect Charge - 1st	UDC2X	\$108.95	\$111.10	NA	NA	\$74.27	\$108.14	NA	NA	NA	
$\Box$		NRC - Disconnect Charge - Add'l	UDC2X	\$57.01	\$18.28	NA	NA	\$39.44	\$57.27	NA	NA	NA	
Ш		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	TBD	
		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	TBD	
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.42	NA	
		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA	
		NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA	
		NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$36.18	\$34.22	\$36.18	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00	
2-1	Wire	ADSL Compatible Loop incl Man Svc Inquiry & Fac Reservation											
H15.		RC - Statewide, per month	UAL2X	NA	NA	NA	NA	NA	NA	\$14.60	NA	NA	
HH		RC - Zone 1, per month (Note 2)	UAL2X	\$12.09	\$12.78	\$11.23	\$8.79	\$11.90	\$10.87	TBD	\$17.10	\$15.93	-
шш		Lond i, por monar (note 2)	UNLLA	Ψ12.00	Ψ12.70	Ψ11.20	ψυ.10	Ψ11.00	Ψ10.01	. 50	Ψιτιίο	ψ10.00	

				AND OTHER SE									
Ш		RC - Zone 2, per month (Note 2)	UAL2X	\$19.64	\$18.72	\$12.97	\$16.46	\$20.43	\$14.40	TBD	\$25.79	\$20.05	
		RC - Zone 3, per month (Note 2)	UAL2X	\$35.59	\$41.29	\$20.62	\$28.40	\$41.73	\$20.58	TBD	\$34.15	\$28.74	
		RC - Zone 4, per month (Note 2)	UAL2X	NA	NA	NA	NA	NA	\$27.16	NA	NA	NA	
		NRC - 1st	UAL2X	\$514.21	\$113.85	\$359.73	\$713.50	\$343.13	\$504.82	\$504.90	\$600.61	\$640.79	
		NRC - Add'l	UAL2X	\$464.58	\$99.61	\$325.15	\$609.44	\$310.03	\$456.24	\$456.17	\$507.33	\$541.94	
Ħ		NRC - Disconnect Charge - 1st	UAL2X	\$106.65	\$154.23	NA	NA	\$72.54	\$105.86	NA	NA	NA	
H		NRC - Disconnect Charge - Add'l	UAL2X	\$56.98	\$35.23	NA	NA	\$39.42	\$57.25	\$26.94	NA	NA	
Ħ		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
H		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
H	+	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA NA	\$3.84	NA NA	\$3.94	NA	NA	NA	NA	TBD	
H	$\vdash$	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$12.76	\$44.42	NA NA	
H		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$12.97	NA	\$8.42	NA NA	\$8.06	\$11.34	NA	\$13.55	NA NA	
H		NRC - Incremental Charge - Manual Service Order - Add 1	SOMAN	\$17.77	NA NA	NA	NA NA	\$11.41	\$16.06	NA NA	NA	NA NA	
-													
H	0.147	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$36.18	\$34.22	\$36.18	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00	
1		e ADSL Compatible Loop without Man Svc Inquiry & Fac Reservation											
4	RC -	Statewide, per month	UAL2W	NA	NA	NA	NA	NA	NA	\$14.60	NA	NA	
Ш		Zone 1, per month	UAL2W	\$12.09	\$12.78	\$11.23	\$8.79	\$11.90	\$10.87	TBD	\$17.10	\$15.93	
		Zone 2, per month	UAL2W	\$19.64	\$18.72	\$12.97	\$16.46	\$20.43	\$14.40	TBD	\$25.79	\$20.05	
Ш		Zone 3, per month	UAL2W	\$35.59	\$41.29	\$20.62	\$28.40	\$41.73	\$20.58	TBD	\$34.15	\$28.74	
		Zone 4, per month	UAL2W	NA	NA	NA	NA	NA	\$27.16	NA	NA	NA	
		NRC - 1st	UAL2W	\$375.21	\$258.86	\$220.73	\$260.95	\$255.34	\$365.82	\$365.90	\$461.60	\$501.79	
		NRC - Add'l	UAL2W	\$325.58	\$175.48	\$186.15	\$176.21	\$170.72	\$317.24	\$317.17	\$368.33	\$402.94	
		NRC - Disconnect Charge - 1st	UAL2W	\$106.65	\$108.29	NA	\$112.86	\$112.32	\$105.86	NA	NA	NA	
		NRC - Disconnect Charge - Add'l	UAL2W	\$56.98	\$15.46	NA	\$15.88	\$15.81	\$57.25	\$26.94	NA	NA	
		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	TBD	
		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
H		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	TBD	
H		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$12.76	\$44.42	NA NA	
H		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	NA NA	\$13.55	NA	
H		NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA NA	NA NA	NA	\$11.41	\$16.06	NA	NA	NA	
H		NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$36.18	\$34.22	\$36.18	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00	
-	2 14/:-	re HDSL Compatible Loop, incl Man Svc Inquiry & Fac Reserv	OCOSL	ψ <del>4</del> 5.99	φ30.10	φ34.ZZ	φ30.10	φ32.11	φ <del>4</del> 5.27	φ45.54	φ45.45	φ55.00	
-	2-7711	RC - Statewide, per month	UHL2X	NA	NA	NA	NA	NA	NA	\$11.98	NA	NA	
H													
H		RC - Zone 1, per month (Note 2)	UHL2X	\$9.41	\$9.80	\$7.88	\$6.29	\$8.97	\$8.50	TBD	\$12.21	\$11.62	
1		RC - Zone 2, per month (Note 2)	UHL2X	\$15.29	\$14.35	\$9.09	\$11.78	\$15.41	\$11.26	TBD	\$18.41	\$14.62	
4		RC - Zone 3, per month (Note 2)	UHL2X	\$27.70	\$31.65	\$14.46	\$20.33	\$31.48	\$16.10	TBD	\$24.39	\$20.96	
$\vdash$	$\vdash \vdash$	RC - Zone 4, per month (Note 2)	UHL2X	NA	NA	NA	NA	NA	\$21.25	NA	NA	NA	
$\perp$	oxdot	NRC - 1st	UHL2X	\$514.21	\$113.85	\$359.73	\$713.50	\$343.13	\$504.82	\$504.90	\$600.61	\$640.79	
Ш		NRC - Add'l	UHL2X	\$464.58	\$99.61	\$325.15	\$609.44	\$310.03	\$456.24	\$456.17	\$507.33	\$541.94	
$\perp$	oxdot	NRC - Disconnect Charge - 1st	UHL2X	\$106.65	\$154.23	NA	NA	\$72.54	\$105.86	NA	NA	NA	
Ш		NRC - Disconnect Charge - Add'l	UHL2X	\$56.98	\$35.23	NA	NA	\$39.42	\$57.25	NA	NA	NA	
Ш		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
Ш		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	TBD	
Ш		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	TBD	
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.06	NA	
		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA	
		NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA	
IT		NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$36.18	\$34.22	\$36.18	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00	
H	2-Wir	e HDSL Compatible Loop, without Man Svc Inquiry & Fac Reserv		¥	<del>+</del>	<del></del>	******	**	<del>*</del> · · · · ·	¥	*	,,,,,,,	
H		wide, per month	UHL2W	NA	NA	NA	NA	NA	NA	\$11.98	NA	NA	
H	1	Zone 1, per month	UHL2W	\$9.41	\$9.80	\$7.88	\$6.29	\$8.97	\$8.50	TBD	\$12.21	\$11.62	
H		Zone 2, per month	UHL2W	\$15.29	\$14.35	\$9.09	\$11.78	\$15.41	\$11.26	TBD	\$18.41	\$14.62	
+	$\vdash$	Zone 3, per month	UHL2W	\$27.70	\$31.65	\$14.48	\$20.33	\$31.48	\$16.10	TBD	\$24.39	\$20.96	
ш		, p	J TT	<b>4</b> 0	Ψ51.00	ψ. n. 10	<b>\$25.00</b>	<b>\$510</b>	ψ.J.10		¥= 1.00	ψ=0.00	

				AND OTHER SE	RVICES								
		Zone 4, per month	UHL2W	NA	NA	NA	NA	NA	\$21.25	NA	NA	NA	
		NRC - 1st	UHL2W	\$375.21	\$276.19	\$220.73	\$278.28	\$272.64	\$365.82	\$365.90	\$461.60	\$501.79	
		NRC - Add'l	UHL2W	\$325.58	\$192.81	\$186.15	\$193.54	\$188.02	\$317.24	\$317.17	\$368.33	\$402.94	
		NRC - Disconnect Charge - 1st	UHL2W	\$106.65	\$108.29	NA	\$112.86	\$112.32	\$105.86	NA	NA	NA	
		NRC - Disconnect Charge - Add'l	UHL2W	\$56.98	\$15.46	NA	\$15.88	\$15.81	\$57.25	NA	NA	NA	
		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	TBD	
H		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	-
H		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA NA	\$3.84	NA NA	\$3.94	NA	NA NA	NA NA	NA	TBD	
+	-	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.06	NA NA	
H	-	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA NA	\$8.42	NA NA	\$8.06	\$11.34	\$12.76	\$13.55	NA NA	
H		NRC - Incremental Charge - Manual Service Order - Add 1	SOMAN	\$17.77	NA NA	NA	NA NA	\$11.41	\$16.06	NA	NA	NA NA	
H		v		\$45.99	\$36.18	\$34.22	\$36.18			\$45.34	\$45.43	\$55.00	
-	4 180	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$30.18	\$34.22	\$30.18	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00	
Ш	4-Wir	e HDSL Compatible Loop, incl Man Svc Inquiry & Fac Reserv	11111 437							<b>010.07</b>		A47.04	
Ш		RC - Statewide, per month	UHL4X	NA • • • • •	NA	NA	NA	NA	NA	\$13.97	NA	\$17.91	
$\mathbf{H}$		RC - Zone 1, per month (Note 2)	UHL4X	\$11.52	\$14.75	\$10.39	\$7.68	\$12.97	\$10.36	TBD	\$16.21	\$15.46	
Ш		RC - Zone 2, per month (Note 2)	UHL4X	\$18.71	\$21.59	\$12.00	\$14.38	\$21.76	\$13.73	TBD	\$24.45	\$19.46	
Ш		RC - Zone 3, per month (Note 2)	UHL4X	\$33.90	\$47.64	\$19.07	\$24.82	\$44.44	\$19.62	TBD	\$32.38	\$27.88	
Ш		RC - Zone 4, per month (Note 2)	UHL4X	NA	NA	NA	NA	NA	\$25.90	NA	NA	NA	
		NRC - 1st	UHL4X	\$541.13	\$116.91	\$378.86	\$748.93	\$361.45	\$531.21	\$531.35	\$625.11	\$666.70	
Ш		NRC - Add'I	UHL4X	\$491.50	\$101.71	\$344.28	\$646.17	\$328.35	\$482.63	\$482.62	\$532.78	\$568.86	
		NRC - Disconnect Charge - 1st	UHL4X	\$106.65	\$161.19	NA	NA	\$72.54	\$105.86	NA	NA	NA	
		NRC - Disconnect Charge - Add'l	UHL4X	\$56.98	\$26.10	NA	NA	\$39.42	\$57.25	NA	NA	NA	
		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
		NRC - Setvice Order submitted Electronically, per LSR - Disconnect NRC - Service Order submitted Manually, per LSR NRC - Service Order submitted Manually, per LSR, Disconnect	SOMEC SOMAN SOMAN	NA NA NA	\$0.42 \$21.56 \$3.84	NA NA NA	NA \$29.24 \$3.94	NA NA NA	NA NA NA	NA NA NA	NA NA NA	TBD \$19.99 TBD	
H	-	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.06	NA NA	
+		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$12.97	NA NA	\$8.42	NA NA	\$8.06	\$11.34	\$12.76	\$13.55	NA NA	
+		NRC - Incremental Charge - Manual Service Order - Add 1	SOMAN	\$17.77	NA NA	NA	NA NA	\$11.41	\$16.06	NA	NA	NA NA	
+		NRC - Incremental Charge - Manual Service Order - Disconnect  NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$36.18	\$34.22	\$36.18	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00	
-	4 180	·	UCUSL	\$45.99	\$30.18	\$34.22	\$30.18	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00	
-		e HDSL Compatible Loop, without Man Svc Inquiry & Fac Res	1 11 11 4147	NA	NA	NA	NA	NA	NA	<b>640.07</b>	NA	\$17.91	
-	State	wide, per month	UHL4W							\$13.97			
+		Zone 1, per month	UHL4W	\$11.52	\$14.75	\$10.39	\$7.68	\$12.97	\$10.36	TBD TBD	\$16.21	\$15.46	
+		Zone 2, per month	UHL4W	\$18.71	\$21.59	\$12.00	\$14.38	\$21.76	\$13.73		\$24.45	\$19.46	$\longrightarrow$
Ш		Zone 3, per month	UHL4W	\$33.90	\$47.64	\$19.07	\$24.82	\$44.44	\$19.62	TBD	\$32.38	\$27.88	-
$\perp$		Zone 4, per month	UHL4W	NA ©400.40	NA COOR 40	NA ************************************	NA ************************************	NA ************************************	\$25.90	NA ************************************	NA ***	NA 0507.70	
Ш		NRC - 1st	UHL4W	\$402.13	\$333.40	\$239.86	\$276.76	\$329.76	\$392.21	\$392.35	\$486.11	\$527.70	
Щ		NRC - Add'l	UHL4W	\$352.50	\$250.01	\$205.28	\$192.02	\$245.15	\$343.63	\$343.62	\$393.78	\$429.86	
		NRC - Disconnect Charge - 1st	UHL4W	\$106.65	\$114.30	NA	\$122.17	\$124.19	\$105.86	NA	NA	NA	
		NRC - Disconnect Charge - Add'l	UHL4W	\$56.98	\$19.58	NA	\$20.64	\$20.98	\$57.25	NA	NA	NA	
Ш		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA NA	\$0.42	NA	NA	NA	NA	NA	NA	TBD	
		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
H		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	TBD	
+		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.06	NA NA	_
+		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA NA	\$8.42	NA NA	\$8.06	\$11.34	\$12.76	\$13.55	NA NA	-
+	+	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA	NA	NA NA	\$11.41	\$16.06	NA	NA	NA NA	
H	$\vdash$	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$36.18	\$34.22	\$36.18	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00	-
H	4-Wir	e DS1 Digital Loop	3000L	ψ10.00	ψου.10	ΨΟ Γ.ΔΔ	ψου.10	Ψ02.11	Ψ10.21	ψ 10.0 <del>1</del>	ψ10.70	Ψ00.00	-
ш			l	l .									

			AND OTHER SE								
	RC - Statewide, per month	USLXX	NA	NA	NA	NA	NA	NA	\$62.78	NA	NA
	RC - Zone 1, per month (Note 2)	USLXX	\$51.74	\$64.69	\$55.53	\$50.26	\$56.32	\$50.99	TBD	\$59.61	\$57.73
	,										
	RC - Zone 2, per month (Note 2)	USLXX	\$84.05	\$94.71	\$64.13	\$94.06	\$96.73	\$67.58	TBD	\$89.90	\$75.40
+++	RC - Zone 3, per month (Note 2)	USLXX	\$152.29	\$208.93	\$101.93	\$162.34	\$197.57	\$96.58	TBD	\$119.06	\$98.59
+++	RC - Zone 4, per month (Note 2)	USLXX	\$152.29 NA	\$206.93 NA	NA	\$162.34 NA		\$127.47	NA	NA	δ96.59 NA
$\vdash$	, , ,						NA				
	NRC - 1st	USLXX	\$610.13	\$540.00	\$429.98	\$849.80	\$410.38	\$599.09	\$714.84	\$715.77	\$313.08
	NRC - Add'I	USLXX	\$380.26	\$465.00	\$268.18	\$523.27	\$255.48	\$373.90	\$421.47	\$421.50	\$219.72
	NRC - Disconnect Charge - 1st	USLXX	\$134.77	\$82.85	NA	NA	\$92.35	\$133.53	NA	NA	\$96.86
	NRC - Disconnect Charge - Add'l	USLXX	\$55.97	\$21.69	NA	NA	\$38.44	\$56.25	NA	NA	\$40.45
	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	TBD
	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99
	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	TBD
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$43.77	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
+++	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA NA	NA	NA NA	\$11.41	\$16.06	NA	NA	NA NA
+++	more more manage manual octyles of del - Disconnect	JOIVIAIN	ψ11.11	INA	INA	INA	ψ11. <del>4</del> 1	ψ10.00	INA	INA	INA
			ĺ								
ШШ	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$49.18	\$36.18	\$34.52	\$36.18	\$33.05	\$48.17	\$45.34	\$48.47	\$55.00
4-Wi	re 56 Kbps Dig Grade Loop										
	RC - Statewide, per month	UDL56	NA	NA	NA	NA	NA	NA	\$32.67	NA	\$42.23
	RC - Zone 1, per month (Note 2)	UDL56	\$27.33	\$39.08	\$25.75	\$35.92	\$27.50	\$25.61	TBD	\$34.26	\$36.45
	RC - Zone 2, per month (Note 2)	UDL56	\$44.40	\$57.21	\$29.74	\$40.32	\$47.24	\$33.94	TBD	\$51.67	\$45.87
	RC - Zone 3, per month (Note 2)	UDL56	\$80.45	\$126.22	\$47.27	\$37.90	\$96.48	\$48.51	TBD	\$68.43	\$65.75
	RC - Zone 4, per month (Note 2)	UDL56	NA	NA	NA	NA	NA	\$64.02	NA	NA	NA
	NRC - 1st	UDL56	\$498.05	\$654.72	\$348.55	\$258.62	\$333.28	\$489.00	\$489.04	\$602.73	\$643.00
++++	NRC - Add'l	UDL56	\$343.70	\$428.45	\$241.20	\$176.11	\$230.50	\$337.93	\$337.51	\$393.50	\$421.26
++++	NRC - Disconnect Charge - 1st	UDL56	\$129.62	\$122.15	NA	\$127.74	\$87.99	\$128.36	NA	\$44.06	NA NA
++++	NRC - Disconnect Charge - 1st	UDL56	\$64.25	\$27.42	NA NA	\$27.90	\$44.24	\$64.35	NA NA	\$13.55	NA NA
+++	· · ·						•				
$\vdash$	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	TBD
	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99
	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	TBD
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	NA	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	NA	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA NA	NA	NA	\$11.41	\$16.06	NA NA	NA	NA
HHH	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$36.18	\$34.22	\$36.18	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00
4-Wi	re 64 Kbps Dig Grade Loop	00001	ψ-σ.σσ	ψ50.10	Ψ07.22	ψου. το	Ψ02.11	ψ-τυ.Σ1	ψ-το.υ-τ	ψτυ.τυ	ψ00.00
1-4-441	RC - Statewide, per month	UDL64	NA	NA	NA	NA	NA	NA	\$32.67	\$41.70	\$42.23
+++		UDL64	\$27.33	\$39.08	\$25.75	\$35.92	\$27.50	\$25.61	TBD	\$34.26	\$36.45
+++	RC - Zone 1, per month (Note 2)										
+++	RC - Zone 2, per month (Note 2)	UDL64	\$44.40	\$57.21	\$29.74	\$40.32	\$47.24	\$33.94	TBD	\$51.67	\$45.87
HH	RC - Zone 3, per month (Note 2)	UDL64	\$80.45	\$126.22	\$47.27	\$37.90	\$96.48	\$48.51	TBD	\$68.43	\$65.75
$\sqcup \! \! \perp \! \! \! \perp$	RC - Zone 4, per month (Note 2)	UDL64	NA	NA	NA	NA	NA	\$64.02	NA	NA	NA
	NRC - 1st	UDL64	\$498.05	\$654.72	\$348.55	\$258.62	\$333.28	\$489.00	\$489.04	\$602.73	\$643.00
ШШ	NRC - Add'l	UDL64	\$343.70	\$428.45	\$241.20	\$176.11	\$230.50	\$337.93	\$337.51	\$393.50	\$421.26
	NRC - Disconnect Charge - 1st	UDL64	\$129.62	\$122.15	NA	\$127.74	\$87.99	\$128.36	NA	\$44.06	NA
	NRC - Disconnect Charge - Add'l	UDL64	\$64.25	\$27.42	NA	\$27.90	\$44.24	\$64.35	NA	\$13.55	NA
	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
+++	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
	price Service State Submitted Electromodity, per Lore Disconnect	OCIVILO	14/3	Ψ0.72	14/-1	14/1	14/3	14/3	11/1	14/-1	13/3

				AND OTHER SE	RVICES								
		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	NA	
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	NA	NA	
		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	NA	NA	
		NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA	
h		NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$36.18	\$34.22	\$36.18	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00	
H	2-Wir	e Unb Copper Loop/Short(< or = 18kft), incl Man SI & Fac Res*	00002	<b>\$10.00</b>	φοσιτο	Ψ0	φοσιτο	Ψ02	ψ.σ. <u>z</u> .	Ψ10.01	Ψ.σσ	φοσισσ	
H	7	RC - Statewide, per month	UCLPB	\$15.11	NA	\$13.97	NA	NA	NA	NA	\$20.81	NA	
+	<del>                                      </del>	RC - Zone 1, per month (Note 2)	UCLPB	TBD	\$18.60	\$19.80	\$14.94	\$16.34	\$16.85	\$12.27	\$18.90	\$19.85	
H	-	RC - Zone 2, per month (Note 2)	UCLPB	TBD	\$27.23	\$22.86	\$15.15	\$17.99	\$22.34	\$20.63	\$28.50	\$24.98	
$\vdash$													
$\vdash$		RC - Zone 3, per month (Note 2)	UCLPB	TBD	\$60.07	\$36.34	\$15.73	\$18.83	\$31.92	\$23.88	\$37.75	\$35.81	
$\vdash$		RC - Zone 4, per month (Note 2)	UCLPB	NA	NA	NA	NA	NA	\$42.13	NA	NA	NA	
		NRC - 1st	UCLPB	\$514.21	\$389.84	\$395.16	\$391.93	\$386.13	\$504.82	\$450.00	\$600.61	\$270.01	
		NRC - Add'l	UCLPB	\$464.58	\$251.26	\$217.39	\$251.98	\$246.38	\$456.24	\$390.00	\$507.33	\$234.63	
		NRC - Disconnect Charge - 1st	UCLPB	TBD	\$154.23	\$142.27	\$160.06	\$159.29	\$105.86	NA	NA	\$74.54	
		NRC - Disconnect Charge - Add'l	UCLPB	TBD	\$35.23	\$37.86	\$36.20	\$36.02	\$57.25	NA	NA	\$39.14	
		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA	
		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
Ħ		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	NA	
H	TT	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$47.00	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$47.00	NA	
H		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$21.00	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$25.52	NA NA	
H		NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	\$142.27	NA	\$11.41	\$16.06	NA	\$21.00	NA NA	
+	<del>                                     </del>	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$17.77	NA NA	\$37.86	NA NA	\$11.41	\$16.06	NA NA	\$21.00	NA NA	
H	$\vdash$												
H		NRC - Incremental Charge - Manual Order Coordination - per loop	UCLMC	\$16.00	\$16.31	\$36.46	\$16.31	\$32.77	\$45.27	\$16.00	\$45.43	\$34.29	
H		Unb Copper Loop/Short (< or = 18kft), without Man SI & Fac Res									****		
$\perp$	RC -	Statewide, per moth	UCLPW	NA	NA	NA	NA	NA	NA	NA	\$20.81	NA	
Ш		Zone 1, per month	UCLPW	TBD	\$18.60	\$11.90	\$14.94	\$16.34	\$16.85	\$12.27	\$18.90	\$19.85	
Ш		Zone 2, per month	UCLPW	TBD	\$27.23	\$13.74	\$15.15	\$17.99	\$22.34	\$20.63	\$28.50	\$24.98	
		Zone 3, per month	UCLPW	TBD	\$60.07	\$21.83	\$15.73	\$18.83	\$31.92	\$23.88	\$37.75	\$35.81	
Ш		Zone 4, per month	UCLPW	NA	NA	NA	NA	NA	\$42.13	NA	NA	NA	
		NRC - 1st	UCLPW	\$375.21	\$257.00	\$154.13	\$259.09	\$253.48	\$365.82	\$311.00	\$461.61	\$131.01	
		NRC - Add'l	UCLPW	\$325.58	\$173.62	\$139.75	\$174.35	\$168.86	\$317.24	\$251.00	\$368.33	\$95.63	
		NRC - Disconnect Charge - 1st	UCLPW	TBD	\$108.29	\$140.73	\$112.86	\$112.32	\$105.86	NA	NA	\$74.54	
		NRC - Disconnect Charge - Add'l	UCLPW	TBD	\$15.46	\$37.45	\$15.88	\$15.81	\$57.25	NA	NA	\$39.14	
Ħ		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
h		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA	
Ħ		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
H	++	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA NA	\$3.84	NA NA	\$3.94	NA NA	NA NA	NA NA	NA	NA NA	
H	++	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$47.00	Ψ3.04 NA	NA NA	Ψ3.94 NA	\$18.14	\$25.52	\$26.94	\$47.00	NA NA	
H	$\vdash\vdash$	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$21.00	NA NA	NA NA	NA NA	\$8.06	\$11.34	\$12.76	\$25.52	NA NA	
+	++	NRC - Incremental Charge - Manual Service Order - Add 1	SOMAN	\$17.77	NA NA	NA NA	NA NA	\$11.41	\$11.3 <del>4</del> \$16.06	\$12.76 NA	\$23.52	NA NA	
$\vdash$	$\vdash\vdash$	ů				NA NA	NA NA					NA NA	
H	$\vdash \vdash$	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$17.77	NA ©10.01			\$11.41	\$16.06	NA ©40.00	\$21.00		-
+	$\sqcup \bot$	NRC - Incremental Charge - Manual Order Coordination - per loop	UCLMC	\$16.00	\$16.31	\$36.46	\$16.31	\$32.77	\$45.27	\$16.00	\$45.43	\$34.29	
$\perp$	L												
	2-Wir	e Unb Copper Loop/Long (> 18kft), incl Man SI & Fac Res											
Ш		RC - Statewide, per month	UCL2L	\$40.00	NA	\$41.61	NA	NA	NA	NA	NA	NA	
Ш		RC - Zone 1, per month (Note 2)	UCL2L	TBD	\$48.79	\$19.80	\$36.19	\$43.92	\$16.85	\$38.66	\$18.90	\$19.85	
Ш		RC - Zone 2, per month (Note 2)	UCL2L	TBD	\$58.13	\$22.86	\$49.31	\$59.76	\$22.34	\$64.03	\$28.50	\$24.98	
		RC - Zone 3, per month (Note 2)	UCL2L	TBD	\$71.17	\$36.34	\$80.78	\$104.74	\$31.92	\$73.89	\$37.75	\$35.81	
		RC - Zone 4, per month (Note 2)	UCL2L	NA	NA	NA	NA	NA	\$42.13	NA	NA	NA	
		, , ,											
		NRC - 1st	UCL2L	\$514.21	\$331.86	\$395.16	\$333.21	\$332.73	\$504.82	\$450.00	\$600.61	\$270.01	
H	++	NRC - Add'l	UCL2L	\$464.58	\$193.27	\$217.39	\$193.26	\$192.99	\$456.24	\$390.00	\$507.33	\$234.63	
ш		1410 7441	UCLZL	ψ+υ+.υυ	Ψ133.27	Ψ <u></u> 11.33	ψ133.20	ψ132.33	Ψ430.24	Ψ000.00	ψυυ1.υυ	ψ <u>2</u> 04.00	

			AND OTHER SE	RVICES								
ШШ	NRC - Disconnect Charge - 1st	UCL2L	NA	\$154.23	\$142.27	\$160.06	\$159.29	\$105.86	NA	NA	\$74.54	
	NRC - Disconnect Charge - Add'l	UCL2L	NA	NA	\$37.86	\$36.20	\$36.02	\$57.25	NA	NA	\$39.14	
	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	TBD	
	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$47.00	NA	\$18.94	\$47.00	\$18.14	\$25.52	\$26.94	\$47.00	NA	
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$21.00	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$25.52	NA	
	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	\$142.27	NA	\$11.41	\$16.06	NA	\$21.00	NA	
	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'L	SOMAN	\$17.77	NA	\$37.86	NA	\$11.41	\$16.06	NA	\$21.00	NA	
	NRC - Incremental Charge - Manual Order Coordination - per loop	UCLMC	\$16.00	\$16.31	\$36.46	\$16.31	\$32.77	\$45.27	\$16.00	\$45.43	\$34.29	
HH	Parisa Pa		¥10.00	Ţ	400110	<b>V</b> 10101	***	¥ 101=1	Ţ	<b>V</b> 10110	70 1120	
2-V	/ire Unb Copper Loop/Long (> 18kft), without Man SI & Fac Res		1		1							
	- Statewide, per month	UCL2W	\$40.00	NA	\$37.00	NA	NA	NA	NA	NA	NA	
H	Zone 1, per month	UCL2W	TBD	\$48.79	TBD	\$36.19	\$43.92	\$16.85	\$38.66	\$18.90	\$19.85	
++++	25.16 1, 501 1101111	0022	.55	ψ.σσ	.55	φοσιιο	ψ.ιο.ιο <u>Σ</u>	ψ.σ.σσ	Ψ00.00	ψ.σ.σσ	ψ.σ.σσ	
	7 0	UCL2W	TBD	ФE0 40	TBD	£40.04	\$59.76	\$22.34	\$64.03	\$28.50	\$24.98	
++++	Zone 2, per month Zone 3, per month	UCL2W UCL2W	TBD	\$58.13 \$71.17	TBD	\$49.31 \$80.78				\$28.50		
++++	* 1				NA NA		\$104.74	\$31.92	\$73.89 NA	\$37.75 NA	\$35.81 NA	
+++	Zone 4, per month  NRC - 1st	UCL2W	NA fore or	NA \$199.01		NA \$200.37	NA Topo on	\$42.13			\$131.01	
+++	NRC - Add'l	UCL2W	\$375.21		\$154.13		\$200.08	\$365.82	\$311.00	\$461.61		
+++		UCL2W	\$325.58	\$115.63	\$139.75	\$115.63	\$115.46	\$317.24	\$251.00	\$368.33	\$95.63	
+++	NRC - Disconnect Charge - 1st	UCL2W	NA	\$108.29	TBD	\$112.86	\$112.32	\$105.86	NA	NA	\$74.54	
+++	NRC - Disconnect Charge - Add'l	UCL2W	NA no.50	\$15.46	TBD	\$15.88	\$15.81	\$57.25	NA 10.50	NA 20.50	\$39.14	
+++	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
$\square$	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA	
$\Box$	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
Ш	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	NA	
$\Box$	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$47.00	NA	TBD	NA	\$18.14	\$25.52	\$26.94	\$47.00	NA	
ш	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$21.00	NA	TBD	NA	\$8.06	\$11.34	\$12.76	\$25.52	NA	
	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	TBD	NA	\$11.41	\$16.06	NA	\$21.00	NA	
	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'L	SOMAN	\$17.77	NA	TBD	NA	\$11.41	\$16.06	NA	\$21.00	NA	
	NRC - Incremental Charge - Manual Order Coordination - per loop	UCLMC	\$16.00	\$16.31	\$36.46	\$16.31	\$32.77	\$45.27	\$16.00	\$45.43	\$34.29	
14.W	/ire Unb Copper Loop/Short (< or = 18kft), incl Man SI & Fac Res	UCLIVIC	\$10.00	\$10.51	φ30.40	\$10.31	φ32.11	Φ <del>4</del> 5.27	\$10.00	φ45.45	φ34.29	
	tewide, per month*	UCL4S	TBD	NA	\$19.34	NA	NA	NA	NA	NA	NA	
Jia	Zone 1, per month	UCL4S	TBD	\$25.56	\$16.65	\$25.26	\$29.14	\$25.00	\$18.49	\$25.00	\$25.00	
++++	Zone 2, per month	UCL4S	TBD	\$30.53	\$19.22	\$23.00	\$26.62	\$27.50	\$29.75	\$27.50	\$27.50	
HH	Zone 3, per month	UCL4S	TBD	\$32.24	\$30.55	\$19.08	\$28.75	\$30.00	\$34.14	\$30.00	\$30.00	
<del>                                     </del>	Zone 4, per month	UCL4S	NA NA	Ψ32.24 NA	NA	NA	\$20.75 NA	\$35.00	934.14 NA	\$30.00 NA	NA	
<del>                                     </del>	NRC - 1st	UCL4S	TBD	\$438.27	\$353.80	\$440.35	\$434.49	\$400.00	\$400.00	\$400.00	\$400.00	
<del>                                     </del>	NRC - Add'l	UCL4S	TBD	\$299.68	\$162.61	\$300.41	\$294.74	\$300.00	\$300.00	\$300.00	\$300.00	
<del>                                     </del>	NRC - Disconnect Charge - 1st	UCL4S	TBD	\$161.19	\$156.25	\$171.58	\$174.43	\$150.00	\$150.00	\$150.00	\$150.00	
<del>                                     </del>	NRC - Disconnect Charge - 1st	UCL4S	TBD	\$39.76	\$41.96	\$41.90	\$42.60	\$40.00	\$40.00	\$40.00	\$40.00	
<del>                                     </del>	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
<del>                                     </del>	NRC - Service Order submitted Electronically, per LSR - NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	\$3.50 NA	\$0.42	\$3.50 NA	\$3.50 NA	φ3.50 NA	\$3.50 NA	φ3.50 NA	\$3.50 NA	\$3.50 NA	
HHH	NRC - Service Order submitted Electronically, per LSR - Disconnect	SOMAN	NA NA	\$0.42	NA NA	\$29.24	NA NA	NA NA	NA NA	NA NA	\$19.99	
HHH	NRC - Service Order submitted Manually, per LSR  NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA NA	\$3.84	NA NA	\$3.94	NA NA	NA NA	NA NA	NA NA	NA	
HH	NRC - Service Order Submitted Manually, per LSR, Disconnect  NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	φ3.0 <del>4</del> NA	NA NA	φ3.94 NA	\$18.14	\$25.52	\$26.94	\$47.00	NA NA	
HH	141/O - Informental Charge - Manual Service Order - 15t	SUIVIAIN	IBD	INA	INA	INA	φ16.14	Φ∠3.5∠	φ∠0.94	φ47.00	INA	
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	N/A	NA	\$8.06	611 24	\$12.76	\$25.52	NA	
HH	NRC - Incremental Charge - Manual Service Order - Add I	SOMAN	TBD	NA NA	NA NA	NA NA	\$8.06	\$11.34 \$16.06	\$12.76 NA	\$25.52 \$21.00	NA NA	
HHH	NRC - Incremental Charge - Manual Service Order - Disconnect  NRC - Incremental Charge - Manual Service Order - Disconnect - Add'L	SOMAN	עפו	INA	INA	NA NA	\$11.41	\$16.06	NA NA	\$21.00	NA NA	
	prince - incremental Charge - Manual Service Order - Disconnect - Add L	SUIVIAIN		L		INA	<b>Ф</b> 11.41	Φ10.00	INA	<b>⊅∠1.∪</b> U	INA	

In INFO - incremental Charger - Manual Colar Coordination - per toop				AND OTHER SE									
RC-Stewards, per month			UCLMC	TBD	\$16.31	\$36.46	\$16.31	\$32.77	\$45.27	\$16.00	\$45.43	\$34.29	
Zone 1, per mouth	4-W	ire Unb Copper Loop/Short (< or = 18kft), <u>without</u> Man SI & Fac Res											
Example   UCLAM   TBD   \$0.0.53   \$19.22   \$32.00   \$36.62   \$27.50   \$29.75   \$77.50   \$27.50   \$20.00   \$30	RC	- Statewide, per month	UCL4W	TBD	NA	\$19.34	NA	NA	NA	NA	NA	NA	
Tome 5, per mounth		Zone 1, per month	UCL4W	TBD	\$25.56	\$16.65	\$25.26	\$29.14	\$25.00	\$18.49	\$25.00	\$25.00	
Tome 5, per mounth		Zone 2, per month	UCL4W	TBD	\$30.53	\$19.22	\$23.00	\$26.62	\$27.50	\$29.75	\$27.50	\$27.50	
CLAY   NA		, ,											
NRC - 1st	++++												
NRC - Actor	+++												
NRC - Disconnect Charge - 1st	+++												
NRC - Desconnect Charge - Add   School   Security   S	$\square$												
NRC - Service Order submitted Electronically, per LSR   SOMEC   S3.50   S3.5													
NRC - Service Order submitted Electronically, per LSR - Disconnect   SOMEC   NA   S0.42   NA   NA   NA   NA   NA   NA   NA   N		NRC - Disconnect Charge - Add'l		TBD	\$19.58	\$41.96	\$20.64	\$20.98	\$40.00	\$40.00	\$40.00	\$40.00	
NRC - Service Order submitted Manually, per LSR   SCMAN   NA   \$21.56   NA   \$29.24   NA   NA   NA   NA   NA   NA   NA   N		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
NRC - Service Order submitted Manually, per LSR   SOMAN   NA   \$21.56   NA   \$29.24   NA   NA   NA   NA   NA   NA   NA   N		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA							
NRC - Service Order submitted Manually, per LSR, Disconnect		7.1											
NRC - Service Order submitted Manually, per LSR, Disconnect													
NRC - Service Order submitted Manually, per LSR, Disconnect		NRC - Service Order submitted Manually, per LSR	SOMAN	NΔ	¢21.56	NΑ	\$20.24	NΑ	NΙΔ	NΑ	NΙΔ	\$10.00	
NRC - Incremental Charge - Manual Service Order - 14st   SOMAN   TBD   NA   NA   NA   S18.14   \$25.52   \$28.94   \$47.00   NA   NA   NA   NA   NA   NA   S18.14   \$25.52   \$28.94   \$47.00   NA   NA   NA   NA   NA   NA   S18.14   \$16.06   NA   \$21.00   NA   NA   NA   NA   S18.14   \$16.06   NA   \$21.00   NA   NA   NA   NA   NA   S18.14   \$16.06   NA   \$21.00   NA   NA   NA   NA   NA   S18.14   \$16.06   NA   \$21.00   NA   NA   NA   NA   S18.14   \$16.06   NA   \$21.00   NA   NA   NA   NA   NA   S18.14   \$16.06   NA   \$21.00   NA   NA   NA   NA   NA   NA   S18.14   \$16.06   NA   \$21.00   NA   NA   NA   NA   NA   NA   NA	++++												
NRC - Incremental Charge - Manual Service Order - AddT   SOMAN   TBD NA NA NA \$8.66   \$11.34   \$12.76   \$25.52   NA   NRC - Incremental Charge - Manual Service Order - Disconnect   SOMAN   TBD NA NA NA \$11.41   \$16.06   NA \$21.00   NA   NA   NRC - Incremental Charge - Manual Service Order - Disconnect - AddT.   SOMAN   NA NA   \$11.41   \$16.06   NA   \$21.00   NA   NA   NRC - Incremental Charge - Manual Service Order - Disconnect - AddT.   SOMAN   NA NA   \$11.41   \$16.06   NA   \$21.00   NA   NA   NRC - Incremental Charge - Manual Service Order - Disconnect - AddT.   SOMAN   NA NA   \$11.41   \$16.06   NA   \$21.00   NA   NA   NA   NA   NA   NA   NA	+++				· ·		*						
NRC - Incremental Charge - Manual Service Order - Disconnect + Add L SOMAN NA NA NA NA S11.41 \$16.06 NA \$21.00 NA NA NRC - Incremental Charge - Manual Service Order - Disconnect + Add L SOMAN NA NA NA NA S11.41 \$16.06 NA \$21.00 NA NRC - Incremental Charge - Manual Order Coordination - per loop UCLMC TBD \$16.31 \$36.46 \$16.31 \$32.77 \$45.27 \$16.00 \$84.32 \$34.29 \$1.49 NRC - Incremental Charge - Manual Order Coordination - per loop UCLMC TBD \$16.31 \$36.46 \$16.31 \$32.77 \$45.27 \$16.00 \$84.32 \$34.29 \$1.49 NRC - Incremental Charge - Manual Order Coordination - per loop UCL4L TBD \$85.36 NA	HH												
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'L   SOMAN   NA   NA   S11.41   S16.08   NA   S21.00   NA	$\Box$												
NRC - Incremental Charge - Menual Order Coordination - per loop	Ш			TBD		NA							
A-Wire Unb Copper Loop/Long (>18kH), incl Man Svc ing & Fac Res   UCL4L   TBD	Ш	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'L	SOMAN		NA		NA	\$11.41	\$16.06	NA	\$21.00	NA	
Statewide, Per month		NRC - Incremental Charge - Manual Order Coordination - per loop	UCLMC	TBD	\$16.31	\$36.46	\$16.31	\$32.77	\$45.27	\$16.00	\$45.43	\$34.29	
Statewide, Per month	4-V	/ire Unb Copper Loop/Long (>18kft), incl Man Syc Ing & Fac Res											
Zone 1, per month			UCI 4I	TBD	NA	\$55.86	NA	NA	NA	NA	NA	NA	
Zone 2, per month													
Zone 3, per month	+++												
Zone 4, per month	++++	7.1											
NRC - 1st	++++												
NRC - Add	+++												
NRC - Disconnect Charge - 1st		NRC - 1st	UCL4L	IBD	\$380.29	\$397.06	\$381.63	\$381.09	\$400.00	\$400.00	\$400.00	\$400.00	
NRC - Disconnect Charge - 1st													
NRC - Disconnect Charge - 1st													
NRC - Disconnect Charge - Add1		NRC - Add'l	UCL4L	TBD	\$241.70	\$227.88	\$241.69	\$241.35	\$300.00	\$300.00	\$300.00	\$300.00	
NRC - Service Order submitted Electronically, per LSR   SOMEC   S3.50   S2.75   S3.50   S3.5		NRC - Disconnect Charge - 1st	UCL4L	TBD	\$161.19	\$156.25	\$171.58	\$174.43	\$150.00	\$150.00	\$150.00	\$150.00	
NRC - Service Order submitted Electronically, per LSR   SOMEC   S3.50   S2.75   S3.50   S3.5		NRC - Disconnect Charge - Add'l	UCL4L	TBD	\$39.76	\$41.96	\$41.90	\$42.60	\$40.00	\$40.00	\$40.00	\$40.00	
NRC - Setvice Order submitted Electronically, per LSR - Disconnect   SOMEC   NA   \$0.42   NA   NA   NA   NA   NA   NA   NA   N													
NRC - Service Order submitted Manually, per LSR   SOMAN   NA   \$21.56   NA   \$29.24   NA   NA   NA   NA   \$19.99		7.1											
NRC - Service Order submitted Manually, per LSR, Disconnect   SOMAN   NA   \$3.84   NA   \$3.94   NA   NA   NA   NA   NA   NA   NA   N	HH												
NRC - Incremental Charge - Manual Service Order - 1st   SOMAN   TBD   NA   NA   NA   \$18.14   \$25.52   \$26.94   \$47.00   NA   NRC - Incremental Charge - Manual Service Order - Add'l   SOMAN   TBD   NA   NA   NA   \$8.06   \$11.34   \$12.76   \$25.52   NA   NRC - Incremental Charge - Manual Service Order - Disconnect   SOMAN   TBD   NA   NA   NA   \$11.41   \$16.06   NA   \$21.00   NA   NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l   SOMAN   NA   NA   NA   \$11.41   \$16.06   NA   \$21.00   NA   NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l   SOMAN   NA   NA   \$11.41   \$16.06   NA   \$21.00   NA   NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l   SOMAN   NA   NA   \$11.41   \$16.06   NA   \$21.00   NA   NA   NRC - Incremental Charge - Manual Order Coordination - per loop   UCLMC   TBD   \$16.31   \$36.46   \$16.31   \$32.77   \$45.27   \$16.00   \$45.43   \$34.29   \$44.44	++++												
NRC - Incremental Charge - Manual Service Order - Add'I   SOMAN   TBD   NA   NA   NA   NA   \$8.06   \$11.34   \$12.76   \$25.52   NA   NRC - Incremental Charge - Manual Service Order - Disconnect   SOMAN   TBD   NA   NA   NA   NA   \$11.41   \$16.06   NA   \$21.00   NA   NA   NA   NA   NA   NA   NA	++++	211 .											
NRC - Incremental Charge - Manual Service Order - Disconnect	+++												
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'L   SOMAN   NA   S11.41   \$16.06   NA   \$21.00   NA     NRC - Incremental Charge - Manual Order Coordination - per loop   UCLMC   TBD   \$16.31   \$36.46   \$16.31   \$32.77   \$45.27   \$16.00   \$45.43   \$34.29     4-Wire Unb Copper Loop/Long (>18kft), without Man SI & Fac Res   UCL4O   TBD   NA   \$55.86   NA   NA   NA   NA   NA   NA   NA   N	$\Box$												
NRC - Incremental Charge - Manual Order Coordination - per loop	Ш			TBD		NA							
A-Wire Unb Copper Loop/Long (>18kft), without Man SI & Fac Res   UCL4O   TBD   NA   \$55.86   NA   NA   NA   NA   NA   NA   NA   N													
Statewide, Per month	$\Pi\Pi\Pi\Pi$	NRC - Incremental Charge - Manual Order Coordination - per loop	UCLMC	TBD	\$16.31	\$36.46	\$16.31	\$32.77	\$45.27	\$16.00	\$45.43	\$34.29	
Statewide, Per month	4-V	/ire Unb Copper Loop/Long (>18kft), without Man SI & Fac Res											
Zone 1, per month			UCL4O	TBD	NA	\$55.86	NA	NA	NA	NA	NA	NA	
Zone 2, per month	H					*							
Zone 2, per month													
Zone 2, per month	$  \cdot  $	Zone 1, per month	UCL4O	TBD	\$82.70	\$47.56	\$61.02	\$77.94	\$50.00	\$54.54	\$50.00	\$50.00	
Zone 3, per month	ПП	Zone 2, per month	UCL4O	TBD	\$119.02	\$54.92	\$55.74	\$115.84	\$60.00	\$90.93	\$60.00	\$60.00	
Zone 4, per month	HH	· ·											
NRC - 1st         UCL4O         TBD         \$247.44         \$397.06         \$248.79         \$248.44         \$261.00         \$261.00         \$261.00           NRC - Add'I         UCL4O         TBD         \$164.06         \$227.88         \$164.05         \$163.82         \$161.00         \$161.00         \$161.00           NRC - Disconnect Charge - 1st         UCL4O         TBD         \$114.30         \$156.25         \$122.17         \$124.19         \$150.00         \$150.00         \$150.00           NRC - Disconnect Charge - Add'I         UCL4O         TBD         \$19.58         \$41.96         \$20.64         \$20.98         \$40.00         \$40.00         \$40.00	HH												
NRC - Add'    UCL4O   TBD   \$164.06   \$227.88   \$164.05   \$163.82   \$161.00   \$161.0	++++												
NRC - Disconnect Charge - 1st         UCL4O         TBD         \$114.30         \$156.25         \$122.17         \$124.19         \$150.00         \$150.00         \$150.00           NRC - Disconnect Charge - Add'l         UCL4O         TBD         \$19.58         \$41.96         \$20.64         \$20.98         \$40.00         \$40.00         \$40.00	++++												
NRC - Disconnect Charge - Add'l UCL4O TBD \$19.58 \$41.96 \$20.64 \$20.98 \$40.00 \$40.00 \$40.00 \$40.00	+++												
	+++												
NRC - Service Order submitted Electronically, per LSR	Ш	· ·											
	1111	INRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$4.50	

				AND OTHER SE	RVICES								
		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA	
		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	NA	
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	NA	NA	\$18.14	\$25.52	\$26.94	\$47.00	NA	
		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	NA	NA	\$8.06	\$11.34	\$12.76	\$25.52	NA	
+++	-	NRC - Incremental Charge - Manual Service Order - Add 1	SOMAN	TBD	NA NA	NA NA	NA NA	\$11.41	\$16.06	NA	\$21.00	NA NA	
+++	-	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'L	SOMAN	160	NA NA	NA NA	NA NA	\$11.41	\$16.06	NA NA	\$21.00	NA NA	
+++	+	NRC - Incremental Charge - Manual Order Coordination - per loop	UCLMC	TBD	\$16.31	\$36.46	\$16.31	\$32.77	\$45.27	\$16.00	\$45.43	\$34.29	
++-		Local Loop	OCLIVIC	100	ψ10.51	ψ50.40	ψ10.51	ψ32.11	ψ45.21	\$10.00	ψ45.45	Ψ54.25	
H-1		DS3 Unbundled Local Loop - per mile	1L5ND	\$10.85	\$11.77	\$8.90	\$43.69	\$11.26	\$54.39	\$32.53	\$15.53	\$30.53	
+++	+	DS3 Unbundled Local Loop- per Facility Termination	UE3PX	\$419.65	\$404.58	\$390.34	\$436.95	\$439.59	\$427.81	\$387.01	\$421.60	\$400.21	
+++	+	NRC - Facility Termination - 1st	UE3PX	\$640.54	\$903.37	\$639.50	\$1,091.00	\$594.70	\$975.22	\$964.04	\$735.42	\$726.16	
+++	-	NRC - Facility Termination - 1st	UE3PX	\$426.82	\$528.05	\$426.40	\$661.23	\$394.70	\$549.17	\$542.73	\$519.31	\$411.64	
+++	-	NRC - Facility Termination - Add 1	UE3PX	\$121.72	\$221.46	\$122.31	NA	\$102.16	\$134.07	NA	NA	\$103.36	
+++		NRC - Facility Termination - Disconnect - 1st	UE3PX	\$121.72	\$154.90	\$119.14	NA NA	\$99.46	\$134.07	NA NA	NA NA	\$100.59	
++	-	NRC - Pacility Termination - Disconnect - Add t	SOMAN	\$118.54 NA	\$154.90	\$119.14 NA	\$19.99	\$99.46 NA	\$130.59 NA	NA NA	NA NA	\$19.99	
++		NRC - Manual Svc Order, per LSR  NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA NA	\$3.84	NA NA	\$19.99 NA	NA NA	NA NA	NA NA	NA NA	\$19.99 NA	
H	+	NRC - Manual Svc Order, per LSR disconnect NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$3.84 \$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
H	+	NRC - Electronic Svc Order, per LSR NRC - Electronic Svc Order, per LSR disconnect	SOMEC	\$3.50 NA	\$2.75	\$3.50 NA	\$3.50 NA	\$3.50 NA	\$3.50 NA	\$3.50 NA	\$3.50 NA	\$3.50 NA	
++	-		SOMAN	\$38.48	\$0.42 NA	\$37.55			\$68.62	\$29.76		NA NA	
++	-	NRC - Incremental ChargeManual Svc Order - 1st					\$50.25 \$50.25	\$34.92			\$54.26	NA NA	
++	-	NRC - Incremental ChargeManual Svc Order - Add'l	SOMAN SOMAN	\$38.48 \$19.03	NA NA	\$37.55 \$18.03		\$34.92	\$68.62	\$29.76	\$54.26 NA	NA NA	
++	-	NRC - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	SOMAN	\$19.03	NA NA	\$18.03	\$20.94 \$20.94	\$20.94 \$20.94	\$28.59 \$28.59	NA NA	NA NA	NA NA	
1	<u> </u>	NRC - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add'l	SUMAN	\$19.03	NA	\$18.03	\$20.94	\$20.94	\$28.59	INA	INA	INA	
		1 Local Loop	41 END	<b>640.05</b>	<b>644.77</b>	<b>#0.00</b>	£40.00	<b>#00.00</b>	ΦE 4.00	<b>#00.50</b>	<b>#45.50</b>	<b>#00.50</b>	
++	_	STS-1 Unbundled Local Loop - per mile	1L5ND	\$10.85	\$11.77	\$8.90	\$43.69	\$38.98	\$54.39	\$32.53	\$15.53	\$30.53	
++	-	STS-1 Unbundled Local Loop- per Facility Termination	UDLS1	\$419.65	\$446.09	\$421.59	\$436.95	\$497.08	\$427.81	\$387.01	\$431.32	\$400.21	
++	-	NRC - STS-1 - Facility Termination - 1st	UDLS1	\$640.54	\$903.37	\$639.50	\$1,091	\$709.14	\$975.22	\$964.04	\$735.42	\$726.16	
++	_	NRC - STS-1 - Facility Termination - Add'I	UDLS1	\$426.82	\$528.05	\$426.40	\$661.23 NA	\$402.63	\$549.17	\$542.73	\$519.31	\$411.64 \$103.36	
++	_	NRC - STS-1 - Facility Termination - Disconnect - 1st	UDLS1	\$121.72	\$221.46	\$122.31		\$102.16	\$134.07	NA	NA		
++	_	NRC - STS-1 - Facility Termination - Disconnect - Add'l	UDLS1	\$118.54	\$154.90	\$119.14	NA Outside	\$99.46	\$130.59	NA	NA	\$100.59	
+++		NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	\$19.99	NA	NA	NA	NA	\$19.99	
++		NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA no.50	\$3.84	NA no.50	NA On Fo	NA no.50	NA Oo 50	NA no.50	NA To 50	NA 00.50	
++	_	NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
++		NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA Doc 10	\$0.42	NA DOZ.55	NA Too 10	NA DEC. OF	NA Tagana	NA Too To	NA DELLOS	NA	
++		NRC - STS-1 - Incremental ChargeManual Svc Order - 1st	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$29.76	\$54.26	NA	
H	+	NRC - STS-1 - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$29.76	\$54.26	NA	
++	+	NRC - STS-1 - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	SOMAN	\$19.03	NA NA	\$18.03	NA	\$20.94	\$28.59	NA	NA	NA	
H-		NRC - STS-1 - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add'l	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	NA	NA	NA	
110	)C3-	Local Loop	41 CND	<b>#0.00</b>	<b>#0.00</b>	ФС 7F	<b>#22.45</b>	\$20 FB	£44.07	<b>#04.00</b>	¢44.70	f00.40	
++		Local Loop - OC3 - per Mile	1L5ND	\$8.23	\$8.93	\$6.75	\$33.15	\$29.58	\$41.27	\$24.69	\$11.78	\$23.16	
++		Local Loop - OC3 - per Facility Termination		\$691.33	\$648.60	\$630.21	\$713.29	\$753.65	\$689.68	\$611.36	\$701.71	\$620.20	
H	+	NRC - OC3 - Facility Termination - 1st		\$949.63	\$966.45	\$947.69	\$1,543	\$1,025	\$1,427	\$1,411	\$1,044	\$1,050	
H	+	NRC - OC3 - Facility Termination - Add'l		\$413.38	\$408.85	\$413.00	\$661.23	\$402.63	\$549.17	\$542.73	\$505.88	\$411.64	
H	+	NRC - OC3 - Facility Termination - Disconnect - 1st		\$121.72	\$11.56	\$122.31	NA	\$102.16	\$134.07	\$131.65	NA	\$103.36	
+++		NRC - OC3 - Facility Termination - Disconnect - Add'l	001411	\$118.54	\$108.34	\$119.14	NA O 10.00	\$99.46	\$130.59	\$128.19	NA	\$100.59	
$\mathbf{H}$	_	NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA NA	\$19.99	NA	NA	NA	NA	\$19.99	
$\mathbf{H}$	_	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA no.50	\$3.84	NA no.50	NA no so	NA no. 50	NA Oo 50	NA no.50	NA To 50	NA DO 50	
H	_	NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
$\mathbf{H}$	_	NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA Doc 10	\$0.42	NA DOT 55	NA Tagada	NA 050.05	NA Tag ag	NA Tagana 4	NA DELLOS	NA	
$\mathbf{H}$	_	NRC - OC3 - Incremental ChargeManual Svc Order - 1st	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$69.34	\$54.26	NA	
$\mathbf{H}$	_	NRC - OC3 - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$69.34	\$54.26	NA	
$\mathbf{H}$	_	NRC - OC3 -Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	\$29.76	NA	NA	
		NRC - OC3 -Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add'l	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	\$29.76	NA	NA	

	AND OTHER SERVICES												
0	C -1:	2 Local Loop											
		Local Loop - OC12 - per Mile	1L5ND	\$10.13	\$10.99	\$8.31	\$40.80	\$36.40	\$50.79	\$30.38	\$14.50	\$28.51	
		Local Loop - OC12 - per Facility Termination		\$2,557	\$2,053.06	\$2,109.00	\$2,457	\$2,571	\$2,371	\$2,122	\$2,663	\$2,079	
		NRC - OC12 - Facility Termination - 1st		\$1,165	\$1,183.46	\$1,162.00	\$1,858	\$1,245	\$1,742	\$1,722	\$1,259	\$1,276	
		NRC - OC12 - Facility Termination - Add'l		\$413.38	\$408.85	\$413.00	\$661.23	\$402.63	\$549.17	\$542.73	\$505.88	\$411.64	
	_	NRC - OC12 - Facility Termination - Disconnect - 1st		\$121.72	\$111.56	\$122.31	NA	\$102.16	\$134.07	\$131.65	NA	\$103.36	
$\Box$		NRC - OC12 - Facility Termination - Disconnect - Add'l		\$118.54	\$108.34	\$119.14	NA	\$99.46	\$130.59	\$128.19	NA	\$100.59	
++-		NRC - Manual Svc Order, per LSR	SOMAN	NA NA	\$21.56	NA NA	\$19.99	NA NA	NA	NA NA	NA	\$19.99	
H		NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA NA	NA NA	NA NA	NA	NA	NA	NA NA	
+++		NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
++-		NRC - Electronic Svc Order, per LSR NRC - Electronic Svc Order, per LSR disconnect	SOMEC	\$3.50 NA	\$0.42	\$3.50 NA	\$3.50 NA	φ3.50 NA	\$3.50 NA	\$3.50 NA	\$3.50 NA	\$3.50 NA	
+++		NRC - C12 - Incremental Charge - Manual Svc Order - 1st	SOMAN	\$38.48	Φ0.42 NA	\$37.55	\$93.12	\$50.25	\$68.62	\$68.62	\$54.26	NA NA	
+++												NA NA	
++-		NRC - OC12 - Incremental Charge - Manual Svc Order - Add'l	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$68.62	\$54.26		
$\Box$		NRC - OC12 - Incremental Cost-Manual Svc. Order vs. Elect-Disconnect-1st	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	\$28.59	NA	NA	
	_	NRC - OC12 - Incremental Cost-Manual Svc. Order vs. Elect-Disconnect-Add'l	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	\$28.59	NA	NA	
100		8 Local Loop		4									
$\Box$		Local Loop - OC48 - per Mile	1L5ND	\$33.22	\$36.04	\$27.25	\$133.84	\$119.40	\$166.59	\$120.02	\$47.57	\$93.50	
		Local Loop - OC48 - per Facility Termination		\$1,713	\$1,685.97	\$1,598.00	\$2,129	\$2,268	\$1,753	\$1,677	\$1,733	\$1,832	
$\coprod$		Local Loop - OC12 interface on OC48 Facility		\$736.71	\$587.71	\$594.80	\$725.77	\$723.29	\$667.00	\$582.66	\$773.40	\$570.54	
		NRC - OC48 - Facility Termination - 1st		\$1,165	\$1,183.46	\$1,162.00	\$1,858	\$1,245	\$1,742	\$1,722	\$1,259	\$1,276	
		NRC - OC48 - Facility Termination - Add'l		\$413.38	\$408.85	\$413.00	\$661.23	\$402.63	\$549.17	\$542.73	\$505.88	\$411.64	
		NRC - OC48 - Interface OC12 on OC48 - 1st		\$121.72	\$543.72	\$539.36	\$844.21	\$532.13	\$729.04	\$720.81	\$635.04	\$544.55	
		NRC - OC48 - Interface OC12 on OC48 - Add'l		\$118.54	\$312.05	\$317.38	\$516.89	\$304.90	\$404.94	\$400.38	\$410.02	\$311.39	
		NRC - OC48 - Facility Termination - Disconnect - 1st		\$121.72	\$111.56	\$122.31	NA	\$102.16	\$134.07	\$131.65	NA	\$103.36	
	_	NRC - OC48 - Facility Termination - Disconnect - Add'l		\$118.54	\$108.34	\$119.14	NA	\$99.46	\$130.59	\$128.19	NA	\$100.59	
		NRC - OC48- Interface OC12 on OC48 - Disconnect - 1st		\$121.72	\$111.56	\$122.31	NA	\$102.16	\$134.07	\$131.65	NA	\$103.36	
		NRC - OC48 - Interface OC12 on OC48 - Disconnect - Add'l		\$118.54	\$108.34	\$119.14	NA	\$99.46	\$130.59	\$128.19	NA	\$100.59	
H		NRC - Manual Svc Order, per LSR	SOMAN	NA NA	\$21.56	NA	\$19.99	NA NA	NA	NA NA	NA	\$19.99	
++-		NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA.	\$3.84	NA NA	NA	NA NA	NA NA	NA	NA	NA NA	
++-		NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
++-		NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA	
+++			SOMAN	\$19.03	NA	\$18.03	\$93.12	\$20.94	\$28.59	NA NA	NA NA	NA NA	
+++	-	NRC - OC48 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-1s NRC - OC48 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-Ad	SOMAN	\$19.03	NA NA	\$18.03	\$93.12	\$20.94 \$20.94	\$28.59	NA NA	NA NA	NA NA	
+++	-				NA NA					NA NA	NA NA	NA NA	
++-	_	NRC - OC48 - Interface - Manual Svc Order vs Electronic-Disconnect-1st	SOMAN	\$19.03		\$18.03	NA	\$20.94	\$28.59				
$\Box$		NRC - OC48 - Interface - Manual Svc Order vs Electronic-Disconnect-Add'l	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	NA	NA	NA	
$\Box$		NRC - OC-48 - Incremental ChargeManual Svc Order-1st	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$69.34	\$54.26	NA	
Ш		NRC - OC-48 - Incremental ChargeManual Svc Order-Add'l	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$69.34	\$54.26	NA	
Ш	Ш	NRC - OC48 - Interface OC12 on OC48 - Incremental ChargeManual Svc Order	SOMAN	\$38.48	NA	\$37.55	NA	\$50.25	\$68.62	\$69.34	NA	NA	
$\coprod$	Ш	NRC - OC48 - Interface OC12 on OC48 - Incremental ChargeManual Svc Order	SOMAN	\$38.48	NA	\$37.55	NA	\$50.25	\$68.62	\$69.34	NA	NA	
Uı	nbui	ndled Loop Modification/Conditioning											
		NRC - Load Coil/Equipment Removal per 2 Wire pair - Loops less than or equal											
Ш		to 18kft **	ULM2L	\$65.23	\$65.40	\$69.28	\$65.40	\$65.30	\$80.55	\$65.49	\$65.20	\$65.40	
		NRC - Load Coil/Equipment Removal per 2 Wire pair - Loops greater than 18kft -											
		1st **	ULM2G	\$716.70	\$710.71	\$757.04	\$710.73	\$709.71	\$880.00	\$719.55	\$716.32	\$710.71	
		NRC - Load Coil/Equipment Removal per 2 Wire pair - Loops greater than 18kft -										İ	
		Add'l **	ULM2G	\$23.55	\$23.77	\$23.49	\$23.77	\$23.74	\$27.30	\$23.65	\$23.54	\$23.77	
		NRC - Load Coil/Equipment Removal per 4 Wire pair - Loops less than or equal											
		to 18kft **	ULM4L	\$65.23	\$65.40	\$69.28	\$65.40	\$65.30	\$80.55	\$65.49	\$65.20	\$65.40	
$\Box$		NRC - Load Coil/Equipment Removal per 4 Wire pair - Loops greater than 18kft -											
		1st **	ULM4G	\$716.70	\$710.71	\$757.04	\$710.73	\$709.71	\$880.00	\$719.55	\$716.32	\$710.71	
H	$\Box$	NRC - Load Coil/Equipment Removal per 4 Wire pair - Loops greater than 18kft -				Ž. Z	J	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	,,,,,,,,	2	J		
		Add'l **	ULM4G	\$23.55	\$23.77	\$23.49	\$23.77	\$23.74	\$27.30	\$23.65	\$23.54	\$23.77	
H	H	NRC - Bridge Tap Removal per pair unloaded **	ULMBT	\$65.28	\$65.44	\$79.99	\$65.44	\$65.35	\$121.14	\$65.64	\$65.24	\$65.44	
+++	H		OLIVID I	ψ00.20	ψου. 1 1	ψ, σ.σσ	ψου. 1 1	Ψ00.00	Ψ121.11	Ψ00.01	ψ00. <b>Σ</b> 1	ψου. 11	
டப													

			AND OTHER SE	RVICES								
	BUNDLED SUB-LOOPS											
SU	B-LOOP DISTRIBUTION											
	Cross-Box Set-Up											
	NRC - Set-Up per Cross Box location in the field - CLEC Feeder Facility set-up	USBSA	\$517.43	\$711.78	\$421.08	\$627.16	\$639.68	TBD	\$498.09	\$510.15	TBD	
	NRC - Set-Up per Cross Box location in the field - per 25 pair panel set-up	USBSB	\$44.87	\$45.28	\$67.10	\$45.28	\$45.22	TBD	\$45.04	\$44.84	TBD	
	NRC - Set-Up per Building Equipment Room - CLEC Feeder Facility set-up	USBSC	\$311.77	\$333.44	\$394.74	\$407.02	\$289.90	TBD	\$313.01	\$311.60	\$314.09	
htt	NRC - Set-Up per Building Equipment Room - per 25 pair panel set-up	USBSD	\$107.63	\$109.85	\$154.57	\$111.55	\$104.26	TBD	\$108.06	\$108.17	\$119.07	
Loc	pp Distribution per 2-Wire Analog VG Sub-Loop, per month	USBN2	\$9.05	NA	\$9.12	NA NA	NA NA	TBD	NA	\$8.34	TBD	
<del>   </del>	Zone 1, per month	USBN2	TBD	\$9.36	TBD	\$9.03	\$10.33	TBD	\$7.99	NA	TBD	
+++	Zone 2, per month	USBN2	TBD	\$12.49	TBD	\$12.25	\$14.43	TBD	\$12.63	NA NA	TBD	
++++	Zone 3, per month	USBN2	TBD	\$16.13	TBD	\$16.71	\$21.11	TBD	\$14.43	NA NA	TBD	
++++	7.1	USBN2	NA	NA	NA NA	NA	NA	TBD	Φ14.43 NA	NA NA	NA NA	
-H-H	Zone 4, per month											
$\sqcup \sqcup$	NRC - 1st	USBN2	\$125.53	\$139.20	\$207.01	\$139.19	\$138.99	TBD	\$126.03	\$195.98	TBD	
$\perp \downarrow \downarrow \downarrow$	NRC - Add'l	USBN2	\$54.32	\$61.94	\$171.32	\$61.93	\$61.84	TBD	\$54.54	\$63.70	TBD	
шш	NRC - Disconnect Charge - 1st	USBN2	\$92.45	\$98.49	TBD	\$101.18	\$100.70	TBD	\$71.13	NA	TBD	
+++	NRC - Disconnect Charge - Add'I	USBN2	\$13.21	\$13.08	TBD	\$13.44	\$13.37	TBD	\$10.16	NA	TBD	
Ш	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD	
	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA	TBD	
	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	TBD	NA	NA	TBD	
	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	TBD	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	\$18.94	NA	\$18.14	TBD	\$26.94	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	\$8.42	NA	\$8.06	TBD	\$12.76	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	TBD	NA	\$11.41	TBD	\$15.12	NA	TBD	
	NRC - Incremental Charge - Manual Order Coordination - per loop	USBMC	\$51.29	\$16.31	\$34.22	\$16.31	\$16.29	TBD	\$45.34	\$45.43	TBD	
100	pp Distribution per 4-Wire Analog VG Sub-Loop, per month	USBN4	\$10.56	NA NA	\$8.32	NA	NA	TBD	NA NA	\$10.22	NA NA	
H	Zone 1, per month	USBN4	TBD	\$10.12	TBD	\$10.18	\$15.67	TBD	\$9.23	NA	TBD	
	Zone 2, per month	USBN4	TBD	\$18.29	TBD	\$9.44	\$20.35	TBD	\$14.63	NA	TBD	
+++	Zone 3, per month	USBN4	TBD	\$26.09	TBD	\$13.38	\$20.33	TBD	\$16.73	NA NA	TBD	
+		USBN4	NA	\$26.09 NA	NA NA	\$13.36 NA	\$24.93 NA	TBD	NA	NA NA	TBD	
	Zone 4, per month										TBD	
+++	NRC - 1st	USBN4	\$155.90	\$165.68	\$219.35	\$165.67	\$165.43	TBD	\$156.52	\$232.76		
+	NRC - Add'l	USBN4	\$79.35	\$88.42	\$72.99	\$88.41	\$88.29	TBD	\$79.66	\$91.92	TBD	
$\bot$	NRC - Disconnect Charge - 1st	USBN4	\$107.24	\$104.31	\$123.72	\$109.94	\$111.76	TBD	\$78.56	NA	TBD	
	NRC - Disconnect Charge - Add'l	USBN4	\$18.46	\$17.15	\$28.77	\$18.08	\$18.38	TBD	\$13.53	NA	TBD	
	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50		
$\perp \downarrow \perp \perp$	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA		
	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	TBD	NA	NA		
	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	TBD	NA	NA		
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	NA	NA	\$18.14	TBD	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	NA	NA	\$8.06	TBD	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	NA	NA	\$11.41	TBD	NA	TBD	TBD	
111	NRC - Incremental Charge - Manual Order Coordination - per loop	USBMC	\$51.29	\$16.31	\$34.22	\$16.31	\$16.29	TBD	\$45.34	\$45.43	TBD	
Loc	pp Distribution per 2 Wire Unbundled Copper Sub-Loop, per month	UCS2X	\$6.55	NA NA	TBD	NA NA	NA	TBD	NA NA	\$6.82	NA NA	
<del>    -~</del>	Zone 1, per month	UCS2X	TBD	\$7.91	TBD	\$8.01	\$8.72	TBD	\$7.33	NA	TBD	
+++	Zone 2, per month	UCS2X	TBD	\$10.37	TBD	\$9.18	\$11.43	TBD	\$10.95	NA NA	TBD	
+++	Zone 3, per month	UCS2X	TBD	\$12.76	TBD	\$11.02	\$14.08	TBD	\$12.36	NA NA	TBD	
+++	Zone 4, per month	UCS2X	NA NA	NA	NA	NA	NA	TBD	912.30 NA	NA NA	TBD	
+++	NRC - 1st	UCS2X	\$136.55	\$139.20	TBD	\$139.19	\$138.99	TBD	\$137.10	\$212.46	TBD	
+++	NRC - 1st NRC - Add'l										TBD	
+++		UCS2X	\$60.00	\$61.94	TBD	\$61.93	\$61.84	TBD	\$60.24	\$70.03		
+	NRC - Disconnect Charge - 1st	UCS2X	\$99.54	\$98.49	TBD	\$101.18	\$100.70	TBD	\$76.58	NA	TBD	
$\bot \bot \bot$	NRC - Disconnect Charge - Add'I	UCS2X	\$14.05	\$13.08	TBD	\$13.44	\$13.37	TBD	\$10.81	NA	TBD	
$\bot \bot \bot$	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD	
$\perp \! \! \perp \! \! \! \perp$	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA	TBD	
	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	TBD	NA	NA	TBD	
	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	TBD	NA	NA	TBD	

				AND OTHER SE	RVICES							
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	TBD	NA	\$18.14	TBD	NA	NA	TBD
		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	TBD	NA	\$8.06	TBD	NA	NA	TBD
		NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	TBD	NA	\$11.41	TBD	NA	NA	TBD
		NRC - Incremental Charge - Manual Order Coordination - per loop	USBMC	\$51.29	\$16.31	TBD	\$16.31	\$16.29	TBD	\$45.34	\$45.43	TBD
Lo	ор	Distribution per 4 Wire Unbundled Copper Sub-Loop, per month	UCS4X	\$8.06	NA	TBD	NA	NA	TBD	NA	\$8.86	NA
H		Zone 1, per month	UCS4X	TBD	\$7.11	TBD	\$10.65	\$10.48	TBD	\$7.14	NA	TBD
		Zone 2, per month	UCS4X	TBD	\$11.26	TBD	\$9.71	\$13.52	TBD	\$11.09	NA	TBD
++	_	Zone 3, per month	UCS4X	TBD	\$16.92	TBD	\$8.45	\$17.66	TBD	\$12.63	NA NA	TBD
		Zone 4, per month	UCS4X	NA	NA	NA	NA	NA	TBD	NA	NA	TBD
HH		NRC - 1st	UCS4X	\$161.59	\$165.68	TBD	\$165.67	\$165.43	TBD	\$162.24	\$238.46	TBD
++-		NRC - Add'l	UCS4X	\$85.04	\$88.42	TBD	\$88.41	\$88.29	TBD	\$85.38	\$97.61	TBD
++-		NRC - Disconnect Charge - 1st	UCS4X	\$107.24	\$104.31	TBD	\$109.94	\$111.76	TBD	\$78.56	NA NA	TBD
++-	_	NRC - Disconnect Charge - Add'l	UCS4X	\$18.46	\$17.15	TBD	\$18.08	\$18.38	TBD	\$13.53	NA NA	TBD
++-		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD
++-		NRC - Service Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	\$3.50 NA	NA	TBD	NA	NA	TBD
++-		NRC - Service Order submitted Manually, per LSR	SOMAN	NA NA	\$21.56	NA NA	\$29.24	NA NA	TBD	NA NA	NA NA	TBD
HH	_	NRC - Service Order submitted Manually, per LSR  NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA NA	\$3.84	NA NA	\$3.94	NA NA	TBD	NA NA	NA NA	TBD
HH		NRC - Service Order submitted Manually, per LSR, Disconnect NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	\$3.84 NA	TBD	\$3.94 NA	NA \$18.14	TBD	NA NA	NA NA	TBD
++-		<u> </u>	SOMAN	TBD	NA NA	TBD	NA NA	\$18.14 \$8.06	TBD	NA NA	NA NA	TBD
++-		NRC - Incremental Charge - Manual Service Order - Add'l										
$\vdash$		NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA O 10 O 1	TBD	NA O10.01	\$11.41	TBD	NA 0.45.04	NA 0.45,40	TBD
1		NRC - Incremental Charge - Manual Order Coordination - per loop	USBMC	\$51.29	\$16.31	TBD	\$16.31	\$16.29	TBD	\$45.34	\$45.43	TBD
Su		oop-Intrabldg Ntwk Cable (aka riser cable), 2W analog, per mo	USBR2	\$1.33	\$3.87	\$1.61	\$3.23	\$1.59	TBD	\$3.50	\$1.60	\$1.47
$\vdash$		NRC - 1st	USBR2	\$113.60	\$113.62	\$137.03	\$113.61	\$106.11	TBD	\$114.05	\$189.52	\$150.35
		NRC - Add'l	USBR2	\$50.44	\$36.36	\$41.59	\$36.35	\$35.12	TBD	\$37.20	\$47.09	\$45.63
$\Box$		NRC - Disconnect Charge - 1st	USBR2	\$99.54	\$98.49	\$115.85	\$101.18	\$93.19	TBD	\$76.58	NA	\$128.85
$\Box$		NRC - Disconnect Charge - Add'l	USBR2	\$14.05	\$13.08	\$19.17	\$13.44	\$13.92	TBD	\$10.81	NA	\$21.32
		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	\$3.50
		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA	TBD
		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	TBD	NA	NA	\$19.99
		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	TBD	NA	NA	TBD
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	NA	NA	\$18.14	TBD	NA	NA	NA
		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	NA	NA	\$8.06	TBD	NA	NA	NA
		NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	NA	NA	\$11.41	TBD	NA	NA	NA
		NRC - Incremental Charge - Manual Order Coordination - per loop	USBMC	\$51.29	\$16.31	\$34.22	\$16.31	\$16.29	TBD	\$45.34	\$45.43	\$55.00
Su	ıb-L	oop-Intrabidg Ntwk Cable (aka riser cable), 4W analog, per mo	USBR4	\$2.17	\$7.20	\$2.96	\$6.29	\$2.83	TBD	\$3.75	\$2.78	\$2.55
		NRC - 1st	USBR4	\$127.17	\$126.10	\$176.46	\$126.10	\$118.69	TBD	\$127.67	\$204.20	\$193.62
		NRC - Add'l	USBR4	\$50.62	\$48.84	\$55.11	\$48.84	\$47.70	TBD	\$50.82	\$60.40	\$60.47
		NRC - Disconnect Charge - 1st	USBR4	\$107.45	\$104.31	\$122.17	\$109.94	\$101.65	TBD	\$78.71	NA	\$135.88
		NRC - Disconnect Charge - Add'l	USBR4	\$14.59	\$17.15	\$19.57	\$18.08	\$14.61	TBD	\$10.69	NA	\$21.76
		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	\$3.50
		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA	TBD
		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	TBD	NA	NA	\$19.99
		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	TBD	NA	NA	TBD
$\Box$		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	NA	NA	\$18.14	TBD	NA	NA	NA
		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	NA	NA	\$8.06	TBD	NA	NA	NA
$\Box$		NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	NA	NA	\$11.41	TBD	NA	NA	NA
		NRC - Incremental Charge - Manual Order Coordination - per loop	USBMC	\$51.29	\$16.31	\$34.22	\$16.31	\$16.29	TBD	\$45.34	\$45.43	\$55.00
HH												****
SU	JB-I	OOP FEEDER				1						
		ess-Box Set-Up										
	1											
		NRC - DS0 Set-Up per Cross Box location - CLEC Distribution Facility set-up	USBFW	517.43	711.78	\$421.08	627.16	\$639.68	TBD	498.09	510.15	TBD
++-		NRC - DS0 Set-Up per Cross Box location - GEEC Distribution 1 admity set-up	USBFX	44.87	45.28	\$67.10	45.28	\$45.22	TBD	45.04	44.84	TBD
HH		200 000 op por oroug box roughlor por 20 pair parior out up	OOD! A	77.07	70.20	ψ07.10	70.20	Ψ-τυ.ΔΔ	100	70.07	77.07	100
		NRC - DS1 Set-Up per Cross Box location - CLEC Distribution Facility set-up	USBFY	517.43	711.78	\$421.08	627.16	\$639.68	TBD	498.09	510.15	TBD
шш		Title Do Tot op per cross box location. OLLO Distribution 1 actiffy set-up	ו ומפט	J17. <del>4</del> J	711.70	ψ+21.00	027.10	ψυσσ.υυ	טטו	+30.03	310.13	טטו

	AND OTHER SERVICES  NRC - DS1 Set-Up per Cross Box location - per pair panel set-up USBFZ 44.87 45.28 \$67.10 45.28 \$45.22 TBD 45.04 44.84 TBD													
		NRC - DS1 Set-Up per Cross Box location - per pair panel set-up	USBFZ	44.87	45.28	\$67.10	45.28	\$45.22	TBD	45.04	44.84	TBD		
	2-Wire	Analog VG Ground-Start Unbundled Sub-Loop Feeder, per month	USBFA	\$13.68	NA	\$8.58	NA	NA	TBD	NA	\$10.06	NA		
		Zone 1, per month	USBFA	TBD	\$10.75	TBD	\$10.36	\$11.01	TBD	\$8.92	NA	TBD		
		Zone 2, per month	USBFA	TBD	\$11.57	TBD	\$13.62	\$13.36	TBD	\$14.10	NA	TBD		
		Zone 3, per month	USBFA	TBD	\$13.51	TBD	\$19.69	\$21.56	TBD	\$16.11	NA	TBD		
++		Zone 4, per month	USBFA	NA NA	NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA	TBD		
+++									TBD			TBD		
$\vdash$		NRC - 1st	USBFA	\$198.42	\$193.62	\$206.44	\$192.57	\$192.30		\$122.52	\$199.27			
		NRC - Add'l	USBFA	\$116.66	\$113.00	\$170.05	\$111.96	\$111.80	TBD	\$46.61	\$60.33	TBD		
		NRC - Disconnect Charge - 1st	USBFA	\$119.95	\$116.59	TBD	\$119.14	\$118.57	TBD	\$77.90	NA	TBD		
		NRC - Disconnect Charge - Add'l	USBFA	\$27.04	\$26.70	TBD	\$26.79	\$26.66	TBD	\$14.68	NA	TBD		
		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD		
		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA	TBD		
		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	TBD	NA	NA	TBD		
		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	TBD	NA	NA	TBD		
H	1	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	\$18.94	NA	\$18.14	TBD	\$26.94	NA	TBD		
HH	_	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	\$8.42	NA	\$8.06	TBD	\$12.76	NA	TBD		
H		NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA NA	TBD	NA NA	\$11.41	TBD	\$15.12	NA NA	TBD		
H		1410 moremental Charge - Manual Gervice Older - Disconnect	SOIVIAIN	טטו	IVA	טטו	INA	φ11. <del>4</del> 1	טטו	ψ13.12	IVA	טטו		
1	) \A/:	Angley VC Lean Ctart Hubundlad Cub Lean Fooder non-	HCDED	£12 C0	NA	¢0 50	NA	NA	TBD	NA	\$10.06	NA		
H	2-VVITE	e Analog VG Loop-Start Unbundled Sub-Loop Feeder, per month	USBFB	\$13.68		\$8.58								
Ш		Zone 1, per month	USBFB	TBD	\$10.75	TBD	\$10.36	\$11.01	TBD	\$8.92	NA	TBD		
		Zone 2, per month	USBFB	TBD	\$11.57	TBD	\$13.62	\$13.36	TBD	\$14.10	NA	TBD		
		Zone 3, per month	USBFB	TBD	\$13.51	TBD	\$19.69	\$21.56	TBD	\$16.11	NA	TBD		
		Zone 4, per month	USBFB	NA	NA	NA	NA	NA	TBD	NA	NA	TBD		
		NRC - 1st	USBFB	\$198.42	\$193.62	\$206.44	\$192.57	\$192.30	TBD	\$122.52	\$199.27	TBD		
		NRC - Add'I	USBFB	\$116.66	\$113.00	\$170.05	\$111.96	\$111.80	TBD	\$46.61	\$60.33	TBD		
		NRC - Disconnect Charge - 1st	USBFB	\$119.95	\$116.59	TBD	\$119.14	\$118.57	TBD	\$77.90	NA	TBD		
		NRC - Disconnect Charge - Add'l	USBFB	\$27.04	\$26.70	TBD	\$26.79	\$26.66	TBD	\$14.68	NA	TBD		
		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD		
H	_	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA	TBD		
+++		NRC - Service Order submitted Manually, per LSR	SOMAN	NA NA	\$21.56	NA	\$29.24	NA NA	TBD	NA NA	NA NA	TBD		
+++		NRC - Service Order submitted Manually, per LSR, Disconnect				NA NA			TBD			TBD		
+++			SOMAN	NA TDD	\$3.84		\$3.94	NA ***		NA COO.O.4	NA			
$\vdash$	_	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	\$18.94	NA	\$18.14	TBD	\$26.94	NA	TBD		
$\perp \perp \downarrow$	_	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	\$8.42	NA	\$8.06	TBD	\$12.76	NA	TBD		
		NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	TBD	NA	\$11.41	TBD	\$15.12	NA	TBD		
Ш														
	2-Wire	e Analog VG Reverse Battery Unb Sub-Loop Feeder, per mo	USBFC	\$13.68	NA	\$8.58	NA	NA	TBD	NA	\$10.06	NA		
		Zone 1, per month	USBFC	TBD	\$10.75	TBD	\$10.36	\$11.01	TBD	\$8.92	NA	TBD		
		Zone 2, per month	USBFC	TBD	\$11.57	TBD	\$13.62	\$13.36	TBD	\$14.10	NA	TBD		
		Zone 3, per month	USBFC	TBD	\$13.51	TBD	\$19.69	\$21.56	TBD	\$16.11	NA	TBD		
		Zone 4, per month	USBFC	NA	NA	NA	NA	NA	TBD	NA	NA	NA		
Ш		NRC - 1st	USBFC	\$198.42	\$193.62	\$206.44	\$192.57	\$192.30	TBD	\$122.52	\$199.27	TBD		
H	1	NRC - Add'I	USBFC	\$116.66	\$113.00	\$170.05	\$111.96	\$111.80	TBD	\$46.61	\$60.33	TBD		
H	_	NRC - Disconnect Charge - 1st	USBFC	\$119.95	\$116.59	TBD	\$119.14	\$118.57	TBD	\$77.90	NA	TBD		
H		NRC - Disconnect Charge - Add'l	USBFC	\$27.04	\$26.70	TBD	\$26.79	\$26.66	TBD	\$14.68	NA	TBD		
H		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD		
H		NRC - Service Order submitted Electronically, per LSR  NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	φ3.50 NA	\$0.42	\$3.50 NA	\$3.50 NA	φ3.50 NA	TBD	\$3.50 NA	\$3.50 NA	TBD		
H		NRC - Service Order submitted Electronically, per LSR - Disconnect	SOMAN	NA NA	\$21.56	NA NA	\$29.24	NA NA	TBD	NA NA	NA NA	TBD		
H												TBD		
H	_	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA D10.01	\$3.94	NA	TBD	NA ************************************	NA			
Ш		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	\$18.94	NA	\$18.14	TBD	\$26.94	NA	TBD		
Ш	_	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	\$8.42	NA	\$8.06	TBD	\$12.76	NA	TBD		
Ш		NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	TBD	NA	\$11.41	TBD	\$15.12	NA	TBD		
Ш														
		Analog VG Ground-Start Unbundled Sub-Loop Feeder, per month	USBFD	\$24.94	NA	\$19.91	NA	NA	NA	NA	\$10.22	NA		
		Zone 1, per month	USBFD	TBD	\$23.35	TBD	\$30.69	\$25.90	TBD	\$21.91	NA	TBD		

	· · · · · · · · · · · · · · · · · · ·		AND OTHER SE						•			
$\perp \downarrow \perp$	Zone 2, per month	USBFD	TBD	\$27.94	TBD	\$36.12	\$27.00	TBD	\$35.92	NA	TBD	
	Zone 3, per month	USBFD	TBD	\$40.51	TBD	\$22.90	\$25.61	TBD	\$41.37	NA	TBD	L
	Zone 4, per month	USBFD	NA	NA	NA	NA	NA	TBD	NA	NA	NA	
	NRC - 1st	USBFD	\$224.21	\$222.74	\$243.41	\$221.19	\$220.87	TBD	\$226.36	\$232.76	TBD	
	NRC - Add'I	USBFD	\$142.45	\$140.22	\$81.32	\$138.67	\$138.47	TBD	\$144.28	\$91.92	TBD	
	NRC - Disconnect Charge - 1st	USBFD	\$131.15	\$127.64	\$134.77	\$133.53	\$135.74	TBD	\$96.50	NA	TBD	
	NRC - Disconnect Charge - Add'l	USBFD	\$34.66	\$32.91	\$33.93	\$33.69	\$34.25	TBD	\$25.82	NA	TBD	
	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD	
	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA	TBD	
	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	TBD	NA	NA	TBD	
	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	TBD	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	NA	NA	\$18.14	TBD	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	NA	NA	\$8.06	TBD	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	NA	NA	\$11.41	TBD	NA	NA	TBD	
$\pm \pm \pm$		00				1	<b>\$</b>					
4-V	-Wire Analog VG Loop-Start Unbundled Sub-Loop Feeder, per month	USBFE	\$24.94	NA	\$19.91	NA	NA	TBD	NA	\$10.22	NA	
<del>++ i</del>	Zone 1, per month	USBFE	TBD	\$23.35	TBD	\$30.69	\$25.90	TBD	\$21.91	NA	TBD	
+	Zone 2, per month	USBFE	TBD	\$27.94	TBD	\$36.12	\$27.00	TBD	\$35.92	NA NA	TBD	
+++	Zone 3. per month	USBFE	TBD	\$40.51	TBD	\$22.90	\$25.61	TBD	\$41.37	NA NA	TBD	<b>—</b>
+++	Zone 4, per month	USBFE	NA NA	NA	NA NA	NA	NA NA	TBD	NA NA	NA NA	NA NA	<b>—</b>
+	NRC - 1st	USBFE	\$224.21	\$222.74	\$243.41	\$221.19	\$220.87	TBD	\$226.36	\$232.76	TBD	
+++	NRC - Add'l	USBFE	\$142.45	\$140.22	\$81.32	\$138.67	\$138.47	TBD	\$144.28	\$91.92	TBD	<del></del>
+	NRC - Disconnect Charge - 1st	USBFE	\$131.15	\$127.64	\$134.77	\$133.53	\$135.74	TBD	\$96.50	NA	TBD	<b>—</b>
+	NRC - Disconnect Charge - Add'l	USBFE	\$34.66	\$32.91	\$33.93	\$33.69	\$34.25	TBD	\$25.82	NA NA	TBD	-
+	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD	<b>—</b>
++-	NRC - Service Order submitted Electronically, per LSR - Disconnect	SOMEC	\$3.50 NA	\$0.42	\$3.50 NA	\$3.50 NA	φ3.50 NA	TBD	\$3.50 NA	Ψ3.50 NA	TBD	<b>-</b>
+	NRC - Service Order submitted Manually, per LSR - Disconnect	SOMAN	NA NA	\$21.56	NA NA	\$29.24	NA NA	TBD	NA NA	NA NA	TBD	<del>                                     </del>
+	NRC - Service Order submitted Manually, per LSR.  NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA NA	\$3.84	NA NA	\$3.94	NA NA	TBD	NA NA	NA NA	TBD	<del></del>
+	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	\$3.84 NA	NA NA	\$3.94 NA	\$18.14	TBD	NA NA	NA NA	TBD	
+		SOMAN	TBD	NA NA	NA NA	NA NA	\$8.06	TBD	NA NA	NA NA	TBD	<b>——</b>
+	NRC - Incremental Charge - Manual Service Order - Add'l		TBD	NA NA	NA NA	NA NA	· · · · · · · · · · · · · · · · · · ·	TBD	NA NA	NA NA	TBD	
+	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	IBD	NA	NA	NA	\$11.41	IBD	NA	INA	IBD	<b>——</b>
	Was IODN Haban die d'Out-Le en Freder au mante	HODEE	<b>#00.00</b>	NIA	<b>047.70</b>	NA NA	N. A.	NIA	NIA	<b>#04.00</b>	NIA	<b>——</b>
2-V	-Wire ISDN Unbundled Sub-Loop Feeder, per month	USBFF	\$23.66	NA Coo.oo	\$17.73	NA 047.75	NA 010.01	NA	NA ©40.00	\$24.33	NA	<b>—</b>
+	Zone 1, per month	USBFF	TBD	\$22.39	TBD	\$17.75	\$19.34	TBD	\$19.63	NA	TBD	1
+	Zone 2, per month	USBFF	TBD	\$25.85	TBD	\$23.67	\$24.09	TBD	\$31.61	NA	TBD	
+	Zone 3, per month	USBFF	TBD	\$26.12	TBD	\$29.90	\$32.27	TBD	36,27	NA	TBD	+
$+\!\!+\!\!\!-\!\!\!\!+$	Zone 4, per month	USBFF	NA Coop oo	NA COOC 74	NA food 50	NA Coro	NA COLO FO	TBD	NA COOC OA	NA ************************************	NA	<del></del>
+	NRC - 1st	USBFF	\$200.26	\$222.74	\$208.50	\$218.90	\$218.59	TBD	\$202.01	\$293.73	TBD	+
+	NRC - Add'I	USBFF	\$104.51	\$140.22	\$62.31	\$136.39	\$136.19	TBD	\$105.88	\$126.38	TBD	<del></del>
+	NRC - Disconnect Charge - 1st	USBFF	\$119.95	\$127.64	\$119.68	\$121.40	\$120.82	TBD	\$92.57	NA	TBD	+
+	NRC - Disconnect Charge - Add'l	USBFF	\$27.04	\$32.91	\$29.58	\$26.04	\$25.91	TBD	\$21.09	NA	TBD	<del></del>
+	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD	<b>—</b>
$\bot \bot \bot$	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA	TBD	<b></b>
$\bot \!\!\!\! \bot \!\!\!\!\! \bot$	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	TBD	NA	NA	TBD	<b></b>
$\perp \! \! \perp \! \! \! \! \! \perp \! \! \! \! \! \! \! \! \! \!$	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	TBD	NA	NA	TBD	
ЩЦ	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	NA	NA	\$18.14	TBD	NA	NA	TBD	L
Ш	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	NA	NA	\$8.06	TBD	NA	NA	TBD	1
ЩЦ	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	NA	NA	\$11.41	TBD	NA	NA	TBD	L
												1
4-V	-Wire DSI Unbundled Sub-Loop Feeder, per month	USBFG	\$49.54	NA	TBD	NA	NA	NA	NA	NA	NA	
	Zone 1, per month	USBFG	TBD	\$56.00	TBD	\$74.60	\$64.67	TBD	\$39.69	TBD	TBD	
	Zone 2, per month	USBFG	TBD	\$80.13	TBD	\$104.09	\$92.67	TBD	\$67.36	TBD	TBD	
	Zone 3, per month	USBFG	TBD	\$156.12	TBD	\$151.77	\$309.79	TBD	\$78.12	TBD	TBD	1
		HODEO	NIA	N I A	NA	NA	NA	TBD	NA	NIA	NA	
ШІ	Zone 4, per month	USBFG USBFG	NA \$272.55	NA \$211.55	TBD	\$209.62	\$209.47	TBD	\$393.01	NA TBD	TBD	

NRC - Add'  USBFG \$133.92 \$129.04 TBD \$127.11 \$127.07 TBD \$153.37 TBD TBD     NRC - Disconnect Charge - 1st USBFG \$155.15 \$127.78 TBD \$133.43 \$135.74 TBD NA TBD TBD													
· · · · · · · · · · · · · · · · · · ·													
NRC - Disconnect Charge - Add'l	USBFG	\$24.59	\$33.06	TBD	\$33.59	\$34.25	TBD	NA	TBD	TBD			
NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD			
NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA	TBD			
NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	TBD	NA	NA	TBD			
NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	TBD	NA	NA	TBD			
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	TBD	NA	\$18.14	TBD	\$42.19	TBD	TBD			
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	TBD	NA	\$8.06	TBD	\$12.76	TBD	TBD			
NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	TBD	NA	\$11.41	TBD	NA	TBD	TBD			
2-Wire Copper Unbundled Sub-Loop Feeder, per month	USBFH	\$10.59	NA	NA	NA	NA	NA	NA	\$10.78	NA			
Zone 1, per month	USBFH	TBD	\$11.01	TBD	\$8.29	\$8.81	TBD	\$10.66	NA	TBD			
Zone 2, per month	USBFH	TBD	\$9.78	TBD	\$7.30	\$7.72	TBD	\$16.44	NA	TBD			
Zone 3, per month	USBFH	TBD	\$7.83	TBD	\$6.03	\$5.93	TBD	\$18.69	NA NA	TBD			
Zone 4, per month	USBFH	NA	NA	NA NA	NA	\$174.93	TBD	NA	NA NA	NA NA	1		
NRC - 1st	USBFH	\$172.20	\$175.18	TBD	\$175.18	\$92.53	TBD	\$172.89	\$259.80	TBD			
NRC - Add'l	USBFH	\$90.45	\$92.66	TBD	\$92.66	\$116.22	TBD	\$90.81	\$106.45	TBD	<del>                                     </del>		
NRC - Add 1 NRC - Disconnect Charge - 1st	USBFH	\$114.73	\$113.67	TBD	\$116.78	\$21.31	TBD	\$88.27	NA	TBD			
NRC - Disconnect Charge - 1st  NRC - Disconnect Charge - Add'l	USBFH	\$114.73	\$20.84	TBD	\$116.78	\$21.31	TBD	\$88.27 \$16.79	NA NA	TBD	-		
NRC - Disconnect Charge - Add I NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$20.84	\$3.50			TBD		\$3.50	TBD	<b> </b>		
7.1	SOMEC	\$3.50 NA	\$2.75 \$0.42	\$3.50 NA	\$3.50 NA	\$3.50 NA	TBD	\$3.50 NA	\$3.50 NA	TBD			
NRC - Setvice Order submitted Electronically, per LSR - Disconnect							TBD			TBD			
NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA		NA	NA	TBD			
NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	TBD	NA	NA				
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	TBD	NA	\$18.14	TBD	NA	NA	TBD			
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	TBD	NA	\$8.06	TBD	NA	NA	TBD			
NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	TBD	NA	\$11.41	TBD	NA	NA	TBD			
NRC - Incremental Charge - Manual Order Coordination - per loop	USBMC	\$51.29	\$16.31	TBD	\$16.31	\$16.29	TBD	\$45.34	\$45.53	TBD			
4-Wire Copper Unbundled Sub-Loop Feeder, per month	USBFJ	\$15.01	NA	NA	NA	NA	NA	NA	\$15.63	NA			
Zone 1, per month	USBFJ	TBD	\$20.59	TBD	\$16.55	\$20.58	TBD	\$14.68	NA	TBD			
Zone 2, per month	USBFJ	TBD	\$21.48	TBD	\$15.35	\$14.96	TBD	\$23.74	NA	TBD			
Zone 3, per month	USBFJ	TBD	\$17.70	TBD	\$12.52	\$13.15	TBD	\$27.26	NA	TBD			
Zone 4, per month	USBFJ	NA	NA	NA	NA	NA	TBD	NA	NA	NA			
NRC - 1st	USBFJ	\$206.32	\$209.61	TBD	\$209.61	\$209.31	TBD	\$207.14	\$294.59	TBD			
NRC - Add'l	USBFJ	\$134.23	\$127.09	TBD	\$127.09	\$126.91	TBD	\$134.77	\$152.62	TBD			
NRC - Disconnect Charge - 1st	USBFJ	\$123.01	\$119.80	TBD	\$126.27	\$128.36	TBD	\$90.12	NA	TBD			
NRC - Disconnect Charge - Add'l	USBFJ	\$26.53	\$25.07	TBD	\$26.43	\$26.87	TBD	\$19.44	NA	TBD			
NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD			
NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA	TBD			
NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	TBD	NA	NA	TBD			
NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	TBD	NA	NA	TBD			
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	TBD	NA	\$18.14	TBD	NA	NA	TBD			
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	TBD	NA	\$8.06	TBD	NA	NA	TBD			
NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	TBD	NA	\$11.41	TBD	NA	NA	TBD			
NRC - Incremental Charge - Manual Order Coordination - per loop	USBMC	\$51.29	\$16.31	TBD	\$16.31	\$16.29	TBD	\$45.34	\$45.43	TBD			
<u> </u>													
4-Wire 2.4 KBPS Digital Unbundled Sub-Loop Feeder, per month	USBFK	\$30.97	NA	NA	NA	NA	NA	NA	\$32.47	NA			
Zone 1, per month	USBFK	TBD	\$24.89	TBD	\$27.38	\$27.16	TBD	\$26.71	NA	TBD			
Zone 2, per month	USBFK	TBD	\$28.83	TBD	\$33.41	\$24.93	TBD	\$44.07	NA	TBD			
Zone 3, per month	USBFK	TBD	\$29.16	TBD	\$24.47	\$25.05	TBD	\$50.83	NA	TBD			
Zone 4, per month	USBFK	NA.	NA NA	NA	NA NA	NA	TBD	NA	NA NA	NA			
NRC - 1st	USBFK	\$212.90	\$211.32	TBD	\$209.77	\$209.47	TBD	\$215.00	\$309.57	TBD	1		
NRC - Add'I	USBFK	\$131.14	\$128.81	TBD	\$127.26	\$127.07	TBD	\$132.92	\$157.93	TBD	<b> </b>		
NRC - Disconnect Charge - 1st	USBFK	\$131.14	\$127.64	TBD	\$133.53	\$135.74	TBD	\$96.50	NA	TBD			
TT T INVO - Disconlinect Charge - 1st	USBER	φισι.ισ	φ121.0 <del>4</del>	טסו	φ133.33	φ133.74	עסו	φ30.50	INA	טסו			

				AND OTHER SE	RVICES								
Ш		NRC - Disconnect Charge - Add'l	USBFK	\$34.66	\$32.91	TBD	\$33.69	\$34.25	TBD	\$25.82	NA	TBD	
		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD	
		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA	TBD	
		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	TBD	NA	NA	TBD	
		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	TBD	NA	NA	TBD	
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	TBD	NA	\$18.14	TBD	NA	NA	TBD	
		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	TBD	NA	\$8.06	TBD	NA	NA	TBD	
		NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	TBD	NA	\$11.41	TBD	NA	NA	TBD	
	4-Wir	e 4.8 KBPS Digital Unbundled Sub-Loop Feeder, per month	USBFL	\$30.97	NA	NA	NA	NA	NA	NA	\$32.47	NA	
		Zone 1, per month	USBFL	TBD	\$24.89	TBD	\$27.38	\$27.16	TBD	\$26.71	NA	TBD	
		Zone 2, per month	USBFL	TBD	\$28.83	TBD	\$33.41	\$24.93	TBD	\$44.07	NA	TBD	
		Zone 3, per month	USBFL	TBD	\$29.16	TBD	\$24.47	\$25.05	TBD	\$50.83	NA	TBD	
H		Zone 4, per month	USBFL	NA NA	NA	NA.	NA	NA	TBD	NA	NA	NA	
$\Box$		NRC - 1st	USBFL	\$212.90	\$211.32	TBD	\$209.77	\$209.47	TBD	\$215.00	\$309.57	TBD	
H		NRC - Add'I	USBFL	\$131.14	\$128.81	TBD	\$127.26	\$127.07	TBD	\$132.92	\$157.93	TBD	
H		NRC - Disconnect Charge - 1st	USBFL	\$131.15	\$127.64	TBD	\$133.53	\$135.74	TBD	\$96.50	NA NA	TBD	
++	+	NRC - Disconnect Charge - Add'l	USBFL	\$34.66	\$32.91	TBD	\$33.69	\$34.25	TBD	\$25.82	NA	TBD	
H	+	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD	
H	+	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA	TBD	
H	-	NRC - Service Order submitted Manually, per LSR	SOMAN	NA NA	\$21.56	NA NA	\$29.24	NA NA	TBD	NA NA	NA NA	TBD	
+++		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA NA	\$3.84	NA NA	\$3.94	NA NA	TBD	NA NA	NA NA	TBD	
++	-	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	TBD	NA	\$18.14	TBD	NA NA	NA NA	TBD	
+++		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA NA	TBD	NA NA	\$8.06	TBD	NA NA	NA NA	TBD	
++	-	NRC - Incremental Charge - Manual Service Order - Add 1	SOMAN	TBD	NA NA	TBD	NA NA	\$11.41	TBD	NA NA	NA NA	TBD	
++		NCC - Inclemental Charge - Manual Service Order - Disconnect	SOMAN	100	INA	IBD	INA	<b>Φ11.41</b>	100	INA	INA	160	
++	4 18/:	o O C KADO Digital Unboundled Sub-Lean Fooder, nor month	USBFM	\$30.97	NA	NA	NA	NA	NA	NA	\$32.47	NA	
	4-VVII	e 9.6 KBPS Digital Unbundled Sub-Loop Feeder, per month	USBFM	Ψ30.97 TBD	\$24.89	TBD	\$27.38	\$27.16	TBD	\$26.71	Ψ32.47 NA	TBD	
+++	-	Zone 1, per month									NA NA	TBD	
++		Zone 2, per month	USBFM	TBD TBD	\$28.83	TBD TBD	\$33.41	\$24.93	TBD TBD	\$44.07	NA NA	TBD	
+++		Zone 3, per month	USBFM	NA NA	\$29.16 NA	NA NA	\$24.47 NA	\$25.05 NA	TBD	\$50.83 NA	NA NA	NA	
+++	-	Zone 4, per month	USBFM			TBD			TBD			TBD	
++		NRC - 1st	USBFM	\$212.90	\$211.32	TBD	\$209.77	\$209.47	TBD	\$215.00	\$309.57	TBD	
++			USBFM	\$131.14	\$128.81		\$127.26	\$127.07	TBD	\$132.92	\$157.93	TBD	
++		NRC - Disconnect Charge - 1st	USBFM	\$131.15	\$127.64	TBD	\$133.53	\$135.74		\$96.50	NA		
++		NRC - Disconnect Charge - Add'l	USBFM	\$34.66	\$32.91	TBD	\$33.69	\$34.25	TBD	\$25.82	NA To 50	TBD	
$\mathbb{H}$	_	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD	
$\mathbb{H}$	-	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA ************************************	NA	TBD	NA	NA	TBD	
$\mathbb{H}$		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	TBD	NA	NA	TBD	
Ш	_	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA 210.11	TBD	NA	NA	TBD	
Ш	_	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	TBD	NA	\$18.14	TBD	NA	NA	TBD	
Ш		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	TBD	NA	\$8.06	TBD	NA	NA	TBD	
Ш	_ _	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	TBD	NA	\$11.41	TBD	NA	NA	TBD	
Ш													
	4-Wir	e 19.2 KBPS Digital Unbundled Sub-Loop Feeder, per month	USBFN	\$30.97	NA	NA	NA	NA	NA	NA	\$32.47	NA	
$\coprod$		Zone 1, per month	USBFN	TBD	\$24.89	TBD	\$27.38	\$27.16	TBD	\$26.71	NA	TBD	
Ш		Zone 2, per month	USBFN	TBD	\$28.83	TBD	\$33.41	\$24.93	TBD	\$44.07	NA	TBD	
Ш		Zone 3, per month	USBFN	TBD	\$29.16	TBD	\$24.47	\$25.05	TBD	\$50.83	NA	TBD	
$\coprod$		Zone 4, per month	USBFN	NA	NA	NA	NA	NA	TBD	NA	NA	NA	
Ш	_ _	NRC - 1st	USBFN	\$212.90	\$211.32	TBD	\$209.77	\$209.47	TBD	\$215.00	\$309.57	TBD	
Ш		NRC - Add'l	USBFN	\$131.14	\$128.81	TBD	\$127.26	\$127.07	TBD	\$132.92	\$157.93	TBD	
Ш		NRC - Disconnect Charge - 1st	USBFN	\$131.15	\$127.64	TBD	\$133.53	\$135.74	TBD	\$96.50	NA	TBD	
Ш		NRC - Disconnect Charge - Add'l	USBFN	\$34.66	\$32.91	TBD	\$33.69	\$34.25	TBD	\$25.82	NA	TBD	
Ш		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	NA	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD	
Ш		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA	TBD	
		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	TBD	NA	NA	TBD	

			AND OTHER SE									
$\sqcup \sqcup$	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	TBD	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	TBD	NA	\$18.14	TBD	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	TBD	NA	\$8.06	TBD	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	TBD	NA	\$11.41	TBD	NA	NA	TBD	
4-\	4-Wire 56 KBPS Digital Unbundled Sub-Loop Feeder, per month	USBFO	\$30.97	NA	NA	NA	NA	NA	NA	\$32.47	NA	
	Zone 1, per month	USBFO	TBD	\$24.89	TBD	\$27.38	\$27.16	TBD	\$26.71	NA	TBD	
	Zone 2, per month	USBFO	TBD	\$28.83	TBD	\$33.41	\$24.93	TBD	\$44.07	NA	TBD	
	Zone 3, per month	USBFO	TBD	\$29.16	TBD	\$24.47	\$25.05	TBD	\$50.83	NA	TBD	
	Zone 4, per month	USBFO	NA	NA	NA	NA	NA	TBD	NA	NA	NA	
H	NRC - 1st	USBFO	\$212.90	\$211.32	TBD	\$209.77	\$209.47	TBD	\$215.00	\$309.57	TBD	
+++	NRC - Add'I	USBFO	\$131.14	\$128.81	TBD	\$127.26	\$127.07	TBD	\$132.92	\$157.93	TBD	
+++	NRC - Disconnect Charge - 1st	USBFO	\$131.15	\$127.64	TBD	\$133.53	\$135.74	TBD	\$96.50	NA NA	TBD	
++-	NRC - Disconnect Charge - Add'l	USBFO	\$34.66	\$32.91	TBD	\$33.69	\$34.25	TBD	\$25.82	NA	TBD	
+++	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	NA NA	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD	
++-	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA NA	NA	NA	TBD	NA	NA	TBD	
HH	NRC - Service Order submitted Electronically, per LSR - Disconnect	SOMAN	NA NA	\$21.56	NA NA	\$29.24	NA NA	TBD	NA NA	NA NA	TBD	
++-	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA NA	\$3.84	NA NA	\$29.24	NA NA	TBD	NA NA	NA NA	TBD	
++-	NRC - Service Order submitted Manually, per LSR, Disconnect  NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	\$3.84 NA	TBD	\$3.94 NA	\$18.14	TBD	NA NA	NA NA	TBD	
++-	NRC - Incremental Charge - Manual Service Order - 1st  NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA NA	TBD	NA NA	\$18.14	TBD	NA NA	NA NA	TBD	
++-	NRC - Incremental Charge - Manual Service Order - Add 1  NRC - Incremental Charge - Manual Service Order - Disconnect		TBD	NA NA	TBD	NA NA		TBD	NA NA	NA NA	TBD	
H	INTO - Incremental Charge - Manual Service Order - Disconnect	SOMAN	ומט	INA	ומט	INA	\$11.41	IBD	INA	INA	IDU	
H	A William OA K DDO District High condition of the design of the second o	HODED	<b>#20.07</b>	NA	NIA	NA	NA	NIA	NIA	<b>#20.47</b>	NIA	
4-1	4-Wire 64 KBPS Digital Unbundled Sub-Loop Feeder, per month	USBFP	\$30.97		NA			NA	NA Table 74	\$32.47	NA	
$\vdash$	Zone 1, per month	USBFP	TBD	\$24.89	TBD	\$27.38	\$27.16	TBD	\$26.71	NA	TBD	
$\Box$	Zone 2, per month	USBFP	TBD	\$28.83	TBD	\$33.41	\$24.93	TBD	\$44.07	NA	TBD	
$\Box$	Zone 3, per month	USBFP	TBD	\$29.16	TBD	\$24.47	\$25.05	TBD	\$50.83	NA	TBD	
$\Box$	Zone 4, per month	USBFP	NA	NA	NA	NA	NA	TBD	NA	NA	NA	
ш	NRC - 1st	USBFP	\$212.90	\$211.32	TBD	\$209.77	\$209.47	TBD	\$215.00	\$309.57	TBD	
ш	NRC - Add'I	USBFP	\$131.14	\$128.81	TBD	\$127.26	\$127.07	TBD	\$132.92	\$157.93	TBD	
ш	NRC - Disconnect Charge - 1st	USBFP	\$131.15	\$127.64	TBD	\$133.53	\$135.74	TBD	\$96.50	NA	TBD	
	NRC - Disconnect Charge - Add'l	USBFP	\$34.66	\$32.91	TBD	\$33.69	\$34.25	TBD	\$25.82	NA	TBD	
	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	NA	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD	
	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	NA	NA	TBD	
	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	\$29.24	TBD	NA	NA	TBD	
	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	\$3.94	TBD	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	TBD	NA	\$18.14	TBD	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	TBD	NA	\$8.06	TBD	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	NA	TBD	NA	\$11.41	TBD	NA	NA	TBD	
												_
Un	Unbundled Sub-Loop Modification											
	NRC - Load Coil/Equipment Removal per 2 Wire pair - 1st	ULM2X	\$358.90	\$357.81	TBD	\$357.82	\$332.79	TBD	\$360.33	\$358.71	TBD	
	NRC - Load Coil/Equipment Removal per 2 Wire pair - Add'l	ULM2X	\$8.08	\$8.15	TBD	\$8.15	\$7.49	TBD	\$8.11	\$8.07	TBD	
	NRC - Load Coil/Equipment Removal per 4 Wire pair - 1st	ULM4X	\$358.90	\$357.81	TBD	\$357.82	\$332.79	TBD	\$360.33	\$358.71	TBD	
Ш	NRC - Load Coil/Equipment Removal per 4 Wire pair - Add'l	ULM4X	\$8.08	\$8.15	TBD	\$8.15	\$7.49	TBD	\$8.11	\$8.07	TBD	
	NRC - Bridge Tap Removal per pair unloaded - 1st	ULMBT	\$561.91	\$562.71	TBD	\$562.73	\$521.03	TBD	\$564.15	\$561.62	TBD	
Ш	NRC - Bridge Tap Removal per pair unloaded - Add'l	ULMBT	\$10.09	\$10.19	TBD	\$10.19	\$9.36	TBD	\$10.14	\$8.07	TBD	
Lo	Loop Make Up											
	NRC - Loop Makeup - Preordering Without Reservation, per working facility											
	queried (Manual) **	UMKLW	\$134.00	\$134.00	\$134.00	\$134.00	\$134.00	\$134.00	\$134.00	\$134.00	\$100.00	
	Loop Makeup - Preordering Without Reservation, per spare facility queried											
	(Manual) Maximum number of spare facilities per manual LMUSI is 3. **	UMKLW	\$134.00	\$134.00	\$134.00	\$134.00	\$134.00	\$134.00	\$134.00	\$134.00	\$100.00	
$\Box$	NRC-Loop Makeup - Preordering With Reservation, per spare facility queried											
	(Manual) Max number of spare facilities per manual LMUSI is 3. **	UMKLP	\$140.00	\$140.00	\$140.00	\$140.00	\$140.00	\$140.00	\$140.00	\$140.00	\$100.00	
ш_	Number of Spare Residue por mandar Empor to C.	CHINE	ψσ.σ	ψ	Ψ	ψσ.σσ	Ψ	Ψ σ. σ σ	ψ	Ψ σ.σσ	Ψ.00.00	

			AND OTHER SE	RVICES	,						
	NRC - Loop Makeup - Preordering Without Reservation, per working facility queried (Mechanized) **		\$1.08	\$1.08	\$1.08	\$1.08	\$1.08	\$1.08	\$1.08	\$1.08	\$0.6888
	Loop Makeup - Preordering Without Reservation, per spare facility queried										
ш	(Mechanized) Max number of spare facilities per mechanized LMUSI is 10.**		\$1.08	\$1.08	\$1.08	\$1.08	\$1.08	\$1.08	\$1.08	\$1.08	\$0.6888
	Loop Makeup - Preordering With Reservation, per spare facility queried										
	(Mechanized) Max number of spare facilities per mechanized LMUSI is 10.**		\$1.08	\$1.08	\$1.08	\$1.08	\$1.08	\$1.08	\$1.08	\$1.08	\$0.6888
ш											
Uı	Inbundled Network Terminating Wire, per pair, per month	UENPP	\$0.49	\$0.46	\$1.37	\$0.64	\$0.35	TBD	\$0.44	\$0.46	TBD
	NRC - UNTW Pair, per pair	UENPP	\$40.02	\$65.35	\$2.48	\$65.35	\$64.77	TBD	\$65.82	\$60.93	TBD
ш	NRC - Disconnect Charge, per pair	UENPP	\$0.87	NA	\$1.74	NA	NA	TBD	NA	NA	TBD
	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	TBD
	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	TBD	TBD	TBD	TBD
	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	\$29.24	TBD	TBD	TBD	TBD
$\mathbb{H}$	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	\$3.94	TBD	TBD	TBD	TBD
S	Sub-Loop Concentration - Channelization Sys (Outside CO)										
H-3	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	TBD	\$3.50	\$3.50	\$3.50
++	NRC - Service Order submitted Electronically, per LSR - Disconnect	SOMEC	\$3.50 NA	\$0.42	NA	NA	NA	TBD	TBD	TBD	NA
HH	NRC - Service Order submitted Electronically, per LSR - Disconnect	SOMAN	NA NA	\$21.56	NA NA	\$29.24	\$29.24	TBD	TBD	TBD	\$19.99
++-	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA NA	\$3.84	NA	\$3.94	\$3.94	TBD	TBD	TBD	NA NA
++-	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	\$18.94	TBD	TBD	TBD	NA NA	NA NA	NA NA
+++	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA NA	\$8.42	TBD	TBD	TBD	NA NA	NA NA	NA NA
++-	TR008 - System A (96 channel capacity - channels 1-96), per month	UCT8A	\$829.76	\$477.76	\$724.79	\$493.89	\$546.21	TBD	\$579.03	\$571.42	\$683.78
++-	NRC - 1st	UCT8A	\$404.46	\$408.22	\$632.36	\$408.23	\$407.65	TBD	\$587.83	\$586.42	\$634.31
+++	NRC - Add'I	UCT8A	\$220.32	\$222.37	\$310.82	\$222.37	\$222.06	TBD	\$278.83	\$277.97	\$311.78
+++	NRC-Disconnect, 1st	UCT8A	\$236.23	\$236.02	Ψ310.62 NA	\$253.87	\$258.08	TBD	Ψ276.63 NA	NA	NA NA
+++	NRC-Disconnect, Add'l	UCT8A	\$74.90	\$74.84	NA NA	\$80.50	\$81.83	TBD	NA NA	NA NA	NA NA
++-	TR008 - System B (96 channel capacity - channels 97-192), per month	UCT8B	\$87.34	\$85.12	\$92.91	\$82.15	\$87.89	TBD	\$85.59	\$87.65	\$102.12
++-	NRC - 1st	UCT8B	\$404.46	\$408.22	\$632.36	\$408.23	\$407.65	TBD	\$587.83	\$586.42	\$634.31
++-	NRC - Add'I	UCT8B	\$220.32	\$222.37	\$310.82	\$222.37	\$222.06	TBD	\$278.83	\$277.97	\$311.78
++-	NRC-Disconnect, 1st	UCT8B	\$236.23	\$236.02	νA	\$253.87	\$258.08	TBD	Ψ276.63 NA	NA	NA
++-	NRC-Disconnect, Add'l	UCT8B	\$74.90	\$74.84	NA NA	\$80.50	\$81.83	TBD	NA NA	NA NA	NA NA
+++	TR303 - System A (96 channel capacity - channels 1-96), per month	UCT3A	\$868.77	\$512.86	\$764.42	\$530.75	\$584.28	TBD	\$615.62	\$611.62	\$726.87
++-	NRC - 1st	UCT3A	\$404.46	\$408.22	\$632.36	\$408.23	\$407.65	TBD	\$587.83	\$586.42	\$634.31
++-	NRC - Add'l	UCT3A	\$220.32	\$222.37	\$310.82	\$222.37	\$222.06	TBD	\$278.83	\$277.97	\$311.78
++-	NRC-Disconnect, 1st	UCT3A	\$220.32	\$236.02	NA	\$253.87	\$258.08	TBD	Φ276.63 NA	Φ277.97 NA	NA
++-	NRC-Disconnect, 1st	UCT3A	\$74.90	\$74.84	NA NA	\$80.50	\$81.83	TBD	NA NA	NA NA	NA NA
++-	TR303 - System B (96 channel capacity - channels 97-192), per month	UCT3B	\$126.35	\$120.21	\$132.54	\$119.01	\$125.95	TBD	\$122.18	\$127.76	\$145.21
++-	NRC - 1st	UCT3B	\$404.46	\$408.22	\$632.36	\$408.23	\$407.65	TBD	\$587.83	\$586.42	\$634.31
++-	NRC - Add'l	UCT3B	\$220.32	\$222.37	\$310.82	\$222.37	\$222.06	TBD	\$278.83	\$277.97	\$311.78
HH	NRC-Disconnect, 1st	UCT3B	\$236.23	\$236.02	Φ310.62 NA	\$253.87	\$258.08	TBD	Φ276.63 NA	- φ277.97 NA	Ψ311.76 NA
++-	NRC-Disconnect, Add'l	UCT3B	\$236.23 \$74.90	\$236.02	NA NA	\$253.87	\$258.08	TBD	NA NA	NA NA	NA NA
++-	DS1 Feeder Interface, per month	UCTFS	\$67.03	\$74.84 NA	\$72.12	\$80.50 NA	\$81.83 NA	TBD	\$64.63	\$69.24	\$76.73
HH	Zone 1, per month	UCTFS	λ67.03 NA	\$56.65	Ψ72.12 TBD	\$60.01	\$73.29	TBD	TBD	769.24 ТВD	TBD
HH	Zone 2, per month	UCTFS	NA NA	\$65.86	TBD	\$60.01	\$88.72	TBD	TBD	TBD	TBD
HH	Zone 3, per month	UCTFS	NA NA	\$107.08	TBD	\$106.11	\$88.72 \$187.11	TBD	TBD	TBD	TBD
HH	Zone 4, per month	UCTFS	NA NA	\$107.08 NA	NA NA	\$106.11 NA	\$187.11 NA	TBD	NA NA	NA NA	NA NA
HH	NRC 1st	UCTFS	\$192.51	\$211.55	\$425.74	\$209.62	\$209.47	TBD	\$237.09	\$236.32	\$418.37
++-	NRC Add'l	UCTFS	\$192.51	\$211.55	\$425.74 \$198.06	\$209.62	\$209.47	TBD	\$237.09 \$116.01	\$236.32	\$198.67
HH	NRC-Disconnect, 1st	UCTFS	\$105.73 \$56.94	\$129.04 \$127.78	\$198.06 NA	\$127.11 \$133.43	\$127.07 \$135.74	TBD	\$116.01 NA	\$115.55 NA	\$198.67 NA
HH	NRC-Disconnect, 1st	UCTFS	\$56.94 \$12.81	\$33.06	NA NA	\$133.43	\$135.74	TBD	NA NA	NA NA	NA NA
HH	Channel Interface-2 Wire Voice-Loop Start or Ground Start, per mo	ULCC2	\$12.81	\$33.06	\$2.38	\$33.59	\$34.25	TBD	\$2.22	\$2.43	\$2.61
HH	NRC 1st	ULCC2		\$2.12 \$21.07	\$2.38 \$41.82	\$2.24 \$21.08		TBD	\$2.22 \$28.66	\$2.43	\$2.61 \$41.95
++	NRC 1st	ULCC2	\$20.88 \$20.77	\$21.07 \$20.96	\$41.82 \$41.58	\$21.08 \$20.96	\$21.05 \$20.93	TBD	\$28.50	\$28.58	\$41.95 \$41.71
	INKC Aud I	ULUUZ	\$2U.//	\$20.96	<b>⊅41.5</b> 8	<b>⊅</b> ∠U.90	ֆ∠∪.ყპ	ופט	ֆ∠ၓ.၁∪	<b>⊅∠</b> 8.4∠	φ41./1

				AND OTHER SE	RVICES								
		NRC-Disconnect, 1st	ULCC2	\$10.00	\$9.99	NA	\$10.75	\$10.92	TBD	NA	NA		
		NRC-Disconnect, Add'l	ULCC2	\$9.94	\$9.93	NA	\$10.68	\$10.86	TBD	NA	NA		
	Ch	nannel Interface - 2 Wire ISDN, per month	ULCC1	\$9.51	\$8.48	\$9.53	\$8.98	\$9.34	TBD	\$8.88	\$9.72	\$10.43	
		NRC 1st	ULCC1	\$20.88	\$21.07	\$41.82	\$21.08	\$21.05	TBD	\$21.24	\$28.58	\$41.95	
		NRC Add'I	ULCC1	\$20.77	\$20.96	\$41.58	\$20.96	\$20.93	TBD	\$21.13	\$28.42	\$41.71	
		NRC-Disconnect, 1st	ULCC1	\$10.00	\$9.99	NA	\$10.75	\$10.92	TBD	NA	NA	NA	
		NRC-Disconnect, Add'l	ULCC1	\$9.94	\$9.93	NA	\$10.68	\$10.86	TBD	NA	NA	NA	
H		nannel Interface - 2 Wire Voice - Reverse Battery, per month	ULCCR	\$14.14	\$12.61	\$14.17	\$13.35	\$13.89	TBD	\$13.20	\$14.46	\$15.51	
H		NRC 1st	ULCCR	\$20.88	\$21.07	\$41.82	\$21.08	\$21.05	TBD	\$28.66	\$28.58	\$41.95	
+++	ı.	NRC Add'l	ULCCR	\$20.77	\$20.96	\$41.58	\$20.96	\$20.93	TBD	\$28.50	\$28.42	\$41.71	
+++		NRC-Disconnect, 1st	ULCCR	\$10.00	\$9.99	NA NA	\$10.75	\$10.92	TBD	NA NA	NA NA	NA	
HH	+	NRC-Disconnect, Add'l	ULCCR	\$9.94	\$9.93	NA NA	\$10.68	\$10.86	TBD	NA NA	NA NA	NA NA	
HH		nannel Interface - 4 Wire Voice (Specials Card), per month	ULCC4	\$8.43	\$7.52	\$8.45	\$7.96	\$8.29	TBD	\$7.87	\$8.62	\$9.26	
++	CI.	NRC 1st	ULCC4	\$20.88	\$21.07	\$41.82	\$21.08	\$21.05	TBD	\$28.66	\$28.58	\$41.95	
++	-	NRC Add'l	ULCC4	\$20.77	\$20.96	\$41.58	\$20.96	\$20.93	TBD	\$28.50	\$28.42	\$41.71	
++	-	NRC-Disconnect, 1st	ULCC4	\$10.00	\$9.99	NA	\$10.75	\$10.92	TBD	NA	Ψ20.42 NA	NA NA	
++		NRC-Disconnect, Add'l											
+++	_	,	ULCC4	\$9.94	\$9.93	NA CALOO	\$10.68	\$10.86	TBD	NA Too 47	NA ©40.44	NA Out-oo	
+++		est Circuit, per month	UCTTC	\$41.21	\$36.76	\$41.30	\$38.90	\$40.49	TBD	\$38.47	\$42.14	\$45.22	
+++		NRC 1st	UCTTC	\$20.88	\$21.07	\$41.82	\$21.08	\$21.05	TBD	\$28.66	\$28.58	\$41.95	
$\mathbb{H}$	-	NRC Add'l	UCTTC	\$20.77	\$20.96	\$41.58	\$20.96	\$20.93	TBD	\$28.50	\$28.42	\$41.71	
$\Box$		NRC-Disconnect, 1st	UCTTC	\$10.00	\$9.99	NA	\$10.75	\$10.92	TBD	NA	NA	NA	
Ш		NRC-Disconnect, Add'l	UCTTC	\$9.94	\$9.93	NA	\$10.68	\$10.86	TBD	NA	NA	NA	
Ш	Ch	nannel Interface - Digital 56Kbps, per month	ULCC5	\$12.49	\$11.14	\$12.51	\$11.79	\$12.27	TBD	\$11.66	\$12.77	\$13.71	
Ш		NRC 1st	ULCC5	\$20.88	\$21.07	\$41.82	\$21.08	\$21.05	TBD	\$28.66	\$28.58	\$41.95	
		NRC Add'l	ULCC5	\$20.77	\$20.96	\$41.58	\$20.96	\$20.93	TBD	\$28.50	\$28.42	\$41.71	
		NRC-Disconnect, 1st	ULCC5	\$10.00	\$9.99	NA	\$10.75	\$10.92	TBD	NA	NA	NA	
		NRC-Disconnect, Add'l	ULCC5	\$9.94	\$9.93	NA	\$10.68	\$10.86	TBD	NA	NA	NA	
	Ch	nannel Interface - Digital 64Kbps, per month	ULCC6	\$12.49	\$11.14	\$12.51	\$14.08	\$12.27	TBD	\$11.66	\$12.77	\$13.71	
		NRC 1st	ULCC6	\$20.88	\$21.07	\$41.82	\$21.08	\$21.05	TBD	\$28.66	\$28.58	\$41.95	
		NRC Add'l	ULCC6	\$20.77	\$20.96	\$41.58	\$20.96	\$20.93	TBD	\$28.50	\$28.42	\$41.71	
		NRC-Disconnect, 1st	ULCC6	\$10.00	\$9.99	NA	\$10.75	\$10.92	TBD	NA	NA	NA	
		NRC-Disconnect, Add'l	ULCC6	\$9.94	\$9.93	NA	\$10.68	\$10.86	TBD	NA	NA	NA	
	oop	Concentration System (Inside C.O.)											
		NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
		NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA	
		NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	\$29.24	NA	NA	NA	\$19.99	
		NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	\$3.94	NA	NA	NA	NA	
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBD	NA	\$18.94	TBD	\$18.14	\$25.52	TBD	\$44.06	TBD	
		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBD	NA	\$8.42	TBD	\$8.06	\$11.34	TBD	\$13.55	TBD	
		R008 -System A (96 channel capacity - channels 1-96), per month	UCT8A	\$327.44	\$470.73	\$316.63	\$522.17	\$432.54	\$454.79	\$375.96	\$399.21	\$380.06	
		NRC - 1st	UCT8A	\$1,115.10	\$651.05	\$1,111.95	\$651.04	\$650.11	\$1,115.10	\$1,113.00	\$1,119.30	\$1,114.05	
+++	+	NRC - Add'I	UCT8A	NA NA	NA	NA	NA	NA	NA NA	NA NA	NA	NA NA	
H		NRC-Disconnect, 1st	UCT8A	NA NA	NA NA	NA.	NA	NA.	NA NA	NA NA	NA NA	NA NA	
	+	NRC-Disconnect, Add'l	UCT8A	NA NA	NA	NA NA	NA NA	NA	NA NA	NA NA	NA	NA NA	
+++	TE	R008 -System B (96 channel capacity - channels 97-192), per month	UCT8B	\$67.41	\$55.96	\$65.27	\$63.59	\$61.71	\$73.30	\$65.98	\$71.91	\$68.71	
+++		NRC - 1st	UCT8B	\$464.57	\$271.27	\$463.37	\$271.27	\$270.88	\$464.71	\$463.74	\$466.38	\$464.21	
+++	-	NRC - Add'l	UCT8B	5464.57 ΝΑ	Φ2/1.2/ NA	NA	Φ2/1.2/ NA	λ270.00 NA	Φ464.71 NA	η463.74 NA	Ψ400.36 NA	NA	
+++	-	NRC-Disconnect, 1st	UCT8B	NA NA	NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA	
+++	+	NRC-Disconnect, Add'l	UCT8B	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	
+++	TE	R303 - System A (96 channel capacity - channels 1-96), per month	UCT3A	\$375.18	\$510.37	\$362.87	\$567.21	\$476.24	\$506.70	\$422.68	\$450.13	\$428.73	
+++		NRC - 1st	UCT3A	\$375.18	\$651.05	\$1,111.95	\$651.04	\$650.11	\$1,115.10	\$1,113.00	\$1,119.30	\$1,114.05	
+++	-	INRC - 1st	UCT3A	\$1,115.10 NA	NA	\$1,111.95 NA	\$651.04 NA	NA	\$1,115.10 NA	\$1,113.00 NA	\$1,119.30 NA	\$1,114.05 NA	
+++	-	NRC-Add I NRC-Disconnect, 1st	UCT3A	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA		NA NA	
++			UCT3A		NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	
+++		NRC-Disconnect, Add'l		NA C444.00									
	II R	R303 - System B (96 channel capacity - channels 97-192), per month	UCT3B	\$111.30	\$94.30	\$110.02	\$107.16	\$103.99	\$123.52	\$111.17	\$121.16	\$115.79	

		AND OTHER SE									
NRC - 1st	UCT3B	\$464.57	\$271.27	\$463.37	\$271.27	\$270.88	\$464.71	\$463.74	\$466.38	\$464.21	
NRC - Add'l	UCT3B	NA	NA	NA	NA	NA	NA	NA	NA	NA	
NRC-Disconnect, 1st	UCT3B	NA	NA	NA	NA	NA	NA	NA	NA	NA	
NRC-Disconnect, Add'I	UCT3B	NA	NA	NA	NA	NA	NA	NA	NA	NA	
DS1 Interface, per month	UCTCO	\$6.42	\$5.28	\$6.15	\$6.04	\$5.91	\$6.99	\$6.27	\$6.79	\$6.49	
NRC 1st	UCTCO	\$367.70	\$126.61	\$366.72	\$126.61	\$126.43	\$367.80	\$367.04	\$369.13	\$367.41	
NRC Add'I	UCTCO	\$132.03	\$92.17	\$130.63	\$92.17	\$92.04	\$132.07	\$131.79	\$132.54	\$131.92	
NRC-Disconnect, 1st		· ·		NA	\$33.46	\$34.02	NA	NA		NA	
	UCTCO	NA	\$31.11						NA		
NRC-Disconnect, Add'l	UCTCO	NA	\$8.71	NA	\$9.37	\$9.52	NA	NA	NA	NA	
Channel Interface-2 Wire Voice-Loop Start or Groun		\$2.55	\$2.10	\$2.44	\$2.40	\$2.35	\$2.77	\$0.89	\$2.69	\$2.58	
NRC 1st	ULCC2	\$35.77	\$21.07	\$35.68	\$21.08	\$21.05	\$35.78	\$35.73	\$35.91	\$35.74	
NRC Add'l	ULCC2	\$35.55	\$20.96	\$35.48	\$20.96	\$20.93	\$35.37	\$35.49	\$35.71	\$35.54	
NRC-Disconnect, 1st	ULCC2	NA	\$9.99	NA	\$10.75	\$10.92	NA	NA	NA	NA	
NRC-Disconnect, Add'l	ULCC2	NA	\$9.93	NA	\$10.68	\$10.86	NA	NA	NA	NA	
Channel Interface - 2 Wire ISDN, per month	ULCC1	\$10.19	\$8.38	\$9.76	\$9.59	\$9.39	\$11.10	\$9.95	\$10.76	\$10.30	
NRC 1st	ULCC1	\$35.77	\$21.07	\$35.68	\$21.08	\$21.05	\$35.78	\$35.71	\$35.91	\$35.74	
NRC Add'I	ULCC1	\$35.55	\$20.96	\$35.48	\$20.96	\$20.93	\$35.37	\$35.51	\$35.71	\$35.54	
NRC-Disconnect, 1st	ULCC1	NA	\$9.99	NA NA	\$10.75	\$10.92	NA	NA	NA NA	NA NA	
NRC-Disconnect, Add'l	ULCC1	NA NA	\$9.93	NA NA	\$10.73	\$10.86	NA NA	NA NA	NA NA	NA NA	
Channel Interface - 2 Wire Voice - Reverse Battery,		\$15.15	\$12.46	\$14.51	\$14.26	\$13.95	\$16.46	\$14.80	\$16.01	\$15.32	
. NRC 1st	ULCCR	\$35.77	\$12.46	\$35.68	\$14.26	\$21.05	\$35.78	\$35.71	\$35.91	\$35.74	
NRC 4dd'l	ULCCR				\$21.08 \$20.96						
NRC-Disconnect, 1st		\$35.55	\$20.96	\$35.48 NA		\$20.93	\$35.37	\$35.51	\$35.71	\$35.54 NA	
<u> </u>	ULCCR	NA	\$9.99		\$10.75	\$10.92	NA	NA	NA		
NRC-Disconnect, Add'l	ULCCR	NA	\$9.93	NA	\$10.68	\$10.86	NA	NA	NA	NA	
Channel Interface - 4 Wire Voice, per month	ULCC4	\$9.04	\$7.43	\$8.65	\$8.51	\$8.32	\$9.83	\$8.82	\$9.55	\$9.13	
NRC 1st	ULCC4	\$35.77	\$21.07	\$35.68	\$21.08	\$21.05	\$35.78	\$35.71	\$35.91	\$35.74	
NRC Add'l	ULCC4	\$35.55	\$20.96	\$35.48	\$20.96	\$20.93	\$35.37	\$35.51	\$35.71	\$35.54	
NRC-Disconnect, 1st	ULCC4	NA	\$9.99	NA	\$10.75	\$10.92	NA	NA	NA	NA	
NRC-Disconnect, Add'l	ULCC4	NA	\$9.93	NA	\$10.68	\$10.86	NA	NA	NA	NA	
Test Circuit, per month	UCTTC	\$44.16	36.31	\$42.30	\$41.58	\$40.67	\$47.85	\$43.13	\$46.66	\$44.65	
NRC 1st	UCTTC	\$35.77	\$21.07	\$35.68	\$21.08	\$21.05	\$35.78	\$35.71	\$35.91	\$35.74	
NRC Add'l	UCTTC	\$35.55	\$20.96	\$35.48	\$20.96	\$20.93	\$35.37	\$35.51	\$35.71	\$35.54	
NRC-Disconnect, 1st	UCTTC	\$35.55	\$9.99	NA	\$10.75	\$10.92	NA	NA	NA NA	NA NA	
NRC-Disconnect, Add'l	UCTTC	\$35.55	\$9.93	NA NA	\$10.68	\$10.86	NA	NA	NA.	NA NA	
Channel Interface - Digital 56Kbps, per month	ULCC5	TBD	\$11.01	TBD	\$12.60	\$12.33	TBD	TBD	TBD	TBD	
NRC 1st	ULCC5	TBD	\$21.07	TBD	\$21.08	\$21.05	TBD	TBD	TBD	TBD	
NRC 4dd'l		TBD		TBD	\$21.08		TBD	TBD	TBD	TBD	
	ULCC5		\$20.96			\$20.93					
NRC-Disconnect, 1st	ULCC5	TBD	\$9.99	NA	\$10.75	\$10.92	TBD	TBD	TBD	TBD	
NRC-Disconnect, Add'l	ULCC5	TBD	\$9.93	NA	\$10.68	\$10.86	TBD	TBD	TBD	TBD	
Channel Interface - Digital 64Kbps, per month	ULCC6	TBD	\$11.01	TBD	\$12.60	\$12.33	TBD	TBD	TBD	TBD	
NRC 1st	ULCC6	TBD	\$21.07	TBD	\$21.08	\$21.05	TBD	TBD	TBD	TBD	
NRC Add'l	ULCC6	TBD	\$20.96	TBD	\$20.96	\$20.93	TBD	TBD	TBD	TBD	
NRC-Disconnect, 1st	ULCC6	TBD	\$9.99	NA	\$10.75	\$10.92	TBD	TBD	TBD	TBD	
NRC-Disconnect, Add'l	ULCC6	TBD	\$9.93	NA	\$10.68	\$10.86	TBD	TBD	TBD	TBD	
LINE SHARING											
2-Wire analog VG (SL1) for Line Sharing											
RC - per month (Note 3) **		TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	\$12.16	
NRC - 1st (Note 3) **		TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	\$31.99	
NRC - Add'l (Note 3) **		TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	\$20.02	
Tatio Additions of		100	טטו	טטו	טטו	טפו	טטו	טטו	טטו	Ψ20.02	
System Splitter, OS Line Conscitu		+								<del>                                     </del>	
System Splitter - 96 Line Capacity		#400 00	<b>#</b> 400.00	M400.00	#40C CC	#400 CC	<b>#</b> 400.00	#40C CC	M400 00	Ø400.00	
RC - Per month **	ULSDA	\$100.00	\$100.00	\$100.00	\$100.00	\$100.00	\$100.00	\$100.00	\$100.00	\$100.00	
NRC - 1st **	ULSDA	\$150.00	\$150.00	\$150.00	\$300.00	\$150.00	\$300.00	\$300.00	\$300.00	\$150.00	
NRC - Addl **	ULSDA	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	

			AND OTHER SE									
	NRC - Disconnect 1st **	ULSDA	\$150.00	\$150.00	\$150.00	NA	\$150.00	NA	NA	NA	\$150.00	
	NRC - Disconnect Add'I **	ULSDA	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	
				,								
	System Splitter - 24 Line Capacity											
++++	RC - Per month **	ULSDB	\$25.00	\$25.00	\$25.00	\$25.00	\$25.00	\$25.00	\$25.00	\$25.00	\$25.00	
	NRC - 1st **											
		ULSDB	\$150.00	\$150.00	\$150.00	\$300.00	\$150.00	\$300.00	\$300.00	\$300.00	\$150.00	
$\Box$	NRC - Addl **	ULSDB	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	
	NRC - Disconnect 1st **	ULSDB	\$150.00	\$150.00	\$150.00	NA	\$150.00	NA	NA	NA	\$150.00	
	NRC - Disconnect Add'I **	ULSDB	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	
	Loop Capacity, Line Activation Per Occurrence											
	RC - Per Month **	ULSDC	\$6.00	\$6.00	\$6.00	\$6.00	\$6.00	\$6.00	\$6.00	\$6.00	\$3.48	
	NRC - 1st **	ULSDC	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	
++++	NRC - Addl **	ULSDC	\$22.00	\$22.00	\$22.00	\$22.00	\$22.00	\$22.00	\$22.00	\$22.00	\$21.39	
++++	INCC - Addi	ULSDC	\$22.00	\$22.00	\$22.00	\$22.00	\$22.00	\$22.00	\$22.00	\$22.00	\$21.39	
+++	NDO 0 : 0   1   1   1   1   1   1   1   1   1						*					
ш	NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	TBD	
IIII	NRC - Service Order submitted Manually, per LSR	SOMAN	NA	\$21.56	NA	\$29.24	NA	NA	NA	NA	\$19.99	
	NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	NA	\$3.84	NA	\$3.94	NA	NA	NA	NA	TBD	
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.22	NA	
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA	
HH	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA NA	NA	NA NA	\$11.41	\$16.06	NA NA	NA	NA NA	
+++	INC - Incientental Charge - Manual Service Order - Disconnect	SOMAIN	Φ17.77	INA	INA	INA	φ11. <del>4</del> 1	\$10.00	INA	INA	INA	
HHH												
	Subsequent Activity - Per Occurrence											
	NRC - 1st **	ULSDS	\$30.00	\$30.00	\$30.00	\$30.00	\$30.00	\$30.00	\$30.00	\$30.00	\$30.00	
	NRC - Addl **	ULSDS	\$15.00	\$15.00	\$15.00	\$15.00	\$15.00	\$15.00	\$15.00	\$15.00	\$15.00	
	* Interim Rates subject to true-up											
++++	** TN rates are interim and subject to true-up.											
NO1												
	In states where a specific NRC for customer transfer, feature additions and											
HH-1.	changes is not stated, the applicable NRC from the appropriate tariff applies.											
-1 $1$ $1$ $1$	2 Effective May 1, 2000 statewide rates will be replaced by Deaveraged Loop											
	Rates by Zone where available. Until approximately December 31, 2000 or until											
	such time that BellSouth billing systems have been developed to handle the new											
	zone rate structure, BellSouth will bill at the Zone 1 Deaveraged Loop rate level											
	only. After December 31, 2000 or such time that the billing systems have been											
	developed to handle the new zone rate structure, BellSouth will begin billing											
	pursuant to CLEC-1's interconnection agreement.											
	The recurring interim and nonrecurring interim rates in TN for 2-Wire analog VG											
	(SL1) for Line Sharing is for a stand-alone loop purchased by CLEC-1 to provide											
	both analog voice service and xDSL services or in the event CLEC-1 wishes to											
	continue providing xDSL services to an end-user who terminates its BellSouth-											
	provided voice service. These rates apply when CLEC-1 purchases the splitter											
	from BellSouth.											
+++	IIUIII DEIIOUUIII.											
+++	<del> </del>		ļ									
+++												

EDCAL EXCHANGE SWITCHING (PORTS)	DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
2. Wire Voice Grade Line Port (Residence), per month	OCAL EXCHANGE SWITCHING (PORTS)										1
2 - wire voice unbundled port - residence	2-Wire Analog Line Port (Res., Bus.), per month										1
2-wive voice unbundled port with caller ID - residence	2-Wire Voice Grade Line Port (Residence), per month										1
LEPRO   \$2.07   \$1.62   \$1.85   \$2.61   \$2.20   \$2.11   \$2.00   \$2.35	2- wire voice unbundled port - residence	UEPRL	\$2.07	\$1.62	1.85 - Note 1	2.61 - Note 1	\$2.20	\$2.11	\$2.19	\$2.35	4.73 - Note 1
Decided particular continuation of Alabama extended local dising parity port with caller ID	2-wire voice unbundled port with caller ID - residence	UEPRC	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.73
D   Caller ID	2-wire voice unbundled port outgoing only - residence	UEPRO	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.73
caller ID  2-wire voice grade unbundled Louisiana extended local dialing parity port with caller to voice grade unbundled Louisiana extended local dialing parity port with caller to voice grade unbundled Mississippi extended local dialing parity port with caller to Voice grade unbundled South Carolina extended local dialing parity port with caller to Voice grade unbundled South Carolina extended local dialing parity port with caller to Voice grade unbundled Tennessee extended local dialing parity port with UEPAU NA			\$2.07	NA	NA	NA	NA	NA	NA	NA	NA
Caller ID   UEPAS   NA   NA   NA   NA   NA   NA   NA	caller ID	UEPRM	NA	NA	NA	\$2.61	NA	NA	NA	NA	NA
Caller ID   Caller October   Caller Oc	caller ID	UEPAS	NA	NA	NA	NA	\$2.20	NA	NA	NA	NA
Caller ID	caller ID		NA	NA	NA	NA	NA	\$2.11	NA	NA	NA
Caller ID   Call	caller ID		NA	NA	NA	NA	NA	NA	NA	\$2.35	NA
2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL) 2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7) 2-wire voice unbundled Louisiana Area Plus with caller ID - residence (LV8) 2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence (LV8) 2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R) 1	caller ID										\$4.73
Z-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)  Z-wire voice unbundled South Carolina Area Calling port with Caller ID - residence (LW8)  Z-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (LW8)  Z-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (LW8)  Z-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (LPAJ)  Z-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (LPAJ)  Z-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (LPAJ)  Z-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (LY8)  Z-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (LY8)  Z-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (LY8)  Z-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (LY8)  Z-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (LY8)  Z-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (LY8)  Z-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (LY8)  Z-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (LY8)  Z-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (LY8)  Z-wire voice unbundled res, low usage line port with Caller ID - residence (LY8)  Z-wire voice unbundled res, low usage line port with Caller ID (LUM)  Z-wire voice unbundled res, low usage line port with Caller ID (LUM)  Z-wire voice unbundled port with unbundled port with Caller ID (LUM)  Z-wire voice unbundled port with unbundled port with Caller ID (LY8)  Z-wire voice unbundled port with unbundled port with Caller ID (LY8)  Z-wire voice unbundled port with unbundled port with Caller ID (LY8)  Z-wire voice unbundled Albama extended local dialing parity port with (LY8)  Z-wire voice grade unbundled Kentucky extended local dialing parity port with (LY8)  Z-wire voice grade unb		_									NA
2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence (L/WB)   2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)   2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)   2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACER)   2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (UEPAL NA	,		NA	NA	NA	NA	\$2.20	NA	NA	NA	NA
L(W8)   Care voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)   Care voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)   Care voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)   Care voice unbundled Tennessee Area Calling port with Caller ID - residence (TACSR)   UEPAL NA			NA	NA	NA	NA	\$2.20	NA	NA	NA	NA
(FZR)	(LW8)		NA	NA	NA	NA	NA	NA	NA	\$2.35	NA
CTACER    UEPAL NA	(F2R)	UEPAK	NA	NA	NA	NA	NA	NA	NA	NA	\$4.73
TACSR)  Quere voice unbundled Tennessee Area Calling port with Caller ID - residence (1MF2X)  Quere voice unbundled Tennessee Area Calling port with Caller ID - residence (2MR)  Quere voice unbundled Tennessee Area Calling port with Caller ID (LUM)  Quere voice unbundled res, low usage line port with Caller ID (LUM)  Quere voice unbundled res, low usage line port with Caller ID (LUM)  Quere voice unbundled port with unbund	(TACER)	UEPAL	NA	NA	NA	NA	NA	NA	NA	NA	\$4.73
Company   Comp	(TACSR)	UEPAM	NA	NA	NA	NA	NA	NA	NA	NA	\$4.73
(2MR) UEPAO NA	(1MF2X)	UEPAN	NA	NA	NA	NA	NA	NA	NA	NA	\$4.73
LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)  LNPCX  2-Wire Voice Grade Line Port(Business), per month  2-wire voice unbundled port without Caller ID  2-wire voice unbundled port with unbundled port with Caller+E484 ID  2-wire voice unbundled opt with unbundled port with Caller +E484 ID  2-wire voice unbundled outgoing only port  2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID  UEPBC  2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID  UEPAW  \$2.07  NA  NA  NA  NA  NA  NA  NA  NA  NA  N	(2MR)										\$4.73
2-Wire Voice Unbundled port without Caller ID 2-wire voice unbundled port with unbundled port with Caller+E484 ID 2-wire voice unbundled port with unbundled port with Caller+E484 ID 2-wire voice unbundled port with unbundled port with Caller+E484 ID 2-wire voice unbundled port with unbundled port with Caller+E484 ID 2-wire voice unbundled outgoing only port 2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID UEPAW 2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID UEPAW 1	2-wire voice unbundled res, low usage line port with Caller ID (LUM)	UEPAP	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.73
2-wire voice unbundled port with unbundled port with Caller ID  2-wire voice unbundled port with unbundled port with Caller +E484 ID  2-wire voice unbundled port with unbundled port with Caller +E484 ID  2-wire voice unbundled outgoing only port  2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID  UEPBC  2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID  UEPBC  2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID  UEPBC  2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID  UEPBM  NA  NA  NA  NA  NA  NA  NA  NA  NA  N	LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	LNPCX									
2-wire voice unbundled port with out Caller ID  2-wire voice unbundled port with unbundled port with Caller+E484 ID  2-wire voice unbundled port with unbundled port with Caller+E484 ID  2-wire voice unbundled outgoing only port  2-wire voice unbundled outgoing only port  2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID  UEPAW  2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID  UEPBC  \$2.07  \$1.62  \$1.85  \$2.61  \$2.20  \$2.11  \$2.00  \$2.35  2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID  UEPAW  \$2.07  NA  NA  NA  NA  NA  NA  NA  NA  NA  N	2-Wire Voice Grade Line Port(Rusiness), per month										+
2-wire voice unbundled port with unbundled port with Caller+E484 ID  2-wire voice unbundled outgoing only port  UEPBC  2-wire voice unbundled outgoing only port  UEPBC  2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID  UEPAW  S2.07  NA  NA  NA  NA  NA  NA  NA  NA  NA  N		LIEPRI	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.73
2-wire voice unbundled outgoing only port  2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID  2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID  UEPAW  \$2.07  NA  NA  NA  NA  NA  NA  NA  NA  NA  N											\$4.73
2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID  2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID  UEPBM  NA  NA  NA  NA  NA  NA  NA  NA  NA  N											\$4.73
2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID  2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID  UEPBM NA	2-wire voice grade unbundled Alabama extended local dialing parity port with called	er	•								NA NA
2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID  UEPAX  NA  NA  NA  NA  NA  NA  NA  NA  NA											NA
2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID  UEPAY  NA  NA  NA  NA  NA  NA  NA  NA  NA											NA
	caller ID		NA	NA	NA	NA	NA	\$2.11	NA	NA	NA
	caller ID	th UEPAZ	NA	NA	NA	NA	NA	NA	NA	\$2.35	NA
2-wire voice grade unbundled Tennessee extended local dialing parity port with caller ID UEPAV NA			NA	NA	NA	NA	NA	NA	NA	NA	\$4.73
2-wire voice unbundled incoming only port with Caller ID UEPB1 \$2.07 \$1.62 \$1.85 \$2.61 \$2.20 \$2.11 \$2.00 \$2.35	2-wire voice unbundled incoming only port with Caller ID	UEPB1	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.73

DES	SCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
TT	2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	UEPAA	NA	NA	NA	NA	\$2.20	NA	NA	NA	NA
11	2-wire voice unbundled SC Bus Area Calling Port with Caller ID (LMB)	UEPAB	NA	NA	NA	NA	NA	NA	NA	\$2.35	NA
t	2-wire voice unbundled TN Bus 2-Way Area Calling Port Economy Option		1							4=.00	
	(TACC1)	UEPAC	NA	NA	NA	NA	NA	NA	NA	NA	\$4.73
	2-wire voice unbundled TN Bus 2-Way Area Calling Port Standard Option										<b>V U</b>
	(TACC2)	UEPAD	NA	NA	NA	NA	NA	NA	NA	NA	\$4.73
11	2-wire voice unbundled TN Bus 2-WAY Collierville and Memphis Local Calling Port										•
	(B2F)	UEPAE	NA	NA	NA	NA	NA	NA	NA	NA	\$4.73
Ħ											•
11	LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	LNPCX									
Ħ	, , , , , , , , , , , , , , , , , , ,										
Ħ	Non-Recurring Charges (NRC) - 1st (Residence)										
Ħ	2- wire voice unbundled port - residence	UEPRL	\$21.93	\$4.76	\$17.16	\$37.78	\$16.43	\$22.98	\$21.60	\$24.98	\$9.93
11	2-wire voice unbundled port with caller ID - residence	UEPRC	\$21.93	\$4.76	\$17.16	\$37.78	\$16.43	\$22.98	\$24.04	\$24.98	\$9.93
	2-wire voice unbundled port outgoing only - residence	UEPRO	\$21.93	\$4.76	\$17.16	\$37.78	\$16.43	\$22.98	\$24.04	\$24.98	\$9.93
Ħ	2-wire voice grade unbundled Alabama extended local dialing parity port with caller	-				*				, , , , ,	*
	ID 31 71	UEPAR	\$21.93	NA	NA	NA	NA	NA	NA	NA	NA
Ħ	2-wire voice grade unbundled Kentucky extended local dialing parity port with										
	caller ID	UEPRM	NA	NA	NA	\$37.78	NA	NA	NA	NA	NA
	2-wire voice grade unbundled Louisiana extended local dialing parity port with										
	caller ID	UEPAS	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
	2-wire voice grade unbundled Mississippi extended local dialing parity port with										
	caller ID	UEPAT	NA	NA	NA	NA	NA	\$22.98	NA	NA	NA
	2-wire voice grade unbundled South Carolina extended local dialing parity port with										
	caller ID	UEPAU	NA	NA	NA	NA	NA	NA	NA	\$24.98	NA
	2-wire voice grade unbundled Tennessee extended local dialing parity port with										
	caller ID	UEPAQ	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
	2-wire voice unbundled Florida area calling with caller ID - residence	UEPAF	NA	\$4.76	NA	NA	NA	NA	NA	NA	NA
	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)	UEPAG	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)	UEPAH	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
	2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence										
	(LW8)	UEPAJ	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
	(F2R)	UEPAK	NA	NA	NA	NA	NA	NA	NA	\$24.98	NA
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
	(TACER)	UEPAL	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
	(TACSR)	UEPAM	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
	(1MF2X)	UEPAN	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
	(2MR)	UEPAO	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
	2-wire voice unbundled Res Low Usage Line Port with Caller+E563 ID (LUM)	UEPAP	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
	NRC - Add'l (Residence)										
	2- wire voice unbundled port - residence	UEPRL	\$21.93	\$4.54	\$17.16	\$37.78	\$16.43	\$22.98	\$21.60	\$24.98	\$9.19
	2-wire voice unbundled port with caller ID - residence	UEPRC	\$21.93	\$4.54	\$17.16	\$37.78	\$16.43	\$22.98	\$9.08	\$24.98	\$9.19
	2-wire voice unbundled port outgoing only - residence	UEPRO	\$21.93	\$4.54	\$17.16	\$37.78	\$16.43	\$22.98	\$9.08	\$24.98	\$9.19
	2-wire voice grade unbundled Alabama extended local dialing parity port with caller										
	ID	UEPAR	\$21.93	NA	NA	NA	NA	NA	NA	NA	NA
	2-wire voice grade unbundled Kentucky extended local dialing parity port with										
	caller ID	UEPRM	NA	NA	NA	\$37.78	NA	NA	NA	NA	NA
	2-wire voice grade unbundled Louisiana extended local dialing parity port with										
	caller ID	UEPAS	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
			<u> </u>		•						

DESCR	IPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	e-wire voice grade unbundled Mississippi extended local dialing parity port with										
	aller ID	UEPAT	NA	NA	NA	NA	NA	\$22.98	NA	NA	NA
	P-wire voice grade unbundled South Carolina extended local dialing parity port with										
	aller ID	UEPAU	NA	NA	NA	NA	NA	NA	NA	\$24.98	NA
	P-wire voice grade unbundled Tennessee extended local dialing parity port with caller ID	UEPAQ	NA	NA	NA	NA	NA	NA	NA	NA	\$9.19
	ealer ID - residence !-wire voice unbundled Florida area calling with caller ID - residence	UEPAF	NA NA	\$4.54	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)	UEPAG	NA NA	NA	NA NA	NA NA	\$16.43	NA NA	NA NA	NA NA	NA NA
	P-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)	UEPAH	NA NA	NA.	NA NA	NA NA	\$16.43	NA.	NA.	NA.	NA
	P-wire voice unbundled South Carolina Area Calling port with Caller ID - residence						Ţ				
('	LW8)	UEPAJ	NA	NA	NA	NA	NA	NA	NA	\$24.98	NA
	P-wire voice unbundled Tennessee Area Calling port with Caller ID - residence F2R)	UEPAK	NA	NA	NA	NA	NA	NA	NA	NA	\$9.19
	P-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										40
	TACER)	UEPAL	NA	NA	NA	NA	NA	NA	NA	NA	\$9.19
	-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
	TACSR)	UEPAM	NA	NA	NA	NA	NA	NA	NA	NA	\$9.19
	P-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
	1MF2X)	UEPAN	NA	NA	NA	NA	NA	NA	NA	NA	\$9.19
	P-wire voice unbundled Tennessee Area Calling port with Caller ID - residence 2MR)	UEPAO	N/A	NA	NA	NIA.	NA	NA	NA	NA	\$9.19
	2-wire voice unbundled Res Low Usage Line Port with Caller ID (LUM)	UEPAD	NA \$21.93	\$4.54	\$17.16	NA \$37.78	\$16.43	\$22.98	\$9.08	\$24.98	\$9.19
	-wire voice driburidied Res Low Osage Line Fort with Caller ID (LOW)	UEFAF	φ21.93	φ4.54	φ17.10	φ37.76	\$10.43	\$22.90	φ9.06	\$24.90	φ9.19
	IRC - Subsequent Activity	USASC	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
<del>                                     </del>	The Subsequent / ourns	OOAGO	Ψ10.00	Ψ10.00	ψ10.00	Ψ10.00	ψ10.00	Ψ10.00	ψ10.00	Ψ10.00	Ψ10.00
I	IRC - 1st (Business)										
	P-wire Voice Unbundled Port without Caller ID	UEPBL	\$21.93	\$4.76	\$17.16	\$37.55	\$16.43	\$22.98	\$21.60	\$24.98	\$9.93
2	e-wire voice unbundled port with Caller ID	UEPBC	\$21.93	\$4.76	\$17.16	\$37.55	\$16.43	\$22.98	\$24.04	\$24.98	\$9.93
	2-wire voice unbundled outgoing only port	UEPBO	\$21.93	\$4.76	\$17.16	\$37.55	\$16.43	\$22.98	\$24.04	\$24.98	\$9.93
	P-wire voice grade unbundled Alabama extended local dialing parity port with caller										
	D	UEPAW	\$21.93	NA	NA	NA	NA	NA	NA	NA	NA
	e-wire voice grade unbundled Kentucky extended local dialing parity port with										
	aller ID	UEPBM	NA	NA	NA	\$37.78	NA	NA	NA	NA	NA
	P-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	UEPAX	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
	P-wire voice grade unbundled Mississippi extended local dialing parity port with	UEFAX	INA	INA	INA	INA	\$10.43	INA	INA	INA	INA
	raller ID	UEPAY	NA	NA	NA	NA	NA	\$22.98	NA	NA	NA
	P-wire voice grade unbundled South Carolina extended local dialing parity port with	02.7	10/	147	147.	107	10.	Ψ22.00	107	10.	10.
	aller ID	UEPAZ	NA	NA	NA	NA	NA	NA	NA	\$24.98	NA
2	e-wire voice grade unbundled Tennessee extended local dialing parity port with										
	aller ID	UEPAV	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
	P-wire voice unbundled Incoming only Port with Caller ID	UEPB1	\$21.93	\$4.76	\$17.16	\$37.55	\$16.43	\$22.98	\$24.04	\$24.98	\$9.93
	P-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	UEPAA	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
	P-wire voice unbundled SC Bus Area Calling Port with Caller ID+E587 (LMB)	UEPAB	NA	NA	NA	NA	NA	NA	NA	\$24.98	NA
2	2-wire voice unbundled TN Bus 2-way Area Calling Port Economy Option (TACC1)	UEPAC	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
l s	2-wire voice unbundled TN Bus 2-way Area Calling Port Standard Option (TACC2)	UEPAD	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
	P-wire voice unbundled TN Bus 2-way Collierville and Memphis Local Calling Port	OLI /ID	14/3	19/3	19/3	19/3	14/3	14/3	19/3	19/3	Ψ0.00
	B2F)	UEPAE	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
1111								ļ	ļ		1
	IRC - Add'l (Rusiness)										
	IRC - Add'I (Business) -wire voice unbundled port without Caller ID	UEPBL	\$21.93	\$4.54	\$17.16	\$37.55	\$16.43	\$22.98	\$21.60	\$24.98	\$9.19

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
2-wire voice unbundled outgoing only port	UEPBO	\$21.93	\$4.54	\$17.16	\$37.55	\$16.43	\$22.98	\$9.08	\$24.98	\$9.19
2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	UEPAW	\$21.93	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	UEPBM	NA	NA	NA	\$37.78	NA	NA	NA	NA	NA
2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	UEPAX	NA NA	NA NA	NA NA	NA NA	\$16.43	NA NA	NA NA	NA NA	NA NA
2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	UEPAY	NA NA	NA NA	NA NA	NA NA	NA NA	\$22.98	NA NA	NA NA	NA NA
2-wire voice grade unbundled South Carolina extended local dialing parity port with caller ID	UEPAZ	NA NA	NA NA	NA NA	NA NA	NA NA	NA	NA NA	\$24.98	NA NA
2-wire voice grade unbundled Tennessee extended local dialing parity port with caller ID	UEPAV	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA	\$9.19
2-wire voice unbundled incoming only port with Caller ID	UEPB1	\$21.93	\$4.54	\$17.16	\$37.55	\$16.43	\$22.98	\$9.08	\$24.98	\$9.19
2-wire voice unburided incoming only port with Caller ID (BUC)	UEPAA	921.93 NA	NA	NA	NA	\$16.43	922.96 NA	NA	924.96 NA	NA
2-wire voice unburidied EA bus Area Calling Fort with Caller ID (bbc)  2-wire voice unburidied EA bus Area Calling Fort with Caller ID (LMB)	UEPAB	NA	NA NA	NA	NA NA	NA	NA	NA	\$24.98	NA
2-wire voice unbundled TN Bus 2-way Area Calling Port Economy Option (TACC1)	UEPAC	NA	NA	NA	NA	NA	NA	NA	NA	\$9.19
2-wire voice unbundled TN Bus 2-way Area Calling Port Standard Option (TACC2)	UEPAD	NA	NA	NA	NA	NA	NA	NA	NA	\$9.19
2-wire voice unbundled TN Bus 2-way Collierville and Memphis Locall Calling Port (B2F)	UEPAE	NA	NA	NA	NA	NA	NA	NA	NA	\$9.19
NRC - Subsequent Activity	USASC	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
NDC Disconnect Charge 4ct		+								
NRC - Disconnect Charge - 1st  2- wire voice unbundled port - residence	UEPRL	\$6.21	\$2.76	NA	NA	\$4.38	\$6.56	NA	NA	\$3.66
2-wire voice unburidled port - residence  2-wire voice unbundled port with caller ID - residence	UEPRC	\$6.21	\$2.76	NA NA	NA NA	\$4.38	\$6.56	NA NA	NA NA	\$3.66
2-wire voice unburided port with caller 15 - residence	UEPRO	\$6.21	\$2.76	NA NA	NA NA	\$4.38	\$6.56	NA NA	NA NA	\$3.66
2-wire voice grade unbundled Alabama extended local dialing parity port with caller	UEPAR	\$6.21	NA	NA NA	NA NA	NA NA	NA	NA NA	NA NA	NA
2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	UEPRM	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	UEPAS	NA NA	NA NA	NA NA	NA NA	\$4.38	NA NA	NA NA	NA NA	NA NA
2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	UEPAT	NA NA	NA NA	NA NA	NA NA	NA NA	\$6.56	NA NA	NA NA	NA NA
2-wire voice grade unbundled South Carolina extended local dialing parity port with caller ID	UEPAU	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
2-wire voice grade unbundled Tennessee extended local dialing parity port with caller ID	UEPAQ	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$3.66
2-wire voice unbundled Florida area calling with caller ID - residence	UEPAF	NA NA	\$2.76	NA NA	NA	NA	NA NA	NA NA	NA NA	NA
2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)	UEPAG	NA NA	NA	NA NA	NA NA	\$4.38	NA NA	NA NA	NA NA	NA NA
2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)	UEPAH	NA	NA NA	NA NA	NA NA	\$4.38	NA NA	NA	NA NA	NA
2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence (LW8)	UEPAJ	NA.	NA	NA	NA	NA	NA	NA NA	NA	NA
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)	UEPAK	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$3.66
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACER)	UEPAL	NA	NA	NA	NA	NA	NA	NA	NA	\$3.66
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACSR)	UEPAM	NA	NA	NA	NA	NA	NA	NA	NA	\$3.66
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (1MF2X)	UEPAN	NA	NA	NA	NA	NA	NA	NA	NA	\$3.66

DES	CRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
	(2MR)	UEPAO	NA	NA	NA	NA	NA	NA	NA	NA	\$3.66
	2-wire voice unbundled Res Low Usage Line Port with Caller ID (LUM)	UEPAP	\$6.21	\$2.76	NA	NA	\$4.38	\$6.56	NA	NA	\$3.66
	2-wire voice unbundled port without Caller ID	UEPBL	\$6.21	\$2.76	NA	NA	\$4.38	\$6.56	NA	NA	\$3.66
	2-wire voice unbundled port with Caller ID	UEPBC	\$6.21	\$2.76	NA	NA	\$4.38	\$6.56	NA	NA	\$3.66
	2-wire voice unbundled outgoing only Port	UEPBO	\$6.21	\$2.76	NA	NA	\$4.38	\$6.56	NA	NA	\$3.66
	2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	UEPAW	\$6.21	NA	NA	NA	NA	NA	NA	NA	NA
	2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	UEPBM	NA	NA	NA	NA	NA	NA	NA	NA	NA
	2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	UEPAX	NA	NA	NA	NA	\$4.38	NA	NA	NA	NA
	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	UEPAY	NA NA	NA NA	NA NA	NA NA	NA NA	\$6.56	NA.	NA NA	NA NA
	2-wire voice grade unbundled South Carolina extended local dialing parity port with caller ID	UEPAZ	NA NA	NA NA	NA NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA
	2-wire voice grade unbundled Tennessee extended local dialing parity port with										
$\vdash$	caller ID	UEPAV	NA \$6.21	NA \$2.76	NA NA	NA	NA ©4.00	NA \$6.56	NA	NA	\$3.66
<del>     </del>	2-wire voice unbundled Incoming only Port with Caller ID	UEPB1	* -			NA	\$4.38		NA	NA	\$3.66
<del>     </del>	2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	UEPAA	NA NA	NA	NA	NA	\$4.38	NA	NA	NA	NA
$\vdash$	2-wire voice unbundles SC Bus Area Calling Port with Caller ID (LMB)	UEPAB	NA	NA	NA	NA	NA	NA	NA	NA	NA
	2-wire voice unbundled TN Bus 2-way Area Calling Port Economy Option (TACC1)	UEPAC	NA	NA	NA	NA	NA	NA	NA	NA	\$3.66
	2-wire voice unbundled TN Bus 2-way Area Calling Port Standard Option (TACC2)	UEPAD	NA	NA	NA	NA	NA	NA	NA	NA	\$3.66
	2-wire voice unbundled TN Bus 2-Way Collierville and Memphis Local Calling Port (B2F)	UEPAE	NA	NA	NA	NA	NA	NA	NA	NA	\$3.66
<u> </u>	NDO DI LOI ALIII										
	NRC - Disconnect Charge - Add'I	LIEDDI	20.01	00.50			<b>A</b> 4 00	<b>\$0.50</b>			<b>*</b> 0.00
	2- wire voice unbundled port - residence	UEPRL	\$6.21	\$2.59	NA	NA	\$4.38	\$6.56	NA	NA	\$2.92
	2-wire voice unbundled port with caller ID - residence	UEPRC	\$6.21	\$2.59	NA	NA	\$4.38	\$6.56	NA	NA	\$2.92
	2-wire voice unbundled port outgoing only - residence	UEPRO	\$6.21	\$2.59	NA	NA	\$4.38	\$6.56	NA	NA	\$2.92
	2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	UEPAR	\$6.21	NA	NA	NA	NA	NA	NA	NA	NA
	2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	UEPRM	NA	NA	NA	NA	NA	NA	NA	NA	NA
	2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	UEPAS	NA	NA	NA	NA	\$4.38	NA	NA	NA	NA
	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	UEPAT	NA	NA	NA	NA	NA	\$6.56	NA	NA	NA
	2-wire voice grade unbundled South Carolina extended local dialing parity port with caller ID	UEPAU	NA NA	NA NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA
	2-wire voice grade unbundled Tennessee extended local dialing parity port with caller ID	UEPAQ	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$2.92
++	2-wire voice unbundled Florida area calling with caller ID - residence	UEPAF	NA NA	\$2.59	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA
++	2-wire voice unbundled Florida area calling with caller ID - residence (RUL)	UEPAG	NA NA	NA	NA NA	NA NA	\$4.38	NA NA	NA	NA NA	NA NA
++	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)	UEPAH	NA NA	NA	NA NA	NA NA	\$4.38	NA NA	NA	NA NA	NA NA
	2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence (LUN8)	UEPAJ	NA NA	NA NA	NA NA	NA NA	νΑ.36	NA NA	NA NA	NA NA	NA NA
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)	UEPAK	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$2.92
-	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence	OLITAN	INA	INA	INA	14/4	INA	INM	INA	INA	ψΖ.3∠

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACSR)	UEPAM	NA	NA	NA	NA	NA	NA	NA	NA	\$2.92
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (1MF2X)	UEPAN	NA	NA	NA	NA	NA	NA	NA	NA	\$2.92
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence	-									, ,
(2MR)	UEPAO	NA	NA	NA	NA	\$4.38	\$6.56	NA	NA	\$2.92
2-wire voice unbundled Res Low Usage Line Port with Caller ID (LUM)	UEPAP	\$6.21	\$2.59	NA	NA	\$4.38	\$6.56	NA	NA	\$2.92
2-wire voice unbundled port without Caller ID	UEPBL	\$6.21	\$2.59	NA	NA	\$4.38	\$6.56	NA	NA	\$2.92
2-wire voice unbundled port with Caler ID	UEPBC	\$6.21	\$2.59	NA	NA	\$4.38	\$6.56	NA	NA	\$2.92
2-wire voice unbundled outgoing only port	UEPBO	\$6.21	\$2.59	NA	NA	\$4.38	\$6.56	NA	NA	\$2.92
2-wire voice grade unbundled Alabama extended local dialing parity port with caller	UEPAW	\$6.21	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	UEPBM	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA.
2-wire voice grade unbundled Louisiana extended local dialing parity port with	OEFBIN	INA	INA	INA	INA	INA	INA	INA	INA	INA
caller ID	UEPAX	NA	NA	NA	NA	\$4.38	NA	NA	NA	NA
2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	UEPAY	NA	NA	NA	NA	NA	\$6.56	NA	NA	NA
2-wire voice grade unbundled South Carolina extended local dialing parity port with caller ID	UEPAZ	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice grade unbundled Tennessee extended local dialing parity port with	LIEDAY	NIA.	N10		NIA	N/A	NIA	N/A		<b>#</b> 0.00
caller ID	UEPAV UEPB1	NA Co.o.t	NA \$2.59	NA	NA	NA \$4.38	NA Co. FO	NA NA	NA	\$2.92
2-wire voice unbundled incoming only port with Caller ID  2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	UEPAA	\$6.21		NA	NA		\$6.56		NA	\$2.92
	UEPAA	NA NA	NA NA	NA NA	NA NA	\$4.38	NA	NA NA	NA NA	NA NA
2-wire voice unbundled SC Bus Area Calling Port with Caller ID (LMB)	UEPAB	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice unbundled TN Bus 2-way Area Calling Port Economy Option (TACC1)	UEPAC	NA	NA	NA	NA	NA	NA	NA	NA	\$2.92
2-wire voice unbundled TN Bus 2-way Area Calling Port Standard Option (TACC2)	UEPAD	NA	NA	NA	NA	NA	NA	NA	NA	\$2.92
2-wire voice unbundled TN Bus 2-way Collierville and Memphis Local Calling Port										
(B2F)	UEPAE	NA	NA	NA	NA	NA	NA	NA	NA	\$2.92
NRC - OSS										
NRC - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces	SOMEC	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.42	\$20.3
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$14.63	\$10.5
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$10.39	\$16.06	NA	NA	\$13.3
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$1.44	NA	NA	NA	NA	NA	NA	NA	\$1.40
All available features, per month	UEPVF	\$5.55	\$3.40	NA	NA	\$8.28	\$6.75	NA	\$6.29	NA
NRC - 1st (all types)		\$24.72	NA	NA	NA	NA	\$21.42	NA	\$36.24	NA
NRC - Add'l (all types)		\$24.72	NA	NA	NA	NA	\$21.42	NA	\$36.24	NA
NRC - Disconnect Charge - 1st		\$18.41	NA	NA	NA	NA	\$19.68	NA	NA	NA
NRC - Disconnect Charge - Add'l		\$18.41	NA	NA	NA	NA	\$19.68	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	NA	NA	NA	\$25.52	NA	\$44.42	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	NA	NA	NA	\$11.34	NA	\$14.63	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	NA	\$16.06	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$1.44	NA	NA	NA	NA	NA	NA	NA	NA
Three available feature, per month	UEPVF	NA	NA	NA	NA	\$8.28	\$3.31	NA	\$3.03	NA
NRC - 1st (all types)		NA	NA.	NA	NA.	NA NA	\$3.06	NA	\$4.53	NA
NRC - Add'l (all types)		NA NA	NA NA	NA NA	NA NA	NA NA	\$3.06	NA NA	\$4.53	NA.

DES	CRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
	NRC - Disconnect Charge - 1st		NA	NA	NA	NA	NA	\$8.20	NA	NA	NA
	NRC - Disconnect Charge - Add'l		NA	NA	NA	NA	NA	\$8.20	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	NA	NA	NA	NA	\$25.52	NA	\$44.42	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	NA	NA	NA	\$11.34	NA	\$14.63	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	NA	NA	NA	NA	NA	\$16.06	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
4-Wii	e Analog VG Port, per month	UEP4A	NA	\$8.74	\$8.47	NA	\$10.13	\$9.60	\$8.69	\$2.28	\$11.11
	NRC - 1st	UEP4A	NA	\$4.76	\$17.16	NA	\$16.43	\$22.98	\$21.69	\$3.50	\$9.93
	NRC - Add'l	UEP4A	NA	\$4.54	\$17.16	NA	\$16.43	\$22.98	\$21.69	\$3.50	\$9.19
	NRC - Disconnect Charge - 1st	BFR	NA	\$2.82	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
	NRC - Disconnect Charge - Add'l	BFR	NA	\$2.64	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	\$21.56	\$18.94	NA	\$18.14	\$25.52	\$26.85	NA	\$20.35
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	\$21.56	\$8.42	NA	\$8.06	\$11.34	\$12.67	NA	\$10.54
	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	NA	\$3.84	NA	NA	\$8.94	\$16.06	NA	NA	\$13.32
	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	NA	\$3.84	NA	NA	\$8.94	\$16.06	NA	NA	\$1.40
				*							* -
2-Wii	e DID Port, per month	UEPP2	\$12.08	\$9.38	\$11.35	NA	\$13.12	\$14.63	\$12.36	\$12.08	\$8.97
	NRC - 1st	UEPP2	\$50.00	\$248.44	\$61.91	NA	\$59.28	\$83.09	\$81.84	\$50.00	\$47.75
	NRC - Add'l	UEPP2	\$18.00	\$37.49	\$61.91	NA	\$59.28	\$83.09	\$81.84	\$50.00	\$47.01
	NRC - Disconnect Charge - 1st	UEPP2	NA	\$113.28	NA	NA	\$9.20	\$13.48	NA	NA	\$9.21
	NRC - Disconnect Charge - Add'l	UEPP2	NA	\$7.12	NA	NA	\$9.20	\$13.48	NA	NA	\$8.47
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	\$21.56	\$18.94	NA	\$18.14	\$25.52	\$26.94	NA	\$20.35
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	\$21.56	\$8.42	NA	\$8.06	\$11.34	\$12.76	NA	\$10.54
	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	NA	\$3.84	NA	NA	\$10.39	\$16.07	NA	NA	\$13.32
	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	NA	\$3.84	NA	NA	\$10.39	\$16.07	NA	NA	\$1.40
4-Wii	e DS1 Port w/DID capability, per month	UEPDD	\$130.23	\$63.31	\$120.80	NA	\$149.27	\$146.46	\$123.65	\$130.23	\$120.00
											To be
	NRC - 1st	UEPDD	\$50.00	\$413.93	\$89.44	NA	\$85.63	\$117.81	\$116.59	\$60.00	negotiated
											To be
	NRC - Add'l	UEPDD	\$18.00	\$191.44	\$52.46	NA	\$50.23	\$71.18	\$69.92	\$60.00	negotiated
	NRC - Disconnect Charge - 1st	UEPDD	NA	\$137.29	NA	NA	\$8.82	\$12.94	NA	NA	NA
	NRC - Disconnect Charge - Add'l	UEPDD	NA	\$4.65	NA	NA	\$8.82	\$12.94	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	\$21.56	\$18.94	NA	\$18.14	\$25.52	\$26.94	NA	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	\$21.56	\$8.42	NA	\$8.06	\$11.34	\$12.76	NA	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	NA	\$3.84	NA	NA	\$10.39	\$16.06	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	NA	\$3.84	NA	NA	\$10.39	\$16.06	NA	NA	NA
2-Wii	e ISDN Port(2) (3), per month	U1PMA	\$16.42	\$10.20	\$13.47	\$12.33	\$23.33	\$51.91	\$24.50	\$33.74	NA
$\perp \downarrow \downarrow$	NRC - 1st	U1PMA	\$63.24	\$155.34	\$47.37	\$90.48	\$45.35	\$63.59	\$62.29	\$65.79	NA
	NRC - Add'l	U1PMA	\$63.24	\$106.00	\$47.37	\$84.53	\$45.35	\$63.59	\$62.29	\$65.79	NA
$\bot \bot \bot$	NRC - Disconnect Charge - 1st	U1PMA	\$5.69	\$93.37	NA	NA	\$4.31	\$7.04	NA	NA	NA
$\perp$	NRC - Disconnect Charge - Add'l	U1PMA	\$5.69	\$20.98	NA	NA	\$4.31	\$7.04	NA	NA	NA
$\bot \bot \bot$	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$56.19	\$21.56	\$39.98	NA	\$38.29	\$53.87	\$55.30	\$67.52	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$56.19	\$21.56	\$39.98	NA	\$38.29	\$53.87	\$55.30	\$67.52	NA
$\perp \! \! \perp$	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$12.97	\$3.84	NA	NA	\$6.65	\$11.34	NA	NA	NA
$\perp \perp$	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$12.97	\$3.84	NA	NA	\$6.65	\$11.34	NA	NA	NA
$\perp \! \! \perp \! \! \! \perp$	NRC - User Profile per B Channel (4)	U1UMA	NA	NA	NA	\$5.61	NA	NA	NA	NA	NA
$\perp \!\!\! \perp \!\!\! \perp$											
0 14"	- IODN D((0) (0) !!!!! - - - - - - -	U1PMA	NA	NA	NA	NA	NA	NA	NA	\$38.68	\$18.40
2-Wii	e ISDN Port(2) (3) including all available features, per month				_						
2-Wii	NRC - 1st	U1PMA	NA	NA	NA	NA	NA	NA	NA	\$106.40	\$30.23
2-Wii					NA NA NA	NA NA NA	NA NA NA	NA NA NA	NA NA NA	\$106.40 \$106.40 NA	\$30.23 \$29.49 \$4.10

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
NRC - Disconnect Charge - Add'I	U1PMA	NA	\$4.10							
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	\$67.52	\$41.43						
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	\$67.52	\$42.17						
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	NA	\$9.80							
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	NA	NA NA	NA	NA NA	NA NA	NA	NA NA	NA.	\$9.80
	00									ψ0.00
2-Wire ISDN Port(2) (3) including three available features, per month	U1PMA	NA	\$36.01	NA						
NRC - 1st	U1PMA	NA	\$70.32	NA						
NRC - Add'I	U1PMA	NA	NA NA	NA	NA NA	NA NA	NA	NA.	\$70.32	NA NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	\$67.52	NA						
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	\$67.52	NA						
4-Wire ISDN DS1 Port, per month	UEPEX	\$186.02	\$95.39	\$163.16	NA	\$194.72	\$213.21	\$179.75	\$214.79	\$308.00
NRC - 1st	UEPEX	\$244.85	\$417.51	\$186.80	NA	\$181.89	\$244.12	\$241.63	\$278.37	\$148.66
NRC - Add'l	UEPEX	\$244.85	\$203.18	\$186.80	NA	\$181.89	\$244.12	\$241.63	\$278.37	\$147.18
NRC - Disconnect Charge - 1st	UEPEX	\$51.19	\$149.75	NA	NA	\$27.11	\$53.32	NA	NA	\$38.46
NRC - Disconnect Charge - Add'l	UEPEX	\$51.19	\$37.93	NA	NA	\$27.11	\$53.32	NA	NA	\$36.98
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$54.75	\$21.56	\$37.88	NA	\$33.18	\$51.03	\$53.89	\$65.48	\$40.69
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$54.75	\$21.56	\$37.88	NA	\$33.18	\$51.03	\$53.89	\$65.48	\$42.17
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$11.53	\$3.84	NA	NA	\$7.73	\$8.51	NA	NA	\$9.07
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$11.53	\$3.84	NA	NA	\$7.73	\$8.51	NA	NA	\$10.54
4-Wire ISDN DS1 Port including all available features, per month	UEPEX	NA	NA	NA	\$275.48	NA	NA	NA	\$251.00	NA
NRC - 1st	UEPEX	NA	NA	NA	\$181.27	NA	NA	NA	\$311.73	NA
NRC - Add'l	UEPEX	NA	NA	NA	\$116.42	NA	NA	NA	\$311.73	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	\$65.48	NA						
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	\$65.48	NA						
i i i										
2-Wire Analog Line Port (PBX), per month										
2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence	UEPRD	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.18	\$2.35	\$4.63
LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS	UEPPC	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS	UEPPO	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS	UEPP1	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
LONG DISTANCE TERMINAL PBX TRUNK-BUSINESS	UEPLD	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
TN 2-WAY CALLING PLAN PBX TRUNK - BUSINESS	UEPT2	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
TN OUTWARD CALLING PLAN PBX TRUNK - BUSINESS	UEPTO	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING										
PORT	UEPA2	\$2.07	NA							
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA										
CALLING PORT	UEPL2	NA	NA	NA	NA	\$2.20	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS	UEPLD	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE										
CALLING PORT	UEPT2	NA	\$4.63							
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING										
PORT	UEPTO	NA	\$4.63							
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT	UEPXA	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS	UEPXB	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT	UEPXC	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT	UEPXD	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD										
CAPABLE PORT	UEPXE	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING										
PORT WITHOUT LUD	UEPXF	NA	NA	NA	\$2.61	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT	UEPXG	NA	NA	NA	\$2.61	NA	NA	NA	NA	NA

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT	UEPXH	NA	NA	NA	\$2.61	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT										
WITHOUT LUD	UEPXJ	NA	NA	NA	\$2.61	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL										
CALLING PORT	UEPXK	NA	NA	NA	NA	\$2.20	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY										
ADMINISTRATIVE CALLING PORT	UEPXL	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY				4				4		4
ROOM CALLING PORT	UEPXM	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
O WIDE VOICE LINDUNDLED A WAY OUTCOING DRY HOTEL /HOSDITAL										
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT	LIEDVN	NIA	NIA	NIA	NIA	NIA	NIA	NIA	NA	¢4.62
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL	UEPXN	NA	NA	NA	NA	NA	NA	NA	INA	\$4.63
DIACOUNT ROOM CALLING PORT	UEPXO	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL	UEPAU	\$2.07	\$1.02	\$1.00	\$2.01	\$2.20	ΦΖ.11	\$2.00	φ2.35	φ4.03
DISCOUNT CALLING PORT	UEPXP	NA	NA	NA	NA	\$2.20	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY	OLI XI	INA	INA	INA	INA	Ψ2.20	INA	INA	INA	INA
CALLING PORT	UEPXQ	NA	NA	NA	NA	NA	\$2.11	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL	OLI AQ	14/3	14/3	14/4	14/3	14/3	Ψ2.11	14/3	14/3	14/-1
CALLING PORT	UEPXR	NA	NA	NA	NA	NA	\$2.11	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT	UEPXS	\$2.07	\$1.62	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$4.63
2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS	02.7.0	<b>\$2.01</b>	Ųo2	ψ1.00	ψ2.01	Ψ2.23	Ψ2	Ψ2.00	Ψ2.00	ψσσ
CALLING PORT	UEPXT	NA	NA	NA	NA	NA	NA	NA	\$2.35	NA
									4=.00	
2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT	UEPXU	NA	NA	NA	NA	NA	NA	NA	NA	\$4.63
2-WIRE VOICE UNBUNDLED 2-WAY PBX TENNESSEE REGIONSERV										
CALLING PORT	UEPXV	NA	NA	NA	NA	NA	NA	NA	NA	\$4.63
UNBUNDLED LOOP BILLING USOC (REQUIRES ONE PER PORT)	UEPLX									
<u> </u>										
LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	LNPCP									
NRC - 1st	UEPPC	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence	UEPRD	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$21.60	\$24.36	\$9.93
LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS	UEPPC	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS	UEPPO	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS	UEPP1	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
LONG DISTANCE TERMINAL PBX TRUNK-BUSINESS	UEPLD	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
TN 2-WAY CALLING PLAN PBX TRUNK - BUSINESS	UEPT2	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
TN OUTWARD CALLING PLAN PBX TRUNK - BUSINESS	UEPTO	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING										
PORT	UEPA2	\$21.93	NA							
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA				1				1		
CALLING PORT	UEPL2	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS	UEPLD	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE				ĺ				ĺ		
CALLING PORT	UEPT2	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING	====									
PORT	UEPTO	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT	UEPXA	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS	UEPXB	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT	UEPXC	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT	UEPXD	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD			†		1		0			
CAPABLE PORT	UEPXE	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALL		4=1100	<b>V</b>	******	400	*******	<b>V</b>	<b>4</b> =	4=	70.00
PORT WITHOUT LUD	UEPXF	NA	NA	NA	\$36.47	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT		NA	NA	NA	\$36.47	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT	UEPXH	NA	NA	NA	\$36.47	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT					*******					
	UEPXJ	NA	NA	NA	\$36.47	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL					*******					
CALLING PORT	UEPXK	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY										
ADMINISTRATIVE CALLING PORT	UEPXL	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY		, , , , ,					•			*
ROOM CALLING PORT	UEPXM	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
	-	, , , , ,					•			*
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL										
ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING POR	RT UEPXN	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL										
DIACOUNT ROOM CALLING PORT	UEPXO	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL										
DISCOUNT CALLING PORT	UEPXP	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY	′									
CALLING PORT	UEPXQ	NA	NA	NA	NA	NA	\$22.98	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL										
CALLING PORT	UEPXR	NA	NA	NA	NA	NA	\$22.98	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT	UEPXS	\$21.93	\$62.56	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	\$9.93
2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS										
CALLING PORT	UEPXT	NA	NA	NA	NA	NA	NA	NA	\$24.36	NA
2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING F	ORT UEPXU	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
2-WIRE VOICE UNBUNDLED 2-WAY PBX TENNESSEE REGIONSERV										
CALLING PORT	UEPXV	NA	NA	NA	NA	NA	NA	NA	NA	\$9.93
Subsequent Activity	USASC	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
NRC - Add'l										
2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Reside	nce UEPRD	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$21.60	\$24.36	\$9.19
LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS	UEPPC	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS	UEPPO	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS	UEPP1	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
LONG DISTANCE TERMINAL PBX TRUNK-BUSINESS	UEPLD	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
TN 2-WAY CALLING PLAN PBX TRUNK - BUSINESS	UEPT2	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
TN OUTWARD CALLING PLAN PBX TRUNK - BUSINESS	UEPTO	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALL	ING									
PORT	UEPA2	\$21.93	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA										
CALLING PORT	UEPL2	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS	UEPLD	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE	-									
CALLING PORT	UEPT2	NA	NA	NA	NA	NA	NA	NA	NA	\$9.19
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALL		1		İ	1	1		1	İ	
PORT	UEPTO	NA	NA	NA	NA	NA	NA	NA	NA	\$9.19
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT	UEPXA	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS	UEPXB	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
1 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	1 01.7.0	, ,	, ,	,		,	Ţ	, ,	,	Ţ

DESCRIPTION		USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
2-WIRE VOICE	E UNBUNDLED PBX LD DDD TERMINALS PORT	UEPXC	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
2-WIRE VOICE	E UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT	UEPXD	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
2-WIRE VOICI	E UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD										
CAPABLE PO	RT	UEPXE	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
2-WIRE VOICE	E UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING										
PORT WITHO		UEPXF	NA	NA	NA	\$36.47	NA	NA	NA	NA	NA
	E UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT	UEPXG	NA	NA	NA	\$37.47	NA	NA	NA	NA	NA
	E UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT	UEPXH	NA	NA	NA	\$38.47	NA	NA	NA	NA	NA
	E UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT										
WITHOUT LUI		UEPXJ	NA	NA	NA	\$39.47	NA	NA	NA	NA	NA
	E UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL										
CALLING POF		UEPXK	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
	E UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY				4		4				
	TIVE CALLING PORT	UEPXL	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
	E UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY	==			<b>.</b>						
ROOM CALLIN	NG POR I	UEPXM	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
2 WIDE VOICE	E UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL						ĺ				
		LIEDVN	NIA	NIA	NIA	NIA	NIA	NIA	NIA	NIA	<b>CO 40</b>
	DMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT E UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL	UEPXN	NA	NA	NA	NA	NA	NA	NA	NA	\$9.19
	OOM CALLING PORT	UEPXO	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
	E UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL	UEFAU	φ21.93	\$29.70	\$17.10	φ30.47	\$10.43	\$22.90	\$9.05	φ24.30	φ3.13
	ALLING PORT	UEPXP	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
	E UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY	ULFAF	INA	INA	INA	INA	\$10.43	INA	INA	INA	INA
CALLING POR		UEPXQ	NA	NA	NA	NA	NA	\$22.98	NA	NA	NA
	E UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL	OLI AQ	INA	INA	INA	INA	INA	Ψ22.90	INA	INA	INA
CALLING POF		UEPXR	NA	NA	NA	NA	NA	\$22.98	NA	NA	NA
	E UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT	UEPXS	\$21.93	\$29.70	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	\$9.19
	E UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS	02.70	Ψ2σσ	<b>\$20.1.0</b>	ψ	ψου	ψ.σσ	<b>\$22.00</b>	φοισσ	Ψ21.00	φοιισ
CALLING POF		UEPXT	NA	NA	NA	NA	NA	NA	NA	\$24.36	NA
2-WIRE VOICE	E UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT	UEPXU	NA	NA	NA	NA	NA	NA	NA	NA	\$9.19
2-WIRE VOICE	E UNBUNDLED 2-WAY PBX TENNESSEE REGIONSERV										
CALLING POF	RT	UEPXV	NA	NA	NA	NA	NA	NA	NA	NA	\$9.19
NRC - Discon	nect Charge - 1st										
2 WIRE VOICE	E UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence	UEPRD	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
	IBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS	UEPPC	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
	IBUNDLED OUTWARD PBX TRUNK - BUSINESS	UEPPO	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
	IBUNDLED INCOMING PBX TRUNK - BUSINESS	UEPP1	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
	NCE TERMINAL PBX TRUNK-BUSINESS	UEPLD	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
	LLING PLAN PBX TRUNK - BUSINESS	UEPT2	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
	D CALLING PLAN PBX TRUNK - BUSINESS	UEPTO	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
	E UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING										
PORT		UEPA2	\$6.21	NA	NA	NA	NA	NA	NA	NA	NA
	E UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA					l			l		
CALLING POF		UEPL2	NA 20.01	NA Too o 7	NA	NA	\$3.77	NA 00.50	NA	NA	NA To ac
	E UNBUNDLED PBX LD TERMINAL PORTS	UEPLD	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
	E UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE	LIEDTO	N. A	N. A	N.A	N. A	N. A	N/A	N. A	N/A	<b>#0.00</b>
CALLING POF		UEPT2	NA	NA	NA	NA	NA	NA	NA	NA	\$3.66
	E UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING	LIEDTO	NIA	NIA.	NIA.	l NA	NIA.	NIA	l NA	NIA	<b>#2.00</b>
PORT	E UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT	UEPTO	NA CC 24	NA COC OZ	NA NA	NA NA	NA Co. 77	NA CC FC	NA NA	NA NA	\$3.66
		UEPXA	\$6.21	\$26.37	NA NA	NA NA	\$3.77	\$6.56	NA NA	NA NA	\$3.66
2-WIKE VOICE	E UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS	UEPXB	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66

2-WIRE VOICE UNBUN	DLED PBX LD DDD TERMINALS PORT	UEPXC	\$6.21	<b>#00.07</b>	NIA.	212	**		111		+
1 1 1			Ψ0.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
2-WIRE VOICE UNBLIN	DLED PBX LD TERMINAL SWITCHBOARD PORT	UEPXD	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
	OLED PBX LD TERMINAL SWITCHBOARD IDD										
CAPABLE PORT		UEPXE	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
2-WIRE VOICE UNBUN	OLED 2-WAY PBX KENTUCKY ROOM AREA CALLING	-	•					• • • • • • • • • • • • • • • • • • • •			
PORT WITHOUT LUD		UEPXF	NA	NA	NA	NA	NA	NA	NA	NA	NA
	DLED PBX KENTUCKY LUD AREA CALLING PORT	UEPXG	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUN	DLED PBX KENTUCKY PREMIUM CALLING PORT	UEPXH	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUN	OLED 2-WAY KENTUCKY AREA CALLING PORT	-									
WITHOUT LUD		UEPXJ	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUN	OLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL										
CALLING PORT		UEPXK	NA	NA	NA	NA	\$3.77	NA	NA	NA	NA
2-WIRE VOICE UNBUN	DLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY										1
ADMINISTRATIVE CALL	ING PORT	UEPXL	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
2-WIRE VOICE UNBUN	DLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY										
ROOM CALLING PORT		UEPXM	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
											1
2-WIRE VOICE UNBUN	DLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL										
ECONOMY ADMINIATR	ATIVE CALLING PORTTENNESSEE CALLING PORT	UEPXN	NA	\$26.37	NA	NA	NA	NA	NA	NA	\$3.66
2-WIRE VOICE UNBUN	DLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL										
DIACOUNT ROOM CAL		UEPXO	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
2-WIRE VOICE UNBUN	DLED 1-WAY OUTGOING PBX LOUISIANA LOCAL										
DISCOUNT CALLING P	DRT	UEPXP	\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUN	DLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY										
CALLING PORT		UEPXQ	NA	NA	NA	NA	NA	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUN	DLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL										
CALLING PORT		UEPXR	NA	NA	NA	NA	NA	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUN	DLED 1-WAY OUTGOING PBXMEASURED PORT	UEPXS	\$6.21	\$26.37	NA	NA	\$3.77	\$6.56	NA	NA	\$3.66
2-WIRE VOICE UNBUN	DLED 2-WAY PBX SOUTH CAROLINA AREA PLUS										
CALLING PORT		UEPXT	NA	NA	NA	NA	NA	NA	NA	NA	NA
	DLED PBX COLLIERVILLE & MEMPHIS CALLING PORT	UEPXU	NA	NA	NA	NA	NA	NA	NA	NA	\$3.66
	DLED 2-WAY PBX TENNESSEE REGIONSERV										
CALLING PORT		UEPXV	NA	NA	NA	NA	NA	NA	NA	NA	\$3.66
NRC - Disconnect Char											
	DLED COMBINATION 2-WAY PBX TRUNK - Residence	UEPRD	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
	COMBINATION 2-WAY PBX TRUNK - BUSINESS	UEPPC	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
	O OUTWARD PBX TRUNK - BUSINESS	UEPPO	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
	) INCOMING PBX TRUNK - BUSINESS	UEPP1	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
LONG DISTANCE TERM	IINAL PBX TRUNK-BUSINESS	UEPLD	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
	AN PBX TRUNK - BUSINESS	UEPT2	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
	S PLAN PBX TRUNK - BUSINESS	UEPTO	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
	DLED 2-WAY COMBINATION PBX ALABAMA CALLING										
PORT		UEPA2	\$6.21	NA	NA	NA	NA	NA	NA	NA	NA
	DLED 2-WAY COMBINATION PBX LOUISIANA										
CALLING PORT		UEPL2	NA	NA	NA	NA	\$3.77	NA	NA	NA	NA
	DLED PBX LD TERMINAL PORTS	UEPLD	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
2-WIRE VOICE UNBUN	DLED 2-WAY COMBINATION PBX TENNESSEE										
CALLING PORT		UEPT2	NA	NA	NA	NA	NA	NA	NA	NA	\$2.92
2-WIRE VOICE UNBUN	DLED 1-WAY OUTGOING PBX TENNESSEE CALLING										
I DODT		UEPTO	NA	NA	NA	NA	NA	NA	NA	NA	\$2.92
PORT							*	*			<b>#2.02</b>
1 1 1 -	DLED 2-WAY COMBINATION PBX USAGE PORT	UEPXA	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92

DES	CRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
TĪ	2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT	UEPXC	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
11	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT	UEPXD	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
$\top \top$	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD		****	*****			*****	40.00		1	<del></del>
	CAPABLE PORT	UEPXE	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
	2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING		****	******			*****	40.00			<del></del>
	PORT WITHOUT LUD	UEPXF	NA	NA	NA	NA	NA	NA	NA	NA	NA
++	2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT	UEPXG	NA NA	NA	NA.	NA NA	NA	NA	NA	NA NA	NA
++	2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT	UEPXH	NA NA	NA	NA.	NA NA	NA NA	NA NA	NA	NA NA	NA
++	2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT	OLIAII	14/3	13/3	14/4	14/3	1973	14/3	14/3	14/3	13/3
	WITHOUT LUD	UEPXJ	NA	NA	NA	NA	NA	NA	NA	NA	NA
++	2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL	OLI XI	INA	INA	INA	INA	INA	INA	INA	INA	INA
	CALLING PORT	UEPXK	NA	NA	NA	NA	\$3.77	NA	NA	NA	NA
++	2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY	OLI AIX	14/3	13/3	14/4	14/3	ψ5.77	14/3	14/3	14/3	19/3
	ADMINISTRATIVE CALLING PORT	UEPXL	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
++	2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY	UEFAL	φυ.21	\$1.09	INA	INA	φ3.77	φ0.50	INA	INA	Ψ2.92
	ROOM CALLING PORT	UEPXM	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
+	ROOM CALLING FOR I	UEPAIVI	\$0.∠1	\$1.09	INA	INA	φ3.77	Φ0.00	INA	INA	\$2.92
	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL				1						
		LIEDVAL	N14			NIA		NIA	NIA	NIA	<b>#0.00</b>
++	ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT	UEPXN	NA	NA	NA	NA	NA	NA	NA	NA	\$2.92
	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL	LIEDVO	<b>#0.04</b>	<b>64.00</b>		NI A	<b>#0.77</b>	<b>0</b> 0 50	NIA	NIA	<b>#0.00</b>
++	DIACOUNT ROOM CALLING PORT	UEPXO	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL		4					4			
+	DISCOUNT CALLING PORT	UEPXP	\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
	2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY										
$\bot \bot$	CALLING PORT	UEPXQ	NA	NA	NA	NA	NA	\$6.56	NA	NA	NA
	2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL										
$\perp \perp$	CALLING PORT	UEPXR	NA	NA	NA	NA	NA	\$6.56	NA	NA	NA
	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT	UEPXS	\$6.21	\$1.69	NA	NA	\$3.77	\$6.56	NA	NA	\$2.92
	2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS										
	CALLING PORT	UEPXT	NA	NA	NA	NA	NA	NA	NA	NA	NA
	2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT	UEPXU	NA	NA	NA	NA	NA	NA	NA	NA	\$2.92
	2-WIRE VOICE UNBUNDLED 2-WAY PBX TENNESSEE REGIONSERV										
	CALLING PORT	UEPXV	NA	NA	NA	NA	NA	NA	NA	NA	\$2.92
											-
$\top$	NRC - OSS										
+	NRC - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of										
	the OSS interactive interfaces	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
++	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	\$21.56	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$41.86	\$20.35
++	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	\$21.56	\$8.42	NA NA	\$8.06	\$11.34	\$12.76	\$14.46	\$10.54
++	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	\$3.84	NA	NA NA	\$8.94	\$16.06	NA	NA NA	\$13.32
++	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$0.48	\$3.84	NA NA	NA NA	NA	NA	NA	NA NA	\$1.40
++	1410 Indianicinal offarge - Maridal Getvice Order - Disconlinect - Add I	SOIVIAIN	φυ.40	φ3.0 <del>4</del>	INA	INA	INA	INA	INA	INA	φ1. <del>4</del> 0
2 18"	re Analog Hunting, per line per month	HTGUX	See features	NA	NA	\$0.29	NA	See features	NA	See features	NA
2-771		HTGUX	See features	NA NA	NA NA	\$2.14	NA NA		NA NA	See features	NA NA
++	NRC - 1st NRC - Add'l	HTGUX		NA NA	NA NA	\$2.14 \$2.14	NA NA	See features See features	NA NA	See features See features	NA NA
$+\!+$	INCO - AUU I	HIGUX	See features	INA	NA	<b>⊅</b> ∠.14	INA	see reatures	NA	see reatures	INA
++	Dout you wouth		<b>60.04</b>	<b>C4</b> 00	#0.0F	<b>#2.04</b>	<b>#0.50</b>	<b>#0.00</b>	N.1.4	<b>₾0.77</b>	<b>60.10</b>
Coin	Port, per month		\$2.34	\$1.62	\$2.05	\$3.04	\$2.50	\$2.32	NA	\$2.77	\$2.16
$+\!+$	NRC - 1st		\$21.93	\$4.76	\$17.16	\$40.71	\$16.43	\$22.98	NA	\$24.75	\$9.93
+	NRC - Add'I		\$21.93	\$4.54	\$17.16	\$40.71	\$16.43	\$22.98	NA	\$24.75	\$9.19
$\bot \bot$	NRC - Disconnect Charge - 1st		\$5.21	\$2.76	NA	NA	\$4.15	\$6.56	NA	NA	\$3.66
$\perp \! \! \perp$	NRC - Disconnect Charge - Add'l		\$5.21	\$2.59	NA	NA	\$4.15	\$6.56	NA	NA	\$2.92
Ш	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$25.93	\$21.56	\$18.94	NA	\$18.14	\$25.52	NA	\$43.48	\$20.35
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	\$21.56	\$8.42	NA	\$8.06	\$11.34	NA	\$14.57	\$10.54

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$16.33	\$3.84	NA NA	NA.	\$9.86	\$16.06	NA NA	NA NA	\$13.32
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$0.48	\$3.84	NA.	NA.	NA NA	NA NA	NA NA	NA.	\$1.40
			40.0							******
VERTICAL FEATURES		1								
			No add'l		No add'l					
Local Switching Features offered with Port, Per month	N/A	NA	charge	NA	charge	\$8.28	NA	NA	See above	NA
Three-Way Calling, per month		\$1.12	NA	NA	NA	NA	\$1.32	\$0.89	\$1.10	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Customer Changeable Speed Calling, per month		\$0.08	NA	NA	NA	NA	\$0.0755	\$0.17	\$0.1247	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Call Waiting		\$0.03	NA	NA	NA	NA	\$0.033	\$0.09	\$0.0665	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Remote Activation of Call Fordwarding, per month		\$0.18	NA	NA	NA	NA	\$0.4859	\$0.85	\$0.3743	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Cancel Call Waiting, per month		\$0.01	NA	NA	NA	NA	\$0.0082	\$0.01	\$0.0099	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Automatic Callback, per month		\$0.29	NA	NA	NA	NA	\$0.9977	\$0.66	\$0.8015	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Automatic Recall, per month		\$0.28	NA	NA	NA	NA	\$0.3164	\$0.29	\$0.3102	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Calling Number Delivery, per month		\$0.22	NA	NA	NA	NA	\$0.1817	\$0.33	\$0.3272	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Calling Number Delivery Blocking, per month		\$1.17	NA	NA	NA	NA	\$0.9913	\$0.02	\$0.3684	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Customer Originated Trace, per month		\$0.14	NA	NA	NA	NA	\$0.1918	\$0.14	\$0.1402	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Selective Call Rejection, per month		\$0.13	NA	NA	NA	NA	\$0.1721	\$0.13	\$0.1528	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Selective Call Forwarding, per month		\$0.05	NA	NA	NA	NA	\$0.1050	\$0.28	\$0.1287	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Selective Call Acceptance, per month		\$0.29	NA	NA	NA	NA	\$0.4010	\$0.33	\$0.3283	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Multiline Hunt Service (Rotary)		00.44					#0.40 <del>7</del> :	00.11	00.1007	
Service per line, (in addition to port) , per month		\$0.11	NA	NA	NA	NA	\$0.1271	\$0.14	\$0.1301	NA
NRC NRC NRC NRC NRC NRC NRC NRC NRC NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA To 40	NA Mo ozoo	NA
Call Forwarding Variable, per month		\$0.05	NA	NA	NA	NA	\$0.0474	\$0.10	\$0.0768	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Call Forwarding Busy Line, per month		\$0.03	NA	NA	NA	NA	\$0.0279	\$0.08	\$0.0603	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
NRC - Disconnect		\$0.55	NA	NA NA	NA NA	NA NA	\$0.5466	NA NA	NA NA	NA NA
Call Forwarding Don't Answer All Calls, per month		\$0.03	NA.	NA.	NA NA	NA	\$0.0308	\$0.09	\$0.0655	NA NA
NRC		\$1.03	NA.	NA.	NA.	NA NA	\$1.02	\$1.51	\$1.51	NA NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Remote Call Forwarding, per month		\$1.36	NA	NA.	NA	NA	\$1.47	\$0.95	\$1.41	NA
INRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Call Transfer, per month		\$0.12	NA	NA	NA	NA	\$0.1404	\$0.14	\$0.1392	NA
NRC ,		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Call Hold, per month		\$0.03	NA	NA	NA	NA	\$0.0190	\$0.15	\$0.0677	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC – Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Toll Restricted Service, per month		\$0.04	NA	NA	NA	NA	\$0.0387	\$0.10	\$0.0743	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Message Waiting Indicator – Stutter Dial Tone, per month		\$0.03	NA	NA	NA	NA	\$0.0356	\$0.03	\$0.0318	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Anonymous Call Rejection, per month		\$0.93	NA	NA	NA	NA	\$0.9519	\$1.29	\$1.13	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Shared Call Appearances of a DN, per month		\$0.41	NA	NA	NA	NA	\$0.5015	\$0.29	\$0.3513	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.47	\$1.47	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Multiple Call Appearances, per month		\$0.09	NA	NA	NA	NA	\$0.0932	\$0.07	\$0.0891	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.47	\$1.47	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
ISDN Bridged Call Exclusion, per month		\$0.00	NA	NA	NA	NA	\$0.0013	\$0.0011	\$0.0013	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.47	\$1.47	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Call by Call Access, per month		\$28.29	NA	NA	NA	NA	\$50.89	\$19.83	\$0.3621	NA
NRC		\$28.94	NA	NA	NA	NA	\$28.61	\$33.33	\$33.36	NA
NRC - Disconnect		\$5.22	NA	NA	NA	NA	\$5.16	NA	NA	NA
Privacy Release, per month		\$0.01	NA	NA	NA	NA	\$0.0030	\$0.0041	\$0.0116	NA
NRC NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Multi Appearance Directory Number Calls, per month		\$0.10	NA	NA	NA NA	NA	\$0.1115	\$0.13	\$0.1048	NA
NRC NRC Rissonant		\$1.03	NA	NA		NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA Co coco	NA Co od od	NA
Make Set Busy, per month		\$0.01	NA	NA	NA NA	NA	\$0.0013	\$0.0020	\$0.0101	NA
		\$1.03	NA	NA		NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect   Teen Service (Res. Dist. Alerting Service), per month		\$0.55 \$0.15	NA NA	NA NA	NA NA	NA NA	\$0.5466 \$0.1071	NA \$0.26	NA \$0.2149	NA NA
NRC   NRC					NA NA					
NRC - Disconnect		\$1.03 \$0.55	NA NA	NA NA	NA NA	NA NA	\$1.02 \$0.5466	\$1.51 NA	\$1.51 NA	NA NA
Code Restriction and Diversion, per month		\$0.55	NA NA	NA NA	NA NA	NA NA	\$0.5466	\$0.09	\$0.0708	NA NA
NRC		\$1.03	NA NA	NA NA	NA NA	NA NA	\$0.0464	\$0.09	\$0.0708	NA NA
NRC - Disconnect				NA NA	NA NA	NA NA		\$1.51 NA	\$1.51 NA	NA NA
Call Park, per month		\$0.55 \$0.04	NA NA	NA NA	NA NA	NA NA	\$0.5466 \$0.0443	\$0.09	\$0.0694	NA NA
NRC		\$1.03	NA NA	NA NA	NA NA	NA NA	\$0.0443	\$0.09	\$0.0694	NA NA
NRC - Disconnect		\$0.55	NA NA	NA NA	NA NA	NA NA	\$0.5466	NA \$0.14	NA \$0.1179	NA NA
Automatic Line, per month		\$0.09	1				\$0.1111			
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
2-WIRE ISDN BRI FEATURES										
Shared Primary Number-First Appr On Each Add'l Terminal	DS1FJ	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Secondary Only Dn (Shared/Non-Shared) First Appearance	LLDSF	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Shared Secondary Only Dn-First Appr On Each Add'l Term	DS1F1	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Shared Non-ISDN DN	DOE	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Privacy Release	DS1FU	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Manual Exclusion	DS1FM	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Variable-Voice Or Voice/Data	LLNCV	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Variable – Data	LLOCD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Variable – Feature Button – Voice	GJXCF	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Variable – Feature Button – Data	LLPCD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Busy Line – Voice Or Voice/Data	LLQCV	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Busy Line – Data	LLRCD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Frwdng Busy Line-Prgrmmbl-Voice Or Voice/Data	M6AVA	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Busy Line – Programmable - Data	M6ADF	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Don't Answer – Voice Or Voice/Data	LLSCV	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Don't Answer – Data	LLUCD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwdng Don't Answer–Prgrmmble Voice Or Voice/Data	M6BVA	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Don't Answer – Programmable - Data	M6BDF	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Frwdng Multiple Simultaneous – Voice Or Voice/Data	M6CV5	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Multiple Simultaneous – Data	M6CD5	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Conference, Drop, Hold And Transfer	DS1FN	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Six-Way Conference, Drop, Hold And Transfer	LLY6P	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Multi-Line Hunt Group – Voice Or Voice/Data	HTG	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Multi-Line Hunt Group – Data	HTGSD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Speed Calling	LLZSU	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Visual Message Waiting Indicator	LLAVP	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Audible Message Waiting Indicator	MWW	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Additional Call Appearance, PDN Or DN	DS1FG	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Tracing	NST	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Return	NSS	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Preferred Call Forwarding	NCE	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Block	NSY	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Repeat Dialing	NSQ	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Per Line Blocking For Agencies/Law Enforcement	NOB	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Per Line Blocking For Non-Pub Customers	NOBNN	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Per Line Blocking For General Public	NOBPC	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Per Line Blocking For Non-Pub, And Non-Listed Customer	NOBPP	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Per Line Blocking For Non-Pub Customers	NOBNP	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Per Line Blocking For Non-Pub Customers	NOBNR	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Return Denial Of, Per Activation	BCR	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Repeat Dialing, Denial Of, Per Activation	BRD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Automatic Line/Direct Connect	M6GN9	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Make Set Busy	M6MPD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Selective Call Acceptance	M6K16	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Park/Call Retrieve	M6HP6	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Transfer System Exception	M6QTD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Make Set Busy – Intragroup	M6MGD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
All Customized Code Restrictions	CREX+	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Additional Listings	CLT	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Additional Listing No Rate	FLT	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Cross Reference Listing	LLT	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Non-Pub Listing No Rate	NP3	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Non-List Listing	NLT	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
Non-List Listing No Rate	NLE	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Alternate Call Listing	FNA	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Manual Service Order Charge	SOMAN	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
All Selective Class Of Call Screening	SRG++	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
ISDN Message Waiting Indication-Lamp, per month		\$0.01	NA	NA	NA	NA	\$0.0105	\$0.0107	\$0.0138	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.47	\$1.47	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
ISDN Feature Function Buttons		NA	NA	NA	NA	NA	NA	NA		
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Subsequent Ordering Charge – (per order, per line)		NA	NA	NA	NA	NA	NA	NA		
NRC - Electronic - 1st		\$2.88	NA	NA	NA	NA	\$2.84	\$5.42	\$1.36	NA
NRC - Electronic - Add'I		\$0.96	NA	NA	NA	NA	\$0.95	\$0.95	\$0.71	NA
NRC - Manual - 1st		\$4.80	NA	NA	NA	NA	\$4.73	\$1.89	\$7.35	NA
NRC - Manual - Add'l		\$0.96	NA	NA	NA	NA	\$0.95	NA	\$0.95	NA
NRC - Disconnect		\$2.88	NA	NA	NA	NA	\$2.84	NA	NA	NA
Unbundled Port Usage Charges										
End Office Switching (Port Usage)										
End Office Switching Function, per mou	N/A	\$0.0018	\$0.0008846	\$0.0016333	\$0.002562	\$0.0021	\$0.0023771	\$0.0017000	\$0.0019295	\$0.0008
End Office Interoffice Trunk Port—Shared, per mou	N/A	\$0.0002		\$0.0001564	NA	\$0.0002	\$0.0001927	NA	\$0.0002581	NA
Tandem Switching (Port Usage) (Local or Access Tandem)										
Tandem Switching Function per mou	N/A	\$0.00063	\$0.0001522	\$0.0006757	\$0.001096	\$0.0008	\$0.0007834	\$0.0009	\$0.0006843	\$0.000978
Tandem Interoffice Trunk Port - Shared per mou			\$0.0002713	\$0.0002126	NA	\$0.0003	\$0.0002834	NA	\$0.0004034	NA
Common (Shared) Transport										
Common (Shared) Transport per mile per mou	N/A	\$0.00001		\$0.000008		\$0.0000083	\$0.0000091	\$0.00001	\$0.0000121	\$0.000064
Common (Shared) Transport Facilities Termination per mou	N/A	\$0.00045	\$0.0004579	\$0.0004152	\$0.000426	\$0.00047	\$0.0004281	\$0.00034	\$0.0004672	\$0.0003871
NOTES:										
Interim rates subject to true-up.										
1 Port rate includes all available features.										
2 Transmission/usage charges associated with POTS circuit switched usage will										
also apply to circuit switched voice and/or circuit switched data transmission by B-										
Channels associated with 2-wire ISDN ports.										
Access to B Channel or D Channel Packet capabilities will be avail- able only										
through BFR/New Business Request Process. Rates for the packet capabilities										
will be determined via the Bona Fide Request/New Business Request Process.										
4 This rate element is for those states which have a specific rate for User Profile per										
B Channel.			<b>_</b>							
5 This rate element is for use in those states with a different rate for additional									1	
minutes of use.		1	1						-	
6 Rates in TN and FL are interim and shall be trued-up when final rates are ordered.		<u> </u>								
									1	

	UNBUNDLED DEDICATED TRANSPORT - Local Channel	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	Local Channel - Dedicated - 2-Wire VG										
	2-wire VG per mile	1L5NC	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	2-wire VG Monthly Recurring per month	ULDV2	\$14.61	\$29.33	\$13.91	\$22.26	\$14.94	\$17.83	\$14.83	\$16.83	\$19.02
	NRC - 2-wire VG - 1st	ULDV2	\$494.65	\$386.34	\$382.95	\$585.15	\$347.49	\$487.62	\$553.80	\$554.00	\$199.33
	NRC - 2-wire VG -Add'l	ULDV2	\$84.44	\$66.36	\$62.40	\$98.53	\$59.75	\$84.35	\$89.69	\$88.58	\$24.16
	NRC - 2-Wire VG - Disconnect Chg - 1st	ULDV2	\$77.81	\$67.91	NA	NA	\$53.68	\$77.69	NA	NA	\$54.81
	NRC - 2-Wire VG - Disconnect Chg - Add'l	ULDV2	\$7.63	\$5.92	NA	NA	\$6.60	\$8.95	NA	NA	\$4.80
	NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	\$19.99	NA	NA	NA	NA	\$19.99
	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire VG - Incremental ChargeManual Svc Order - 1st	SOMAN	\$27.37	NA	\$18.94	\$41.46	\$18.14	\$25.50	\$42.17	\$43.75	NA
	NRC - 2-Wire VG - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$18.37	NA	\$8.42	\$11.99	\$8.06	\$11.34	\$12.76	\$13.55	NA
	NRC - 2-Wire VG - Incremental ChargeManual Svc Order-Disconnect	SOMAN	\$17.75	NA	NA	NA	\$11.40	\$16.05	NA	NA	NA
	Local Channel - Dedicated - 2-Wire VG Rev. Bat.										
ШΤ	2-wire VG per mile	1L5NC	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	2-wrire VG Monthly Recurring per month	ULDV4	\$14.61	\$29.33	\$13.91	\$22.26	\$14.94	\$17.83	\$14.83	\$16.83	\$19.02
$\Box$	NRC - 2-wire VG - 1st	ULDV4	\$494.65	\$3,865.34	\$382.95	\$585.15	\$347.49	\$487.62	\$553.80	\$554.00	\$199.33
	NRC - 2-wire VG -Add'l	ULDV4	\$84.44	\$66.36	\$62.40	\$98.53	\$59.75	\$84.35	\$89.69	\$88.58	\$24.16
	NRC - 2-Wire VG - Disconnect Chg - 1st	ULDV4	\$77.81	\$67.91	NA	NA	\$53.68	\$77.69	NA	NA	\$54.81
	NRC - 2-Wire VG - Disconnect Chg - Add'l	ULDV4	\$7.63	\$5.92	NA	NA	\$6.60	\$8.95	NA	NA	\$4.80
	NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	\$19.99	NA	NA	NA	NA	\$19.99
	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire VG - Incremental ChargeManual Svc Order - 1st	SOMAN	\$27.37	NA	\$18.94	\$41.46	\$18.14	\$25.50	\$42.17	\$43.75	NA
	NRC - 2-Wire VG - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$18.37	NA	\$8.42	\$11.99	\$8.06	\$11.34	\$12.76	\$13.55	NA
	NRC - 2-Wire VG - Incremental ChargeManual Svc Order-Disconnect	SOMAN	\$17.75	NA	NA	NA	\$11.40	\$16.05	NA	NA	NA
	Local Channel - Dedicated - 4-Wire VG										
	4-wire VG per mile	1L5NC	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	4-wire VG Monthly Recurring per month	ULDV4	\$15.77	\$30.50	\$14.99	\$23.38	\$16.21	\$19.03	\$15.87	\$18.05	\$20.14
	NRC - 4-Wire VG - 1st	ULDV4	\$502.43	\$387.21	\$368.44	\$585.15	\$352.75	\$495.25	\$562.23	\$562.46	\$201.53
	NRC - 4-Wire VG - Add'l	ULDV4	\$86.68	\$67.22	NA	\$98.53	\$61.33	\$86.56	\$92.67	\$91.57	\$24.83
	NRC - 4-Wire VG - Disconnect Chg - 1st	ULDV4	\$78.71	\$68.78	NA	NA	\$54.36	\$78.58	NA	NA	\$55.52
	NRC - 4-Wire VG - Disconnect Chg - Add'l	ULDV4	\$8.53	\$6.79	NA	NA	\$7.28	\$9.84	NA	NA	\$5.51
++	NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	\$19.99	NA	NA	NA	NA	\$19.99
++	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	NA
++	NRC - Electronic Svc Order, per LSR	SOMEC	NA	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
++	NRC - Electronic Svc Order, per LSR disconnect	SOMEC	\$3.50	\$0.42	NA	NA	NA	NA Pos. so	NA 0 10 17	NA	NA
HH	NRC - 4-Wire VG - Incremental ChargeManual Svc Order - 1st	SOMAN	\$27.37	NA	\$18.94	\$41.46	\$18.14	\$25.52	\$42.17	\$43.64	NA
$\vdash \vdash$	NRC - 4-Wire VG - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$18.73	NA	\$8.42	\$11.99	\$8.06	\$11.34	\$12.76	\$13.55	NA
$\vdash \vdash$	NRC - 4-Wire VG - Incremental ChargeManual Svc Order-Disconnect	SOMAN	\$17.75	NA	NA	NA	\$11.40	\$17.25	NA	NA	NA
$\vdash \vdash$	Level Observed Budderford DO4										
$\vdash\vdash$	Local Channel - Dedicated - DS1	41.5110	<b>#0.00</b>	<b>#</b> 0.00	<b>#</b> 0.00	#0.00	#0.00	<b>#0.00</b>	<b>#0.00</b>	<b>#</b> 0.00	<b>#0.00</b>
$\vdash\vdash$	DS1 per mile	1L5NC	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
$\vdash\vdash$	DS1 per month	ULDF1	\$35.52	\$43.53	\$38.36	\$43.80	\$43.80	\$38.91	\$35.68	\$37.20	\$40.27
$\vdash\vdash$	NRC - DS1 - 1st	ULDF1	\$503.57	\$242.45	\$356.15	\$538.95	\$348.56	\$494.83	\$534.48	\$534.81	\$277.35
++	NRC - DS1 - Add'l	ULDF1	\$442.84	\$226.44	\$312.89	\$464.94	\$300.30	\$435.28	\$462.69	\$462.81	\$233.26
++	NRC - DS1 - Disconnect Chg - 1st	ULDF1	\$46.28	\$41.13	\$122.31	NA NA	\$24.15	\$46.85	NA NA	NA	\$33.18
++	NRC - DS1 - Disconnect Chg - Add'l	ULDF1	\$32.18	\$28.28	\$119.14	NA ©40.00	\$21.31	\$33.02	NA NA	NA	\$22.30
++	NRC - Manual Svc Order, per LSR	SOMAN	NA NA	\$21.56	NA NA	\$19.99	NA NA	NA NA	NA NA	NA	\$19.99
++	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA ©2.50	\$3.84		NA ©2.50				NA Co. FO	NA CO. FO
$\Box$	NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50

MRC   Electronic Six Orders, part 158 (decorrent)   SOMAN												
NRC - DS1 - Incommental Charges—Manual Size Order - 1411 SOMAN S0.00 NA NA NA NA NA S177 S3.11 NA NA NA NA NA S177 S3.11 NA NA NA NA S177 S3.11 NA NA NA NA S177 S3.11 NA NA NA NA S177 S3.11 NA NA NA NA S177 S3.11 NA NA NA NA S177 S3.11 NA NA NA NA S177 S3.11 NA NA NA NA S177 S3.11 NA NA NA NA S177 S3.11 NA NA NA NA NA NA NA NA NA NA NA NA NA	$\vdash$	NPC - Flectronic Syc Order, per LSP disconnect	SOMEC	NΔ	\$0.42	NΙΔ	NΙΔ	NΙΔ	NΔ	NΙΔ	NΙΔ	NΙΔ
NRC - DS1 - Incremental ChargeManual Svc Order - Add   SOMAN   SOMAN   SOMAN   NA   NA   NA   S14.08   \$27.70   NA   NA   NA   NA   NA   NA   S14.08   \$27.70   NA   NA   NA   NA   NA   NA   NA   N	++											
NRC - DS1 - Incornemental Charges—Name   SOLAMN   S0.22   NA   S10.00   S0.04   S0.04   S0.05   S0.04   S0.05   S0.04   S0.05   S0.04   S0.05   S0.04   S0.05   S0.04   S0.05   S0.0	++											
Local Channel - Dedicated - D93	++											
OSS-per mile per month	+++	NRC - DST - Incremental ChargeManual SVC Order-Disconnect	SOMAN	φ29.21	INA	\$16.03	INA	\$19.40	\$27.40	INA	INA	INA
OSS-per mile per month	H	Local Channel - Dedicated - DS3										1
DS3 - Facility Termination per month			1L5NC	\$8.44	\$9.16	\$6.92	\$34.00	\$30.34	NA	NA	\$12.08	\$23.76
NRC - DS3 - Facility Termination - 14st												
NRC - DS3 - Facility Termination - Disconnect - 1st	+++											
NRC - C93 - Facility Termination - Disconnect - 14st												
NRC - DS3 - Facility Termination - Disconnect - AddT   ULDFS   S118.54   S14.90   S119.14   NA   S99.46   S40.87   NA   NA   NA   NA   NA   NA   NA   N												
NRC - Manual Svo Order, per LSR disconnent   SOMAN NA   \$319.99   NA   NA   NA   NA   NA   NA   NA												
NRC - Manual Svc Order, per LSR disconnect   SOMAN   NA   S3.34   NA   NA   NA   NA   NA   NA   NA   N												\$19.99
NRC - Electronic Svc Order, per LSR   SOMEC   S3.50											NA	NA
NRC - Electronic Swc Order, pet LSR disconnect   SOMEC   NA   S0.42   NA   NA   NA   NA   NA   NA   NA   N												
NRC - DS3 - Incremental ChargeManual Svc Order - 1st												
NRC - DS3 - Incremental Charge-Manual Swc Order - Add1	+++											
NRC - DSS - Incremental Charge - Manual Svc Order-Disconnect - 181   SOMAN   S19.03   NA   S18.03   NA   \$20.94   \$25.35   NA   NA   NA   NA   NA   NA   NA   N												
NRC - DS3 - Incremental Charge-Manual Svc Order-Disconnect-Add1   SOMAN   \$19.03   NA   \$18.03   NA   \$20.94   \$25.35   NA   NA   NA   NA   NA   NA   NA   N	+++											
Local Channel - Dedicated - STS-1   1L5NC   \$8.44   \$9.16   \$6.92   \$34.00   \$9.77   \$38.98   NA   \$12.08   \$25.11	H											
STS-1- per mile per month	H	Titto Doo moremental entarge manual eve erael Discombet raar	00.00	ψ.σ.σσ		ψ.σ.σσ		Ψ20.0.	Ψ20.00			
STS-1- per mile per month	H	Local Channel - Dedicated - STS-1										
STS-1- Facility Termination per month	H		1L5NC	\$8 44	\$9.16	\$6.92	\$34.00	\$8.77	\$38.98	NA	\$12.08	\$25.11
NRC - ST5-1 - Facility Termination - 1st	++											
NRC - STS-1 - Facility Termination - Obsconnect - 1st	++											
NRC - STS-1 - Facility Termination - Disconnect - 1st   ULDFS   \$121.7Z   \$221.46   \$122.31   NA   \$113.75   \$42.41   NA   NA   \$103.36	+++											
NRC - STS-1 - Facility Termination - Disconnect - Add'  ULDFS   \$118.54   \$154.90   \$119.14   NA   \$110.80   \$40.87   NA   NA   \$10.59   NRC - Manual Svc Order, per LSR   SOMAN   NA   \$21.56   NA   \$19.99   NA   NA   NA   NA   NA   NA   NA	++											
NRC - Manual Svc Order, per LSR	++	,										
NRC - Manual Svc Order, per LSR disconnect   SOMAN   NA   S3.84   NA   NA   NA   NA   NA   NA   NA   N	+											
NRC - Electronic Svc Order, per LSR   SOMEC   \$3.50	+											
NRC - Electronic Svc Order, per LSR disconnect   SOMEC   NA   \$0.42   NA   NA   NA   NA   NA   NA   NA   N	++											
NRC - STS-1 - Incremental ChargeManual Svc Order - 1st   SOMAN   \$38.48   NA   \$37.55   \$93.12   \$34.92   \$96.10   \$56.25   \$54.26   NA   NRC - STS-1 - Incremental ChargeManual Svc Order- Add'  SOMAN   \$38.48   NA   \$37.55   \$93.12   \$34.92   \$96.10   \$56.25   \$54.26   NA   NRC - STS-1 - Incremental ChargeManual Svc Order-Disconnect -1st   SOMAN   \$19.03   NA   \$18.03   NA   \$16.77   \$25.35   NA   NA   NA   NA   NA   NA   NA   N	++											
NRC - STS-1 - Incremental ChargeManual Svc Order - Add'  SOMAN \$38.48 NA \$37.55 \$93.12 \$34.92 \$96.10 \$56.25 \$54.26 NA NRC - STS-1 - Incremental ChargeManual Svc Order-Disconnect-1st SOMAN \$19.03 NA \$18.03 NA \$16.77 \$25.35 NA NA NA NA NA NA NA NA NA NA NA NA NA	++											
NRC - STS-1 - Incremental Charge - Manual Svc Order-Disconnect - 1st   SOMAN   \$19.03   NA   \$18.03   NA   \$16.77   \$25.35   NA   NA   NA   NA   NRC - STS-1 - Incremental Charge - Manual Svc Order-Disconnect-Add'  SOMAN   \$19.03   NA   \$18.03   NA   \$16.77   \$25.35   NA   NA   NA   NA   NA   NA   NA   N	+											
NRC - STS-1 - Incremental ChargeManual Svc Order-Disconnect-Add'  SOMAN \$19.03 NA \$18.03 NA \$16.77 \$25.35 NA NA NA NA NA NA NA NA NA NA NA NA NA	++											
Local Channel - Dedicated - OC3	++											
OC3 per mile per month	H	1410 010 1 Indicational Charge Manage CVC Order Discontinue Add 1	0011111111	Ψ10.00	101	ψ10.00	147.	Ψ10.77	Ψ20.00	100	10.	101
OC3 Facility Termination per month		Local Channel - Dedicated - OC3										1
NRC - OC3 - Facility Termination - 1st		OC3 per mile per month	1L5NC	\$7.09	\$8.93	\$5.82	\$28.56	\$25.48	\$35.55	\$21.27	\$10.15	\$19.95
NRC - OC3 - Facility Termination - 1st				\$1,123	\$648.60	\$914.22	\$1,493	\$1,179	\$873.23	\$914.18	\$493.31	\$1,263
NRC - OC3 - Facility Termination - Add'    NRC - OC3 - Facility Termination - Disconnect Chg - 1st   \$11.38   \$408.85   \$413.00   \$661.23   \$402.63   \$549.17   \$670.92   \$519.31   \$411.64     NRC - OC3 - Facility Termination - Disconnect Chg - 1st   \$121.72   \$11.56   \$122.31   NA   \$102.16   \$134.07   NA   NA   \$103.36     NRC - OC3 - Facility Termination - Disconnect Chg - Add'    \$118.54   \$108.34   \$119.14   NA   \$99.46   \$130.59   NA   NA   \$100.59     NRC - Manual Svc Order, per LSR   SOMAN   NA   \$21.56   NA   \$19.99   NA   NA   NA   NA   NA   NA   NA	ПТ											
NRC - OC3 - Facility Termination - Disconnect Chg - Add'l   \$118.54 \$108.34 \$119.14 NA \$99.46 \$130.59 NA NA \$100.59 NRC - Manual Svc Order, per LSR   SOMAN NA \$21.56 NA \$19.99 NA NA NA NA NA NA NA NA NA NA NA NA NA	ПТ											
NRC - Manual Svc Order, per LSR   SOMAN NA   \$21.56 NA   \$19.99 NA NA NA NA NA NA NA NA NA NA NA NA NA		NRC - OC3 - Facility Termination - Disconnect Chg - 1st		\$121.72	\$11.56	\$122.31	NA	\$102.16	\$134.07	NA	NA	\$103.36
NRC - Manual Svc Order, per LSR   SOMAN NA   \$21.56 NA   \$19.99 NA NA NA NA NA NA NA NA NA NA NA NA NA		NRC - OC3 - Facility Termination - Disconnect Chg - Add'l		\$118.54	\$108.34	\$119.14	NA	\$99.46	\$130.59	NA	NA	\$100.59
NRC - Manual Svc Order, per LSR disconnect   SOMAN   NA   \$3.84   NA   NA   NA   NA   NA   NA   NA   N			SOMAN	NA	\$21.56		\$19.99			NA	NA	
NRC - Electronic Svc Order, per LSR			SOMAN	NA	\$3.84							
NRC - OC3 - Incremental ChargeManual Svc Order - 1st   SOMAN   \$38.48   NA   \$37.55   \$93.12   \$50.25   \$68.62   \$99.10   \$54.26   NA   NRC - OC3 - Incremental ChargeManual Svc Order - Add'l   SOMAN   \$38.48   NA   \$37.55   \$93.12   \$50.25   \$68.62   \$99.10   \$54.26   NA   NRC - OC3 - Incremental ChargeManual Svc Order-Disconnect-1st   SOMAN   \$19.03   NA   \$18.03   NA   \$20.94   \$28.59   NA   NA   NA   NA   NA   NA   NA   N		NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
NRC - OC3 - Incremental ChargeManual Svc Order - 1st   SOMAN   \$38.48   NA   \$37.55   \$93.12   \$50.25   \$68.62   \$99.10   \$54.26   NA   NRC - OC3 - Incremental ChargeManual Svc Order - Add'l   SOMAN   \$38.48   NA   \$37.55   \$93.12   \$50.25   \$68.62   \$99.10   \$54.26   NA   NRC - OC3 - Incremental ChargeManual Svc Order-Disconnect-1st   SOMAN   \$19.03   NA   \$18.03   NA   \$20.94   \$28.59   NA   NA   NA   NRC - OC3 - Incremental ChargeManual Svc Order-Disconnect-Add'l   SOMAN   \$19.03   NA   \$18.03   NA   \$20.94   \$28.59   NA   NA   NA   NA   NA   NA   NA   N		NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42		NA	NA	NA	NA	NA	NA
NRC - OC3 - Incremental ChargeManual Svc Order - Add'l   SOMAN   \$38.48   NA   \$37.55   \$93.12   \$50.25   \$68.62   \$99.10   \$54.26   NA   NRC - OC3 - Incremental ChargeManual Svc Order-Disconnect-1st   SOMAN   \$19.03   NA   \$18.03   NA   \$20.94   \$28.59   NA   NA   NA   NA   NA   NA   NA   N			SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$99.10	\$54.26	NA
NRC - OC3 - Incremental ChargeManual Svc Order-Disconnect-Add'l SOMAN \$19.03 NA \$18.03 NA \$20.94 \$28.59 NA NA NA NA NA NA Local Channel - Dedicated - OC12    OC12 per mile per month			SOMAN	\$38.48	NA		\$93.12	\$50.25	\$68.62	\$99.10	\$54.26	NA
NRC - OC3 - Incremental ChargeManual Svc Order-Disconnect-Add'l SOMAN \$19.03 NA \$18.03 NA \$20.94 \$28.59 NA NA NA NA NA NA Local Channel - Dedicated - OC12    OC12 per mile per month	ПТ		SOMAN	\$19.03	NA		NA	\$20.94	\$28.59	NA	NA	NA
Local Channel - Dedicated - OC12			SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	NA	NA	NA
OC12 per mile per month 1L5NC \$10.13 \$2,053.06 \$8.31 \$40.80 \$36.40 \$50.79 \$30.38 \$14.50 \$28.51		· ·										
OC12 per mile per month 1L5NC \$10.13 \$2,053.06 \$8.31 \$40.80 \$36.40 \$50.79 \$30.38 \$14.50 \$28.51		Local Channel - Dedicated - OC12										1
		OC12 per mile per month	1L5NC	\$10.13	\$2,053.06	\$8.31	\$40.80	\$36.40	\$50.79	\$30.38	\$14.50	\$28.51
							\$4,492					

Attachment 2

Exhibit C Rates - Page 42

			1				1	Т	1	1	
	NRC - OC12 - Facility Termination - 1st		\$1,165	\$408.85	\$1,162.00	\$1,858	\$1,245	\$1,742.00	\$1,853	\$1,259	\$1,276
	NRC - OC12 - Facility Termination - Add'l		\$413.38	\$111.56	\$413.00	\$661.23	\$402.63	\$549.17	\$670.92	\$505.88	\$411.64
	NRC - OC12 - Facility Termination -Disconnect Chg - 1st		\$121.72	\$108.34	\$122.31	NA	\$102.16	\$134.07	NA	NA	\$103.36
	NRC - OC12 - Facility Termination - Disconnect Chg - Add'l		\$118.54	\$21.56	\$119.14	NA	\$99.46	\$130.59	NA	NA	\$100.59
	NRC - Manual Svc Order, per LSR	SOMAN	NA	\$3.84	NA	\$19.99	NA	NA	NA	NA	\$19.99
	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$2.75	NA	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$0.42	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.43	NA	NA	NA	NA	NA	NA	NA
	NRC - OC12 - Incremental ChargeManual Svc Order - 1st	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$99.10	\$54.26	NA
	NRC - OC12 - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$99.10	\$54.26	NA
	NRC - OC12 - Incremental ChargeManual Svc Order-Disconnect-1st	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	NA	NA	NA
	NRC - OC12 - Incremental ChargeManual Svc Order-Disconnect-Add'l	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	NA	NA	NA
	Local Channel - Dedicated - OC48										
	OC48 per mile per month	1L5NC	\$33.22	\$36.04	\$27.25	\$133.84	\$119.40	\$166.59	\$99.66	\$47.57	\$93.50
	OC48 Facility Termination per month		\$1,947	\$1,685.97	\$1,671	\$2,156	\$2,311	\$1,768.00	\$1,837	\$1,842	\$1,853
	OC48 - Interface OC12 on OC48 per month		\$699.62	\$587.71	\$558.07	\$728.81	\$706.85	\$668.36	\$584.78	\$773.40	\$572.61
	NRC - OC48 - Facility Termination - 1st		\$1,165	\$1,183.46	\$1,162.00	\$1,858	\$1,245	\$1,742.00	\$1,853	\$1,259	\$1,276
	NRC - OC48 - Facility Termination -Add'l		\$413.38	\$408.85	\$413.00	\$661.23	\$402.63	\$549.17	\$670.92	\$505.88	\$411.64
	NRC - OC48 - Interface OC12 on OC48 - 1st		\$540.10	\$543.72	\$539.36	\$844.21	\$532.13	\$729.04	\$852.47	\$635.04	\$544.55
	NRC -OC48 - Interface OC12 on OC48 -Add'l		\$317.48	\$312.05	\$317.38	\$516.89	\$304.90	\$404.94	\$528.57	\$410.02	\$311.39
	NRC - OC48 - Facility Termination - Disconnect Chg - 1st		\$121.72	\$111.56	\$122.31	NA	\$102.16	\$134.07	NA	NA	\$103.36
	NRC - OC48 - Facility Termination - Disconnect Chg - Add'l		\$118.54	\$108.34	\$119.14	NA	\$99.46	\$130.59	NA	NA	\$100.59
	NRC - OC48 - Interface OC12 on OC48 - Disconnect Chg - 1st		\$121.72	\$111.56	\$122.31	NA	\$102.16	\$134.07	NA	NA	\$103.36
	NRC - OC48 - Interface OC12 on OC48 - Disconnect Chg - Add'l		\$118.54	\$108.34	\$119.14	NA	\$99.46	\$130.59	NA	NA	\$100.59
	NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	\$19.99	NA	NA	NA	NA	\$19.99
++	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
++	NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
	NRC - OC48 - Incremental ChargeManual Svc Order - 1st	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$99.10	\$54.26	NA
++	NRC - OC48 - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$38.48	NA NA	\$37.55	\$93.12	\$50.25	\$68.62	\$99.10	\$54.26	NA
++	NRC - OC48 -Interface-Incremental Cost-Manual Svc. Order vs. Electronic-1	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$99.10	\$54.26	NA
++-	NRC - OC48 -Interface-Incremental Cost-Manual Svc. Order vs. Electronic-A	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$99.10	\$54.26	NA
++-+	NRC - OC48 - Incremental ChargeManual Svc Order-Disconnect-1st	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	NA	NA	NA
++	NRC - OC48 - Incremental ChargeManual Svc Order-Disconnect-Add'l	SOMAN	\$19.03	NA	\$18.03	NA NA	\$20.94	\$28.59	NA	NA	NA NA
++-+	NRC - OC48 -Interface-Incremental Cost-Manual Svc. Order vs. Electronic-D		\$19.03	NA NA	\$18.03	NA NA	\$20.94	\$28.59	NA NA	NA NA	NA NA
++	NRC - OC46 -Interface-Incremental Cost-Manual Svc. Order vs. Electronic-D	SOMAN	\$19.03	NA NA	\$18.03	NA NA	\$20.94	\$28.59	NA NA	NA NA	NA NA
++	NRC - 0046 -Interface-incremental cost-Marida 3vc. Order vs. Electronic-D	SOMAN	\$19.03	INA	φ10.03	INA	\$20.94	\$20.59	INA	INA	INA
++	UNBUNDLED DEDICATED TRANSPORT - Interoffice Channel										
++	Interoffice Transport - Dedicated - 2-wire VG										
++-+	2-Wire VG - per mile per month	1L5XX	\$0.0339	\$0.0098	\$0.0222	\$0.0301	\$0.0384	\$0.0323	\$0.0282	\$0.0373	\$0.0173
++-+	2-Wire VG - Facility Termination per month	U1TV2	\$18.49	\$26.52	\$17.07	\$27.66	\$19.10	\$21.33	\$18.01	\$21.42	\$18.33
H	NRC - 2-wire VG - Facility Termination -1st	U1TV2	\$107.11	\$81.09	\$79.61	\$142.31	\$76.20	\$106.72	\$137.48	\$136.44	\$55.39
H	NRC - 2-wire VG - Facility Termination - 1st  NRC - 2-wire VG - Facility Termination - Add'l	U1TV2	\$48.27	\$54.83	\$36.08	\$142.31	\$76.20	\$48.83	\$52.58	\$136.44	\$17.37
H		U1TV2			\$36.08 NA	\$56.21 NA					\$17.37
++	NRC - 2-wire VG - Facility Termination - Disconnect Charge -1st	U1TV2	\$37.16 \$5.88	\$31.01 \$12.78	NA NA	NA NA	\$28.03 \$5.37	\$38.05 \$7.23	NA NA	NA NA	\$27.96
H	NRC - 2-wire VG - Facility Termination - Disconnect Charge -Add'l									NA NA	
H +	NRC - Manual Svc Order, per LSR	SOMAN	NA NA	\$21.56	NA	\$19.99	NA NA	NA NA	NA NA		\$19.99
HH	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA Co.50	\$3.84	NA To 50	NA TO 50	NA To 50	NA Co. FO	NA TO 50	NA To 50	NA Co. 50
HH	NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
H	NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA 005.50	NA	NA Tables	NA
++	NRC - 2-wire VG - Incremental ChargeManual Svc Order - 1st	SOMAN	\$27.37	NA	\$18.94	\$37.21	\$18.14	\$25.52	\$38.07	\$39.63	NA
++	NRC - 2-wire VG - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$27.57	NA	\$18.94	\$37.21	\$18.14	\$25.52	\$38.07	\$39.63	NA
$\coprod$	NRC - 2-wire VG - Incremental ChargeManual Svc Order-Disconnect1st	SOMAN	\$12.97	NA	NA	NA	\$8.06	\$11.34	NA	NA	NA
$\Box$	NRC - 2-wire VG - Incremental ChargeManual Svc Order-DisconnectAdd'l	SOMAN	\$12.97	NA	NA	NA	\$8.06	\$11.34	NA	NA	NA
$\vdash$								ļ			
111	Interoffice Transport - Dedicated - 2-wire VG - Rev Bat.		1	1			Ī	1	1		1

$\Box$				I			I	1		1	1
	2-Wire VG - per mile per month	1L5XX	\$0.0339	\$0.0098	\$0.0222	\$0.0301	\$0.0384	\$0.0323	\$0.0282	\$0.0373	\$0.0173
	2-Wire VG - Facility Termination per month	U1TR2	\$18.49	\$26.52	\$17.07	\$27.66	\$19.10	\$21.33	\$18.00	\$21.42	\$18.33
	NRC - 2-wire VG - Facility Termination -1st	U1TR2	\$107.11	\$81.09	\$79.61	\$142.31	\$76.20	\$106.72	\$137.48	\$136.44	\$55.39
	NRC - 2-wire VG - Facility Termination - Add'l	U1TR2	\$48.27	\$54.83	\$36.08	\$56.21	\$34.54	\$48.83	\$52.58	\$51.37	\$17.37
	NRC - 2-wire VG -Facility Termination - Disconnect Charge -1st	U1TR2	\$37.16	\$31.01	NA	NA	\$28.03	\$38.05	NA	NA	\$27.96
	NRC - 2-wire VG - Facility Termination - Disconnect Charge -Add'l	U1TR2	\$5.88	\$12.78	NA	NA	\$5.37	\$7.23	NA	NA	\$3.51
	NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	\$19.99	NA	NA	NA	NA	\$19.99
	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
	NRC - 2-wire VG - Incremental ChargeManual Svc Order - 1st	SOMAN	\$27.37	NA	\$18.94	\$37.21	\$18.14	\$25.52	\$38.07	\$39.63	NA
	NRC - 2-wire VG - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$27.57	NA	\$18.94	\$37.21	\$18.14	\$25.52	\$38.07	\$39.63	NA
	NRC - 2-wire VG - Incremental ChargeManual Svc Order-Disconnect1st	SOMAN	\$12.97	NA	NA	NA	\$8.06	\$11.34	NA	NA	NA
	NRC - 2-wire VG - Incremental ChargeManual Svc Order-DisconnectAdd'l	SOMAN	\$12.97	NA	NA	NA	\$8.06	\$11.34	NA	NA	NA
	•										
	Common (Shared) Transport										
	Common (Shared) Transport per mile per mou	NA	\$0.00001	\$0.000012	\$0.000008	\$0.0000049	\$0.0000083	\$0.0000091	\$0.00001	\$0.0000121	\$0.00004
	Common (Shared) Transport Facilities Termination per mou	NA	\$0.00045	\$0.0005	\$0.0004152	\$0.000426	\$0.00047	\$0.0004281	\$0.00034	\$0.0004672	\$0.00036
ldot											
	Interoffice Transport - Dedicated - 4-wire VG										
	4-Wire VG - per mile per month	1L5XX	NA	\$0.0098	NA	NA	NA	NA	NA	NA	NA
	4-Wire VG - Facility Termination per month	U1TV4	NA	\$23.64	NA	NA	NA	NA	NA	NA	NA
	NRC - 4-wire VG - Facility Termination -1st	U1TV4	NA	\$81.09	NA	NA	NA	NA	NA	NA	NA
	NRC - 4-wire VG - Facility Termination - Add'l	U1TV4	NA	\$54.63	NA	NA	NA	NA	NA	NA	NA
	NRC - 4-wire VG -Facility Termination - Disconnect Charge -1st	U1TV4	NA	\$31.01	NA	NA	NA	NA	NA	NA	NA
	NRC - 4-wire VG - Facility Termination - Disconnect Charge -Add'l	U1TV4	NA	\$12.78	NA	NA	NA	NA	NA	NA	NA
	NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	NA	NA	NA	NA	NA	NA
	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	NA	\$2.77	NA	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
	NRC - 4-wire VG - Incremental ChargeManual Svc Order - 1st	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
	NRC - 4-wire VG - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
	NRC - 4-wire VG - Incremental ChargeManual Svc Order-Disconnect1st	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
	NRC - 4-wire VG - Incremental ChargeManual Svc Order-DisconnectAdd'l	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
	Interoffice Transport - Dedicated - DS0 - 56kbps										
	DS0 - per mile per month	1L5XX	\$0.0339	\$0.0098	\$0.0222	\$0.0301	\$0.0384	\$0.0323	\$0.0282	\$0.0373	\$0.1730
	DS0 - Facility Termination per month	U1TD5	\$17.81	\$19.31	\$16.45	\$26.95	\$18.37	\$20.64	\$17.40	\$20.71	\$17.74
	NRC - DS0 - Facility Termination - 1st	U1TD5	\$107.11	\$81.11	\$79.61	\$142.31	\$76.20	\$106.72	\$137.48	\$136.44	\$55.39
Ш	NRC - DS0 - Facility Termination - Add'l	U1TD5	\$48.27	\$54.83	\$36.08	\$56.21	\$34.54	\$48.83	\$52.58	\$51.37	\$17.37
Ш	NRC - DS0 -Facility Termination - Disconnect Charge - 1st	U1TD5	\$37.16	\$31.01	NA	NA	\$28.03	\$38.05	NA	NA	\$27.96
	NRC - DS0 - Facility Termination - Disconnect Charge - Add'l	U1TD5	\$5.88	\$12.78	NA	NA	\$5.37	\$7.23	NA	NA	\$3.51
	NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	\$19.99	NA	NA	NA	NA	\$19.99
ldot	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	NA
ПП	NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.77	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
ПП	NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
$\Box\Box$	NRC - DS0 -Incremental ChargeManual Svc Order - 1st	SOMAN	\$27.37	NA	\$18.94	\$37.21	\$18.14	\$25.52	\$38.07	\$39.63	NA
ldot	NRC -DS0 - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$27.37	NA	\$18.94	\$37.21	\$18.14	\$25.52	\$38.07	\$39.63	NA
山口	NRC - DS0 -Incremental ChargeManual Svc Order-Disconnect1st	SOMAN	\$12.97	NA	NA	NA	\$8.06	\$11.34	NA	NA	NA
ПП	NRC - DS0 -Incremental ChargeManual Svc Order-Disconnect—Add'l	SOMAN	\$12.97	NA	NA	NA	\$8.06	\$11.34	NA	NA	NA
$\Box\Box$											
ПП	Interoffice Transport - Dedicated -DS0 - 64 kbps										
			<b>#0.0000</b>	\$0.0098	\$0.0222	\$0.0301	\$0.0384	\$0.0323	\$0.0282	\$0.0373	\$0.1730
	DS0 - per mile per month	1L5XX	\$0.0339	φ0.0096	Ψ0.0222	Ψ0.0001	Ψ0.0001				
	DS0 - per mile per month DS0 - Facility Termination per month	1L5XX U1TD6	\$0.0339	\$19.31	\$16.45	\$26.95	\$18.37	\$20.64	\$17.40	\$20.71	\$17.74
									\$17.40 \$137.48 \$52.58	\$20.71 \$136.44 \$51.37	\$17.74 \$55.39 \$17.37

$\perp \perp \perp$	NRC - DS0 -Facility Termination - Disconnect Charge - 1st	U1TD6	\$37.16	\$31.01	NA	NA	\$28.03	\$38.05	NA	NA	\$27.96
$\square$	NRC - DS0 - Facility Termination - Disconnect Charge - Add'l	U1TD6	\$5.88	\$12.78	NA	NA	\$5.37	\$7.23	NA	NA	\$3.51
$\square$	NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	\$19.99	NA	NA	NA	NA	\$19.99
$\square$	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	NA
$\vdash$	NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.77	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
$\vdash$	NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
$\vdash$	NRC - DS0 -Incremental ChargeManual Svc Order - 1st	SOMAN	\$27.37	NA	\$18.94	\$37.21	\$18.14	\$25.52	\$38.07	\$39.63	NA
$\vdash$	NRC -DS0 - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$27.37	NA	\$18.94	\$37.21	\$18.14	\$25.52	\$38.07	\$39.63	NA
++-	NRC - DS0 -Incremental ChargeManual Svc Order-Disconnect1st	SOMAN	\$12.97	NA	NA	NA	\$8.06	\$11.34	NA	NA	NA
++	NRC - DS0 -Incremental ChargeManual Svc Order-Disconnect—Add'l	SOMAN	\$12.97	NA	NA	NA	\$8.06	\$11.34	NA	NA	NA
++-	Interoffice Transport - Dedicated - DS1										-
++-	DS1 - per mile per month	1L5XX	\$0.6920	\$0.6013	\$0.4523	\$0.4500	\$0.7831	\$0.6598	\$0.0783	\$0.7598	\$0.3525
++-	DS1 - Facility Termination per month	U1TF1	\$79.69	\$99.79	\$78.47	\$55.05	\$93.40	\$74.40	\$71.29	\$94.98	\$75.83
++-	NRC - DS1-Facility Termination - 1st	U1TF1	\$198.15	\$45.91	\$147.07	\$298.18	\$140.49	\$196.28	\$217.17	\$216.27	\$145.98
++-	NRC - DS1 - Facility Termination - Add'l	U1TF1	\$148.18	\$44.18	\$111.75	\$231.23	\$106.69	\$147.31	\$163.75	\$162.70	\$109.85
++-	NRC - DS1 - Facility Termination - Add1	U1TF1	\$25.44	\$30.30	NA NA	Ψ231.23 NA	\$20.00	\$26.56	NA	NA	\$19.55
++	NRC - DS1 - Facility Termination - Disconnect Charge - 1st	U1TF1	\$20.42	\$26.76	NA NA	NA NA	\$16.34	\$20.50	NA NA	NA NA	\$14.99
++	NRC - Manual Svc Order, per LSR	SOMAN	NA	\$20.76	NA NA	\$19.99	NA	NA	NA NA	NA NA	\$19.99
++	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA NA	\$3.84	NA NA	NA	NA NA	NA NA	NA NA	NA NA	NA
++	NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.77	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
++	NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
HH	NRC - DS1 - Incremental ChargeManual Svc Order - 1st	SOMAN	\$27.37	NA NA	\$18.94	NA	\$18.14	\$25.52	\$38.07	\$39.63	NA NA
	NRC -DS1 - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$27.37	NA	\$18.94	NA.	\$18.14	\$25.52	\$38.07	\$39.63	NA
	NRC - DS1 - Incremental ChargeManual Svc Order-Disconnect1st	SOMAN	\$12.97	NA	NA	NA	\$8.06	\$11.31	NA	NA	NA
	NRC - DS1 - Incremental ChargeManual Svc Order-Disconnect—Add'l	SOMAN	\$12.97	NA	NA	NA	\$8.06	\$11.34	NA	NA	NA
	, and the second										
	Interoffice Transport - Dedicated - DS3										
	DS3 - per mile per month	1L5XX	\$4.98	\$4.17	\$2.75	\$12.62	\$14.04	\$15.02	\$12.98	\$8.13	\$5.89
	DS3 -Facility Termination per month	U1TF3	\$898.15	\$1,121.93	\$788.00	\$1,204	\$1,101	\$744.38	\$720.38	\$967.70	\$760.20
	NRC - DS3 - Facility Termination -1st	U1TF3	\$511.77	\$557.69	\$511.10	\$946.23	\$611.41	\$686.74	\$794.94	\$606.72	\$625.91
	NRC - DS3 - Facility Termination - Add'l	U1TF3	\$330.92	\$325.61	\$330.77	\$516.89	\$304.90	\$477.76	\$579.55	\$423.45	\$311.39
	NRC - DS3 - Facility Termination - Disconnect Charge - 1st	U1TF3	\$121.72	\$111.56	\$122.31	NA	\$102.16	\$125.56	NA	NA	\$103.36
$\perp \perp \perp$	NRC - DS3 - Facility Termination - Disconnect Charge - Add'l	U1TF3	\$118.54	\$108.34	\$119.14	NA	\$99.46	\$118.79	NA	NA	\$100.59
$\square$	NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	\$19.99	NA	NA	NA	NA	\$19.99
$\vdash$	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	NA
++	NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.77	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
++	NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA COLAD	\$0.42	NA COZ OC	NA CO2.42	NA ©E0.2E	NA Coa oz	NA COLOC	NA ©E4.00	NA NA
++	NRC - DS3 - Incremental ChargeManual Svc Order - 1st	SOMAN	\$38.48	NA NA	\$37.96	\$93.12	\$50.25	\$64.97	\$91.26	\$54.26	NA NA
++	NRC - DS3 - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$38.48	NA NA	\$37.96	\$93.12	\$50.25	\$64.97	\$91.26	\$54.26	NA NA
++	NRC - DS3 - Incremental ChargeManual Svc Order-Disconnect1st NRC - DS3 - Incremental ChargeManual Svc Order-Disconnect—Add'l	SOMAN SOMAN	\$19.03 \$19.03	NA NA	\$18.23 \$18.23	NA NA	\$20.94 \$20.94	\$27.08 \$27.08	NA NA	NA NA	NA NA
++	NAC - DOS - INCIENTENTAL CHARGEIMANUAL SVC ORGET-DISCONNECT—Add I	SUIVIAIN	φ19.03	INA	Φ10.∠3	INA	<b>⊅∠∪.94</b>	Φ∠1.00	INA	INA	INA
++-	Interoffice Transport - Dedicated - STS-1	+						<del> </del>			<del>                                     </del>
++	STS-1 - per mile per month	1L5XX	\$4.98	\$4.17	\$2.72	\$12.62	\$14.04	\$13.48	\$6.29	\$8.13	\$6.88
++	STS-1 -Facility Termination per month	U1TFS	\$895.41	\$1,105.98	\$783.63	\$1,204	\$1,101	\$692.52	\$800.94	\$967.58	\$838.65
HH	NRC - STS-1 - Facility Termination -1st	U1TFS	\$511.77	\$557.69	\$449.91	\$946.23	\$611.41	\$858.15	\$624.86	\$606.72	\$858.26
	NRC - STS-1 - Facility Termination - Add'l	U1TFS	\$330.92	\$325.61	\$119.14	\$516.89	\$304.90	\$524.58	\$436.36	\$423.45	\$525.25
	NRC - STS-1 - Facility Termination - Disconnect Charge - 1st	U1TFS	\$121.72	\$111.56	\$137.17	NA	\$102.16	\$125.56	NA	NA NA	\$103.36
+-	NRC - STS-1 - Facility Termination - Disconnect Charge - Add'l	U1TFS	\$118.54	\$108.34	\$119.14	NA	\$99.46	\$118.79	NA	NA	\$100.59
111		SOMAN	NA	\$21.56	NA	\$19.99	NA	NA	NA	NA	\$19.99
++	NRC - Manual Svc Order, per LSR				NA	NA	NA	NA	NA	NA	NA
#	NRC - Manual Svc Order, per LSR NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	INA	14/3	13/3	14/3	INA	INA	
	NRC - Manual Svc Order, per LSR disconnect	SOMAN SOMEC	NA \$3.50	\$3.84 \$2.77	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	NRC - Manual Svc Order, per LSR disconnect NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.77	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50

NRC - DS3 - Incremental ChargeManual Svc Order-Disconnect1st	SOMAN	\$19.03	NA	\$3.17	NA	\$20.94	\$27.08	NA	NA	
NRC - DS3 - Incremental ChargeManual Svc Order-Disconnect—Add'l	SOMAN	\$19.03	NA	\$3.17	NA NA	\$20.94	\$27.08	NA	NA	
Titto 200 more on ango managa or or or or or or or or or or or or or	00	ψ.σ.σσ		Ψοιιι		Ψ20.0.	ψ2σ			
Interoffice Transport - Dedicated - OC3										
OC3 -per mile per month	1L5XX	\$7.35	\$8.24	\$4.37	\$27.97	\$23.89	\$18.35	\$14.10	\$9.75	\$
OC3 -Facility Termination per month		\$2,475	\$3,020.08	\$2,187.00	\$3,390	\$2,990	\$1,892.00	\$2,071	\$2,802	\$
NRC - OC-3 - Facility Termination - 1st		\$820.85	\$869.65	\$819.29	\$1,399	\$927.35	\$1,283.00	\$1,381	\$915.64	\$
NRC - OC-3 - Facility Termination - Add'l		\$317.48	\$312.05	\$317.38	\$516.89	\$304.90	\$404.94	\$509.93	\$410.02	\$
NRC - OC-3 - Facility Termination - Disconnect Charge - 1st		\$121.72	\$111.56	\$122.31	NA	\$102.16	\$134.07	\$131.65	NA	\$
NRC - OC-3 - Facility Termination - Disconnect Charge - Add'l		\$118.54	\$108.34	\$119.14	NA	\$99.46	\$130.59	\$128.19	NA	\$
NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	\$19.99	NA	NA	NA	NA	\$
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	
NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.77	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	9
NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.43	NA	NA	NA	NA	NA	NA	
NRC - OC3 - Incremental Cost - Manual Svc Order vs. Electronic-1st	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$94.77	\$54.26	
NRC - OC3 - Incremental Cost - Manual Svc Order vs. Electronic-Add'l	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$94.77	\$54.26	
	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	\$29.76	NA	
NRC - OC3 - Incremental Cost - Manual Svc Order vs. Electronic-Disconnect-	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	\$29.76	NA	1
Interoffice Transport - Dedicated - OC12										+-
OC12 -per mile per month	1L5XX	\$19.26	\$26.45	\$15.05	\$84.88	\$74.44	\$60.42	\$48.55	\$32.52	\$
OC12 -Facility Termination	120701	\$9,763	\$11,599.14	\$8,202.00	\$12,344	\$11,517	\$7,182.00	\$2,071	\$11,132	\$
NRC - OC12- Facility Termination - 1st		\$1.036	\$1.086.66	\$1,034.00	\$1.713	\$1.147	\$1,598.00	\$1,381	\$1,131	9
NRC - OC12- Facility Termination - Add'l		\$317.48	\$312.05	\$317.38	\$516.89	\$304.90	\$404.94	\$509.93	\$410.02	\$
NRC - OC12 - Facility Termination - Disconnect Chg - 1st		\$121.72	\$111.56	\$122.31	NA	\$102.16	\$134.07	\$131.65	NA	\$
NRC - OC12 - Facility Termination - Disconnect Chg - Add'l		\$118.54	\$108.34	\$119.14	NA	\$99.46	\$130.59	\$128.19	NA	\$
NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	\$19.99	NA	NA	NA	NA	\$
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	
NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.77	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	
NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.43	NA	NA	NA	NA	NA	NA	
NRC - OC12 - Incremental Cost - Manual Svc Order vs. Electronic-1st	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$94.77	\$54.26	
NRC - OC12 - Incremental Cost - Manual Svc Order vs. Electronic-Add'l	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$94.77	\$54.26	
NRC - OC12 - Incremental Cost - Manual Svc Order vs. Elect-Disconnect-1st	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	\$29.76	NA	
NRC - OC12 - Incremental Cost - Manual Svc Order vs. Elect-Disconnect-Add	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	\$29.76	NA	
Interoffice Transport - Dedicated - OC48										
OC48 -per mile per month	1L5XX	\$30.65	\$34.07	\$25.70	\$138.02	\$128.59	\$102.43	\$120.02	\$45.92	\$
OC48 -Facility Termination per month	ILSAA	\$11.691		\$11,134.00	\$16,017	\$14,950	\$11,480.00	\$1,677	\$967.58	\$
OC48 -per Interface OC12 on OC48 per month		\$1,424	\$1,199.42	\$1,137.00	\$1,497	\$1,451	\$1,351.00	\$582.66	\$1,561	\$
NRC - OC48 - Facility Termination - 1st		\$1,036	\$1,086.66	\$1,034.00	\$1,713	\$1,437	\$1,598.00	\$1,722	\$1,131	\$
NRC - OC48 - Facility Termination - Add'l		\$317.48	\$312.05	\$317.38	\$516.89	\$304.90	\$404.94	\$542.73	\$410.02	\$:
NRC - OC48 - Interface OC12 on OC48 - 1st		\$540.10	\$543.72	\$539.36	\$844.21	\$532.13	\$729.04	\$720.81	\$635.04	\$
NRC - OC48 - Interface OC12 on OC48 - Add'l		\$317.48	\$312.05	\$317.38	\$516.89	\$304.90	\$404.94	\$400.38	\$410.02	\$
NRC - OC48 - Facility Termination - Disconnect Chg - 1st		\$121.72	\$111.56	\$122.31	NA	\$102.16	\$134.07	\$131.65	NA NA	\$
NRC - OC48 - Facility Termination - Disconnect Chg - Add'l		\$118.54	\$108.34	\$119.14	NA	\$99.46	\$130.59	\$128.19	NA	\$
NRC - OC48 - Interface OC12 on OC48 - Disconnect - 1st		\$121.72	\$111.56	\$122.31	NA	\$102.16	\$134.07	\$131.65	NA	\$
NRC - OC48 - Interface OC12 on OC48 - Disconnect - Add'l		\$118.54	\$108.34	\$119.14	NA NA	\$99.46	\$130.59	\$128.19	NA NA	\$
NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	\$19.99	NA	NA	NA	NA	\$
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA NA	NA	NA	NA	NA	NA	┪
NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.77	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	1
NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.43	NA	NA	NA	NA	NA	NA	1
NRC - OC48 - Incremental Cost - Manual Svc. Order vs. Electronic-1st	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$69.34	\$54.26	1
NRC - OC48 - Incremental Cost - Manual Svc. Order vs. Electronic-Add'l	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$69.34	\$54.26	1
		\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$69.34	\$54.26	1
NRC - OC48 - Interface- Incremental Cost - Manual Svc. Order vs. Electronic-1	SOMAN	φ30. <del>4</del> 0	IN/A							
NRC - OC48 - Interface- Incremental Cost - Manual Svc. Order vs. Electronic-1 NRC - OC48 - Interface- Incremental Cost - Manual Svc. Order vs. Electronic-A		\$38.48	NA	\$37.55	\$93.12	\$50.25	\$68.62	\$69.34	\$54.26	T

-	NRC - OC48 - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	\$28.59	\$29.76	NA	NA
	NRC - OC48-Interface-Incremental Cost-Manual Svc. Order vs. Elect-Disconne		\$19.03	NA NA	\$18.03	NA NA	\$20.94	\$28.59	\$29.76	NA NA	NA NA
++	NRC - OC48-Interface-Incremental Cost-Manual Svc. Order vs. Elec-Disconne		\$19.03	NA NA	\$18.03	NA NA	\$20.94	\$28.59	\$29.76	NA	NA
+++	1410 0040 Interface incremental oost Manual Ove. Order vs. Elec Disconne	OOWAIN	ψ13.03	19/3	ψ10.00	13/3	Ψ20.54	Ψ20.00	Ψ23.70	19/3	14/3
	UNBUNDLED CHANNELIZATION										
	DS3 Channelization (DS3 to DS1)										
	per Channelized System (28 DS1) per month	MQ3	\$225.36	\$220.97	\$182.04	\$236.32	\$245.84	\$229.30	\$226.81	\$200.01	\$222.98
	NRC - 1st	MQ3	\$265.87	\$356.40	\$265.91	\$425.41	\$259.76	\$356.80	\$351.95	\$321.54	\$265.08
	NRC - Add'l	MQ3	\$188.51	\$188.00	\$188.78	\$303.33	\$182.64	\$247.40	\$243.76	\$234.30	\$185.94
	NRC -1st - Disconnect	MQ3	\$71.76	\$61.64	\$72.50	NA	\$60.96	\$79.94	\$77.90	NA	\$61.09
	NRC -Add'l - Disconnect	MQ3	\$52.03	\$58.98	\$59.96	NA	\$50.46	\$65.20	\$63.32	NA	\$50.31
	per Interface per month (COCI)	UC1D1	\$17.22	\$14.40	\$11.02	\$8.52	\$7.55	\$5.58	\$4.61	\$11.99	\$3.91
	NRC - 1st	UC1D1	\$12.05	\$13.16	\$12.02	\$15.86	\$12.29	\$15.85	\$15.76	\$12.05	\$12.61
	NRC - Add'l	UC1D1	\$8.69	\$9.43	\$8.66	\$11.36	\$8.80	\$11.35	\$11.28	\$8.68	\$9.03
	NRC - Manual Svc Order, per LSR	SOMAN	NA NA	\$21.56	NA NA	\$19.99	NA NA	NA	NA NA	NA NA	\$19.99
	NRC - Manual Svc Order, per LSR disconnect	SOMAN		\$3.84		NA Co. co		NA ©2.50			NA Co.co
++-	NRC - Electronic Svc Order, per LSR NRC - Electronic Svc Order, per LSR disconnect	SOMEC	\$3.50 NA	\$2.77 \$0.43	\$3.50 NA	\$3.50 NA	\$3.50 NA	\$3.50 NA	\$3.50 NA	\$3.50 NA	\$3.50 NA
++-	Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -1st	SOMAN	\$15.61	\$0.43 NA	\$14.75	\$41.47	\$19.74	\$26.95	\$28.13	\$25.59	\$21.71
	Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -1st	SOMAN	\$7.39	NA NA	\$6.55	\$11.99	\$8.77	\$11.98	\$13.33	\$8.92	\$10.46
	Incremental Cost-Manual Svc. Order vs. Elect -Disconnect - 1st	SOMAN	\$11.67	NA NA	\$10.60	NA NA	\$12.43	\$16.97	\$18.26	NA	\$14.21
	Incremental Cost-Manual Svc. Order vs. Elect -Disconnect - Add'l	SOMAN	\$0.9469	NA	NA NA	NA	NA	NA NA	\$1.48	NA NA	\$1.46
	interestrictive Cook Mariada CVC. Craci Vo. Eloci Bioconnoct Add i	CONTRACT	ψο.ο τοο	107	101	1471	10/	1471	ψιιιο	10,	Ψ1.10
	DS1 Channelization (DS1 to DS0)										
	per Channelized System (24 DS0) per month	MQ1	\$136.82	\$153.60	\$126.22	\$200.01	\$209.87	\$146.87	\$177.72	\$147.51	\$165.21
	NRC - 1st	MQ1	\$197.98	\$182.14	\$198.22	\$302.82	\$193.63	\$271.52	\$267.19	\$220.89	\$197.21
	NRC - Add'l	MQ1	\$123.12	\$125.18	\$123.59	\$184.20	\$118.37	\$164.56	\$161.43	\$137.15	\$119.99
	NRC -1sr - Disconnect	MQ1	\$30.18	\$19.52	\$31.03	NA	\$26.44	\$36.38	\$34.55	NA	\$25.66
	NRC -Add'l - Disconnect	MQ1	\$18.86	\$18.14	\$19.75	NA	\$16.83	\$11.98	\$21.14	NA	\$15.81
	- Interface (COCI)										
	per OCU-DP(data) card per month (2.4-64kbs)	1D1DD	\$1.66	\$2.20	\$1.86	\$2.94	\$3.12	\$2.86	\$2.88	\$2.34	\$2.46
	NRC - 1st	1D1DD	\$12.05	\$13.16	\$12.02	\$15.86	\$12.29	\$15.85	\$15.76	\$12.05	\$12.61
	NRC - Add'l	1D1DD	\$8.69	\$9.43	\$8.66	\$11.36	\$8.80	\$11.35	\$11.28	\$8.68	\$9.03
	per BRITE card per month	UC1CA	\$3.41	\$3.83	\$3.71	\$4.04	\$4.18	\$3.88	\$3.76	\$4.21	\$3.33
	NRC - 1st	UC1CA	\$12.05	\$13.16	\$12.02	\$15.86	\$12.29	\$15.85	\$15.76	\$12.05	\$12.61
	NRC - Add'l per VG card per month (DS0)	UC1CA 1D1VG	\$8.69 \$0.8586	\$9.43 \$1.45	\$8.66 \$1.17	\$11.36 \$1.40	\$8.80 \$1.62	\$11.35 \$1.45	\$11.28 \$1.64	\$8.68 \$1.47	\$9.03 \$1.25
$\vdash$	NRC - 1st	1D1VG	\$12.05	\$13.16	\$1.17	\$1.40 \$15.86	\$1.62	\$1.45 \$15.85	\$1.64	\$1.47	\$1.25
	NRC - Add'l	1D1VG	\$8.69	\$9.43	\$8.66	\$11.36	\$8.80	\$11.35	\$11.28	\$8.68	\$9.03
++	NRC - Add 1 NRC - Manual Svc Order, per LSR	SOMAN	Φ0.09 NA	\$21.56	φο.00 NA	\$19.99	φο.ου NA	NA NA	Ψ11.20 NA	Ф0.00 NA	\$19.99
++-	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA NA	\$3.84	NA NA	NA	NA NA	NA NA	NA NA	NA NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.77	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	NRC - Electronic Svc Order, per LSR disconnect	SOMAN	NA	\$0.43	NA	NA	NA	NA	NA	NA NA	NA
	Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -1st	SOMAN	\$15.61	NA	\$14.75	\$41.47	\$19.74	\$26.95	\$28.13	\$25.59	\$25.66
	Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -Add'l	SOMAN	\$7.39	NA	\$6.55	\$11.99	\$8.77	\$11.98	\$13.33	\$8.92	\$15.81
	Incremental Cost-Manual Svc. Order vs. Elect -Disconnect - 1st	SOMAN	\$11.67	NA	\$10.70	NA	\$12.43	\$16.97	\$18.26	NA	\$14.21
	Incremental Cost-Manual Svc. Order vs. Elect -Disconnect - Add'l	SOMAN	\$0.9469	NA	\$0.00	NA	NA	NA	\$1.48	NA	\$1.46
	_										
	UNBUNDLED DARK FIBER										
	Dark Fiber - Interoffice (four fiber strands) per route mile or fraction thereof, per mor		\$25.80	\$28.82	\$24.96	\$31.95	\$32.28	\$33.93	\$29.86	\$36.75	\$28.60
	NRC - Per each four-fiber dark fiber arrangement - 1st	UDF14	\$1,739.00		\$1,737.00	\$1,741.00	\$1,746.00	\$1,741.00	\$1,738.00	\$1,747.00	\$1,742.00
oxdot	NRC - Per each four-fiber dark fiber arrangement - Add'l	UDF14	\$563.09	\$275.82	\$562.39	\$563.75	\$565.20	\$563.79	\$562.82	\$565.53	\$564.08
$\Box$	NRC -Disconnect1st	UDF14	NA	\$587.64	NA	NA	NA	NA	NA	NA	NA
$\Box$	NRC -DisconnectAdd'l	UDF14	NA	\$366.34	NA	NA	NA	NA	NA	NA	NA
+	Ded They Level Observe Way Character 1	41.550	<b>#70.00</b>	<b>#</b> 50.05	<b>#</b> 54.00	0.40.07	004.70	074.55	ΦEC 17	#400 0 <del>7</del>	<b>#00.00</b>
	Dark Fiber - Local Channel(four fiber strands) per route mile or fraction thereof, per	TL5DC	\$70.82	\$58.35	\$54.63	\$49.07	\$64.72	\$71.55	\$56.47	\$100.37	\$60.06

NRC - Per each	n four-fiber dark fiber arrangement - 1st	UDFC4	\$1,739.00	\$1,278.62	\$1,737.00	\$1,741.00	\$1,746.00	\$1,741.00	\$1,738.00	\$1,747.00	\$1,742.00
NRC - Per each	n four-fiber dark fiber arrangement - Add'l	UDFC4	\$563.09	\$275.82	\$562.39	\$563.75	\$565.20	\$563.79	\$562.82	\$565.53	\$564.08
NRC -Disconne	ct1st	UDFC4	NA	\$587.64	NA						
NRC -Disconne	ctAdd'l	UDFC4	NA	\$366.34	NA						
Dark Fiber - Local L	oop (four fiber strands) per route mile or fraction thereof, per mo	1L5DL	\$70.82	\$58.35	\$54.63	\$49.07	\$64.72	\$71.55	\$56.47	\$100.37	\$60.06
NRC - Per each	n four-fiber dark fiber arrangement - 1st	UDFL4	\$1,739.00	\$1,278.62	\$1,737.00	\$1,741.00	\$1,746.00	\$1,741.00	\$1,738.00	\$1,747.00	\$1,742.00
NRC - Per each	n four-fiber dark fiber arrangement - Add'l	UDFL4	\$563.09	\$275.82	\$562.39	\$563.75	\$565.20	\$563.79	\$562.82	\$565.53	\$564.08
NRC -Disconne	ct1st	UDFL4	NA	\$587.64	NA						
NRC -Disconne	ctAdd'l	UDFL4	NA	\$366.34	NA						
NOTES:											
Interim rates subject	t to true-up.										

Exhibit C

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
UNBUNDLED LOOP COMBINATIONS							0			
ONDONDEED EOOF COMBINATIONS										
Unbundled Loop/Port Combinations (Note 4)										
MARKET RATES (INCLUDING ALL VERTICAL FEATURES) (Note 1)										
Density Zone 1 / Top 8 MSAs in BellSouth Region			Orlando, Ft. Lauderdale, Miami	Atlanta		New Orleans		Greensboro- Winston Salem- Highpoint/ Charlotte- Gastonia-Rock Hill		Nashville
Customers with 4 or more DS0 Equivalent										
Currently Combined (Note2)								1		
2-Wire Voice Grade Loop with 2-Wire Line Port (Res. and Bus.)										
2-Wire Voice Grade Line Port (Res.), per month										
2- wire voice unbundled port - residence	UEPRL	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-wire voice unbundled port with caller ID - residence	UEPRC	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-wire voice unbundled port outgoing only - residence	UEPRO	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	UEPAR	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	UEPRM	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	UEPAS	NA	NA	NA	NA	\$14.00	NA	NA	NA	NA
2-wire voice grade unbundled Mississippi extended local dialing parity port with caller										
ID   2-wire voice grade unbundled South Carolina extended local dialing parity port with	UEPAT	NA	NA	NA	NA	NA	NA	NA	NA	NA
caller ID	UEPAU	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice grade unbundled Tennessee extended local dialing parity port with caller ID	UEPAQ	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-wire voice unbundled Florida area calling with caller ID - residence	UEPAF	NA	\$14.00	NA	NA	NA	NA	NA	NA	NA
2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)	UEPAG	NA	NA	NA	NA	\$14.00	NA	NA	NA	NA
2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)	UEPAH	NA	NA	NA	NA	\$14.00	NA	NA	NA	NA
2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence (LW8)	UEPAJ	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)	UEPAK	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence								l l		
(TACER)	UEPAL	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACSR)	UEPAM	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (1MF2X)	UEPAN	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (2MR)	UEPAO	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-wire voice unbundled res, low usage line port with Caller ID (LUM)	UEPAP	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-Wire Voice Grade Line Port (Bus.), per month								İ		
2-wire voice unbundled port without Caller ID	UEPBL	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-wire voice unbundled port with unbundled port with Caller+E484 ID	UEPBC	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-wire voice unbundled outgoing only port	UEPBO	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	UEPAW	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	UEPBM	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	UEPAX	NA	NA	NA	NA	\$14.00	NA	NA	NA	NA
2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	UEPAY	NA	NA	NA	NA	NA	NA	NA	NA	NA

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
2-wire voice grade unbundled South Carolina extended local dialing parity port with			<u> </u>							
caller ID	UEPAZ	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice grade unbundled Tennessee extended local dialing parity port with caller										
I ID	UEPAV	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-wire voice unbundled incoming only port with Caller ID	UEPB1 UEPAA	NA NA	\$14.00	\$14.00	NA NA	\$14.00 \$14.00	NA NA	\$14.00	NA NA	\$14.00 NA
2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC) 2-wire voice unbundled SC Bus Area Calling Port with Caller ID (LMB)	UEPAA	NA NA	NA NA	NA NA	NA NA	\$14.00 NA	NA NA	NA NA	NA NA	NA NA
2-wife voice driburidied SC Bus Area Calling Fort with Caller ID (LIVIB)	UEFAB	INA	INA	INA	INA	INA	INA	INA	INA	INA
2-wire voice unbundled TN Bus 2-Way Area Calling Port Economy Option (TACC1)	UEPAC	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-wire voice unbundled TN Bus 2-Way Area Calling Port Standard Option (TACC2)	UEPAD	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-wire voice unbundled TN Bus 2-WAY Collierville and Memphis Local Calling Port (B2F)	UEPAE	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-Wire Voice Grade Loop (SL1) (Res. and Bus.)										
RC - 2-Wire Voice Grade Loop - Statewide	UEPLX	NA	NA	NA	NA	NA	NA	\$14.18	NA	NA
RC - 2-Wire Voice Grade Loop Zone 1	UEPLX	NA	\$14.90	\$10.80	NA	\$14.05	NA	NA	NA	\$15.92
RC - 2-Wire Voice Grade Loop Zone 2	UEPLX	NA	\$18.51	\$12.47	NA	\$24.14	NA	NA	NA	\$20.79
RC - 2-Wire Voice Grade Loop Zone 3	UEPLX	NA	\$24.25	\$19.83	NA	\$49.30	NA	NA	NA	\$27.18
Combination Rates	N		<del></del>					005 15	<b>—</b>	NI.
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Statewide	Note 8	NA	NA ************************************	NA Double	NA	NA Toolog	NA	\$28.18	NA	NA 100.00
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 1 (Note 6)	Note 8	NA	\$28.90	\$24.80	NA	\$28.05	NA	NA	NA	\$29.92
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 2 (Note 6)	Note 8	NA NA	\$32.51	\$26.47	NA NA	\$38.14	NA NA	NA NA	NA NA	\$34.79
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)  Nonrecurring Charges	Note 8	NA	\$38.25	\$33.83	NA	\$63.30	NA	NA	NA	\$41.18
			<b>├</b>							
2-Wire Voice Grade Line Port (Res. And Bus.)			<del></del>							
NRC - 2- wire voice grade unbundled port/loop combination - 1st, with change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.50
NRC - 2- wire voice grade unbundled port/loop combination - Add'l, with change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.50
NRC - 2- wire voice grade driburided porvious combination - Add i, with change		INA	\$41.50	\$41.50	INA	\$41.50	NA	\$41.50	INA	\$41.5U
NRC - 2- wire voice grade unbundled port/loop combination - 1st, no change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.50
NRC - 2- wire voice grade unbundled port/loop combination - Add'l, no change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.50
INTO - 2- wire voice grade unburidied politicop combination - Add i, no change		INA	Ψ41.50	ψ41.50	INA	ψ <del>4</del> 1.50	INA	Ψ41.30	INA	ψ41.50
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent		NA	\$10.00	\$10.00	NA	\$10.00	NA	\$10.00	NA	\$10.00
1440 - 2-4416 Voice Grade Edop/Elife Fort Combination - Cabacquent		INA	\$10.00	\$10.00	INA	\$10.00	INA	\$10.00	INA	ψ10.00
NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic,										
per LSR received from the CLEC by one of the OSS interactive interfaces	SOMEC	NA	\$2.75	\$3.50	NA	\$3.50	NA	\$3.50	NA	\$3.50
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual	2220	1	<del>+=</del>	Ţ2.00	1	Ţ2.00		<b>45.00</b>	'	+1.00
Svc.Order vs. Electronic - 1st	SOMAN	NA	\$21.56	\$33.76	NA	\$31.92	NA	\$40.18	NA	\$30.89
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual			*	*		*		*		*
Svc.Order vs. Electronic - Add'l	SOMAN	NA	\$21.56	\$7.86	NA	\$7.32	NA	\$9.45	NA	\$7.03
NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update									1	
Electronic		NA	TBD	TBD	NA	\$2.11	NA	\$1.42	NA	TBD
NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update	•									
Manual Service Order		NA	TBD	TBD	NA	\$5.12	NA	\$10.27	NA	TBD
NRC - Electronic Service Order Disconnect		NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
NRC - Incremental Manual Service Order Disconnect		NA	\$3.84	\$20.00	NA	\$20.00	NA	\$20.00	NA	\$20.00
2-Wire Voice Grade Loop with 2-Wire Line Port PBX										·
2-Wire Analog Line Port (PBX), per month										
2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence	UEPRD	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS	UEPPC	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS	UEPPO	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
	UEPP1	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS										
LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS	02			*******		*		*		•

BELLSOUTH/BTI RATES NETWORK ELEMENTS AND OTHER SERVICES LOOP/PORT COMBINATIONS

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING										
PORT	UEPL2	NA	NA	NA	NA	\$14.00	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS	UEPLD	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING										
PORT	UEPT2	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING										
PORT	UEPTO	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT	UEPXA	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS	UEPXB	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT	UEPXC	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT	UEPXD	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE										
PORT	UEPXE	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING										
PORT WITHOUT LUD	UEPXF	NA	NA	NA	NA	NA	NA	NA	NA	NA
A WIDE VOICE INDIVIDUED DRY WELFT CONTROL OF THE										
2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT	UEPXG	NA	NA NA	NA NA	NA	NA	NA	NA	NA	NA NA
2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT	UEPXH	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT WITHOUT LUD	HERVI	N.A	NIA.	NI A	N.A	NI A	NI A	NI A	N/A	NI A
LUD   2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL CALLING	UEPXJ	NA	NA	NA	NA	NA	NA	NA	NA	NA
PORT	HEDVI	NIA.	NIA	NIA.	NIA	C4400	NIA	NIA.	NIA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY	UEPXK	NA	NA	NA	NA	\$14.00	NA	NA	NA	NA
ADMINISTRATIVE CALLING PORT	UEPXL	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ROOM	UEPAL	INA	\$14.00	\$14.00	INA	\$14.00	INA	\$14.00	NA	\$14.00
CALLING PORT	UEPXM	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL	UEFAIVI	INA	\$14.00	\$14.00	INA	\$14.00	INA	\$14.00	INA	\$14.00
ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT	UEPXN	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL	UEFAIN	INA	INA	INA	INA	INA	INA	INA	INA	\$14.00
DIACOUNT ROOM CALLING PORT	UEPXO	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL	OLI AO	IVA	Ψ14.00	ψ14.00	IVA	Ψ14.00	IVA	ψ14.00	14/3	ψ14.00
DISCOUNT CALLING PORT	UEPXP	NA	NA	NA	NA	\$14.00	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY	OLI XII	1471	1471	1471	1471	ψ14.00	1471	1471	14/7	107
CALLING PORT	UEPXQ	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL	02. AQ									
CALLING PORT	UEPXR	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT	UEPXS	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS			·							•
CALLING PORT	UEPXT	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT	UEPXU	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-WIRE VOICE UNBUNDLED 2-WAY PBX TENNESSEE REGIONSERV CALLING										
PORT	UEPXV	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	LNPCP									
2-Wire Voice Grade Loop (SL1)										
RC - 2- Wire Voice Grade Loop - Statewide	UEPLX	NA	NA	NA	NA	NA	NA	\$14.18	NA	NA
RC - 2- Wire Voice Grade Loop - Zone 1	UEPLX	NA	\$14.90	\$10.80	NA	\$14.05	NA	NA	NA	\$15.92
RC - 2- Wire Voice Grade Loop - Zone 2	UEPLX	NA	\$18.51	\$12.47	NA	\$24.14	NA	NA	NA	\$20.79
RC - 2- Wire Voice Grade Loop - Zone 3	UEPLX	NA	\$24.25	\$19.83	NA	\$49.30	NA	NA	NA	\$27.18
RC - 2- Wire Voice Grade Loop - Zone 4	UEPLX	NA	NA	NA	NA	NA	NA	NA	NA	NA
Combination Rates		<u> </u>								
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Statewide	Note 8	NA	NA	NA	NA	NA	NA	\$28.18	NA	NA
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 1 (Note 6)	Note 8	NA	\$28.90	\$24.80	NA	\$28.05	NA	NA	NA	\$29.92
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 2 (Note 6)	Note 8	NA	\$32.51	\$26.47	NA	\$38.14	NA	NA	NA	\$34.79
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)	Note 8	NA	\$38.25	\$33.83	NA	\$63.30	NA	NA	NA	\$41.18

DE	SCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	Nonrecurring Charges										
	Jan 19 Ja										
	NRC - 2- wire voice grade unbundled port/loop combination - 1st, with change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.50
1	INCC - 2- wire voice grade unburided porvioup combination - 1st, with change		INA	\$41.50	\$41.50	INA	\$41.50	INA	\$41.50	INA	\$41.5U
	NRC - 2- wire voice grade unbundled port/loop combination - Add'l, with change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.50
	NRC - 2- wire voice grade unbundled port/loop combination - 1st, no change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.50
	NRC - 2- wire voice grade unbundled port/loop combination - Add'l, no change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.50
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent		NA	\$10.00	\$10.00	\$10.00	\$10.00	NA	\$10.00	NA	\$10.00
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	COMEO	NIA	<b>0.75</b>	<b>PO 50</b>	NIA	<b>#0.50</b>		<b>CO 50</b>		<b>CO. FO</b>
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual	SOMEC	NA	\$2.75	\$3.50	NA	\$3.50	NA	\$3.50	NA	\$3.50
	Svc.Order vs. Electronic - 1st	SOMAN	NA	\$21.56	\$33.67	NA	\$31.92	NA	\$40.18	NA	\$30.89
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual	SOIVIAIN	INM	Ψ21.00	φυυ.υτ	INA	ψυ1.84	INA	φ40.10	INA	ψ50.03
	Svc.Order vs. Electronic - Add'l	SOMAN	NA	\$21.56	\$7.88	NA	\$7.32	NA	\$9.45	NA	\$7.03
	NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update	001111111		<b>\$2</b> 1.00	ψ1.00		ψ1.0 <u>2</u>		ψ0.10		ψ1.00
	Electronic		NA	TBD	TBD	NA	\$2.11	NA	\$1.42	NA	TBD
	NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update										
	Manual Service Order		NA	TBD	TBD	NA	\$5.12	NA	\$10.27	NA	TBD
	NRC - Electronic Service Order Disconnect		NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
	NRC - Incremental Manual Service Order Disconnect		NA	\$3.84	\$20.00	NA	\$20.00	NA	\$20.00	NA	\$20.00
0	ST BASED RATES (Notes 2 & 3)										
Cı	urrently Combined										
	2-Wire Voice Grade Loop with 2-Wire Line Port										
	2-Wire Voice Grade Line Port (Res.), per month										
	2- wire voice unbundled port - residence	UEPRL	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
	2-wire voice unbundled port with caller ID - residence	UEPRC	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
	2-wire voice unbundled port outgoing only - residence	UEPRO	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
		LIEDAD	<b>#</b> 0.00								
	2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	UEPAR	\$2.20	NA	NA	NA	NA	NA	NA	NA	NA
	2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	UEPRM	NA	NA	NA	\$2.61	NA	NA	NA	NA	NA
	2-will voice grade disputition residucity extended local dialing parity port with caller ID	OLF KIVI	INA	INA	INA	φ∠.01	INA	INA	INA	INA	INA
	2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	UEPAS	NA	NA	NA	NA	\$2.55	NA	NA	NA	NA
	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller					1.2.	<del>+</del> =.00	1		1	
	ID	UEPAT	NA	NA	NA	NA	NA	\$2.12	NA	NA	NA
	2-wire voice grade unbundled South Carolina extended local dialing parity port with										
	caller ID	UEPAU	NA	NA	NA	NA	NA	NA	NA	\$3.69	NA
	2-wire voice grade unbundled Tennessee extended local dialing parity port with caller										
	ID III	UEPAQ	NA	NA	NA	NA	NA	NA	NA	NA	\$4.54
	2-wire voice unbundled area plus port with caller ID - residence	UEPRM	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
	2-wire voice unbundled Florida area calling with caller ID - residence	UEPAF	NA	\$1.35	NA NA	NA NA	NA CO. E.E.	NA NA	NA NA	NA NA	NA NA
	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)  2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)	UEPAG UEPAH	NA NA	NA NA	NA NA	NA NA	\$2.55 \$2.55	NA NA	NA NA	NA NA	NA NA
	2-wire voice unbundled Louisiana Area Pius with Caller ID - residence (AC7)  2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence	UEPAR	INA	INA	INA	INA	Φ∠.33	INA	INA	INA	INA
	(LW8)	UEPAJ	NA	NA	NA	NA	NA	NA	NA	\$3.69	NA
		OLI AU	INA	INA	INA	INA	INA	INA	INA	φυ.υσ	INA
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)	UEPAK	NA	NA	NA	NA	NA	NA	NA	NA	\$4.54
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence					1		1		1 1	¥ ··• ·
	(TACER)	UEPAL	NA	NA	NA	NA	NA	NA	NA	NA	\$4.54
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
	(TACSR)	UEPAM									\$4.54

	CRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
$\perp \perp$	(1MF2X)	UEPAN	NA	NA	NA	NA	NA	NA	NA	NA	\$4.54
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (2MR)	UEPAO	NA	NA	NA	NA	NA	NA	NA	NA	\$4.54
++	2-wire voice unburidied refinessee Area Calling port with Caller ID - residence (2MK)  2-wire voice unbundled res, low usage line port with Caller ID (LUM)	UEPAP	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54 \$4.54
++	2-Wire Voice Grade Line Port (Bus.), per month	OLFAF	Φ2.20	φ1.33	\$1.79	\$2.01	\$2.55	ΦΖ.1Ζ	\$2.20	\$3.09	φ4.54
++	2-wire voice Grade Line Fort (Bus.), per month  2-wire voice unbundled port without Caller ID	UEPBL	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
+-	2-wire voice unbundled port without caller ib  2-wire voice unbundled port with unbundled port with Caller+E484 ID	UEPBC	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
+-	2-wire voice unbundled outgoing only port	UEPBO	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
++	2 Wile voice unburialed outgoing only port	OLI DO	ψ2.20	ψ1.55	ψ1.73	Ψ2.01	Ψ2.55	ΨΖ.1Ζ	ψ2.20	ψ5.05	Ψτ.υτ
	2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	UEPAW	\$2.20	NA	NA	NA	NA	NA	NA	NA	NA
	0 01 71					İ				†	
	2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	UEPBM	NA	NA	NA	\$2.61	NA	NA	NA	NA	NA
	2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	UEPAX	NA	NA	NA	NA	\$2.55	NA	NA	NA	NA
	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller										
	ID	UEPAY	NA	NA	NA	NA	NA	\$2.12	NA	NA	NA
	2-wire voice grade unbundled South Carolina extended local dialing parity port with	· · · · · · · · · · · · · · · · · · ·									
	caller ID	UEPAZ	NA	NA	NA	NA	NA	NA	NA	\$3.69	NA
	2-wire voice grade unbundled Tennessee extended local dialing parity port with caller										
$\sqcup \! \! \! \! \! \perp$	ID	UEPAV	NA	NA	NA	NA	NA	NA	NA	NA	\$4.54
oxdot	2-wire voice unbundled incoming only port with Caller ID	UEPB1	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
$oldsymbol{oldsymbol{oldsymbol{eta}}}$	2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	UEPAA	NA	NA	NA	NA	\$2.55	NA	NA	NA	NA
Ш	2-wire voice unbundled SC Bus Area Calling Port with Caller ID (LMB)	UEPAB	NA	NA	NA	NA	NA	NA	NA	\$3.69	NA
	0 :	LIEDAG									
Щ.	2-wire voice unbundled TN Bus 2-Way Area Calling Port Economy Option (TACC1)	UEPAC	NA	NA	NA	NA	NA	NA	NA	NA	\$4.54
	Control value and and TN Don Collins of Collins Don Collins of Control Collins (TACCO)	LIEDAD									04.54
$\vdash$	2-wire voice unbundled TN Bus 2-Way Area Calling Port Standard Option (TACC2)	UEPAD	NA	NA	NA	NA	NA	NA	NA	NA	\$4.54
	2-wire voice unbundled TN Bus 2-WAY Collierville and Memphis Local Calling Port (B2F)	UEPAE	NA	NIA	NA	NIA.	NIA		NI A	NA	\$4.54
++	2-Wire Voice Grade Loop (SL1)	UEFAE	INA	NA	INA	NA	NA	NA	NA	INA	\$4.54
$\vdash$	RC - 2- Wire Voice Grade Loop (SLT)	UEPLX	NA	NA	NA	NA	NA	NA	\$14.18	NA	NA
++	RC - 2- Wire Voice Grade Loop - Statewide  RC - 2- Wire Voice Grade Loop - Zone 1	UEPLX	\$14.35	\$14.90	\$10.80	\$14.79	\$14.05	\$14.59	NA	\$17.02	\$12.48
++	RC - 2- Wire Voice Grade Loop - Zone 2	UEPLX	\$23.31	\$18.51	\$12.47	\$27.68	\$24.14	\$19.33	NA NA	\$25.66	\$14.42
++	RC - 2- Wire Voice Grade Loop - Zone 3	UEPLX	\$42.24	\$24.25	\$19.83	\$47.78	\$49.30	\$27.63	NA NA	\$33.99	\$21.77
++	RC - 2- Wire Voice Grade Loop - Zone 4	UEPLX	NA	NA	NA	NA NA	NA	\$36.47	NA NA	NA NA	NA NA
++	Combination Rates	02. 27	1471	1471	1471	1471	1471	ψου.+1	14/ (	1471	101
H	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Statewide	Note 8	NA	NA	NA	NA	NA	NA	\$16.46	NA	NA
$\vdash$	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 1 (Note 6)	Note 8	\$16.55	\$16.25	\$12.59	\$17.40	\$16.60	\$16.71	NA	\$20.71	\$17.02
$\vdash$			+ . 5.00					\$21.45	NA NA	\$29.35	\$18.96
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 2 (Note 6)	Note 8	\$25.51	\$19.86	\$14.26	\$30.29	\$26.69				
$\vdash \vdash$	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 2 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)	Note 8 Note 8	\$25.51 \$44.44	\$19.86 \$25.60	\$14.26 \$21.62	\$30.29 \$50.39	\$26.69 \$51.85	\$29.75	NA NA	\$37.68	\$26.31
Ħ											
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)	Note 8	\$44.44	\$25.60	\$21.62	\$50.39	\$51.85	\$29.75	NA	\$37.68	\$26.31
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)	Note 8	\$44.44	\$25.60	\$21.62	\$50.39	\$51.85	\$29.75	NA	\$37.68	\$26.31
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6) Nonrecurring Charges	Note 8 Note 8	\$44.44 NA	\$25.60 NA	\$21.62 NA	\$50.39 NA	\$51.85 NA	\$29.75 \$38.59	NA NA	\$37.68 NA	\$26.31 NA
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is	Note 8 Note 8 USAC2	\$44.44 NA \$2.80	\$25.60 NA \$0.1964	\$21.62 NA \$2.01	\$50.39 NA \$10.00	\$51.85 NA \$3.80	\$29.75 \$38.59 \$5.20	NA NA \$2.77	\$37.68 NA \$1.59	\$26.31 NA \$1.03
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is	Note 8 Note 8 USAC2 USAC2	\$44.44 NA \$2.80 \$0.41	\$25.60 NA \$0.1964 \$0.1964	\$21.62 NA \$2.01 \$0.3108	\$50.39 NA \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29	\$29.75 \$38.59 \$5.20 \$0.41	NA NA \$2.77 \$0.40	\$37.68 NA \$1.59 \$0.40	\$26.31 NA \$1.03 \$0.2886
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change	Note 8 Note 8 USAC2 USAC2 USACC USACC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41	\$25.60 NA \$0.1964 \$0.1964 \$0.1964 \$0.1964	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108	\$50.39 NA \$10.00 \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41	NA NA \$2.77 \$0.40 \$2.77	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40	\$26.31 NA \$1.03 \$0.2886 \$1.03
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change	Note 8 Note 8 USAC2 USAC2 USACC	\$44.44 NA \$2.80 \$0.41 \$2.80	\$25.60 NA \$0.1964 \$0.1964 \$0.1964	\$21.62 NA \$2.01 \$0.3108 \$2.01	\$50.39 NA \$10.00 \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29 \$3.80	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20	NA NA \$2.77 \$0.40 \$2.77	\$37.68 NA \$1.59 \$0.40 \$1.59	\$26.31 NA \$1.03 \$0.2886 \$1.03
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	Note 8 Note 8 USAC2 USAC2 USACC USACC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41	\$25.60 NA \$0.1964 \$0.1964 \$0.1964 \$0.1964	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108	\$50.39 NA \$10.00 \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41	NA NA \$2.77 \$0.40 \$2.77	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40	\$26.31 NA \$1.03 \$0.2886 \$1.03
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic,	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC USACC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00	\$25.60 NA \$0.1964 \$0.1964 \$0.1964 \$0.1964 \$10.00	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00	\$50.39 NA \$10.00 \$10.00 \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00	NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	Note 8 Note 8 USAC2 USAC2 USACC USACC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41	\$25.60 NA \$0.1964 \$0.1964 \$0.1964 \$0.1964	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108	\$50.39 NA \$10.00 \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41	NA NA \$2.77 \$0.40 \$2.77	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40	\$26.31 NA \$1.03 \$0.2886 \$1.03
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7) NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual	Note 8 Note 8 USAC2 USAC2 USAC2 USACC USACC USACC SOMEC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00 \$3.50	\$25.60 NA \$0.1964 \$0.1964 \$0.1964 \$0.1964 \$10.00 \$2.75	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00 \$3.50	\$50.39 NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$3.50	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00 \$3.50	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00 \$3.50	NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00 \$3.50	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00 \$3.50
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7) NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc. Order vs. Electronic - 1st	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC USACC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00	\$25.60 NA \$0.1964 \$0.1964 \$0.1964 \$0.1964 \$10.00	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00	\$50.39 NA \$10.00 \$10.00 \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00	NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add¹l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add¹l, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC USACC USAS2 SOMEC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00 \$3.50	\$25.60 NA \$0.1964 \$0.1964 \$0.1964 \$10.00 \$2.75	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00 \$3.50	\$50.39 NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.90 \$10.90	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00 \$3.50	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00 \$3.50 \$43.52	NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00 \$3.50	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00 \$3.50 \$43.19	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00 \$3.50
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7) NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc. Order vs. Electronic - 1st	Note 8 Note 8 USAC2 USAC2 USAC2 USACC USACC USACC SOMEC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00 \$3.50	\$25.60 NA \$0.1964 \$0.1964 \$0.1964 \$0.1964 \$10.00 \$2.75	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00 \$3.50	\$50.39 NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$3.50	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00 \$3.50	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00 \$3.50	NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00 \$3.50	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update										
Manual Service Order		\$8.25	TDD	TBD	TBD	<b>05.40</b>	ФС 00	¢40.07	CO 04	<b>\$7.07</b>
NRC - Electronic Service Order Disconnect		NA	TBD \$0.42	NA	NA NA	\$5.12 NA	\$6.88 NA	\$10.27 NA	\$8.91 NA	\$7.97 NA
NRC - Incremental Manual Service Order Disconnect		\$20.00	\$3.84	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00
		\$20.00	\$3.84	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00
NRCs for New (not Currently Combined) as ordered in Georgia:	LIEDDI	NIA	NIA	COO 44	NIA	NIA.	NIA	NIA	NIA	NIA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPRL	NA	NA	\$22.14	NA	NA	NA	NA NA	NA	NA NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPRL	NA	NA	\$15.25	NA	NA	NA	NA NA	NA	NA NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPRC	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPRC	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPRO	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPRO	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPAP	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPAP	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPBL	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPBL	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPBC	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPBC	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPBO	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPBO	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPB1	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
					1	ĺ				
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPB1	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
THE 2 THE FOLSE STAGE 2009 THAT 2 THE 2HOT ON THEM THAT	02. 5.	1471	1471	ψ10.20	1471	14/1	147.	14/1	101	101
					1	ĺ	1	1		
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	USAS2	NA	NA	\$10.00	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Disconnect - 1st		NA	NA	\$8.45	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Disconnect - Add'l		NA	NA	\$3.91	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic,										
per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - Incremental Cost Manual vs.										
Electronic - New - 1st		NA	NA	\$37.06	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - Incremental Cost Manual vs.				4						
Electronic - New - Add'l		NA	NA	\$8.19	NA	NA	NA	NA	NA	NA
NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update	•									
Electronic		NA	NA	TBD	NA	NA	NA	NA	NA	NA
NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update	•									
Manual Service Order		NA	NA	TBD	NA	NA	NA	NA	NA	NA
NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - Incremental Cost Manual vs.		l			l	l		l		
Electronic - New - Disconnect		NA	NA	\$11.17	NA	NA	NA	NA	NA	NA
					1	ĺ				
2- Wire Voice Grade Loop - Bus Only with 2 -Wire DID Trunk Port					<u> </u>					
2 - Wire Line Port - DID Trunk Port, per month	UEPD1	TBD	\$9.36	\$11.35	TBD	\$13.12	\$14.63	\$12.12	TBD	\$8.78
2-Wire Voice Grade Loop (SL2)										
RC - 2- Wire Voice Grade Loop - Statewide	UECD1	NA	NA	NA	NA	NA	NA	\$11.76	NA	NA
RC - 2- Wire Voice Grade Loop - Zone 1	UECD1	\$17.95	\$18.48	\$16.84	\$17.27	\$17.65	\$18.35	NA	\$21.57	\$9.60
RC - 2- Wire Voice Grade Loop - Zone 2	UECD1	\$29.16	\$22.43	\$19.45	\$32.32	\$30.32	\$24.33	NA	\$32.53	\$11.09
RC - 2- Wire Voice Grade Loop - Zone 3	UECD1	\$52.84	\$27.87	\$30.92	\$55.78	\$61.93	\$34.77	NA	\$43.08	\$16.74
RC - 2- Wire Voice Grade Loop - Zone 4	UECD1	NA	NA	NA	NA	NA	\$45.88	NA	NA	NA
Combination Rates										
RC - 2-Wire Voice Grade Loop with 2-Wire DID Port, Statewide	Note 8	NA	NA	NA	NA	NA	NA	\$23.79	NA	NA
RC - 2-Wire Voice Grade Loop with 2-Wire DID Port, Zone 1 (Note 6)	Note 8	TBD	\$27.84	\$28.19	TBD	\$30.77	TBD	NA	TBD	\$18.38
RC - 2-Wire Voice Grade Loop with 2-Wire DID Port, Zone 2 (Note 6)	Note 8	TBD	\$31.79	\$30.80	TBD	\$43.44	TBD	NA	TBD	\$19.87
RC - 2-Wire Voice Grade Loop with 2-Wire DID Port, Zone 3 (Note 6)	Note 8	TBD	\$37.23	\$42.27	TBD	\$75.05	TBD	NA	TBD	\$25.52
RC - 2-Wire Voice Grade Loop with 2-Wire DID Port, Zone4 (Note 6)	Note 8	NA	NA	NA	NA	NA	TBD	NA	NA	NA
NRC- 2- Wire Voice Grade Loop with 2- Wire DID Port - 1st	USACC	TBD	\$14.62	\$166.08	TBD	TBD	TBD	\$13.26	TBD	\$8.76

Attachment 2 Exhibit C

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
NRC- 2- Wire Voice Grade Loop with 2- Wire DID Port - Addl	USACC	TBD	\$3.73	\$140.01	TBD	TBD	TBD	\$8.39	TBD	\$5.75
NRC - 2-Wire DID Subsequent Activity - Add Trunks, Per Trunk	USAS2	NA NA	\$53.57	NA	NA	NA NA	NA NA	NA	NA NA	NA
			400.00							
NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic,										
per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
NRC- 2- Wire Voice Grade Loop with 2- Wire DID Port - Incremental Cost- Manual										
Service Order - 1st	SOMAN	\$19.99	\$21.56	\$37.88	\$19.99	\$19.99	\$19.99	\$53.89	\$19.99	\$41.43
NRC- 2- Wire Voice Grade Loop with 2- Wire DID Port - Incremental Cost- Manual										
Service Order - Addl	SOMAN	\$19.99	\$21.56	\$16.84	\$19.99	\$19.99	\$19.99	\$11.34	\$19.99	\$9.80
NRC - Electronic Service Order Disconnect		NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
NRC - Incremental Manual Service Order Disconnect		\$20.00	\$3.84	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00
2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Port										
2-wire ISDN Digital Port per month	UEPPB	\$16.42	\$8.51	\$13.47	\$12.33	\$23.33	\$51.91	\$24.37	\$33.74	\$18.21
2-Wire ISDN Digital Grade Loop										
RC - 2-Wire ISDN Digital Grade Loop - Statewide	USL2X	NA	NA	NA	NA	NA	NA	\$19.08	NA	NA
RC - 2-Wire ISDN Digital Grade Loop - Statewide	USL2X	\$23.23	\$22.48	\$21.89	\$23.66	\$21.15	\$21.86	NA	\$26.68	\$16.20
RC - 2-Wire ISDN Digital Grade Loop - Zone 2	USL2X	\$37.74	\$27.90	\$25.27	\$44.28	\$36.32	\$28.97	NA NA	\$40.24	\$18.71
RC - 2-Wire ISDN Digital Grade Loop - Zone 3	USL2X	\$68.38	\$30.78	\$40.17	\$76.42	\$74.19	\$41.40	NA	\$53.29	\$28.25
RC - 2-Wire ISDN Digital Grade Loop - Zone 4	USL2X	NA NA	NA NA	NA	NA	NA	\$54.64	NA	NA	NA NA
i i										
Combination Rates								A 10 15		
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Port - Statewide	Note 8	NA Too of	NA Tagana	NA Dos co	NA Dos co	NA	NA ATO TT	\$43.45	NA Doc 40	NA
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Port - Zone 1	Note 8	\$39.65	\$30.99	\$35.36	\$35.99	\$44.48	\$73.77	NA	\$60.42	\$34.41
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Port - Zone 2  RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Port - Zone 3	Note 8	\$54.16	\$36.41	\$38.74	\$56.61	\$59.65	\$80.78	NA	\$73.98	\$36.92
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Port - Zone 3  RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Port - Zone 4	Note 8	\$84.80	\$39.30	\$53.64	\$88.75	\$97.52	\$93.31	NA NA	\$87.03	\$46.46
NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Port - 1st conversion	Note 8 USACB	NA TBD	NA \$86.79	NA TBD	NA TBD	NA TBD	\$106.55 TBD	\$174.35	NA TBD	NA \$117.23
INRC - 2-VVIIe ISDIN DIGITAL GIAGE LOOP/2-WIFE ISDIN DIGITAL PORT - 1St conversion	USACB	IBD	\$86.79	IBD	IBD	IBD	IBD	\$174.35	IBD	\$117.23
NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Port - Add'l conversion	USACB	TBD	\$54.04	TBD	TBD	TBD	TBD	\$174.35	TBD	\$117.23
NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Port - Non Feature	USACB	טפו	φ34.04	טפו	IBD	IDD	עסו	\$174.33	טפו	\$117.23
Subsequent Activity	USASB	TBD	TBD	TBD	TBD	TBD	TBD	\$286.15	TBD	\$212.88
- Cubscoquent / touvity	OOAOD	100	100	100	100	100	100	Ψ200.13	100	ΨΖ12.00
NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic,										
per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Port - Incremental Cost-		40.00	<del></del>	40.00	70.00	40.00	40.00	40.00	70.00	*****
Manual Service Order - 1st		\$19.99	\$21.56	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99
NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Port - Incremental Cost-										
Manual Service Order - Addl		\$19.99	\$21.56	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99
NRC - Electronic Service Order Disconnect		TBD	\$0.42	TBD	TBD	TBD	TBD	TBD	TBD	TBD
NRC - Incremental Manual Service Order Disconnect		\$20.00	\$3.84	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00
			· · · · · · · · · · · · · · · · · · ·							
4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port										
4 - Wire ISDN DS1 Digital Trunk Port	UEPPP	\$186.02	\$95.39	\$163.16	\$275.48	\$194.72	\$213.21	\$179.01	\$214.79	\$78.40
4 - Wire DS1 Digital Loop										
RC - 4- Wire DS1 Digital Loop- Statewide	USL4P	NA	NA	NA	NA	NA	NA	\$62.71	NA	NA
RC - 4- Wire DS1 Digital Loop- Zone 1	USL4P	\$51.74	\$92.48	\$55.53	\$50.26	\$56.32	\$50.99	NA	\$59.61	\$57.73
RC - 4- Wire DS1 Digital Loop- Zone 2	USL4P	\$84.05	\$119.68	\$64.13	\$94.06	\$96.73	\$67.58	NA	\$89.90	\$75.40
RC - 4- Wire DS1 Digital Loop- Zone 3	USL4P	\$152.29	\$194.70	\$101.93	\$162.34	\$197.57	\$96.58	NA	\$119.06	\$98.59
RC - 4-Wire DS1 Digital Loop - Zone 4	USL4P	NA	NA	NA	NA	NA	\$127.47	NA	NA	NA
Combination Rates			· · · · · · · · · · · · · · · · · · ·							
RC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port - Statewide	Note 8	NA	NA	NA	NA	NA	NA	\$241.72	NA	NA
		0007.70	A407.07	0040.00	\$325.74	\$251.04	\$264.20	NA	CO74 40	\$136.13
RC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port - Zone 1	Note 8	\$237.76	\$187.87	\$218.69					\$274.40	
RC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port - Zone 1  RC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port - Zone 2  RC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port - Zone 3	Note 8 Note 8 Note 8	\$237.76 \$270.07 \$338.31	\$187.87 \$215.07 \$290.08	\$218.69 \$227.29 \$265.09	\$369.54 \$437.82	\$291.45 \$392.29	\$280.79 \$309.79	NA NA	\$274.40 \$304.69 \$333.85	\$153.80 \$176.99

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
RC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port - Zone 4	Note 8	NA	NA	NA	NA	NA	\$340.68	NA	NA	NA
NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port Combination - 1st		1								
conversion	USACP	TBD	\$247.97	TBD	TBD	TBD	TBD	\$481.51	TBD	\$328.53
NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port Combination - Add'l										
conversion	USACP	TBD	\$157.17	TBD	TBD	TBD	TBD	\$481.51	TBD	\$328.53
NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port -										
Subsequent Channel Activation - Per Channel	USASP	TBD	\$29.06	TBD	TBD	TBD	TBD	\$36.92	TBD	\$28.39
NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port Combination -										
Subsequent Inward/2-way Telephone Numbers	PR7TG	TBD	\$0.9804	TBD	TBD	TBD	TBD	\$1.17	TBD	\$0.9353
NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port Combination -										
Subsequent Outward Telephone numbers	PR7TP	TBD	\$23.02	TBD	TBD	TBD	TBD	\$28.17	TBD	\$22.36
NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port Combination -										<b>.</b>
Subsequent Inward Telephone Numbers	PR7ZT	TBD	\$46.05	TBD	TBD	TBD	TBD	\$56.33	TBD	\$44.71
NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port Combination -	110400	TDD	TDD	TDD	TDD	TDD	TDD	0055.05	TDD	<b>0.100 70</b>
Subsequent Service Order Per Order	USASP	TBD	TBD	TBD	TBD	TBD	TBD	\$255.25	TBD	\$189.76
NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic,										
per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN Digital Port - Incremental Cost-	SOMEC	\$3.50	<b>\$2.75</b>	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
Manual Service Order - 1st		\$19.99	\$21.56	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99
NRC - 4-Wire ISDN Digital Loop with 4-wire ISDN Digital Port - Incremental Cost-		ψ13.33	ψ2 1.30	ψ13.33	ψ13.33	ψι.συ συ.συ	ψ13.33	ψ13.33	ψ13.33	ψ13.33
Manual Service Order - Addl		\$19.99	\$21.56	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99
NRC - Electronic Service Order Disconnect		TBD	\$0.42	TBD	TBD	TBD	TBD	TBD	TBD	TBD
NRC - Incremental Manual Service Order Disconnect		\$20.00	\$3.84	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00
1410 Indianativalida ecivide etda bisconiect		Ψ20.00	ψ5.04	Ψ20.00	Ψ20.00	Ψ20.00	Ψ20.00	Ψ20.00	Ψ20.00	Ψ20.00
4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port		+								
4 - Wire DDITS Digital Coop With 4 - Wire DDITS Trunk Port  4 - Wire DDITS Digital Trunk Port (Formerly DID Trunk Port)	UDD1T	\$130.23	\$63.31	\$120.80	\$83.28	\$149.27	\$146.46	\$123.65	TBD	\$120.00
4 - Wire DS1 Digital Loop	USLDC	\$130.23	φυσ.σ ι	\$120.00	φου.20	\$149.27	\$140.40	\$123.00	IBD	\$120.00
4 - Wire DS1 Digital Loop - Statewide	USLDC	NA	NA		NA	NA	NA	\$62.71	NA	NA
4 - Wire DS1 Digital Loop - Statewate	USLDC	\$51.74	\$64.69	\$55.53	\$106.04	\$56.32	\$50.99	NA	\$59.61	\$57.73
4 - Wire DS1 Digital Loop - Zone 2	USLDC	\$84.05	\$94.71	\$64.13	\$135.15	\$96.73	\$67.58	NA NA	\$89.90	\$75.40
4 - Wire DS1 Digital Loop - Zone 3	USLDC	\$152.29	\$208.93	\$101.93	\$186.26	\$197.57	\$96.58	NA NA	\$119.06	\$98.59
4 - Wire DS1 Digital Loop - Zone 4	USLDC	NA	NA	NA	NA	NA	\$127.47	NA NA	NA	NA
Combination Rates	USLDC	INA	INA	INA	INA	INA	\$127.47	INA	INA	INA
4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Statewide	Note 8	NA	NA	NA	NA	NA	NA	\$186.36	NA	NA
4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Zone 1	Note 8	\$181.97	\$128.00	\$176.33	\$189.32	\$205.59	\$197.45	NA	TBD	\$93.28
4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Zone 2	Note 8	\$214.28	\$158.02	\$184.93	\$218.43	\$246.00	\$214.04	NA NA	TBD	\$110.95
4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Zone 3	Note 8	\$282.52	\$272.24	\$222.73	\$269.54	\$346.84	\$243.04	NA NA	TBD	\$134.14
4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Zone 4	Note 8	NA	NA	NA NA	NA	NA	\$273.93	NA NA	NA NA	NA NA
1 So I signal book was I thin boll of family of Long 4	14016-0	14/3	11/3	14/5	14/5	14/7	Ψ210.00	14/5	14/7	14/7
Local number Portability per DSO Activated	LNPCP	\$3.15	\$3.15	\$3.15	\$3.15	\$3.15	\$3.15	\$3.15	\$3.15	\$3.15
Central Office Terminating Point	CTG	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	0	Ţ-1.00	<del>+00</del>	Ţ3.00	+ 5.00	Ţ2.00	+2.00	+5.00	Ţ2.00	+5.00
Telephone Number / Trunk Group establishment		†			İ					
Telephone Number for 2-Way Trunk Group	UDTGX	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Telephone Number for 1-Way Outward Trunk Group	UDTGY	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Telephone Number for 1-Way Inward Trunk Group Without DID	UDTGZ	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers (FL,			** **			*		*		****
GA, NC, & SC only)	NDZ	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers (AL,										
KY, LA, MS, & TN). In addition, Provides Additional DID Numbers for each Group of 20		1								
DID Numbers (Valid in All States)	ND4	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
DID Numbers, Non- consecutive DID Numbers , Per Number	ND5	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Interoffice Channel Mileage - (Dedicated DS1) FX/FCO for 4 - Wire DS1 Digital		1 7								
Loop with 4 - Wire DDITS Trunk Port										
Fixed cost 0-8 miles (Facilities Termination)	1LNO1	\$79.69	\$92.62	\$63.39	\$55.05	\$93.40	\$74.40	\$71.29	\$94.98	\$75.83
Additional costs per mile 0-8 miles	1LNOA	\$0.6920	\$0.2000	\$0.3068	\$0.4500	\$0.7831	\$0.6598	\$0.0783	\$0.7598	\$0.3525
Fixed cost 9-25 miles (Facilities Termination)	1LN02	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00

LI	RIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	additional costs per mile 9-25 miles	1LNOB	\$0.6920	\$0.2000	\$0.3068	\$0.4500	\$0.7831	\$0.6598	\$0.0783	\$0.7598	\$0.3525
	Fixed cost 25 + miles (Facilities Termination)	1LNO3	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
lacksquare	Additional costs 25 + miles	1LNOC	\$0.6920	\$0.2000	\$0.3068	\$0.4500	\$0.7831	\$0.6598	\$0.0783	\$0.7598	\$0.3525
$\vdash$	Enhanced Performance Charges										
	Enhanced Performance Charges - as negotiated in contract	UDTPC	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
++	Elimanosa i circimanos charges de negotiatea in contract	ODITO	IBN	IDIA	TDIA	TDIN	IDIN	TDIA	IDIA	TDIA	IDIN
H	Non-recurring Charges									1	
	NRC - 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port Combination - OSS LSR										
	Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	NRC- 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port - Incremental Cost- Manual Service Order - 1st	SOMAN	\$19.99	\$21.56	\$37.88	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99
	NRC- 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port - Incremental Cost-										
	Manual Service Order - Add'l	SOMAN	\$19.99	\$21.56	\$16.84	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99
	NRC - Electronic Service Order Disconnect		TBD	\$0.42	TBD	TBD	TBD	TBD	TBD	TBD	TBD
	NRC - Incremental Manual Service Order Disconnect		\$20.00	\$3.84	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port -Conversion - Switch										
	as is - 1st	USAC4	TBD	\$268.82	\$269.96	\$261.15	\$266.76	TBD	\$490.38	TBD	\$312.91
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Conversion - Switch as is - Additional	USAC4	TBD	\$134.07	\$269.96	\$134.08	\$123.16	TBD	\$490.38	TBD	\$312.91
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Conversion with DS1 changes - 1st	USAWA	TBD	\$268.82	\$269.96	\$261.15	\$266.76	TBD	\$490.38	TBD	\$312.91
+	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Conversion with DS1	USAWA	IBD	\$200.02	\$209.90	Φ201.13	\$200.70	160	φ <del>4</del> 90.36	100	φ312.91
	Changes - Additional	USAWA	TBD	\$134.07	\$269.96	\$134.08	\$123.16	TBD	\$490.38	TBD	\$312.91
+	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Conversion with		1	********	<b>4</b>	Ţ	<b>V</b> .= <b>U</b>		***************************************	1	*******
	Change - Trunks - 1st	USAWB	TBD	\$268.82	\$269.96	\$261.15	\$266.76	TBD	\$490.38	TBD	\$312.91
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Conversion with										
	Change - Trunks - Additional	USAWB	TBD	\$134.07	\$269.96	\$134.08	\$123.16	TBD	\$490.38	TBD	\$312.91
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Digital Trunk Port - Subsequent				4						
—	Service Activity Per Service Order	USAS4	TBD	TBD	\$147.47	TBD	\$68.57	TBD	\$127.63	TBD	\$94.88
+	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - New - 1st	UDDIT	NA	NA	\$858.30	NA	NA	NA	NA	NA	NA
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - New - Additional	UDDIT	NA	NA	\$514.02	NA	NA	NA	NA	NA	NA
+	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - New - 1st -	ODDII	14/4	IVA	ψ514.02	IVA	14/3	INA	IVA	14/5	IVA
	Disconnect	UDDIT	NA	NA	TBD	NA	NA	NA	NA	NA	NA
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - New - Additional		1							1	
	Disconnect	UDDIT	TBD	NA	TBD	TBD	NA	TBD	\$11.98	TBD	\$15.29
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Subsequent Channel										
丄	Activation - Per Channel - 2-Way Trunk	UDTTA	TBD	\$28.96	\$28.71	\$28.96	\$26.60	TBD	\$146.91	TBD	\$108.67
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Subsequent Channel								<u> </u>		
	Activation - Per Channel - 1-Way Outward Trunk	UDTTB	TBD	\$28.96	\$28.71	\$28.96	\$26.60	TBD	\$146.91	TBD	\$108.67
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Subsequent Channel Activation - Per Channel - 1-Way Inward Trunk Without DID	UDTTC	TBD	\$28.96	\$28.71	\$28.96	\$26.60	TBD	\$146.91	TBD	\$108.67
+	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Subsequent Channel	UDITO	IDU	φ <b>∠</b> 0.90	φ20./ Ι	Φ∠0.90	φ20.00	IBD	\$140.91	IDU	φ100.01
	Activation - Per Channel - 1-Way Inward Trunk With DID	UDTTD	TBD	\$28.96	\$28.71	\$28.96	\$26.60	TBD	\$146.91	TBD	\$108.67
+	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Subsequent Channel	ODITO	100	Ψ20.30	Ψ20.7 1	Ψ20.30	Ψ20.00	100	ψ170.31	100	ψ100.07
	Activation - Per Channel - 2-Way DID with User Transfer	UDTTE	TBD	\$28.96	\$28.71	\$28.96	\$26.60	TBD	\$146.91	TBD	\$108.67
1	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Digital Trunk Port - Subsequent		1					İ		1	
	Signaling Changes		TBD	TBD	TBD	TBD	TBD	TBD	\$29.65	TBD	\$22.92
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Digital Trunk Port - Subsequent										
$\perp$	Telephone Numbers		TBD	TBD	TBD	TBD	TBD	TBD	\$120.96	TBD	\$88.68
T	NRC - Interoffice Channel Mileage - (Dedicated DS1) FX/FCO for 4 - Wire DS1 Digital			· · · · · · · · · · · · · · · · · · ·							<del></del>
	Loop with 4 - Wire DDITS Trunk PortFixed cost 0-8 miles (Facilities Termination) - 1st -	41.110.4				l l		l	l	I I	
1	New Only	1LNO1	\$198.15	NA	\$147.07	NA	NA	NA	NA	NA	NA
$\perp$	NDO 1-4				•	1		1	ı		
+	NRC - Interoffice Channel Mileage - (Dedicated DS1) FX/FCO for 4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk PortFixed cost 0-8 miles (Facilities Termination) -										

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
NRC - Interoffice Channel Mileage - (Dedicated DS1) FX/FCO for 4 - Wire DS1 Digital				_			_	_		
Loop with 4 - Wire DDITS Trunk PortFixed cost 0-8 miles (Facilities Termination) -										
Disconnect - 1st - New Only	1LNO3	\$25.44	NA							
NRC - Interoffice Channel Mileage - (Dedicated DS1) FX/FCO for 4 - Wire DS1 Digital										
Loop with 4 - Wire DDITS Trunk PortFixed cost 0-8 miles (Facilities Termination) -										
Disconnect Additional - New Only	1LNO4	\$20.42	NA							
BIPOLAR 8 ZERO SUSTITUTION									1	
NRC - Superframe Format - Conversion or new install 1st	CCOSF	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
NRC - Superframe Format - Conversion or new install Additional	CCOSF	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
NRC - Extended Superframe Format - Change or Subsequent Activity - 1st	CCOSF	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
NRC - Extended Superframe Format - Change or Subsequent Activity - Additional	CCOSF	\$600.00	\$655.00	\$600.00	\$730.00	\$605.00	\$600.00	\$615.00	\$605.00	\$590.00
NRC - Extended Superframe Format - Conversion or New Install 1st	CCOEF	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
NRC - Extended Superframe Format - Conversion or New Install - Additional	CCOEF	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
NRC - Extended Superframe Format - Change or Subsequent Activity - 1st	CCOEF	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
NRC - Extended Superframe Format - Change or Subsequent Activity - Additional	CCOEF	\$600.00	\$655.00	\$600.00	\$730.00	\$605.00	\$600.00	\$615.00	\$605.00	\$590.00
Alternate Mark Inversion (AMI)					1				+	
NRC - Superframe Format - 1st	MCOSF	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
NRC - Superframe Format - Additional	MCOSF	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
NRC - Extended Superframe Format - 1st	MCOSF	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
NRC - Extended Superframe Format - Additional	MCOSF	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
The Extended experimenter similar radiations.		ψο.σσ	ψ0.00	ψ0.00	ψο.σσ	ψ0.00	ψ0.00	ψο.σσ	ψ0.00	Ψοίου
2-Wire Voice Grade Loop with 2-Wire Line Port PBX										
2-Wire Analog Line Port (PBX), per month										
2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence	UEPRD	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS	UEPPC	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS	UEPPO	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS	UEPP1	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING										
PORT	UEPA2	\$2.20	NA							
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING										
PORT	UEPL2	NA	NA	NA	NA	\$2.55	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS	UEPLD	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING										
PORT	UEPT2	NA	\$4.54							
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING	LIERTO	N/A	NIA		,,,	NIA.	l	NIA.	N10	04.54
PORT  2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT	UEPTO UEPXA	NA \$2.20	NA \$1.35	NA C4.70	NA \$2.61	NA \$2.55	NA Co.40	NA \$2.28	NA \$3.69	\$4.54
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT				\$1.79			\$2.12	7		\$4.54
2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS  2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT	UEPXB UEPXC	\$2.20 \$2.20	\$1.35 \$1.35	\$1.79 \$1.79	\$2.61 \$2.61	\$2.55 \$2.55	\$2.12 \$2.12	\$2.28 \$2.28	\$3.69 \$3.69	\$4.54 \$4.54
2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT	UEPXC	\$2.20 \$2.20	\$1.35 \$1.35	\$1.79 \$1.79	\$2.61 \$2.61	\$2.55 \$2.55	\$2.12 \$2.12	\$2.28 \$2.28	\$3.69	\$4.54 \$4.54
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD FOR I	UEPAD	φ∠.∠∪	φ1.30	φ1./9	φ∠.01	φ∠.33	φ∠. Ι∠	φ∠.∠0	φ3.09	φ4.54
PORT	UEPXE	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING	OLFAL	Ψ2.20	ψ1.33	ψ1.13	ΨΖ.ΟΙ	Ψ2.00	ΨΕ. ΙΖ	Ψ2.20	ψ5.05	ψ+.J+
PORT WITHOUT LUD	UEPXF	NA	NA	NA	\$2.61	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT	UEPXG	NA NA	NA NA	NA NA	\$2.61	NA NA	NA NA	NA NA	NA NA	NA NA
2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT	UEPXH	NA NA	NA NA	NA NA	\$2.61	NA NA	NA NA	NA NA	NA NA	NA NA
2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT WITHOUT		1			Ţ		l		1	
LUD	UEPXJ	NA	NA	NA	\$2.61	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL CALLING		1			<del></del>		1		1	
	UEPXK	NA	NA	NA	NA	\$2.55	NA	NA	NA	NA
PORT										
PORT   2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY	OLI XIII									

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
	0000	7.2		<u> </u>					- 55	
CALLING PORT	UEPXM	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL		¥====	******	*	<b>V</b>	<del></del>	<b>*</b>	<b>4</b> ==0	- Variati	****
ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT	UEPXN	NA	NA	NA	NA	NA	NA	NA	NA	\$4.54
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL										-
DIACOUNT ROOM CALLING PORT	UEPXO	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL										
DISCOUNT CALLING PORT	UEPXP	NA	NA	NA	NA	\$2.55	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY										
CALLING PORT	UEPXQ	NA	NA	NA	NA	NA	\$2.12	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL										
CALLING PORT	UEPXR	NA	NA	NA	NA	NA	\$2.12	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT	UEPXS	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS										
CALLING PORT	UEPXT	NA	NA	NA	NA	NA	NA	NA	\$3.69	NA
2 WIDE VOICE LINDUNDLED DRY COLLIEDVILLE & MEMBLIE CALLING DODT	LIEDVII	NIA	NIA	NIA	NIA	NIA	NIA.	N10	NIA.	04.54
2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY PBX TENNESSEE REGIONSERV CALLING	UEPXU	NA	NA	NA	NA	NA	NA	NA	NA	\$4.54
PORT	UEPXV	NA	NA	NA	NA	NA	NA	NA	NA	\$4.54
	UEFAV	INM	INA	INA	INA	INA	INA	INA	INA	φ4.54
LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	LNPCP	+			<u> </u>		1		<u> </u>	
ECOAL NOMBERT ORTABIETT (REQUIRES ONE FERT ORT)	LITT OI	+								
2-Wire Voice Grade Loop (SL1)		1							<b>†</b>	
RC - 2- Wire Voice Grade Loop - Statewide	UEPLX	NA	NA	NA	NA	NA	NA	\$14.18	NA	NA
RC - 2- Wire Voice Grade Loop - Zone 1	UEPLX	\$14.35	\$14.90	\$10.80	\$14.79	\$14.05	\$14.59	NA	\$17.02	\$12.48
RC - 2- Wire Voice Grade Loop - Zone 2	UEPLX	\$23.31	\$18.51	\$12.47	\$27.68	\$24.14	\$19.33	NA	\$25.66	\$14.42
RC - 2- Wire Voice Grade Loop - Zone 3	UEPLX	\$42.24	\$24.25	\$19.83	\$47.78	\$49.30	\$27.63	NA	\$33.99	\$21.77
RC - 2- Wire Voice Grade Loop - Zone 4	UEPLX	NA	NA	NA	NA	NA	\$36.47	NA	NA	NA
Combination Rates										
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Statewide	Note 8	NA	NA	NA	NA	NA	NA	\$16.46	NA	NA
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 1 (Note 6)	Note 8	\$16.55	\$16.25	\$12.59	#VALUE!	\$16.60	\$16.71	NA	\$20.71	\$17.02
		1								
PC - 2 Wire Voice Grade Leep with 2 Wire Line Bott Zone 2 (Note 6)	Note 8	\$25.51	\$10.86	\$14.26	#\/^   E	\$26.60	¢21.45	NΛ	\$20.35	\$18.06
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 2 (Note 6)	Note 8	\$25.51	\$19.86 \$25.60	\$14.26 \$21.62	#VALUE!	\$26.69 \$51.85	\$21.45 \$20.75	NA NA	\$29.35 \$37.68	\$18.96 \$26.31
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)	Note 8	\$44.44	\$25.60	\$21.62	#VALUE!	\$51.85	\$29.75	NA	\$37.68	\$26.31
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)										
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6) Nonrecurring Charges	Note 8 Note 8	\$44.44 NA	\$25.60 NA	\$21.62 NA	#VALUE! NA	\$51.85 NA	\$29.75 \$38.59	NA NA	\$37.68 NA	\$26.31 NA
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)	Note 8	\$44.44	\$25.60	\$21.62	#VALUE!	\$51.85	\$29.75	NA	\$37.68	\$26.31
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6) Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is	Note 8 Note 8 USAC2	\$44.44 NA \$2.80	\$25.60 NA \$15.82	\$21.62 NA \$2.01	#VALUE! NA \$10.00	\$51.85 NA \$3.80	\$29.75 \$38.59 \$5.20	NA NA \$2.77	\$37.68 NA \$1.59	\$26.31 NA \$1.03
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6) Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is	Note 8 Note 8 USAC2 USAC2	\$44.44 NA \$2.80 \$0.41	\$25.60 NA \$15.82 \$3.80	\$21.62 NA \$2.01 \$0.3108	#VALUE! NA \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29	\$29.75 \$38.59 \$5.20 \$0.41	NA NA \$2.77 \$0.40	\$37.68 NA \$1.59 \$0.40	\$26.31 NA \$1.03 \$0.2886
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6) Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is	Note 8 Note 8 USAC2 USAC2	\$44.44 NA \$2.80 \$0.41	\$25.60 NA \$15.82 \$3.80	\$21.62 NA \$2.01 \$0.3108	#VALUE! NA \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29	\$29.75 \$38.59 \$5.20 \$0.41	NA NA \$2.77 \$0.40	\$37.68 NA \$1.59 \$0.40	\$26.31 NA \$1.03 \$0.2886
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change	Note 8 Note 8 USAC2 USAC2 USACC	\$44.44 NA \$2.80 \$0.41 \$2.80	\$25.60 NA \$15.82 \$3.80 \$15.82	\$21.62 NA \$2.01 \$0.3108 \$2.01	#VALUE! NA \$10.00 \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29 \$3.80	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20	NA NA \$2.77 \$0.40 \$2.77	\$37.68 NA \$1.59 \$0.40 \$1.59	\$26.31 NA \$1.03 \$0.2886 \$1.03
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)  RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	Note 8 Note 8 USAC2 USAC2 USACC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20	NA NA \$2.77 \$0.40 \$2.77	\$37.68 NA \$1.59 \$0.40 \$1.59	\$26.31 NA \$1.03 \$0.2886 \$1.03
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)  RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic,	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC USACC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80 \$10.00	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00	NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)  RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	Note 8 Note 8 USAC2 USAC2 USACC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20	NA NA \$2.77 \$0.40 \$2.77	\$37.68 NA \$1.59 \$0.40 \$1.59	\$26.31 NA \$1.03 \$0.2886 \$1.03
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6) Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7) NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC USACC SOMEC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80 \$10.00 \$2.75	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00 \$3.50	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$3.50	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00 \$3.50	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00 \$3.50	NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00 \$3.50	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$1.00 \$3.50
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)  RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC USACC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80 \$10.00	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00	NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)  RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC SOMEC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00 \$3.50	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80 \$10.00 \$2.75	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00 \$3.50	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.90 \$10.90	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00 \$3.50	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00 \$3.50	NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00 \$3.50	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00 \$3.50	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00 \$3.50
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)  RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC USACC SOMEC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80 \$10.00 \$2.75	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00 \$3.50	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$3.50	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00 \$3.50	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00 \$3.50	NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00 \$3.50	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$1.00 \$3.50
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6) Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7) NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - Add'l	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC SOMEC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00 \$3.50	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80 \$10.00 \$2.75	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00 \$3.50	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.90 \$10.90	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00 \$3.50	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00 \$3.50	NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00 \$3.50	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00 \$3.50	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00 \$3.50
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Norrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7) NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - Add'l  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC SOMEC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00 \$3.50 \$40.71 \$9.58	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80 \$10.00 \$2.75 \$21.56	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00 \$3.50 \$33.67	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.90 \$10.90 \$10.90	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00 \$3.50 \$31.92	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00 \$3.50 \$43.52	NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00 \$3.50 \$40.18	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00 \$3.50 \$43.19	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00 \$3.50 \$3.50
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)  RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - Add'l  NRC - 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Electronic	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC SOMEC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00 \$3.50	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80 \$10.00 \$2.75	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00 \$3.50	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.90 \$10.90	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00 \$3.50	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00 \$3.50	NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00 \$3.50	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00 \$3.50	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00 \$3.50
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)  RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - Add'l  NRC - 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update Electronic  NRC - 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update Electronic	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC SOMEC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00 \$3.50 \$40.71 \$9.58	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80 \$10.00 \$2.75 \$21.56 \$TBD	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00 \$3.50 \$33.67 \$7.88	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00  \$10.00	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00 \$3.50 \$31.92 \$7.32	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00 \$3.50 \$43.52 \$0.99	NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00 \$3.50 \$40.18 \$9.45	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00 \$3.50 \$43.19 \$9.91	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00 \$3.50 \$30.89 \$7.03
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6) Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7) NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - Add'l NRC - 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Electronic NRC - 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Manual Service Order	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC SOMEC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00 \$3.50 \$40.71 \$9.58	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80 \$10.00 \$2.75 \$21.56 \$21.56 TBD TBD	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00 \$3.50 \$33.67 \$7.88	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$19.90 \$19.99  TBD  TBD	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00 \$3.50 \$31.92 \$7.32 \$2.11 \$5.12	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00 \$3.50 \$43.52 \$0.99	NA NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00 \$3.50 \$40.18 \$9.45	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00 \$3.50 \$43.19 \$9.91 \$0.71 \$8.91	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00 \$3.50 \$3.50 \$30.89 \$7.03
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6) Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7) NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - Add'l  NRC - 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update Electronic NRC - 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update Manual Service Order NRC - Electronic Service Order Disconnect	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC SOMEC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00 \$3.50 \$40.71 \$9.58 \$1.44 \$8.25 TBD	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80 \$10.00 \$2.75 \$21.56  TBD TBD \$0.42	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00 \$3.50 \$33.67 \$7.88 TBD	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$19.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00 \$3.50 \$31.92 \$7.32 \$2.11 \$5.12 TBD	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00 \$3.50 \$43.52 \$0.99	NA NA NA NA NA NA NA NA NA NA NA NA NA N	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00 \$3.50 \$43.19 \$9.91 \$0.71 \$8.91 TBD	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00 \$3.50 \$3.50 \$30.89 \$7.03
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)  RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent  NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - Add'l  NRC - 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Electronic  NRC - 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Manual Service Order  NRC - Electronic Service Order Disconnect  NRC - Incremental Manual Service Order Disconnect	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC SOMEC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00 \$3.50 \$40.71 \$9.58	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80 \$10.00 \$2.75 \$21.56 \$21.56 TBD TBD	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00 \$3.50 \$33.67 \$7.88	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$19.90 \$19.99  TBD  TBD	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00 \$3.50 \$31.92 \$7.32 \$2.11 \$5.12	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00 \$3.50 \$43.52 \$0.99	NA NA NA \$2.77 \$0.40 \$2.77 \$0.40 \$10.00 \$3.50 \$40.18 \$9.45	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00 \$3.50 \$43.19 \$9.91 \$0.71 \$8.91	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00 \$3.50 \$3.50 \$30.89 \$7.03
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6) Nonrecurring Charges NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7) NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - Add'l  NRC - 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update Electronic NRC - 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update Manual Service Order NRC - Electronic Service Order Disconnect	Note 8 Note 8 USAC2 USAC2 USACC USACC USACC SOMEC	\$44.44 NA \$2.80 \$0.41 \$2.80 \$0.41 \$10.00 \$3.50 \$40.71 \$9.58 \$1.44 \$8.25 TBD	\$25.60 NA \$15.82 \$3.80 \$15.82 \$3.80 \$10.00 \$2.75 \$21.56  TBD TBD \$0.42	\$21.62 NA \$2.01 \$0.3108 \$2.01 \$0.3108 \$10.00 \$3.50 \$33.67 \$7.88 TBD	#VALUE! NA \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$19.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00 \$10.00	\$51.85 NA \$3.80 \$0.29 \$3.80 \$0.29 \$10.00 \$3.50 \$31.92 \$7.32 \$2.11 \$5.12 TBD	\$29.75 \$38.59 \$5.20 \$0.41 \$5.20 \$0.41 \$10.00 \$3.50 \$43.52 \$0.99	NA NA NA NA NA NA NA NA NA NA NA NA NA N	\$37.68 NA \$1.59 \$0.40 \$1.59 \$0.40 \$10.00 \$3.50 \$43.19 \$9.91 \$0.71 \$8.91 TBD	\$26.31 NA \$1.03 \$0.2886 \$1.03 \$0.2886 \$10.00 \$3.50 \$30.89 \$7.03

LIDEO	ODIDTION	11000				101					
DES	CRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
$\vdash$	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPRD	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
$\perp \perp \perp$	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPPC	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
$\perp$	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPPC	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPPO	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPPO	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPP1	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPP1	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPLD	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPLD	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPXA	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPXA	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPXB	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
	NIDO O Miles Visios Conde Languista C Miles Line Book Name Addill	LIEDVO			045.05						
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPXB	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPXC	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
H	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPXC	NA	NA NA	\$15.25	NA	NA NA	NA	NA NA	NA	NA NA
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPXD	NA.	NA NA	\$22.14	NA.	NA NA	NA.	NA NA	NA.	NA NA
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPXD	NA.	NA NA	\$15.25	NA.	NA NA	NA.	NA NA	NA.	NA NA
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPXE	NA NA	NA NA	\$22.14	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPXE	NA NA	NA NA	\$15.25	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPXL	NA NA	NA NA	\$22.14	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPXL	NA NA	NA NA	\$15.25	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add 1	UEPXM	NA NA	NA NA	\$22.14	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPXM	NA NA	NA NA	\$15.25	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NICO - 2-VVIII VOICE GIAGE LOOP WITH 2-VVIIIE LINE FOR - NEW - Add I	UEFAIVI	INA	INA	\$10.20	INA	INA	INA	INA	INA	INA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPXO	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPXO	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPXS	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPXS	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
	'										
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	USAS2	NA	NA	\$10.00	NA	NA	NA	NA	NA	NA
+	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Disconnect - 1st	00/102	NA.	NA NA	\$8.45	NA	NA NA	NA	NA NA	NA	NA NA
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Disconnect - Add'l		NA	NA	\$3.91	NA	NA NA	NA	NA	NA	NA NA
+++	THE E THE TOISE CHARLE CONTROL AND THE BROCK HOW BROCK HOLD		147.	100	ψ0.01	1471	1471	1471	14/1	1471	1471
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic,										
	per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
H +	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - Incremental Cost Manual vs.	OOMEO	IVA	IVA	ψ5.50	IVA	IVA	19/5	14/4	INA	IVA
	Electronic - New - 1st		NA	NA	\$37.06	NA	NA	NA	NA	NA	NA
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - Incremental Cost Manual vs.		INA	INA	ψ57.00	INA	INA	INA	11/7	INA	14/2
	Electronic - New - Add'l		NA	NA	\$8.19	NA	NA	NA	NA	NA	NA
++	NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update		INA	INA	ψ0.13	INA	INA	INA	11/7	INA	14/2
	Electronic		NA	NA	TBD	NA	NA	NA	NA	NA	NA
++	NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update		INA	INA	100	INA	INA	INA	11/7	INA	ING
	Manual Service Order		NA	NA	TBD	NA	NA	NA	NA	NA	NA
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - Incremental Cost Manual vs.		147	14/7	יטטי	11/	11/1	147	14/7	19/7	INC
	Electronic - New - Disconnect		NA	NA	\$11.17	NA	NA	NA	NA	NA	NA
++	Elocatorilo 146W - Discottificot		INA	INA	ψ11.11	INA	INA	INM	INM	INA	INA
++	All Other Loop/Port Combinations		TBD	TBD	Note 2	TBD	TBD	TBD	TBD	TBD	TBD
++	All Other Ecopifort Combinations		עמו	עמו	Note 2	עמו	עמו	עמו	עפו	IDD	עמו
++	LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	LNPCX	-		-	-			-	<b> </b>	
++	LOCAL NUMBER FOR I ADILIT I (REQUIRES ONE PER PORT)	LINCUX	1		<del>                                     </del>	<b>-</b>			-	1	
NOT	FC.		1		1	<del>                                     </del>			<del>                                     </del>	<del>                                     </del>	
NOT			1		1	<del>                                     </del>			<del>                                     </del>	ļ	
H+.	Interim rates subject to true-up.		1		1	<b>!</b>		1	<b>!</b>	1	
1	Market Rates will apply in those areas where BellSouth is not required to provide circuit					1					
++	switching pursuant to FCC rules.		ļ		<b>_</b>	-		ļ	ļ	-	
111			1			1		l	1	1	

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
In Georgia, rates will apply for Currently Combined as well as not Currently Combined loop/port combinations unless otherwise identified.										
In the absence of ordered rates by a State Commission, the recurring rates for Currently Combined combinations of loop/port network elements will be the sum of the recurring rates for the UNEs which make up the combinations, and the nonrecurring rates shall be as set forth in this section.										
4 End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section of this rate exhibit shall apply to all combinations of loop/port network elements.										
5 Deleted										
6 Effective May 1, 2000 statewide rates will be replaced by Deaveraged Loop Rates by Zone where available. Until approximately December 31, 2000 or until such time that BellSouth billing systems have been developed to handle the new zone rate structure, BellSouth will bill at the Zone 1 Deaveraged Loop rate level only. After December 31, 2000 or such time that the billing systems have been developed to handle the new zone rate structure, BellSouth will begin billing pursuant to CLEC-1's interconnection agreement.										
7 In the absence of ordered OSS rates by a state commission, BellSouth will offer regionwic	de rates									
8 There is not a unique combination USOC. CLEC should submit the loop and port USOCs	3.									
9 Rates in TN and FL are interim and shall be trued-up when final rates are ordered.	·									

E	NHANCED EXTENDED LINKS (EELs)		L			<u> </u>					
	New EEL rates are the sum of the individual UNE network elements (interoffice		I			i '		l			
	transport and loop [channelization if applicable].					<u> </u>		<u> </u>			
			I			i '		ł			
	DS1 Interoffice Channel and 2-wire VG Local Loop EEL:	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	Recurring Charges					i					
	2-wire VG Loop per month, statewide	UEAL2	NA	\$20.52	NA	NA	NA	NA	\$19.50	NA	NA
	2-wire VG Loop per month, Zone 1 (Note 1)	UEAL2	NA	TBD	\$15.40	NA	\$17.65	NA	TBD	NA	\$15.54
	2-wire VG Loop per month, Zone 2 (Note 1)	UEAL2	NA	TBD	\$17.78	NA	\$30.32	NA	TBD	NA	\$19.55
	2-wire VG Loop per month, Zone 3 (Note 1)	UEAL2	NA	TBD	\$28.26	NA	\$61.93	NA	TBD	NA	\$28.02
	2-wire VG Loop per month, Zone 4 (Note 1)	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	Interoffice Channel - Dedicated - DS1 - per mile per month	1L5XX	NA	\$0.6013	\$0.4523	NA	\$0.7831	NA	\$0.5753	NA	\$0.3525
	Interoffice Channel - Dedicated - DS1 - Facility Termination per month	U1TF1	NA	\$99.79	\$78.47	NA	\$93.40	NA	\$71.29	NA	\$75.83
	DS1 Channelized System per month	MQ1	NA	\$153.60	\$18.23	NA	\$209.87	NA	\$177.72	NA	\$165.21
	VG (COCI) interface card per month	1D1VG	NA	\$1.45	\$2.20	NA	\$1.62	NA	\$1.64	NA	\$1.25
	Non-Recurring Charges - New EEL (Note 2)(Note 3)										
	NRC- DS1 interoffice Facility Termination - 1st	U1TF1	NA	\$171.01	\$142.64	NA	\$160.49	NA	\$217.17	NA	\$165.53
Ħ	NRC-DS1 interoffice Facility Termination - Add'I	U1TF1	NA	\$101.84	\$94.87	NA	\$123.03	NA	\$163.75	NA	\$124.84
ΙŢ	NRC-2-wire VG Local Loop - 1st	UEAL2	NA	\$38.02	\$77.54	NA	\$128.42	NA	\$142.97	NA	\$103.76
TT	NRC-2-wire VG Local Loop - Add'l	UEAL2	NA	\$35.15	\$51.57	NA	\$93.60	NA	\$106.56	NA	\$65.84
Ħ	NRC-DS1 Channelization System -1st	MQ1	NA	\$275.18	\$206.09	NA	\$220.07	NA	\$193.63	NA	\$222.87
Ħ	NRC-DS1 Channelization System - Add'l	MQ1	NA	\$137.77	\$137.06	NA	\$135.20	NA	\$118.37	NA	\$135.80
Ħ	NRC-VG(COCI)interface card -1st	1D1VG	NA	\$12.16	\$12.02	NA	\$12.29	NA	\$15.76	NA	\$12.61
Ħ	NRC-VG(COCI)interface card - Add'	1D1VG	NA	\$8.77	\$8.66	NA	\$8.80	NA	\$11.28	NA	\$9.03
Ħ	NRC- 2-wire VG Local Loop and Channelized DS1 Interoffice Combination - Electronic	SOMEC	NA	\$3.17	\$3.50	NA	\$3.50	NA	\$3.50	NA	\$3.50
tt	NRC- 2-wire VG Local Loop and Channelized DS1 Interoffice Combination - Manual S	SOMAN	NA	\$25.40	NA	NA NA	NA NA	NA	NA NA	NA	\$19.99
Ħ	NRC- 2-wire VG Local Loop and Channelized DS1 Interoffice Combination - Manual S	SOMAN	NA	NA	\$33.63	NA	\$242.20	NA	\$66.20	NA	NA
tt	NRC- 2-wire VG Local Loop and Channelized DS1 Interoffice Combination - Manual S	SOMAN	NA	NA	\$27.49	NA	\$153.37	NA	\$51.40	NA	NA
tt	NRC- 2-wire VG Local Loop and Channelized DS1 Interoffice Combination - Manual S	SOMAN	NA	NA	\$19.88	NA	\$45.91	NA NA	NA	NA	NA
tt	NRC- 2-wire VG Local Loop and Channelized DS1 Interoffice Combination - Manual S	SOMAN	NA	NA	\$11.85	NA	\$8.06	NA	NA NA	NA	NA
tt					<b>V</b>		70.00		†		+
Ħ	DS1 Interoffice Channel and 4-wire VG Local Loop EEL:					i			1		
tt	Recurring Charges								1 1		+
tt	4-wire VG Loop per month	UEAL4	NA	NA	NA	NA	NA	NA	\$27.49	NA	NA
tt	4-wire VG Loop per month, Zone 1 (Note 1)	UEAL4	NA NA	\$24.26	\$22.88	NA NA	\$24.36	NA NA	NA NA	NA NA	\$15.92
tt	4-wire VG Loop per month, Zone 2 (Note 1)	UEAL4	NA NA	\$35.51	\$26.42	NA NA	\$41.85	NA	NA NA	NA NA	\$20.79
tt	4-wire VG Loop per month, Zone 3 (Note 1)	UEAL4	NA NA	\$78.35	\$41.99	NA NA	\$86.47	NA NA	NA NA	NA	\$27.18
tt	4-wire VG Loop per month, Zone 4 (Note 1)	NA	NA NA	NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA
tt	Interoffice Channel - Dedicated - DS1 - per mile per month	1L5XX	NA.	\$0.6013	\$0.4523	NA NA	\$0.7831	NA NA	\$0.0783	NA NA	\$0.3525
tt	Interoffice Channel - Dedicated - DS1 - Facility Termination per month	U1TF1	NA NA	\$99.79	\$78.47	NA NA	\$93.40	NA	\$71.29	NA NA	\$75.83
tt	DS1 Channelized System per month	MQ1	NA NA	\$153.60	\$18.23	NA NA	\$209.87	NA NA	\$177.72	NA NA	\$165.21
tt	VG (COCI) interface card per month	1D1VG	NA	\$1.45	\$2.67	NA	\$1.62	NA NA	\$1.64	NA	\$1.25
tt	Non-Recurring Charges - New EEL (Note 2) (Note 3)	.5		<b>\$11.10</b>	Ψ2.01		ψo2		<b>\$1.0</b> 1		<b>V.1.20</b>
tt	NRC- DS1 interoffice Facility Termination - 1st	U1TF1	NA	\$237.34	\$142.64	NA	\$160.49	NA	\$534.48	NA	\$165.53
tt	NRC-DS1 interoffice Facility Termination - Add'l	U1TF1	NA NA	\$145.26	\$94.87	NA NA	\$123.03	NA NA	\$462.69	NA	\$124.84
tt	NRC-4-wire VG Local Loop - 1st	UEAL4	NA NA	\$329.76	\$228.99	NA NA	\$128.42	NA NA	\$288.47	NA	\$103.76
++	NRC-4-wire VG Local Loop - Add'l	UEAL4	NA NA	\$178.91	\$203.26	NA NA	\$93.60	NA NA	\$237.45	NA NA	\$65.84
tt	NRC-DS1 Channelization System -1st	MQ1	NA NA	\$208.85	\$206.09	NA NA	\$220.07	NA NA	\$301.74	NA NA	\$222.87
++	NRC-DS1 Channelization System - 13t	MQ1	NA	\$94.35	\$137.06	NA NA	\$135.20	NA NA	\$182.57	NA NA	\$135.80
++	NRC-VG(COCI)interface card -1st	1D1VG	NA	\$12.16	\$12.02	NA NA	\$12.29	NA NA	\$15.76	NA NA	\$12.61
++	NRC-VG(COCI)interface card - 1st  NRC-VG(COCI)interface card - Add'	1D1VG	NA NA	\$8.77	\$8.66	NA NA	\$8.80	NA NA	\$11.28	NA NA	\$9.03
++	NRC-DS1 interoffice channel and 4-wire VG Local Loop Combination - Electronic Svc	SOMEC	NA NA	\$3.17	\$3.50	NA NA	\$3.50	NA NA	\$3.50	NA NA	\$3.50
++	NRC-DS1 interoffice channel and 4-wire VG Local Loop Combination - Electronic SVC	SOMAN	NA NA	\$25.40	NA	NA NA	Ψ3.50 NA	NA NA	93.50 NA	NA NA	\$19.99
++	NRC-DS1 interoffice channel and 4-wire VG Local Loop Combination - Manual Svc O	SOMAN	NA NA	\$25.40 NA	\$33.63	NA NA	\$242.20	NA NA	\$66.20	NA NA	\$19.99 NA
	NRC-DS1 interoffice channel and 4-wire VG Local Loop Combination - Manual Svc Of NRC-DS1 interoffice channel and 4-wire VG Local Loop Combination - Manual Svc Of	SOMAN	NA NA	NA NA	\$27.49	NA NA	\$153.37	NA NA	\$51.40	NA NA	NA NA
					UL1.40	IN/A	w100.01	11//	U-01.4U	INA	INA
H	NRC-DS1 interoffice channel and 4-wire VG Local Loop Combination - Manual Svc Or	SOMAN	NA	NA	\$19.88	NA	\$45.91	NA	NA	NA	NA

	New EEL rates are the sum of the individual UNE network elements (interoffice				<del></del>						T
	transport and loop [channelization if applicable].		I		, ,		!			ł	
		-									†
	DS1 Interoffice Channel and 2-wire ISDN Local Loop:	-								i	
	Recurring Charges	-								i	
	2-wire ISDN Loop per month	U1L2X	NA	NA	NA	NA	NA	NA	\$24.98	NA	NA
	2-wire ISDN Loop per month, Zone 1 (Note 1)	U1L2X	NA	\$32.34	\$21.89	NA	\$21.15	NA	TBD	NA	\$15.54
	2-wire ISDN Loop per month, Zone 2 (Note 1)	U1L2X	NA	\$47.35	\$25.27	NA	\$36.22	NA	TBD	NA	\$19.55
	2-wire ISDN Loop per month, Zone 3 (Note 1)	U1L2X	NA	\$104.47	\$40.17	NA	\$74.19	NA	TBD	NA	\$28.02
	2-wire ISDN Loop per month, Zone 4 (Note 1)	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	Interoffice Channel - Dedicated - DS1 - per mile per month	1L5XX	NA	\$0.6013	\$0.4523	NA	\$0.7831	NA	\$0.5753	NA	\$0.3525
	Interoffice Channel - Dedicated - DS1 - Facility Termination per month	U1TF1	NA	\$99.79	\$78.47	NA	\$93.40	NA	\$71.29	NA	\$75.83
	DS1 Channelized System per month	MQ1	NA	\$153.60	\$18.23	NA	\$209.87	NA	\$177.72	NA	\$165.21
	2-wire ISDN(BRITE COCI) per month	UC1CA	NA	\$3.83	\$3.71	NA	\$4.18	NA	\$3.76	NA	\$3.33
	Non-Recurring Charges - New EEL (Note 2)(Note 3)									<u> </u>	
	NRC- DS1 interoffice Facility Termination - 1st	U1TF1	NA	\$237.34	\$142.64	NA	\$160.49	NA	\$217.17	NA	\$165.53
	NRC-DS1 interoffice Facility Termination - Add'l	U1TF1	NA	\$145.26	\$94.87	NA	\$123.03	NA	\$163.75	NA	\$124.84
	NRC- 2-wire ISDN Local Loop - 1st	U1L2X	NA	\$329.76	\$77.54	NA	\$223.27	NA	\$325.91	NA	\$58.50
$\coprod$	NRC- 2-wire ISDN Local Loop - Add'l	U1L2X	NA	\$148.55	\$51.57	NA	\$172.63	NA	\$251.31	NA	\$31.00
	NRC-DS1 Channelization System -1st	MQ1	NA	\$208.85	\$206.09	NA	\$220.07	NA	\$301.74	NA	\$222.87
$\Box$	NRC-DS1 Channelization System - Add'l	MQ1	NA	\$94.35	\$137.06	NA	\$135.20	NA	\$182.57	NA	\$135.80
	NRC-2-wire BRITE(COCI)interface card -1st	UC1CA	NA	\$12.16	\$12.02	NA	\$12.29	NA	\$15.76	NA	\$12.61
	NRC-2-wire BRITE(COCI)interface card -Add'l	UC1CA	NA	\$8.77	\$8.66	NA	\$8.80	NA	\$11.28	NA	\$9.03
$\perp \perp \perp$	NRC-DS1 interoffice channel and 2-wire ISDN Local Loop Combination - Electronic S	/ SOMEC	NA	\$3.17	\$3.50	NA	\$3.50	NA	\$3.50	NA	\$3.50
$\vdash$	NRC-DS1 interoffice channel and 2-wire ISDN Local Loop Combination - Manual Svc	SOMAN	NA	\$25.40	NA	NA	NA	NA	NA	NA	\$19.99
$\sqcup \sqcup \bot$	NRC-DS1 interoffice channel and 2-wire ISDN Local Loop Combination - Manual Svc	SOMAN	NA	NA	\$33.63	NA	\$57.58	NA	\$38.07	NA	NA
++	NRC-DS1 interoffice channel and 2-wire ISDN Local Loop Combination - Manual Svc	SOMAN	NA	NA	\$27.49	NA	\$36.31	NA	\$38.07	NA	NA
++	NRC-DS1 interoffice channel and 2-wire ISDN Local Loop Combination - Manual Svc	SOMAN	NA	NA	\$19.88	NA	\$16.12	NA	NA	NA	NA
+++	NRC-DS1 interoffice channel and 2-wire ISDN Local Loop Combination - Manual Svc	SOMAN	NA	NA	\$11.85	NA	\$8.06	NA	NA	NA	NA
++	DS1 Interoffice Channel and 4-wire 56 kbps Local Loop:			-	<b></b>	<b></b>	-	<u> </u>	<b></b>	<del></del>	+
++	Recurring Charges			-		<b> </b>		<b> </b>	<b></b>		+
++	4-wire 56kbps Loop per month	UDL56	NA	NA	NA	NA	NA	NA	\$32.67	NA	NA
+++	4-wire 56kbps Loop per month  4-wire 56kbps Loop per month, Zone 1 (Note 1)	UDL56	NA NA	\$39.08	\$26.44	NA NA	\$27.50	NA NA	\$32.67 TBD	NA NA	\$36.45
+++	4-wire 56kbps Loop per month, Zone 2 (Note 1)	UDL56	NA NA	\$57.21	\$26.44	NA NA	\$47.24	NA NA	TBD	NA NA	\$45.87
++	4-wire 56kbps Loop per month, Zone 3 (Note 1)	UDL56	NA NA	\$126.22	\$46.53	NA NA	\$96.48	NA NA	TBD	NA NA	\$65.75
++	4-wire 56kbps Loop per month, Zone 4 (Note 1)	NA	NA NA	NA	NA	NA NA	NA	NA NA	NA NA	NA NA	NA
++	Interoffice Channel - Dedicated - DS1 - per mile per month	1L5XX	NA NA	\$0.6013	\$0.4523	NA NA	\$0.7831	NA NA	\$0.5753	NA NA	\$0.3525
+++	Interoffice Channel - Dedicated - DS1 - Facility Termination per month	U1TF1	NA NA	\$99.79	\$78.47	NA NA	\$93.40	NA NA	\$71.29	NA NA	\$75.83
+++	DS1 Channelized System per month	MQ1	NA	\$153.60	\$18.23	NA NA	\$209.87	NA NA	\$177.72	NA	\$165.21
HH	4-wire 56kbps card COCI per month	1D1DD	NA NA	\$2.20	\$1.06	NA NA	\$3.12	NA NA	\$2.88	NA NA	\$2.46
	Non-Recurring Charges - New EEL (Note 2) (Note 3)			<b>\$2.20</b>	ψσσ		ψ02		<b>\$2.00</b>		<u> </u>
++	NRC- DS1 interoffice Facility Termination - 1st	U1TF1	NA	\$237.34	\$142.64	NA	\$160.49	NA	\$217.17	NA	\$165.53
++	NRC-DS1 interoffice Facility Termination - Add'l	U1TF1	NA	\$145.26	\$94.87	NA NA	\$123.03	NA	\$163.75	NA NA	\$124.84
	NRC-4-wire 56kbps Local Loop - 1st	UDL56	NA NA	\$329.76	\$395.14	NA NA	\$333.28	NA NA	\$489.04	NA NA	\$643.00
$\Box$	NRC-4-wire 56kbps Local Loop - Add'l	UDL56	NA	\$148.55	\$206.98	NA	\$230.50	NA	\$337.51	NA	\$421.26
	NRC-DS1 Channelization System -1st	MQ1	NA	\$208.85	\$206.09	NA	\$220.07	NA	\$301.74	NA	\$222.87
	NRC-DS1 Channelization System - Add'l	MQ1	NA	\$94.35	\$137.06	NA	\$135.20	NA	\$182.57	NA	\$135.80
	NRC-4-wire 56kbps(COCI)interface card -1st	1D1DD	NA	\$12.16	\$12.02	NA	\$12.29	NA	\$15.76	NA	\$12.61
	NRC-4-wire 56kbps(COCI)interface card -Add'I	1D1DD	NA	\$8.77	\$8.66	NA	\$8.80	NA	\$11.28	NA	\$9.03
	NRC-DS1 interoffice channel and 4-wire 56kbps Local Loop Combination - Electronic	SOMEC	NA	\$3.17	\$3.50	NA	\$3.50	NA	\$3.50	NA	\$3.50
	NRC-DS1 interoffice channel and 4-wire 56kbps Local Loop Combination - Manual Sv	SOMAN	NA	\$25.40	NA	NA	NA	NA	NA	NA	\$19.99
	NRC-DS1 interoffice channel and 4-wire 56kbps Local Loop Combination - Manual Sv	SOMAN	NA	NA	\$33.63	NA	\$242.20	NA	\$65.01	NA	NA
	NRC-DS1 interoffice channel and 4-wire 56kbps Local Loop Combination - Manual Sv	SOMAN	NA	NA	\$27.49	NA	\$153.37	NA	\$50.83	NA	NA
	NRC-DS1 interoffice channel and 4-wire 56kbps Local Loop Combination - Manual SV	SOMAN	NA	NA	\$19.88	NA	\$45.91	NA	NA	NA	NA
$\perp$											
Ш	NRC-DS1 interoffice channel and 4-wire 56kbps Local Loop Combination - Manual Sv	SOMAN	NA	NA	\$11.85	NA	\$8.06	NA	NA	NA	NA

New EEL rates are the sum of the individual UNE network elements (interoffice transport and loop [channelization if applicable].	•									
DS1 Interoffice Channel and 4-wire 64 kbps Local Loop:										
Recurring Charges										
4-wire 64kbps Loop per month	UDL64	NA	NA	NA	NA	NA	NA	\$32.67	NA	NA
4-wire 64kbps Loop per month, Zone 1 (Note 1)	UDL64	NA	\$39.08	\$26.44	NA	\$27.50	NA	TBD	NA	\$36.45
4-wire 64kbps Loop per month, Zone 2 (Note 1)	UDL64	NA	\$57.21	\$30.53	NA	\$47.24	NA	TBD	NA	\$45.87
4-wire 64kbps Loop per month, Zone 3 (Note 1)	UDL64	NA	\$126.22	\$46.53	NA	\$96.48	NA	TBD	NA	\$65.7
4-wire 64kbps Loop per month, Zone 4 (Note 1)	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Interoffice Channel - Dedicated - DS1 - per mile per month	1L5XX	NA	\$0.6013	\$0.4523	NA	\$0.7831	NA	\$0.5753	NA	\$0.352
Interoffice Channel - Dedicated - DS1 - Facility Termination per month	U1TF1	NA	\$99.79	\$78.47	NA	\$93.40	NA	\$71.29	NA	\$75.8
DS1 Channelized System per month	MQ1	NA	\$153.60	\$18.23	NA	\$209.87	NA	\$177.72	NA	\$165.
4-wire 64kbps card COCI per month	1D1DD	NA	\$1.06	\$1.06	NA	\$3.12	NA	\$2.88	NA	\$2.40
Non-Recurring Charges - New EEL (Note 2) (Note 3)										
NRC- DS1 interoffice - 1st	U1TF1	NA	\$237.34	\$142.64	NA	\$160.49	NA	\$217.17	NA	\$165.
NRC- DS1 interoffice - Add'l	U1TF1	NA	\$145.26	\$94.87	NA	\$123.03	NA	\$163.75	NA	\$124.
NRC-4-wire 64kbps Local Loop - 1st	UDL64	NA	\$329.76	\$395.14	NA	\$333.28	NA	\$489.04	NA	\$103.7
NRC-4-wire 64kbps Local Loop - Add'l	UDL64	NA	\$148.55	\$206.98	NA	\$230.50	NA	\$337.51	NA	\$65.8
NRC-DS1 Channelization System -1st	MQ1	NA	\$208.85	\$206.09	NA	\$220.07	NA	\$301.74	NA	\$222.
NRC-DS1 Channelization System - Add'l	MQ1	NA	\$94.35	\$137.06	NA	\$135.20	NA	\$288.33	NA	\$135.
NRC-4-wire 64kbps(COCI)interface card -1st	1D1DD	NA	\$12.16	\$12.02	NA	\$12.29	NA	\$15.76	NA	\$12.6
NRC-4-wire 64kbps(COCI)interface card -Add'I	1D1DD	NA	\$8.77	\$8.66	NA	\$8.80	NA	\$11.28	NA	\$9.0
NRC-DS1 interoffice channel and 4-wire 64kbps Local Loop Combination - Electroni	c SOMEC	NA	\$3.17	\$3.50	NA	\$3.50	NA	\$3.50	NA	\$3.5
NRC-DS1 interoffice channel and 4-wire 64kbps Local Loop Combination - Manual S		NA	\$25.40	NA	NA	NA	NA	NA	NA	\$19.9
NRC-DS1 interoffice channel and 4-wire 64kbps Local Loop Combination - Manual S		NA	NA	\$33.63	NA	\$242.20	NA	\$65.01	NA	NA.
NRC-DS1 interoffice channel and 4-wire 64kbps Local Loop Combination - Manual S		NA	NA	\$27.49	NA	\$153.37	NA	\$50.83	NA	NA
NRC-DS1 interoffice channel and 4-wire 64kbps Local Loop Combination - Manual S		NA	NA	\$19.88	NA	\$45.91	NA	NA	NA	N/
NRC-DS1 interoffice channel and 4-wire 64kbps Local Loop Combination - Manual S		NA	NA	\$11.85	NA	\$8.06	NA	NA	NA	NA
DC4 Interesting Channel and DC4 Interesting Local Local								1		
DS1 Interoffice Channel and DS1 Interoffice Local Loop:	+							+		
Recurring Charges	1101.777	NI A	NIA.	NIA	NIA.	NIA	NIA	<b>#00.70</b>	NIA	NI A
DS1 Loop per month	USLXX	NA	NA Control	NA OFFO. 40	NA NA	NA ©50.00	NA	\$62.78	NA	NA
DS1 Loop per month, Zone 1 (Note 1)	USLXX	NA NA	\$64.69	\$52.40	NA NA	\$56.32	NA	TBD	NA NA	NA NA
DS1 Loop per month, Zone 2 (Note 1)	USLXX	NA NA	\$94.71	\$60.51	NA NA	\$96.73	NA	TBD TBD		NA NA
DS1 Loop per month, Zone 3 (Note 1)	USLXX	NA	\$208.93	\$96.18	NA NA	\$197.57	NA		NA	
DS1 Loop per month, Zone 4 (Note 1)	NA 41.5VV	NA NA	NA Co codo	NA Co. 4500	NA NA	NA Co 7004	NA NA	NA O 5750	NA	NA NA
Interoffice Channel - Dedicated - DS1 - per mile per month	1L5XX U1TF1	NA NA	\$0.6013	\$0.4523	NA NA	\$0.7831 \$93.40	NA NA	\$0.5753	NA NA	NA NA
Interoffice Channel - Dedicated - DS1 - Facility Termination per month	UTIFT	NA	\$99.79	\$78.47	NA	\$93.40	INA	\$71.29	INA	IN/A
Non-Recurring Charges - New EEL (Note 2) (Note 3)	LIATEA	NIA	#007.04	<b>#</b> 440.04	NIA.	<b>#</b> 400.40	NIA	0047.47	NIA	NIA.
NRC- DS1 interoffice - 1st	U1TF1	NA	\$237.34	\$142.64	NA NA	\$160.49	NA	\$217.17	NA	NA NA
NRC- DS1 interoffice - Add'l	U1TF1	NA NA	\$145.26	\$94.87	NA NA	\$123.03	NA	\$163.75	NA	NA NA
NRC-DS1 Local Loop - 1st NRC-DS1 Local Loop - Add'l	USLXX	NA NA	\$561.45 \$334.01	\$627.44 \$231.49	NA NA	\$502.73 \$293.92	NA NA	\$714.84 \$421.47	NA NA	NA NA
NRC-DS1 Local Loop - Add1  NRC-DS1 interoffice channel and DS1 Local Loop Combination - Electronic Svc Ord		NA NA	\$334.01	\$3.50	NA NA	\$293.92	NA NA	\$3.50	NA NA	\$3.5
		NA NA	\$3.17	\$3.50 NA	NA NA	\$3.50 NA	NA NA	\$3.50 NA	NA NA	\$3.5 \$19.9
NRC-DS1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order NRC-DS1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order		NA NA	\$25.40 NA	\$33.63	NA NA	\$242.20	NA NA	\$65.01	NA NA	\$19.5 NA
NRC-DS1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order  NRC-DS1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order		NA NA	NA NA	\$27.49	NA NA	\$242.20 \$153.37	NA NA	\$50.83	NA NA	NA NA
NRC-DS1 Interoffice channel and DS1 Local Loop Combination - Manual Svc Order  NRC-DS1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order		NA NA	NA NA	\$27.49 \$19.88	NA NA	\$153.37 \$45.91	NA NA	\$50.83 NA	NA NA	NA NA
		NA NA	NA NA	\$19.88 \$11.85	NA NA	\$45.91	NA NA	NA NA	NA NA	NA NA
NRC-DS1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order	- SUIVIAIN	INA	INA	\$11.65	INA	<b>⊅</b> δ.∪δ	INA	INA	NA	I NA
DS3 Interoffice Channel and DS3 Local Loop:	1							1		
Recurring Charges										
DS3 Loop per Facility Termination per month	UE3PX	NA	\$404.58	\$390.34	NA	\$669.01	NA	\$387.01	NA	\$607
		NA.	\$11.77	\$8.90	NA.	\$30.34	NA	\$32.53	NA	\$23.
DS3 Loop per mile	1L5ND									
DS3 Loop per mile	1L5ND 1L5XX	NA NA		\$717.60	NA		NA		NA	\$760
			\$1,121.93 \$4.17			\$1,101 \$14.04		\$720.38 \$12.98	NA NA	\$760. \$5.8

BELLSOUTH/BTI RATES NETWORK ELEMENTS AND OTHER SERVICES ENHANCED EXTENDED LINKS

Attachment 2 Exhibit C Rates - Page 64

	New EEL rates are the sum of the individual UNE network elements (interoffice										
	transport and loop [channelization if applicable].										
	NRC- DS3 interoffice - 1st	U1TF3	NA	\$154.30	\$633.41	NA	\$713.57	NA	\$794.94	NA	\$729.27
	NRC- DS3 interoffice - Add'l	U1TF3	NA	\$77.50	\$449.91	NA	\$404.36	NA	\$579.55	NA	\$411.98
	NRC-DS3 Local Loop - 1st	UE3PX	NA	\$1,020.45	\$761.81	NA	\$811.30	NA	\$964.04	NA	\$829.52

#### BELLSOUTH/BTI RATES NETWORK ELEMENTS AND OTHER SERVICES ENHANCED EXTENDED LINKS

		New EEL rates are the sum of the individual UNE network elements (interoffice										
		transport and loop [channelization if applicable].										
		NRC-DS3 Local Loop - Add'l	UE3PX	NA	\$513.74	\$545.54	NA	\$502.09	NA	\$542.73	NA	\$512.23
		NRC-DS3 interoffice channel and DS3 Local Loop Combination - Electronic Svc Orde	SOMEC	NA	\$3.17	\$3.50	NA	\$3.50	NA	\$3.50	NA	\$3.50
		NRC-DS3 interoffice channel and DS3 Local Loop Combination - Manual Svc Order,	SOMAN	NA	\$25.40	NA	NA	NA	NA	NA	NA	\$19.99
		NRC-DS3 interoffice channel and DS3 Local Loop Combination - Manual Svc Order -	SOMAN	NA	NA	\$37.55	NA	\$100.50	NA	\$83.19	NA	NA
		NRC-DS3 interoffice channel and DS3 Local Loop Combination - Manual Svc Order -	SOMAN	NA	NA	\$37.55	NA	\$100.50	NA	\$69.01	NA	NA
		NRC-DS3 interoffice channel and DS3 Local Loop Combination - Manual Svc Order -	SOMAN	NA	NA	\$18.03	NA	\$41.88	NA	NA	NA	NA
		NRC-DS3 interoffice channel and DS3 Local Loop Combination - Manual Svc Order -	SOMAN	NA	NA	\$18.03	NA	\$41.88	NA	NA	NA	NA
		STS-1 Interoffice Channel and STS-1 Local Loop:										
		Recurring Charges										
		STS-1 Loop per Facility Termination per month	UDLS1	NA	\$446.09	\$421.59	NA	\$497.08	NA	\$387.01	NA	\$400.21
		STS-1 Loop per mile	1L5ND	NA	\$11.77	\$8.90	NA	\$38.98	NA	\$32.53	NA	\$30.53
Ш		Interoffice Channel - Dedicated - STS-1 - FacilityTermination per month	U1TFS	NA	\$1,105.98	\$788.00	NA	\$1,101	NA	\$800.94	NA	\$838.65
Ш		Interoffice Channel - Dedicated - STS-1 - per mile per month	1L5XX	NA	\$4.17	\$2.72	NA	\$14.04	NA	\$6.29	NA	\$6.88
		Non-Recurring Charges - New EEL (Note 2)(Note 3)										
		NRC- STS-1 interoffice - 1st	U1TFS	NA	\$682.89	\$633.41	NA	\$713.57	NA	\$624.86	NA	\$961.62
		NRC- STS-1 interoffice - Add'l	U1TFS	NA	\$288.32	\$449.91	NA	\$404.36	NA	\$436.36	NA	\$625.84
		NRC-STS-1 Local Loop - 1st	UDLS1	NA	\$154.30	\$761.81	NA	\$811.30	NA	\$964.04	NA	\$829.52
		NRC-STS-1 Local Loop - Add'l	UDLS1	NA	\$77.50	\$545.54	NA	\$502.09	NA	\$542.73	NA	\$512.23
		NRC-STS-1 interoffice channel and STS-1 Local Loop Combination - Electronic Svc Q	SOMEC	NA	\$1,020.45	\$3.50	NA	\$3.50	NA	\$3.50	NA	\$3.50
Ш		NRC-STS-1 interoffice channel and STS-1 Local Loop Combination - Manual Svc Ord	SOMAN	NA	\$513.74	NA	NA	NA	NA	NA	NA	\$19.99
		NRC-STS-1 interoffice channel and STS-1 Local Loop Combination - Manual Svc Ord	SOMAN	NA	\$3.17	\$37.96	NA	\$100.50	NA	\$123.62	NA	NA
Ш		NRC-STS-1 interoffice channel and STS-1 Local Loop Combination - Manual Svc Ord	SOMAN	NA	\$25.40	\$37.96	NA	\$100.50	NA	\$84.76	NA	NA
Ш		NRC-STS-1 interoffice channel and STS-1 Local Loop Combination - Manual Svc Ord	SOMAN	NA	NA	\$18.23	NA	\$41.88	NA	NA	NA	NA
Ш	1	NRC-STS-1 interoffice channel and STS-1 Local Loop Combination - Manual Svc Ord	SOMAN	NA	NA	\$18.23	NA	\$41.88	NA	NA	NA	NA
Ш	1											
Ш		DS3 Interoffice Channel and DS1 Local Loop:										
Ш		Recurring Charges										
		DS1 Loop per month	USLXX	NA	NA	NA	NA	NA	NA	\$62.78	NA	NA

	New EEL rates are the sum of the individual UNE network elements (interoffice										
	transport and loop [channelization if applicable].										
	DS1 Loop per month, Zone 1 (Note 1)	USLXX	NA	\$64.69	\$52.40	NA	\$56.32	NA	TBD	NA	NA
	DS1 Loop per month, Zone 2 (Note 1)	USLXX	NA	\$94.71	\$60.51	NA	\$96.73	NA	TBD	NA	NA
	DS1 Loop per month, Zone 3 (Note 1)	USLXX	NA	\$208.93	\$96.18	NA	\$197.57	NA	TBD	NA	NA
11	DS1 Loop per month, Zone 4 (Note 1)	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	Interoffice Channel - Dedicated - DS3 - FacilityTermination per month	U1TF3	NA	\$1,121.93	\$717.60	NA	\$1,101	NA	\$720.38	NA	NA
	Interoffice Channel - Dedicated - DS3 - per mile per month	1L5XX	NA	\$4.17	\$6.46	NA	\$14.04	NA	\$12.98	NA	NA
	DS3 Channelized System per month	MQ3	NA	\$220.97	\$202.91	NA	\$245.84	NA	\$226.81	NA	NA
	DS3 Interface per month (DS1 COCI)	UC1D1	NA	\$14.40	\$0.67	NA	\$7.55	NA	\$4.61	NA	NA
	Non-Recurring Charges - New EEL (Note 2)(Note 3)				*		*		,		
	NRC- DS3 interoffice - 1st	U1TF3	NA	\$154.30	\$520.09	NA	\$713.57	NA	\$794.94	NA	NA
	NRC- DS3 interoffice - Add'l	U1TF3	NA	\$77.50	\$282.38	NA	\$404.36	NA	\$579.55	NA	NA
	NRC-DS1 Local Loop - 1st	USLXX	NA	\$561.45	\$142.64	NA	\$502.73	NA	\$714.84	NA	NA
	NRC-DS1 Local Loop - Add'l	USLXX	NA	\$334.01	\$94.87	NA	\$293.92	NA	\$421.47	NA	NA
-1-1-	NRC-DS3 Channelization System -1st	MQ3	NA	\$404.85	\$316.28	NA	\$320.72	NA	\$351.95	NA	NA
-1-1-	NRC-DS3 Channelization System - Add'l	MQ3	NA	\$168.26	\$171.72	NA	\$233.10	NA	\$243.76	NA	NA
+	NRC-DS1(COCI)interface card -1st	UC1D1	NA NA	\$12.16	\$12.02	NA NA	\$12.29	NA NA	\$15.76	NA NA	NA NA
+	NRC-DS1(COCI)interface card -Add'l	UC1D1	NA	\$8.77	\$8.66	NA	\$8.80	NA	\$11.28	NA NA	NA NA
+	NRC-DS3 interoffice channel and DS1 Local Loop Combination - Electronic Svc Orde	SOMEC	NA NA	\$3.17	\$3.50	NA NA	\$3.50	NA NA	\$3.50	NA NA	NA
+	NRC-DS3 interoffice channel and DS1 Local Loop Combination - Liectionic Svc Order, p	SOMAN	NA NA	\$25.40	\$33.63	NA NA	NA	NA NA	NA	NA NA	NA NA
+	NRC-DS3 interoffice channel and DS1 Local Loop Combination - Manual Svc Order -	SOMAN	NA NA	NA	\$27.49	NA NA	\$36.28	NA NA	\$118.20	NA NA	NA NA
+	NRC-DS3 interoffice channel and DS1 Local Loop Combination - Manual Svc Order -	SOMAN	NA	NA NA	\$19.88	NA	\$26.20	NA	\$104.02	NA.	NA
+	NRC-DS3 interoffice channel and DS1 Local Loop Combination - Manual Svc Order -	SOMAN	NA NA	NA NA	\$11.85	NA	\$19.47	NA NA	NA	NA NA	NA
+++	NRC-DS3 interoffice channel and DS1 Local Loop Combination - Manual Svc Order -	SOMAN	NA NA	NA NA	GA GA	NA	\$8.06	NA NA	NA NA	NA NA	NA NA
+	Wandar ove order	OOWAIN	14/3	19/3	OΛ	1973	ψ0.00	19/3	INA	14/3	14/3
++	STS-1 Interoffice Channel and DS1 Local Loop:										
++	Recurring Charges										
+	DS1 Loop per month, Zone 1 (Note 1)	USLXX	NA	\$92.48	\$52.40	NA	\$56.32	NA	TBD	NA	NA
	DS1 Loop per month, Zone 2 (Note 1)	USLXX	NA NA	\$119.68	\$60.51	NA NA	\$96.73	NA NA	TBD	NA	NA NA
++	DS1 Loop per month, Zone 3 (Note 1)	USLXX	NA NA	\$194.70	\$96.18	NA NA	\$197.57	NA NA	TBD	NA NA	NA NA
+	DS1 Loop per month, Zone 4 (Note 1)	NA	NA NA	NA	NA	NA NA	NA	NA NA	NA	NA NA	NA NA
+	Interoffice Channel - Dedicated - STS-1 - FacilityTermination per month	U1TFS	NA NA	\$1,105.98	\$717.60	NA NA	\$1.101	NA NA	\$387.01	NA NA	NA NA
++	Interoffice Channel - Dedicated - STS-1 - Pacinity reminiation per month	1L5XX	NA NA	\$4.17	\$6.46	NA NA	\$14.04	NA NA	\$32.53	NA NA	NA NA
++	DS3 Channelized System per month	MQ3	NA NA	\$220.97	\$202.91	NA NA	\$245.84	NA NA	\$226.81	NA	NA NA
+	DS3 Interface per month (DS1 COCI)	UC1D1	NA NA	\$14.40	\$0.67	NA NA	\$7.55	NA NA	\$4.61	NA NA	NA NA
+	Non-Recurring Charges - New EEL (Note 2)(Note 3)	OCIDI	INA	\$14.40	\$0.07	INA	\$1.55	INA	φ4.01	INA	INA
++	NRC-DS1 Local Loop - 1st	USLXX	NA	\$561.45	\$520.09	NA	\$502.73	NA	\$714.84	NA	NA
+	NRC-DS1 Local Loop - 1st	USLXX	NA NA	\$334.01	\$282.38	NA NA	\$293.92	NA NA	\$421.47	NA NA	NA NA
++	NRC- STS-1 interoffice - 1st	U1TFS	NA NA	\$154.30	\$202.30 \$142.64	NA NA	\$713.57	NA NA	\$624.86	NA NA	NA NA
++	NRC- STS-1 interoffice - 1St  NRC- STS-1 interoffice - Add'l	U1TFS	NA NA	\$77.50	\$94.87	NA NA	\$404.36	NA NA	\$436.36	NA NA	NA NA
++	NRC-DS3 Channelization System -1st	MQ3	NA NA	\$404.85	\$316.28	NA NA	\$320.72	NA NA	\$351.95	NA NA	NA NA
	NRC-DS3 Channelization System - Add'l	MQ3	NA NA	\$168.26	\$171.72	NA NA	\$233.10	NA NA	\$243.76	NA NA	NA NA
++	NRC-DS3 Charmelization System - Add I	UC1D1	NA NA	\$108.26	\$171.72	NA NA	\$233.10 \$12.29	NA NA	\$243.76 \$15.76	NA NA	NA NA
++	NRC-DS1(COCI)interface card -1st  NRC-DS1(COCI)interface card -Add'l	UC1D1	NA NA	\$8.77	\$8.66	NA NA	\$8.80	NA NA	\$15.76	NA NA	NA NA
++	NRC-STS-1 interoffice channel and DS1 Local Loop Combination - Electronic Svc Ord	SOMEC	NA NA	\$8.77	\$8.66	NA NA	\$8.80	NA NA	\$11.28 \$3.50	NA NA	NA NA
+	NRC-STS-1 interoffice channel and DS1 Local Loop Combination - Electronic Svc Order	SOMAN	NA NA	\$3.17	\$3.50 NA	NA NA	\$3.50 NA	NA NA	\$3.50 NA	NA NA	NA NA
++	NRC-STS-1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order	SOMAN	NA NA	\$25.40 NA	\$33.63	NA NA	\$68.39	NA NA	\$81.94	NA NA	NA NA
++		SOMAN	NA NA	NA NA	\$33.63	NA NA		NA NA	\$67.76	NA NA	NA NA
++	NRC-STS-1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order	SOMAN		NA NA			\$58.31	NA NA		NA NA	NA NA
++	NRC-STS-1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order	SOMAN	NA NA	NA NA	\$19.88 \$11.85	NA NA	\$50.49 \$29.00	NA NA	NA NA	NA NA	NA NA
++	NRC-STS-1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order	SUMAN	INA	INA	C8.11¢	INA	\$∠9.00	INA	INA	INA	INA
++	2 mins VC Interesting Channel and 2 mins VC I again			<del> </del>					-		1
+	2-wire VG Interoffice Channel and 2-wire VG Local Loop:			-					1		
+ +	Recurring Charges	LIEALO	NIA.	040.46	<b>#45.40</b>	NIA	047.05	N10	NIA	NIA	045.54
+	2-wire VG Loop per month, Zone 1 (Note 1)	UEAL2	NA	\$18.48	\$15.40	NA	\$17.65	NA	NA	NA	\$15.54
+	2-wire VG Loop per month, Zone 2 (Note 1)	UEAL2	NA	\$22.43	\$17.78	NA	\$30.32	NA	NA	NA	\$19.55
1 1	2-wire VG Loop per month, Zone 3 (Note 1)	UEAL2	NA	\$27.87	\$28.26	NA	\$61.93	NA	NA	NA	\$28.02

	New EEL rates are the sum of the individual UNE network elements (interoffice		I								
	transport and loop [channelization if applicable].										
H	2-wire VG Loop per month, Zone 4 (Note 1)	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
$\vdash$	Interoffice Channel - Dedicated - 2-wire VG - FacilityTermination per month	U1TV2	NA	\$26.52	\$17.07	NA	\$19.10	NA	NA.	NA	\$18.33
$\Box$	Interoffice Channel - Dedicated - 2-wire VG - per mile per month	1L5XX	NA	\$0.01	\$0.0222	NA	\$0.0384	NA	NA.	NA	\$0.0173
$\Box$	Non-Recurring Charges - New EEL (Note 2)(Note 3)	120701		ψο.σ.	Ψ0.0222		ψ0.000.				ψοιστι σ
	NRC- 2-wire VG interoffice - 1st	U1TV2	NA	\$112.10	\$79.61	NA	\$104.23	NA	NA	NA	\$83.35
	NRC- 2-wire VG interoffice - Add'l	U1TV2	NA	\$67.61	\$36.08	NA	\$39.91	NA	NA	NA	\$20.88
	NRC-2-wire VG Local Loop - 1st	UEAL2	NA	\$38.02	\$104.17	NA	\$128.42	NA	NA	NA	\$192.97
	NRC-2-wire VG Local Loop - Add'l	UEAL2	NA	\$35.15	\$78.10	NA	\$93.60	NA	NA	NA	\$140.72
	NRC-2-wire VG interoffice channel and 2-wire VG Local Loop Combination - Electroni	SOMEC	NA	\$3.17	\$3.50	NA	\$3.50	NA	NA	NA	\$3.50
	NRC-2-wire VG interoffice channel and 2-wire VG Local Loop Combination - Manual S	SOMAN	NA	\$25.40	NA	NA	NA	NA	NA	NA	\$19.99
	NRC-2-wire VG interoffice channel and 2-wire VG Local Loop Combination - Manual S	SOMAN	NA	NA	\$37.88	NA	\$36.28	NA	NA	NA	NA
	NRC-2-wire VG interoffice channel and 2-wire VG Local Loop Combination - Manual \$	SOMAN	NA	NA	\$27.36	NA	\$26.20	NA	NA	NA	NA
	NRC-2-wire VG interoffice channel and 2-wire VG Local Loop Combination - Manual \$	SOMAN	NA	NA	NA	NA	\$19.47	NA	NA	NA	NA
	NRC-2-wire VG interoffice channel and 2-wire VG Local Loop Combination - Manual \$	SOMAN	NA	NA	NA	NA	\$8.06	NA	NA	NA	NA
	4-wire VG Interoffice Channel and 4-wire VG Local Loop:										
	Recurring Charges										
	4-wire VG Loop per month, Zone 1 (Note 1)	UEAL4	NA	\$24.26	NA	NA	NA	NA	NA	NA	NA
	4-wire VG Loop per month, Zone 2 (Note 1)	UEAL4	NA	\$35.51	NA	NA	NA	NA	NA	NA	NA
	4-wire VG Loop per month, Zone 3 (Note 1)	UEAL4	NA	\$78.35	NA	NA	NA	NA	NA	NA	NA
	4-wire VG Loop per month, Zone 4 (Note 1)	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	Interoffice Channel - Dedicated - 4-wire VG - FacilityTermination per month	U1TV4	NA	\$23.64	NA	NA	NA	NA	NA	NA	NA
	Interoffice Channel - Dedicated - 4-wire VG - per mile per month	1L5XX	NA	\$0.0098	NA	NA	NA	NA	NA	NA	NA
	Non-Recurring Charges - New EEL (Note 2)(Note 3)										
	NRC- 4-wire VG interoffice - 1st	U1TV4	NA	\$160.33	NA	NA	NA	NA	NA	NA	NA
	NRC- 4-wire VG interoffice - Add'l	U1TV4	NA	\$73.44	NA	NA	NA	NA	NA	NA	NA
	NRC-4-wire VG Local Loop - 1st	UEAL4	NA	\$329.76	NA	NA	NA	NA	NA	NA	NA
Щ	NRC-4-wire VG Local Loop - Add'l	UEAL4	NA	\$148.55	NA	NA	NA	NA	NA	NA	NA
Ш	NRC-4-wire VG interoffice channel and 4-wire VG Local Loop Combination - Electroni	SOMEC	NA	\$3.17	NA	NA	NA	NA	NA	NA	NA
oxdot	NRC-4-wire VG interoffice channel and 4-wire VG Local Loop Combination - Manual S	SOMAN	NA	\$25.40	NA	NA	NA	NA	NA	NA	NA
oxdot	NRC-4-wire VG interoffice channel and 4-wire VG Local Loop Combination - Manual S	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
oxdot	NRC-4-wire VG interoffice channel and 4-wire VG Local Loop Combination - Manual S		NA	NA	NA	NA	NA	NA	NA	NA	NA
<b>-</b>	NRC-4-wire VG interoffice channel and 4-wire VG Local Loop Combination - Manual S	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
	NRC-4-wire VG interoffice channel and 4-wire VG Local Loop Combination - Manual S	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
<b>-</b>	4 . 5011										
<u> </u>	4-wire 56 kbps Interoffice Channel and 4-wire 56kbps Local Loop:										
<b>-</b>	Recurring Charges	LIBLES		000.00	000.44		00100		TD D		<b>01500</b>
$\vdash \vdash$	4-wire 56kbps Loop per month, Zone 1 (Note 1)	UDL56	NA NA	\$39.08 \$57.21	\$26.44	NA NA	\$24.36	NA NA	TBD TBD	NA	\$15.92
	4-wire 56kbps Loop per month, Zone 2 (Note 1)	UDL56			\$26.42		\$41.85			NA NA	\$20.79
$\vdash\vdash$	4-wire 56kbps Loop per month, Zone 3 (Note 1)	UDL56 NA	NA NA	\$126.22 NA	\$46.53 NA	NA NA	\$86.47 NA	NA NA	TBD NA	NA NA	\$27.18 NA
$\vdash \vdash$	4-wire 56kbps Loop per month, Zone 4 (Note 1)	U1TD5	NA NA	\$23.64	\$16.45	NA NA	\$18.37	NA NA	\$17.40	NA NA	\$17.74
$\vdash \vdash$	Interoffice Channel - Dedicated - 4-wire 56kbps - FacilityTermination per month Interoffice Channel - Dedicated - 4-wire 56kbps - per mile per month	1L5XX	NA NA	\$23.64	\$16.45	NA NA	\$18.37	NA NA	\$17.40	NA NA	\$17.74
$\vdash \vdash$	Non-Recurring Charges - New EEL (Note 2)(Note 3)	ILOAA	INA	φυ.υυθδ	φυ.υ∠∠∠	INA	φυ.υ384	INA	φυ.υ∠δ∠	INA	φυ.1730
$\vdash \vdash$	NRC- 4-wire 56kbps interoffice - 1st	U1TD5	NA	\$160.33	\$79.61	NA	\$104.23	NA	\$137.48	NA	\$83.35
$\vdash$	NRC- 4-wire 56kbps interoffice - 1st  NRC- 4-wire 56kbps interoffice - Add'l	U1TD5	NA NA	\$73.44	\$36.08	NA NA	\$39.91	NA NA	\$137.48	NA NA	\$20.88
$\vdash\vdash$	NRC- 4-wire 56kbps Local Loop - 1st	UDL56	NA NA	\$73.44	\$348.55	NA NA	\$421.27	NA NA	\$52.58 \$489.04	NA NA	\$20.88
$\vdash$	NRC-4-wire 56kbps Local Loop - 1st  NRC-4-wire 56kbps Local Loop - Add'l	UDL56	NA NA	\$148.55	\$241.20	NA NA	\$274.74	NA NA	\$337.51	NA NA	\$421.28
H	NRC-4-wire 56kbps interoffice channel and 4-wire 56kbps Local Loop Combination - B	SOMEC	NA NA	\$3.17	\$3.50	NA NA	\$3.50	NA NA	\$3.50	NA NA	\$3.50
H	NRC-4-wire 56kbps interoffice channel and 4-wire 56kbps Local Loop Combination - N	M SOMAN	NA NA	\$25.40	NA	NA NA	NA	NA NA	\$3.50 NA	NA NA	\$19.99
H	NRC-4-wire 56kbps interoffice channel and 4-wire 56kbps Local Loop Combination - N	1 SOMAN	NA NA	Ψ23.40 NA	\$37.88	NA NA	\$36.28	NA NA	\$65.01	NA NA	NA
$\vdash \vdash$	NRC-4-wire 56kbps interoffice channel and 4-wire 56kbps Local Loop Combination - N	1 SOMAN	NA NA	NA	\$27.36	NA NA	\$26.20	NA NA	\$50.83	NA NA	NA NA
$\vdash \vdash$	NRC-4-wire 56kbps interoffice channel and 4-wire 56kbps Local Loop Combination - N	1 SOMAN	NA NA	NA NA	NA	NA NA	\$11.41	NA NA	NA	NA NA	NA NA
H	NRC-4-wire 56kbps interoffice channel and 4-wire 56kbps Local Loop Combination - N	1 SOMAN	NA NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA
$\vdash \vdash$	The state of the s										1
				1					1		i

	New EEL rates are the sum of the individual UNE network elements (interoffice										
4	transport and loop [channelization if applicable].										
_	4-wire 64 kbps Interoffice Channel and 4-wire 64 kbps Local Loop:										
4	Recurring Charges										
1	4-wire 64kbps Loop per month, Zone 1 (Note 1)	UDL64	NA	\$39.08	\$26.44	NA	\$27.50	NA	TBD	NA	\$36.4
1	4-wire 64kbps Loop per month, Zone 2 (Note 1)	UDL64	NA	\$57.21	\$30.53	NA	\$47.24	NA	TBD	NA	\$45.8
1	4-wire 64kbps Loop per month, Zone 3 (Note 1)	UDL64	NA	\$126.22	\$46.53	NA	\$96.48	NA	TBD	NA	\$65.7
1	4-wire 64kbps Loop per month, Zone 4 (Note 1)	NA	NA	NA	NA	NA	NA	NA	NA	NA	N/
	Interoffice Channel - Dedicated - 4-wire 64kbps - FacilityTermination per month	U1TD6	NA	\$19.31	\$16.45	NA	\$18.37	NA	\$17.40	NA	\$17.
1	Interoffice Channel - Dedicated - 4-wire 64kbps - per mile per month	1L5XX	NA	\$0.0098	\$0.0222	NA	\$0.0384	NA	\$0.0282	NA	\$0.17
	Non-Recurring Charges - New EEL (Note 2)(Note 3)										
	NRC- 4-wire 64kbps interoffice - 1st	U1TD6	NA	\$160.33	\$79.61	NA	\$713.57	NA	\$137.48	NA	\$729
	NRC- 4-wire 64kbps interoffice - Add'l	U1TD6	NA	\$73.44	\$36.08	NA	\$404.36	NA	\$52.58	NA	\$411
	NRC-4-wire 64kbps Local Loop - 1st	UDL64	NA	\$329.76	\$348.55	NA	\$811.30	NA	\$489.04	NA	\$829
	NRC-4-wire 64kbps Local Loop - Add'l	UDL64	NA	\$148.55	\$241.20	NA	\$502.09	NA	\$337.51	NA	\$512
	NRC-4-wire 64kbps interoffice channel and 4-wire 64kbps Local Loop Combination - #	SOMEC	NA	\$3.17	\$3.50	NA	\$3.50	NA	\$3.50	NA	\$3.
	NRC-4-wire 64kbps interoffice channel and 4-wire 64kbps Local Loop Combination - M	SOMAN	NA	\$25.40	NA	NA	NA	NA	NA	NA	\$19
	NRC-4-wire 64kbps interoffice channel and 4-wire 64kbps Local Loop Combination - M	SOMAN	NA	NA	\$37.55	NA	\$100.50	NA	\$65.01	NA	N
	NRC-4-wire 64kbps interoffice channel and 4-wire 64kbps Local Loop Combination - M	SOMAN	NA	NA	\$37.55	NA	\$100.50	NA	\$50.83	NA	N
	NRC-4-wire 64kbps interoffice channel and 4-wire 64kbps Local Loop Combination - M	1 SOMAN	NA	NA	\$18.03	NA	\$41.88	NA	NA	NA	N
	NRC-4-wire 64kbps interoffice channel and 4-wire 64kbps Local Loop Combination - IV	1 SOMAN	NA	NA	\$18.03	NA	\$41.88	NA	NA	NA	N
	letwork Elements used in Existing Combinations at UNE Rates (Note4)	USOC	AL	FL	GA	KY	LA	MS	NC	SC	Т
N	Local Loop:	0300	AL	FL	GA	Νī	LA	IVIO	NC	30	<del></del>
											-
	2-Wire Analog Voice Grade Loop - Service Level 2	LINOVA	41 END	<b>#</b> 0.00	<b>#</b> 0.00	<b>#</b> 0.00	<b>#0.00</b>	<b>#</b> 0.00	<b>#0.00</b>	<b>#</b> 0.00	•
	2-Wire Analog Voice Grade Loop - per mile per month	UNCVX	1L5ND	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0
	2-Wire Analog Voice Grade Loop -per month	UEAL2	NA 047.05	\$20.52	NA O45.40	NA 047.07	NA MAZ OF	NA ©40.05	\$19.50	NA COL 57	N
	Zone 1	UEAL2	\$17.95	TBD	\$15.40	\$17.27	\$17.65	\$18.35	TBD	\$21.57	\$15
	Zone 2	UEAL2	\$29.16	TBD TBD	\$17.78	\$32.32	\$30.32	\$24.33	TBD TBD	\$32.53	\$19
	Zone 3	UEAL2	\$52.84 NA	NA	\$28.26 NA	\$55.78 NA	\$61.93 NA	\$34.77	NA	\$43.08 NA	\$28
	Zone 4	UEAL2	NA NA	NA	NA	INA	INA	\$45.88	INA	NA	N
	NRC - Ordinarily Combined in GA (Note 5)	LIEALO			<b>#</b> 10117						-
	NRC - 1st NRC - Add'l	UEAL2 UEAL2	NA NA	NA NA	\$104.17 \$78.10	NA NA	NA NA	NA NA	NA NA	NA NA	N N
		UEAL2	NA NA				NA NA	NA NA	NA NA		+
	NRC - Disconnect Charge - 1st			NA	NA NA	NA				NA	N
	NRC - Disconnect Charge - Add'l	UEAL2	NA	NA	NA no.50	NA	NA	NA	NA	NA	N
	NRC - Electronic Svc Order, per LSR	SOMEC	NA NA	NA NA	\$3.50	NA NA	NA NA	NA	NA NA	NA NA	N
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA NA	NA NA	\$18.94	NA NA	NA NA	NA NA	NA NA	NA NA	N
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA NA	NA NA	\$8.42	NA NA	NA NA	NA NA	NA NA	NA NA	N
	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	N
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	LINCCC	<b>#</b> E4.02	<b>#22.7</b> 5	PEO 42	<b>#</b> E4.00	<b>#</b> E4.00	<b>\$54.00</b>	<b>\$54.00</b>	<b>\$54.00</b>	0.5
	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54
	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add'I	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32
	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0
	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'I	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0
-	2-Wire Analog Voice Grade Loop - Service Level 2 - (reverse battery)										<del>                                     </del>
	2-Wire Analog Voice Grade Loop - Loop[ Start - per mile per month	1L5ND	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0
	2-Wire Analog Voice Grade Loop - Rev Bat - per month	UEAR2	NA	\$20.52	NA	NA	NA	NA	\$19.50	NA	N
	Zone 1	UEAR2	\$17.95	\$23.23	\$16.84	\$17.27	\$17.65	\$18.35	TBD	\$21.57	\$15
	Zone 2	UEAR2	\$29.16	\$22.43	\$19.45	\$32.32	\$30.32	\$24.33	TBD	\$32.53	\$19
	Zone 3	UEAR2	\$52.84	\$27.87	\$30.92	\$55.78	\$61.93	\$34.77	TBD	\$43.08	\$28
	Zone 4	UEAR2	NA	NA	NA	NA	NA	\$45.88	NA	NA	I V
	NRC - Ordinarily Combined in GA (Note 5)										
			NA	NA	\$104.17	NA	NA	NA	NA	NA	N
	NRC - 1st	UEAR2	NA NA	INA	Φ10 <del>4</del> .17	INA	INA	INA	INA		
	NRC - 1st NRC - Add'l	UEAR2 UEAR2	NA NA	NA NA	\$78.10	NA NA	NA NA	NA NA	NA NA	NA NA	I N

	New EEL rates are the sum of the individual UNE network elements (interoffice										
	transport and loop [channelization if applicable].										
	NRC - Disconnect Charge - Add'l	UEAR2	NA	NA	NA	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	NA	\$18.94	NA	NA	NA	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	\$8.42	NA	NA	NA	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)										
$\sqcup \sqcup$	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
$\vdash$	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
$\vdash \vdash \vdash$	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
$\vdash \vdash \vdash$	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'l	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
$\vdash$	A Mina Anglan Vaiga Crada Lagr			-					-		
$\vdash$	4-Wire Analog Voice Grade Loop	4L END	<b>#</b> 0.00	<b>#0.00</b>	<b>#</b> 0.00	<b>#</b> 0.00	<b>#0.00</b>	<b>#</b> 0.00	<b>#0.00</b>	<b>#</b> 0.00	<b>#0.00</b>
$\vdash$	4-Wire Analog Voice Grade Loop per mile per month	1L5ND	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
$\vdash \vdash \vdash$	Zone 1	UEAL4 UEAL4	\$24.01	\$24.26	\$22.88 \$26.42	\$20.92	\$24.36	\$22.38	TBD TBD	\$49.47 \$44.44	\$15.54
+++	Zone 2 Zone 3	UEAL4	\$39.00 \$70.67	\$35.51 \$78.35	\$26.42 \$41.99	\$39.14 \$67.56	\$41.85 \$85.47	\$29.67 \$42.40	TBD	\$44.44 \$58.85	\$19.55 \$28.02
+++	Zone 4	UEAL4	\$70.67 NA	NA	- 541.99 NA	NA	Φ05.47 NA	\$55.96	NA	, ф36.63 NA	Φ20.02 NA
+++	NRC - Ordinarily Combined in GA (Note 5)	UEAL4	INA	INA	INA	INA	INA	φ55.90	INA	INA	INA
+++	NRC - 1st	UEAL4	NA	NA	\$206.95	NA	NA	NA	NA	NA	NA
+++	NRC - Add'l	UEAL4	NA NA	NA NA	\$170.57	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - Disconnect Charge - 1st	UEAL4	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+++	NRC - Disconnect Charge - Add'l	UEAL4	NA	NA NA	NA NA	NA NA	NA NA	NA	NA NA	NA	NA NA
+++	NRC - Electronic Svc Order, per LSR	SOMEC	NA NA	NA NA	\$3.50	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	NA.	NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA.	\$8.42	NA	NA	NA	NA.	NA	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)										
	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'l	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	2-Wire ISDN Digital Grade Loop										
	2-Wire ISDN Loop per mile per month	1L5ND	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	2-Wire ISDN Digital Grade Loop per month	U1L2X	NA	NA	NA	NA	NA	NA	\$24.98	NA	NA
	Zone 1	U1L2X	\$23.23	\$32.34	\$21.89	\$23.66	\$21.15	\$21.86	TBD	\$26.68	\$15.54
	Zone 2	U1L2X	\$37.74	\$47.35	\$25.27	\$44.28	\$36.32	\$28.97	TBD	\$40.24	\$19.55
Ш	Zone 3	U1L2X	\$68.38	\$104.47	\$40.17	\$76.42	\$74.19	\$41.40	TBD	\$53.29	\$28.02
$\sqcup \sqcup$	Zone 4	U1L2X	NA	NA	NA	NA	NA	\$54.64	NA	NA	NA
$\sqcup \sqcup$	NRC - Ordinarily Combined in GA (Note 5)			<u> </u>					<u> </u>		
$\sqcup \!\!\!\! \perp$	NRC - 1st	U1L2X	NA	NA	\$233.38	NA	NA	NA	NA	NA	NA
$\vdash \vdash$	NRC - Add'l	U1L2X	NA	NA	\$180.35	NA	NA	NA	NA	NA	NA
$\vdash \vdash$	NRC - Disconnect Dharge - 1st	U1L2X	NA	NA	NA NA	NA NA	NA	NA	NA	NA	NA NA
$\vdash \vdash$	NRC - Disconnect Charge - Add'l	U1L2X	NA	NA	NA To 50	NA	NA	NA	NA	NA	NA
$\vdash \vdash$	NRC - Electronic Svc Order, per LSR	SOMEC	NA NA	NA NA	\$3.50	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
$\vdash \vdash$	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	NA NA	\$18.94	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA NA	NA NA	\$8.42	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
$\vdash \vdash$	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
H	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
++	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st		\$54.03 \$32.11	\$32.75 \$32.75		\$54.09 \$32.16	\$54.23 \$32.24	\$54.09 \$32.16	\$54.00		\$54.13 \$32.17
++	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add"I	UNCCC	\$32.11	\$32.75 \$16.77	\$26.99 \$12.61	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25 \$0.00	\$32.17
++	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'l	UNCCC	\$0.00	\$16.77	\$12.61 \$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
++	INNO- 2/4-VVINE CONIDINATION - SWIICH AS IS CONVERSION Charge - Disconnect - Add'i	UNCCC	φυ.υυ	φιυ.//	φ12.01	φυ.υυ	φυ.υυ	φυ.υυ	φυ.υυ	φυ.υυ	φυ.υυ
++	4-Wire 56 kbps Digital Grade Loop			<del>                                     </del>			1		<del>                                     </del>		1
$\sqcup \sqcup$	14-111116 OU NUPS DIGITAL GLADE LOUP			1	l		l .		1		l

New EEL rates are the sum of the individual UNE network elements (interoffice transport and loop [channelization if applicable].										
4-Wire 56 kbps Digital Grade Loop per mile per month	1L5ND	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
4-Wire 56 kbps Digital Grade Loop per month	UDL56	NA	NA	NA	NA	NA	NA	\$32.67	NA	NA
Zone 1	UDL56	\$27.33	\$33.90	\$26.44	NA	\$27.50	\$25.61	TBD	\$34.26	\$36.45
Zone 2	UDL56	\$44.40	\$44.72	\$30.53	NA	\$47.25	\$33.94	TBD	\$51.67	\$45.87
Zone 3	UDL56	\$80.45	\$50.85	\$46.53	NA	\$96.48	\$48.51	TBD	\$68.43	\$65.75
Zone 4	UDL56	NA	NA	NA	NA	NA	\$64.02	NA	NA	NA
NRC - Ordinarily Combined in GA (Note 5)										
NRC - 1st	UDL56	NA	NA	\$348.55	NA	NA	NA	NA	NA	NA
NRC - Add'l	UDL56	NA	NA	\$241.20	NA	NA	NA	NA	NA	NA
NRC - Disconnect Dharge - 1st	UDL56	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Disconnect Charge - Add'l	UDL56	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	NA	\$18.94	NA	NA	NA	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	\$8.42	NA	NA	NA	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)										
NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.1
NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.1
NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.0
NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'l	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.0
		******	<b>V</b> 10111	*	40.00	40.00	40.00	70.00	******	****
4-Wire 64 kbps Digital Grade Loop								1		
4-Wire 64 kbps Digital Grade Loop per mile per month	1L5ND	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.0
4-Wire 64 kbps Digital Grade Loop per month	UDL64	NA NA	NA	NA	NA	NA	NA	\$32.67	NA	NA
Zone 1	UDL64	\$27.33	\$33.90	\$26.44	NA	\$27.50	\$25.61	TBD	\$34.26	\$36.4
Zone 2	UDL64	\$44.40	\$44.72	\$30.53	NA	\$47.25	\$33.94	TBD	\$51.67	\$45.8
Zone 3	UDL64	\$80.45	\$50.85	\$46.53	NA	\$96.48	\$48.51	TBD	\$68.43	\$65.7
Zone 4	UDL64	NA	Ψ30.63 NA	Ψ <del>4</del> 0.55	NA NA	NA	\$64.02	NA	NA	NA
NRC - Ordinarily Combined in GA (Note 5)	ODLOT	14/3	14/-1	14/3	1973	14/4	ψ04.02	14/3	1973	11/1
NRC - 1st	UDL64	NA	NA	\$348.55	NA	NA	NA	NA	NA	NA
NRC - Add'l	UDL64	NA NA	NA NA	\$241.20	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - Disconnect Dharge - 1st	UDL64	NA NA	NA NA	9241.20 NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - Disconnect Charge - 1st	UDL64	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - Electronic Svc Order, per LSR	SOMEC	NA NA	NA NA	\$3.50	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - Electronic SVC Order, per LSR  NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA NA	NA NA	\$3.50 \$18.94	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA NA	NA NA	\$8.42	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - Incremental Charge - Manual Service Order - Add 1  NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	NA NA	NA NA	Φ0.42 NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	SOMAN	INA	INA	INA	INA	INA	INA	INA	INA	INA
NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)  NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.
<del>-</del>	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$34.00	\$32.25	\$32.
NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add"I	UNCCC			\$26.99	\$0.00	\$32.24	\$32.16	\$32.10	\$32.25	\$32.
NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00 \$0.00	\$16.77 \$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.0
NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'l	UNCCC	\$0.00	\$10.77	\$12.01	φυ.υυ	φυ.υυ	\$0.00	\$0.00	\$0.00	φ0.0
4-Wire DS1 Digital Loop	-	-					-	-	-	1
	1L5ND	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.0
4-Wire DS1 Digital Loop per mile per month										
4-Wire DS1 Digital Loop per month	USLXX	NA OF4.74	NA COA CO	NA ©50.40	NA TEO OO	NA ©50.00	NA ©50.00	\$62.78	NA OFFICIAL	NA
Zone 1	USLXX	\$51.74	\$64.69	\$52.40	\$50.28	\$56.32	\$50.99	TBD	\$59.61	NA
Zone 2	USLXX	\$84.05	\$94.71	\$60.51	\$94.06	\$96.73	\$67.58	TBD	\$89.90	NA NA
Zone 3	USLXX	\$152.29	\$208.93	\$96.18	\$162.34	\$197.57	\$96.58	TBD	\$119.06	NA NA
Zone 4	USLXX	NA	NA	NA	NA	NA	\$127.47	NA	NA	NA
NRC - Ordinarily Combined in GA (Note 5)										1
NRC - 1st	USLXX	NA	NA	\$429.98	NA	NA	NA	NA	NA	NA
NRC - Add'l	USLXX	NA	NA	\$268.18	NA	NA	NA	NA	NA	NA
NRC - Disconnect Charge - 1st	USLXX	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Disconnect Charge - Add'l	USLXX	NA	NA	NA	NA	NA	NA	NA	NA	N/

1 1	П	New EEL rates are the sum of the individual UNE network elements (interoffice		I	T	1	1	I			1	1
		transport and loop [channelization if applicable].										
+		NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
+		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA NA	NA NA	\$18.94	NA NA	NA NA	NA	NA NA	NA NA	NA NA
+	$\dashv$	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA NA	NA NA	\$8.42	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	$\dashv$	NRC - Incremental Charge - Manual Service Order - Add 1	SOMAN	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+		NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	JOINAIN	INA	INA	INA	INA	INA	INA	INA	INA	INA
+		NRC-DS1 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
+		NRC-DS1 COMBINATION - Switch As Is Conversion Charge - 1st  NRC-DS1 COMBINATION - "Switch As Is" Conversion Charge - Add"I	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
+		NRC-DS1 COMBINATION - Switch As Is Conversion Charge - Add1  NRC-DS1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
+		NRC- DS1 COMBINATION - Switch As is Conversion Charge - Disconnect - 1st  NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'I	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
+	-	NRC-DST COMBINATION - SWICH AS IS CONVEISION Charge - Disconnect - Add t	UNCCC	φ0.00	\$10.77	\$12.01	φυ.υυ	φυ.υυ	φυ.υυ	\$0.00	φ0.00	\$0.00
$\pm$	-	DS3 Local Loop										
+	$\dashv$	per mile per month	1L5ND	\$10.85	\$11.77	\$8.90	\$43.69	\$38.98	\$54.39	\$32.53	\$15.53	\$30.53
+	$\dashv$	facitility termination per month	UE3PX	\$419.65	\$404.58	\$390.34	\$436.95	\$497.08	\$427.81	\$387.01	\$421.60	\$400.21
+		NRC - Ordinarily Combined in GA (Note 5)	OLSI X	Ψ+15.05	ψ+0+.50	ψ550.54	ψ-30.33	Ψ-37.00	ψ+21.01	ψ507.01	Ψ-21.00	ψ+00.21
$\pm$	$\dashv$	NRC - Facility Termination - 1st	UE3PX	NA	NA	\$639.50	NA	NA	NA	NA	NA	NA
+	$\dashv$	NRC - Facility Termination - 1st	UE3PX	NA NA	NA NA	\$426.40	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	$\dashv$	NRC - Facility Termination - Add 1	UE3PX	NA NA	NA NA	\$122.31	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	$\dashv$	NRC - Facility Termination - Disconnect - Add'l	UE3PX	NA NA	NA NA	\$119.14	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	$\dashv$	NRC - Manual Svc Order, per LSR	SOMAN	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	$\dashv$	NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	$\dashv$	NRC - Electronic Svc Order, per LSR	SOMEC	NA NA	NA NA	\$3.50	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
$\pm$	$\dashv$	NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	$\dashv$	NRC - Incremental ChargeManual Svc Order - 1st	SOMAN	NA NA	NA NA	\$37.55	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	$\dashv$	NRC - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA NA	NA NA	\$37.55	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	$\dashv$	NRC - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	SOMAN	NA NA	NA NA	\$18.03	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	$\dashv$	NRC - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add'l	SOMAN	NA NA	NA NA	\$18.03	NA NA	NA	NA NA	NA NA	NA NA	NA NA
+	$\dashv$	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	OOMAN	13/3	11/7	ψ10.05	14/3	INA	14/3	1973	11/7	14/3
$\pm$		NRC-DS3 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
+		NRC-DS3 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
$\pm$		NRC- DS3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
$\pm$	I	NRC- DS3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'I	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	T	The Boo Compliant The Compliant Charge Blockmost Add.	0.1000	ψ0.00	ψ.σ	ψ12.01	ψ0.00	ψσ.σσ	ψσ.σσ	ψ0.00	ψ0.00	ψ0.00
11		STS-1 Local Loop										
	T	per mile per month	1L5ND	\$10.85	\$11.77	\$8.90	\$43.69	\$38.98	\$54.39	\$32.53	\$15.53	\$30.53
	7	facitility termination per month	UDLS1	\$419.65	\$446.09	\$421.59	\$436.95	\$497.08	\$427.81	\$387.01	\$431.32	\$400.21
		NRC - Ordinarily Combined in GA (Note 5)										
	T	NRC - STS-1 - Facility Termination - 1st	UDLS1	NA	NA	\$639.50	NA	NA	NA	NA	NA	NA
		NRC - STS-1 - Facility Termination - Add'l	UDLS1	NA	NA	\$426.40	NA	NA	NA	NA	NA	NA
		NRC - STS-1 - Facility Termination - Disconnect - 1st	UDLS1	NA	NA	\$122.31	NA	NA	NA	NA	NA	NA
	T	NRC - STS-1 - Facility Termination - Disconnect - Add'l	UDLS1	NA	NA	\$119.14	NA	NA	NA	NA	NA	NA
	T	NRC - Manual Svc Order, per LSR	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
		NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
		NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
		NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	NA	NA	NA	NA	NA	NA	NA	NA
		NRC - STS-1 - Incremental ChargeManual Svc Order - 1st	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
		NRC - STS-1 - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
		NRC - STS-1 - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
		NRC - STS-1 - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
		NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)										
		NRC-STS-1 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
		NRC-STS-1 COMBINATION - "Switch As Is" Conversion Charge - Add'I	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
		NRC- STS-1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
		NRC- STS-1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'I	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
			·									
		OC-3 Local Loop	1L5ND	\$1,123	\$8.93	\$6.75	\$33.15	\$29.58	\$41.27	\$24.69	\$11.78	\$23.16

New EEL rates are the sum of the individual UNE network elements (interoffice transport and loop [channelization if applicable].										
per mile per month		\$7.09	\$648.60	\$630.21	\$436.95	\$753.65	\$689.68	\$611.36	\$701.71	\$620.20
facility termiantion per month										
NRC - Ordinarily Combined in GA (Note 5)										
NRC - OC3 - Facility Termination - 1st		NA	NA	\$947.69	NA	NA	NA	NA	NA	NA
NRC - OC3 - Facility Termination - Add'l		NA	NA	\$413.00	NA	NA	NA	NA	NA	NA
NRC - OC3 - Facility Termination - Disconnect - 1st		NA	NA	\$122.31	NA	NA	NA	NA	NA	NA
NRC - OC3 - Facility Termination - Disconnect - Add'l		NA	NA	\$119.14	NA	NA	NA	NA	NA	NA
NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
NRC - OC3 - Incremental ChargeManual Svc Order - 1st	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
NRC - OC3 - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
NRC - OC3 -Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
NRC - OC3 -Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add'l	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)										
NRC-OC-3 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
NRC-OC-3 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.1
NRC- OC-3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
NRC- OC-3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'l	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.0
		*			*	*				
OC-12 Local Loop										
per mile per month	1L5ND	\$10.13	\$11.18	\$8.31	\$40.80	\$36.40	\$50.79	\$30.38	\$14.50	\$28.5
facility termination per month		\$5,630	\$2,068	\$2,109.00	\$2,457	\$2,571	\$2,371	\$2,122	\$2,663	\$2,07
NRC - Ordinarily Combined in GA (Note 5)		φο,σσσ	<b>\$2,000</b>	ψ2,:00:00	ψ2,	Ψ2,0	ΨΞ,σ:	<b>42,.22</b>	<b>\$2,000</b>	ΨΞ,σ.
NRC - OC12 - Facility Termination - 1st		NA	NA	\$1.162.00	NA	NA	NA	NA	NA	NA
NRC - OC12 - Facility Termination - Add'l		NA NA	NA NA	\$413.00	NA NA	NA NA	NA NA	NA NA	NA NA	NA
NRC - OC12 - Facility Termination - Disconnect - 1st		NA NA	NA NA	\$122.31	NA NA	NA NA	NA	NA NA	NA NA	NA NA
NRC - OC12 - Facility Termination - Disconnect - Add'l		NA NA	NA NA	\$119.14	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - Electronic Svc Order, per LSR	SOMEC	NA NA	NA NA	\$3.50	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - Clectronic Svc Order, per ESK  NRC -OC12 - Incremental Charge - Manual Svc Order - 1st	SOMAN	NA NA	NA NA	\$37.55	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - OC12 - Incremental Charge - Manual Svc Order - 1st	SOMAN	NA NA	NA NA	\$37.55	NA NA	NA NA	NA	NA NA	NA NA	NA
NRC - OC12 - Incremental Cost-Manual Svc. Order vs. Elect-Disconnect-1st	SOMAN	NA NA	NA NA	\$18.03	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - OC12 - Incremental Cost-Manual Svc. Order vs. Elect-Disconnect-Add'l	SOMAN	NA NA	NA NA	\$18.03	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	SOMAN	INA	INA	\$10.03	INA	INA	INA	INA	INA	INA
NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	ΦE4.00	\$54.26	\$54.
NRC-OC-12 COMBINATION - "Switch As Is" Conversion Charge - 1st								\$54.00		
NRC-OC-12 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.1
NRC- OC-12 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.0
NRC- OC-12 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'I	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.0
00.401.00011.000										
OC-48 Local Loop	41.5110	<b>#</b> 00.00	000.07	007.05	<b>0.100.50</b>	<b>0.110.10</b>	A400 F0	<b>#</b> 400.00	0.47.57	000
per mile per month	1L5ND	\$33.22	\$36.67	\$27.25	\$166.59	\$119.40	\$166.59	\$120.02	\$47.57	\$93.
facility termination per month		\$1,947	\$1,699	\$1,598.00	\$2,129	\$2,268	\$1,753	\$1,677	\$1,733	\$1,83
OC-12 Interface on OC-48 Loop per month		\$699.62	\$592.09	\$594.80	\$725.77	\$723.29	\$667.00	\$582.66	\$773.40	\$570.
NRC - Ordinarily Combined in GA (Note 5)										
NRC - OC48 - Facility Termination - 1st		NA	NA	\$1,162.00	NA	NA	NA	NA	NA	NA
NRC - OC48 - Facility Termination - Add'l		NA	NA	\$413.00	NA	NA	NA	NA	NA	NA
NRC - OC48 - Interface OC12 on OC48 - 1st		NA	NA	\$539.36	NA	NA	NA	NA	NA	NA
NRC - OC48 - Interface OC12 on OC48 - Add'l		NA	NA	\$317.38	NA	NA	NA	NA	NA	NA
NRC - OC48 - Facility Termination - Disconnect - 1st		NA	NA	\$122.31	NA	NA	NA	NA	NA	NA
NRC - OC48 - Facility Termination - Disconnect - Add'l		NA	NA	\$119.14	NA	NA	NA	NA	NA	NA
NRC - OC48- Interface OC12 on OC48 - Disconnect - 1st		NA	NA	\$122.31	NA	NA	NA	NA	NA	N/
NRC - OC48 - Interface OC12 on OC48 - Disconnect - Add'l		NA	NA	\$119.14	NA	NA	NA	NA	NA	N/
NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	N/
NRC - OC48 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
NRC - OC48 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-	A SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
NRC - OC48 - Interface - Manual Svc Order vs Electronic-Disconnect-1st	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
NRC - OC48 - Interface - Manual Svc Order vs Electronic-Disconnect-Add'l	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA

	New EEL rates are the sum of the individual UNE network elements (interoffice										
	transport and loop [channelization if applicable].										
	NRC - OC-48 - Incremental ChargeManual Svc Order-1st	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC - OC-48 - Incremental ChargeManual Svc Order-Add'l	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC - OC48 - Interface OC12 on OC48 - Incremental ChargeManual Svc Orde	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC - OC48 - Interface OC12 on OC48 - Incremental ChargeManual Svc Orde	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
$\sqcup \sqcup$	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)										
$\sqcup \sqcup$	NRC-OC-48 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
$\perp \perp \perp$	NRC-OC-48 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
$\perp \perp \perp$	NRC- OC-48 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
$\vdash$	NRC- OC-48 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'l	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Local Channels:										
	Local Channel - Dedicated - 2-Wire VG										
	Monthly Recurring per month	ULDV2	\$14.61	\$29.33	\$13.91	\$22.26	\$14.94	\$17.83	\$14.83	\$16.83	\$19.02
	Monthly Recurring per mile per month	1L5NC	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC - Ordinarily Combined in GA (Note 5)										*
	NRC - 2-wire VG Local Channel - 1st	ULDV2	NA	NA	\$245.43	NA	NA	NA	NA	NA	NA
	NRC - 2-wire VG Local Channel -Add'l	ULDV2	NA	NA	\$33.90	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire VG - Incremental ChargeManual Svc Order - 1st	SOMAN	NA	NA	\$33.63	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire VG - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA	NA	\$27.49	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire VG - Incremental ChargeManual Svc Order - Disconnect - 1st	SOMAN	NA	NA	\$22.24	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire VG - Incremental ChargeManual Svc Order - Disconnect - Add'l	SOMAN	NA	NA	\$8.08	NA	NA	NA	NA	NA	NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)										
	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
ш	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add'I	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'I	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
$\vdash \vdash$											
$\vdash \vdash \vdash$	Local Channel - Dedicated - 2-Wire VG - Rev Bat.		****				*				
$\vdash \vdash$	Monthly Recurring per month	ULDR2	\$14.61	\$26.31	\$13.91	\$22.26	\$14.94	\$17.83	\$14.82	\$16.83	\$19.02
$\vdash \vdash \vdash$	Monthly Recurring per mile per month	1L5NC	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
oxdot	NRC - Ordinarily Combined in GA (Note 5)				****						
$\vdash \vdash \vdash$	NRC - 2-wire VG Local Channel - 1st	ULDR2	NA	NA	\$245.43	NA	NA	NA	NA	NA	NA
$\vdash \vdash \vdash$	NRC - 2-wire VG Local Channel -Add'l	ULDR2	NA	NA	\$33.90	NA	NA	NA	NA	NA	NA
$\vdash \vdash \vdash$	NRC - Electronic Svc Order, per LSR	SOMEC SOMAN	NA NA	NA	\$3.50 \$33.63	NA NA	NA	NA NA	NA NA	NA NA	NA NA
$\vdash\vdash\vdash$	NRC - 2-Wire VG - Incremental ChargeManual Svc Order - 1st	SOMAN	NA NA	NA NA	\$33.63	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++-	NRC - 2-Wire VG - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA NA	NA NA	\$27.49	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++-	NRC - 2-Wire VG - Incremental ChargeManual Svc Order - Disconnect - 1st NRC - 2-Wire VG - Incremental ChargeManual Svc Order - Disconnect - Add'l	SOMAN	NA NA	NA NA	\$8.08	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
$\vdash$	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	SOMAN	INA	INA	\$0.00	INA	INA	INA	INA	INA	INA
++-	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
++-	NRC-2/4-WIRE COMBINATION - Switch As Is Conversion Charge - 1st  NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add"	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$34.13
++	NRC-2/4-WIRE COMBINATION - Switch As is Conversion Charge - Add I  NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$32.75 \$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
++-	NRC- 2/4-WIRE COMBINATION - Switch As Is Conversion Charge - Disconnect - Add'l	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
$\vdash$	The State of the S	011000	ψυ.υυ	ψ10.77	ψ12.01	ψο.σο	ψο.οο	ψ0.00	Ψ0.00	ψ0.00	Ψ0.00
	Local Channel - Dedicated - 4-Wire VG										
	Monthly Recurring per month	ULDV4	\$15.77	\$30.50	\$14.99	\$23.38	\$16.21	\$19.03	\$15.87	\$18.05	\$20.14
	Monthly Recurring per mile per month	1L5NC	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC - Ordinarily Combined in GA (Note 5)										
	NRC-4-wire VG Local Channel - 1st	ULDV4	NA	NA	\$245.43	NA	NA	NA	NA	NA	NA
	NRC-4-wire VG Local Channel - Add'l	ULDV4	NA	NA	\$33.90	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
	NRC - 4-Wire VG Local Channel - Incremental ChargeManual Svc Order - 1st	SOMAN	NA	NA	\$33.63	NA	NA	NA	NA	NA	NA
	NRC - 4-Wire VG Local Channel - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA	NA	\$27.49	NA	NA	NA	NA	NA	NA
	NRC - 4-Wire VG Local Channel - Incremental ChargeManual Svc Order - Disconne	SOMAN	NA	NA	\$22.24	NA	NA	NA	NA	NA	NA

#### BELLSOUTH/BTI RATES NETWORK ELEMENTS AND OTHER SERVICES ENHANCED EXTENDED LINKS

	New EEL rates are the sum of the individual UNE network elements (interoffice transport and loop [channelization if applicable].										
	NRC - 4-Wire VG Local Channel - Incremental ChargeManual Svc Order - Disconne	SOMAN	NA	NA	\$8.08	NA	NA	NA	NA	NA	NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)				70.00						
	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'l	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
$\pm$	INTO-2/4-WINE COMBINATION - Switch As is Conversion Charge - Disconnect - Add i	011000	ψ0.00	ψ10.77	Ψ12.01	ψ0.00	ψ0.00	Ψ0.00	ψ0.00	Ψ0.00	ψ0.00
	Local Channel - Dedicated - DS1										
	DS1 Monthly Recurring per month	ULDF1	\$35.52	\$43.53	\$38.36	\$43.80	\$43.80	\$38.91	\$35.68	\$37.20	\$40.27
$\pm$	Monthly Recurring per mile per month	1L5NC	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC - Ordinarily Combined in GA (Note 5)	120110	ψ0.00	ψ0.00	ψ0.00	ψ0.00	ψ0.00	Ψ0.00	ψ0.00	ψο.σσ	ψ0.00
+	NRC - DS1 Local Channel - 1st	ULDF1	NA	NA	\$166.88	NA	NA	NA	NA	NA	NA
+	NRC - DS1 Local Channel - Add'l	ULDF1	NA NA	NA NA	\$84.14	NA NA	NA NA	NA NA	NA NA	NA NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	NA NA	NA NA	\$3.50	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - DS1 Local Channel - Incremental ChargeManual Svc Order - 1st	SOMAN	NA NA	NA NA	\$65.60	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - DS1 Local Channel - Incremental ChargeManual Svc Order - 1st  NRC - DS1 Local Channel - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA NA	NA NA	\$33.53	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
-				NA NA					NA NA	NA NA	NA NA
++	NRC - DS1 Local Channel - Incremental ChargeManual Svc Order - Disconnect - 1s	SOMAN	NA		\$35.76	NA NA	NA	NA			
-	NRC - DS1 Local Channel - Incremental ChargeManual Svc Order - Disconnec - Ad	SOMAN	NA	NA	\$14.05	NA	NA	NA	NA	NA	NA
++	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	1111000	05400	000 75	050.40	05400	<b>A</b> = 4.00	<b>\$54.00</b>	05400	<b>05400</b>	<b>#</b> 54.40
	NRC-DS1 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
	NRC-DS1 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
	NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
+	NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'I	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
11	Local Channel - Dedicated - DS3	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	DS3 Local Channel - per mile per month	1L5NC	\$8.44	\$9.32	\$6.92	\$34.00	\$30.34	NA	NA	\$12.08	\$23.76
	DS3 Local Channel - Facility Termination per month	ULDF3	\$525.40	\$560.39	\$515.91	\$635.09	\$558.00	\$531.39	\$512.00	\$481.14	\$615.65
	NRC - Ordinarily Combined in GA (Note 5)	OLDIO	Ψ020.10	Ψ000.00	φ010.01	Ψ000.00	Ψ000.00	φοσ1.00	φ012.00	φισι.ιι	ψ010.00
	NRC - DS3 Local Channel Facility Termination - 1st	ULDF3	NA	NA	\$639.50	NA	NA	NA	NA	NA	NA
	NRC - DS3 Local Channel - Facility Termination - Add'l	ULDF3	NA NA	NA NA	\$426.40	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
$\pm$	NRC - Electronic Svc Order, per LSR	SOMEC	NA NA	NA NA	\$3.50	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - DS3 Local Channel - Incremental ChargeManual Svc Order - 1st	SOMAN	NA NA	NA NA	\$37.55	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - DS3 Local Channel - Incremental ChargeManual Svc Order - 1st	SOMAN	NA NA	NA NA	\$37.55	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - DS3 Local Channel - Incremental ChargeManual Svc Order - Add 1	SOMAN	NA NA	NA NA	\$18.03	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - DS3 Local Channel - Incremental ChargeManual Svc Order - Disconnec - Ad		NA NA	NA NA	\$18.03	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	SOMAIN	INA	INA	\$10.03	INA	INA	INA	INA	INA	INA
++	NRC-DS3 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$71.04	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
	NRC-DS3 COMBINATION - Switch As Is Conversion Charge - 1st	UNCCC	\$32.11	\$32.75	\$39.60	\$32.16	\$32.24	\$32.16	\$34.00	\$32.25	\$32.17
++	NRC-DS3 COMBINATION - Switch As Is Conversion Charge - Disconnect - 1st					\$0.00			\$0.00		\$0.00
++	NRC- DS3 COMBINATION - Switch As is Conversion Charge - Disconnect - Add'l	UNCCC	\$0.00 \$0.00	\$16.77 \$16.77	\$0.00 \$0.00	\$0.00	\$0.00 \$0.00	\$0.00 \$0.00	\$0.00	\$0.00 \$0.00	\$0.00
+	INRC- D53 COMBINATION - Switch As is Conversion Charge - Disconnect - Add I	UNCCC	\$0.00	\$10.77	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Local Channel - Dedicated - STS-1										
$\dashv \dagger$	STS-1 Local Channel - per mile per month	1L5NC	\$8.44	\$9.32	\$6.92	\$34.00	\$8.77	\$38.98	NA	\$12.08	\$25.11
$\dashv \dagger$	STS-1 Local Channel - Facility Termination per month	ULDFS	\$525.40	\$569.67	\$517.56	\$635.09	\$558.00	\$531.39	\$512.00	\$481.14	\$615.65
+	NRC - Ordinarily Combined in GA (Note 5)	02510	Ψ020.70	ψυσυ.στ	ψο 17.00	ψοσο.σσ	Ψ000.00	ψοσ1.00	ψο 12.00	ΨΙΟΙ.ΙΤ	ψο 10.00
++	NRC - STS-1 Local Channel Facility Termination - 1st	ULDFS	NA	NA	\$761.81	NA	NA	NA	NA	NA	NA
++	NRC - STS-1 Local Channel - Facility Termination - Add'l	ULDFS	NA NA	NA NA	\$545.54	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - Electronic Svc Order, per LSR	SOMEC	NA NA	NA NA	\$3.50	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - STS-1 Local Channel - Incremental ChargeManual Svc Order - 1st	SOMAN	NA NA	NA NA	\$37.55	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+		SOMAN	NA NA	NA NA		NA NA	NA NA	NA NA	NA NA	NA NA	
++	NRC - STS-1 Local Channel - Incremental ChargeManual Svc Order - Add'l				\$37.55						NA NA
++	NRC - STS-1 Local Channel - Incremental ChargeManual Svc Order - Disconnect -	SOMAN	NA	NA NA	\$18.03	NA NA	NA	NA	NA	NA	NA
+	NRC - STS-1 Local Channel - Incremental ChargeManual Svc Order - Disconnec - A	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
++	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	LINICOO	ΦΕ 4 OO	#00.75	<b>#FO 40</b>	<b>#</b> F4.00	<b>05400</b>	<b>#</b> E4.00	ΦE4.00	<b>#</b> E4.00	05440
+	NRC-STS-1 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
	NRC-STS-1 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17

ПП	New EEL rates are the sum of the individual UNE network elements (interoffice										
	transport and loop [channelization if applicable].										
	NRC- STS-1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC- STS-1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'I	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Local Channel - OC3										
	Local Channel - OC3 - per Mile	1L5NC	\$8.23	\$7.83	\$6.75	\$33.15	\$29.58	\$41.27	\$24.69	\$11.78	\$23.16
	Local Channel - OC3 - per Facility Termination	TBA	\$691.33	\$940.35	\$630.21	\$713.29	\$753.65	\$689.68	\$611.36	\$701.71	\$620.20
	NRC - Ordinarily Combined in GA (Note 5)										
	NRC - OC3 - Facility Termination - 1st	TBA	NA	NA	\$947.69	NA	NA	NA	NA	NA	NA
	NRC - OC3 - Facility Termination - Add'I	TBA	NA	NA	\$413.00	NA	NA	NA	NA	NA	NA
	NRC - OC3 - Facility Termination - Disconnect - 1st	TBA	NA	NA	\$122.31	NA	NA	NA	NA	NA	NA
	NRC - OC3 - Facility Termination - Disconnect - Add'l	TBA	NA	NA	\$119.14	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
	NRC - OC3 - Incremental ChargeManual Svc Order - 1st	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
	NRC - OC3 - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
	NRC - OC3 -Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC - OC3 -Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add'l	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)				·						
	NRC-OC-3 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
	NRC-OC-3 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
	NRC- OC-3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC- OC-3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'I	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Local Channel - OC12	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	Local Channel - OC12 - per Mile	1L5NC	\$10.13	\$11.18	\$8.31	\$40.80	\$36.40	\$50.79	\$30.38	\$14.50	\$28.51
	Local Channel - OC12 - per Facility Termination	TBA	\$2,557	\$2,753	\$2,109.00	\$2.457	\$2,571	\$2,371	\$2.122	\$2,663	\$2,079
	NRC - Ordinarily Combined in GA (Note 5)		, , , , , ,	, , , , ,	, ,	,	. ,-	. , .	* /	, ,	, ,
	NRC - OC12 - Facility Termination - 1st	TBA	NA	NA	\$1.162.00	NA	NA	NA	NA	NA	NA
	NRC - OC12 - Facility Termination - Add'l	TBA	NA	NA	\$413.00	NA	NA	NA	NA	NA	NA
	NRC - OC12 - Facility Termination - Disconnect - 1st	TBA	NA	NA	\$122.31	NA	NA	NA	NA	NA	NA
	NRC - OC12 - Facility Termination - Disconnect - Add'l	TBA	NA	NA	\$119.14	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
	NRC -OC12 - Incremental Charge - Manual Svc Order - 1st	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
	NRC - OC12 - Incremental Charge - Manual Svc Order - Add'l	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
	NRC - OC12 - Incremental Cost-Manual Svc. Order vs. Elect-Disconnect-1st	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC - OC12 - Incremental Cost-Manual Svc. Order vs. Elect-Disconnect-Add'l	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)										
	NRC-OC-12 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
	NRC-OC-12 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
	NRC- OC-12 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC- OC-12 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
			* * * * * * * * * * * * * * * * * * * *			• • • • • • • • • • • • • • • • • • • •	*	*		*	*
	Local Channel - OC48	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	Local Channel - OC48 - per Mile	1L5NC	\$33.22	\$36.67	\$27.25	\$133.84	\$119.40	\$166.59	\$120.02	\$47.57	\$93.50
	Local Channel - OC48 - per Facility Termination	TBA	\$1,713	\$1,944	\$1,598.00	\$2,129	\$2,268	\$1,753	\$1,677	\$1,733	\$1,832
	Local Channel - OC12 interface on OC48 Facility	TBA	\$736.71	\$586.28	\$594.80	\$725.77	\$723.29	\$667.00	\$582.66	\$773.40	\$570.54
	NRC - Ordinarily Combined in GA (Note 5)				,						,
H	NRC - OC48 - Facility Termination - 1st	TBA	NA	NA	\$1,162.00	NA	NA	NA	NA	NA	NA
	NRC - OC48 - Facility Termination - Add'l	TBA	NA NA	NA.	\$413.00	NA NA	NA NA	NA	NA NA	NA	NA
$\top \!$	NRC - OC48 - Interface OC12 on OC48 - 1st	TBA	NA NA	NA.	\$539.36	NA	NA NA	NA	NA NA	NA	NA
ĦĦ	NRC - OC48 - Interface OC12 on OC48 - Add'I	TBA	NA	NA.	\$317.38	NA	NA	NA	NA.	NA	NA
$\dashv \dagger$	NRC - OC48 - Facility Termination - Disconnect - 1st	TBA	NA	NA	\$122.31	NA	NA	NA	NA.	NA	NA
	NRC - OC48 - Facility Termination - Disconnect - Add'l	TBA	NA NA	NA.	\$119.14	NA	NA NA	NA	NA NA	NA	NA
	NRC - OC48- Interface OC12 on OC48 - Disconnect - 1st	TBA	NA NA	NA NA	\$122.31	NA	NA	NA	NA	NA	NA
ĦĦ	NRC - OC48 - Interface OC12 on OC48 - Disconnect - Add'l	TBA	NA	NA.	\$119.14	NA	NA	NA	NA	NA	NA
ſĦ	NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA.	\$3.50	NA	NA	NA	NA.	NA	NA
					, ,,,,,,,,						

	New EEL rates are the sum of the individual UNE network elements (interoffice										
	transport and loop [channelization if applicable].										
	NRC - OC48 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-1st	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC - OC48 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-Add'l	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC - OC48 - Interface - Manual Svc Order vs Electronic-Disconnect-1st	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC - OC48 - Interface - Manual Svc Order vs Electronic-Disconnect-Add'l	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC - OC-48 - Incremental ChargeManual Svc Order-1st	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
	NRC - OC-48 - Incremental ChargeManual Svc Order-Add'l	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
	NRC - OC48 - Interface OC12 on OC48 - Incremental ChargeManual Svc Order-1st	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
	NRC - OC48 - Interface OC12 on OC48 - Incremental ChargeManual Svc Order-Ad	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)										
	NRC-OC-48 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
	NRC-OC-48 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
	NRC- OC-48 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC- OC-48 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Ĭ										
	Interoffice Channels:										
	Interoffice Channel - Dedicated - 2-wire VG										
	Interoffice Channel - Dedicated 2-wire VG - per mile per month	1L5XX	\$0.0339	\$0.0100	\$0.0222	\$0.0301	\$0.0384	\$0.0323	\$0.0282	\$0.0373	\$0.0173
	Interoffice Channel - Dedicated 2-wire VG - Facility Termination per month	U1TV2	\$18.49	\$26.72	\$17.07	\$27.66	\$19.10	\$21.33	\$18.01	\$21.42	\$18.33
	NRC - Ordinarily Combined in GA (Note 5)	02	ψ.σσ	Ψ202	<b>V</b>	Ψ27.00	ψ.σσ	Ψ21100	ψ.σ.σ.	Ψ22	ψ.σ.σσ
	NRC - 2-wire VG Interoffice Channel - Facility Termination - 1st	U1TV2	NA	NA	\$79.61	NA	NA	NA	NA	NA	NA
	NRC - 2-wire VG Interoffice Channel - Facility Termination - Add'l	U1TV2	NA	NA.	\$36.08	NA	NA	NA	NA.	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	NA NA	NA NA	\$3.50	NA NA	NA NA	NA NA	NA NA	NA	NA
	NRC - 2-wire VG Interoffice Channel - Incremental ChargeManual Svc Order - 1st	SOMAN	NA NA	NA NA	\$18.94	NA NA	NA NA	NA NA	NA NA	NA	NA
	NRC - 2-wire VG Interoffice Channel - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA NA	NA NA	\$18.94	NA	NA	NA	NA NA	NA	NA
+	NRC - 2-wire VG Interoffice Channel - Incremental ChargeManual Svc Order - Disco	SOMAN	NA NA	NA	NA NA	NA	NA	NA	NA	NA	NA
	NRC - 2-wire VG Interoffice Channel - Incremental ChargeManual Svc Order - Disco	SOMAN	NA NA	NA NA	NA NA	NA NA	NA	NA NA	NA NA	NA	NA NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	00101111	1471	1471	1471	1471	10/	1471	100	1471	10.0
++	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
	NRC-2/4-WIRE COMBINATION - Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
++	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
++	NRC- 2/4-WIRE COMBINATION - "Switch As is" Conversion Charge - Disconnect - Add"l	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
+	NRC-2/4-WIRE COMBINATION - SWICH AS IS CONVEISION Charge - Disconnect - Add 1	UNCCC	ψ0.00	ψ10.77	Ψ12.01	ψ0.00	ψ0.00	ψ0.00	ψ0.00	\$0.00	ψ0.00
	Interoffice Channel - Dedicated - 4-wire VG										
	Interoffice Channel - Dedicated 4-wire VG - per mile per month	1L5XX	NA	\$0.0100	NA	NA	NA	NA	NA	NA	NA
	Interoffice Channel - Dedicated 4-wire VG - Facility Termination per month	U1TV4	NA NA	\$23.82	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - Ordinarily Combined in GA (Note 5)	01174	INA	\$23.02	INA	INA	INA	INA	INA	INA	INA
++	NRC - 4-wire VG Interoffice Channel - Facility Termination - 1st	U1TV4	NA	NA	NA	NA	NA	NA	NA	NA	NA
	NRC - 4-wire VG Interoffice Channel - Facility Termination - Add'l	U1TV4	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - Electronic Svc Order, per LSR	SOMEC	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - 4-wire VG Interoffice Channel - Incremental ChargeManual Svc Order - 1st	SOMAN	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	NRC - 4-wire VG Interoffice Channel - Incremental ChargeManual Svc Order - 1st  NRC - 4-wire VG Interoffice Channel - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - 4-wire VG Interoffice Channel - Incremental ChargeManual Svc Order - Add I	SOMAN	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	NRC - 4-wire VG Interoffice Channel - Incremental ChargeManual Svc Order - Disco	SOMAN	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	SOIVIAIN	INA	INA	INA	INA	INA	INA	INA	INA	INA
+	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
+	- ·	UNCCC		*							\$54.13 \$32.17
+	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add'I		\$32.11	\$32.75 \$16.77	\$26.99 \$12.61	\$32.16 \$0.00	\$32.24	\$32.16	\$32.10 \$0.00	\$32.25 \$0.00	\$32.17
+	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00 \$0.00	\$16.77 \$16.77	\$12.61 \$12.61	\$0.00	\$0.00 \$0.00	\$0.00 \$0.00	\$0.00	\$0.00	\$0.00
+	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'l	UNCCC	φυ.υυ	φ10.//	φ1Z.01	φυ.υυ	φυ.υυ	φυ.υυ	φυ.υυ	φυ.υυ	φυ.υυ
+	Interesting Channel Dedicated DCO ECI-time		-						<del>                                     </del>		
+	Interoffice Channel - Dedicated - DS0 - 56kbps	41.53/3/	#0.000C	<b>#0.0400</b>	#0.000C	#0.0004	#0.000 <i>t</i>	#0.000C	#0.000C	<b>#0.0070</b>	<b>₽0.470</b> 0
+	Interoffice Channel - Dedicated - DS0 - 56kbps - per mile per month	1L5XX	\$0.0339	\$0.0100	\$0.0222	\$0.0301	\$0.0384	\$0.0323	\$0.0282	\$0.0373	\$0.1730
+	Interoffice Channel - Dedicated - DS0 - 56 kbps - Facility Termination per month	U1TD5	\$17.81	\$19.46	\$16.45	\$26.95	\$18.37	\$20.64	\$17.40	\$20.71	\$17.74
44	NRC - Ordinarily Combined in GA (Note 5)										
$\bot \bot$	NRC - 4-wire 56kbps Interoffice Channel - Facility Termination - 1st	U1TD5	NA	NA	\$79.61	NA	NA	NA	NA	NA	NA
	NRC - 4-wire 56 kbps Interoffice Channel - Facility Termination - Add'l	U1TD5	NA	NA	\$36.08	NA	NA	NA	NA	NA	NA

	Now EEL action on the course of the building LINE action dealers and the course			1	ı		1				
	New EEL rates are the sum of the individual UNE network elements (interoffice										
$\vdash$	transport and loop [channelization if applicable].										
	NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
	NRC - 4-wire 56 kbps Interoffice Channel - Incremental ChargeManual Svc Order - 1	SOMAN	NA	NA	\$18.94	NA	NA	NA	NA	NA	NA
	NRC - 4-wire 56 kbps Interoffice Channel - Incremental ChargeManual Svc Order - A	SOMAN	NA	NA	\$18.94	NA	NA	NA	NA	NA	NA
	NRC - 4-wire 56 kbps Interoffice Channel - Incremental ChargeManual Svc Order - I	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
	NRC - 4-wire 56 kbps Interoffice Channel - Incremental ChargeManual Svc Order - I	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)										
	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'I	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Interoffice Channel - Dedicated - DS0 - 64kbps										
	Interoffice Channel - Dedicated - DS0 - 64kbps - per mile per month	1L5XX	\$0.0339	\$0.0100	\$0.0222	\$0.0301	\$0.0384	\$0.0323	\$0.0282	\$0.0373	\$0.1730
	Interoffice Channel - Dedicated - DS0 - 64 kbps - Facility Termination per month	U1TD6	\$17.81	\$19.46	\$16.45	\$26.95	\$18.37	\$20.64	\$17.40	\$20.71	\$17.74
	NRC - Ordinarily Combined in GA (Note 5)										
	NRC - 4-wire 64kbps Interoffice Channel - Facility Termination - 1st	U1TD6	NA	NA	\$79.61	NA	NA	NA	NA	NA	NA
	NRC - 4-wire 64 kbps Interoffice Channel - Facility Termination - Add'l	U1TD6	NA	NA	\$36.08	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
	NRC - 4-wire 64 kbps Interoffice Channel - Incremental ChargeManual Svc Order - 1	SOMAN	NA	NA	\$18.94	NA	NA NA	NA	NA	NA	NA
	NRC - 4-wire 64 kbps Interoffice Channel - Incremental ChargeManual Svc Order - A	SOMAN	NA	NA NA	\$18.94	NA	NA NA	NA	NA NA	NA	NA NA
	NRC - 4-wire 64 kbps Interoffice Channel - Incremental ChargeManual Svc Order - I	SOMAN	NA	NA	NA NA	NA	NA	NA	NA.	NA	NA
	NRC - 4-wire 64 kbps Interoffice Channel - Incremental ChargeManual Svc Order - I	SOMAN	NA	NA NA	NA NA	NA NA	NA	NA	NA NA	NA	NA NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	OOMAN	19/3	14/3	13/3	14/3	19/3	14/3	14/3	19/3	14/3
	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
+++		UNCCC		\$32.75	\$26.99	\$32.16	\$32.24		\$34.00	\$32.25	\$32.17
++	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add'I		\$32.11					\$32.16			
+++	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
++	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'l	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
$\vdash$	Interesting Channel Dedicated DC4										
$\vdash$	Interoffice Channel - Dedicated - DS1	41.5107	<b>A</b> 0.0000	<b>#</b> 0.0040	A0 4500	<b>#</b> 0.4500	A0 7004	<b>A</b> 0.0500	<b>#</b> 0.0700	<b>A</b> 0.7500	A0.0505
+++	Interoffice Channel - Dedicated - DS1 - per mile per month	1L5XX	\$0.6920	\$0.6013	\$0.4523	\$0.4500	\$0.7831	\$0.6598	\$0.0783	\$0.7598	\$0.3525
+++	Interoffice Channel - Dedicated - DS1 - Facility Termination per month	U1TF1	\$79.69	\$99.79	\$78.47	\$55.05	\$93.40	\$74.40	\$71.29	\$94.98	\$75.83
	NRC - Ordinarily Combined in GA (Note 5)										
	NRC - DS1 Interoffice Channel - Facility Termination - 1st	U1TF1	NA	NA	\$169.57	NA	NA	NA	NA	NA	NA
	NRC - DS1 Interoffice Channel - Facility Termination - Add'l	U1TF1	NA	NA	\$112.77	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
	NRC - DS1 Interoffice Channel - Incremental ChargeManual Svc Order - 1st	SOMAN	NA	NA	\$23.98	NA	NA	NA	NA	NA	NA
	NRC - DS1 Interoffice Channel - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA	NA	\$17.77	NA	NA	NA	NA	NA	NA
$\Box$	NRC - DS1 Interoffice Channel - Incremental ChargeManual Svc Order - Disconnect	SOMAN	NA	NA	\$15.13	NA	NA	NA	NA	NA	NA
	NRC - DS1 Interoffice Channel - Incremental ChargeManual Svc Order - Disconnec	SOMAN	NA	NA	\$7.02	NA	NA	NA	NA	NA	NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)										
	NRC-DS1 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
	NRC-DS1 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
	NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'l	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	·										
	Interoffice Channel - Dedicated - DS3 - per mile per month										
	Interoffice Channel - Dedicated - DS3 - per mile per month	1L5XX	\$4.98	\$4.25	\$6.46	\$12.06	\$16.15	\$13.48	\$12.98	\$19.14	\$6.88
	Interoffice Channel - Dedicated - DS3 - Facility Termination per month	U1TF3	\$898.15	\$1,130	\$717.60	\$1,112.02	\$1,131.09	\$686.84	\$720.38	\$904.49	\$840.61
	NRC - Ordinarily Combined in GA (Note 5)	<u> </u>	<b>\$0000</b>	<b>\$</b> .,	Ţ	Ţ., <u>Z.</u>	Ţ.,.oo	<b>\$000.01</b>	Ţ. <u>2</u> 0.00	<b>\$000</b>	\$0.0.0.
	NRC - DS3 Interoffice Channel - Facility Termination - 1st	U1TF3	NA	NA	\$578.97	NA	NA	NA	NA	NA	NA
+++	NRC - DS3 Interoffice Channel - Facility Termination - 1st	U1TF3	NA NA	NA NA	\$312.17	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+++	NRC - Electronic Svc Order, per LSR	SOMEC	NA NA	NA NA	\$3.50	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
<del>                                      </del>	NRC - DS3 Interoffice Channel - Incremental ChargeManual Svc Order - 1st	SOMAN	NA NA	NA NA	\$51.27	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+++	NRC - DS3 Interoffice Channel - Incremental ChargeManual Svc Order - 1st	SOMAN	NA NA	NA NA	\$38.87	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+++					*						
1 1 1	NRC - DS3 Interoffice Channel - Incremental ChargeManual Svc Order - Disconnect	SOMAN	NA	NA	\$30.42	NA	NA	NA	NA	NA	NA

	New EEL rates are the sum of the individual UNE network elements (interoffice										
	transport and loop [channelization if applicable].										
	NRC - DS3 Interoffice Channel - Incremental ChargeManual Svc Order - Disconnect	SOMAN	NA	NA	\$18.76	NA	NA	NA	NA	NA	NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)										
	NRC-DS3 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
	NRC-DS3 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
	NRC- DS3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC- DS3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'I	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Interoffice Channel - Dedicated - STS-1										
	Interoffice Channel - Dedicated - STS-1 - per mile per month	1L5XX	\$4.98	\$9.32	\$2.72	\$12.62	\$14.04	\$15.02	\$12.98	\$8.13	\$5.89
	Interoffice Channel - Dedicated - STS-1 - Facility Termination per month	U1TFS	\$898.15	\$569.67	\$788.00	\$1,204	\$1,101	\$744.38	\$720.38	\$967.70	\$760.20
	NRC - Ordinarily Combined in GA (Note 5)										
	NRC - STS-1 Interoffice Channel - Facility Termination - 1st	U1TFS	NA	NA	\$587.08	NA	NA	NA	NA	NA	NA
	NRC - STS-1 Interoffice Channel - Facility Termination - Add'l	U1TFS	NA	NA	\$238.28	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
	NRC - STS-1 Interoffice Channel - Incremental ChargeManual Svc Order - 1st	SOMAN	NA	NA	\$61.19	NA	NA	NA	NA	NA	NA
+	NRC - STS-1 Interoffice Channel - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA	NA	\$61.19	NA	NA	NA	NA	NA	NA
	NRC - STS-1 Interoffice Channel - Incremental ChargeManual Svc Order - Disconne	SOMAN	NA	NA	\$3.17	NA	NA	NA	NA	NA	NA
	NRC - STS-1 Interoffice Channel - Incremental ChargeManual Svc Order - Disconne	SOMAN	NA	NA	\$3.17	NA	NA	NA	NA	NA	NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	1111000	<b>0</b> 5400	A00 75	050.40	<b>\$54.00</b>	05100	<b>A5400</b>	05400	<b>\$54.00</b>	05440
-++	NRC-STS-1 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
-++	NRC-STS-1 COMBINATION - "Switch As Is" Conversion Charge - Add'I	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
-++	NRC- STS-1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC- STS-1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add"	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Interoffice Channel - OC3										
	Interoffice Channel - OC3 - per Mile	1L5XX	\$7.35	\$8.38	\$4.37	\$27.97	\$23.89	\$18.35	\$14.10	\$9.75	\$13.45
	Interoffice Channel - OC3 - per Facility Termination	TBA	\$2,475	\$3,043	\$2,187.00	\$3,390	\$2,990	\$1,892.00	\$2,071	\$2,802	\$2,124
	NRC - Ordinarily Combined in GA (Note 5)										
	NRC - OC3 - Facility Termination - 1st	TBA	NA	NA	\$947.69	NA	NA	NA	NA	NA	NA
	NRC - OC3 - Facility Termination - Add'l	TBA	NA	NA	\$413.00	NA	NA	NA	NA	NA	NA
	NRC - OC3 - Facility Termination - Disconnect - 1st	TBA	NA	NA	\$122.31	NA	NA	NA	NA	NA	NA
	NRC - OC3 - Facility Termination - Disconnect - Add'l	TBA	NA	NA	\$119.14	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
	NRC - OC3 - Incremental ChargeManual Svc Order - 1st	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
	NRC - OC3 - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
	NRC - OC3 -Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC - OC3 -Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add'l	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
-++	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	LINIOOO	<b>#</b> 54.00	<b>#00.75</b>	Ø50.40	ΦE 4.00	<b>#</b> E4.00	<b>#</b> 54.00	<b>#</b> 54.00	<b>#</b> F4.00	05440
	NRC-OC-3 COMBINATION - "Switch As Is" Conversion Charge - 1st NRC-OC-3 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$54.03 \$32.11	\$32.75 \$32.75	\$58.43 \$26.99	\$54.09 \$32.16	\$54.23 \$32.24	\$54.09 \$32.16	\$54.00 \$32.10	\$54.26 \$32.25	\$54.13 \$32.17
++	NRC- OC-3 COMBINATION - Switch As Is Conversion Charge - Add 1  NRC- OC-3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$32.75 \$16.77	\$26.99	\$0.00	\$0.00	\$32.16	\$0.00	\$32.25	\$32.17
++	NRC- OC-3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'I	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
+	NAC- 00-3 COMBINATION - SWIGH AS IS CONVERSION Charge - Disconnect - Add I	UNCCC	φυ.υυ	φιυ.//	φι∠.01	φυ.υυ	φυ.υυ	φυ.υυ	φυ.υυ	φυ.υυ	φυ.υυ
	Interoffice Channel - OC12										
	Interoffice Channel - OC12 - per Mile	1L5XX	\$19.26	\$26.91	\$15.05	\$84.88	\$74.44	\$60.42	\$30.38	\$32.52	\$49.80
	Interoffice Channel - OC12 - per Facility Termination	TBA	\$9,763	\$11,685	\$8,202.00	\$12,344	\$11,517	\$7,182.00	\$2,122	\$11,132	\$8,015
	NRC - Ordinarily Combined in GA (Note 5)										
	NRC - OC12 - Facility Termination - 1st	TBA	NA	NA	\$1,034.00	NA	NA	NA	NA	NA	NA
	NRC - OC12 - Facility Termination - Add'l	TBA	NA	NA	\$317.38	NA	NA	NA	NA	NA	NA
	NRC - OC12 - Facility Termination - Disconnect - 1st	TBA	NA	NA	\$122.31	NA	NA	NA	NA	NA	NA
$\perp \! \! \perp$	NRC - OC12 - Facility Termination - Disconnect - Add'l	TBA	NA	NA	\$119.14	NA	NA	NA	NA	NA	NA
$\perp$	NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
+	NRC -OC12 - Incremental Charge - Manual Svc Order - 1st	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
+	NRC - OC12 - Incremental Charge - Manual Svc Order - Add'l	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
1 1	NRC - OC12 - Incremental Cost-Manual Svc. Order vs. Elect-Disconnect-1st	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA

	New EEL rates are the sum of the individual UNE network elements (interoffice transport and loop [channelization if applicable].										
	NRC - OC12 - Incremental Cost-Manual Svc. Order vs. Elect-Disconnect-Add'l	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
+	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	CONSTIT	1471	147.	Ψ10.00	1471	1471	107	1471	107	1471
+	NRC-OC-12 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
++	NRC-OC-12 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
++	NRC- OC-12 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
+	NRC- OC-12 COMBINATION - Switch As Is Conversion Charge - Disconnect - Add'	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NC-00-12 COMBINATION - SWIICH AS IS CONVERSION CHARGE - DISCONNECT - Add	UNCCC	φ0.00	\$10.77	\$12.01	\$0.00	φυ.υυ	φυ.υυ	\$0.00	φυ.υυ	φυ.υυ
	Interoffice Channel - OC48										
+	Interoffice Channel - OC48 - per Mile	1L5XX	\$30.65	\$34.66	\$25.70	\$138.02	\$128.59	\$102.43	\$120.02	\$45.92	\$106.55
++	Interoffice Channel - OC48 - per Facility Termination	TBA	\$11,691	\$12,554	\$11,134.00	\$16,017	\$14,950	\$11,480.00	\$1,677	\$967.58	\$11,632
++	Interoffice Channel - OC12 interface on OC48 Facility	TBA	\$1.424	\$1,208	\$1,137.00	\$1.497	\$1,451	\$1,351.00	\$582.66	\$1,561	\$1,170
+	NRC - Ordinarily Combined in GA (Note 5)	IDA	Ψ1,424	Ψ1,200	ψ1,137.00	Ψ1,431	Ψ1,401	ψ1,331.00	ψ302.00	ψ1,501	ψ1,170
++	NRC - OC48 - Facility Termination - 1st	TBA	NA	NA	\$1,034.00	NA	NA	NA	NA	NA	NA
++	NRC - OC46 - Facility Termination - Add'l	TBA	NA NA	NA NA	\$317.38	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++			NA NA	NA NA		NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - OC48 - Interface OC12 on OC48 - 1st	TBA			\$539.36						
++	NRC - OC48 - Interface OC12 on OC48 - Add'l	TBA	NA	NA	\$317.38	NA	NA	NA	NA	NA	NA
+	NRC - OC48 - Facility Termination - Disconnect - 1st	TBA	NA	NA	\$122.31	NA NA	NA	NA	NA	NA	NA
++	NRC - OC48 - Facility Termination - Disconnect - Add'l	TBA	NA	NA	\$119.14	NA	NA	NA	NA	NA	NA
+	NRC - OC48- Interface OC12 on OC48 - Disconnect - 1st	TBA	NA	NA	\$122.31	NA NA	NA	NA	NA	NA	NA
	NRC - OC48 - Interface OC12 on OC48 - Disconnect - Add'l	TBA	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
$\bot \bot$	NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
	NRC - OC48 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-1st	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
$\bot \bot$	NRC - OC48 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-Add'l	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
	NRC - OC48 - Interface - Manual Svc Order vs Electronic-Disconnect-1st	SOMAN	NA	NA	\$37.55	NA	NA	NA	NA	NA	NA
	NRC - OC48 - Interface - Manual Svc Order vs Electronic-Disconnect-Add'l	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC - OC-48 - Incremental ChargeManual Svc Order-1st	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC - OC-48 - Incremental ChargeManual Svc Order-Add'l	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC - OC48 - Interface OC12 on OC48 - Incremental ChargeManual Svc Order-1st	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC - OC48 - Interface OC12 on OC48 - Incremental ChargeManual Svc Order-Add	SOMAN	NA	NA	\$18.03	NA	NA	NA	NA	NA	NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)										
	NRC-OC-48 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
	NRC-OC-48 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
	NRC- OC-48 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC- OC-48 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Channelization:										
	DS3 Channelization										
	DS3 Channelized System per month	MQ3	\$225.36	\$222.61	\$182.04	\$236.32	\$245.84	\$229.30	\$226.81	\$200.01	\$222.98
	DS3 Interface per month (DS1 COCI)	UC1D1	\$17.22	\$14.51	\$11.02	\$8.52	\$7.55	\$5.58	\$4.61	\$11.99	\$3.91
	NRC - Ordinarily Combined in GA (Note 5)										
	NRC - DS3 Channelization - 1st	MQ3	NA	NA	\$316.28	NA	NA	NA	NA	NA	NA
11	NRC - DS3 Channelization - Add'l	MQ3	NA	NA	\$171.72	NA	NA	NA	NA	NA	NA
TT	NRC - Channel Activation - 1st	UC1D1	NA	NA	\$12.02	NA	NA	NA	NA	NA	NA
tt	NRC - Channel Activation - Addl	UC1D1	NA	NA	\$8.66	NA	NA	NA	NA	NA	NA
ff	NRC - Electronic Svc Order, per LSR	SOMEC	NA.	NA.	\$3.50	NA NA	NA NA	NA NA	NA NA	NA.	NA NA
$\dagger \dagger$	NRC - DS3 Channelization - Incremental ChargeManual Svc Order - 1st	SOMAN	NA NA	NA NA	\$50.72	NA	NA	NA	NA NA	NA NA	NA
+ +	NRC - DS3 Channelization - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA NA	NA	\$38.45	NA NA	NA	NA NA	NA NA	NA NA	NA NA
++	NRC - DS3 Channelization - Incremental ChargeManual Svc Order - Add1	SOMAN	NA NA	NA NA	\$30.09	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	NRC - DS3 Channelization - Incremental ChargeManual Svc Order - Disconnect - A		NA NA	NA NA	\$18.56	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	JOINAIN	19/3	INA	ψ10.50	11/1	19/3	19/3	11/3	INA	13/3
++	NRC-DS3/STS-1 COMBINATION - "Switch As Is" Conversion Charge (Note of	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
++	NRC-DS3/STS-1 COMBINATION - Switch As Is Conversion Charge - 1st  NRC-DS3/STS-1 COMBINATION - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75		\$32.16	\$32.24		\$32.10	\$32.25	\$32.17
++	NRC-DS3/STS-1 COMBINATION - "Switch As Is" Conversion Charge - Add"  NRC- DS3/STS-1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect -	UNCCC	\$32.11	\$32.75 \$16.77	\$26.99 \$12.61	\$32.16 \$0.00		\$32.16	\$32.10 \$0.00	\$32.25	\$32.17
+							\$0.00	\$0.00	*		
++	NRC- DS3/STS-1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect -	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
1 1											

	New EEL rates are the sum of the individual UNE network elements (interoffice										
$\vdash$	transport and loop [channelization if applicable].  DS1 Channelization										<b> </b>
-		MQ1	<b>#</b> 400.00	<b>045474</b>	<b>#</b> 400.00	<b>#</b> 000 04	#000 07	£440.07	<b>#477.70</b>	<b>64.47.54</b>	<b>#</b> 405.04
H	DS1 Channelized System per month OCU-DP(data) interface card per month (2.4-64kbs)	1D1DD	\$136.82 \$1.66	\$154.74 \$2.22	\$126.22 \$1.86	\$200.01 \$2.94	\$209.87 \$3.12	\$146.87 \$2.86	\$177.72 \$2.88	\$147.51 \$2.34	\$165.21 \$2.46
H	VG interface card per month (2.4-64kbs)	1D1VG	\$0.8586	\$2.22	\$1.86	\$2.94 \$1.40	\$3.12	\$2.86 \$1.45	\$2.88	\$2.34 \$1.47	\$2.46
H	2-wire ISDN(BRITE card) per month	UC1CA	\$3.41	\$3.86	\$3.71	\$4.04	\$4.18	\$3.88	\$3.76	\$4.21	\$3.33
H	NRC - Ordinarily Combined in GA (Note 5)	UCTCA	φ3.41	φ3.00	φ3.7 Ι	φ4.04	φ4.10	φ3.00	φ3.70	Ψ4.∠1	φ3.33
	NRC - DS1 Channelization - 1st	MQ1	NA	NA	\$206.09	NA	NA	NA	NA	NA	NA
$\vdash$	NRC - DS1 Channelization - 1st	MQ1	NA NA	NA NA	\$137.06	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - Channel Activation VG - 1st	1D1VG	NA NA	NA NA	\$12.02	NA	NA NA	NA	NA NA	NA NA	NA.
	NRC - Channel Activation VG - Add'l	1D1VG	NA	NA	\$8.66	NA	NA	NA	NA	NA NA	NA
	NRC - Channel Activation OCU-DP- 1st	1D1DD	NA	NA.	\$12.02	NA	NA	NA	NA.	NA.	NA
	NRC - Channel Activation OCU-DP- Add'l	1D1DD	NA	NA	\$8.66	NA	NA	NA	NA	NA	NA
	NRC - Channel Activation BRITE - 1st	UCICA	NA	NA	\$12.02	NA	NA	NA	NA	NA	NA
	NRC - Channel Activation BRITE - Add'I	UCICA	NA	NA	\$8.66	NA	NA	NA	NA	NA	NA
	NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
	NRC - DS1 Channelization - Incremental ChargeManual Svc Order - 1st	SOMAN	NA	NA	\$14.75	NA	NA	NA	NA	NA	NA
╧	NRC - DS1 Channelization - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA	NA	\$6.55	NA	NA	NA	NA	NA	NA
	NRC - DS1 Channelization - Incremental ChargeManual Svc Order - Disconnect - 1s	SOMAN	NA	NA	\$10.70	NA	NA	NA	NA	NA	NA
	NRC - DS1 Channelization - Incremental ChargeManual Svc Order - Disconnect - Ad	SOMAN	NA	NA	\$0.00	NA	NA	NA	NA	NA	NA
	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)										
	NRC-DS1 COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
	NRC-DS1 COMBINATION - "Switch As Is" Conversion Charge - Add'I	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
	NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'l	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
											<b>!</b>
	Access to DCS - Customer Reconfiguration (FlexServ)										<b>!</b>
	DS1 DSC Termination with DS0 Switching		TBD	\$28.72	\$22.86	TBD	TBD	TBD	TBD	TBD	TBD
	DS1 DSC Termination with DS1 Switching		TBD	\$12.23	\$8.64	TBD	TBD	TBD	TBD	TBD	TBD
_	DS3 DSC Termination with DS1 Switching		TBD	\$154.31	\$151.85	TBD	TBD	TBD	TBD	TBD	TBD
_	NRC - Ordinarily Combined in GA:		TDD	00.07	00.04	TD D		T00	TDD	<b>TDD</b>	
_	NRC - Customer Configuration Establishment		TBD TBD	\$2.97 \$3.44	\$2.91	TBD TBD	TBD TBD	TBD TBD	TBD TBD	TBD TBD	TBD TBD
	NRC - Customer Configuration Establishment - Disconnect				\$3.36		TBD	TBD	TBD	TBD	TBD
-	NRC- DS1 DSC Termination with DS0 Switching - 1st NRC- DS1 DSC Termination with DS0 Switching - Add'l		TBD TBD	\$51.50 \$39.64	\$32.07 \$31.49	TBD TBD	TBD	TBD	TBD	TBD	TBD
-	NRC- DS1 DSC Termination with DS0 Switching - Add1  NRC- DS1 DSC Termination with DS0 Switching - Disconnect - 1st		TBD	\$39.64	\$20.16	TBD	TBD	TBD	TBD	TBD	TBD
-	NRC- DS1 DSC Termination with DS0 Switching - Disconnect - 1st  NRC- DS1 DSC Termination with DS0 Switching - Disconnect - Add'l		TBD	\$24.98	\$20.16	TBD	TBD	TBD	TBD	TBD	TBD
-	NRC- DS1 DSC Termination with DS0 Switching - Dscorinect - Add1		TBD	\$37.23	\$18.07	TBD	TBD	TBD	TBD	TBD	TBD
-	NRC- DS1 DSC Termination with NRC- DS1 Switching - 1st		TBD	\$25.36	\$17.49	TBD	TBD	TBD	TBD	TBD	TBD
-	NRC- DS1 DSC Termination with NRC- DS1 Switching - Disconnect - 1st		TBD	\$23.30	\$17.49	TBD	TBD	TBD	TBD	TBD	TBD
+	NRC- DS1 DSC Termination with NRC- DS1 Switching - Disconnect - Add'l		TBD	\$16.73	\$12.10	TBD	TBD	TBD	TBD	TBD	TBD
-	NRC- DS3 DSC Termination with DS1 Switching - 1st		TBD	\$51.50	\$32.07	TBD	TBD	TBD	TBD	TBD	TBD
+	NRC- DS3 DSC Termination with DS1 Switching - Add'l		TBD	\$39.64	\$31.49	TBD	TBD	TBD	TBD	TBD	TBD
	NRC- DS3 DSC Termination with DS1 Switching - Disconnect - 1st		TBD	\$31.06	\$20.16	TBD	TBD	TBD	TBD	TBD	TBD
	NRC- DS3 DSC Termination with DS1 Switching - Disconnect - Add'l		TBD	\$24.98	\$20.16	TBD	TBD	TBD	TBD	TBD	TBD
$\vdash$	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)			Ţ	<del>+</del> =30					1	
$\vdash$	NRC-DCS COMBINATION - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
-	NRC-DCS COMBINATION - "Switch As Is" Conversion Charge - Add'I	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
	NRC- DCS COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
+	NRC- DCS COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'l	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
1											
1	Node (Synchronet)										ſ
+	Node per month	UNCNT	\$15.77	\$16.35	\$13.98	\$18.11	\$15.43	\$16.15	\$16.00	\$14.55	\$17.11
	NRC - Ordinarily Combined in GA:			1					1		
	NRC - Node - 1st	UNCNT	NA	NA	\$47.19	NA	NA	NA	NA	NA	NA
-	NRC - Node - Add'l	UNCNT	NA	NA	\$4.47	NA	NA	NA	NA	NA	NA
	[1410 14040 1444]	CITOITI	14/-1	14/3	ψ <b>τ.</b> τι	14/-1	14/-1	1.4/-7	14/1	14/-1	1177

т т	New EEL rates are the sum of the individual UNE network elements (interoffice		ı	1	1			ı	1		
	transport and loop [channelization if applicable].										
+	NRC - Node - Disconnect - 1st	UNCNT	NA	NA	\$13.81	NA	NA	NA	NA	NA	NA
+	NR C - Node - Disconnect - Add'l	SOMAN	NA NA	NA NA	\$3.95	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - Node - Disconnect - Add i  NRC - Node - Incremental Charge - Manual Service Order - 1st	SOMAN	NA NA	NA NA	\$21.73	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - Node - Incremental Charge - Manual Service Order - 1st  NRC - Node - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA NA	NA NA	Φ21.73 NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - Node - Incremental Charge - Manual Service Order - Add 1	SOMAN	NA NA	NA NA	\$3.87	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - Node - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	SUMAN	NA	NA	NA	INA	INA	NA	INA	INA	INA
+	NRC-Node - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$32.75	\$58.43	\$54.09	\$54.23	\$54.09	\$54.00	\$54.26	\$54.13
+	NRC-Node - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$32.75	\$26.99	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
++	NRC-Node - "Switch As Is" Conversion Charge - Disconnect - 1st	UNCCC	\$0.00	\$32.75 \$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
++	NRC- Node - "Switch As Is" Conversion Charge - Disconnect - Add"	UNCCC	\$0.00	\$16.77	\$12.61	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
+	NRC- Node - Switch As is Conversion Charge - Disconnect - Add i	UNCCC	\$0.00	\$10.77	\$12.01	\$0.00	φυ.υυ	\$0.00	\$0.00	φυ.υυ	\$0.00
+	Optional Features & Functions:										
+	NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel	CCOEF	\$184.85	\$184.92	\$184.62	\$184.91	\$184.65	\$184.60	\$184.76	\$185.26	\$185.16
+	NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel	CCOEF	\$23.81	\$23.82	\$23.78	\$23.82	\$23.70	\$23.78	\$23.60	\$23.86	\$23.85
++	NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel	CCOEF	\$1.99	\$23.02	\$23.76	\$1.99	\$1.97	\$1.96	\$1.99	\$1.99	\$2.03
+	NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel	CCOEF	\$0.77	\$0.80	\$0.79	\$0.78	\$0.77	\$0.76	\$0.78	\$0.78	\$0.79
+	NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - Manual Service Q	SOMAN	\$29.23	\$21.73	\$29.33	\$29.24	\$29.20	\$29.33	\$29.33	\$29.33	\$29.33
+	NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - Manual Service C	SOMAN	\$3.93	\$3.87	\$3.93	\$3.94	\$3.92	\$3.93	\$3.93	\$3.93	\$3.93
+	NAC - Clear Charmer Capability (B623/E3F) Option - Subsequent - Manual Service C	SOMAN	φ3.93	φ3.07	φ3.93	φ3.94	ψ3.92	φ3.93	φ3.93	φ3.93	φ3.93
+	NRC - Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Channel -	CCOSF	\$184.85	\$184.92	\$184.62	\$184.91	\$184.65	\$184.60	\$184.76	\$185.26	\$185.16
$\pm$	NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel	CCOSF	\$23.81	\$23.82	\$23.78	\$23.82	\$23.70	\$23.78	\$23.60	\$23.86	\$23.85
+	NRC - Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Channel -	CCOSF	\$1.99	\$2.07	\$2.03	\$1.99	\$1.97	\$1.96	\$1.99	\$1.99	\$2.03
	NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel	CCOSF	\$0.77	\$0.80	\$0.79	\$0.78	\$0.77	\$0.76	\$0.78	\$0.78	\$0.79
++	NRC - Clear Channel Capability (B8ZS/SF) Option - Subsequent - Manual Service Or		\$29.23	\$21.73	\$29.33	\$29.24	\$29.20	\$29.33	\$29.33	\$29.33	\$29.33
	NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - Manual Service C	SOMAN	\$3.93	\$3.87	\$3.93	\$3.94	\$3.92	\$3.93	\$3.93	\$3.93	\$3.93
+	1410 Ocal Onamici Capability (B020/201) Option Cabacquent Mandar Cervice C	OOMAN	ψ0.55	ψ5.07	ψ5.55	ψ5.54	ψ5.52	ψ0.00	ψ0.00	ψ5.55	ψ0.00
	otes :										
++:	Interim rates subject to true-up.										
##	Deaveraged Rates will be effective May 1, 2000										
	New EELs will only be available in the State of Georgia and in density Zone 1 of the fo	llowing MSAs	in the BellSou	th Region:							
ĦĖ	Florida - Miami, Orlando, Ft. Lauderdale		20000	l							
$\pm$	Louisiana - New Orleans										
+	N. Carolina - Greensboro, Charlotte										
+	Tennessee - Nashville										
11:	Unapproved rates are subject to true up.										
_	4 Add together the recurring rates of all the applicable network elements in order to obta	in total month	lv recurring ra	te.							
+	* Examples:		, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,								
+	- 2-wire VG Loop + Voice Grade Interface Card + DS1 Channelization System + DS	I Interoffice C	hannel								
	- DS1 Loop + DS1 Interface Card + DS3 Channelization System + DS3 Interoffice C										
+	- DS3 Local Channel + DS3 Interoffice Channel + DS3 Channelization System + DS		ard								
11:	The Ordinarily Combined in GA NRC applies to new combinations within the State of C										
	The "Switch As Is" NRC is a conversion charge. One SAI charge is applicable per circ								1		

DE000:	101		T			101	T				
DESCRIPTI		USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	Support Systems										<b> </b>
Reco	overy of incremental OSS costs, per CLP, per month		NA	NA	NA	NA	NA	NA	\$305.00	NA	NA
	OSS OLEC Daily Usage File: Recording, Per Message		\$0.0002	\$0.008	\$0.0001275	\$0.0008611	\$0.00019	\$0.0001179	\$0.0003	\$0.0002862	\$0.008
	OSS OLEC Daily Usage File: Message Processing, Per Message		\$0.0033	\$0.004	\$0.0082548	\$0.0032357	\$0.0024	\$0.0032089	\$0.0032	\$0.0032344	\$0.004
	OSS OLEC Daily Usage File: Message Distribution, Per Magnetic Tape		\$55.19	\$54.95	\$28.85	\$55.68	\$47.3000	\$54.62	\$54.61	\$54.72	\$54.95
	OSS OLEC Daily Usage File: Data Transmission (CONNECT:DIRECT), Per		\$0.00004	\$0.001	\$0.0000434	\$0.0000365	\$0.0000300	\$0.0000354	\$0.00004	\$0.0000357	\$0.001
	ily Usage File (ADUF)										
	ADUF, Message Processing, per message		\$0.004	\$0.004	\$0.0136327	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004
	ADUF, Message Distribution, per Magnetice Tape provisioned		NA	NA	NA	NA	NA	NA	NA	NA	NA
	ADUF, Data Transmision (CONNECT:DIRECT), per message		\$0.001	\$0.001	\$0.0000434	\$0.001	\$0.0000305	\$0.001	\$0.001	\$0.001	\$0.001
	Optional Daily Usage File (EODUF)										
	anced Optional Daily Usage File: Message Processing , Per Message		\$0.004	\$0.004	\$0.0034555	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004
	anced Optional Daily Usage File: Message Processing, per magnetic tape		NA	NA	NA	NA	NA	NA	NA	NA	NA
Enha	anced Optional Daily Usage File: Data Transmision (CONNECT:DIRECT), per		NA	NA	NA	NA	NA	NA	NA	NA	NA
											i
	II Free Dialing Ten Digit Screening Service (Note 1)										
8XX Access	s Ten Digit Screening (all types), per call (Note 2)	N/A	\$0.0005	NA	\$0.0004868	NA	\$0.0005305	\$0.0005321	\$0.00050	\$0.0005227	NA
	s Ten Digit Screening Svc. W/8XX No. Delivery										
per q	query	N/A	NA	\$0.004	NA	\$0.0010	NA	NA	\$0.00365	NA	\$0.004
for 8	XX Numbers, with Optional Complex Features, per query	N/A	NA	\$0.0045	NA	\$0.0011	NA	NA	\$0.00431	NA	\$0.0045
8XX Access	s Ten Digit Screening Svc. W/POTS No. Delivery										ĺ .
per q	query	N/A	NA	\$0.004	NA	\$0.0010	NA	NA	\$0.00383	NA	\$0.004
with 0	Optional Complex Features, per query	N/A	NA	\$0.0045	NA	\$0.0011	NA	NA	\$0.00431	NA	\$0.0045
8XX Access	s Ten Digit Screening Svc. W/800 No. Delivery										
per m	nessage	N/A	NA	NA	NA	NA	NA	NA	NA	NA	NA
for 8	XX Numbers, w/Optional Complex Features, per message	N/A	NA	NA	NA	NA	NA	NA	NA	NA	NA
	s Ten Digit Screening Svc. W/POTS No. Delivery	•									i
	nessage	N/A	NA	NA	NA	NA	NA	NA	NA	NA	NA
	Optional Complex Features, per message	N/A	NA	NA	NA	NA	NA	NA	NA	NA	NA
	n Charge per 8XX number reserved	.,,.,									
NRC		N8R1X	\$7.13	NA	\$6.57	\$10.05	\$6.29	\$8.46	\$7.05	\$6.38	\$30.00
	- Addl'I	N8R1X	\$0.97	NA	\$0.76	\$1.19	\$0.73	\$0.96	\$0.96	\$0.9583	\$0.50
	- Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$27.84	NA
	- Incremental Charge - Manual Service Order - Add'l	SOMAN	NA NA	NA	NA NA	NA.	NA	NA NA	NA NA	NA NA	NA
	Established w/o POTS (w/8XX No.) Translations	00									
NRC	,	N/A	\$15.88	NA	\$12.81	\$30.59	\$12.27	\$17.04	\$23.82	\$22.63	\$67.50
	ISI Addl'I	N/A	\$1.97	NA NA	\$1.45	\$3.22	\$1.39	\$17.04	\$23.02	\$2.73	\$1.50
	Addit	N/A N/A	\$10.04	NA NA	\$1.45 NA	φ3.22 NA	\$8.30	\$11.32	Φ2.73 NA	\$42.95	\$1.50 NA
	- Disconnect Charge - Add'l	N/A	\$0.97	NA NA	NA C10.04	NA NA	\$0.73	\$0.96	NA ©44.25	NA NA	NA NA
	- Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$41.35	NA	NA
	- Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
	- Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.75	NA	NA	NA	\$11.40	\$16.05	NA	NA	NA
	Established with POTS Translations		<u> </u>		ļ		1	L		<u> </u>	<b>.</b>
NRC	* 1	N8FTX	\$15.88	NA	\$12.81	\$30.59	\$12.27	\$17.04	\$23.82	\$22.63	\$67.50
	- Addl'I	N8FTX	\$1.97	NA	\$1.45	\$3.22	\$1.39	\$1.93	\$2.73	\$2.73	\$1.50
NRC	- Disconnect Charge - 1st	N8FTX	\$10.04	NA	NA	NA	\$8.30	\$11.32	NA	\$42.95	NA
NRC	- Disconnect Charge - Add'l	N8FTX	\$0.97	NA	NA	NA	\$0.73	\$0.96	NA	NA	NA
	- Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$41.35	NA	NA
	- Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
	- Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.75	NA NA	NA NA	NA NA	\$11.40	\$16.05	NA	NA NA	NA NA
	d Area of Service per 8XX Number	JOIVIAIN	ψ17.73	INA	INA	INA	Ψ11.40	ψ10.03	INA	INA	11/7
NRC		N8FCX	\$5.69	NA	\$4.46	\$6.97	\$4.27	\$5.63	\$5.63	\$5.64	\$3.00
	15t Addi'l	N8FCX	\$2.85	NA NA	\$2.23	\$3.49	\$2.14	\$2.81	\$2.82	\$2.82	\$3.00
	Add I - Incremental Charge - Manual Service Order - 1st	SOMAN	\$2.65 NA	NA NA	Ψ2.23 NA	\$3.49 NA	Ψ2.14 NA	Φ2.01 NA	Ψ2.02 NA	Φ2.02 NA	\$1.50 NA
	ŭ	SOMAN	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
INRC	- Incremental Charge - Manual Service Order - Add'l	SOMAIN	INA	INA	INA	INA	INA	INA	INA	INA	INA

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
Multiple Inter LATA Carrier Routing per Carrier Requested per 8XX #										
NRC - 1st	N8FMX	\$6.66	NA	\$5.22	\$8.16	\$5.00	\$6.59	\$6.59	\$6.60	\$3.50
NRC - Addl'I	N8FMX	\$3.81	NA	\$2.99	\$4.67	\$2.86	\$3.77	\$3.77	\$3.78	\$2.00
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
Change Charge per request										
NRC - 1st	N8FAX	\$8.10	NA	\$7.33	\$11.24	\$7.01	\$9.42	\$8.01	\$7.34	\$48.50
NRC - Addi'l	N8FAX	\$0.97	NA	\$0.76	\$1.19	\$0.73	\$0.96	\$0.96	\$0.9583	\$0.50
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA NA	\$18.14	\$25.52	\$26.94	\$27.84	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
Call Handling and Destination Features  NRC - 1st	N8FDX	\$5.69	NA	\$4.72	\$6.97	\$4.27	\$5.63	\$5.63	\$5.64	\$3.00
NRC - Add'l	N8FDX N8FDX	\$5.69 NA	NA NA	\$4.72	\$6.97 \$6.97	\$4.27 \$4.27	\$5.63	\$5.63 NA	\$5.64	\$3.00
INIC - Add I	NOFDA	INA	INA	φ4.40	φ0.97	φ4.Z1	φ3.03	INA	φ5.04	φ3.00
LINE INFORMATION DATABASE ACCESS (LIDB)										
LIDB Common Transport per query	OQT	\$0.00004	\$0.0003	\$0.0000338	\$0.00006	\$0.0000418	\$0.0000446	\$0.0003	\$0.0000442	\$0.0003
LIDB Validation per query	OQU	\$0.041003	\$0.041003	\$0.0105974	\$0.00938	\$0.0103774		\$0.013400	\$0.0141003	\$0.041003
LIDB Originating Point Code Establishment or Change - NRC	N/A	\$64.36	NA	\$50.30	\$107.60	\$48.17	\$63.63	\$91.00	\$61.62	NA
NRC - Incremental Charge - Electronic Service Order	1	NA	NA	NA	NA	NA	NA	\$62.26	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$25.93	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$27.84	\$91.00
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	NA	NA	NA	NA	NA	\$27.84	NA
CCS7 SIGNALING TRANSPORT SERVICE										
CCS7 Signaling Connection, per link (A link) per month		\$18.79	\$5.00	\$17.05	\$16.31	\$19.48	\$21.58	\$155.00	\$21.79	\$155.00
NRC NRC		\$171.98	\$400.00	\$131.96	\$354.95	\$126.34	\$169.72	\$510.00	\$277.07	\$510.00
NRC - Disconnect		\$135.70	NA	NA	NA	\$101.10	\$134.08	NA	\$42.95	NA
NRC - Incremental Charge - Manual Service Order	SOMAN	\$25.93	NA	\$18.94	NA	\$18.14	\$25.52	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$16.31	NA	NA	NA	\$11.40	\$16.05	NA	NA	NA
CCS7 Signaling Connection, per link (B link) (also known as D link) per month		\$18.79	\$5.00	\$17.05	\$16.31	\$19.48	\$21.58	\$155.00	<b>4</b>	Not available
NRC		\$171.98	\$400.00	\$131.96	\$354.95	\$126.34	\$169.72	\$510.00	\$277.07	\$510.00
NRC - Disconnect		\$135.70	NA	NA	NA	\$101.10	\$134.08	NA	\$42.95	NA
NRC - Incremental Charge - Manual Service Order	SOMAN	\$25.93	NA	\$18.94	NA	\$18.14	\$25.52	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$16.31	NA	NA	NA	\$11.40	\$16.05	NA	NA	NA
CCS7 Signaling Termination, per STP port per month		\$148.72	\$113.00	\$133.99	\$174.08	\$161.99	\$161.12	\$132.88	\$156.33	\$355.00
CCS7 Signaling Usage, per ISUP message		\$0.00004	\$0.00001	\$0.0000354	\$0.000037893	\$0.0000430	\$0.0000456	\$0.00004	\$0.0000452	\$0.000023
(applicable when measurement and billing capability exists.)										
CCS7 Signaling Usage, per TCAP message		\$0.0001	\$0.00004	\$0.0000870	\$0.000102042	\$0.0001052	\$0.0001115	\$0.00009	\$0.0001108	\$0.00005
(applicable when measurement and billing capability exists.)										
CCS7 Signaling Usage Surrogate, per link per LATA per mo (9)		\$376.12	\$64.00	\$340.67	\$329.98	\$406.71	\$406.53	\$338.98	\$396.55	\$395.00
CCS7 Signaling Point Code, Establishment or Change, per STP affected	1	#co.oo	<b>#</b> 00.00	<b>#</b> 00.00	<b>#</b> 00.00	<b>#</b> 00.00	<b>#</b> 00.00	<b>#</b> 00.00	<b>#</b> 00.00	<b>CO 00</b>
I INC		\$62.00	\$62.00	\$62.00	\$62.00	\$62.00	\$62.00	\$62.00	\$62.00	\$62.00
OPERATOR CALL PROCESSING		_								
Operator Provided Call Handling per min - Using BST LIDB	N/A	\$1.21	\$1.00	\$0.9680296	\$1.6016	\$0.91	\$1.19	\$1.20	\$1.21	NA
Call Completion Access Termination Charge per call attempt	N/A	\$0.08	NA	NA	NA	NA	NA NA	NA	\$0.08	NA NA
Operator Provided Call Handling per min - Using Foreign LIDB	N/A	\$1.25	\$1.00	\$1.02	\$1.6249	\$0.96	\$1.24	\$1.24	\$1.25	NA NA
Call Completion Access Termination Charge per call attempt	N/A	\$0.08	NA	NA NA	NA	NA	NA	NA	\$0.08	NA NA
Operator Provided Call Handling, per call	N/A	NA	NA NA	NA.	NA NA	NA.	NA.	NA NA	NA NA	\$0.30
Fully Automated Call Handling per call - Using BST LIDB	N/A	\$0.11	\$0.10	\$0.0776409	\$0.0856	\$0.10	\$0.1072884	\$0.11	\$0.1115808	\$0.15
Fully Automated Call Handling per call - Using Foreign LIDB	N/A	\$0.13	\$0.10	\$0.0976984	\$0.1071	\$0.12	\$0.1253666	\$0.12	\$0.1293459	\$0.15
Professional recording of name (OCP alone)	USOD1	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00
Professional recording of name (DA and OCP alone)	USOD1	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00
DRAM or front-end loading, per TOPS switch	USOD2	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00
AABS or back-end loading, per IVS	USOD2	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00
EBAS or 0- automation loading, per NAV shelf	USOD2	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00
Recording Charge per Branded Announcement – Disconnect – Initial	N/A	\$9.61	NA	NA	NA	NA	NA	NA	NA	NA

Attachment 2

Exhibit C

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
Recording Charge per Branded Announcement – Disconnect – Subsequent	N/A	\$9.61	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	IN/A	ψ3.01	INA	INA	INA	INA	INA	INA	INA	INA
INWARD OPERATOR SERVICES										
Verification, per minute		\$1.16	NA	\$0.921083	NA	\$0.86	\$1.14	\$1.15	\$1.15	NA
Verification and Emergency Interrupt, per minute	N/A N/A	\$1.16	NA NA	\$0.921083	NA NA	\$0.86	\$1.14	\$1.15	\$1.15	NA NA
Verification, per call	VIL	NA	\$0.80	NA NA	\$1.00	NA NA	NA	\$0.54	NA	\$0.90
Verification and Emergency Interrupt, per call	N/A	NA	\$1.00	NA NA	\$1.111	NA	NA.	\$0.65	NA.	\$1.95
DIRECTORY ASSISTANCE SERVICES	14/71	147	ψ1.00	14/ (	Ψι.ιιι	1471	1471	ψ0.00	10.0	ψ1.00
Directory Assist Call Completion Access Svc (DACC), per call attempt	N/A	\$0.10	\$0.10	\$0.10	\$0.10	\$0.10	\$0.10	\$0.062	\$0.10	\$0.10
Call Completion Access Term charge per completed call	N/A	NA	NA	NA	NA	NA	NA	NA	\$0.08	NA
Number Services Intercept per query	N/A	\$0.0235	\$0.01	\$0.0097497	\$0.0086	\$0.02	\$0.0188268	\$0.0110	\$0.0124036	\$0.15
Number Services Intercept per Intercept Query Update	N/A	NA	NA	NA	\$0.0055	NA	NA	NA	NA	NA
Directory Assistance Access Service Calls, per call	N/A	\$0.275	\$0.275	\$0.275	\$0.275	\$0.275	\$0.275	\$0.260000	\$0.275	\$0.275
Professional recording of name (DA alone)	N/A	\$3,000.00	\$3,000.00	\$3,000.00	\$3,000.00	\$3,000.00	\$3,000.00	\$3,000.00	\$3,000.00	\$3,000.00
Professional recording of name (DA and OCP alone)	N/A	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00
DRAM or front-end loading, per TOPS switch	N/A	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00
AABS or back-end loading, per IVS	N/A	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00
EBAS or 0- automation loading, per NAV shelf	N/A	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00
Recording Charge per Branded Announcement – Disconnect – Initial	N/A	\$9.61	NA	NA	NA	NA	NA	NA	NA	NA
Recording Charge per Branded Announcement – Disconnect – Subsequent	N/A	\$9.61	NA	NA	NA	NA	NA	NA	NA	NA
Directory Transport										
Directory Transport - Local Channel DS1, per month	N/A	\$35.52	\$43.64	\$38.36	\$36.32	\$43.83	\$38.91	\$35.68	\$37.20	\$133.81
NRC - 1st	N/A	\$503.57	\$242.45	\$356.15	\$637.46	\$339.69	\$494.83	\$534.48	\$534.81	\$868.97
NRC - Add'l	N/A	\$442.84	\$226.44	\$312.89	\$546.94	\$298.29	\$435.28	\$462.69	\$462.81	\$486.83
NRC - Disconnect Charge - 1st	N/A	\$46.28	NA	NA	NA	\$33.02	\$46.85	NA	NA	NA
NRC - Disconnect Charge - Add'l	N/A	\$32.18	NA	NA	NA	\$23.32	\$33.02	NA	NA	NA
NRC - Incremental Charge-Manual Svc Order - NRC - 1st	SOMAN	\$61.99	NA	\$44.22	NA	\$42.34	\$59.58	\$86.15	\$87.99	NA
NRC - Incremental Charge-Manual Svc Order - NRC -addl		NA	NA	NA	NA	NA	NA	\$1.77	NA	NA
NRC - Incremental Charge-Manual Svc Order - NRC-Disconnect	SOMAN	\$29.27	NA	NA	NA	\$19.48	\$27.41	NA	\$3.11	NA
Directory Transport - Dedicated DS1 Level Interoffice per mile per mo	N/A	\$0.6923	\$0.6013	\$0.4523	\$0.45	\$0.78	\$0.6598	\$0.5753	\$0.7598	\$23.00
Directory Transport - Dedicated DS1 Level Interoffice per facility termination per mo	N/A	\$79.69	\$99.79	\$78.47	\$55.05	\$93.40	\$74.40	\$71.29	\$94.98	\$90.00
NRC - 1st	N/A	\$198.15	\$45.91	\$147.07	\$298.18	\$140.49	\$196.28	\$217.17	\$216.27	\$100.49
NRC - Add'l	N/A	\$148.18	\$44.18	\$111.75	\$231.18	\$106.69	\$147.31	\$163.75	\$162.70	\$100.49
NRC - Disconnect Charge - 1st	N/A	\$25.44	NA	NA NA	NA NA	\$20.00	\$26.56	NA	NA NA	NA NA
NRC - Disconnect Charge - Add'l NRC - Incremental Charge - Manual Service Order - 1st	N/A SOMAN	\$20.42	NA NA	NA ©4.0.04	NA NA	\$16.34	\$21.61	NA \$38.07	NA \$39.63	NA NA
NRC - Incremental Charge - Manual Service Order - 1st  NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$27.37 \$27.37	NA NA	\$18.94 NA	NA NA	\$18.14 \$18.14	\$25.52 \$25.52	\$38.07	\$39.63 \$39.63	NA NA
NRC - Incremental Charge - Manual Service Order - Add I  NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$27.37	NA NA	NA NA	NA NA	\$18.14	\$25.52	\$38.07 NA	\$39.63 NA	NA NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$12.97	NA NA	NA NA	NA NA	\$8.06	\$11.34	NA NA	NA NA	NA NA
Switched Common Transport per DA Access Service per call	N/A	\$0.0003	\$0.0003	\$0.0002906	\$0.000175	\$0.0003274		\$0.00020	\$0.000327	NA NA
Switched Common Transport per DA Access Service per call per mile	N/A	\$0.0003	\$0.0003	\$0.0002906	\$0.000175	\$0.0003274		\$0.00020	\$0.000327	NA NA
Access Tandem Switching per DA Access Service per call	N/A	\$0.0003	\$0.00055	\$0.000100	\$0.000783		\$0.0000202	\$0.0003	\$0.0024809	NA NA
DA Interconnection, per DA Access Service Call	N/A	\$0.00269	NA	\$0.00269	NA	NA	NA	\$0.00	\$0.00024003	NA NA
Directory Transport-Installation NRC, per trunk or signaling connection	N/A	\$0.00200		\$0.00 <u>2</u> 00		1		*****	Ţ0.000 <u>2</u> 00	
NRC - 1st	N/A	\$260.69	\$206.06	\$204.23	\$501.98	\$195.54	\$257.73	NA	\$407.81	NA
NRC - Add'I	N/A	\$5.95	\$4.71	\$4.42	\$13.32	\$4.23	\$5.85	NA	\$11.00	NA
NRC - Disconnect Charge - 1st	N/A	\$173.46	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Disconnect Charge - Add'I	N/A	\$5.95	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	NA	\$44.22	NA	\$130.05	\$171.49	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	NA	NA	\$4.23	\$5.85	NA	NA	NA
NRC - Manual Service Order - 1st		NA	NA	NA	NA	NA	NA	\$407.53	NA	NA
NRC - Manual Service Order - Add'l		NA	NA	NA	NA	NA	NA	\$10.98	NA	NA
		1								
Directory Assistance Database Service (DADS)										
Directory Assistance Database Service charge per listing	N/A	\$0.0446	\$0.001	\$0.0445	\$0.0193	\$0.0443	\$0.0447	\$0.04460	\$0.0444	NA
		•	•			•	•	•		

DESCRIPTION	usoc	AL	FL	GA	KY	LA	MS	NC	sc	TN
Directory Assistance Database Service, per month	DBSOF	\$128.55	\$100.00	\$95.50	\$120.76	\$90.54	\$126.17	\$126.26	\$127.23	NA
AIN (Note 4)										TBD
AIN, per message	CAM	NA	\$0.00004	NA	NA	NA	NA	NA	NA	NA
AIN - BellSouth AIN SMS Access Service	CAM								NA	NA
Service Establishment Charge, per state, initial set-up										
NRC	CAMSE	\$197.49	NA	\$90.25	NA	\$153.31	\$174.03	\$294.77	\$296.16	NA
NRC - Disconnect	CAMSE	\$114.22	NA	NA	NA	\$78.06	\$135.96	NA	NA	NA
Port Connection - Dial/Shared Access										
NRC	CAMDP	\$64.05	NA	\$29.66	NA	\$50.07	\$53.47	\$86.94	\$87.29	NA
NRC - Disconnect	CAMDP	\$27.04	NA	NA	NA	\$18.61	\$37.70	NA	NA	NA
Port Connection - ISDN Access	0.4144.5	004.05		<b>#</b> 00.00		<b>450.07</b>	<b>0</b> 50.47	000.04	007.00	
NRC NRC - Disconnect	CAM1P	\$64.05	NA	\$29.66	NA NA	\$50.07	\$53.47	\$86.94	\$87.29	NA NA
	CAM1P	\$27.04	NA	NA	NA	\$18.61	\$37.70	NA	NA	NA
User ID Codes - per User ID Code	CAMAU	¢141.04	NA	\$84.43	NA	\$104.05	\$420.02	\$200 B2	\$202.08	NA
NRC NRC - Disconnect	CAMAU	\$141.84 \$70.05	NA NA	\$84.43 NA	NA NA	\$104.95 \$48.95	\$129.83 \$79.91	\$200.83 NA	\$202.08 NA	NA NA
Security Card per User ID Code, initial or replacement	CAIVIAU	φ/0.05	INA	INA	INA	φ46.95	φ/9.91	INA	INA	INA
	CAMRC	\$142.13	NA	\$35.44	NA	\$125.33	\$131.54	\$172.05	\$172.26	NA
NRC - Disconnect	CAMRC	\$35.26	NA NA	935.44 NA	NA NA	\$24.40	\$45.77	NA	NA	NA NA
Storage, per unit (100Kb)	N/A	\$0.0026	NA NA	\$0.0023	NA NA	\$0.0029	\$0.0029	\$0.0023	\$0.0028	NA NA
Session per minute	N/A	\$0.0020	NA NA	\$0.0023	NA NA	\$0.0029	\$0.0029	\$0.0023	\$0.0942966	NA NA
C0. Performed Session, per minute	IN/A	ψ0.0032	INA	ψ0.0793004	NA NA	\$1.97	\$2.09	\$2.08	\$2.07	NA NA
AIN - BellSouth AIN Toolkit Service					14/4	Ψ1.57	Ψ2.00	Ψ2.00	Ψ2.01	14/-1
AIN, Service Creation Tools	CAMBP	NA	TBD	NA	NA	NA	NA	NA	NA	NA
Service Establishment Charge, per state, initial set-up	07 111121									
NRC	BAPSC	\$192.69	NA	\$86.74	NA	\$153.25	\$169.31	\$290.05	\$291.41	NA
NRC - Disconnect	BAPSC	\$114.22	NA	NA	NA	\$78.05	\$135.96	NA	NA	NA
Training Session, per customer		Ť				, , , , , ,				
NRC NRC	BAPVX	\$8,363.00	NA	\$8,348.00	NA	\$8,315.00	\$8,379.00	\$8,363.00	\$8,333.00	NA
NRC - Disconnect	BAPVX	NA	NA	NA	NA	NA	NA	NA	NA	NA
Trigger Access Charge, per trigger, per DN, Term. Attempt										
NRC	BAPTT	\$49.64	NA	\$19.13	NA	\$41.08	\$39.30	\$72.76	\$73.02	NA
NRC - Disconnect	BAPTT	\$27.04	NA	NA	NA	\$18.60	\$37.70	NA	NA	NA
Trigger Access Charge, per trigger per DN, Off-Hook Delay										
NRC	BAPTD	\$49.64	NA	\$114.80	NA	\$41.08	\$39.30	\$72.76	\$73.02	NA
NRC - Disconnect	BAPTD	\$27.04	NA	NA	NA	\$18.60	\$37.70	NA	NA	NA
Trigger Access Charge, per trigger, per DN, Off-Hook Immediate	ļ	1								
NRC	BAPTM	\$49.64	NA	\$19.13	NA	\$41.08	\$39.30	\$72.76	\$73.02	NA
NRC - Disconnect	BAPTM	\$27.04	NA	NA	NA	\$18.60	\$37.70	NA	NA	NA
Trigger Access Charge, per trigger, per DN, 10-Digit PODP	D.4===	011===		<b>ATO</b> 5.5		005.77	0405.55	0.4.1.	<b>0.155.55</b>	
NRC NRC	BAPTO	\$117.98	NA	\$70.06	NA	\$92.99	\$106.90	\$149.95	\$150.25	NA
NRC - Disconnect	BAPTO	\$37.90	NA	NA	NA	\$26.73	\$48.44	NA	NA	NA
Trigger Access Charge, per trigger, per DN, CDP	DADTO	C447.00	NIA	¢70.00	NIA	\$00.00	£400.00	£4.40.05	¢450.05	NIA
NRC NRC Picconnect	BAPTC	\$117.98	NA	\$70.06	NA	\$92.99	\$106.90	\$149.95	\$150.25	NA
NRC - Disconnect	BAPTC	\$37.90	NA	NA	NA	\$26.73	\$48.44	NA	NA	NA
Trigger Access Charge, per trigger, per DN, Feature Code NRC	BAPTF	\$117.98	NA	\$70.06	NA	\$92.99	\$106.00	\$149.95	\$150.25	NA
NRC - Disconnect	BAPTE	\$117.98	NA NA	\$70.06 NA	NA NA	\$92.99 \$26.73	\$106.90 \$48.44	\$149.95 NA	\$150.25 NA	NA NA
Query Charge, per query	DAFIF	\$37.90	NA NA	\$0.0209223	NA NA	\$26.73	\$0.0256138	\$0.02	\$0.0250662	NA NA
Type 1 Node Charge, per AIN Toolkit Subscription, per node, per query	1	\$0.024	NA NA	\$0.0209223	NA NA	\$0.0065	\$0.0256136	\$0.02	\$0.0250662	NA NA
17,500 1 14000 Onlargo, por mile Toolikit Gubsonption, per node, per query		ψυ.υυυ	1477	ψυ.υυσσ1σ1	14/7	ψυ.υυυυ	\$0.0000101	ψυ.υυυ	ψ0.0002313	14/7
SCP Storage Charge, per SMS Access Acct, per 100 Kb	N/A	\$1.63	NA	\$1.46	NA	\$1.79	\$1.79	\$1.45	\$1.73	NA
Monthly Report - per AIN Toolkit Service Subscription	BAPMS	\$16.00	NA	\$15.96	NA	\$15.89	\$16.01	\$15.98	\$15.93	NA
NRC	BAPMS	\$44.56	NA	\$22.64	NA	\$34.61	\$44.02	\$71.80	\$72.15	NA

DESCRIPTION	USOC	AL	FL	GA	кү	LA	MS	NC	sc	TN
NRC - Disconnect	BAPMS	\$31.84	NA	NA	NA NA	\$21.97	\$31.28	NA	NA NA	NA
Special Study - per AIN Toolkit Service Subscription	BAPLS	\$0.10	NA	\$0.0861109	NA	\$0.08	\$0.0810536	\$0.08	\$0.0872769	NA
I INRC	BAPLS	\$47.74	NA	\$22.64	NA	\$37.77	\$47.21	\$47.20	\$47.35	NA
NRC - Disconnect	BAPLS	\$15.90	NA NA	NA NA	NA NA	NA	NA	NA NA	NA	NA
Call Event Report - per AIN Toolkit Service Subscription	BAPDS	\$15.90	NA	\$15.87	NA	\$15.81	\$15.93	\$15.90	\$15.84	NA
NRC	BAPDS	\$44.56	NA NA	\$22.64	NA	\$34.61	\$44.02	\$71.80	\$72.15	NA
NRC - Disconnect	BAPDS	\$31.84	NA NA	NA NA	NA	\$21.97	\$31.28	NA	NA NA	NA
Call Event special Study - per AIN Toolkit Service Subscription	BAPES	\$0.003	NA NA	\$0.0028704	NA NA	\$0.0026	\$0.0027018	\$0.003	\$0.0029092	NA
NRC	BAPES	\$47.74	NA NA	\$22.64	NA NA	\$37.77	\$47.21	\$47.20	\$47.35	NA
NRC - Disconnect	BAPES	\$15.90	NA NA	NA NA	NA NA	\$37.77	NA	NA NA	NA	NA
	5, 11 20	ψ.ο.οο				ψο				
CALLING NAME (CNAM) QUERY SERVICE										
CNAM (Database Owner), Per Query	N/A	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016
CNAM (Non-Database Owner), Per Query *	N/A	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01
NRC, applicable when CLEC-1 uses the Character Based User Interface (CHUI)	N/A	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00
* Volume and term arrangements are also available.	14// (	Ψ000.00	φοσο.σσ	φοσο.σσ	ψ000.00	Ψ000.00	ψοσο.σσ	φοσο.σσ	φοσσ.σσ	φοσσ.σσ
voiding and term arrangements are also available.		+								
SELECTIVE ROUTING (Note 5)		+		<b>†</b>			<b>†</b>			
Per Line or PBX Trunk, each		NA	NA	NA	\$10.00 (Interim	NA	NA	NA	NA	TBD
NRC		NA	NA NA	NA.	NA NA	NA NA	NA.	NA	NA NA	TBD
Customized routing per unique line class code, per request, per switch		14/3	14/3	14/7	14/3	NA NA	NA NA	NA NA	NA NA	NA NA
NRC	USRCR	\$230.60	\$229.65	\$180.62	\$229.65	\$229.65	\$227.99	\$229.65	\$226.22	\$229.65
NRC - Incremental Charge - Manual Service Order	CONTON	\$25.93	NA NA	\$18.94	NA	NA NA	\$25.52	NA NA	\$27.84	NA
Tarke moraliar charge manda corvice craci		Ψ20.00	14/1	Ψ10.01	14/ (	1471	Ψ20.02	10/1	Ψ21.01	1471
VIRTUAL COLLOCATION										
2-wire Cross-Connect		-								
RC	UEAC2	\$0.28	\$0.524	\$0.30	\$0.31	\$0.26	\$0.3996	\$0.09	\$0.3648	\$0.30
NRC - 1st	UEAC2	\$30.76	\$11.57	\$12.60	\$54.21	\$23.04	\$30.93	\$41.78	\$41.50	\$19.20
NRC - Add'l	UEAC2	\$29.40	\$11.57	\$12.60	\$51.07	\$22.11	\$29.59	\$39.23	\$38.94	\$19.20
NRC - 1st - Manual Service Order	OLAOZ	NA	NA	NA	NA NA	NA	NA	\$4.75	NA	NA
NRC - Add'l - Manual Service Order		NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$4.75	NA NA	NA NA
NRC - Disconnect - 1st	UEAC2	\$12.75	NA NA	NA NA	NA NA	\$9.48	\$12.76	NA NA	NA NA	NA NA
NRC - Disconnect - Add'l	UEAC2	\$11.38	NA NA	NA NA	NA NA	\$8.54	\$11.43	NA NA	NA NA	NA NA
4-wire Cross-Connect	OLAOZ	ψ11.00	14/3	IVA	14/1	ψ0.0-	ψ11.40	14/-1	14/3	14/-1
RC	UEAC4	\$0.56	\$0.524	\$0.50	\$0.62	\$0.52	\$0.7992	\$0.18	\$0.7297	\$0.50
NRC - 1st	UEAC4	\$66.71	\$11.57	\$12.60	\$54.23	\$23.23	\$31.17	\$41.91	\$41.56	\$19.20
NRC - Add'l	UEAC4	\$50.43	\$11.57	\$12.60	\$50.96	\$22.24	\$29.77	\$39.25	\$38.90	\$19.20
NRC - Add 1	01/04	NA	NA	NA	NA	Ψ22.24 NA	NA	\$4.73	NA	NA
NRC - Add'l - Manual Service Order		NA NA	NA NA	NA NA	NA NA	NA	NA NA	\$4.73	NA NA	NA NA
NRC - Disconnect - 1st	UEAC4	\$12.82	NA NA	NA NA	NA NA	\$9.53	\$12.83	NA	NA NA	NA NA
NRC - Disconnect - Add'l	UEAC4	\$11.39	NA	NA NA	NA NA	\$8.55	\$11.43	NA	NA NA	NA NA
2-fiber Cross-Connect	02/101	\$.1.00	. •, •	· • • • • • • • • • • • • • • • • • • •	, .	ψ0.00	<b>\$.1.10</b>	, .	,, .	, .
RC	CNC2F	\$12.10	NA	\$15.64	\$15.64	\$19.13	\$15.64	\$15.99	\$15.06	\$15.64
NRC - 1st	CNC2F	\$55.46	NA NA	\$41.56	\$41.56	\$41.07	\$41.56	\$67.34	\$69.28	\$41.56
NRC - Add'l	CNC2F	\$39.18	NA NA	\$29.82	\$29.82	\$29.63	\$29.82	\$48.55	\$48.89	\$29.82
NRC - Disconnect - 1st	CNC2F	\$16.83	NA NA	NA	NA	\$12.84	\$12.96	NA	NA NA	NA
NRC - Disconnect - Add'l	CNC2F	\$13.27	NA NA	NA NA	NA NA	\$10.29	\$10.34	NA NA	NA NA	NA NA
4-fiber Cross-Connect	OHOLI	ψ10.21	14/1	1471	14/1	ψ10.20	Ψ10.07	14/1	1471	14/1
RC	CNC4F	\$21.75	NA	\$28.11	\$28.11	\$34.38	\$28.11	\$28.74	\$27.08	\$28.11
NRC - 1st	CNC4F	\$66.71	NA NA	\$50.53	\$50.53	\$49.81	\$50.53	\$82.35	\$84.07	\$50.53
NRC - Add'l	CNC4F	\$50.43	NA	\$38.78	\$38.78	\$38.37	\$38.78	\$63.56	\$63.68	\$38.78
NRC - Disconnect - 1st	CNC4F	\$21.86	NA NA	NA	NA	\$16.75	\$16.97	NA	NA	NA
NRC - Disconnect - 1st	CNC4F	\$18.31	NA NA	NA NA	NA NA	\$14.20	\$14.35	NA NA	NA NA	NA NA
DS1 Cross-Connects	011041	ψ10.51	INA	INA	INA	ψ14.20	ψ14.55	INA	INA	INA
RC		NA	NA	NA	NA	NA	NA	\$0.97	NA	NA
NRC - 1st		NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$71.02	NA NA	NA NA
		INA	IN/A	INA	INA	INA	INA	φ <i>ι</i> 1.U∠	INA	INA

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
NRC - Add'l		NA	NA	NA	NA	NA	NA	\$51.08	NA	NA
NRC - Manual Service Order - 1st		NA	NA	NA	NA	NA	NA	\$4.70	NA	NA
NRC - Manual Service Order - Add'l		NA	NA	NA	NA	NA	NA	\$4.70	NA	NA
DS3 Cross-Connects										
RC		NA	NA	NA	NA	NA	NA	\$12.33	NA	NA
NRC - 1st		NA	NA	NA	NA	NA	NA	\$69.84	NA	NA
NRC - Add'l		NA	NA	NA	NA	NA	NA	\$49.43	NA	NA
NRC - Manual Service Order - 1st		NA	NA	NA	NA	NA	NA	\$4.70	NA	NA
NRC - Manual Service Order - Add'l		NA	NA	NA	NA	NA	NA	\$4.70	NA	NA
If no rate is identified in the contract, the rate for the specif	ic service or function will be as set forth in applical	ble BellSouth ta	riff or as negot	iated by the pa	arties upon reque	st by either pa	rty.			
interim rates subject to true-up.										
1 BellSouth and CLEC shall negotiate rates for this of										
reached within sixty (60) days of the Effective Date,										
Florida PSC to settle the disputed charge or charge	s. (FL)									
2 This rate element is for those states w/o separate ra										
Delivery vs. POTS No. Delivery and calls with Option	nal Complex Features vs. w/o									
Optional Complex Features.										
3 This charge is only applicable where signaling usag	e measurement or billing									
capability does not exist.										
4 Prices for AIN to be determined upon development	of mediation device. (TN)									
5 Price for Line Class Codes for Selective Routing sh	all be determined by the TRA.									
I I I(TN)	*				1					

### AMENDMENT TO THE

# INTERCONNECTION AGREEMENT BETWEEN BUSINESS TELECOM, INC.

### AND

# BELLSOUTH TELECOMMUNICATIONS, INC. DATED February 21, 2000

This Amendment to Interconnection Agreement ("Amendment") is entered into by and between BellSouth Telecommunications, Inc., a Georgia corporation ("BellSouth"), and Business Telecom, Inc., a North Carolina corporation ("BTI"). BellSouth and BTI may be individually referred to herein as "Party" and collectively as "Parties".

WHEREAS, BellSouth and BTI executed an Interconnection Agreement dated February 21, 2000 (the "Interconnection Agreement"); and

WHEREAS, the Parties desire to amend the Interconnection Agreement as set forth herein;

NOW, THEREFORE, for and in consideration of the promises herein contained, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree to amend the Interconnection Agreement as follows:

- 1. The term of the Amendment shall begin on January 1, 2001 and shall terminate on September 30, 2002.
- 2. The existing Interconnection Agreement is hereby amended to add the following, which shall be a new Attachment 1A to the Interconnection Agreement:
  - 1.0 Volume and Term
  - 1.1 Scope

This Amendment applies to the services specified in Appendix I provisioned within the states of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee.

## 1.2 Purpose

BellSouth and BTI intend for BTI to purchase the services set forth in Appendix I at a Volume & Term Rate ("V&T Rate") as set forth in Section 1.3.3 and 1.6. The specified services shall be

purchased in sufficient quantities so that BTI will comply with the minimum quantity set forth in Section 1.4 below.

- 1.3 Definitions
- 1.3.1 "V&T Eligible Services" are those services listed in Appendix I.
- 1.3.1.1 BellSouth shall upon BTI's request, consider the addition of new rates, terms and conditions for business lines, vertical features and rotary/hunting service tariffed after the effective date of this Amendment ("Newly Tariffed Service").
- 1.3.1.2 BellSouth shall use best efforts to notify BTI when a Newly Tariffed Service is added that falls into one of the three categories listed in paragraph 1.3.1.1. BTI shall submit, in writing to BellSouth, a request to add the Newly Tariffed Service to Appendix I. If accepted by BellSouth as a V&T Eligible Service, BellSouth shall propose an effective V&T within 45 days of receipt of BTI's request to add the Newly Tariffed. The parties will amend this Amendment to add such new services and the effective date for the V&T rate for the Newly Tariffed Service shall be the date such service first appears on BTI's bill.
- 1.3.1.3 BellSouth shall upon BTI's request, consider the addition of new rates, terms and conditions for business lines, vertical features and rotary/hunting service tariffed before the effective date of this Amendment ("Previously Tariffed Service").
- 1.3.1.4 BTI shall submit, in writing to BellSouth, a request to add the Previously Tariffed Service to Appendix I. If accepted by BellSouth as a V&T Eligible Service, BellSouth shall propose an effective V&T Rate within 45 days of receipt of BTI's request to add the Previously Tariffed Service. The parties will amend this Amendment to add such new services and the effective date for the V&T rate for the Previously Tariffed Service shall be the date last Party signs the Amendment.
- 1.3.2 "Business Lines" under this Amendment are calculated using the following V&T Eligible Services or any additional services added as per Section 1.3.1.1:
  - Flat Rate Business = 1 Line
  - Business Plus/ Business Choice = 1 line
  - Expanded Area Calling Business = 1 Line
  - Measured Rate Business = 1 line

- 1.3.3 V&T Rate is the state specific flat rate for the recurring charges for V&T Eligible Services. The V&T Rate shall be applied in lieu of the Tariff Rate less the Resale Applicable Discounts as specified in Exhibit A, for the recurring charges for V&T Eligible Services. The Tariff Rate less the Resale Applicable Discounts shall be applied, however, to the non-recurring charges for V&T Eligible Services.
- 1.4 Resale Commitment
- 1.4.1 To be eligible for the V&T Rate, BTI shall purchase a minimum of 24,500 Business Lines ("Minimum Commitment").
- 1.4.1.1 To be eligible for the V&T Rates, BTI shall maintain at least 9,000 of BTI's total resale Business Lines in Zone 1 and Zone 2 ("Zone Requirements"). Zones 1 and 2 are defined in the attached Appendix II.
- 1.4.1.2 To be eligible for the V&T Rates, BTI shall have on the Effective Date of the Amendment at least 14,700 of BTI's total resale Business Lines in Alabama, Georgia, Kentucky, Mississippi, North Carolina, South Carolina and Tennessee ("State Requirements").
- 1.4.2 Not later than August 15, 2001, and every 6 months thereafter, BellSouth will review BTI's resale business lines in-service to determine whether BTI achieved or otherwise maintained its Minimum Commitment and Zone requirements.
- 1.4.3 If as a result of these reviews, BellSouth determines that BTI did not meet its Minimum Commitment or Zone Requirements, BTI shall pay a Commitment Shortfall Charge; provided, however, that if BTI disagrees with BellSouth's determination, BTI may invoke the Billing Dispute procedures of Section 8.6.3 of the General Terms and Conditions. Upon request, BellSouth will provide documentation supporting its determination.
- 1.4.3.1 If the number of Business Lines BTI purchases for resale falls below 22,500 business lines, BTI will be assessed a \$60 per line Commitment Shortfall Charge ("Commitment Shortfall Charge") for each line and each month BTI is below the 22,500 Minimum Commitment. Or, if the number of Business Lines BTI purchases for resale falls below 9,000 Business Lines in Zone 1 and Zone 2, BTI will be assessed a \$12 per line Zone Requirement Shortfall Charge ("Zone Requirement Shortfall Charge") for each line and each month BTI is below the 9,000 Business Lines in Zone 1 and Zone 2 Zone Requirements. If BTI falls below the 22,500 Minimum Commitment and falls below the 9,000 Business Lines in Zone 1

- and Zone 2, BTI will only be subject to the Commitment Shortfall Charge.
- 1.4.3.2 If the number of Business Lines BTI purchases for resale falls below 20,250 business lines, BTI shall be subject to the termination liability as defined in Section 1.8 of this Amendment.
- 1.4.3.3 BellSouth will issue BTI a bill for the Shortfall Charge; BTI shall pay the charge within thirty (30) days of receipt; provided, however, that if BTI disagrees with BellSouth's assessment of Commitment Shortfall Charge, BTI may invoke the Billing Dispute procedures of Section 8.6.3 of the General Terms and Conditions.
- 1.5 Resale Commitment Reward
- 1.5.1 For each Business Line target BTI meets, on or before the specified Business Line target date as set forth below, BTI will receive a \$75,000 credit on BellSouth's CRIS bill to BTI. If BTI does not meet the Business Line target by the Business Line target date, BTI is not eligible to receive the \$75,000 credit.
  - 26,000 Business Lines on or before August 1, 2001
  - 28,000 Business Lines on or before September 1, 2001
  - 30,000 Business Lines on or before March 1, 2002
  - 32,000 Business Lines on or before May 1, 2002
- 1.6 Volume & Term Rate
- 1.6.1 The Volume & Term ("V&T") Rates for V&T Eligible Services shall be determined by the rates set forth in Appendix III.
- 1.6.2 In the event that BellSouth, either before or after the effective date of the Amendment, enters into an agreement with any other telecommunications carrier ("Other Volume Agreement") which provides for the provision within the States of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina, or Tennessee of any of the arrangements covered by this Agreement upon rates, terms or conditions that differ in any material respect from rates, terms and conditions for such arrangements set forth in the Amendment ("Other Terms"), BellSouth shall be deemed thereby to have offered such Other Volume Agreement to BTI in its entirety. In the event BTI elects to adopt such Other Volume Agreement in its entirety, the Parties shall execute an amendment adopting such Other Volume Agreement. The Parties will work to ensure that such adoption occurs within 60 days. The effective date of the adopted

Other Volume Agreement shall be the date such adoption is executed by both Parties.

- 1.7 Secondary Service Charge Waiver
- 1.7.1 During such times as BTI maintains the Minimum Commitment as described in Section 1.4.1, BellSouth will waive the secondary service charges, as described in Section A4 of the General Subscriber Service Tariff, for the addition of Custom Calling and Touchstar Services and Rotary/Hunting Services. Further, BTI acknowledges that due to the functionality of BellSouth's billing systems, BellSouth is or may be unable to reflect applicable discounts on BTI's initial bill for secondary service order charges. All credits issued pursuant to the secondary service charge waiver shall be in arrears such that the applicable credit for a given billing cycle will be reflected on a quarterly basis.
- 1.8 Termination
- 1.8.1 Notwithstanding anything to the contrary in the General Terms and Conditions, BTI shall have the right to terminate the purchase of Volume and Term eligible services, as defined in Appendix I at Volume & Term Rates, prior to expiration by providing BellSouth written notice of such termination 60 days prior to the effective date of the termination.
- 1.8.2 Notwithstanding anything to the contrary in the General Terms and Conditions, BellSouth shall have the right to terminate the purchase of Volume and Term eligible services, as defined in Appendix I at Volume & Term Rates, prior to expiration by providing BTI written notice of such termination 60 days prior to the effective date of termination when the following condition applies:
  - If the number of Business Lines BTI purchases for resale falls below 20,250 Business Lines.
- 1.8.3 In the event BTI or BellSouth terminates the Amendment, pursuant to Section 1.8, BTI will be charged a termination charge equal to the year-to-date accumulated credits received, excluding any resale applicable discounts ("Accumulated Credits") multiplied by a Termination factor ("Termination Factor"). The Termination Factors applicable if BTI or BellSouth terminate the Amendment are as follows:
  - Effective Date through November 30, 2001 Termination Factor shall be 100%

- December 1, 2001 through expiration or termination -Termination Factor shall be 95%
- 1.8.4 Any termination liability due shall be reduced by the Commitment Shortfall Charge and Zone Requirement Shortfall Charges already paid by BTI pursuant to section 1.4.3.1.
- 1.9 Reciprocal Compensation
- 1.9.1 No compensation will be due from BellSouth for the transport and termination of resold services purchased by BTI from BellSouth even if BTI purchases such services on behalf of or for the use of another entity, whether or not affiliated with BTI. Further, the Parties agree that no compensation will be due from BTI to BellSouth for transport and termination of Local Traffic over resold services.
- 1.10 Tariff Changes
- 1.10.1 If during the term of the Amendment, BellSouth receives regulatory approval for price reductions or increases on tariff services purchased by BTI, V&T Rates described in Appendix II shall not be affected by such reduction or increase and shall still apply: provided however, BTI may at its option elect to purchase resold services at the new tariffed rate less the Resale Applicable Discounts set forth in Exhibit A of Attachment 1. BTI shall notify BellSouth in writing of such election, and the Parties shall agree to amend this Amendment to reflect such change. The new rate shall become effective on the date such amendment is executed by both Parties. BTI shall have sole discretion to include lines resold under such tariffed rates in the calculation of the Minimum Commitment.
- 3. The term of the Interconnection Agreement is extended to September 30, 2002.
- 4. Sections 2.3 and 2.4 of the General Terms and Conditions of the Interconnection Agreement are deleted in their entirety and replaced with the following Sections 2.3 and 2.4 below:
  - 2.3 If, within one hundred and thirty-five (135) to one hundred and sixty (160) days of commencing the negotiation referred to in Sections 2.1 and 2.2 above, the Parties are unable to satisfactorily negotiate new resale and/or local interconnection terms, conditions and prices, either Party may petition the Commission to establish appropriate local interconnection and /or resale arrangements pursuant to 47 U.S.C. 252. The Parties agree that, in such event, they shall encourage the

Commission to issue its order regarding the appropriate local interconnection and/or resale arrangements no later than the expiration date of this Agreement. The Parties further agree that in the event the Commission does not issue its order prior to the expiration date of this Agreement, or if the Parties continue beyond the expiration date of this Agreement to negotiate the local interconnection and/or resale arrangements without Commission intervention, the terms, conditions and prices ultimately ordered by the Commission or negotiated by the Parties, will be effective retroactive to the day following the expiration date of this Agreement.

- 2.4 In the event that as of the date of expiration of this Agreement and conversion of this Agreement to a month-to-month term, the Parties have not entered into a Subsequent Agreement and (i) the Parties are not involved in active good faith negotiations or mediation for a Subsequent Agreement and (ii) no arbitration proceeding has been filed in accordance with Section 2.3 above, then either Party may terminate this Agreement upon sixty (60) days written notice to the other Party. In the event that BellSouth terminates this Agreement as provided above, BellSouth shall continue to offer services to BTI pursuant to the terms, conditions and rates set forth in BellSouth's Statement of Generally Available Terms (SGAT) to the extent an SGAT has been approved by the applicable Commission(s). If any state Commission has not approved a BellSouth SGAT, then upon BellSouth's termination of this Agreement as provided herein, BellSouth will continue to provide services to BTI pursuant to BellSouth's then current standard interconnection agreement. In the event that the SGAT or BellSouth's standard interconnection agreement becomes effective as between the Parties, the Parties may continue to negotiate a Subsequent Agreement, and the terms of such Subsequent Agreement shall be effective retroactive to the day following expiration of this Agreement.
- 5. Attachment 4 of the Interconnection Agreement is deleted in its entirety and is hereby replaced with a new Attachment 4, attached hereto as Exhibit 1.
- 6. Attachment 9 of the Interconnection Agreement is deleted in its entirety and is hereby replaced with a new Attachment 9, attached hereto as Exhibit 2.
- 7. All of the other provisions of the Interconnection Agreement shall remain in full force and effect.
- 8. Either or both of the Parties is authorized to submit this Amendment to the appropriate Commission or other regulatory body having jurisdiction over the subject matter of this Amendment, for approval subject Section 252(e) of the federal Telecommunications Act of 1996.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representative on the date indicated below.

Business Telecom, Inc. Signature on File	<b>BellSouth Telecommunications, Inc.</b> Signature on File
Signature	Signature
Mike Newkirk Name	Gregory R. Follensbee Name
COO Title	Senior Director Title
<u>5/23/01</u> Date	6/5/01 Date

USOC Code	Description
113	REGIONSERVE BUSINESS INDI+. 2 WAY
113CL	REGIONSERVE BUSINDIV+W/CALLER ID
1FB	
1FBCL	Flat rate line, business, two-way
1LD1E	Flat Rate business line, caller ID PRIMARY RATE ISDN NON-DIS, month to month
1MB	
1MBCL	Message rate line, business  Measure Rate Business Line with Caller ID
1MBGE	Message rate line, business, Georgia Community Calling
1MG	BUSINESS MEASURED RATE LINE BUSINESS, TWO WAY, NON HUNTING
1MGCL	Measure Rate Business Line with Caller ID
1MH	Measure Rate Business Line with Caller ID
1S8	BUSINESS LINE, ECONOMY SERVICE, LOCAL OPTION SERVICE RATE, TWO WAY
1S8CL	BUSINESS LINE, ECONOMY SERVICE, LOCAL OPTION SERVICE RATE, TWO WAY, with Caller ID
1ZJ	BUSINESS LINE STANDARD LOCAL OPERATIONAL SERVICE RATE, BUSINESS, STD OPTION, TWO WAY
1ZJCL	BUSINESS LINE STANDARD LOCAL OPERATIONAL SERVICE RATE, BUSINESS, STD OPTION, TWO WAY w/Caller ID
3QM1X	ESSX NAR RegionServ, msd rate inward only
3QMCX	ESSX NAR RegionServ, msd rate both way
3QMOX	ESSX NAR RegionServ, msd rate outward only
7FB	Business Line - Auxiliary Line
7FBCL	Business Line - Auxiliary Line with Caller ID
ACB	AREA CALLING SERVICE BUSINESS, EXPANDED LOCAL AND CALLING, ECONOMY OPTION
ACBCL	AREA CALLING SERVICE BUSINESS, EXPANDED LOCAL AND CALLING, ECONOMY OPTION with Caller ID
ASB	AREA CALLING SERVICE, EXPANDED LOCAL AREA CALLING SERVICE, STD OPTION
ASBCL	Expanded Local Area Calling Service, Standard Option with Caller ID
B1M	Business Line - Measured
B1MCL	Business Line - Measured
B2K1D	Flat rate usage charge, business inward (w/ LUD)
B2K1K	Flat rate usage charge, business inward
B2K2D	Flat rate usage charge, business both way (w/ LUD)
B2K2K	Flat rate usage charge, business both way (w/o LUD)
B2K2P	Flat rate usage charge, business both way, premium calling
B6P	AREA PLUS SERVICE BUSINESS
B6PCL	AREA PLUS SERVICE BUSINESS WITH CALLER ID
BC1	BUSINESS CHOICE
BC2	BUSINESS CHOICE
BD1	Business Plus Service, Florida, option 1, flat rated plan

USOC Code	Description
BD1CL	BUSINESS LINE WITH TOUCHTONE+
BD2	Business Plus Service, Florida, option 2, combination rated plan
BD2CL	BUSINESS LINE WITH TOUCH+
BF1	Business Plus Service, Alabama, option 1, flat rated plan
BF1CL	BUSINESS LINE W/TT OPT 1 W/CALLER ID
BF2	Business Plus Service, Alabama, option 2, combination rated plan
BF2CL	BUSINESS LINE WITH TT OPTION 2 WITH CALLER ID
BG1	Business Plus Service, Georgia, option 1, flat rated plan
BG1CL	Business Plus Service, Georgia, option 1, flat rated plan with Caller ID
BG2	Business Plus Service, Georgia, option 2, combination rated plan
BG2CL	BUSINESS LINE W/TOUCH-TONE W/ CALLER ID
BK1	Business Plus Service, Kentucky, option 1, flat rated plan
BK1CL	Business Plus Service, Kentucky, option 1, flat rated plan, with Caller ID
BK2	Business Plus Service, Kentucky, option 2, combination rated plan
BK2CL	Business Plus Service, Kentucky, option 2, combination rated plan, with Caller ID
BL1	Business Plus Service, Louisiana, option 1, flat rated plan
BL1CL	Business Plus Service, Louisiana, option 1, flat rated plan, with Caller ID
BL2	Business Plus Service, Louisiana, option 2, combination rated plan
BL2CL	Business Plus Service, Louisiana, option 2, combination rated plan, with Caller ID
BMU	PBX Service, Measured Rate, Inward Only
BO8	Local Option service rate, business economy option inward only
BS1	Business Plus Service, South Carolina, option 1, flat rated plan
BS1CL	Busines Line with Touch+
BS2	Business Plus Service, South Carolina, option 2, combination rated plan
BS2CL	Busines Line with Touch+
BT1	Business Plus Service, Tennessee, option 1, flat rated plan
BT1CL	Business Plus Service, Tennessee, option 1, flat rated plan with Caller ID
BT2	Business Plus Service, Tennessee, option 2, combination rated plan
BT2CL	Busines Line with Touch+
BU1	Business Plus Service, Mississippi, option 1, flat rated plan
BU1CL	Business Plus Service, Mississippi, option 1, flat rated plan, with Caller ID
BU2	Business Plus Service, Mississippi, option 2, combination rated plan
BU2CL	Business Plus Service, Mississippi, option 2, combination rated plan, with Caller ID
BUA	BUSINESS MEASURED LINE FLAT MEASURED BUS LINE, TWO WAY
BV1	Business Plus Service, North Carolina, option 1, flat rated plan
BV1CL	Business Plus Service, North Carolina, option 1, flat rated plan, with Caller ID

USOC Code	Description			
BV2	Business Plus Service, North Carolina, option 2, combination rated plan with Caller ID			
BV2CL	Business Plus Service, North Carolina, option 2, combination rated plan			
CCS	Class Room Communication			
COM11	COMPLETE CHOICE FOR BUSINESS, BUS PLUS SRVC OPTION 1, 1 LINE PKG OF MILEAGE			
COM12	COMPLETE COICE FOR BUS 2-LINE PACKAGE			
COM13	Complete Choice for Business			
COM14	Complete choice for bus, bus pus srvc opt 1			
COM15	Complete choice for bus, bus pus srvc opt 2			
COM16	Complete choice for bus, bus pus srvc opt 3			
COM17	Complete choice for bus, bus pus srvc opt 4			
COM18	Complete choice for bus, bus pus srvc opt 5			
COM19	Complete choice for bus, bus pus srvc opt 6			
COM1H	Complete choice for bus, bus pus srvc opt 1			
COM21	Complete Choice for Business			
COM22	Complete Choice for Business			
COM24	Complete Choice for Business			
COM25	Complete Choice for Business			
COM26	Complete Choice for Business			
COM27	Complete Choice for Business			
COM28	Complete Choice for Business			
COM29	Complete Choice for Business			
COM2H	Complete choice for bus, bus pus srvc opt 1			
COMF1	Complete Choice, Flat Rate Business Service			
COMF2	Complete Choice, Flat Rate Business Service			
COMF3	Complete Choice, Flat Rate Business Service			
COMF5	Complete Choice, Flat Rate Business Service			
COMF6	Complete Choice, Flat Rate Business Service			
COMF7	Complete Choice, Flat Rate Business Service			
COMF8	Complete Choice, Flat Rate Business Service			
COMF9	Complete Choice, Flat Rate Business Service			
COMP1	Complete Choice for Business			
COMP2	Complete Choice for Business			
COMP3	Complete Choice for Business			
COMPF	Complete Choice for Business Package			
COMPH	Complete Choice for Business			
E4O	Call forwarding variable			

USOC Code	Description			
EQ4GE	MESSAGE RATE ESSC NAR, One way incoming, EXPANDED LOCAL RATE 3 MILE CALLING (GA)			
EQA	NETWORK ACCESS REGISTER, ESSX, combination			
EQB	NETWORK ACCESS REGISTER, ESSX, one way incoming			
EQC	NETWORK ACCESS REGISTER, ESSX, one way incoming  NETWORK ACCESS REGISTER, ESSX, one way outgoing			
EQE	NETWORK ACCESS REGISTER, ESSX, combination, initial			
EQEGE	MESSAGE RATE ESSC NAR, COMBINATION EXPANDED LOCAL RATE 3 MILE CALLING (GA)			
EQF	NETWORK ACCESS REGISTER, ESSX, one way outgoing, initial			
EQFGE	MESSAGE RATE ESSC NAR, One way outgoing, EXPANDED LOCAL RATE 3 MILE CALLING (GA)			
EQG	NETWORK ACCESS REGISTER, ESSX, combination, additional			
EQK	NETWORK ACCESS REGISTER, ESSX, one way outgoing, additional			
EQP	NETWORK ACCESS REGISTER, Measured Rate, both way			
EQR	NETWORK ACCESS REGISTER, Measured Rate, inward only			
EQS	NETWORK ACCESS REGISTER, Measured Rate, outward only			
ER3	Call forwarding Variable with Speed calling (8-code)			
ER5	Call forwarding Variable with Three way calling			
ER6	Three way calling with speed calling (8-code)			
ER9	Call waiting with Three way calling			
ES3	Call waiting, call forwarding variable, three way calling and speed calling (8-code)			
ES6	Call Waiting with Speed Calling (8-code)			
ES7	Call Waiting with call forwarding variable			
ESA	Call waiting with Call forwarding Variable and Speed Calling( 8-code)			
ESC	Three way Calling			
ESCWT	Three way calling with transfer			
ESF	Speed calling 30 code			
ESFTK	Speed calling 30 code			
ESFWT	Speed calling 30 code			
ESL	Speed Calling 8 code			
ESLTK	Speed Calling 8 code			
ESLWT	Speed Calling 8 code			
ESM	Call forwarding variable			
ESR	Call forwarding variable with three way calling and speed calling (8-code)			
ESX	Call waiting			
ETB	Call waiting with Three way calling and speed calling (8-code)			
ETC	Call waiting with call forwarding variable and three way calling			
EXMEX	MAIN STATION LINE, DISTANCE ESSX EXCHANGE CIRCUIT, SCB-FLAT SBT-FLAT, MESS. 01 MEAS RATE 1 1/2 MILES			
EXMFA	MILEAGE ESSX/DIGITAL ESS ESSX EXCHANGE CIRCUIT, MAIN STATION LINE, MSG/MEAS RATE 2 MILES, NC			

USOC Code	Description			
EXMFX	MAIN STATION LINE, DISTANCE SENSITIVE 2 MILES			
EXMGX	MAIN STATION LINE DISTANCE, ESSX EXCHANGE CIRCUIT, MAIN STATION LINE, SCB-FLAT RATE, SBT FLAT RATE, MSG/MEAS RATE 2 1/2 MILES			
EXMHX	MAIN STATION LINE ESSX EXCHANGE CIRCUITE, MAIN STATION LINE MSG/MEAS 3 1/2 MILES			
EXMJX	MAIN STATION LINE, DISTANT ESSX EXCHANGE CIRCUITE, MAIN STATION LINE/MEAS RATE			
EXMOX	MAIN STATION LINE, DISTANT ESSX EXCHANGE CIRCUITE, MAIN STATION LINE			
EXMPX	MAIN STATION LINE, DISTANCE ESSX EXCHANGE CIRCUIT, MSG/MEASSURE RATE 1 1/4 MILES			
FCS	Flexible call forwarding			
FCSCN	Flexible call forwarding with Audio calling name			
GC1	Flat Rate Service			
GCE	Call forwarding busy line			
GCJ	Call forwarding don't answer			
GCJRC	Call forwarding don't answer - ring control			
GCZ	Remote access - call forwarding variable			
GJC	Customer control call forwarding don't answer			
GJP	Call control of call forwarding busy line			
HBY	Anonymous Call Rejection			
HTG	Rotary Hunting, business flat rate lines			
HTG1B	Hunting/Rotary/Grouping features used w/Louisiana local optional srvc			
HTGAP	Hunting/Rotary used with Area Plus			
HTGDB	Hunting/Rotary/Grouping features thrifty caller service			
HTGGE	Hunting/Rotary Service, Georgia Community Calling			
HTGKX	Hunting/Rotary Service, Measured Rate MS			
HTGLE	Hunting/Rotary Service, Area Calling Plan MS			
HTGLO	Hunting/Rotary Service, Message Rate MS			
HTGME	Hunting/Rotary Service, Flat Rate MS			
HTGMS	Rotary Hunting, business message rate lines			
HTGNS	Rotary Hunting with EAS Differentials			
HTGTR	Rotary Hunting			
L3B	LOCAL OPTION B GROUPING SERVICE LOCAL OPTIONAL SERVICE, OPTION B, BUSINESS DISTRICT (OPTION 2)			
LPRFX	CHANNELS ACTIVATED: "B" CHANNEL			
LPSMX	ISDN individual business service			
LPWHX	ISDN individual business service			
LPXOX	ISDN individual business service			
LPYLX	ISDN individual business service			
LQ1AA	ISDN individual business service			
LQ1AB	ISDN individual business service			

USOC Code	Description
LQ1AC	ISDN individual business service
LQ1AD	ISDN individual business service
LQ1BA	ISDN individual business service
LQ1BA	BASIC RATE ISDN DSL ARRANGEMENT OPT USAGE PLAN, IND BUSINESS SRVE LINES
LQ1BB	ISDN individual business service
LQ1BC	ISDN individual business service
LQ1BD	ISDN individual business service
LQGFX	ISDN individual business service
LQGFX	ISDN B CHANNEL FLAT RATE
LQGHX	ISDN individual business service
LQGMB	ISDN individual business service
LQGOX	ISDN individual business service
LQRDB	ISDN individual business service
LQRLB	ISDN individual business service
LQTDB	ISDN individual business service
LQTEB	ISDN THRESHOLD PRICING, month to month
LQTHB	ISDN individual business service
LQTTB	ISDN THRESHOLD PRICING B DSL BASIC RATE ISPN ACCESS (5ESS/DMS) BUSINESS, month to month
LTBEB	ISDN BASIC RATE ACCESS ARRANGEMENT, month to month
LTBLB	INDIVIDUAL LINE ISDN BUSINESS, month to month
LTCEB	ISDN individual business service
LTCHC	INDIVIDUAL LINE ISDN BUSINESS PLUS
LTQDX	ISDN individual business service
M1LCA	MULTISERV STATION LINK C
M1LCE	MultiServ, station link
M1LFA	MULTISERV STATION LINE
M1LFH	MultiServ, station link
M1LRA	MULTISERV STATION LINK M
M1LSA	MULTISERV STATION LINK MSG RATE
M1M1A	MultiServ, feature group
M1M52	MultiServ, feature group
M1M56	MultiServ, feature group
M1M59	MultiServ, feature group
M1M5A	MultiServ, feature group
M1MD2	MultiServ, feature group
M1MD3	MultiServ, feature group

USOC Code	Description
M1MD6	MultiServ, feature group
M1MDA	MultiServ, feature group
M1MDB	MultiServ, feature group
M9Q1R	NARS, MultiServ/Centrex, one way inward
M9Q1S	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, one way inward
M9Q1X	NARS, MultiServ/Centrex, one way inward
M9QCR	NARS, MultiServ/Centrex, combination
M9QCS	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, both way
M9QCX	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, both way
M9QOR	NARS, MultiServ/Centrex, one way outward
M9QOS	NARS, MultiServ Plus, one way outward, message rate
M9QOX	NARS, MultiServ/Centrex, one way inward
MNCCF	ISDN individual business service
MNCCR	ISDN individual business service
MWW	Message waiting indication
MWWAV	Message waiting indication audio/visual
NCE	Preferred Call Forwarding - per line
NDT	PBX SVC: DID TRUNK TERMINATION
NSD	Caller ID - Basic per line
NSK	Call Selector - per line
NSQ	Repeat Dialing - per line
NSS	Call Return - Per Line
NST	Call Tracing - per line
NSY	Call Block - per line
NWE1X	COMMUNITY CALLER PLUS SERVICE, NAR, ESSX, Inward only
NWECX	COMMUNITY CALLER PLUS SERVICE, NAR, ESSX, Combination
NWEOX	COMMUNITY CALLER PLUS SERVICE, NAR, ESSX, Outdial
NWO1X	COMMUNITY CALLER PLUS SERVICE, NAR, Inward only
NWOCX	COMMUNITY CALLER PLUS SERVICE, NAR, Combination
NWOOX	COMMUNITY CALLER PLUS SERVICE, NAR, Outdial
NX2	Call Selector, Business, Additional Service, per line
NX5	Call block business and residential service per line
NX6	Preferred call forwarding, business, additional service, per line
NX8	Call return business and residential, additional service per line
NX9	Repeat dial business and residential additional service per line
NXECR	Enhanced Caller ID with ACR - per line

USOC Code	Description			
NXMCR	Caller ID - Deluxe with ACR per line			
NXMMN	Caller ID - Deluxe without ACR per line for multi-line hunt group arrangement			
PBC	Community Caller Plus Service, complex business, without inward call billing			
PBCCL	COMMUNITY CALLER PLUS			
PCE	Community Caller Plus Service, simple business, without inward call billing			
PCX	EXPANDED LOCAL SERVICE, THRIFTY CALLER			
PCXCL	Thrifty caller service, comples business, w/out inward call billing			
PMBCB	Community Circle Plan, message rate service complex business			
PMBSB	Community Circle Plan, message rate service simple business			
PPB	EXPANDED LOCAL SERVICE SIMPLE BUSINESS			
PPBCL	Thrifty Caller Service, simple business, without inward call billing			
PR71C	PRIMARY RATE ISDN B-CHANN			
PR71D	Primary Rate ISDN, interface, data only			
PR71E	PRIMARY RATE ISDN INWARD			
PR71U	PRIMARY RATE ISDN B-CHANN			
PR71V	PRIMARY RATE ISDN VOICE D-CHANNEL INTERFACE, VOICE/DATA STANDARD			
PR7BD	PRIMARY RATE ISDN B-CHANN			
PR7BE	PRIMARY RATE ISDN B-CHANNEL			
PR7BF	PRIMARY RATE ISDN B-CHANNEL			
PR7BL	PRIMARY RATE ISDN B-CHANNEL			
PR7BS	PRIMARY RATE ISDN			
PR7BU	PRIMARY RATE ISDN			
PR7BV	PRIMARY RATE ISDN B-CHANNEL			
RXRTX	MAIN STATION LINE TERMINATION, CENTREX/ESSX OE ONLY TERMINATED AS A PBX TRUNK			
SM1	EXPANDED LOCAL MEASUSRED SERVICE			
SMGP1	SIMPLIFIED MESSAGE DESK INTERFACE (SMDI) PER LINK, ESSA PER LINK, ANALOG/DIGITAL ESSX SERVICE			
SWMCX	EXPANDED LOCAL SERVICE THRIFTY CALLER SERVICE, SHARING/RESALE, MEAS SRVC			
T13CX	REGION SERVICE-BUSINESS TRUNK WITH OUT DISCT COMBINATION			
TACC1	TENNESSEE AREA CALLING PLAN ECONOMY LINE, BUSINESS LINE			
TACC2	TENNESSEE AREA CALLING PLL+ STD BUSINESS LINE			
TACCS	TENNESSEE AREA CALLING PLAN TN AREA CALLING PLAN BOTHWAY, STD, BUSINESS TRUNK			
TAKC1	REGIONSERV BUSINESS INDIV			
TAKCE	REGIONSERV TRUNKS BUSINESS WITH DIST BOTHWAY BUSINESS TRUNK			
TDD1X	PBX SERVICE, DID INWARD TRUNK			
TDDCX	PBX SERVICE, DID Combination			
TFB	PBX SERVICE, COMBINATION BUS FLAT RATE COMBINATION TRUNK			

USOC Code	Description
TFC	PBX SERVICE, COMBINATION BUS FLAT RATE COMBINATION TRUNK
TFN	Business, flat rate inward only trunk
TFU	Business, flat rate inward only trunk
TKG	PBX Service, Measured Rate, Combination
TKV	PBX Service, Measured Rate, Outward Only
TM2	Message rate, PBX trunk, both way, additional
TM3	Message rate, PBX trunk, outdial, additional
TM5	PBX/CENTREX, TOUCH-TONE C, MESSAGE RATE, PBX TRUNK, COMBINATION, ADDTL
TM6	PBX/CENTREX, TOUCH-TONE C, MESSAGE RATE, PBX TRUNK, COMBINATION, ADDTL
TM7	PBX/CENTREX, TOUCH-TONE C, MESSAGE RATE, PBX TRUNK, COMBINATION, ADDTL
TMB	PBX SERVICE, COMBINATION, MSG RATE, BOTH WAY FIRST TRUNK
TMC	PBX SERVICE, COMBINATION, MSG RATE, FIRST TRUNK
TMCGE	PBX SERVICE, Georgia Community Calling, Combination
TMD	PBX SERVICE, inward, first
TMDGE	PBX SERVICE, Georgia Community Calling, Inward
TMU	Message rate, PBX trunk, outdial, first
TMUGE	PBX SERVICE, Georgia Community Calling, Out Dial
TTT	LONG DISTANCE TRUNKS CONN+
TTTXB	LONG DISTANCE TRUNKS
TWC1X	COMMUNITY CALLING PLUS SERVICE, PBX Trunk, Inward Only
TWCCX	COMMUNITY CALLING PLUS SERVICE, PBX Trunk, Combination
TWCND	COMMUNITY CALLING PLUS SERVICE, PBX Trunk, DID Combination
TWCNX	COMMUNITY CALLING PLUS SERVICE, PBX Trunk, DID
TWCOX	COMMUNITY CALLING PLUS SERVICE, PBX Trunk, Outdial

WC CLLI	ZONE	STATE	Rate Group
ALBSALMA	3	AL	6
ALCYALMT	3	AL	4
ALVLALMA	3	AL	4
ANTNALLE	3	AL	6
ANTNALMT	3	AL	6
ANTNALOX	3	AL	6
ATHNALER	3	AL	4
ATHNALMA	3	AL	4
ATTLALNM	3	AL	5
AUBNALMA	3	AL	5
BLFNALMA	3	AL	6
BOAZALMA	3	AL	4
BRHMALCH	1	AL	6
BRHMALCP	1	AL	6
BRHMALEL	1	AL	6
BRHMALEN	1	AL	6
BRHMALEW	1	AL	6
BRHMALFO	1	AL	6
BRHMALFS	1	AL	6
BRHMALHW	1	AL	6
BRHMALMT	1	AL	6
BRHMALOM	1	AL	6
BRHMALOX	1	AL	6
BRHMALRC	1	AL	6
BRHMALTA	1	AL	6
BRHMALVA	1	AL	6
BRHMALWE	1	AL	6
BRHMALWL	1	AL	6
BRPTALMA	3	AL	2
BRTOALMA	3	AL	3
BSMRALBP	3	AL	6
BSMRALBU	3	AL	6
BSMRALHT	3	AL	6
BSMRALMA	3	AL	6
BYMNALMA	3	AL	3
CALRALMA	3	AL	6
CHBGALMA	3	AL	4
CHLSALMA	3	AL	6
CLANALMA	3	AL	3
CLMBALMA	3	AL	6
CLMNALFA	3	AL	5
CLMNALJC	3	AL	5
CLMNALMA	3	AL	5
CNVIALMA	3	AL	2
CRDVALMA	3	AL	4
CRHLALNM	3	AL	4
CRLDALMA	3	AL	5
CTRNALNM	3	AL	6
CYTNALMA	3	AL	3
DCTRALMT	3	AL	6

WC CLLI	ZONE	STATE	Rate Group
DDVLALMA	3	AL	4
DMPLALMA	3	AL	3
DORAALMA	3	AL	6
EUFLALMA	3	AL	3
EUTWALBO	3	AL	2
EUTWALMA	3	AL	2
EVRGALMA	3	AL	2
FLRNALMA	3	AL	6
FMTNALNM	3	AL	3
FRHPALMA	3	AL	6
FTDPALMA	3	AL	1
FTPYALMA	3	AL	3
GDSDALHS	3	AL	5
GDSDALMT	3	AL	5
GDSDALRD	3	AL	5
GDWRALMA	3	AL	4
GNBOALMA	3	AL	2
GRDLALNM	3	AL	6
GRLYALMA	3	AL	6
GTVLALNM	3	AL	4
GYVLALNM	3	AL	6
HLVIALMA	3	AL	6
HNVIALLW	2	AL	6
HNVIALMT	2	AL	6
HNVIALPW	2	AL	6
HNVIALRA	2	AL	6
HNVIALRW	2	AL	6
HNVIALSD	2	AL	6
HNVIALUN	2	AL	6
HNVLALBR	3	AL	5
HNVLALNM	3	AL	5
HRBOALOM	3	AL	4
HRTSALNM	3	AL	5
HRTSALPE	3	AL	5
HZGRALMA	3	AL	6
JCSNALNM	3	AL	2
JCVLALMA	3	AL	6
JSPRALMT	3	AL	4
KLLNALMA	3	AL	6
LFYTALRS	3	AL	1
LGTNALMA	3	AL	6
LNDNALMA	3	AL	3
LVTNALMA	3	AL	2
LXTNALMA	3	AL	6
MARNALNM	3	AL	1
MCINALMA	3	AL	1
MDSNALNM	3	AL	6
MNFDALMA	3	AL	3
MNTVALNM	3	AL	6
MOBLALAP	2	AL	6

WC CLLI	ZONE	STATE	Rate Group
MOBLALAZ	2	AL	6
MOBLALBF	2	AL	6
MOBLALOS	2	AL	6
MOBLALPR	2	AL	6
MOBLALSA	2	AL	6
MOBLALSE	2	AL	6
MOBLALSF	2	AL	6
MOBLALSH	2	AL	6
MOBLALSK	2	AL	6
MOBLALTH	2	AL	6
MOLTALNM	3	AL	5
MPVLALMA	3	AL	4
MTGMALDA	1	AL	6
MTGMALMB	1	AL	6
MTGMALMT	1	AL	6
MTGMALNO	1	AL	6
MTVRALMA	3	AL	6
OHTCALMA	3	AL	6
OPLKALMT	3	AL	5
PDMTALMA	3	AL	6
PHCYALFM	3	AL	6
PHCYALMA	3	AL	6
PNSNALMA	3	AL	6
PRSHALNM	3	AL	4
PRVLALMA	3	AL	6
RDBAALMA	3	AL	3
RLVLALMA	3	AL	6
RRVLALMA	3	AL	6
SELMALMT	3	AL	4
SHFDALMT	3	AL	6
STSNALMA	3	AL	2
SYLCALMT	3	AL	4
THVLALMA	3	AL	2
TLDGALMA	3	AL	4
TLDGALRF	3	AL	4
TROYALMA	3	AL	3
TSCLALDH	3	AL	6
TSCLALMT	3	AL	6
TSCLALNO	3	AL	6
TSKGALMA	3	AL	3
TWCKALMA	3	AL	5
UNTWALNM	3	AL	1
VNCNALMA	3	AL	6
WBTNALNM	3	AL	6
WRRRALNM	3	AL	6
WTMPALMA	3	AL	6
YORKALMA	3	AL	2

WC CLLI	ZONE	SIAIE	Rate Group	
Rate Group: Rates for basic local exchange service are applied on the basis of				
statewide groupings according to the total number of Main Station Lines, PBX				
trunks and Centrex Type Services Main Station Lines within the Local Calling				
Area, including those of other telephone companies within the same Local Calling				
Area.				

WC CLLI	ZONE	STATE	Rate Group
ARCHFLMA	3	FL	5
BCRTFLBT	1	FL	10
BCRTFLMA	1	FL	10
BCRTFLSA	1	FL	10
BGPIFLMA	3	FL	4
BKVLFLJF	3	FL	5
BLDWFLMA	3	FL	9
BLGLFLMA	3	FL	3
BNNLFLMA	3	FL	3
BRSNFLMA	3	FL	5
BYBHFLMA	3	FL	10
CCBHFLAF	3	FL	7
CCBHFLMA	3	FL	7
CDKYFLMA	3	FL	1
CFLDFLMA	3	FL	3
CHPLFLJA	3	FL	3
CNTMFLLE	3	FL	6
COCOFLMA	3	FL	7
COCOFLME	3	FL	7
CSCYFLBA	3	FL	2
DBRYFLDL	3	FL	5
DBRYFLMA	3	FL	5
DELDFLMA	3	FL	5
DLBHFLKP	3	FL	8
DLBHFLMA	3	FL	8
DLSPFLMA	3	FL	4
DNLNFLWM	3	FL	6
DRBHFLMA	1	FL	12
DYBHFLFN	2	FL	6
DYBHFLMA	2	FL	6
DYBHFLOB	2	FL	6
DYBHFLOS	2	FL	6
DYBHFLPO	2	FL	6
EGLLFLBG	3	FL	7
EGLLFLIH	3	FL	7
EORNFLMA	3	FL	11
FLBHFLMA	3	FL	3
FRBHFLFP	3	FL	3
FTGRFLMA	3	FL	9
FTLDFLCR	1	FL	12
FTLDFLCY	1	FL	12
FTLDFLJA	1	FL	12
FTLDFLMR	1	FL	12
FTLDFLOA	1	FL	12
FTLDFLPL	1	FL	12
FTLDFLSG	1	FL	12
FTLDFLSU	1	FL	12
FTLDFLWN	1	FL	12
FTPRFLMA	3	FL	5
GCSPFLCN	3	FL	3

WC CLLI	ZONE	STATE	Rate Group
GCVLFLMA	3	FL	3
GENVFLMA	3	FL	7
GLBRFLMC	3	FL	6
GSVLFLMA	2	FL	6
GSVLFLNW	2	FL	6
HAVNFLMA	3	FL	6
HBSDFLMA	3	FL	6
HLNVFLMA	3	FL	6
HLWDFLHA	2	FL	12
HLWDFLMA	2	FL	12
HLWDFLPE	2	FL	12
HLWDFLWH	2	FL	12
HMSTFLEA	3	FL	12
HMSTFLHM	3	FL	12
HMSTFLNA	3	FL	12
HTISFLMA	3	FL	6
HWTHFLMA	3	FL	5
ISLMFLMA	3	FL	4
JAY FLMA	3	FL	6
JCBHFLAB	1	FL	10
JCBHFLMA	1	FL	9
JCBHFLSP	1	FL	10
JCVLFLAR	1	FL	10
JCVLFLBW	1	FL	10
JCVLFLCL	1	FL	10
JCVLFLFC	1	FL	10
JCVLFLIA	1	FL	10
JCVLFLJT	1	FL	10
JCVLFLLF	1	FL	10
JCVLFLNO	1	FL	10
JCVLFLOW	1	FL	10
JCVLFLRV	1	FL	10
JCVLFLSJ	1	FL	10
JCVLFLSM	1	FL	10
JCVLFLWC	1	FL	10
JPTRFLMA	3	FL	9
KYHGFLMA	3	FL	3
KYLRFLLS	3	FL	4
KYLRFLMA	3	FL	4
KYWSFLMA	3	FL	4
LKCYFLMA	3	FL	4
LKMRFLAB	3	FL	8
LKMRFLMA	3	FL	8
LYHNFLOH	3	FL	5
MCNPFLMA	3	FL	5
MDBGFLPM	3	FL	9
MIAMFLAE	1	FL	12
MIAMFLAL	1	FL	12
MIAMFLAP	1	FL	12
MIAMFLBA	1	FL	12

WC CLLI	ZONE	STATE	Rate Group
MIAMFLBC	1	FL	12
MIAMFLBR	1	FL	12
MIAMFLCA	1	FL	12
MIAMFLFL	1	FL	12
MIAMFLGR	1	FL	12
MIAMFLHL	2	FL	12
MIAMFLIC	1	FL	12
MIAMFLKE	1	FL	12
MIAMFLME	1	FL	12
MIAMFLNM	1	FL	12
MIAMFLNS	1	FL	12
MIAMFLOL	1	FL	12
MIAMFLPB	1	FL	12
MIAMFLPL	1	FL	12
MIAMFLRR	1	FL	12
MIAMFLSH	1	FL	12
MIAMFLSO	1	FL	12
MIAMFLWD	1	FL	12
MIAMFLWM	1	FL	12
MICCFLBB	3	FL	6
MLBRFLMA	3	FL	7
MLTNFLRA	3	FL	6
MNDRFLAV	3	FL	10
MNDRFLLO	1	FL	10
MNDRFLLW	3	FL	9
MNSNFLMA	3	FL	6
MRTHFLVE	3	FL	3
MXVLFLMA	3	FL	9
NDADFLAC	1	FL	12
NDADFLBR	1	FL	12
NDADFLGG	1	FL	12
NDADFLOL	1	FL	12
NKLRFLMA	3	FL	3
NSBHFLMA	3	FL	4
NWBYFLMA	3	FL	5
OKHLFLMA	3	FL	4
OLTWFLLN	3	FL	2
ORLDFLAP	1	FL	11
ORLDFLCL	1	FL	11
ORLDFLMA	1	FL	11
ORLDFLPC	1	FL	11
ORLDFLPH	1	FL	11
ORLDFLSA	1	FL	11
ORPKFLMA	3	FL	9
ORPKFLRW	3	FL	9
OVIDFLCA	3	FL	11
PACEFLPV	3	FL	6
PAHKFLMA	3	FL	3
PCBHFLNT	3	FL FL	<u> </u>
PLCSFLMA	3	FL FL	3
r LUST LIVIA	J	Į FL	J

WC CLLI	ZONE	STATE	Rate Group
PLTKFLMA	3	FL	4
PMBHFLCS	3	FL	12
PMBHFLFE	2	FL	12
PMBHFLMA	2	FL	12
PMBHFLTA	2	FL	12
PMPKFLMA	3	FL	4
PNCYFLCA	3	FL	5
PNCYFLMA	3	FL	5
PNSCFLBL	2	FL	6
PNSCFLFP	2	FL	6
PNSCFLHC	2	FL	6
PNSCFLPB	2	FL	6
PNSCFLWA	2	FL	6
PNVDFLMA	3	FL	9
PRRNFLMA	3	FL	12
PRSNFLFD	3	FL	4
PTSLFLMA	3	FL	6
PTSLFLSO	3	FL	6
SBSTFLFE	3	FL	6
SBSTFLMA	3	FL	6
SGKYFLMA	3	FL	4
SNFRFLMA	3	FL	8
STAGFLBS	3	FL	4
STAGFLMA	3	FL	4
STAGFLSH	3	FL	4
STAGFLWG	3	FL	4
STRTFLMA	3	FL	6
SYHSFLCC	3	FL	3
TRENFLMA	3	FL	5
TTVLFLMA	3	FL	5
VERNFLMA	3	FL	3
VRBHFLBE	1	FL	5
VRBHFLMA	1	FL	5
WELKFLMA	3	FL	4
WPBHFLAN	1	FL	9
WPBHFLGA	1	FL	9
WPBHFLGR	1	FL	9
WPBHFLHH	1	FL	9
WPBHFLLE	1	FL	9
WPBHFLRB	1	FL	9
WPBHFLRP	1	FL	9
WWSPFLHI	3	FL	5
WWSPFLSH	3	FL	5
YNFNFLMA	3	FL	5
YNTWFLMA	3	FL	4
YULEFLMA	3	FL	8

	WC CLLI	ZONE	STATE	Rate Group		
	Rate Group: Rates for basic local exchange service are applied on the basis of					
	statewide groupings according to the total number of exchange access lines, PBX					
	runks and network access registers in the Local Calling Area excluding the					
	Extended Calling Service exchanges.					
ı						

WC CLLI	ZONE	STATE	Rate Group
ACWOGAMA	3	GA	12
AGSTGAAU	2	GA	7
AGSTGAFL	2	GA	7
AGSTGAMT	2	GA	7
AGSTGATH	2	GA	7
AIVLGAMA	3	GA	12
ALBYGAMA	2	GA	5
ALPRGAMA	2	GA	12
AMRCGAMA	3	GA	2
APNGGAES	3	GA	7
ARTNGAES	3	GA	2
ASTLGAMA	3	GA	12
ATHNGAMA	3	GA	5
ATLNGAAD	1	GA	12
ATLNGABH	1	GA	12
ATLNGABU	1	GA	12
ATLNGACD	1	GA	12
ATLNGACS	1	GA	12
ATLNGAEL	1	GA	12
ATLNGAEP	1	GA	12
ATLNGAFP	1	GA	12
ATLNGAGR	1	GA	12
ATLNGAHR	1	GA	12
ATLNGAIC	1	GA	12
ATLNGALA	1	GA	12
ATLNGAPP	1	GA	12
ATLNGASS	1	GA	12
ATLNGATH	1	GA	12
ATLNGAWD	1	GA	12
ATLNGAWE	1	GA	12
BCHNGAES	3	GA	12
BCTNGAMA	3	GA	5
BGRTGAMA	3	GA	12
BLCSGAES	3	GA	2
BNBRGAMA	3	GA	2
BRMNGAES	3	GA	12
BRVIGAMA	3	GA	12
BRWKGAMA	3	GA	2
BUFRGABH	3	GA	12
BWDNGAMA	3	GA	12
BXLYGAES	3	GA	2
CCHRGAMA	3	GA	2
CDTWGAMA	3	GA	12
CHMBGAMA	1	GA	12
CLHNGAES	3	GA	5
CLMBGABV	2	GA	7
CLMBGAMT	2	GA	7
CLMBGAMW	2	GA	7
CLMTGAMA	3	GA	12
CLQTGAES	3	GA	2
014:0/120			

WC CLLI	ZONE	STATE	Rate Group
CMLLGAMA	3	GA	5
CMNGGAMA	3	GA	12
CNCRGAMA	3	GA	12
CNYRGAMA	3	GA GA	12
CORDGAMA	3	GA	2
CRTNGAMA	3	GA	12
CRVLGAMA	3	GA	12
CSSTGAMA	3	GA	7
CVSPGAMA	3	GA	5
CVTNGAMT	3	GA GA	12
CXTNGAMA	3	GA	2
	3		2
DBLNGAMA	3	GA	12
DGVLGAMA		GA	
DLLSGAES	3	GA	12
DLTHGAHS	3	GA	12
DNWDGAMA	1	GA	12
EBTNGAMA	3	GA	2
ESMNGAES	3	GA	2
ETTNGAES	3	GA	2
FKLNGAMA	3	GA	5
FLBRGAMA	3	GA	12
FRBNGAEB	3	GA	12
FRSYGAMA	3	GA	5
FTVYGAMA	3	GA	5
FYVLGASG	3	GA	12
GAY GAMA	3	GA	12
GBSNGAES	3	GA	2
GNBOGAES	3	GA	2
GNVLGAMA	3	GA	5
GRFNGAMA	3	GA	12
GSVLGAMA	3	GA	12
GTVLGAMA	3	GA	12
HGVLGAMA	3	GA	2
HMPNGAJW	3	GA	12
HMTNGAMA	3	GA	7
HPHZGAES	3	GA	7
HRLMGAMA	3	GA	7
HZLHGAMA	3	GA	2
JCSNGAMA	3	GA	12
JESPGAES	3	GA	2
JHCRGAES	3	GA	2
JKISGAMA	3	GA	2
JNBOGAMA	3	GA	12
KGTNGAMA	3	GA GA	12
LERYGAMA	3	GA	5
LGRNGAMA	3	GA	2
LGVLGACS	3	GA	12
LKPKGAMA	3	GA	2
LLBNGAMA	1	GA GA	12
LMCYGAMA	3	GA	2

WC CLLI	ZONE	STATE	Rate Group
LMKNGAMA	3	GA	2
LRVLGAOS	3	GA	12
LSBGGAMA	3	GA	5
LSVLGAMA	3	GA	2
LTHNGAJS	3	GA	12
LTVLGACS	3	GA	12
LULAGAMA	3	GA	12
LYNSGAMA	3	GA	2
MACNGAGP	2	GA	7
MACNGAMT	2	GA GA	7
MACNGAVN	2	GA	7
MCDNGAGS	3	GA GA	12
	3		2
MDSNGAMA		GA	
MLLNGAMA	3	GA	2
MNTIGAMA	3	GA	2
MRRWGAMA	3	GA	12
MRTTGAEA	2	GA	12
MRTTGAMA	2	GA	12
NRCRGAMA	1	GA	12
NWNNGAMA	3	GA	12
NWTNGAHD	3	GA	5
PANLGAMA	3	GA	12
PLHMGAMA	3	GA	2
PLMTGAMA	3	GA	12
PNMTGAMA	3	GA	7
POLRGAMA	3	GA	7
PTCYGAMA	3	GA	12
PWSPGAAS	3	GA	12
RCKMGAES	3	GA	12
RCLDGAMA	3	GA	2
ROMEGATL	3	GA	5
RPVLGAMA	3	GA	12
RSWLGAMA	3	GA	12
RTLGGAMA	3	GA	2
RVDLGAMA	3	GA	12
RYTNGAMA	3	GA	2
SCCRGAMA	3	GA	12
SENOGAMA	3	GA	12
SMVLGAMA	3	GA	5
SMYRGAMA	1	GA	12
SMYRGAPF	1	GA	12
SNLVGAMA	3	GA	12
SNMTGALR	3	GA	12
SNVLGAES	3	GA	2
SPRKGAMA	3	GA	2
SPRTGAMA	3	GA	2
SRDSGAES	3	GA	2
SSISGAES	3	GA	2
STBRGANH	3	GA	12
SVNHGABS	2	GA	7
SVINUGABS		GA	/

WC CLLI	ZONE	STATE	Rate Group
SVNHGADE	2	GA	7
SVNHGAGC	2	GA	7
SVNHGASI	2	GA	7
SVNHGAWB	2	GA	7
SVNHGAWI	2	GA	7
SWBOGAES	3	GA	2
SYLVGAES	3	GA	5
TBISGAMA	3	GA	7
TFTNGAMA	3	GA	2
THSNGAMA	3	GA	2
THVLGAMA	3	GA	2
TLLPGAES	3	GA	12
TMPLGAMA	3	GA	12
TUKRGAMA	1	GA	12
VDALGAMA	3	GA	2
VLDSGAMA	3	GA	2
VLRCGAES	3	GA	12
WDBYGAES	3	GA	2
WDLYGAMA	3	GA	2
WDSTGACR	3	GA	12
WGVLGAES	3	GA	2
WRNSGAMA	3	GA	2
WRRBGAMA	3	GA	7
WRTNGAMA	3	GA	2
WTVLGAES	3	GA	5
WYBOGAES	3	GA	2
WYCRGAMA	3	GA	2
ZBLNGAMA	3	GA	12

**Rate Group:** Rates for basic local exchange service are applied on the basis of statewide groupings according to the total number of exchange access lines and PBX trunks in the Local Calling Area.

ALLNKYMA         3         KY         2           AURRKYMA         3         KY         1           BDFRKYMA         3         KY         1           BGDDKYMA         3         KY         1           BLFDKYMA         3         KY         1           BLSPKYMA         3         KY         1           BNTNKYMA         3         KY         1           BRTNKYMA         3         KY         1           BRMNKYMA         3         KY         1           BRTWKYES         3         KY         1           BVDMKYMA         3         KY         1           BWLGKYMA         3         KY         1           BWLGKYMA         3         KY         1           BWLGKYRWA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLTNKYBA         3         KY         1           CLTNKYES         3         KY         1           CLTNKYMA         3         KY	oup
BDFRKYMA         3         KY         1           BGDDKYMA         3         KY         1           BLFDKYMA         3         KY         1           BLSPKYMA         3         KY         3           BNLYKYMA         3         KY         1           BNTNKYMA         3         KY         1           BRGNKYMA         3         KY         1           BRMNKYMA         3         KY         1           BRTWKYES         3         KY         1           BVDMKYMA         3         KY         1           BWLGKYMA         3         KY         3           BWLGKYRV         3         KY         3           BYVLKYMA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLTNKYMA         3         KY         1           CLTNKYES         3         KY         1           CLTNKYBA         3         KY         1           CNTNKYMA         3         KY         1	
BDFRKYMA         3         KY         1           BGDDKYMA         3         KY         1           BLSPKYMA         3         KY         1           BLSPKYMA         3         KY         1           BNTNKYMA         3         KY         1           BRTNKYMA         3         KY         1           BRMNKYMA         3         KY         1           BRMNKYMA         3         KY         1           BVDMKYMA         3         KY         1           BVDMKYMA         3         KY         1           BWLGKYMA         3         KY         3           BWLGKYRV         3         KY         3           BYVLKYMA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLTNKYMA         3         KY         1           CLTNKYES         3         KY         1           CLTNKYBA         3         KY         1           CNTNKYMA         3         KY         1	
BGDDKYMA         3         KY         1           BLFDKYMA         3         KY         1           BLSPKYMA         3         KY         3           BNTYKYMA         3         KY         1           BRTNKYMA         3         KY         1           BRGNKYMA         3         KY         1           BRMNKYMA         3         KY         1           BRTWKYES         3         KY         1           BVDMKYMA         3         KY         1           BWLGKYMA         3         KY         3           BWLGKYRV         3         KY         3           BYVLKYMA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLTNKYMA         3         KY         1           CLTNKYES         3         KY         1           CLTNKYBA         3         KY         1           CNTNKYMA         3         KY         1           CNTNKYMA         3         KY         1	
BLFDKYMA         3         KY         1           BLSPKYMA         3         KY         3           BNLYKYMA         3         KY         1           BNTNKYMA         3         KY         1           BRGNKYMA         3         KY         1           BRMNKYMA         3         KY         1           BRTWKYES         3         KY         1           BVDMKYMA         3         KY         1           BWLGKYMA         3         KY         3           BWLGKYRV         3         KY         3           BYVLKYMA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLTNKYBS         3         KY         1           CLTNKYBS         3         KY         1           CNTNKYMA         3         KY         1           CNTNKYMA         3         KY         1	
BLSPKYMA         3         KY         3           BNLYKYMA         3         KY         1           BNTNKYMA         3         KY         1           BRGNKYMA         3         KY         1           BRMNKYMA         3         KY         1           BRTWKYES         3         KY         1           BVDMKYMA         3         KY         1           BWLGKYMA         3         KY         3           BWLGKYRV         3         KY         3           BYVLKYMA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLTNKYES         3         KY         1           CLTNKYES         3         KY         1           CNTNKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTNKYMA         3         KY         1	
BNLYKYMA         3         KY         1           BNTNKYMA         3         KY         1           BRGNKYMA         3         KY         1           BRMNKYMA         3         KY         1           BRTWKYES         3         KY         1           BVDMKYMA         3         KY         1           BWLGKYMA         3         KY         3           BWLGKYRV         3         KY         3           BYVLKYMA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLHNKYMA         3         KY         1           CLTNKYMA         3         KY         1           CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTNKYMA         3         KY         1           COTNKYMA         3         KY         3	
BNTNKYMA         3         KY         1           BRGNKYMA         3         KY         1           BRMNKYMA         3         KY         1           BRTWKYES         3         KY         1           BVDMKYMA         3         KY         1           BWLGKYMA         3         KY         3           BWLGKYRV         3         KY         3           BYVLKYMA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLPTKYMA         3         KY         1           CLPTKYMA         3         KY         1           CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTNKYMA         3         KY         1           COTNKYMA         3         KY         3	
BRGNKYMA         3         KY         1           BRMNKYMA         3         KY         1           BRTWKYES         3         KY         1           BVDMKYMA         3         KY         1           BWLGKYMA         3         KY         3           BWLGKYRV         3         KY         3           BYVLKYMA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLHNKYMA         3         KY         1           CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3	
BRMNKYMA         3         KY         1           BRTWKYES         3         KY         1           BVDMKYMA         3         KY         1           BWLGKYMA         3         KY         3           BWLGKYRV         3         KY         3           BYVLKYMA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLHNKYMA         3         KY         1           CLPTKYMA         3         KY         1           CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3	
BRTWKYES         3         KY         1           BVDMKYMA         3         KY         1           BWLGKYMA         3         KY         3           BWLGKYRV         3         KY         3           BYVLKYMA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLHNKYMA         3         KY         1           CLPTKYMA         3         KY         1           CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3	
BVDMKYMA         3         KY         1           BWLGKYMA         3         KY         3           BWLGKYRV         3         KY         3           BYVLKYMA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLHNKYMA         3         KY         1           CLPTKYMA         3         KY         1           CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3	
BWLGKYMA         3         KY         3           BWLGKYRV         3         KY         3           BYVLKYMA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLHNKYMA         3         KY         1           CLPTKYMA         3         KY         1           CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3	
BWLGKYRV         3         KY         3           BYVLKYMA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLHNKYMA         3         KY         1           CLPTKYMA         3         KY         1           CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3	
BYVLKYMA         3         KY         1           CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLHNKYMA         3         KY         1           CLPTKYMA         3         KY         1           CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3	
CADZKYMA         3         KY         1           CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLHNKYMA         3         KY         1           CLPTKYMA         3         KY         1           CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3	
CHPLKYMA         3         KY         1           CLAYKYMA         3         KY         1           CLHNKYMA         3         KY         1           CLPTKYMA         3         KY         1           CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3           COTNKYMA         3         KY         3	
CLAYKYMA         3         KY         1           CLHNKYMA         3         KY         1           CLPTKYMA         3         KY         1           CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3           COTNKYMA         3         KY         3	
CLHNKYMA         3         KY         1           CLPTKYMA         3         KY         1           CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3           COTNKYMA         3         KY         3	
CLPTKYMA         3         KY         1           CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3	
CLTNKYES         3         KY         1           CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3           COTNKYMA         3         KY         3	
CMBGKYMA         3         KY         1           CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3           COTNKYMA         3         KY         3	
CNCYKYMA         3         KY         1           CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3	
CNTNKYMA         3         KY         1           CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3	
CNTWKYMA         3         KY         1           COTNKYMA         3         KY         3	
COTNKYMA 3 KY 3	
CRBOKYMA 3 KY 1	
CRLSKYMA 3 KY 1	
CRTNKYMA 3 KY 1	
CYDNKYMA 3 KY 2	
CYNTKYMA 3 KY 1	
DAVLKYMA 3 KY 1	
DIXNKYMA 3 KY 1	
DRBOKYES 3 KY 1	
DWSPKYES 3 KY 2	
EDVLKYMA 3 KY 1	
EKTNKYMA 3 KY 1	
ELCYKYES 3 KY 2	
EMNNKYES 3 KY 1	
EMNNKYPL 3 KY 1	
ENSRKYMA 3 KY 3	
ERTNKYMA 3 KY 2	
FDCKKYES 3 KY 2	
FDVLKYMA 3 KY 1	
FEBRKYMA 3 KY 1	
FKLNKYMA 3 KY 1	
FLTNKYMA 3 KY 1	
FNVLKYMA 3 KY 1	
FORDKYMA 3 KY 3	

FRDNKYMA	3	KY	1
FRFTKYES	3	KY	3
FRFTKYMA	3	KY	3
GBVLKYMA	3	KY	1
GHNTKYMA	3	KY	1
GNVLKYMA	3	KY	1
GRACKYMA	3	KY	3
GRTWKYMA	3	KY	4
GTHRKYMA	3	KY	1
HABTKYMA	3	KY	3
HANSKYMA	3	KY	2
HBVLKYMA	3	KY	2
HCMNKYMA	3	KY	1
HDBGKYMA	3	KY	1
HNSNKYMA	3	KY	2
HPVLKYMA	3	KY	3
HRBGKYES	3	KY	1
HRFRKYMA	3	KY	1
HRLNKYMA	3	KY	1
HWVLKYMA	3	KY	1
INEZKYMA	3	KY	1
ISLDKYMA	3	KY	1
JCSNKYMA	3	KY	1
JNCYKYMA	3	KY	1
KKVLKYMA	3	KY	3
LBJTKYMA	3	KY	1
LFYTKYMA	3	KY	3
LGRNKYES	3	KY	5
LOUSKYES	3	KY	1
LRBGKYMA	3	KY	1
LSVLKY26	1	KY	5
LSVLKYAN	1	KY	5
LSVLKYAP	1	KY	5
LSVLKYBE	1	KY	5
LSVLKYBR	1	KY	5
LSVLKYCW	1	KY	5
LSVLKYFC	1	KY	5
LSVLKYHA	1	KY	5
LSVLKYJT	1	KY	5
LSVLKYOA	1	KY	5
LSVLKYSH	1	KY	5
LSVLKYSL	1	KY	5
LSVLKYSM	1	KY	5
LSVLKYTS	1	KY	5
LSVLKYVS	1	KY	5
LSVLKYWE	1	KY	5
LVMRKYMA	3	KY	1
MACEKYMA	3	KY	3
MARNKYMA	3	KY	1
MARTKYMA	3	KY	2
MCDNKYMA	3	KY	1

MCWLKYMA         3         KY         2           MDDOKYMA         3         KY         2           MDVIKYMA         3         KY         1           MGFDKYMA         3         KY         1           MGTWKYMA         3         KY         1           MLBGKYMA         3         KY         1           MLTNKYMA         3         KY         1           MRPKYMA         3         KY         2           MRRYKYMA         3         KY         2           MREDKYMA         3         KY         1           MTEDKYMA         3         KY         1           MYFDKYMA         3         KY         1           MYFDKYMA         3         KY         1           MYFDKYMA         3         KY         2           MYVLKYMA         3         KY         1           NEONKYES         3         KY         1           NEONKYES         3         KY         1           NWHNKYMA         3         KY         1           OKGVKYES         3         KY         1           OWBOKYMA         3         KY         1<	
MDVIKYMA         3         KY         2           MGFDKYMA         3         KY         1           MGTWKYMA         3         KY         1           MLBGKYMA         3         KY         1           MLTNKYMA         3         KY         1           MRGPKYMA         3         KY         2           MRRYKYMA         3         KY         2           MRTEDKYMA         3         KY         1           MTEDKYMA         3         KY         1           MTSTKYMA         3         KY         1           MYFDKYMA         3         KY         1           MYPDKYMA         3         KY         1           NEBOKYMA         3         KY         1           NEBOKYMA         3         KY         1           NEONKYES         3         KY         1           NEVLKYMA         3         KY         1           NEVLKYMA         3         KY         4           OWBOKYMA         3         KY         1           OWTNKYMA         3         KY         1           POCHKYIP         3         KY	
MGFDKYMA         3         KY         1           MGTWKYMA         3         KY         1           MLTNKYMA         3         KY         1           MLTNKYMA         3         KY         1           MRGPKYMA         3         KY         2           MRRYKYMA         3         KY         2           MRYENKYMA         3         KY         1           MTSTKYMA         3         KY         1           MYFDKYMA         3         KY         1           MYFDKYMA         3         KY         1           MYFDKYMA         3         KY         1           NEBOKYMA         3         KY         1           NEONKYES         3         KY         1           NEONKYES         3         KY         1           NEVLKYMA         3         KY         1           NEVLKYMA         3         KY         4           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY	
MGTWKYMA         3         KY         1           MLBGKYMA         3         KY         1           MLTNKYMA         3         KY         1           MRGPKYMA         3         KY         2           MRRYKYMA         3         KY         2           MTEDKYMA         3         KY         1           MTSTKYMA         3         KY         1           MYFDKYMA         3         KY         2           MYVLKYMA         3         KY         1           NEBOKYMA         3         KY         1           NEONKYES         3         KY         1           NRVLKYMA         3         KY         1           NRVLKYMA         3         KY         1           NRVLKYMA         3         KY         1           NRVLKYMA         3         KY         4           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYMA         3         KY         3	
MLBGKYMA         3         KY         1           MLTNKYMA         3         KY         1           MRGPKYMA         3         KY         2           MRRYKYMA         3         KY         2           MTEDKYMA         3         KY         1           MTSTKYMA         3         KY         1           MYPDKYMA         3         KY         2           MYVLKYMA         3         KY         1           NEBOKYMA         3         KY         1           NEONKYES         3         KY         1           NRVLKYMA         3         KY         1           NRVLKYMA         3         KY         1           NRVLKYMA         3         KY         1           NRVLKYMA         3         KY         1           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           POCHKYIP         3         KY         3           PDCHKYID         3         KY         3           PDCHKYMA         3         KY         3	
MLTNKYMA         3         KY         1           MRGPKYMA         3         KY         2           MRRYKYMA         3         KY         2           MTEDKYMA         3         KY         1           MTSTKYMA         3         KY         1           MYFDKYMA         3         KY         2           MYVLKYMA         3         KY         1           NEBOKYMA         3         KY         1           NEONKYES         3         KY         1           NRVLKYMA         3         KY         1           NRVLKYMA         3         KY         1           NRVLKYMA         3         KY         1           OKGVKYES         3         KY         4           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYRL         3         KY         3           PDCHKYRL         3         KY         3           PDCHKYMA         3         KY         3	
MRGPKYMA         3         KY         2           MRRYKYMA         3         KY         2           MTEDKYMA         3         KY         1           MTSTKYMA         3         KY         1           MYFDKYMA         3         KY         2           MYVLKYMA         3         KY         1           NEBOKYMA         3         KY         2           NEONKYES         3         KY         1           NRVLKYMA         3         KY         2           NWHNKYMA         3         KY         1           OKGVKYES         3         KY         4           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYIP         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYRL         3         KY         3           PDCHKYMA         3         KY         3           PKVLKYMA         3         KY         3	
MRRYKYMA         3         KY         2           MTEDKYMA         3         KY         1           MTSTKYMA         3         KY         1           MYFDKYMA         3         KY         2           MYVLKYMA         3         KY         1           NEBOKYMA         3         KY         2           NEONKYES         3         KY         1           NRVLKYMA         3         KY         1           NRVLKYMA         3         KY         1           NRVLKYMA         3         KY         1           NRVLKYMA         3         KY         1           OKGVKYES         3         KY         4           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYIP         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYMA         3         KY         3           PKVLKYMA         3         KY         3	
MTEDKYMA         3         KY         1           MTSTKYMA         3         KY         1           MYFDKYMA         3         KY         2           MYVLKYMA         3         KY         1           NEBOKYMA         3         KY         2           NEONKYES         3         KY         1           NRVLKYMA         3         KY         2           NWHNKYMA         3         KY         1           OKGVKYES         3         KY         4           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYID         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYMA         3         KY         3           PIVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PMBRKYMA         3         KY         3	
MTSTKYMA         3         KY         1           MYFDKYMA         3         KY         2           MYVLKYMA         3         KY         1           NEBOKYMA         3         KY         2           NEONKYES         3         KY         1           NRVLKYMA         3         KY         2           NWHNKYMA         3         KY         1           OKGVKYES         3         KY         4           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYLO         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYMA         3         KY         3           PVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PKVLKYMT         3         KY         3           PMBRKYMA         3         KY         3           PNVLKYMA         3         KY         1<	
MYFDKYMA         3         KY         2           MYVLKYMA         3         KY         1           NEBOKYMA         3         KY         2           NEONKYES         3         KY         1           NRVLKYMA         3         KY         2           NWHNKYMA         3         KY         1           OKGVKYES         3         KY         4           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYLO         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYMA         3         KY         3           PIVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PKYLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNVLKYMA         3         KY         1	
MYVLKYMA         3         KY         1           NEBOKYMA         3         KY         2           NEONKYES         3         KY         1           NRVLKYMA         3         KY         2           NWHNKYMA         3         KY         1           OKGVKYES         3         KY         4           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYLO         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYRL         3         KY         3           PIVLKYMA         3         KY         3           PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         1           PRBGKYES         3         KY         2	
NEBOKYMA         3         KY         2           NEONKYES         3         KY         1           NRVLKYMA         3         KY         2           NWHNKYMA         3         KY         1           OKGVKYES         3         KY         4           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYLO         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYRL         3         KY         3           PIVLKYMA         3         KY         3           PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         1           PRBGKYES         3         KY         1           PRBGKYES         3         KY         2	
NEONKYES         3         KY         1           NRVLKYMA         3         KY         2           NWHNKYMA         3         KY         1           OKGVKYES         3         KY         4           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYLO         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYRL         3         KY         3           PIVLKYMA         3         KY         1           PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         1           PRBGKYES         3         KY         2	
NRVLKYMA         3         KY         2           NWHNKYMA         3         KY         1           OKGVKYES         3         KY         4           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYLO         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYRL         3         KY         3           PIVLKYMA         3         KY         1           PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         3           PNULKYMA         3         KY         1           PRBGKYES         3         KY         2	
NWHNKYMA         3         KY         1           OKGVKYES         3         KY         4           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYLO         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYRL         3         KY         3           PIVLKYMA         3         KY         1           PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         3           PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
OKGVKYES         3         KY         4           OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYLO         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYRL         3         KY         3           PIVLKYMA         3         KY         1           PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         3           PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
OWBOKYMA         3         KY         3           OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYLO         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYRL         3         KY         3           PIVLKYMA         3         KY         1           PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         3           PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
OWTNKYMA         3         KY         1           PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYLO         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYRL         3         KY         3           PIVLKYMA         3         KY         1           PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         3           PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
PARSKYMA         3         KY         1           PDCHKYIP         3         KY         3           PDCHKYLO         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYRL         3         KY         3           PIVLKYMA         3         KY         1           PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         3           PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
PDCHKYIP         3         KY         3           PDCHKYLO         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYRL         3         KY         3           PIVLKYMA         3         KY         1           PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         1           PRBGKYES         3         KY         2	
PDCHKYLO         3         KY         3           PDCHKYMA         3         KY         3           PDCHKYRL         3         KY         3           PIVLKYMA         3         KY         1           PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         3           PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
PDCHKYMA         3         KY         3           PDCHKYRL         3         KY         3           PIVLKYMA         3         KY         1           PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         3           PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
PDCHKYRL         3         KY         3           PIVLKYMA         3         KY         1           PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         3           PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
PIVLKYMA         3         KY         1           PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         3           PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
PKVLKYMA         3         KY         3           PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         3           PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
PKVLKYMT         3         KY         3           PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         3           PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
PLRGKYMA         3         KY         3           PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         3           PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
PMBRKYMA         3         KY         3           PNTHKYMA         3         KY         3           PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
PNTHKYMA         3         KY         3           PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
PNVLKYMA         3         KY         1           PRBGKYES         3         KY         2	
PRBGKYES 3 KY 2	
PRTNKYES 3 KY 1	
PRVDKYMA 3 KY 1	
PRVLKYMA 3 KY 1	
PTRYKYMA 3 KY 1	
RBRDKYMA 3 KY 2	
RCMDKYMA 3 KY 3	
RLVLKYMA 3 KY 1	
RSTRKYES 3 KY 3	
SCRMKYMA 3 KY 1	
SDVLKYMA 3 KY 4	
SEBRKYMA 3 KY 1	
SHGVKYMA 3 KY 1	
SHVLKYMA 3 KY 1	
SLGHKYMA 3 KY 1	
SLPHKYMA 3 KY 1	
SLVSKYMA 3 KY 1	
SNTNKYMA 3 KY 1	
SPFDKYMA 3 KY 1	

3	KY	3
3	KY	1
3	KY	2
3	KY	1
3	KY	4
3	KY	3
3	KY	1
	KY	1
3	KY	2
3	KY	1
	KY	1
3	KY	3
3	KY	2
3	KY	3
	KY	1
3	KY	1
3	KY	3
	KY	2
3	KY	1
	KY	3
3	KY	2
3	KY	2
	KY	1
3	KY	1
3	KY	5
3	KY	2
	3 3 3 3 3 3 3 3 3 3 3	3 KY 3 KY 3 KY 3 KY 3 KY 3 KY 3 KY 3 KY

Rate Group: Rates for basic local exchange service are applied on the basis of statewide groupings according to the total number of Main Station Lines (including main station lines, except as provided following), and PBX trunks (including trunks providing Service to U.S. Government-owned systems certified military bases) in the Limited Local Calling Area.

WC CLLI	ZONE	STATE	Rate Group
ABVLLAMA	3	LA	4
ALBYLAMA	3	LA	7
ALXNLADV	3	LA	9
ALXNLAMA	3	LA	9
ALXNLATG	3	LA	9
AMITLAMA	3	LA	1
ANGILAMA	3	LA	6
ARCDLABW	3	LA	1
ARCDLAMA	3	LA	1
BERNLAMA	3	LA	1
BERNLASP	3	LA	1
BGLSLAMA	3	LA	3
BLDWLAMA	3	LA	2
BLNCLAMA	3	LA	9
BNTNLAMA	3	LA	9
BOYCLAMA	3	LA	8
BRSSLAMA	3	LA	9
BSTRLAMA	3	LA	3
BTRGLABK	2	LA	9
BTRGLABS	2	LA	9
BTRGLAGW	2	LA	9
BTRGLAHR	2	LA	9
BTRGLAIS	2	LA	9
BTRGLAMA	2	LA	9
BTRGLAOH	2	LA	9
BTRGLASB	2	LA	9
BTRGLASW	2	LA	9
BTRGLAWN	2	LA	9
BUNKLAMA	3	LA	1
BURSLAMA	3	LA	1
BUSHLAMA	3	LA	3
CASTLAMA	3	LA	1
CHBYLAMA	3	LA	4
CLFXLAMA	3	LA	1
CLHNLAMA	3	LA	9
CLMALAMA	3	LA	1
CLTNLAMA	3	LA	1
CNVLLAMA	3	LA	2
CNVNLAMA	3	LA	3
CNVRLAMA	3	LA	1
CRNCLAMA	3	LA	9
CRWYLAMA	3	LA	3
CSHTLAMA	3	LA	1
CVTNLAMA	3	LA	8
CWVLLAMA	3	LA	1
DBCHLAMA	3	LA	4
DELHLAMA	3	LA	1
DLCXLAMA	3	LA	9
DNSPLAMA	3	LA	9
DNVLLAMA	3	LA	1

DRDRLAMA	3	LA	3
DRPGLAMA	3	LA	1
DULCLAMA	3	LA	8
DUSNLAMA	3	LA	9
DYLNLAMA	3	LA	3
EDGRLAMA	3	LA	1
EPPSLAMA	3	LA	2
ERTHLAMA	3	LA	4
EUNCLAMA	3	LA	2
FKLNLAMA	3	LA	2
FKTNLAMA	3	LA	1
FLRNLAMA	3	LA	1
FLSMLAMA	3	LA	8
FRDYLAMA	3	LA	1
FRVLLADV	3	LA	1
FRVLLAMA	3	LA	1
FTNCLAMA	3	LA	1
GBLDLAMN	3	LA	1
GBSNLAMA	3	LA	7
GNWDLAMA	3	LA	9
GRCNLAMA	3	LA	1
GRNGLAMA	3	LA	5
GRTWLAMA	3	LA	1
GYDNLAMA	3	LA	1
HGTNLAKN	3	LA	9
HGTNLAMA	3	LA	9
HMNDLAMA	3	LA	7
HOMRLAMA	3	LA	1
HOUMLAMA	3	LA	8
HRBGLAMA	3	LA	1
HRNBLAMA	3	LA	1
HYVLLAMA	3	LA	1
INDPLAMA	3	LA	6
JCSNLAMA	3	LA	1
JNBOLAMA	3	LA	1
JNGSLAMA	3	LA	2
JNRTLAMA	3	LA	1
JNVLLAMA	3	LA	1
JSBNLAMA	3	LA	9
KNNRLABR	3	LA	9
KNNRLAHN	3	LA	9
KNWDLAMA	3	LA	1
KRSPLAMA	3	LA	1
KTCHLAMA	3	LA	9
KTVLLAMA	3	LA	9
LBVLLAMA	3	LA	1
LCMBLAMA	3	LA	9
LCMPLAMA	3	LA	9
LCPTLAMA	3	LA	1
LEVLLABF	3	LA	5
LEVLLAFP	3	LA	5

LEVLLAMA	3	LA	5
LEVLLASN	3	LA	5
LFTTLAMA	3	LA	9
LFYTLAMA	2	LA	9
LFYTLAVM	2	LA	9
LGPTLAMA	3	LA	1
LKARLAMA	3	LA	2
LKCHLADT	3	LA	9
LKCHLAMB	3	LA	9
LKCHLAMW	3	LA	9
LKCHLAUN	3	LA	9
LKCTLAMA	3	LA	9
LKPRLAAL	3	LA	1
LKPRLAMA	3	LA	1
LLNGLABU	3	LA	9
LLNGLAHV	3	LA	9
NWORLAAR	1	LA	9
NWORLAAV	1	LA	9
NWORLABM	1	LA	9
NWORLACA	1	LA	9
NWORLACM	1	LA	9
NWORLAFR	1	LA	9
NWORLALK	1	LA	9
NWORLAMA	1	LA	9
NWORLAMC	1	LA	9
NWORLAMR	1	LA	9
NWORLAMT	1	LA	9
NWORLAMU	1	LA	9
NWORLARV	1	LA	9
NWORLASC	1	LA	9
NWORLASK	1	LA	9
NWORLASW	1	LA	9
NWRDLAMA	3	LA	2
NWTNLAMA	3	LA	1
OKDLLAMA	3	LA	1
OKGVLAMA	3	LA	1
OLCYLAMA	3	LA	1
OPLSLATL	3	LA	5
PASNLABV	3	LA	5
PASNLAMN	3	LA	5
PINELAMA	3	LA	1
PLLCLAMA	3	LA	1
PLQMLACR	3	LA	2
PLQMLAMA	3	LA	2
PNALLAMA	3	LA	1
PNCHLAMA	3	LA	7
PRDSLAMA	3	LA	9
PRPRLAMA	3	LA	1
PRRVLAMA		LA	7
PTBRLAMA	3	LA	4
PTSLLAMA	3	LA	1
PNALLAMA PNCHLAMA PRDSLAMA PRPRLAMA PRRVLAMA PTBRLAMA	3 3 3 3 3 3	LA LA LA LA LA	1 7 9 1 7

RAYNLAMA	3	LA	2
RBLNLAMA	3	LA	4
RCLDLAMA	3	LA	1
ROGNLAMA	3	LA	9
RSTNLAMA	3	LA	6
RYVLLAMA	3	LA	1
SALNLAMA	3	LA	1
SCISLAMA	3	LA	1
SFVLLAMA	3	LA	1
SHPTLABS	2	LA	9
SHPTLACL	2	LA	9
SHPTLAHD	2	LA	9
SHPTLAMA	2	LA	9
SHPTLAQB	2	LA	9
SHPTLASG	2	LA	9
SLIDLAMA	3	LA	7
SLPHLAMA	3	LA	9
SMVLLAMA	3	LA	1
SPFDLAMA	3	LA	7
STBRLAMA	3	LA	9
STGBLAMA	3	LA	9
STJSLAMA	3	LA	1
STLNLAMA	3	LA	1
STTNLAMA	3	LA	9
SWLKLAMA	3	LA	9
THBDLAMA	3	LA	4
TLLHLAMA	3	LA	1
TUNCLAMA	3	LA	1
VCHRLAMA	3	LA	1
VDALLAMA	3	LA	5
VENCLAMA	3	LA	1
VNTNLAMA	3	LA	4
WASHLAMA	3	LA	4
WHCSLAMA	3	LA	1
WKISLAMA	3	LA	1
WLSNLAMA	3	LA	1
WNBOLAMA	3	LA	1
WNFDLACA	3	LA	1
WNFDLAMA	3	LA	1
WTPRLAMA	3	LA	1
YNVLLAMA	3	LA	9
YSCLLAMA	3	LA	9
ZCHRLAMA	3	LA	9
ZWLLLAMA	3	LA	1
Data Craum, Dat			liad an that hasis of

Rate Group: Rates for basic local exchange service are applied on the basis of statewide groupings according to the total number of Main Station Lines, PBX trunks and Centrex Type Services Main Station Lines within the Basic Local Calling Area.

WC CLLI	ZONE	STATE	Rate Group
ABRDMSES	3	MS	5
AMRYMSMA	3	MS	5
ASLDMSMA	3	MS	1
BCTNMSMA	3	MS	5
BEMTMSMA	3	MS	2
BENTMSSU	3	MS	4
BGCHMSSU	3	MS	5
BILXMSDI	3	MS	13
BILXMSED	3	MS	13
BILXMSMA	3	MS	13
BKVLMSSU	3	MS	1
BLDWMSMF	3	MS	9
BLMTMSMA	3	MS	3
BLZNMSMA	3	MS	1
BNITMSMA	3	MS	7
BNTNMSSU	3	MS	4
BNVLMSMA	3	MS	3
BOTNMSMA	3	MS	13
BRHNMSMA	3	MS	5
BRNDMSES	3	MS	13
BRVLMSMA	3	MS	3
BRWDMSMA	3	MS	9
BSLSMSMA	3	MS	8
BTVLMSDS	3	MS	4
BUMTMSMA	3	MS	1
CAFBMSMA	3	MS	8
CDWRMSMA	3	MS	6
CFVLMSMA	3	MS	1
CHNKMSSU	3	MS	9
CHTNMSMA	3	MS	1
CLDNMSMA	3	MS	8
CLEVMSMA	3	MS	4
CLMAMSMA	3	MS	4
CLMBMSMA	3	MS	8
CLNSMSMA	3	MS	3
CNTNMSMA	3	MS	3
CNVIMSMA	3	MS	1
COMOMSMA	3	MS	1
COVLMSSU	3	MS	9
CRNSMSMA	3	MS	1
CRNTMSMA	3	MS	6
CRSPMSMA	3	MS	1
CRTHMSMA	3	MS	3
CRTNMSMA	3	MS	6
CSDLMSMA	3	MS	5
CSVLMSSU	3	MS	9
DCHLMSMA	3	MS	2
DFFEMSMA	3	MS	9
DKLBMSMA	3	MS	1
DNCNMSMA	3	MS	1

WC CLLI	ZONE	STATE	Rate Group
DREWMSMA	3	MS	1
DRNTMSMA	3	MS	3
EDWRMSDS	3	MS	13
ELVLMSMA	3	MS	8
ENTRMSMA	3	MS	5
ETHLMSMA	3	MS	4
EUPRMSFA	3	MS	1
FLORMSMA	3	MS	13
FORSMSMA	3	MS	3
FRPNMSMA	3	MS	5
FYTTMSMA	3	MS	1
GDMNMSMA	3	MS	3
GLPTMSLY	3	MS	13
GLPTMSTS	3	MS	13
GLSTMSMA	3	MS	1
GNSNMSMA	3	MS	1
GNVLMSMA	3	MS	8
GNWDMSMA	3	MS	7
GRNDMSMA	3	MS	4
HDLBMSMA	3	MS	1
HLSPMSMA	3	MS	4
HMTNMSSU	3	MS	5
HODLMSMA	3	MS	8
HPVLMSSU	3	MS	3
HRLYMSMA	3	MS	10
HRNNMSDS	3	MS	5
HSTNMSMA	3	MS	3
HTBGMSMA	3	MS	10
HTBGMSWE	3	MS	10
HZLHMSMA	3	MS	1
INDNMSMA	3	MS	3
INDPMSSU	3	MS	6
INVRMSMA	3	MS	3
ITBNMSMA	3	MS	6
IUKAMSES	3	MS	3
JCSNMSBL	1	MS	13
JCSNMSCB	1	MS	13
JCSNMSCP	1	MS	13
JCSNMSMB	1	MS	13
JCSNMSNR	1	MS	13
JCSNMSPC	1	MS	13
JCSNMSRW	1	MS	13
JNTWMSMA	3	MS	5
KLMCMSMA	3	MS	2
KSCSMSMA	3	MS	4
LAKEMSMA	3	MS	3
LARLMSMA	3	MS	8
LBRTMSMA	3	MS	1
LCDLMSMA	3	MS	3
LELDMSMA	3	MS	8

WC CLLI	ZONE	STATE	Rate Group
LENAMSSU	3	MS	3
LMTNMSSS	3	MS	1
LSVLMSMA	3	MS	3
LULAMSMA	3	MS	5
LXTNMSMA	3	MS	3
LYVLMSMA	3	MS	1
MABNMSMA	3	MS	1
MACNMSMA	3	MS	1
MAGEMSMA	3	MS	4
MCCLMSMA	3	MS	4
MCCMMSMA	3	MS	6
MCCMMSSM	3	MS	6
MCLNMSMA	3	MS	1
MDSNMSES	3	MS	13
MGNLMSMA	3	MS	6
MIZEMSMA	3	MS	1
MNASMSMA	3	MS	9
MNDNMSMA	3	MS	4
MNTIMSMA	3	MS	1
MRDNMSTL	3	MS	9
MRHDMSMA	3	MS	3
MRKSMSHW	3	MS	1
MRTNMSMA	3	MS	1
MSPNMSMA	3	MS	10
MTOLMSMA	3	MS	4
NTCHMSMA	3	MS	7
NTTNMSMA	3	MS	9
NWALMSMA	3	MS	4
NWTNMSHC	3	MS	2
NWTNMSMA	3	MS	2
OBDHMSMA	3	MS	9
OCSPMSGO	3	MS	12
OKLDMSMA	3	MS	1
OKLNMSMA	3	MS	3
OSYKMSMA	3	MS	6
OXFRMSMA	3	MS	6
PACEMSMA	3	MS	1
PCKNMSMA	3	MS	3
PCYNMSMA	3	MS	5
PGSNMSMA	3	MS	2
PHLAMSMA	3	MS	4
PLHTMSMA	3	MS	7
PLTNMSMA	3	MS	8
PNTTMSMA	3	MS	3
PPVLMSMA	3	MS	1
PRVSMSMA	3	MS	1
PSCGMSGA	3	MS	10
PSCGMSMA	3	MS	10
PSCHMSLT	3	MS	12
PSCHMSMA	3	MS	12

WC CLLI	ZONE	STATE	Rate Group
PTCMMSSU	3	MS	4
QTMNMSMA	3	MS	5
RCTNMSMA	3	MS	1
RLFKMSMA	3	MS	1
RLGHMSMA	3	MS	1
RLVLMSMA	3	MS	1
ROXIMSMA	3	MS	1
RPLYMSMA	3	MS	3
RSDLMSMA	3	MS	1
RYMNMSDS	3	MS	13
SCOBMSMA	3	MS	1
SHAWMSES	3	MS	1
SHBTMSMA	3	MS	5
SHLBMSDS	3	MS	1
SHNNMSMA	3	MS	9
SHQLMSMA	3	MS	1
SKVLMSMA	3	MS	7
SLCKMSMA	3	MS	1
SLTLMSSU	3	MS	9
SMDLMSSU	3	MS	6
SMNRMSMA	3	MS	1
SMRLMSMA	3	MS	1
SNRYMSMA	3	MS	2
SNSDMSSU	3	MS	7
SNTBMSPS	3	MS	4
SRDSMSMA	3	MS	1
SRISMSMA	3	MS	10
STRGMSSU	3	MS	7
TCHLMSMA	3	MS	3
TMSBMSMA	3	MS	9
TRRYMSMA	3	MS	13
TTWLMSMA	3	MS	1
TUNCMSMA	3	MS	1
TUPLMSMA	3	MS	9
TYTWMSMC	3	MS	2
TYVLMSMA	3	MS	1
UNINMSDS	3	MS	1
UTICMSDS	3	MS	1
VADNMSMA	3	MS	2
VCBGMSMA	3	MS	7
VNCLMSMA	3	MS	12
VRNAMSMA	3	MS	9
WDVLMSMA	3	MS	1
WESTMSMA	3	MS	3
WGNSMSMA	3	MS	2
WINOMSMA	3	MS	2
WLGVMSSU	3	MS	3
WLNTMSMA	3	MS	3
WNRDMSSU	3	MS	2
WSPNMSMA	3	MS	4

WC CLLI	ZONE	STATE	Rate Group
WSSNMSMA	3	MS	1
WTVYMSMA	3	MS	1
WYBOMSMA	3	MS	5
YZCYMSMA	3	MS	4

**Rate Group:** Rates for basic local exchange service are applied on the basis of statewide groupings according to the total Main Station Lines, PBX trunks and Centrex Type Services Main Station Lines in the Local Calling Area.

WC CLLI	ZONE	STATE	Rate Group
ACMENCMA	3	NC	9
AHVLNCBI	2	NC	8
AHVLNCOH	2	NC	8
AHVLNCOT	2	NC	8
APEXNCCE	3	NC	10
ARDNNCCE	3	NC	10
ARSNNCMA	3	NC	7
ATSNNCMA	3	NC	2
BCMTNCCE	3	NC	9
BLMTNCCE	3	NC	10
BLRKNCCE	3	NC	4
BOONNCKI	3	NC	4
BRGWNCMA	3	NC	2
BSCYNCMA	3	NC	7
BURLNCDA	3	NC	7
BURLNCEL	3	NC NC	7
BURLNCHA	3	NC	7
CARYNCCE	3	NC	10
CARYNCWS	3	NC	10
CHRLNCBO	1	NC	10
CHRLNCCA	1	NC	10
CHRLNCCE	1	NC NC	10
CHRLNCCR	1	NC	10
CHRLNCDE	1	NC	10
CHRLNCER	1	NC	10
CHRLNCLP	1	NC	10
CHRLNCMI	1	NC	10
CHRLNCOD	1	NC NC	10
CHRLNCRE	1	NC	10
CHRLNCSH	1	NC	10
CHRLNCTH	1	NC	10
CHRLNCUN	1	NC	10
CHVLNCCE	3	NC	6
CLEVNCMA	3	NC	5
CLMTNCMA	3	NC	7
CLYDNCMA	3	NC	4
CNTNNCMA	3	NC	4
CPHLNCRO	3	NC	5
CRBHNCCE	3	NC	7
CRLNNCMA	3	NC	4
			7
			4
CRENNCMA CSHYNCMA DNVRNCMA DVSNNCPO ELBONCMA ENKANCMA FAMTNCMA FRCYNCCE FRVWNCMA GBSNNCMA GLBONCAD	3 3 3 3 3 3 3 3 3 3 3	NC NC NC NC NC NC NC NC NC NC NC NC NC N	7 10 10

WC CLLI	ZONE	STATE	Rate Group
GLBONCMA	3	NC	6
GNBONCAP	1	NC	10
GNBONCAS	1	NC	10
GNBONCEU	1	NC	10
GNBONCHO	1	NC	10
GNBONCLA	1	NC	10
GNBONCMC	1	NC	10
GNBONCPG	1	NC	10
GNHMNCMA	3	NC	5
GRVRNCMA	3	NC	5
GSTANCDA	3	NC	8
GSTANCSO	3	NC	8
GTWDNCMA	3	NC	5
HMLTNCMA	3	NC	4
HNVLNCCH	3	NC	6
HNVLNCED	3	NC	6
HNVLNCMI	3	NC	6
HSVLNCCE	3	NC	10
JULNNCMA	3	NC	10
KGMTNCMA	3	NC	8
KNDLNCCE	3	NC	10
LCSRNCMA	3	NC	9
LCSTNCMA	3	NC	8
LENRNCHA	3	NC	8
LENRNCHU	3	NC	8
LKLRNCCE	3	NC	6
LMTNNCMA	3	NC	5
LNBHNCMA	3	NC	2
LNTNNCMA	3	NC	5
LNTNNCVA	3	NC	5
LRBGNCMA	3	NC	3
LTMRNCCE	3	NC	5
LWDLNCCE	3	NC	5
LWLLNCMA	3	NC	8
MADNNCCE	3	NC	6
MGTNNCGL	3	NC	5
MGTNNCGR	3	NC	5
MGVANCCE	3	NC	4
MLTNNCMA	3	NC	6
MNTINCMA	3	NC	10
MTHLNCMA	3	NC	10
MTOLNCCE	3	NC	5
NWLDNCCE	3	NC	3
NWTNNCMA	3	NC	7
PMBRNCCE	3	NC	6
RCHMNCMA	3	NC	4
RDVLNCMA	3	NC	5
RDVLNCSI	3	NC	5
RFFNNCMA	3	NC	5
RLGHNCDU	1	NC	10

WC CLLI	ZONE	STATE	Rate Group
RLGHNCGA	1	NC	10
RLGHNCGL	1	NC	10
RLGHNCHO	1	NC	10
RLGHNCJO	1	NC	10
RLGHNCMO	1	NC	10
RLGHNCSB	1	NC	10
RLGHNCSI	1	NC	10
RTTNNCCE	3	NC	5
RWLDNCMA	3	NC	6
SCHLNCHA	3	NC	9
SCHLNCMA	3	NC	7
SELMNCMA	3	NC	10
SHLBNCMA	3	NC	6
SLBRNCMA	3	NC	6
SOPTNCCE	3	NC	2
SPPNNCMA	3	NC	3
SRFDNCCE	3	NC	10
SSVLNCJE	3	NC	5
SSVLNCMA	3	NC	5
STNLNCCE	3	NC	10
STPNNCMA	3	NC	5
SWNNNCMA	3	NC	9
SXPHNCMA	3	NC	9
TRMNNCMA	3	NC	7
TYVLNCMA	3	NC	6
WGVLNCMA	3	NC	7
WLMGNCFO	2	NC	7
WLMGNCLE	2	NC	7
WLMGNCWI	2	NC	7
WNDLNCPI	3	NC	10
WNSLNCAR	2	NC	10
WNSLNCCL	2	NC	10
WNSLNCFI	2	NC	10
WNSLNCGL	2	NC	10
WNSLNCLE	2	NC	10
WNSLNCVI	2	NC	10
WNSLNCWA	2	NC	10
WNSLNCWH	2	NC	10
WYVLNCMA	3	NC	4
ZBLNNCCE	3	NC	10

**Rate Group:** Rates for basic local exchange service are applied on the basis of statewide groupings according to the total number of exchange access lines and PBX trunks in the Local Calling Area.

WC CLLI	ZONE	STATE	Rate Group
AIKNSCMA	3	SC	5
ALDLSCMA	3	SC	2
ARSNSCAH	3	SC	5
ARSNSCMA	3	SC	5
ARSNSCTV	3	SC	5
BATHSCMA	3	SC	7
BAVLSCMA	3	SC	2
BETNSCMA	3	SC	5
BEVLSCMA	3	SC	2
BHISSCMA	3	SC	7
BLBGSCMA	3	SC	3
BLNHSCMA	3	SC	2
BLRGSCMA	3	SC	7
BMBRSCMA	3	SC	4
BRWLSCBE	3	SC	2
BTBGSCMA	3	SC	2
CENTSCWS	3	SC	4
CHAPSCCL	3	SC	7
CHRWSCES	3	SC	2
CHTNSCDP	2	SC	7
CHTNSCDT	2	SC	7
CHTNSCJM	2	SC	7
CHTNSCJN	2	SC	7
CHTNSCLB	2	SC	7
CHTNSCNO	2	SC	7
CHTNSCWA	2	SC	7
CLIOSCMA	3	SC	2
CLMASCAR	1	SC	7
CLMASCBQ	1	SC	•
CLMASCCH	1	SC	7
CLMASCDF	1	SC	7
CLMASCPA	1	SC	7
CLMASCSA	1	SC	7
CLMASCSC	1	SC	7
CLMASCSH	1	SC	7
CLMASCSN	1	SC	7
CLMASCSU	1	SC	7
CLMASCSW	1	SC	7
CLSNSCMA	3	SC	4
CLTNSCMA	3	SC	3
CLVRSCES	3	SC	5
CMDNSCLG	3	SC	3
CMDNSCMA	3	SC	3
CWPNSCMA	3	SC	5
DLLNSCMA	3	SC	2
DNMKSCES	3	SC	2
DRTNSCMA	3	SC	5
EDBHSCMA	3	SC	1
EDFDSCMA	3	SC	1
EOVRSCMA	3	SC	7
LOVINOUNA	1 3	. 50	1

WC CLLI	ZONE	STATE	Rate Group
ESLYSCMA	3	SC	7
FLBHSCMA	3	SC	7
FLRNSCMA	2	SC	6
FNINSCES	3	SC	7
FNVLSCMA	2	SC	6
GFNYSCMA	3	SC	3
GIVLSCMA	3	SC	5
GNVLSCBE	2	SC	7
GNVLSCCH	2	SC	7
GNVLSCCR	2	SC	7
GNVLSCDT	2	SC	7
GNVLSCWE	2	SC	7
GNVLSCWP	2	SC	7
GNVLSCWR	2	SC	7
GRERSCMA	3	SC	7
HCGVSCMA	3	SC	2
HNPHSCMA	3	SC	5
HTVLSCMA	3	SC	5
ISPLSCIS	3	SC	7
JHTNSCMA	3	SC	2
JNVLSCMA	3	SC	2
JONNSCES	3	SC	2
LATTSCLS	3	SC	2
LBRTSCMA	3	SC	4
LKVWSCMA	3	SC	2
LKWLSCRS	3	SC	5
LYMNSCES	3	SC	6
LYMNSCIP	3	SC	6
MARNSCBN	3	SC	5
MARNSCMA	3	SC	5
MCCLSCMA	3	SC	2
MLNSSCWP	3	SC	5
MNPLSCES	3	SC	7
MRTTSCMA	3	SC	7
NAGSSCMA	3	SC	7
NWBYSCMA	3	SC	3
NWELSCMA	3	SC	4
ORBGSCMA	3	SC	4
PCKNSCES	3	SC	4
PCLTSCES	3	SC	5
PDMTSCES	3	SC	7
PNTNSCMA	3	SC	5
PRSRSCMA	3	SC	2
SALMSCMA	3	SC	3
SBRKSCSK	3	SC	7
SCHLSCES	3	SC	5
SENCSCMA	3	SC	4
SHRNSCMA	3	SC	2
SPBGSCBS	2	SC	6
SPBGSCCV	2	SC	6

WC CLLI	ZONE	STATE	Rate Group
SPBGSCHW	2	SC	6
SPBGSCMA	2	SC	6
SPBGSCWV	2	SC	6
SPFDSCMA	3	SC	1
STGRSCMA	3	SC	1
SUVLSCMA	3	SC	7
SXMLSCMA	3	SC	4
TKNASCST	3	SC	4
TMVLSCMA	3	SC	5
TRRSSCMA	3	SC	7
UNINSCMA	JNINSCMA 3 SC	2	
WCLMSCMA	1	SC	7
WHTMSCMA	3	SC	2
WLHLSCES	3	SC	3
WMNSSCES	3	SC	3
WMTNSCPW	3	SC	5
YORKSCMA	3	SC	5

**Rate Group:** Rates for basic local exchange service are applied on the basis of statewide groupings according to the total number of exchange access lines and PBX trunks in the Local Calling Area.

WC CLLI	ZONE	STATE	Rate Group
ACHLTNMT	3	TN	5
ARTNTNMT	3	TN	5
ASCYTNMA	3	TN	5
ATHNTNMA	3	TN	2
BGSNTNMA	3	TN	2
BLGPTNMA	3	TN	3
BLLSTNMA	3	TN	1
BLNCTNMT	3	TN	2
BLVRTNMA	3	TN	1
BNTNTNMT	3	TN	3
BTSPTNMA	3	TN	1
BWVLTNMA	3	TN	1
CHRLTNMT	3	TN	5
CHTGTNDT	2	TN	4
CHTGTNHT	2	TN	4
CHTGTNMV	2	TN	4
CHTGTNNS	2	TN	4
CHTGTNRB	2	TN	4
CHTGTNRO	2	TN	4
CHTGTNSE	2	TN	4
CHTGTNSM	2	TN	4
CHTNTNMT	3	TN	4
CLDGTNMA	3	TN	2
CLEVTNMA	3	TN	4
CLMATNMA	3	TN	2
CLTNTNMA	3	TN	4
CLVLTNMA	3	TN	3
CMCYTNMT	3	TN	1
CMDNTNMA	3	TN	1
CNHMTNMA	3	TN	3
CNVLTNMA	3	TN	1
CRHLTNCB	3	TN	1
CRPLTNMA	3	TN	5
CRTHTNMA	3	TN	1
CRVLTNMA	3	TN	5
CULKTNMA	3	TN	2
CVTNTNMT	3	TN	5
DCTRTNMT	3	TN	4
DKSNTNMT	3	TN	5
DNRGTNMA	3	TN	4
DOVRTNMT	3	TN	1
DYBGTNMA	3	TN	2
DYERTNMT	3	TN	2
DYTNTNMA	3	TN	4
EAVLTNMA	3	TN	5
ETWHTNMT	3	TN	2
FIVLTNMA	3	TN	4
FKLNTNCC	3	TN	5
FKLNTNMA	3	TN	5
FLVLTNMA	3	TN	1

WC CLLI	ZONE	STATE	Rate Group					
FRDNTNMA	3	TN	3					
FRVWTNMT	3	TN	5					
FYVLTNMA	3	TN	1					
GALLTNMA	3	TN	5					
GBSNTNMT	3	TN	2					
GDJTTNMA	3	TN	1					
GDVLTNMA	3	TN	5					
GLSNTNMA	3							
GNBRTNMA	3	TN	5					
GNFDTNMT	3	TN	1					
GRNBTNMA	3	TN	4					
GTBGTNMT	3	TN	4					
GTWSTNSW	1	TN	5					
HDVLTNMA	3	TN	5					
HHNWTNMA	3	TN	1					
HIMNTNMA	3	TN	4					
HLLSTNMT	3	TN	1					
HMBLTNMA	3	TN	2					
HMPSTNMA	3	TN	2					
HNLDTNMA	3	TN	1					
HNNGTNMA	3	TN	1					
HNSNTNMT	3	TN	1					
HNTGTNMA	3	TN	1					
HRFRTNMA	3	TN	1					
HRNBTNMT	3	TN	1					
HTVLTNMA	3	TN	1					
JCSNTNMA	3	TN	3					
JCSNTNNS	3	TN	3					
JFCYTNMA	3	TN	4					
JLLCTNMA	3	TN	3					
JSPRTNMT	3	TN	4					
KGTNTNMT	3	TN	4					
KNTNTNMA	3	TN	2					
KNVLTNBE	2	TN	4					
KNVLTNFC	2	TN	4					
KNVLTNMA	2	TN	4					
KNVLTNWH	2	TN	4					
KNVLTNYH	2	TN	4					
LBNNTNMA	3	TN	5					
LFLTTNMA	3	TN	2					
LKCYTNMA	3	TN	4					
LNCYTNMA	3	TN	4					
LODNTNMA	3	TN	4					
LRBGTNMA	3	TN	2					
LWBGTNMA	3	TN	1					
LXTNTNMA	3	TN	1					
LYBGTNMT	3	TN	1					
LYLSTNMA	3	TN	2					
LYVLTNMA	3	TN	1					
MAVLTNMA	3	TN	4					
	<u> </u>							

WC CLLI	ZONE	STATE	Rate Group
MCKNTNMA	3	TN	1
MCWNTNMT	3	TN	1
MDTNTNMA	3	TN	1
MDVITNMT	3	TN	1
MEDNTNMA	3	TN	2
MILNTNMA	3	TN	2
MMPHTNBA	1	TN	5
MMPHTNCK	1	TN	5
MMPHTNCT	1	TN	5
MMPHTNEL	1	TN	5
MMPHTNFR	1	TN	5
MMPHTNGT	1	TN	5
MMPHTNHP	1	TN	5
MMPHTNMA	1	TN	5
MMPHTNMT	1	TN	5
MMPHTNOA	1	TN	5
MMPHTNSL	1	TN	5
MMPHTNST	1	TN	5
MMPHTNWW	1	TN	5
MNCHTNMA	3	TN	2
MNPLTNMA	3	TN	2
MRBOTNMA	3	TN	5
MRTWTNMA	3	TN	2
MSCTTNMT	3	TN	4
MSCWTNMA	3	TN	5
MYVLTNMA	3	TN	4
NRRSTNMA	3	TN	4
NSVLTNAA	1	TN	5
NSVLTNAP	1	TN	5
NSVLTNBH	1	TN	5
NSVLTNBV	1	TN	5
NSVLTNBW	1	TN	5
NSVLTNCD	1	TN	5
NSVLTNCH	1	TN	5
NSVLTNDO	1	TN	5
NSVLTNHH	1	TN	5
NSVLTNIN	1	TN	5
NSVLTNMC	1	TN	5
NSVLTNMT	1	TN	5
NSVLTNST	1	TN	5
NSVLTNUN	1	TN	5
NSVLTNWC	1	TN	5
NSVLTNWM	1	TN	5
NWBRTNMA	3	TN	2
NWPTTNMT	3	TN	1
OKRGTNMT	3	TN	4
OLHCTNMA	3	TN	5
OLSPTNMA	3	TN	4
PARSTNMA	3	TN	2
PLMYTNMA	3	TN	3
FLIVITITIVIM	ر ا	LIN	ა

WC CLLI	ZONE	STATE	Rate Group
PLSKTNMA	3	TN	1
PSVWTNMT	3	TN	5
PTBGTNMA	3	TN	1
PTLDTNMA	3	TN	5
RDGLTNMA	3	TN	1
RKWDTNMA	3	TN	4
RPLYTNMA	3	TN	1
RRVLTNMA	3	TN	1
SANGTNMT	3	TN	3
SDDSTNMA	3	TN	4
SEWNTNMW	3	TN	1
SHVLTNMA	3	TN	1
SLMRTNMT	3	TN	1
SMTWTNMA	3	TN	2
SMYRTNMA	3	TN	5
SNTFTNMA	3	TN	2
SNVLTNMA	3	TN	1
SOVLTNMT	3	TN	5
SPBGTNMA	3	TN	4
SPCYTNMT	3	TN	4
SPFDTNMA	3	TN	5
SPHLTNMT	3	TN	2
SRVLTNMA	3	TN	1
SVNHTNMT	3	TN	1
SVVLTNMT	3	TN	4
SWTWTNMT	3	TN	1
TLLHTNMA	3	TN	2
TPVLTNMA	3	TN	1
TRINTNMA	3	TN	5
TROYTNMT	3	TN	1
TRTNTNMA	3	TN	2
TWNSTNMA	3	TN	4
UNCYTNMA	3	TN	1
VNLRTNMA	3	TN	5
WHBLTNMT	3	TN	5
WHHSTNMA	3	TN	5
WHPITNMA	3	TN	4
WHVLTNMT	3	TN	1
WHWLTNMA	3	TN	4
WLPTTNMA	3	TN	2
WNCHTNMA	3	TN	1
WRTRTNMT	3	TN	1
WTTWTNMA	3	TN	5
WVRLTNMT	3	TN	1

**Rate Group:** Rates for basic local exchange service are applied on the basis of statewide groupings according to the total number of Main Station Lines, PBX trunks, ESSX-1 Main Service Station Lines and Centrex Type Services Main

USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
113	REGIONSERVE BUSINESS INDI+, 2 WAY																		\$ 24.50	\$ 18.52
113CL	REGIONSERVE-BUS INDIV+W/CALLER ID																		\$ 24.50	\$ 18.52
1FB	Flat rate line, business, two-way	1	\$ 35.76	\$ 29.93	\$ 19.80	\$ 16.47			\$ 35.00	\$ 28.97	\$ 28.68	\$ 22.74	\$ 35.78	\$ 27.13			\$ 32.55	\$ 24.96	\$ 27.05	\$ 20.45
1FB	Flat rate line, business, two-way	2	\$ 36.23	\$ 30.32	\$ 20.80	\$ 17.30	\$ 24.90	\$ 17.50	\$ 35.90	\$ 29.71		\$ 23.28	\$ 36.95	\$ 28.02			\$ 34.25	\$ 26.26	\$ 30.80	\$ 23.28
1FB	Flat rate line, business, two-way	3	\$ 36.23	\$ 30.32	\$ 21.90	\$ 18.22			\$ 35.90	\$ 29.71		\$ 23.82	\$ 38.12	\$ 28.90			\$ 35.95	\$ 27.57	\$ 32.75	\$ 24.76
1FB	Flat rate line, business, two-way	4	\$ 36.23	\$ 30.32	\$ 22.90	\$ 19.05			\$ 35.90	\$ 29.71	\$ 30.72	\$ 24.35	\$ 39.29	\$ 29.79			\$ 37.65	\$ 28.87	\$ 39.05	\$ 27.88
1FB	Flat rate line, business, two-way	5	\$ 36.23	\$ 30.32	\$ 23.85	\$ 19.84	\$ 31.80	\$ 22.35	\$ 35.25	\$ 23.82	\$ 31.40	\$ 24.89	\$ 40.46	\$ 30.68			\$ 39.35	\$ 30.17	\$ 39.70	\$ 26.68
1FB	Flat rate line, business, two-way	6	\$ 36.23	\$ 25.78	\$ 24.90	\$ 19.68					\$ 32.08	\$ 25.43	\$ 41.00	\$ 31.09			\$ 41.05	\$ 31.48		
1FB	Flat rate line, business, two-way	7			\$ 25.75	\$ 21.42	\$ 37.30	\$ 23.14			\$ 32.76	\$ 25.97	\$ 42.00	\$ 31.85			\$ 42.75	\$ 29.14		
1FB	Flat rate line, business, two-way	8			\$ 26.60	\$ 22.13					\$ 32.87	\$ 26.06	\$ 43.00	\$ 32.60						
1FB	Flat rate line, business, two-way	9			\$ 27.40	\$ 20.51					\$ 33.00	\$ 23.55	\$ 43.00	\$ 32.60						
1FB	Flat rate line, business, two-way	10			\$ 28.00	\$ 20.96							\$ 43.00	\$ 32.60						
1FB	Flat rate line, business, two-way	11			\$ 28.60	\$ 21.41														
1FB	Flat rate line, business, two-way	12			\$ 29.10	\$ 21.79	\$ 48.30	\$ 27.96												
1FBCL	Flat Rate business line, caller ID	1	\$ 35.76	\$ 29.93	\$ 19.80	\$ 16.47			\$ 35.00	\$ 28.97	\$ 28.68	\$ 22.74	\$ 35.78	\$ 27.13			\$ 32.55	\$ 24.96	\$ 27.05	\$ 20.45
1FBCL	Flat Rate business line, caller ID	2	\$ 36.23	\$ 30.32	\$ 20.80	\$ 17.30	\$ 24.90	\$ 17.50	\$ 35.90	\$ 29.71	\$ 29.36	\$ 23.28	\$ 36.95	\$ 28.02			\$ 34.25	\$ 26.26	\$ 30.80	\$ 23.28
1FBCL	Flat Rate business line, caller ID	3	\$ 36.23	\$ 30.32	\$ 21.90	\$ 18.22			\$ 35.90	\$ 29.71	\$ 30.04	\$ 23.82	\$ 38.12	\$ 28.90			\$ 35.95	\$ 27.57	\$ 32.75	\$ 24.76
1FBCL	Flat Rate business line, caller ID	4	\$ 36.23	\$ 30.32	\$ 22.90	\$ 19.05			\$ 35.90	\$ 29.71	\$ 30.72	\$ 24.35	\$ 39.29	\$ 29.79			\$ 37.65	\$ 28.87	\$ 39.05	\$ 27.88
1FBCL	Flat Rate business line, caller ID	5	\$ 36.23	\$ 30.32	\$ 23.85	\$ 19.84	\$ 31.80	\$ 22.35	\$ 35.25	\$ 23.82	\$ 31.40	\$ 24.89	\$ 40.46	\$ 30.68			\$ 39.35	\$ 30.17	\$ 39.70	\$ 26.68
1FBCL	Flat Rate business line, caller ID	6	\$ 36.23	\$ 25.78	\$ 24.90	\$ 19.68					\$ 32.08	\$ 25.43	\$ 41.00	\$ 31.09			\$ 41.05	\$ 31.48		
1FBCL	Flat Rate business line, caller ID	7			\$ 25.75	\$ 21.42	\$ 37.30	\$ 23.14			\$ 32.76	\$ 25.97	\$ 42.00	\$ 31.85			\$ 42.75	\$ 29.14		
1FBCL	Flat Rate business line, caller ID	8			\$ 26.60	\$ 22.13					\$ 32.87	\$ 26.06	\$ 43.00	\$ 32.60						
1FBCL	Flat Rate business line, caller ID	9			\$ 27.40	\$ 20.51					\$ 33.00	\$ 23.55	\$ 43.00	\$ 32.60						
1FBCL	Flat Rate business line, caller ID	10			\$ 28.00	\$ 20.96							\$ 43.00	\$ 32.60						
1FBCL	Flat Rate business line, caller ID	11			\$ 28.60	\$ 21.41														
1FBCL	Flat Rate business line, caller ID	12			\$ 29.10	\$ 21.79	\$ 48.30	\$ 27.96												
	PRIMARY RATE ISDN NON-DIS, month to																			
1LD1E	month	1	\$ 140.00	\$ 105.46	\$ 140.00	\$ 104.82	\$ 140.00	\$ 104.20	\$ 140.00	\$ 106.42	\$ 140.00	\$ 99.89	\$ 140.00	\$ 106.16	\$ 140.00	\$ 103.82	Ψ 1.0.00	\$ 107.35	\$ 140.00	\$ 105.84
1MB	Message rate line, business	2			\$ 14.71	\$ 12.24			\$ 26.17	\$ 21.66			\$ 23.19	\$ 17.58	\$ 32.60	\$ 20.15	\$ 25.74	\$ 19.74	\$ 18.95	\$ 14.33
1MB	Message rate line, business				\$ 15.46	\$ 12.86			\$ 28.52	\$ 23.61			\$ 23.97	\$ 18.18			\$ 27.10	\$ 20.78	\$ 21.55	\$ 16.29
1MB	Message rate line, business	3			\$ 16.29	\$ 13.55			\$ 30.52	\$ 25.26			\$ 24.76	\$ 18.77			\$ 28.46	\$ 21.82	\$ 22.95	\$ 17.35
1MB	Message rate line, business	5			\$ 17.04	\$ 14.18			\$ 32.06	\$ 26.54			\$ 25.54	\$ 19.37			\$ 29.82	\$ 22.87	\$ 27.35	\$ 19.53
1MB	Message rate line, business	6			\$ 17.75	\$ 14.77			\$ 38.17	\$ 25.79			\$ 26.32	\$ 19.96			\$ 31.18	\$ 23.91	\$ 27.80	\$ 18.68
1MB	Message rate line, business	7			\$ 18.54	\$ 14.65							\$ 27.11	\$ 20.56			\$ 32.54	\$ 24.95		
1MB	Message rate line, business	8			\$ 19.18	\$ 15.96							\$ 27.89	\$ 21.15			\$ 33.90	\$ 23.11		
1MB	Message rate line, business	9			\$ 19.81	\$ 16.48							\$ 28.68	\$ 21.75						
1MB	Message rate line, business	10			\$ 20.41	\$ 15.28							\$ 29.46	\$ 22.34						
1MB	Message rate line, business	11			\$ 20.86	\$ 15.62							\$ 30.24	\$ 22.93						
1MB	Message rate line, business  Message rate line, business	12			\$ 21.31	\$ 15.96														
1MB		13			\$ 21.69	\$ 16.24							\$ 31.81	\$ 24.12						
1MB	Message rate line, business	13											\$ 31.81	\$ 21.44				2 12 71	10.05	1100
1MBCL	Measure Rate Business Line with Caller ID	2			\$ 14.71	\$ 12.24			\$ 26.17	\$ 21.66			\$ 23.19	\$ 17.58	\$ 32.60	\$ 20.15	\$ 25.74	\$ 19.74	\$ 18.95	\$ 14.33
1MBCL	Measure Rate Business Line with Caller ID	3			\$ 15.46	\$ 12.86			\$ 28.52	\$ 23.61			\$ 23.97	\$ 18.18			\$ 27.10	\$ 20.78	\$ 21.55	\$ 16.29
1MBCL	Measure Rate Business Line with Caller ID				\$ 16.29	\$ 13.55			\$ 30.52	\$ 25.26			\$ 24.76	\$ 18.77			\$ 28.46	\$ 21.82	\$ 22.95	\$ 17.35
1MBCL	Measure Rate Business Line with Caller ID	4 5			\$ 17.04	\$ 14.18			\$ 32.06	\$ 26.54			\$ 25.54	\$ 19.37			\$ 29.82	\$ 22.87	\$ 27.35	\$ 19.53
1MBCL	Measure Rate Business Line with Caller ID	6			\$ 17.75	\$ 14.77			\$ 38.17	\$ 25.79			\$ 26.32	\$ 19.96			\$ 31.18	\$ 23.91	\$ 27.80	\$ 18.68
1MBCL	Measure Rate Business Line with Caller ID				\$ 18.54	\$ 14.65							\$ 27.11	\$ 20.56			\$ 32.54	\$ 24.95		
1MBCL	Measure Rate Business Line with Caller ID	7			\$ 19.18	\$ 15.96							\$ 27.89	\$ 21.15			\$ 33.90	\$ 23.11		
1MBCL	Measure Rate Business Line with Caller ID	8			\$ 19.81	\$ 16.48							\$ 28.68	\$ 21.75						

Attachment 1A Version 1Q01

1

						<b>51.1405</b>			10V = 1V	10/1/0=					No = '''			00.110.7		
USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
1MBCL	Measure Rate Business Line with Caller ID	9			\$ 20.41	\$ 15.28							\$ 29.46	\$ 22.34						
1MBCL	Measure Rate Business Line with Caller ID	10			\$ 20.86	\$ 15.62							\$ 30.24	\$ 22.93						
1MBCL	Measure Rate Business Line with Caller ID	11			\$ 21.31	\$ 15.96														
1MBCL	Measure Rate Business Line with Caller ID	12			\$ 21.69	\$ 16.24							\$ 31.81	\$ 24.12						
1MBGE	Message rate line, business, Georgia Community Calling	2					\$ 20.38	\$ 14.33					\$ 31.81	\$ 21.44						
IIVIDOL	Message rate line, business, Georgia	+ -					ψ 20.30	ψ 14.55					ψ 31.01	ψ 21. <del>44</del>						
1MBGE	Community Calling Message rate line, business, Georgia	5					\$ 25.90	\$ 18.21												
1MBGE	Community Calling	7					\$ 30.30	\$ 18.79												
	Message rate line, business, Georgia	10																		
1MBGE	Community Calling Message rate line, business, Georgia	12					\$ 39.10	\$ 22.63												
1MBGE	Community Calling	12A					\$ 39.98	\$ 23.14												
1MBGE	Message rate line, business, Georgia Community Calling	12B					\$ 39.98	\$ 23.14												
	Message rate line, business, Georgia						y 33.30													
1MBGE	Community Calling Message rate line, business, Georgia	12C					\$ 39.62	\$ 22.94												
1MBGE	Community Calling	12D					\$ 40.50	\$ 23.45												
	Message rate line, business, Georgia	405																		
1MBGE	Community Calling Message rate line, business, Georgia	12E					\$ 40.50	\$ 23.45												
1MBGE	Community Calling	2A					\$ 21.26	\$ 14.94												
1MBGE	Message rate line, business, Georgia Community Calling	2B					\$ 20.90	\$ 14.69												
TIVIDGE	Message rate line, business, Georgia						\$ 20.90	φ 14.09												
1MBGE	Community Calling	2C					\$ 21.78	\$ 15.31												
1MBGE	Message rate line, business, Georgia Community Calling	2D					\$ 21.82	\$ 15.34												
	Message rate line, business, Georgia																			
1MBGE	Community Calling Message rate line, business, Georgia	2E					\$ 20.86	\$ 14.66												
1MBGE	Community Calling	5A					\$ 26.78	\$ 18.83												
1MBGE	Message rate line, business, Georgia Community Calling	5B					\$ 26.94	\$ 18.94												
TWIDGE	Message rate line, business, Georgia						Ψ 20.54	ψ 10.04												
1MBGE	Community Calling Message rate line, business, Georgia	5C					\$ 26.42	\$ 18.57												
1MBGE	Community Calling	5D					\$ 27.30	\$ 19.19												
	Message rate line, business, Georgia	70																		
1MBGE	Community Calling Message rate line, business, Georgia	7A					\$ 34.74	\$ 21.55												
1MBGE	Community Calling	7B					\$ 31.18	\$ 19.34												
1MBGE	Message rate line, business, Georgia Community Calling	7C					\$ 31.66	\$ 19.64												
	Message rate line, business, Georgia						Ψ 01.00													
1MBGE	Community Calling Message rate line, business, Georgia	7D					\$ 30.82	\$ 19.12												
1MBGE	Community Calling	7E					\$ 31.70	\$ 19.66												
	Message rate line, business, Georgia	7F																		
1MBGE	Community Calling BUSINESS MEASURED RATE LINE	/F					\$ 32.38	\$ 20.08												
1MG	BUSINESS, TWO WAY, NON HUNTING	Α															\$ 22.95	\$ 17.60		
1MG	BUSINESS MEASURED RATE LINE BUSINESS, TWO WAY, NON HUNTING	В															\$ 25.35	\$ 19.44		
	BUSINESS MEASURED RATE LINE																			
1MG	BUSINESS, TWO WAY, NON HUNTING BUSINESS MEASURED RATE LINE	С															\$ 27.75	\$ 21.28		
1MG	BUSINESS, TWO WAY, NON HUNTING	D															\$ 30.15	\$ 20.55		
	BUSINESS MEASURED RATE LINE	NA														0 45.5				
1MG	BUSINESS, TWO WAY, NON HUNTING BUSINESS MEASURED RATE LINE	NA													\$ 24.49	\$ 15.13				
1MGCL	BUSINESS, TWO WAY, NON HUNTING	Α															\$ 22.95	\$ 17.60		
1MGCL	BUSINESS MEASURED RATE LINE BUSINESS, TWO WAY, NON HUNTING	В															\$ 25.35	\$ 19.44		
HVIGGE	DOS200, 1110 11111, 11011 11011 1110																ψ 20.05	ψ 15.44		

usoc		_	AL-Tariff	AL-V&T	FL-Tariff	FL-V&T	GA-Tariff	GA-V&T	KY-Tariff	KY-V&T	LA-Tariff	LA-V&T	MS-Tariff		NC-Tariff	NC-V&T	SC-Tariff	SC-V&T	TN-Tariff	TN-V&T
Code	Description BUSINESS MEASURED RATE LINE	Rate Group	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	MS-V&T Rate	Rate	Rate	Rate	Rate	Rate	Rate
1MGCL	BUSINESS, TWO WAY, NON HUNTING	С															\$ 27.75	\$ 21.28		
	BUSINESS MEASURED RATE LINE	D																		
1MGCL	BUSINESS, TWO WAY, NON HUNTING	NA NA															\$ 30.15	\$ 20.55		
1MGCL	Measure Rate Business Line with Caller ID														\$ 24.49	\$ 15.13				
1MH	Measure Rate Business Line with Caller ID	A															\$ 30.29	\$ 23.23		
1MH	Measure Rate Business Line with Caller ID	В															\$ 33.41	\$ 25.62		
1MH	Measure Rate Business Line with Caller ID	С															\$ 36.55	\$ 28.03		
1MH	Measure Rate Business Line with Caller ID BUSINESS MEASURED LINE MEASURE	D															\$ 39.65	\$ 27.03		
1MH	RATE, BUSINESS TWO WAY HUNTING	NA			\$ 19.22	\$ 14.39														
	BUSINESS LINE, ECONOMY SERVICE,																			
1S8	LOCAL OPTION SERVICE RATE, TWO WAY												\$ 29.90	\$ 22.67						
100	BUSINESS LINE, ECONOMY SERVICE,												ψ 25.50	ψ 22.07						
	LOCAL OPTION SERVICE RATE, TWO																			
1S8CL	WAY, with Caller ID BUSINESS LINE STANDARD LOCAL												\$ 29.90	\$ 22.67						
	OPERATIONAL SERVICE RATE,																			
1ZJ	BUSINESS, STD OPTION, TWO WAY BUSINESS LINE STANDARD LOCAL												\$ 29.90	\$ 22.67						
	OPERATIONAL SERVICE RATE,																			
	BUSINESS, STD OPTION, TWO WAY																			
1ZJCL	w/Caller ID												\$ 29.90	\$ 22.67						
3QM1X	ESSX NAR RegionServ, msd rate inward only																		\$ 10.00	\$ 6.72
3QMCX	ESSX NAR RegionServ, msd rate both way																		\$ 10.00	\$ 6.72
3QMOX	ESSX NAR RegionServ, msd rate outward only																		\$ 10.00	\$ 6.72
7FB	Business Line - Auxiliary Line	1				\$ 16.47												\$ 25.73		
7FB	Business Line - Auxiliary Line	2			\$ 20.80	\$ 17.30	\$ 22.60	\$ 15.89									\$ 35.25	\$ 27.03		
7FB	Business Line - Auxiliary Line	3			\$ 21.90	\$ 18.22											\$ 36.95	\$ 28.33		
7FB	Business Line - Auxiliary Line	4			\$ 22.90	\$ 19.05											\$ 38.65	\$ 29.64		
7FB	Business Line - Auxiliary Line	5			\$ 23.85	\$ 19.84	\$ 29.50	\$ 20.74									\$ 40.35	\$ 30.94		
7FB	Business Line - Auxiliary Line	6			\$ 24.90	\$ 19.68											\$ 42.05	\$ 32.24		
7FB	Business Line - Auxiliary Line	7			\$ 25.75	\$ 21.42	\$ 35.00	\$ 21.71									\$ 43.75	\$ 29.82		
7FB	Business Line - Auxiliary Line	8			\$ 26.60	\$ 22.13														
7FB	Business Line - Auxiliary Line	9			\$ 27.40	\$ 20.51														
7FB	Business Line - Auxiliary Line	10			\$ 28.00	\$ 20.96														
7FB	Business Line - Auxiliary Line	11			\$ 28.60	\$ 21.41														
7FB	Business Line - Auxiliary Line	12			\$ 29.10	\$ 21.79	\$ 46.00	\$ 26.63												
7FB	Business Line - Auxiliary Line	12A					\$ 47.10	\$ 27.27												
7FB	Business Line - Auxiliary Line	12B					\$ 47.10	\$ 27.27												
7FB	Business Line - Auxiliary Line	12C					\$ 46.65	\$ 27.01												
7FB	Business Line - Auxiliary Line	12D					\$ 47.75	\$ 27.64												
7FB	Business Line - Auxiliary Line	12E					\$ 47.75	\$ 27.64												
7FB	Business Line - Auxiliary Line	2A					\$ 23.70	\$ 16.66												
7FB	Business Line - Auxiliary Line	2B					\$ 23.25	\$ 16.34												
7FB	Business Line - Auxiliary Line	2C					\$ 24.35	\$ 17.12												
7FB	Business Line - Auxiliary Line	2D					\$ 24.40	\$ 17.15												
7FB	Business Line - Auxiliary Line	2E					\$ 23.20	\$ 16.31												
7FB	Business Line - Auxiliary Line	5A					\$ 30.60	\$ 21.51												
7FB	Business Line - Auxiliary Line	5B					\$ 30.80	\$ 21.65												
7FB	Business Line - Auxiliary Line	5C					\$ 30.15	\$ 21.19												
7FB	Business Line - Auxiliary Line	5D					\$ 31.25	\$ 21.97												

USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
7FB	Business Line - Auxiliary Line	7A					\$ 40.55	\$ 25.15												
7FB	Business Line - Auxiliary Line	7B					\$ 36.10	\$ 22.39												
7FB	Business Line - Auxiliary Line	7C					\$ 36.70	\$ 22.76												
7FB	Business Line - Auxiliary Line	7D					\$ 35.65	\$ 22.11												
7FB	Business Line - Auxiliary Line	7E					\$ 36.75	\$ 22.79												
7FB	Business Line - Auxiliary Line	7F					\$ 37.60	\$ 23.32												
7FBCL	Business Line - Auxiliary Line with Caller ID	1			\$ 19.80	\$ 16.47	ψ 07.00	ψ 20.02									\$ 33.55	\$ 25.73		
7FBCL	Business Line - Auxiliary Line with Caller ID	2			\$ 20.80	\$ 17.30	\$ 22.60	\$ 15.89									\$ 35.25	\$ 27.03		
7FBCL	Business Line - Auxiliary Line with Caller ID	3			\$ 21.90	\$ 18.22	ψ 22.00	ψ 10.00									\$ 36.95	\$ 28.33	$\vdash$	
7FBCL	Business Line - Auxiliary Line with Caller ID	4			\$ 22.90	\$ 19.05											\$ 38.65	\$ 29.64		
7FBCL	Business Line - Auxiliary Line with Caller ID	5			\$ 23.85	\$ 19.84	\$ 29.50	\$ 20.74									\$ 40.35	\$ 30.94		
_	Business Line - Auxiliary Line with Caller ID	6				\$ 19.68	\$ 29.50	\$ 20.74										\$ 32.24		
7FBCL	•	-			\$ 24.90			0 0171									* .=			
7FBCL	Business Line - Auxiliary Line with Caller ID	7			\$ 25.75	\$ 21.42	\$ 35.00	\$ 21.71									\$ 43.75	\$ 29.82		
7FBCL	Business Line - Auxiliary Line with Caller ID	8			\$ 26.60	\$ 22.13													$\vdash$	
7FBCL	Business Line - Auxiliary Line with Caller ID	9			\$ 27.40	\$ 20.51														
7FBCL	Business Line - Auxiliary Line with Caller ID	10			\$ 28.00	\$ 20.96													<u> </u>	
7FBCL	Business Line - Auxiliary Line with Caller ID	11			\$ 28.60	\$ 21.41													<u>'</u>	
7FBCL	Business Line - Auxiliary Line with Caller ID	12			\$ 29.10	\$ 21.79	\$ 46.00	\$ 26.63											'	
	AREA CALLING SERVICE BUSINESS, EXPANDED LOCAL AND CALLING,																			
ACB	ECONOMY OPTION		\$ 32.50	\$ 21.76															,	1
	AREA CALLING SERVICE BUSINESS,																			
ACBCL	EXPANDED LOCAL AND CALLING, ECONOMY OPTION with Caller ID		\$ 32.50	\$ 21.76															,	1
ACBCL	AREA CALLING SERVICE, EXPANDED		\$ 32.50	) \$ 21.76																
	LOCAL AREA CALLING SERVICE, STD																			
ASB	OPTION  Expanded Local Area Calling Service, Standard		\$ 32.50	\$ 21.76															<u> </u>	
ASBCL	Option with Caller ID		\$ 32.50	\$ 21.76															,	1
B1M	Business Line - Measured		\$ 23.55						\$ 26.17	\$ 17.68	\$ 20.06	\$ 14.31	\$ 23.19	\$ 15.63					\$ 18.45	\$ 12.40
B1MCL	Business Line - Measured		\$ 24.55						\$ 26.17	\$ 17.68	\$ 20.06	\$ 14.31	\$ 23.19						\$ 18.45	\$ 12.40
DIIVICE	Flat rate usage charge, business inward (w/		Ψ 24.50	υ 10.44					ψ 20.17	ψ 17.00	ψ 20.00	ψ 14.51	ψ 25.19	ψ 15.05					ψ 10.43	ψ 12.40
B2K1D	LUD)	1							\$ 29.70	\$ 24.58									<u> </u>	
B2K1D	Flat rate usage charge, business inward (w/ LUD)	2							e 20.70	\$ 24.58										
B2K1D	Flat rate usage charge, business inward (w/								\$ 29.70	\$ 24.58										
B2K1D	LUD)	3							\$ 29.70	\$ 24.58										
Baltip	Flat rate usage charge, business inward (w/	4																		
B2K1D	LUD)   Flat rate usage charge, business inward (w/	4							\$ 29.70	\$ 24.58										
B2K1D	LUD)	5							\$ 33.00	\$ 22.30										
B2K1K	Flat rate usage charge, business inward	1							\$ 29.70	\$ 24.58										
B2K1K	Flat rate usage charge, business inward	2							\$ 29.70	\$ 24.58										
B2K1K	Flat rate usage charge, business inward	3							\$ 29.70	\$ 24.58										
B2K1K	Flat rate usage charge, business inward	4							\$ 29.70	\$ 24.58										
B2K1K	Flat rate usage charge, business inward	5							\$ 33.00	\$ 22.30									$\vdash$	
DZIVIIV	Flat rate usage charge, business both way (w/	⊢ Ť							ψ 33.00	Ψ 22.30									$\vdash$	
B2K2D	LUD)	1							\$ 35.70	\$ 29.55										
B2K2D	Flat rate usage charge, business both way (w/ LUD)	2							\$ 35.70	\$ 29.55										
DZNZU	Flat rate usage charge, business both way (w/								ψ 35.70	φ 29.55										
B2K2D	LUD)	3					<u> </u>		\$ 35.70	\$ 29.55	<u> </u>		<u> </u>		<u> </u>		<u> </u>		<u> </u>	
DOLCOR	Flat rate usage charge, business both way (w/	4								<b>6</b> 22 ==										
B2K2D	LUD) Hat rate usage charge, business both way (w/	4							\$ 35.70	\$ 29.55									$\vdash$	
B2K2D	LUD)	5							\$ 38.35	\$ 25.91										
	Flat rate usage charge, business both way (w/o																			
B2K2K	LUD)	1							\$ 32.70	\$ 27.07										

USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
	Flat rate usage charge, business both way (w/o				****						****									
B2K2K	LUD) Flat rate usage charge, business both way (w/o	2							\$ 32.70	\$ 27.07										
B2K2K	LUD)	3							\$ 32.70	\$ 27.07										
B2K2K	Flat rate usage charge, business both way (w/o LUD)	4							\$ 32.70	\$ 27.07										
	Flat rate usage charge, business both way (w/o	-																		
B2K2K	LUD) Flat rate usage charge, business both way,	5							\$ 35.35	\$ 23.89										
B2K2P	premium calling	1							\$ 32.70	\$ 27.07										
B2K2P	Flat rate usage charge, business both way, premium calling	2							\$ 32.70	\$ 27.07										
	Flat rate usage charge, business both way,								-											
B2K2P	premium calling Flat rate usage charge, business both way,	3							\$ 32.70	\$ 27.07										
B2K2P	premium calling	4							\$ 32.70	\$ 27.07										
B2K2P	Flat rate usage charge, business both way, premium calling	5							\$ 35.35	\$ 23.89										
B6P	AREA PLUS SERVICE BUSINESS								,								\$ 33.00	\$ 22.49		
Dobol	AREA PLUS SERVICE BUSINESS WITH CALLER ID																£ 00.00	¢ 00.40		
B6PCL BC1	BUSINESS CHOICE		\$ 81.95	\$ 54.87	\$ 77.95	\$ 58.36	\$ 96.95	\$ 56.12	\$ 76.85	\$ 51.93	\$ 77.95	\$ 55.62	\$ 83.95	\$ 56.58	\$ 77.95	\$ 48.17	\$ 33.00 \$ 86.95	\$ 22.49 \$ 59.27	\$ 77.95	\$ 52.38
BC2	BUSINESS CHOICE		\$ 58.95	\$ 39.47	\$ 48.95	\$ 36.65	\$ 67.95	\$ 39.34	\$ 76.65	\$ 35.71		\$ 38.49		\$ 50.56	\$ 53.95	\$ 33.34	\$ 61.95	\$ 42.23	\$ 58.95	\$ 39.61
	Business Plus Service, Florida, option 1, flat		ψ 30.93	ψ 33.47			Ψ 07.95	ψ 39.34	ψ 32.03	ψ 33.71	ψ 55.95	ψ 30.49	ψ 01.95	\$ 41.75	ψ 55.95	ψ 55.54	ψ 01.95	Ψ 42.23	ψ 30.93	ψ 33.01
BD1	rated plan				\$ 64.10	\$ 47.99														
BD1CL	BUSINESS LINE WITH TOUCHTONE+ Business Plus Service, Florida, option 2,				\$ 64.10	\$ 47.99														
BD2	combination rated plan				\$ 35.10	\$ 26.28														
BD2CL	BUSINESS LINE WITH TOUCH+				\$ 35.10	\$ 26.28														
BF1	Business Plus Service, Alabama, option 1, flat rated plan		\$ 68.00	\$ 45.53																
BF1CL	BUSINESS LINE W/TT OPT 1 W/CALLER ID		\$ 68.00	\$ 45.53																
BF2	Business Plus Service, Alabama, option 2, combination rated plan		\$ 45.00	\$ 30.13																
	BUSINESS LINE WITH TT OPTION 2 WITH																			
BF2CL	CALLER ID  Business Plus Service, Georgia, option 1, flat		\$ 45.00	\$ 30.13																
BG1	rated plan						\$ 83.30	\$ 48.22												
BG1CL	Business Plus Service, Georgia, option 1, flat rated plan with Caller ID						\$ 83.30	\$ 48.22												
DOTOL	Business Plus Service, Georgia, option 2,						ψ 05.50	Ψ 40.22												
BG2	combination rated plan BUSINESS LINE W/TOUCH-TONE W/						\$ 54.30	\$ 31.43												
BG2CL	CALLER ID						\$ 54.30	\$ 31.43												
BK1	Business Plus Service, Kentucky, option 1, flat rated plan								\$ 62.90	\$ 42.50										
	Business Plus Service, Kentucky, option 1, flat								-											
BK1CL	rated plan, with Caller ID Business Plus Service, Kentucky, option 2,								\$ 62.90	\$ 42.50										
BK2	combination rated plan								\$ 38.90	\$ 26.28										
BK2CL	Business Plus Service, Kentucky, option 2, combination rated plan, with Caller ID								\$ 38.90	\$ 26.28										
DNZUL	Business Plus Service, Louisiana, option 1, flat								\$ 30.90	\$ 20.28										
BL1	rated plan										\$ 64.00	\$ 45.67								
BL1CL	Business Plus Service, Louisiana, option 1, flat rated plan, with Caller ID										\$ 64.00	\$ 45.67								
	Business Plus Service, Louisiana, option 2,										e 40.00									
BL2	combination rated plan Business Plus Service, Louisiana, option 2,										\$ 40.00	\$ 28.54								
BL2CL	combination rated plan, with Caller ID										\$ 40.00	\$ 28.54								
BMU	PBX Service, Measured Rate, Inward Only	1	\$ 23.24	\$ 19.46					\$ 23.17	\$ 19.18			\$ 35.78	\$ 27.13					\$ 18.95	\$ 14.33
BMU	PBX Service, Measured Rate, Inward Only	2	\$ 23.55	\$ 19.71					\$ 25.52	\$ 21.12			\$ 36.95	\$ 28.02					\$ 21.55	\$ 16.29
BMU	PBX Service, Measured Rate, Inward Only	3	\$ 23.55	\$ 19.71					\$ 27.52	\$ 22.78			\$ 38.12	\$ 28.90					\$ 22.95	\$ 17.35

USOC			AL-Tariff	AL-V&T	FL-Tariff	FL-V&T	GA-Tariff	GA-V&T	KY-Tariff	KY-V&T	LA-Tariff	LA-V&T	MS-Tarif		NC-Tariff	NC-V&T	SC-Tariff	SC-V&T	TN-Tariff	TN-V&T
Code	Description	Rate Group	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	MS-V&T Rate		Rate	Rate	Rate	Rate	Rate
BMU	PBX Service, Measured Rate, Inward Only	4	\$ 23.55	\$ 19.71					\$ 29.46	\$ 24.38			\$ 39.2	9 \$ 29.79					\$ 27.35	\$ 19.53
BMU	PBX Service, Measured Rate, Inward Only	5	\$ 23.55	\$ 19.71					\$ 35.82	\$ 24.20			\$ 40.4	6 \$ 30.68					\$ 27.80	\$ 18.68
BMU	PBX Service, Measured Rate, Inward Only	6	\$ 23.55	\$ 16.75																
BO8	Local Option service rate, business economy option inward only				\$ 21.90	¢ 16.40														
БОб	Business Plus Service, South Carolina, option				\$ 21.90	\$ 16.40														
BS1	1, flat rated plan																\$ 73.00	\$ 49.76		
BS1CL	Busines Line with Touch+																\$ 73.00	\$ 49.76		
BS2	Business Plus Service, South Carolina, option 2, combination rated plan																\$ 48.75	\$ 33.23		
BS2CL	Busines Line with Touch+																\$ 48.75	\$ 33.23		
50202	Business Plus Service, Tennessee, option 1,																ψ .σσ	ψ 00.20		
BT1	flat rated plan Business Plus Service, Tennessee, option 1,																		\$ 64.00	\$ 43.01
BT1CL	flat rated plan with Caller ID																		\$ 64.00	\$ 43.01
	Business Plus Service, Tennessee, option 2,																			
BT2	combination rated plan																		\$ 45.70	
BT2CL	Busines Line with Touch+ Business Plus Service, Mississippi, option 1,																		\$ 45.70	\$ 30.71
BU1	flat rated plan												\$ 64.0	0 \$ 43.14						
	Business Plus Service, Mississippi, option 1,																			
BU1CL	flat rated plan, with Caller ID Business Plus Service, Mississippi, option 2,												\$ 64.0	0 \$ 43.14						
BU2	combination rated plan												\$ 42.5	0 \$ 28.65						
D. I.O.O.I	Business Plus Service, Mississippi, option 2,																			
BU2CL	combination rated plan, with Caller ID BUSINESS MEASURED LINE FLAT												\$ 42.5	0 \$ 28.65						
BUA	MEASURED BUS LINE, TWO WAY				\$ 24.30	\$ 18.19														
BV1	Business Plus Service, North Carolina, option 1, flat rated plan														\$ 64.00	\$ 39.55				
DVI	Business Plus Service, North Carolina, option														\$ 64.00	\$ 39.55				
BV1CL	1, flat rated plan, with Caller ID														\$ 64.00	\$ 39.55				
	Business Plus Service, North Carolina, option																			
BV2	2, combination rated plan with Caller ID														\$ 40.44	\$ 24.99				
BV2CL	Business Plus Service, North Carolina, option 2, combination rated plan														\$ 40.44	\$ 24.99				
CCS	Class Room Communication														\$ 40.44	\$ 24.99	\$ 8.00	\$ 5.45		
CCS	COMPLETE CHOICE FOR BUSINESS, BUS																\$ 6.00	\$ 5.45		
	PLUS SRVC OPTION 1, 1 LINE PKG OF								_											
COM11	MILEAGE COMPLETE COICE FOR BUS 2-LINE		\$ 87.00	\$ 55.08	\$ 87.00	\$ 61.33	\$ 102.00	\$ 57.50	\$ 81.00	\$ 51.67	\$ 81.00	\$ 54.43	\$ 81.0	0 \$ 51.73	\$ 81.00	\$ 47.96	\$ 87.00	\$ 57.51	\$ 81.00	\$ 51.58
COM12	PACKAGE		\$ 160.00	\$ 100.62	\$ 159.00	\$ 111.42	\$ 194.00	\$ 108.56	\$ 150.00	\$ 95.06	\$ 150.00	\$ 100.19	\$ 148.0	0 \$ 93.89	\$ 154.00	\$ 90.54	\$ 170.00	\$ 111.65	\$ 150.00	\$ 94.88
COM13	Complete Choice for Business																		\$ 260.00	\$ 218.40
COM14	Complete choice for bus, bus pus srvc opt 1		\$ 312.00	\$ 193.59	\$ 305.00	\$ 211.19	\$ 412.00	\$ 227.15	\$ 284.00	\$ 177.58	\$ 312.00	\$ 205.93	\$ 264.0	0 \$ 165.26	\$ 309.00	\$ 179.13	\$ 337.00	\$ 218.46	\$ 335.00	\$ 209.09
COM15	Complete choice for bus, bus pus srvc opt 2		\$ 385.00	\$ 237.28	\$ 371.00	\$ 255.35	\$ 508.00	\$ 277.98	\$ 350.00	\$ 217.37	\$ 384.00	\$ 251.93	\$ 357.0	0 \$ 221.98	\$ 380.00	\$ 218.72	\$ 415.00	\$ 267.26	\$ 424.00	\$ 262.85
COM16	Complete choice for bus, bus pus srvc opt 3		\$ 461.00	\$ 282.19	\$ 440.00	\$ 301.01	\$ 612.00	\$ 332.36	\$ 417.00	\$ 257.22	\$ 461.00	\$ 300.62	\$ 429.0	0 \$ 264.94	\$ 455.00	\$ 260.02	\$ 497.00	\$ 317.95	\$ 512.00	\$ 315.26
COM17	Complete choice for bus, bus pus srvc opt 4		\$ 537.00	\$ 326.46	\$ 508.00	\$ 345.41	\$ 716.00	\$ 385.87	\$ 485.00	\$ 297.11	\$ 538.00	\$ 348.70	\$ 500.0	0 \$ 306.68	\$ 530.00	\$ 300.69	\$ 578.00	\$ 367.31	\$ 599.00	\$ 366.31
COM18	Complete choice for bus, bus pus srvc opt 5		\$ 613.00	\$ 370.10	\$ 575.00	\$ 388.58	\$ 820.00	\$ 438.53	\$ 552.00	\$ 335.83	\$ 616.00	\$ 396.81	\$ 573.0	0 \$ 349.04	\$ 605.00	\$ 340.75	\$ 659.00	\$ 415.97	\$ 687.00	\$ 417.24
COM19	Complete choice for bus, bus pus srvc opt 6		\$ 689.00			\$ 432.53	\$ 924.00	\$ 490.33		\$ 374.58	\$ 693.00		\$ 645.0	-		\$ 380.19	\$ 741.00	\$ 464.58	\$ 775.00	
COM19	Complete choice for bus, bus pus sive opt 6  Complete choice for bus, bus pus sive opt 1		\$ 236.00	•		\$ 432.53 \$ 165.09		\$ 490.33 \$ 171.08		\$ 374.58	\$ 693.00					\$ 137.20		\$ 464.58 \$ 167.05	\$ 775.00	\$ 467.43 \$ 162.11
	Complete Choice for Business		φ 230.00	φ 141.42	\$ 237.00	\$ 39.03	φ 300.00	φ 1/1.08	φ 217.00	φ 130.60	φ 235.00	φ 156.04	φ 201.C	υ φ 12b.b/	φ 235.00	φ 137.20	φ ∠56.00	φ 107.05	φ ∠56.00	φ 10∠.11
COM21 COM22	Complete Choice for Business		<del> </del>			\$ 67.38														
COM22 COM24	Complete Choice for Business  Complete Choice for Business		\$ 205.00	\$ 125.29	\$ 99.00 \$ 182.00	\$ 67.38 \$ 122.36	\$ 288.00	\$ 157.30	\$ 179.00	\$ 109.44	\$ 206.00	\$ 132.87	\$ 198.0	0 \$ 121.22	\$ 205.00	\$ 117.11	\$ 233.00	\$ 148.05	\$ 260.00	\$ 159.96
	Complete Choice for Business  Complete Choice for Business		\$ 205.00	\$ 125.29 \$ 149.32	\$ 182.00 \$ 214.00	\$ 122.36 \$ 142.98	\$ 288.00 \$ 350.00	\$ 157.30 \$ 189.71	\$ 179.00 \$ 215.00	\$ 109.44 \$ 130.55	\$ 206.00	\$ 132.87 \$ 158.33	\$ 198.0	-		\$ 117.11 \$ 139.52	\$ 233.00	\$ 148.05 \$ 177.99	\$ 260.00 \$ 315.00	
COM25 COM26	Complete Choice for Business  Complete Choice for Business		\$ 246.00		-					1	\$ 247.00	_	-						1	\$ 192.48
COM26	Complete Choice for business	1	ф 290.00	\$ 174.81	\$ 247.00	\$ 164.01	\$ 420.00	\$ 225.92	\$ 252.00	\$ 151.95	<b>э</b> 294.00	\$ 187.30	<b>321.0</b>	υ \$ 193.82	⇒ ∠90.00	\$ 163.28	\$ 334.00	\$ 209.39	\$ 378.00	\$ 229.39

USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V8		FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tar Rate	iff	KY-V&T Rate	LA-Tari Rate	iff	LA-V&T Rate	_	S-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
COM27	Complete Choice for Business		\$ 335.00	\$ 200.	53 \$	280.00	\$ 184.75	\$ 490.00	\$ 261.54	\$ 290	.00 \$	173.64	\$ 340.	.00	\$ 215.25	\$	372.00	\$ 223.05	\$ 335.00	\$ 187.24	\$ 386.00	\$ 240.34	\$ 442.00	\$ 266.37
COM28	Complete Choice for Business		\$ 379.00	\$ 225.	29 \$	313.00	\$ 205.22	\$ 560.00	\$ 296.59	\$ 327	.00 \$	194.41	\$ 386.	.00	\$ 242.85	\$	423.00	\$ 251.85	\$ 380.00	\$ 210.83	\$ 439.00	\$ 271.47	\$ 505.00	\$ 302.21
COM29	Complete Choice for Business		\$ 423.00	\$ 249.	67 \$	346.00	\$ 225.42	\$ 630.00	\$ 331.06	\$ 365	.00 \$	215.46	\$ 433.	.00	\$ 270.70	\$	474.00	\$ 280.21	\$ 425.00	\$ 234.04	\$ 490.00	\$ 300.92	\$ 569.00	\$ 338.12
СОМ2Н	Complete choice for bus, bus pus srvc opt 1				\$	147.00	\$ 99.44																	
COMF1	Complete Choice, Flat Rate Business Service							\$ 72.00	\$ 39.03															
COMIT								Ψ 12.00	ψ 00.00		_			1										
COMF2	Complete Choice, Flat Rate Business Service							\$ 134.00	\$ 72.08					_										
COMF3	Complete Choice, Flat Rate Business Service							\$ 218.00	\$ 116.37															
COMF5	Complete Choice, Flat Rate Business Service				\$	218.00	\$ 143.62	\$ 350.00	\$ 183.94															
COMF6	Complete Choice, Flat Rate Business Service				\$	242.00	\$ 158.42	\$ 420.00	\$ 218.99															
COMF7	Complete Choice, Flat Rate Business Service				\$	275.00	\$ 178.88	\$ 490.00	\$ 253.46															
COMF8	Complete Choice, Flat Rate Business Service				\$	308.00	\$ 199.07	\$ 560.00	\$ 287.35															
COMF9	Complete Choice, Flat Rate Business Service				\$	341.00	\$ 218.98	\$ 630.00	\$ 320.67															
COMP1	Complete Choice for Business		\$ 62.00	\$ 38.	67 \$	52.00	\$ 35.12	\$ 72.00	\$ 40.22	\$ 56	.00 \$	34.95	\$ 56.	.00	\$ 36.79	\$	60.00	\$ 37.49	\$ 56.00	\$ 32.68	\$ 66.00	\$ 42.78	\$ 62.00	\$ 38.93
COMP2	Complete Choice for Business		\$ 110.00	\$ 68.	15 \$	94.00	\$ 63.10	\$ 134.00	\$ 74.29	\$ 100	.00 \$	61.99	\$ 100.	.00	\$ 65.29	\$	106.00	\$ 65.79	\$ 104.00	\$ 60.27	\$ 120.00	\$ 77.27	\$ 114.00	\$ 71.09
COMP3	Complete Choice for Business																						\$ 203.00	\$ 170.52
COMPF	Complete Choice for Business Package				\$	177.00	\$ 117.35	\$ 288.00	\$ 152.54															
COMPH	Complete Choice for Business		\$ 161.00	\$ 99.	07 \$	142.00	\$ 94.73	\$ 218.00	\$ 119.97	\$ 142	.00 \$	87.42	\$ 160.	.00	\$ 103.83	\$	156.00	\$ 96.16	\$ 160.00	\$ 92.07	\$ 181.00	\$ 115.78	\$ 197.00	\$ 122.03
E40	Call forwarding variable				\$	7.00	\$ 3.49												\$ 6.40	\$ 2.64	\$ 12.00	\$ 5.62		
	MESSAGE RATE ESSC NAR, One way incoming, EXPANDED LOCAL RATE 3 MILE	_																						
EQ4GE	CALLING (GA) MESSAGE RATE ESSC NAR, One way	2						\$ 17.36	\$ 12.20					4										
EQ4GE	incoming, EXPANDED LOCAL RATE 3 MILE CALLING (GA)	5						\$ 22.66	\$ 15.93															
	MESSAGE RATE ESSC NAR, One way incoming, EXPANDED LOCAL RATE 3 MILE																							
EQ4GE	CALLING (GA) MESSAGE RATE ESSC NAR, One way	7						\$ 26.88	\$ 16.67															
EQ4GE	incoming, EXPANDED LOCAL RATE 3 MILE CALLING (GA)	12						\$ 39.24	\$ 22.72															
	NETWORK ACCESS REGISTER, ESSX,	1	• 040									47.00				_	04.00							
EQA	combination NETWORK ACCESS REGISTER, ESSX,	'	\$ 24.00	\$ 20.	09 \$	11.74	\$ 9.77			\$ 21	.50 \$	17.80	\$ 25.	.08	\$ 19.88	\$	24.00	\$ 18.20			\$ 16.60	\$ 12.73	\$ 52.25	\$ 39.50
EQA	combination	2	\$ 24.00	\$ 20.	09 \$	12.34	\$ 10.27	\$ 17.36	\$ 12.20	\$ 21	.50 \$	17.80	\$ 25.	.08	\$ 19.88	\$	24.00	\$ 18.20			\$ 18.03	\$ 13.83	\$ 52.25	\$ 39.50
EQA	NETWORK ACCESS REGISTER, ESSX, combination	3	\$ 24.00	\$ 20.	09 \$	12.98	\$ 10.80			\$ 21	.50 \$	17.80	\$ 25.	.08	\$ 19.88	\$	24.00	\$ 18.20			\$ 19.46	\$ 14.92	\$ 52.25	\$ 39.50
EQA	NETWORK ACCESS REGISTER, ESSX, combination	4	\$ 24.00	\$ 20.	09 \$	3 13.58	\$ 11.30			\$ 21	.50 \$	17.80	\$ 25.	.08	\$ 19.88	\$	24.00	\$ 18.20			\$ 20.89	\$ 16.02	\$ 52.25	\$ 37.31
EQA	NETWORK ACCESS REGISTER, ESSX, combination	5	\$ 24.00				\$ 11.76	\$ 22.66	\$ 15.93			14.53			\$ 19.88	s	24.00				\$ 22.31	\$ 17.11	\$ 52.25	
EQA	NETWORK ACCESS REGISTER, ESSX, combination	6	\$ 24.00				\$ 11.67	· ==:00	, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,				·			Ť		*			\$ 23.74	\$ 18.20	, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
	NETWORK ACCESS REGISTER, ESSX,		Ψ 21.0	J	<u>.</u>									1										
EQA	combination	7			\$	15.28	\$ 12.71	\$ 26.88	\$ 16.67		_			_							\$ 25.17	\$ 17.16		
EQA		8			\$	15.78	\$ 13.13							_										
EQA		9			\$	16.25	\$ 12.17				_			_		-								
EQA		10			\$	16.60	\$ 12.43				_			_		-								
EQA	NETWORK ACCESS REGISTER, ESSX,	11			\$	16.96	\$ 12.70			-	_			_									-	
EQA	combination  NETWORK ACCESS REGISTER, ESSX,	12			\$	17.26	\$ 12.92	\$ 39.24	\$ 22.72															
EQA	combination  NETWORK ACCESS REGISTER, ESSX,	12A						\$ 41.00	\$ 23.73															
EQA	combination	12B						\$ 41.00	\$ 23.73															

Control   Cont																							
Column   C	USOC	Description	Data Craum	AL-Tarif			FL-Tariff	FL-V&T	GA-Tariff			KY-Tariff	KY-V&T	LA-Tariff	LA-V&T	MS-Tariff	MC VOT Date	NC-Tariff	NC-V&T	SC-Tariff	SC-V&T	TN-Tariff	TN-V&T
Column   C	Code		Rate Group	Rate	Ra	ate	Kate	Rate	Kate	Rat	te	Rate	Rate	Rate	Rate	Rate	WS-V&I Rate	Kate	Rate	Rate	Rate	Rate	Rate
Control   Cont	EQA		12C						\$ 40.2	3 \$ 2	23.32												
Column   C																							
Column   C	EQA		12D						\$ 42.0	\$ 2	24.34												
March   Marc	EΟΛ		12F						¢ 42.0	1 6 2	04.34												i I
Column	LQA		122						Ψ 42.0	+ ψ 2	4.54												
See Controllation	EQA		2A						\$ 19.1	2 \$ 1	3.44												
Manual Content			op.																				
March   Marc	EQA		ZB						\$ 18.4	) \$ 1	2.93												
Fig. 1. STATEMENT CLUSS PRESIDENT CSSS. 25	FOA		2C						\$ 20.1	5 8 1	4 17												i I
EXPONENT ACCESS REQUISITE, ESSA.  ON PRIMARY ACCESS REQUISITE, ESS									Ψ 20.1	, ,													
COMPANIENCE STREET STRE	EQA		2D						\$ 20.2	\$ 1	4.23												i l
FINAL MACAGES RECURSTER, ESSA.  OF INTERMEDIAL PLANE ACCURSTS RECURSTER, ESSA.  OF INTERMEDIAL PLANE ACCURSTS RECURSTER, ESSA.  OF INTERMEDIAL PLANE ACCURSTS RECURSTER, ESSA.  OF INTERMEDIAL PLANE ACCURSTS RECURSTER, ESSA.  OF INTERMEDIAL PLANE ACCURSTS RECURSTER, ESSA.  OF INTERMEDIAL PLANE ACCURSTS RECURSTER, ESSA.  OF INTERMEDIAL PLANE ACCURSTS RECURSTER, ESSA.  OF INTERMEDIAL PLANE ACCURSTS RECURSTER, ESSA.  OF INTERMEDIAL PLANE ACCURSTS RECURSTER, ESSA.  OF INTERMEDIAL PLANE ACCURSTS RECURSTER, ESSA.  OF INTERMEDIAL PLANE ACCURSTS RECURSTS R			٦٦																				
SOL OMINISTANCE PRESIDENT	EQA		ZE						\$ 18.3	2 \$ 1	2.88												$\longrightarrow$
EN MUNICA MULES REGISTER, ESSA, 58	EQA		5A						\$ 24.4	2 \$ 1	7.17												
EMUMOR ACCESS REGISTER, ESSA. Ordination Ord		NETWORK ACCESS REGISTER, ESSX,																					i d
On Manufacing Control And Manufacing Control	EQA		5B						\$ 24.7	\$ 1	7.39												
ECA OF STRUCK ALCESS RESISTER, ESSX.  FOA STRUCK AL	FO4		50						e 22.7		0.00												i I
COM confinition of membrane and provided by the confinition of the con	EQA		30		_				\$ 23.7	) \$ I	0.00												<del></del>
COMMONING MUCHS REUST IN LESS NOT BETWEEN PLUSTERS SERVICE FROM SERVIC	EQA		5D						\$ 25.4	5 \$ 1	7.90												i I
EGA OPINIORIA AUCESS REGISTER, ESSX, 70																							i d
COMPANIENCE SERVICE STRICE SSX, PROPERTY OF THE PROPERTY OF TH	EQA		7A						\$ 35.7	\$ 2	22.18												
EAA Combination FINDMR ACCURS REGISTER, ESSX, FO STANDARD SERVICE STANDARD	FO.4		7D						e 00.0		7.70												i I
COMPANION CLUSS REGISTER, ESSX, OR COMPANION CLUSS	EQA		76						\$ 28.6	1 5 1	7.76												<del></del>
COMPARISON FROM PROMISE REGISTER, ESSX, ORG No. 1 1 5 24.00 \$ 20.00 \$ 11.74 \$ 9.07 \$ 1.02 \$ 1.20 \$ 1	EQA		7C						\$ 29.6	\$ 1	8.36												i I
EQA Combination 7E																							
Combination 7E	EQA		7D						\$ 27.9	2 \$ 1	7.32												
EGA combination 7F   S 41.36 \$ 25.65   S 41.36 \$	FO4		7E						e 20.6		0.44												i I
Combination 7F   S   S   S   S   S   S   S   S   S	EQA		- '-						ş 29.0	)	0.41												$\longrightarrow$
See See See See See See See See See Se	EQA		7F						\$ 41.3	\$ \$ 2	25.65												i I
NETWORK ACCESS REGISTER, ESSX, one way incoming NETWORK ACCESS REGISTER, ESSX, one way incoming NETWORK ACCESS REGISTER, ESSX, one way incoming NETWORK ACCESS REGISTER, ESSX, one way incoming NETWORK ACCESS REGISTER, ESSX, one way incoming network way incoming network way incoming network way incoming network way incoming network way incoming network and the many incoming network way																							i d
EGB way incoming 2 \$ 240 \$ 200 \$ 1234 \$ 10.27 \$ 17.36 \$ 12.20 \$ 21.50 \$ 17.80 \$ 2.00 \$ 18.88 \$ 24.00 \$ 18.20 \$ 18.20 \$ 18.03 \$ 1.88 \$ 5.22 \$ 3.95.0 \$ 1.80 \$ 1.80 \$ 5.25 \$ 3.95.0 \$ 1.80	EQB		1	\$ 24.0	0 \$ :	20.09	\$ 11.74	\$ 9.77				\$ 21.50	\$ 17.80	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 16.60	\$ 12.73	\$ 52.25	\$ 39.50
NETWORK ACCESS REGISTER, ESSX, one way incoming NETWORK ACCESS REGISTER, ESSX, one way incoming way incoming Network ACCESS REGISTER, ESSX, one way incoming Network ACCESS REGISTER, ESSX, one way incoming Network ACCESS REGISTER, ESSX, one way incoming network ACCESS REGISTER, ESSX, one way incoming Network ACCESS REGISTER, ESSX, one way incoming network ACCESS RE	EOR.		2	\$ 240	0 8	20.00	¢ 1234	\$ 10.27	¢ 173	C 1	2 20	¢ 21.50	\$ 17.80	\$ 25.08	¢ 10.88	\$ 24.00	\$ 18.20			\$ 18.03	¢ 13.83	¢ 52.25	\$ 30.50
NETWORK ACCESS REGISTER, ESSX, one way incoming NETWORK ACCESS REGISTER, ESSX, one way incoming NetWORK ACCESS REGISTER, ESSX, one way incoming NetWORK ACCESS REGISTER, ESSX, one way incoming NetWORK ACCESS REGISTER, ESSX, one New your incoming NetWORK ACCESS REGISTER, ESSX, one New Your incoming NetWORK ACCESS REGISTER, ESSX, one NetWORK ACCESS RE	EQB			φ 24.0	U \$ .	20.09	φ 12.3 <del>4</del>	Ф 10.27	\$ 17.3	) \$ I	2.20	φ 21.50	φ 17.00	φ 25.06	φ 19.00	\$ 24.00	\$ 16.20			φ 10.03	ф 13.03	φ 52.25	φ 39.30
EGB way incoming	EQB		3	\$ 24.0	0 \$ :	20.09	\$ 12.98	\$ 10.80				\$ 21.50	\$ 17.80	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 19.46	\$ 14.92	\$ 52.25	\$ 39.50
NETWORK ACCESS REGISTER, ESSX, one way incoming																							
EQB way incoming 5 \$ 24.00 \$ 20.09 \$ 14.14 \$ 11.76 \$ 22.66 \$ 15.93 \$ 21.50 \$ 14.53 \$ 25.08 \$ 19.88 \$ 24.00 \$ 18.20 \$ \$ 22.31 \$ 17.11 \$ 52.25 \$ 35.11 NETWORK AUCESS REGISTER, ESSX, one way incoming and	EQB		4	\$ 24.0	0 \$ :	20.09	\$ 13.58	\$ 11.30				\$ 21.50	\$ 17.80	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 20.89	\$ 16.02	\$ 52.25	\$ 37.31
NETWORK ACCESS REGISTER, ESSX, one	EOR.		5	\$ 240	0 8	20.00	\$ 1414	¢ 11.76	¢ 22.6	C 1	5 03	¢ 21.50	¢ 1/153	\$ 25.08	¢ 10.88	\$ 24.00	\$ 18.20			\$ 22.31	¢ 1711	¢ 52.25	¢ 35.11
NETWORK ACCESS REGISTER, ESSX, one	LQD			Ψ 24.0	υ ψ .	20.03	ψ 14.14	Ψ 11.70	Ψ 22.0	)	0.33	ÿ 21.50	ψ 14.55	ψ 25.00	ψ 13.00	Ψ 24.00	ψ 10.20			Ψ 22.51	Ψ 17.11	ψ 32.23	ψ 33.11
EQB	EQB	way incoming	6	\$ 24.0	0 \$	17.07	\$ 14.77	\$ 11.67												\$ 23.74	\$ 18.20		
NETWORK ACCESS REGISTER, ESSX, one			_																				
Second   S	EQB		7				\$ 15.28	\$ 12.71	\$ 26.8	3 \$ 1	6.67									\$ 25.17	\$ 17.16		
NETWORK ACCESS REGISTER, ESSX, one   9   \$ 16.25 \$ 12.17	FOR		8				\$ 15.78	\$ 13.13															
NETWORK ACCESS REGISTER, ESSX, one   10	-40		<del>                                     </del>				ų 15.70	ψ 10.13															
EQB	EQB		9				\$ 16.25	\$ 12.17															
NETWORK ACCESS REGISTER, ESSX, one   11			40																				
EQB	EQB						\$ 16.60	\$ 12.43															
NETWORK ACCESS REGISTER, ESSX, one   12	FOR						\$ 16.06	\$ 12.70															
EQB	LQD		<u> </u>				ψ 10.30	Ψ 12.70															
EQB         way incoming         12A         \$ 41.00         \$ 23.73           NETWORK ACCESS REGISTER, ESSX, one         12B         \$ 41.00         \$ 23.73           NETWORK ACCESS REGISTER, ESSX, one         12B         \$ 41.00         \$ 23.73	EQB	way incoming	12				\$ 17.26	\$ 12.92	\$ 39.2	\$ 2	22.72												
NETWORK ACCESS REGISTER, ESSX, one			40.4																				
EQB         way incoming         12B         \$ 41.00         \$ 23.73           NETWORK ACCESS REGISTER, ESSX, one         \$ 41.00         \$ 23.73	EQB		12A						\$ 41.0	\$ 2	23.73												
NETWORK ACCESS REGISTER, ESSX, one	FOR		12B						\$ 410	\$ 2	23.73												
EQB   way incoming   12C       \$ 40.28   \$ 23.32	-00								¥ -1.0	Ψ 2	0.70												
	EQB	way incoming	12C						\$ 40.2	3 \$ 2	23.32												

South   Sout																					
MINISTRATION OF PROPERTY CONTINUES AND PROP	USOC	Description	Poto Group		AL-V&T		FL-V&T		GA-V&T	KY-Tariff	KY-V&T	LA-Tariff	LA-V&T		MC VET Date		NC-V&T		SC-V&T		TN-V&T
THE PARTICLE STRUCTURE STR	Code		Rate Group	Kate	Rate	Rate	Rate	Kate	Rate	Rate	Rate	Rate	Rate	Kate	MS-V&I Kate	Rate	Kate	Kate	Kate	Rate	Rate
Second   S	EQB		12D					\$ 42.04	\$ 24.34												
COUNTY PROVINCE SENSE COUNTY PROVINCE SENSE	EOB.		12E					\$ 42.04	\$ 24.34												
12 INJUNE ALLES STREETING LESS OF AUX STREET	LQD							Ψ 42.04	ψ 24.04												
Second Processing	EQB		2A					\$ 19.12	\$ 13.44												
Columbia   Columbia	EOB		2B					\$ 18.40	\$ 12.93												Ì
EXPONENT ACCRESS PRESIDENT, PRISE CON 22		NETWORK ACCESS REGISTER, ESSX, one						-	,												
Column   C	EQB		2C					\$ 20.16	\$ 14.17												
Column   C	EQB		2D					\$ 20.24	\$ 14.23												Ì
REFERENCE ACLESS RECIPIENT, EDS. COM  SET TOWNS ACLESS RECIPIENT, EDS.	505		ar.																		
Separate   Separate	EQB		ZE					\$ 18.32	\$ 12.88												
Second   S	EQB	way incoming	5A					\$ 24.42	\$ 17.17												
EGB NETWORK ACCESS REQUISIER, ESSX, 006  GB NE	EOB		5B					¢ 24.74	¢ 17.20												
ENTRY NUMBER ACCESS REGISTER, ESSA, 509  608  809 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  70 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  70 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  70 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  70 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  70 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  70 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  70 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  70 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  70 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  71 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  72 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  73 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  74 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  75 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  76 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  77 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  78 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  79 morning  Fig. 1908 AND ACCESS REGISTER, ESSA, 509  Fig. 19	EQB		00					\$ 24.74	φ 17.39												
Second   S	EQB		5C					\$ 23.70	\$ 16.66												
EGB May incoming  FINAL PRINCIPLE SESS, COM  FOR TOWARK ACCURS SERGISTER, ESSS, COM  F	EOB		5D					\$ 25.46	\$ 17.90												Ì
EGB May incoming:  EGB May incom	LQD							ψ 20.40	ψ 17.50												
See   See	EQB	., 5	7A					\$ 35.76	\$ 22.18												
EN FYVORK ACLESS REGISTER, ESSX, 006  608 way incoming  609 more of the control o	EQB		7B					\$ 28.64	\$ 17.76												Ì
NETWORK ACCESS REGISTER, ESSX, one Company of Company o			_																		
May Incoming   May	EQB	,g	7C					\$ 29.60	\$ 18.36												
Way Incoming   FE	EQB		7D					\$ 27.92	\$ 17.32												
RETWORK AULESS REGISTER, ESSK, one Color way outgoing a support of the Color way outgoing a support of																					
EGB   way incoming   7F	EQB	,g	/E					\$ 29.68	\$ 18.41												<del>                                     </del>
EQC Way outgoing EQC WAY OUTGO EQC WAY OUTGO EQC WAY OUTGO EQC WAY OUTGO EQC WAY OUTGO EQC WAY OUTGO EQC WAY OUTGO EQC WAY OUTGO EQC WAY OUTGO EQC WAY OUTGO EQC WAY OUTGO EQC WAY OUTGO EQC WAY OUTGO EQC WAY OUTGO EQC WAY OUTGO EQC WAY OUTGO EQC	EQB	way incoming	7F					\$ 41.36	\$ 25.65												
NETWORK ACLESS REGISTER, ESSX, one 2 2 \$ 24.00 \$ 20.09 \$ 12.34 \$ 10.27 \$ 17.36 \$ 12.20 \$ 21.50 \$ 17.80 \$ 25.08 \$ 19.88 \$ 24.00 \$ 18.20 \$ 18.00 \$ 18.00 \$ \$ 1	F00		1	¢ 24.00	¢ 20.00	¢ 11.74	¢ 0.77			e 24 E0	¢ 17.00	¢ 25.00	¢ 40.00	¢ 24.00	e 10.20			£ 40.00	¢ 40.70	€ F2.2F	¢ 20.50
NE INVIKE AUCESS REUSI ELESS, one EQC  Way outgoing  EQC  Way outgoing	EQC		'	\$ 24.00	\$ 20.08	\$ 11.74	\$ 9.77			\$ 21.50	\$ 17.00	\$ 25.08	ф 19.00	\$ 24.00	\$ 16.20			\$ 16.60	\$ 12.73	\$ 52.25	\$ 39.50
EQC way outgoing 3 \$ 24.00 \$ 20.00 \$ 12.98 \$ 10.80 \$ \$ 21.50 \$ 17.80 \$ 25.08 \$ 19.88 \$ 24.00 \$ 18.20 \$ 19.46 \$ 14.02 \$ 52.25 \$ 39.50 \$ 19.00 \$ 10.00 \$	EQC		2	\$ 24.00	\$ 20.09	\$ 12.34	\$ 10.27	\$ 17.36	\$ 12.20	\$ 21.50	\$ 17.80	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 18.03	\$ 13.83	\$ 52.25	\$ 39.50
NETWORK ACCESS REGISTER, ESSX, one EQC way outgoing 120 w	FOC		3	\$ 24.00	\$ 20.00	\$ 12.98	\$ 10.80			\$ 21.50	\$ 17.80	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 1946	\$ 14.92	\$ 52.25	\$ 39.50
NETWORK ACCESS REGISTER, ESSX, one ECC Way outgoing NETWORK ACCESS REGISTER, ESSX, one way outgoing NetWork ACCESS REGISTER, ESSX, one way outgoing NetWork ACCESS REGISTER, ESSX, one way outgoing NetWork ACCESS REGISTER, ESSX, one way outgoing NetWork ACCESS REGISTER, ESSX, one way outgoing NetWork ACCESS REGISTER, ESSX, one way outgoing NetWork ACCESS REGISTER, ESSX, one way outgoing NetWork ACCESS REGISTER, ESSX, one way outgoing NetWork ACCESS REGISTER, ESSX, one way outgoing NetWork ACCESS REGISTER, ESSX, one way outgoing NetWork ACCESS REGISTER, ESSX, one way outgoing NetWork ACCESS REGISTER, ESSX, one way outgoing NetWork ACCESS REGISTER, ESSX, one way outgoing NetWork ACCESS REGISTER, E		NETWORK ACCESS REGISTER, ESSX, one	_	Ψ 21.00	Ψ 20.00	Ψ 12.00	ψ 10.00			ψ 21.00	Ψ 11.00	Ψ 20.00	ψ 10.00	Ų 2oo	Ų 10.20			ψ 10.10	Ψ 1.1.02	ψ 02.20	ψ 00.00
EQC	EQC		4	\$ 24.00	\$ 20.09	\$ 13.58	\$ 11.30			\$ 21.50	\$ 17.80	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 20.89	\$ 16.02	\$ 52.25	\$ 37.31
EQC   way outgoing   6   \$ 24.00   \$ 17.07   \$ 14.77   \$ 11.67	EQC		5	\$ 24.00	\$ 20.09	\$ 14.14	\$ 11.76	\$ 22.66	\$ 15.93	\$ 21.50	\$ 14.53	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 22.31	\$ 17.11	\$ 52.25	\$ 35.11
NETWORK ACCESS REGISTER, ESSX, one			_																		
EQC	EQC	, , ,	6	\$ 24.00	\$ 17.07	\$ 14.77	\$ 11.67											\$ 23.74	\$ 18.20		
EQC way outgoing 8 \$ \$ 15.78 \$ 13.13 \$ \$ \$ \$ 15.78 \$ \$ 13.13 \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	EQC	way outgoing	7					\$ 26.88	\$ 16.67									\$ 25.17	\$ 17.16		
NETWORK ACCESS REGISTER, ESSX, one   9			0																		
NETWORK ACCESS REGISTER, ESSX, one   10   \$ 16.60 \$ 12.43	EQC		٥			\$ 15.78	\$ 13.13														
EQC   way outgoing   10   \$ 16.80   \$ 12.43	EQC	way outgoing	9			\$ 16.25	\$ 12.17														
NETWORK ACCESS REGISTER, ESSX, one   11	FOC		10			\$ 16.60	¢ 12.42														Ì
NETWORK ACCESS REGISTER, ESSX, one   12   \$ 39.24 \$ 22.72	LQU					ψ 10.00	ψ 12.43														
EQC   way outgoing   12   \$ 39.24 \$ 22.72	EQC		11			\$ 16.96	\$ 12.70														
NETWORK ACCESS REGISTER, ESSX, one   12A	FOC		12					\$ 39.24	\$ 22.72												
NETWORK ACCESS REGISTER, ESSX, one   12B		NETWORK ACCESS REGISTER, ESSX, one																			
EQC         way outgoing         12B         \$ 41.00 \$ 23.73           NETWORK ACCESS REGISTER, ESSX, one         12C         \$ 40.28 \$ 23.32           NETWORK ACCESS REGISTER, ESSX, one         12C         \$ 40.28 \$ 23.32	EQC		12A					\$ 41.00	\$ 23.73												
NETWORK ACCESS REGISTER, ESSX, one EQC way outgoing 12C \$ 40.28 \$ 23.32  NETWORK ACCESS REGISTER, ESSX, one	EQC		12B					\$ 41.00	\$ 23.73												
NETWORK ACCESS REGISTER, ESSX, one			400																		
	EQC		12C					\$ 40.28	\$ 23.32												
EQC   way outgoing   12D     \$ 42.04   \$ 24.34	EQC	way outgoing	12D					\$ 42.04	\$ 24.34												

usoc			AL-Tariff	AL-V&T	FL-Tariff	FL-V&T	GA-Tariff	GA-V&T	KY-Tariff	KY-V&T	LA-Tariff	LA-V&T	MS-Tariff		NC-Tariff	NC-V&T	SC-Tariff	SC-V&T	TN-Tariff	TN-V&T
Code	Description	Rate Group	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	MS-V&T Rate	Rate	Rate	Rate	Rate	Rate	Rate
EQC	NETWORK ACCESS REGISTER, ESSX, one way outgoing	12E					\$ 42.04	\$ 24.34												
EQC	NETWORK ACCESS REGISTER, ESSX, one	122					\$ 42.04	φ 24.34												
EQC	way outgoing	2A					\$ 19.12	\$ 13.44												
	NETWORK ACCESS REGISTER, ESSX, one																			
EQC	way outgoing	2B					\$ 18.40	\$ 12.93												ı
	NETWORK ACCESS REGISTER, ESSX, one	2C																		
EQC	way outgoing NETWORK ACCESS REGISTER, ESSX, one	20					\$ 20.16	\$ 14.17												
EQC	way outgoing	2D					\$ 20.24	\$ 14.23												
LQU	NETWORK ACCESS REGISTER, ESSX, one	<del>                                     </del>					Ψ 20.24	ψ 14.20												<del></del>
EQC	way outgoing	2E					\$ 18.32	\$ 12.88												1
	NETWORK ACCESS REGISTER, ESSX, one																			
EQC	way outgoing	5A					\$ 24.42	\$ 17.17												
	NETWORK ACCESS REGISTER, ESSX, one	- FD																		1
EQC	way outgoing NETWORK ACCESS REGISTER, ESSX, one	5B					\$ 24.74	\$ 17.39												
EQC	way outgoing	5C					\$ 23.70	\$ 16.66												1
LQU	NETWORK ACCESS REGISTER, ESSX, one						ψ 20.70	Ψ 10.00												
EQC	way outgoing	5D					\$ 25.46	\$ 17.90												1
	NETWORK ACCESS REGISTER, ESSX, one																			
EQC	way outgoing	7A					\$ 35.76	\$ 22.18												ı
	NETWORK ACCESS REGISTER, ESSX, one	70																		1
EQC	way outgoing	7B					\$ 28.64	\$ 17.76												
EQC	NETWORK ACCESS REGISTER, ESSX, one way outgoing	7C					\$ 29.60	\$ 18.36												1
EQC	NETWORK ACCESS REGISTER, ESSX, one	,,,					\$ 29.00	\$ 10.30												
EQC	way outgoing	7D					\$ 27.92	\$ 17.32												1
	NETWORK ACCESS REGISTER, ESSX, one																			
EQC	way outgoing	7E					\$ 29.68	\$ 18.41												1
	NETWORK ACCESS REGISTER, ESSX, one																			
EQC	way outgoing	7F					\$ 41.36	\$ 25.65												
FOF	NETWORK ACCESS REGISTER, ESSX, combination, initial	1			¢ 6.70	¢ 504											\$ 7.90	\$ 6.06		1
EQE	NETWORK ACCESS REGISTER, ESSX,	- '			\$ 6.78	\$ 5.64											\$ 7.90	\$ 6.06		
EQE	combination, initial	2			\$ 6.78	\$ 5.64											\$ 8.29	\$ 6.36		1
- 4-	NETWORK ACCESS REGISTER, ESSX,				ψ 0.10	ψ 0.01											Ψ 0.20	ψ 0.00		
EQE	combination, initial	3			\$ 6.78	\$ 5.64											\$ 8.69	\$ 6.66		
	NETWORK ACCESS REGISTER, ESSX,																			
EQE	combination, initial	4			\$ 7.02	\$ 5.84											\$ 9.08	\$ 6.96		
	NETWORK ACCESS REGISTER, ESSX, combination, initial	5			6 700	<b>6</b> 004											<b>6</b> 0.40	¢ 7.07		1
EQE	NETWORK ACCESS REGISTER, ESSX,	5			\$ 7.26	\$ 6.04											\$ 9.48	\$ 7.27		
EQE	combination, initial	6			\$ 7.52	\$ 5.94											\$ 9.87	\$ 7.57		1
	NETWORK ACCESS REGISTER, ESSX,	-			ψ 1.02	ψ 0.01											ψ 0.07	Ψ 1.01		
EQE	combination, initial	7			\$ 7.73	\$ 6.43											\$ 10.27	\$ 7.00		1
	NETWORK ACCESS REGISTER, ESSX,																			
EQE	combination, initial	8			\$ 7.94	\$ 6.61														
	NETWORK ACCESS REGISTER, ESSX,	9																		1
EQE	combination, initial NETWORK ACCESS REGISTER, ESSX,	9			\$ 8.14	\$ 6.09														
EQE	combination, initial	10			\$ 8.29	\$ 6.21														
-~-	NETWORK ACCESS REGISTER, ESSX,	<del>  </del>			7 0.20	7 0.21														
EQE	combination, initial	11			\$ 8.43	\$ 6.31														
	NETWORK ACCESS REGISTER, ESSX,																			
EQE	combination, initial	12			\$ 8.56	\$ 6.41														
	MESSAGE RATE ESSC NAR,																			
FOFCE	COMBINATION EXPANDED LOCAL RATE 3 MILE CALLING (GA)	2					e 0.04	e = 0=												
EQEGE	MESSAGE RATE ESSC NAR,						\$ 8.04	\$ 5.65												
	COMBINATION EXPANDED LOCAL RATE 3																			
EQEGE	MILE CALLING (GA)	5					\$ 10.49	\$ 7.37												
	MESSAGE RATE ESSC NAR,							-												
	COMBINATION EXPANDED LOCAL RATE 3																			
EQEGE	MILE CALLING (GA)	7			l		\$ 12.45	\$ 7.72					l							

USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
Code	MESSAGE RATE ESSC NAR,	Kate Group	Rate	Nate	Rate	Nate	Rate	Nate	Nate	Nate	Rate	Nate	Rate	IVIO-VOI Kate	Nate	Nate	Nate	Nate	Nate	Nate
EQEGE	COMBINATION EXPANDED LOCAL RATE 3 MILE CALLING (GA)	12					\$ 19.47	\$ 11.27												
	NETWORK ACCESS REGISTER, ESSX, one						ψ 13.47	Ψ 11.27												
EQF	way outgoing, initial NETWORK ACCESS REGISTER, ESSX, one	1			\$ 6.78	\$ 5.64											\$ 7.90	\$ 6.06		
EQF	way outgoing, initial	2			\$ 6.78	\$ 5.64											\$ 8.29	\$ 6.36		
EQF	NETWORK ACCESS REGISTER, ESSX, one way outgoing, initial	3			\$ 6.78	\$ 5.64											\$ 8.69	\$ 6.66		
	NETWORK ACCESS REGISTER, ESSX, one																			
EQF	way outgoing, initial NETWORK ACCESS REGISTER, ESSX, one	4			\$ 7.02	\$ 5.84											\$ 9.08	\$ 6.96		
EQF	way outgoing, initial	5			\$ 7.26	\$ 6.04											\$ 9.48	\$ 7.27		
EQF	NETWORK ACCESS REGISTER, ESSX, one way outgoing, initial	6			\$ 7.52	\$ 5.94											\$ 9.87	\$ 7.57		
	NETWORK ACCESS REGISTER, ESSX, one																			
EQF	way outgoing, initial NETWORK ACCESS REGISTER, ESSX, one	7			\$ 7.73	\$ 6.43											\$ 10.27	\$ 7.00		
EQF	way outgoing, initial	8			\$ 7.94	\$ 6.61														
EQF	NETWORK ACCESS REGISTER, ESSX, one way outgoing, initial	9			\$ 8.14	\$ 6.09														
	NETWORK ACCESS REGISTER, ESSX, one																			
EQF	way outgoing, initial NETWORK ACCESS REGISTER, ESSX, one	10			\$ 8.29	\$ 6.21														
EQF	way outgoing, initial	11			\$ 8.43	\$ 6.31														
EQF	NETWORK ACCESS REGISTER, ESSX, one way outgoing, initial	12			\$ 8.56	\$ 6.41														
LQI	MESSAGE RATE ESSC NAR, One way				ψ 0.00	ψ 0.41														
EQFGE	outgoing, EXPANDED LOCAL RATE 3 MILE CALLING (GA)	2					\$ 8.04	\$ 5.65												
EGIOL	MESSAGE RATE ESSC NAR, One way	_					ψ 0.04	ψ 0.00												
EQFGE	outgoing, EXPANDED LOCAL RATE 3 MILE CALLING (GA)	5					\$ 10.49	\$ 7.37												
EQIOL	MESSAGE RATE ESSC NAR, One way						ψ 10.45	Ψ 7.07												
EQFGE	outgoing, EXPANDED LOCAL RATE 3 MILE CALLING (GA)	7					\$ 12.45	\$ 7.72												
EQIOL	MESSAGE RATE ESSC NAR, One way	-					ψ 12.40	Ψ 7.72												
EQFGE	outgoing, EXPANDED LOCAL RATE 3 MILE CALLING (GA)	12					\$ 19.47	\$ 11.27												
	NETWORK ACCESS REGISTER, ESSX,						ψ 15.47	Ψ 11.27												
EQG	combination, additional NETWORK ACCESS REGISTER, ESSX,	1			\$ 6.78	\$ 5.64											\$ 7.90	\$ 6.06		
EQG	combination, additional	2			\$ 6.78	\$ 5.64											\$ 8.29	\$ 6.36		
EQG	NETWORK ACCESS REGISTER, ESSX, combination, additional	3			\$ 6.78	\$ 5.64											\$ 8.69	\$ 6.66		
	NETWORK ACCESS REGISTER, ESSX,																			
EQG	combination, additional NETWORK ACCESS REGISTER, ESSX,	4			\$ 7.02	\$ 5.84											\$ 9.08	\$ 6.96		
EQG	combination, additional	5			\$ 7.26	\$ 6.04											\$ 9.48	\$ 7.27		
EQG	NETWORK ACCESS REGISTER, ESSX, combination, additional	6			\$ 7.52	\$ 5.94											\$ 9.87	\$ 7.57		
	NETWORK ACCESS REGISTER, ESSX,																			
EQG	combination, additional NETWORK ACCESS REGISTER, ESSX,	7			\$ 7.73	\$ 6.43											\$ 10.27	\$ 7.00		
EQG	combination, additional	8			\$ 7.94	\$ 6.61			_											
EQG	NETWORK ACCESS REGISTER, ESSX, combination, additional	9			\$ 8.14	\$ 6.09	·													
	NETWORK ACCESS REGISTER, ESSX,																			
EQG	combination, additional NETWORK ACCESS REGISTER, ESSX,	10			\$ 8.29	\$ 6.21														
EQG	combination, additional	11			\$ 8.43	\$ 6.31														
EQG	NETWORK ACCESS REGISTER, ESSX, combination, additional	12			\$ 8.56	\$ 6.41														
	NETWORK ACCESS REGISTER, ESSX, one																			
EQK	way outgoing, additional NETWORK ACCESS REGISTER, ESSX, one	1			\$ 6.78	\$ 5.64											\$ 7.90	\$ 6.06		
EQK	way outgoing, additional	2			\$ 6.78	\$ 5.64											\$ 8.29	\$ 6.36		

USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
	NETWORK ACCESS REGISTER, ESSX, one		naio	- Tulo			riaio	ruio	ruio	ruio	itato	11410	italo	o var maio	rtuto				nuio	Hulo
EQK	way outgoing, additional NETWORK ACCESS REGISTER, ESSX, one	3			\$ 6.78	\$ 5.64											\$ 8.69	\$ 6.66		
EQK	way outgoing, additional	4			\$ 7.02	\$ 5.84											\$ 9.08	\$ 6.96		
EQK	NETWORK ACCESS REGISTER, ESSX, one way outgoing, additional	5			\$ 7.26	\$ 6.04											\$ 9.48	\$ 7.27		
EQK	NETWORK ACCESS REGISTER, ESSX, one way outgoing, additional	6			\$ 7.52	\$ 5.94											\$ 9.87	\$ 7.57		
	NETWORK ACCESS REGISTER, ESSX, one																			
EQK	way outgoing, additional NETWORK ACCESS REGISTER, ESSX, one	7			\$ 7.73	\$ 6.43											\$ 10.27	\$ 7.00		
EQK	way outgoing, additional	8			\$ 7.94	\$ 6.61														
EQK	NETWORK ACCESS REGISTER, ESSX, one way outgoing, additional	9			\$ 8.14	\$ 6.09														
EQK	NETWORK ACCESS REGISTER, ESSX, one way outgoing, additional	10			\$ 8.29	\$ 6.21														
EQK	NETWORK ACCESS REGISTER, ESSX, one way outgoing, additional	11			\$ 8.43	\$ 6.31														
	NETWORK ACCESS REGISTER, ESSX, one	12																		
EQK	way outgoing, additional NETWORK ACCESS REGISTER, Measured	12			\$ 8.56	\$ 6.41														
EQP	Rate, both way NETWORK ACCESS REGISTER, Measured	1	\$ 6.57	\$ 5.50																
EQP	Rate, both way	2	\$ 6.57	\$ 5.50																
EQP	NETWORK ACCESS REGISTER, Measured Rate, both way	3	\$ 6.57	\$ 5.50																
EQP	Rate, both way	4	\$ 6.57	\$ 5.50																
EQP	NETWORK ACCESS REGISTER, Measured Rate, both way	5	\$ 6.57	\$ 5.50																
	NETWORK ACCESS REGISTER, Measured																			
EQR	Rate, inward only NETWORK ACCESS REGISTER, Measured	1	\$ 6.57	\$ 5.50																
EQR	Rate, inward only NETWORK ACCESS REGISTER, Measured	2	\$ 6.57	\$ 5.50																
EQR	Rate, inward only	3	\$ 6.57	\$ 5.50																
EQR	NETWORK ACCESS REGISTER, Measured Rate, inward only	4	\$ 6.57	\$ 5.50																
EQR	NETWORK ACCESS REGISTER, Measured Rate, inward only	5	\$ 6.57	\$ 5.50																
FOC	NETWORK ACCESS REGISTER, Measured Rate, outward only	1	¢ 6.57																	
EQS	NETWORK ACCESS REGISTER, Measured	·	\$ 6.57	\$ 5.50																
EQS	Rate, outward only NETWORK ACCESS REGISTER, Measured	2	\$ 6.57	\$ 5.50																
EQS	Rate, outward only NETWORK ACCESS REGISTER, Measured	3	\$ 6.57	\$ 5.50																
EQS	Rate, outward only	4	\$ 6.57	\$ 5.50																
EQS	NETWORK ACCESS REGISTER, Measured Rate, outward only	5	\$ 6.57	\$ 5.50																
ER3	Call forwarding Variable with Speed calling (8-code)		•	•							\$ 9.70	\$ 4.23	\$ 8.00	\$ 3.37	\$ 5.20	\$ 2.14				
	Call forwarding Variable with Three way calling																			
ER5 ER6	Three way calling with speed calling (8-code)										\$ 10.60 \$ 9.70		\$ 8.00 \$ 8.00	\$ 3.37 \$ 3.37	\$ 6.10 \$ 6.10	\$ 2.51 \$ 2.51				
ER9	Call waiting with Three way calling										\$ 10.10		\$ 8.00	\$ 3.37	\$ 7.85	\$ 3.23				
ES3	Call waiting, call forwarding variable, three way calling and speed calling (8-code)										\$ 18.70			\$ 5.90	\$ 10.90	\$ 4.49				
ES6	Call Waiting with Speed Calling (8-code)										\$ 9.20	\$ 4.01	\$ 8.00	\$ 3.37	\$ 7.00	\$ 2.88				
ES7	Call Waiting with call forwarding variable										\$ 10.10	\$ 4.40	\$ 9.50	\$ 4.00	\$ 7.00	\$ 2.88				
ESA	Call waiting with Call forwarding Variable and Speed Calling( 8-code)										\$ 14.50	\$ 6.32	\$ 11.25	\$ 4.74	\$ 8.50	\$ 3.50				
ESC	Three way Calling		\$ 4.50	\$ 1.88	\$ 5.00	\$ 2.50	\$ 5.25	\$ 2.17	\$ 4.40	\$ 1.86	\$ 5.30	\$ 2.31	\$ 11.25	\$ 1.79	\$ 3.85	\$ 1.59	\$ 6.00	\$ 2.81	\$ 4.50	\$ 1.89
ESCWT	Three way calling with transfer		\$ 6.00	\$ 2.51	\$ 6.00	\$ 2.99	\$ 6.00	\$ 2.48	\$ 6.00	\$ 2.53	\$ 6.00	\$ 2.62	\$ 6.00	\$ 2.53	\$ 6.00	\$ 2.47	\$ 6.00	\$ 2.81	\$ 6.00	\$ 2.52
ESF	Speed calling 30 code		\$ 5.50	\$ 2.30	\$ 5.00	\$ 2.50	\$ 5.50	\$ 2.27	\$ 5.50	\$ 2.32	\$ 5.50	\$ 2.40	\$ 5.50	\$ 2.32	\$ 5.35	\$ 2.20	\$ 8.00	\$ 3.75	\$ 5.50	\$ 2.31

USOC Code	Description	Rate Group		-Tariff Rate	AL-V Rat		FL-Tariff Rate	FL-V&T Rate	1	GA-Tariff Rate	GA-V&T Rate		Tariff ate	KY-V&	۲   I	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	_	Tariff ate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
ESFTK	Speed calling 30 code						\$ 5.00	\$ 2.50	\$	5.50	\$ 2.27									\$	5.35	\$ 2.20	\$ 8.00	\$ 3.75		
ESFWT	Speed calling 30 code						\$ 5.00	\$ 2.50	\$	5.50	\$ 2.27									\$	5.35	\$ 2.20	\$ 8.00	\$ 3.75		
ESL	Speed Calling 8 code		\$	4.50	\$	.88	\$ 3.00	\$ 1.50	\$	4.00	\$ 1.65	\$	4.40	\$ 1.8	6 \$	4.40	\$ 1.92	\$ 4.25	\$ 1.79	\$	2.90	\$ 1.19	\$ 6.00	\$ 2.81	\$ 4.50	\$ 1.89
ESLTK	Speed Calling 8 code						\$ 3.00	\$ 1.50	\$	4.00	\$ 1.65									\$	2.90	\$ 1.19	\$ 6.00	\$ 2.81		
ESLWT	Speed Calling 8 code						\$ 3.00	\$ 1.50	\$	4.00	\$ 1.65	\$	2.90	\$ 1.2	2 \$	6.00	\$ 2.62									
ESM	Call forwarding variable		\$	4.95	\$ 2	2.49	\$ 5.00	\$ 2.9	1 \$	5.00	\$ 2.48	\$	4.40	\$ 2.2	3 \$	5.30	\$ 2.73	\$ 5.00	\$ 2.53	\$	3.40	\$ 1.68	\$ 6.00	\$ 3.32	\$ 4.75	\$ 2.39
	Call forwarding variable with three way calling															45.00				_						
ESR	and speed calling (8-code)  Call waiting		\$		•			• • • •		0.05		-	4.40	• • •	\$	15.00	\$ 6.54	\$ 11.25		_	7.70	\$ 3.17	• • • •	• • • • •	• • • • • •	0 407
ESX	Call waiting  Call waiting with Three way calling and speed		\$	5.95	\$ 2	2.99	\$ 6.25	\$ 3.64	1 \$	6.25	\$ 3.10	\$	4.40	\$ 2.2	3 \$	4.80	\$ 2.47	\$ 6.00	\$ 3.03	\$	5.70	\$ 2.82	\$ 6.00	\$ 3.32	\$ 3.90	\$ 1.97
ETB	calling (8-code)														\$	14.50	\$ 6.32	\$ 11.25	\$ 4.74	\$	9.30	\$ 3.83				
ETC	Call waiting with call forwarding variable and three way calling															15.10	¢ 6.50	e 44.0F	¢ 474	e	0.20	\$ 3.83				
ETC	MAIN STATION LINE, DISTANCE ESSX					-			-	-					3	15.10	\$ 6.58	\$ 11.25	\$ 4.74	\$	9.30	\$ 3.83				
	EXCHANGE CIRCUIT, SCB-FLAT SBT-FLAT,																									
EXMEX	MESS. 01 MEAS RATE 1 1/2 MILES MILEAGE ESSX/DIGITAL ESS ESSX		\$	15.25	\$ 10	).85			_			\$	13.45	\$ 10.2	2 \$	19.40	\$ 13.84	\$ 12.40	\$ 7.84	\$ 1	2.40	\$ 7.66			\$ 15.60	\$ 10.48
	EXCHANGE CIRCUIT, MAIN STATION LINE,																									
EXMFA	MSG/MEAS RATE 2 MILES, NC						\$ 14.80	\$ 11.70	)											\$ 1	3.10	\$ 8.10	\$ 12.20	\$ 9.35		
EXMFX	MAIN STATION LINE, DISTANCE SENSITIVE 2 MILES		¢	21.00	¢ 1/	1.94						s	17.40	\$ 13.2	2 6	26.00	\$ 18.55	\$ 16.10	\$ 10.17	¢ 1	8 40	\$ 11.37			\$ 19.10	\$ 13.64
LXIVII X	MAIN STATION LINE DISTANCE, ESSX		¥	21.00	Ψ 1.	1.04			-			Ψ	17.40	ψ 13.2	.υ ψ	20.00	ψ 10.55	¥ 10.10	ψ 10.17	ΨΙ	0.40	ψ 11.57			ψ 13.10	ψ 13.04
	EXCHANGE CIRCUIT, MAIN STATION LINE,																									
EXMGX	SCB-FLAT RATE, SBT FLAT RATE, MSG OR REAS RATE 2 1/2 MILES		\$	21 15	\$ 15	5.05						s	23.35	\$ 17.7	5 \$	32.60	\$ 23.26	\$ 25.05	\$ 15.83	\$ 1	8 60	\$ 11.49			\$ 19.15	\$ 13.67
EXMOX	MAIN STATION LINE ESSX EXCHANGE		Ψ	21.10	Ψι	,.00						Ψ	20.00	Ψ 17.7	υ ψ	02.00	Ψ 20.20	ψ 20.00	ψ 10.00	Ψ.	0.00	Ψ 11.43			ψ 13.10	Ψ 10.07
	CIRCUITE, MAIN STATION LINE												00.50							١.,						
EXMHX	MSG/MEAS 3 1/2 MILES MAIN STATION LINE, DISTANT ESSX		\$	21.30	\$ 15	0.15			+			\$	23.50	\$ 17.8	6 \$	47.00	\$ 33.54	\$ 26.10	\$ 16.49	\$ 1	8.75	\$ 11.59			\$ 19.25	\$ 13.74
	EXCHANGE CIRCUITE, MAIN STATION																									
EXMJX	LINE/MEAS RATE MAIN STATION LINE, DISTANT ESSX		\$	21.45	\$ 15	5.26						\$	23.75	\$ 18.0	5 \$	47.65	\$ 34.00	\$ 26.40	\$ 16.68	\$ 1	8.95	\$ 11.71			\$ 19.35	\$ 13.82
	EXCHANGE CIRCUITE, MAIN STATION																									
EXMOX	LINE						\$ 17.25	\$ 13.63	3														\$ 21.10	\$ 16.18		
	MAIN STATION LINE, DISTANCE ESSX EXCHANGE CIRCUIT, MSG/MEASSURE																									
EXMPX	RATE 1 1/4 MILES																			\$ 1	0.70	\$ 6.61				
FCS	Flexible call forwarding		\$	9.00	\$ 3	3.77	\$ 9.00	\$ 4.49	9 \$	8.95	\$ 3.70	\$	9.90	\$ 4.1	8 \$	9.00	\$ 3.92	\$ 8.95	\$ 3.77	_	9.00	\$ 3.71	\$ 8.95	\$ 4.19	\$ 10.00	\$ 4.20
FCSCN	Flexible call forwarding with Audio calling name		6	11.00	6	1.60	\$ 11.00	\$ 5.49	8	10.95	\$ 4.53	e	11.00	\$ 4.6	5 9	11.00	\$ 4.80	\$ 10.95	\$ 4.61	¢ 1	1.00	\$ 4.53	\$ 10.95	\$ 5.13	\$ 11.00	\$ 4.62
GC1	Flat Rate Service		Ψ	11.00	Ψ	1.00	Ψ 11.00	Ψ 0.4	Ψ	10.55	Ψ 4.00	Ψ	11.00	Ψ	υ ψ	11.00	Ψ 4.00	Ψ 10.55	Ψ 4.01	Ψ.	1.00	Ψ 4.00	ψ 10.55	ψ 0.10	Ψ 11.00	Ψ 4.02
GCE	Call forwarding busy line		\$	3.50	\$	.76	\$ 3.25	\$ 1.89	\$	4.00	\$ 1.98	s	3.85	\$ 1.9	5 \$	3.50	\$ 1.80	\$ 3.50	\$ 1.77	\$	3.25	\$ 1.61	\$ 3.50	\$ 1.94	\$ 4.25	\$ 2.14
GCJ	Call forwarding don't answer		\$	3.50	_	.76	\$ 3.25	\$ 1.89	_	4.00	\$ 1.98		3.85	\$ 1.9	_	3.50	\$ 1.80	\$ 3.50	\$ 1.77	_	3.25	\$ 1.61	\$ 3.50	\$ 1.94	\$ 4.25	\$ 2.14
GCJRC	Call forwarding don't answer - ring control		\$	3.50		.76	\$ 3.25	\$ 1.89		4.00	\$ 1.98		3.85	\$ 1.9	_	3.50	\$ 1.80	\$ 3.50	\$ 1.77	_	3.25	\$ 1.61	\$ 3.50	\$ 1.94	\$ 4.25	\$ 2.14
GCZ	Remote access - call forwarding variable		\$	9.00		3.77	\$ 9.00	\$ 4.49	\$	8.95	\$ 3.70	\$	9.35	\$ 3.9	5 \$	9.00	\$ 3.92	\$ 8.95	\$ 3.77		6.90	\$ 2.84	\$ 10.00	\$ 4.69	\$ 9.00	\$ 3.78
	Customer control call forwarding don't answer										\$ 2.58			\$ 2.9												
GJC GJP	Call control of call forwarding busy line		\$	6.50		2.72	\$ 6.25 \$ 6.25	\$ 3.12	·	6.25 6.25	\$ 2.58 \$ 2.58		7.00 7.40	\$ 2.9		6.25	\$ 2.73 \$ 2.73	\$ 6.25 \$ 6.25	\$ 2.63 \$ 2.63	<u> </u>	6.25 6.25	\$ 2.58 \$ 2.58	\$ 8.00 \$ 8.00	\$ 3.75 \$ 3.75	\$ 6.25 \$ 6.25	\$ 2.63 \$ 2.63
	Anonymous Call Rejection		φ	0.50	Φ 4	12	\$ 4.00	\$ 2.00	_	4.00	\$ 1.65			\$ 1.8	_	0.23	φ 2.13	\$ 0.25	ş 2.03	<u> </u>	3.50	\$ 1.44		\$ 2.81	\$ 3.50	\$ 1.47
HBY HTG	Rotary Hunting, business flat rate lines	1	¢	10.00	e (	3.37	\$ 4.00 \$ 7.40	\$ 6.16	_	4.00	ψ 1.05		4.40 12.00	\$ 9.6	_	16.50	\$ 13.08	\$ 10.00	\$ 6.74	_	3.68		\$ 6.00 \$ 10.65	\$ 7.26	\$ 3.50 \$ 12.50	\$ 1.47
HTG	Rotary Hunting, business flat rate lines	2	φ	10.00		3.37	\$ 7.40	\$ 6.43	_	12.43	\$ 7.71			\$ 9.0	_		\$ 13.08			_	3.68	\$ 6.76		\$ 7.26	\$ 12.50	\$ 8.40
HTG	Rotary Hunting, business flat rate lines  Rotary Hunting, business flat rate lines	3	9	10.00		3.37	\$ 7.73	\$ 6.72	-	12.43	7./1 پ		11.25	\$ 9.0	_	16.50	<u> </u>	\$ 10.00 \$ 10.00		+		\$ 6.76	\$ 10.65	\$ 7.26	\$ 12.50 \$ 12.50	\$ 8.40
HTG	Rotary Hunting, business flat rate lines  Rotary Hunting, business flat rate lines	4	9	10.00		3.37			_						_		\$ 13.16			_	3.68		\$ 10.65	\$ 7.26		\$ 7.35
	Rotary Hunting, business flat rate lines  Rotary Hunting, business flat rate lines	5	Ф			_	\$ 8.41	\$ 7.00	_	16.00	¢ 40.07		10.00	\$ 8.0	_	16.60	\$ 13.16	\$ 10.00	\$ 6.74	_	3.68	\$ 6.76	\$ 10.65		\$ 12.50 \$ 12.50	
HTG	· · · · · · · · · · · · · · · · · · ·		\$	10.00		3.37	\$ 8.71	\$ 7.25	_	16.23	\$ 10.07	Ъ	5.70	\$ 3.1	3 \$	16.60	\$ 13.16	\$ 10.00	\$ 6.74	_	3.68	\$ 6.76	\$ 10.65	\$ 7.26	\$ 12.50	\$ 6.83
HTG	Rotary Hunting, business flat rate lines	6 7	Ф	10.00	<b>\$</b>	5.86	\$ 9.05	\$ 7.53	_	40.05	A 0.70				\$	16.60	\$ 13.16	\$ 10.00	\$ 6.74	_	3.68	\$ 6.76	\$ 10.65	\$ 6.35		
HTG	Rotary Hunting, business flat rate lines					-	\$ 9.33	\$ 7.76	_	19.25	\$ 8.76				\$		\$ 13.16	\$ 10.00	\$ 6.74	_	3.68	\$ 6.76	\$ 10.65	\$ 5.90		
HTG	Rotary Hunting, business flat rate lines	8					\$ 9.60	\$ 7.99	_						\$	16.60	\$ 13.16			_	3.68	\$ 6.76				
HTG	Rotary Hunting, business flat rate lines	9					\$ 9.86	\$ 8.20	)						\$	16.60	\$ 13.16	\$ 10.00	\$ 6.74	\$ 1	3.68	\$ 6.76				

USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
HTG	Rotary Hunting, business flat rate lines	10			\$ 10.00	\$ 8.32							\$ 10.00	\$ 6.74	\$ 13.68	\$ 6.76				
HTG	Rotary Hunting, business flat rate lines	11			\$ 10.00	\$ 8.32														
HTG	Rotary Hunting, business flat rate lines	12			\$ 10.00	\$ 8.32	\$ 23.9	5 \$ 7.92					\$ 10.00	\$ 6.74						
HTG	Rotary Hunting with EAS Differentials	12A					\$ 25.9	1 \$ 8.57	•											
HTG	Rotary Hunting with EAS Differentials	12B					\$ 25.9	1 \$ 8.57												
HTG	Rotary Hunting with EAS Differentials	12C					\$ 23.9	5 \$ 7.92												
HTG	Rotary Hunting with EAS Differentials	12D					\$ 26.2	6 \$ 8.69												
HTG	Rotary Hunting with EAS Differentials	12E					\$ 26.2	6 \$ 8.69												
HTG	Rotary Hunting with EAS Differentials	2A					\$ 13.0	4 \$ 8.09												
HTG	Rotary Hunting with EAS Differentials	2B					\$ 12.7	9 \$ 7.93												
HTG	Rotary Hunting with EAS Differentials	2C					\$ 123.3	9 \$ 76.53												
HTG	Rotary Hunting with EAS Differentials	2D					\$ 13.4	2 \$ 8.32												
HTG	Rotary Hunting with EAS Differentials	2E					\$ 12.7	6 \$ 7.91												
HTG	Rotary Hunting with EAS Differentials	5A					\$ 16.8													
HTG	Rotary Hunting with EAS Differentials	5B					\$ 16.9	4 \$ 10.51												
HTG	Rotary Hunting with EAS Differentials	5C					\$ 16.5													
HTG	Rotary Hunting with EAS Differentials	5D					\$ 17.1													
HTG	Rotary Hunting with EAS Differentials	7A					\$ 22.3													
HTG	Rotary Hunting with EAS Differentials	7B					\$ 19.8													
HTG	Rotary Hunting with EAS Differentials	7C					\$ 20.1													
HTG	Rotary Hunting with EAS Differentials	7D					\$ 19.6													
HTG	Rotary Hunting with EAS Differentials	7E					\$ 20.2													
HTG	Rotary Hunting with EAS Differentials	7E					\$ 20.2													
піб	Hunting/Rotary/Grouping features used	/ 1					\$ 20.0	э э 9.41												
HTG1B	w/Louisiana local optional srvc	1 thru 5									\$ 15.50	\$ 12.29								
HTGAP	Hunting/Rotary used with Area Plus																\$ 10.65	\$ 5.90		
HTGDB	Hunting/Rotary/Grouping features thrifty caller service														ф э.о.е.	\$ 1.90				
пібрв	Hunting/Rotary Service, Georgia Community														\$ 3.85	\$ 1.90				
HTGGE	Calling	12A					\$ 18.8	4 \$ 6.23												
HTGGE	Hunting/Rotary Service, Georgia Community Calling	12B					\$ 18.8	4 \$ 6.23												
HIGGE	Hunting/Rotary Service, Georgia Community	120					<b>ў</b> 10.0	+ φ 0.23												
HTGGE	Calling	12C					\$ 18.6	6 \$ 6.17	•											
HTGGE	Hunting/Rotary Service, Georgia Community Calling	12D					\$ 19.1	0 \$ 6.32												
HIGGE	Hunting/Rotary Service, Georgia Community	120					J 19.1	J \$ 0.32												
HTGGE	Calling	12E					\$ 19.1	0 \$ 6.32												
HTGGE	Hunting/Rotary Service, Georgia Community Calling	2A					\$ 9.4	8 \$ 5.88												
IIIGGE	Hunting/Rotary Service, Georgia Community	1					ψ 3.4	J \$ 3.00												
HTGGE	Calling	2B					\$ 9.3	5.77												
HTGGE	Hunting/Rotary Service, Georgia Community Calling	2C					\$ 9.7	4 \$ 6.04												
IIIGGE	Hunting/Rotary Service, Georgia Community	1					ψ 3.1	Ψ 0.04												
HTGGE	Calling	2D					\$ 9.7	6 \$ 6.05												
HTGGE	Hunting/Rotary Service, Georgia Community Calling	2E					\$ 9.2	8 \$ 5.76												
III GGE	Hunting/Rotary Service, Georgia Community	1					φ 9.2	υ ψ 5.76												
HTGGE	Calling	5A					\$ 12.2	4 \$ 7.59												
HTGGE	Hunting/Rotary Service, Georgia Community Calling	5B					\$ 12.3	2 \$ 7.64												
III GGE	Hunting/Rotary Service, Georgia Community	1					12.3 پ	μ 1.04												
HTGGE	Calling	5C					\$ 12.0	6 \$ 7.48												
HTGGE	Hunting/Rotary Service, Georgia Community Calling	5D					\$ 12.5	0 \$ 7.75												
111002	Hunting/Rotary Service, Georgia Community						12.5 پ	υ φ 1.15												
HTGGE	Calling	7A					\$ 16.2	2 \$ 7.38												

					_															
USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V		f FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
Code	Hunting/Rotary Service, Georgia Community	Kate Group	Nate	itati	Nate	Nate	Nate	itate	Nate	Nate	Nate	Nate	Nate	WIS-VOT Kate	Nate	Nate	Rate	Nate	Nate	Nate
HTGGE	Calling	7B					\$ 14.44	\$ 6.57												
HTGGE	Hunting/Rotary Service, Georgia Community Calling	7C					\$ 14.68	\$ 6.68											1	
	Hunting/Rotary Service, Georgia Community																			
HTGGE	Calling Hunting/Rotary Service, Georgia Community	7D					\$ 14.26	\$ 6.49											<b></b>	$\vdash$
HTGGE	Calling	7E					\$ 14.70	\$ 6.69											1	
HTGGE	Hunting/Rotary Service, Georgia Community Calling	7F					\$ 15.04	\$ 6.84											1	
HIGGE	Caming						\$ 15.04	\$ 0.04												
HTGKX	Hunting/Rotary Service, Measured Rate MS	1							\$ 12.00	\$ 9.63										
HTGKX	Hunting/Rotary Service, Measured Rate MS	2							\$ 11.25	\$ 9.03										
HTGKX	Hunting/Rotary Service, Measured Rate MS	3							\$ 10.50	\$ 8.42										
HTGKX	Hunting/Rotary Service, Measured Rate MS	4							\$ 10.00	\$ 8.02									1	
HTGKX	Hunting/Rotary Service, Measured Rate MS	5							\$ 5.70	\$ 3.13									ļ	
HTGLE	Hunting/Rotary Service, Area Calling Plan MS												\$ 10.00	\$ 5.48						
HTGLO	Hunting/Rotary Service, Message Rate MS	1	\$ 10.0	) \$ 8	.37								\$ 10.00							
HTGLO	Hunting/Rotary Service, Message Rate MS	2	\$ 10.0	) \$ 8	.37															
HTGLO	Hunting/Rotary Service, Message Rate MS	3	\$ 10.0	) \$ 8	.37															
HTGLO	Hunting/Rotary Service, Message Rate MS	4	\$ 10.0	) \$ 8	.37														1	
HTGLO	Hunting/Rotary Service, Message Rate MS	5	\$ 10.0	) \$ 8	.37															
HTGLO	Hunting/Rotary Service, Message Rate MS	6	\$ 10.0	) \$ 5	.86															
HTGME	Hunting/Rotary Service, Flat Rate MS	1	\$ 10.0	) \$ 8	.37				\$ 12.00	\$ 9.63	\$ 16.50	\$ 13.08	\$ 10.00	\$ 6.74					\$ 14.21	\$ 9.55
HTGME	Hunting/Rotary Service, Flat Rate MS	2	\$ 10.0	) \$ 8	.37				\$ 11.25	\$ 9.03			\$ 10.00	\$ 6.74					\$ 16.16	\$ 10.86
HTGME	Hunting/Rotary Service, Flat Rate MS	3	\$ 10.0	\$ 8	.37				\$ 10.50	\$ 8.42			\$ 10.00	\$ 6.74					\$ 17.21	\$ 11.57
HTGME	Hunting/Rotary Service, Flat Rate MS	4	\$ 10.0	) \$ 8	.37				\$ 10.00	\$ 8.02			\$ 10.00	\$ 6.74					\$ 20.51	\$ 12.06
HTGME	Hunting/Rotary Service, Flat Rate MS	5	\$ 10.0	) \$ 8	.37				\$ 5.70	\$ 3.13			\$ 10.00	\$ 6.74					\$ 12.50	\$ 6.83
HTGME	Hunting/Rotary Service, Flat Rate MS	6	\$ 10.0	) \$ 5	.86															
HTGMS	Rotary Hunting, business message rate lines	1			\$ 7.4	0 \$ 6.16			\$ 12.00	\$ 9.63			\$ 10.00	\$ 6.74					\$ 14.21	\$ 9.55
HTGMS	Rotary Hunting, business message rate lines	2			\$ 7.7	3 \$ 6.43			\$ 11.25	\$ 9.03			\$ 10.00	\$ 6.74					\$ 16.16	\$ 10.86
HTGMS	Rotary Hunting, business message rate lines	3			\$ 8.0	8 \$ 6.72			\$ 10.50	\$ 8.42			\$ 10.00	\$ 6.74					\$ 17.21	\$ 11.57
HTGMS	Rotary Hunting, business message rate lines	4			\$ 8.4	1 \$ 7.00			\$ 10.00	\$ 8.02			\$ 10.00	\$ 6.74					\$ 20.51	\$ 12.06
HTGMS	Rotary Hunting, business message rate lines	5			\$ 8.7	1 \$ 7.25			\$ 5.70	\$ 3.13			\$ 10.00	\$ 6.74					\$ 12.50	\$ 6.83
HTGMS	Rotary Hunting, business message rate lines	6			\$ 9.0	5 \$ 7.53							\$ 10.00	\$ 6.74						
HTGMS	Rotary Hunting, business message rate lines	7			\$ 9.3								\$ 10.00							
HTGMS	Rotary Hunting, business message rate lines	8			\$ 9.0								\$ 10.00							
HTGMS	Rotary Hunting, business message rate lines	9			\$ 9.8								\$ 10.00							
HTGMS	Rotary Hunting, business message rate lines	10			\$ 10.0															
													\$ 10.00	\$ 6.74						
HTGMS	Rotary Hunting, business message rate lines	11			\$ 10.0						-		-							
HTGMS	Rotary Hunting, business message rate lines	12			\$ 10.0	0 \$ 8.32							\$ 10.00	\$ 6.74						
HTGMS	Rotary Hunting, business message rate lines	13											\$ 10.00	\$ 5.48					ļ	
HTGNS	Rotary Hunting with EAS Differentials	12A					\$ 9.89	\$ 3.27											ļ——— <sup>'</sup>	
HTGNS	Rotary Hunting with EAS Differentials	12B					\$ 9.89	\$ 3.27											<u> </u>	

USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
HTGNS	Rotary Hunting with EAS Differentials	12C	Nate	ivate	Nate	Nate	\$ 9.80	\$ 3.24	Nate	ivate	Nate	Nate	ivate	WO-VOT IVALE	Nate	Nate	Nate	Nate	Nate	Nate
HTGNS	Rotary Hunting with EAS Differentials	12D					\$ 10.03	\$ 3.32												$\vdash$
HTGNS	Rotary Hunting with EAS Differentials	12E					\$ 10.03	\$ 3.41												
	Rotary Hunting with EAS Differentials	2A																		
HTGNS	Rotary Hunting with EAS Differentials  Rotary Hunting with EAS Differentials	2B					,	\$ 3.09												$\vdash$
HTGNS	Rotary Hunting with EAS Differentials	2C					\$ 4.88	\$ 3.03												
HTGNS	, ,	2D					\$ 5.11	\$ 3.17												
HTGNS	Rotary Hunting with EAS Differentials	2E					\$ 5.12	\$ 3.18												$\vdash$
HTGNS	Rotary Hunting with EAS Differentials	5A					\$ 4.87	\$ 3.02												
HTGNS	Rotary Hunting with EAS Differentials	5A 5B					\$ 6.43 \$ 6.47	\$ 3.99												$\vdash$
HTGNS	Rotary Hunting with EAS Differentials						<b>Q</b> 0.11	\$ 4.01												
HTGNS	Rotary Hunting with EAS Differentials	5C 5D					\$ 6.33	\$ 3.93												$\vdash$
HTGNS	Rotary Hunting with EAS Differentials	7A					\$ 6.56	1												$\vdash$
HTGNS	Rotary Hunting with EAS Differentials						\$ 8.52	\$ 3.88												<b>—</b>
HTGNS	Rotary Hunting with EAS Differentials	7B					\$ 7.58	\$ 3.45												<b>——</b>
HTGNS	Rotary Hunting with EAS Differentials	7C					\$ 7.71	\$ 3.51												
HTGNS	Rotary Hunting with EAS Differentials	7D					\$ 7.49	\$ 3.41												
HTGNS	Rotary Hunting with EAS Differentials	7E					\$ 7.72	\$ 3.51												
HTGNS	Rotary Hunting with EAS Differentials	7F					\$ 7.90	\$ 3.59												
HTGTR	Rotary Hunting	1																	\$ 18.38	\$ 12.35
HTGTR	Rotary Hunting	2																	\$ 18.38	\$ 12.35
HTGTR	Rotary Hunting	3																	\$ 18.38	\$ 12.35
HTGTR	Rotary Hunting	4																	\$ 18.38	\$ 10.81
HTGTR	Rotary Hunting	5																	\$ 12.50	\$ 6.83
L3B	LOCAL OPTION B GROUPING SERVICE LOCAL OPTIONAL SERVICE, OPTION B, BUSINESS DISTRICT (OPTION 2)										\$ 26.25	\$ 18.73								
LPRFX	CHANNELS ACTIVATED: "B" CHANNEL		\$ 13.75	\$ 10.36	\$ 16.25	\$ 12.17					\$ 16.25		\$ 16.25	\$ 12.32	\$ 16.25	\$ 12.05	\$ 53.70	\$ 41.18	\$ 16.25	\$ 12.29
LPSMX	ISDN individual business service		*		¥	*					\$ 6.50	\$ 4.64			7 10:20			•	\$ 6.50	\$ 4.91
LPWHX	ISDN individual business service										\$ 120.00	\$ 85.62			\$ 120.00	\$ 88.99			\$ 120.00	\$ 90.72
LPXOX	ISDN individual business service										\$ 110.00	\$ 78.49			\$ 110.00	\$ 81.58			\$ 110.00	\$ 83.16
LPYLX	ISDN individual business service										\$ 15.50	\$ 11.06			\$ 12.00	\$ 8.90			\$ 13.50	\$ 10.21
LQ1AA	ISDN individual business service		\$ 69.95	\$ 52.69	\$ 69.95	\$ 52.37	\$ 69.95	\$ 52.06	\$ 69.95	\$ 53.17	\$ 69.95	\$ 49.91	\$ 69.95	\$ 53.04	\$ 69.95	\$ 51.87	\$ 69.95	\$ 53.64	\$ 69.95	\$ 52.88
LQ1AB	ISDN individual business service		\$ 69.95	\$ 52.69	\$ 69.95	\$ 52.37	\$ 69.95	\$ 52.06		\$ 53.17	\$ 69.95	\$ 49.91	\$ 69.95		\$ 69.95	\$ 51.87	\$ 69.95	\$ 53.64	\$ 69.95	\$ 52.88
LQ1AC	ISDN individual business service		\$ 69.95	\$ 52.69	\$ 69.95	\$ 52.37	\$ 69.95	\$ 52.06		\$ 53.17	\$ 69.95	\$ 49.91	\$ 69.95		\$ 69.95	\$ 51.87	\$ 69.95	\$ 53.64	\$ 69.95	\$ 52.88
LQ1AD	ISDN individual business service		\$ 69.95	\$ 52.69	\$ 69.95	\$ 52.37	\$ 69.95	\$ 52.06		\$ 53.17	\$ 69.95	\$ 49.91	\$ 69.95		\$ 69.95	\$ 51.87	\$ 69.95	\$ 53.64	\$ 69.95	\$ 52.88
	ISDN individual business service		\$ 249.95	\$ 188.29	\$ 249.95		\$ 249.95			\$ 190.00	\$ 249.95				\$ 249.95	\$ 185.36			\$ 249.95	\$ 188.96
LQ1BA	BASIC RATE ISDN DSL ARRANGEMENT		\$ 249.95	\$ 100.29	\$ 249.95	\$ 187.14	\$ 249.95	\$ 186.04	\$ 249.95	\$ 190.00	\$ 249.95	\$ 176.34	\$ 249.95	\$ 169.52	\$ 249.95	\$ 165.36	\$ 249.95	\$ 191.66	\$ 249.95	\$ 100.90
	OPT USAGE PLAN, IND BUSINESS SRVE																			
LQ1BA	LINES		\$ 249.95	\$ 188.29	\$ 249.95	\$ 187.14	\$ 249.95	\$ 186.04		\$ 190.00			\$ 249.95				\$ 249.95	\$ 191.66	\$ 249.95	\$ 188.96
LQ1BB	ISDN individual business service		\$ 249.95	\$ 188.29	\$ 249.95	\$ 187.14	\$ 249.95	\$ 186.04		\$ 190.00	1	\$ 178.34	\$ 249.95		\$ 249.95	\$ 185.36	\$ 249.95	\$ 191.66	\$ 249.95	\$ 188.96
LQ1BC	ISDN individual business service		\$ 249.95	\$ 188.29	\$ 249.95	\$ 187.14	\$ 249.95	\$ 186.04		\$ 190.00	1	\$ 178.34	\$ 249.95		\$ 249.95	\$ 185.36	\$ 249.95	\$ 191.66	\$ 249.95	\$ 188.96
LQ1BD	ISDN individual business service		\$ 249.95	\$ 188.29	\$ 249.95	\$ 187.14	\$ 249.95	\$ 186.04	\$ 249.95	\$ 190.00	\$ 249.95	\$ 178.34	\$ 249.95	\$ 189.52	\$ 249.95	\$ 185.36	\$ 249.95	\$ 191.66	\$ 249.95	\$ 188.96
LQGFX	ISDN individual business service		\$ 13.75	\$ 10.36	\$ 16.25	\$ 12.17	\$ 16.25	\$ 12.09	\$ 16.25	\$ 12.35			\$ 16.25				\$ 16.25	\$ 12.46		
LQGFX	ISDN B CHANNEL FLAT RATE		\$ 13.75	\$ 10.36	\$ 16.25	\$ 12.17	\$ 16.25	\$ 12.09			\$ 16.25	\$ 11.59	\$ 16.25		\$ 16.25	\$ 12.05	\$ 53.70	\$ 41.18		
LQGHX	ISDN individual business service		\$ 120.00	\$ 90.40	\$ 120.00	\$ 224.96	\$ 120.00	\$ 89.32	\$ 120.00	\$ 91.22			\$ 120.00	\$ 90.99			\$ 120.00	\$ 92.02		
LQGMB	ISDN individual business service		\$ 12.25	\$ 9.23	\$ 12.25	\$ 224.96	\$ 12.25	\$ 9.12	\$ 12.25	\$ 9.31			\$ 12.25	\$ 9.29			\$ 12.25	\$ 9.39		
LQGOX	ISDN individual business service		\$ 110.00	\$ 82.86	\$ 110.00	\$ 224.96	\$ 110.00	\$ 81.87	\$ 110.00	\$ 83.62			\$ 110.00	\$ 83.41			\$ 110.00	\$ 84.35		
LQRDB	ISDN individual business service		\$ 13.50	\$ 10.17	\$ 13.50	\$ 224.96	\$ 13.50	\$ 10.05	\$ 13.50	\$ 10.26			\$ 13.50	\$ 10.24			\$ 13.50	\$ 10.35		
LQRLB	ISDN individual business service		\$ 13.50	\$ 10.17	\$ 13.50	\$ 224.96	\$ 13.50	\$ 10.05	\$ 13.50	\$ 10.26			\$ 13.50	\$ 10.24			\$ 13.50	\$ 10.35	-	
LQTDB	ISDN individual business service		\$ 60.00	\$ 45.20	\$ 55.00	\$ 41.18	\$ 55.00	\$ 40.94	\$ 55.00	\$ 41.81			\$ 55.00	\$ 41.70			\$ 55.00	\$ 42.17		

USOC			Al Tariff	AL-V&T	El Toriff	FL-V&T	CA Touiss	GA-V&T	KV Tariff	KY-V&T	I A Touiss	LA-V&T	MC Taviff		NC-Tariff	NC-V&T	SC-Tariff	SC-V&T	TNI Towlff	TN-V&T
Code	Description	Rate Group	AL-Tariff Rate	Rate	FL-Tariff Rate	Rate	GA-Tariff Rate	Rate	KY-Tariff Rate	Rate	LA-Tariff Rate	Rate	MS-Tariff Rate	MS-V&T Rate		Rate	Rate	Rate	TN-Tariff Rate	Rate
LQTEB	ISDN THRESHOLD PRICING, month to month		\$ 60.00	\$ 45.20	\$ 55.00	\$ 41.18	\$ 55.00	\$ 40.94	\$ 55.00	\$ 41.81	\$ 55.00	\$ 39.24	\$ 55.00	\$ 41.70	\$ 55.00	\$ 40.79	\$ 55.00	\$ 42.17		
LQTHB	ISDN individual business service		\$ 60.00	\$ 45.20	\$ 55.00	\$ 41.18	\$ 55.00	\$ 40.94	\$ 55.00	\$ 41.81			\$ 55.00	\$ 41.70			\$ 55.00	\$ 42.17		
	RATE ISPN ACCESS (5ESS/DMS)																			
LQTTB	BUSINESS, month to month		\$ 60.00	\$ 45.20	\$ 55.00	\$ 41.18	\$ 55.00	\$ 40.94	\$ 55.00	\$ 41.81	\$ 55.00	\$ 39.24	\$ 55.00	\$ 41.70	\$ 55.00	\$ 40.79	\$ 55.00	\$ 42.17		$\leftarrow$
LTBEB	ARRANGEMENT, month to month INDIVIDUAL LINE ISDN BUSINESS, month to		\$ 60.00	\$ 45.20	\$ 55.00	\$ 41.18					\$ 55.00	\$ 39.24	\$ 55.00	\$ 41.70	\$ 55.00	\$ 40.79	\$ 55.00	\$ 42.17	\$ 55.00	\$ 41.58
LTBLB	month		\$ 60.00	\$ 45.20	\$ 55.00	\$ 41.18					\$ 55.00	\$ 39.24	\$ 55.00	\$ 41.70	\$ 55.00	\$ 40.79	\$ 55.00	\$ 42.17	\$ 55.00	\$ 41.58
LTCEB	ISDN individual business service										\$ 55.00	\$ 39.24			\$ 55.00	\$ 40.79			\$ 55.00	\$ 41.58
LTCHC	INDIVIDUAL LINE ISDN BUSINESS PLUS		\$ 60.00	\$ 45.20	\$ 55.00	\$ 41.18					\$ 55.00	\$ 39.24	\$ 55.00	\$ 41.70	\$ 55.00	\$ 40.79	\$ 55.00	\$ 42.17	\$ 55.00	\$ 41.58
LTQDX	ISDN individual business service										\$ 13.50	\$ 9.63			\$ 12.00	\$ 8.90			\$ 13.50	\$ 10.21
M1LCA	MULTISERV STATION LINK C														\$ 43.00	\$ 26.57				
M1LCE	MultiServ, station link														\$ 43.00	\$ 26.57				
M1LFA	MULTISERV STATION LINE		\$ 58.00	\$ 41.26	\$ 36.00	\$ 28.45			\$ 49.50	\$ 37.63	\$ 50.00	\$ 35.68	\$ 55.00	\$ 34.75	\$ 43.00	\$ 26.57	\$ 47.50	\$ 36.42	\$ 53.00	\$ 37.84
M1LFH	MultiServ, station link														\$ 43.00	\$ 26.57				
M1LRA	MULTISERV STATION LINK M		\$ 43.00	\$ 30.59					\$ 43.00	\$ 32.69	\$ 39.50	\$ 28.18	\$ 43.00	\$ 27.17	\$ 43.00	\$ 26.57	\$ 37.00	\$ 28.37	\$ 33.00	\$ 23.56
M1LSA	MULTISERV STATION LINK MSG RATE				\$ 19.75	\$ 15.61														
M1M1A	MultiServ, feature group	monthly	\$ 6.20	\$ 4.41	\$ 6.20	\$ 4.90	\$ 6.20	\$ 3.85	\$ 6.20	\$ 4.71	\$ 6.20	\$ 4.42	\$ 6.20	\$ 3.92	\$ 7.00	\$ 4.33	\$ 6.20	\$ 4.75	\$ 6.20	\$ 4.43
M1M52	MultiServ, feature group	monthly	\$ 8.40	\$ 5.98	\$ 8.40	\$ 6.64	\$ 8.40	\$ 5.21	-			\$ 5.99	\$ 8.40		\$ 9.50	\$ 5.87	\$ 8.40	\$ 6.44		
M1M56	MultiServ, feature group	monthly	\$ 7.05	\$ 5.02	\$ 7.05	\$ 5.57	\$ 7.05	\$ 4.37	\$ 7.05	\$ 5.36	\$ 7.05	\$ 5.03	\$ 7.05	\$ 4.45	\$ 8.00	\$ 4.94	\$ 7.05	\$ 5.41	\$ 7.05	\$ 5.03
M1M59	MultiServ, feature group	monthly	\$ 8.40	\$ 5.98	\$ 8.40	\$ 6.64	\$ 8.40	\$ 5.21	\$ 8.40	\$ 6.39	\$ 8.40	\$ 5.99	\$ 8.40	\$ 5.31	\$ 9.50	\$ 5.87	\$ 8.40	\$ 6.44	\$ 8.40	\$ 6.00
M1M5A	MultiServ, feature group	monthly	\$ 6.20	\$ 4.41	\$ 6.20	\$ 4.90	\$ 6.20	\$ 3.85	\$ 6.20	\$ 4.71	\$ 6.20	\$ 4.42	\$ 6.20	\$ 3.92	\$ 7.00	\$ 4.33	\$ 6.20	\$ 4.75	\$ 6.20	\$ 4.43
M1MD2	MultiServ, feature group	monthly	\$ 8.40	\$ 5.98	\$ 8.40	\$ 6.64	\$ 8.40	\$ 5.21	\$ 8.40	\$ 6.39			\$ 8.40	\$ 5.31	\$ 9.50	\$ 5.87	\$ 8.40	\$ 6.44	\$ 8.40	\$ 6.00
M1MD3	MultiServ, feature group	monthly	\$ 8.35	\$ 5.94	\$ 8.35	\$ 6.60	\$ 8.35	\$ 5.18	\$ 8.35	\$ 6.35	\$ 8.35	\$ 5.96	\$ 8.35	\$ 5.28	\$ 9.45	\$ 5.84	\$ 8.35	\$ 6.40	\$ 8.35	\$ 5.96
M1MD6	MultiServ, feature group	monthly	\$ 7.05	\$ 5.02	\$ 7.05	\$ 5.57	\$ 7.05	\$ 4.37	\$ 7.05	\$ 5.36			\$ 7.05	\$ 4.45	\$ 8.00	\$ 4.94	\$ 7.05	\$ 5.41	\$ 7.05	\$ 5.03
M1MDA	MultiServ, feature group	monthly	\$ 6.20	\$ 4.41	\$ 6.20	\$ 4.90	\$ 6.20	\$ 3.85		\$ 4.71		· -	\$ 6.20	\$ 3.92	\$ 7.00	\$ 4.33	\$ 6.20	\$ 4.75	\$ 6.20	\$ 4.43
M1MDB	MultiServ, feature group	monthly	\$ 7.20	\$ 5.12	\$ 7.20	\$ 5.69	\$ 7.20	\$ 4.47	\$ 7.20	\$ 5.47			\$ 7.20	\$ 4.55	<u> </u>	\$ 4.76	\$ 7.20	\$ 5.52	\$ 7.20	\$ 5.14
M9Q1R	NARS, MultiServ/Centrex, one way inward	1	\$ 6.57	\$ 5.50					\$ 10.00	\$ 8.28					\$ 24.42	\$ 15.09			\$ 10.00	\$ 7.56
M9Q1R	NARS, MultiServ/Centrex, one way inward	2	\$ 6.57	\$ 5.50					\$ 10.00	\$ 8.28					\$ 25.04	\$ 15.47			\$ 10.00	\$ 7.56
M9Q1R	NARS, MultiServ/Centrex, one way inward	3	\$ 6.57	\$ 5.50					\$ 10.00	\$ 8.28					\$ 25.68	\$ 15.87			\$ 10.00	\$ 7.56
M9Q1R	NARS, MultiServ/Centrex, one way inward	4	\$ 6.57	\$ 5.50					\$ 10.00	\$ 8.28					\$ 26.23	\$ 16.21			\$ 10.00	\$ 7.14
M9Q1R	NARS, MultiServ/Centrex, one way inward	5	\$ 6.57	\$ 5.50					\$ 10.00	\$ 6.76					\$ 26.89	\$ 16.62			\$ 10.00	\$ 6.72
M9Q1R	NARS, MultiServ/Centrex, one way inward	6	\$ 6.57	\$ 4.67							\$ 7.00	+			\$ 27.64	\$ 17.08				$\vdash$
M9Q1R	NARS, MultiServ/Centrex, one way inward	7									\$ 7.00				\$ 28.29	\$ 17.48				$\vdash$
M9Q1R	NARS, MultiServ/Centrex, one way inward	8									\$ 7.00				\$ 28.97	\$ 17.90				
M9Q1R	NARS, MultiServ/Centrex, one way inward	9									\$ 7.00	\$ 4.99			\$ 29.38	\$ 18.16				
M9Q1R	NARS, MultiServ/Centrex, one way inward NETWORK ACCESS REGISTER, MultiServ	10													\$ 29.38	\$ 18.16				
M9Q1S	Plus/Centrex, one way inward NETWORK ACCESS REGISTER, MultiServ	1			\$ 6.78	\$ 5.64														
M9Q1S	Plus/Centrex, one way inward NETWORK ACCESS REGISTER, MultiServ	2			\$ 6.78	\$ 5.64	\$ 17.36	\$ 12.20												
M9Q1S	Plus/Centrex, one way inward	3			\$ 6.78	\$ 5.64														
M9Q1S	Plus/Centrex, one way inward	4			\$ 7.02	\$ 5.84														
M9Q1S	Plus/Centrex, one way inward	5			\$ 7.26	\$ 6.04	\$ 22.66	\$ 15.93												
M9Q1S	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, one way inward	6			\$ 7.52	\$ 5.94														
M9Q1S	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, one way inward	7			\$ 7.73	\$ 6.43	\$ 26.88	\$ 16.67												
	NETWORK ACCESS REGISTER, MultiServ	8					ψ ∠0.68	φ 10.07												
M9Q1S	Plus/Centrex, one way inward	ď			\$ 7.94	\$ 6.61														

USOC			AL-Tariff	AL-V&T	FL-Tariff	FL-V&T	GA-Tariff	GA-V&T	KY-Tariff	KY-V&T	LA-Tariff	LA-V&T	MS-Tariff		NC-Tariff	NC-V&T	SC-Tariff	SC-V&T	TN-Tariff	TN-V&T
Code	Description	Rate Group	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	MS-V&T Rate	Rate	Rate	Rate	Rate	Rate	Rate
M9Q1S	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, one way inward	9			\$ 8.14	\$ 6.09														
M9Q1S	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, one way inward	10			\$ 8.29	\$ 6.21														
M9Q1S	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, one way inward NETWORK ACCESS REGISTER, MultiServ	11			\$ 8.43	\$ 6.31														
M9Q1S	Plus/Centrex, one way inward	12			\$ 8.56	\$ 6.41	\$ 39.24	\$ 22.72												
M9Q1X	NARS, MultiServ/Centrex, one way inward NETWORK ACCESS REGISTER, MultiServ	1	\$ 24.00	\$ 20.09	\$ 11.74	\$ 9.77			\$ 21.58	\$ 17.86	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 16.60	\$ 12.73	\$ 52.25	\$ 39.50
M9Q1X	Plus/Centrex, one way inward	2	\$ 24.00	\$ 20.09	\$ 12.34	\$ 10.27	\$ 17.36	\$ 12.20	\$ 21.58	\$ 17.86	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20					\$ 52.25	\$ 39.50
M9Q1X	NARS, MultiServ/Centrex, one way inward	3	\$ 24.00	\$ 20.09	\$ 12.98	\$ 10.80			\$ 21.58	\$ 17.86	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 19.46	\$ 14.92	\$ 52.25	\$ 39.50
M9Q1X	NARS, MultiServ/Centrex, one way inward	4	\$ 24.00	\$ 20.09	\$ 13.58	\$ 11.30			\$ 21.58	\$ 17.86	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 20.89	\$ 16.02	\$ 52.25	\$ 37.31
M9Q1X	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, one way inward	5	\$ 24.00	\$ 20.09	\$ 14.14	\$ 11.76	\$ 22.66	\$ 15.93	\$ 21.58	\$ 14.58	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20					\$ 52.25	\$ 35.11
M9Q1X	NARS, MultiServ/Centrex, one way inward	6	\$ 24.00	\$ 17.07	\$ 14.77	\$ 11.67	Ψ 22.00	Ψ 10.50	Ψ 21.50	ψ 14.50	Ψ 20.00	ψ 13.00	ψ 24.00	Ψ 10.20			\$ 23.74	\$ 18.20	ψ 02.20	ψ 30.11
M9Q1X	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, one way inward	7			\$ 16.78	\$ 13.96	\$ 26.88	\$ 16.67												
M9Q1X	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, one way inward	8			\$ 15.78	\$ 13.13														
M9Q1X	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, one way inward	9			\$ 16.25	\$ 12.17														
M9Q1X	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, one way inward	10			\$ 16.60	\$ 12.43														
M9Q1X	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, one way inward	11			\$ 16.96	\$ 12.70														
M9Q1X	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, one way inward	12			\$ 17.26	\$ 12.92														
M9QCR	NARS, MultiServ/Centrex, combination	1	\$ 6.57	\$ 5.50					\$ 10.00	\$ 8.28	\$ 7.00	\$ 5.55			\$ 24.42	\$ 15.09			\$ 10.00	\$ 7.56
M9QCR	NARS, MultiServ/Centrex, combination	2	\$ 6.57	\$ 5.50					\$ 10.00	\$ 8.28	\$ 7.00	\$ 5.55			\$ 25.04	\$ 15.47			\$ 10.00	\$ 7.56
M9QCR	NARS, MultiServ/Centrex, combination	3	\$ 6.57	\$ 5.50					\$ 10.00	\$ 8.28	\$ 7.00	\$ 5.55			\$ 25.68	\$ 15.87			\$ 10.00	\$ 7.56
M9QCR	NARS, MultiServ/Centrex, combination	4	\$ 6.57	\$ 5.50					\$ 10.00	\$ 8.28	\$ 7.00	\$ 5.55			\$ 26.23	\$ 16.21			\$ 10.00	\$ 7.14
M9QCR	NARS, MultiServ/Centrex, combination	5	\$ 6.57	\$ 5.50					\$ 10.00	\$ 6.76	\$ 7.00	\$ 5.55			\$ 26.89	\$ 16.62			\$ 10.00	\$ 6.72
M9QCR	NARS, MultiServ/Centrex, combination	6	\$ 6.57	\$ 4.67							\$ 7.00	\$ 5.55			\$ 27.64	\$ 17.08				
M9QCR	NARS, MultiServ/Centrex, combination	7									\$ 7.00	\$ 5.55			\$ 28.29	\$ 17.48				
M9QCR	NARS, MultiServ/Centrex, combination	8									\$ 7.00	\$ 5.55			\$ 28.97	\$ 17.90				
M9QCR	NARS, MultiServ/Centrex, combination	9									\$ 7.00	\$ 4.99			\$ 29.38	\$ 18.16				
M9QCR	NARS, MultiServ/Centrex, combination NETWORK ACCESS REGISTER, MultiServ	10													\$ 29.38	\$ 18.16				
M9QCS	Plus/Centrex, both way	1			\$ 6.78	\$ 5.64														
M9QCS	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, both way	2			\$ 6.78	\$ 5.64	\$ 8.04	\$ 5.65												
M9QCS	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, both way	3			\$ 6.78	\$ 5.64														
M9QCS	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, both way	4			\$ 7.02	\$ 5.84														
M9QCS	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, both way	5			\$ 7.26	\$ 6.04	\$ 10.49	\$ 7.37												
M9QCS	Plus/Centrex, both way	6			\$ 7.52	\$ 5.94														
M9QCS	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, both way	7			\$ 7.73	\$ 6.43	\$ 12.45	\$ 7.72												
M9QCS	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, both way	8			\$ 7.94	\$ 6.61														
M9QCS	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, both way	9			\$ 8.14	\$ 6.09														
M9QCS	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, both way	10			\$ 8.29	\$ 6.21														
M9QCS	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, both way	11			\$ 8.43	\$ 6.31														
M9QCS	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, both way	12			\$ 8.56	\$ 6.41	\$ 19.47	\$ 11.27												

USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
M9QCX	NARS, MultiServ/Centrex, combination	1	\$ 24.00	\$ 20.09	\$ 11.74	\$ 9.77			\$ 21.50	\$ 17.80	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 16.60	\$ 12.73	\$ 52.25	\$ 39.50
M9QCX	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, both way	2	\$ 24.00	\$ 20.09	\$ 12.34	\$ 10.27	\$ 17.36	\$ 12.20	\$ 21.50	\$ 17.80	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20					\$ 52.25	\$ 39.50
M9QCX	NARS, MultiServ/Centrex, combination	3	\$ 24.00	\$ 20.09	\$ 12.98	\$ 10.80	•		\$ 21.50	\$ 17.80	\$ 25.08		\$ 24.00	•			\$ 19.46	\$ 14.92	\$ 52.25	\$ 39.50
M9QCX	NARS, MultiServ/Centrex, combination	4	\$ 24.00	\$ 20.09	\$ 13.58	\$ 11.30			\$ 21.50	\$ 17.80	\$ 25.08						\$ 20.89	\$ 16.02	\$ 52.25	\$ 37.31
	NETWORK ACCESS REGISTER, MultiServ	_																		
M9QCX	Plus/Centrex, both way	5	\$ 24.00	\$ 20.09		•	\$ 22.66	\$ 15.93			\$ 25.08		\$ 24.00	\$ 18.20					\$ 52.25	\$ 35.11
M9QCX	NARS, MultiServ/Centrex, combination NETWORK ACCESS REGISTER, MultiServ	6	\$ 24.00	\$ 17.07	\$ 14.77	\$ 11.67					\$ 25.08	\$ 19.88					\$ 23.74	\$ 18.20		
м9QCX	Plus/Centrex, both way	7			\$ 16.78	\$ 13.96	\$ 26.88	\$ 16.67			\$ 25.08	\$ 19.88								
MOOOV	NETWORK ACCESS REGISTER, MultiServ	8			e 45.70	f 40.40					¢ 05.00	f 40.00								
M9QCX	Plus/Centrex, both way NETWORK ACCESS REGISTER, MultiServ	0			\$ 15.78	\$ 13.13					\$ 25.08	\$ 19.88								
M9QCX	Plus/Centrex, both way	9			\$ 16.25	\$ 12.17					\$ 25.08	\$ 17.90								
M9QCX	NETWORK ACCESS REGISTER, MultiServ Plus/Centrex, both way	10			\$ 16.60	\$ 12.43														
Modox	NETWORK ACCESS REGISTER, MultiServ				ψ 10.00	ψ 12.40														
M9QCX	Plus/Centrex, both way NETWORK ACCESS REGISTER, MultiServ	11			\$ 16.96	\$ 12.70														
M9QCX	Plus/Centrex, both way	12			\$ 17.26	\$ 12.92	\$ 39.24	\$ 22.72												
M9QOR	NARS, MultiServ/Centrex, one way outward	1	\$ 6.57	\$ 5.50	*		* ****	*	\$ 10.00	\$ 8.28	\$ 7.00	\$ 5.55			\$ 24.42	\$ 15.09			\$ 10.00	\$ 7.56
M9QOR	NARS, MultiServ/Centrex, one way outward	2	\$ 6.57	\$ 5.50					\$ 10.00	\$ 8.28	\$ 7.00	\$ 5.55			\$ 25.04	\$ 15.47			\$ 10.00	\$ 7.56
M9QOR	NARS, MultiServ/Centrex, one way outward	3	\$ 6.57	\$ 5.50					\$ 10.00	\$ 8.28	\$ 7.00	\$ 5.55			\$ 25.68	\$ 15.87			\$ 10.00	\$ 7.56
M9QOR	NARS, MultiServ/Centrex, one way outward	4	\$ 6.57	\$ 5.50					\$ 10.00	\$ 8.28	\$ 7.00	\$ 5.55			\$ 26.23	\$ 16.21			\$ 10.00	\$ 7.14
M9QOR	NARS, MultiServ/Centrex, one way outward	5	\$ 6.57	\$ 5.50					\$ 10.00	\$ 6.76	\$ 7.00	\$ 5.55			\$ 26.89	\$ 16.62			\$ 10.00	\$ 6.72
M9QOR	NARS, MultiServ/Centrex, one way outward	6	\$ 6.57	\$ 4.67							\$ 7.00	\$ 5.55			\$ 27.64	\$ 17.08				
M9QOR	NARS, MultiServ/Centrex, one way outward	7									\$ 7.00	\$ 5.55			\$ 28.29	\$ 17.48				
M9QOR	NARS, MultiServ/Centrex, one way outward	8									\$ 7.00	\$ 5.55			\$ 28.97	\$ 17.90				
M9QOR	NARS, MultiServ/Centrex, one way outward	9									\$ 7.00	\$ 4.99			\$ 29.38	\$ 18.16				
M9QOR	NARS, MultiServ/Centrex, one way outward	10													\$ 29.38	\$ 18.16				
140000	NARS, MultiServ Plus, one way outward, message rate	1			e 0.70	A 504														
M9QOS	NARS, MultiServ Plus, one way outward,	'			\$ 6.78	\$ 5.64														
M9QOS	message rate	2			\$ 6.78	\$ 5.64														
M9QOS	NARS, MultiServ Plus, one way outward, message rate	3			\$ 6.78	\$ 5.64														
WieQOS	NARS, MultiServ Plus, one way outward,				ψ 0.70	ψ 5.04														
M9QOS	message rate	4			\$ 7.02	\$ 5.84														
M9QOS	NARS, MultiServ Plus, one way outward, message rate	5			\$ 7.26	\$ 6.04														
	NARS, MultiServ Plus, one way outward,	_																		
M9QOS	message rate  NARS, MultiServ Plus, one way outward,	6			\$ 7.52	\$ 5.94														
M9QOS	message rate	7			\$ 7.73	\$ 6.43														
	NARS, MultiServ Plus, one way outward,	8																		
M9QOS	message rate  NARS, MultiServ Plus, one way outward,	0			\$ 7.94	\$ 6.61														
M9QOS	message rate	9			\$ 8.14	\$ 6.09														
M9QOS	MARS, MultiServ Plus, one way outward, message rate	10			\$ 8.29	\$ 6.21														
WISQUS	NARS, MultiServ Plus, one way outward,	10			\$ 0.29	Φ 0.21														
M9QOS	message rate	11			\$ 8.43	\$ 6.31														
M9QOS	MARS, MultiServ Plus, one way outward, message rate	12			\$ 8.56	\$ 6.41														
M9QOX	NARS, MultiServ/Centrex, one way inward	1	\$ 24.00	\$ 20.09	\$ 11.74	\$ 9.77			\$ 21.50	\$ 17.80	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 16.60	\$ 12.73	\$ 52.25	\$ 39.50
	NETWORK ACCESS REGISTER, MultiServ						_													
M9QOX	Plus/Centrex, one way outward	2	\$ 24.00	\$ 20.09	\$ 12.34	\$ 10.27	\$ 17.36	\$ 12.20		\$ 18.16	\$ 25.08								\$ 52.25	\$ 39.50
M9QOX	NARS, MultiServ/Centrex, one way inward	3	\$ 24.00	\$ 20.09	\$ 12.98	\$ 10.80			\$ 21.50	\$ 17.80	\$ 25.08		\$ 24.00				\$ 19.46	\$ 14.92	\$ 52.25	\$ 39.50
M9QOX	NARS, MultiServ/Centrex, one way inward	4	\$ 24.00	\$ 20.09	\$ 13.58	\$ 11.30	_		\$ 21.50	\$ 17.80	\$ 25.08		\$ 24.00	\$ 18.20			\$ 20.89	\$ 16.02	\$ 52.25	\$ 37.31
M9QOX	NARS, MultiServ/Centrex, one way inward	5	\$ 24.00	\$ 20.09	\$ 14.14	\$ 11.76	\$ 22.66	\$ 15.93	\$ 21.50	\$ 14.53	\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 22.31	\$ 17.11	\$ 52.25	\$ 35.11

USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
M9QOX	NARS, MultiServ/Centrex, one way inward	6	\$ 24.00	\$ 17.07	\$ 14.77	\$ 11.67					\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 23.74	\$ 18.20		
M9QOX	NARS, MultiServ/Centrex, one way inward	7			\$ 16.78	\$ 13.96	\$ 26.88	\$ 16.67			\$ 25.08	\$ 19.88	\$ 24.00	\$ 18.20			\$ 25.17	\$ 17.16		
M9QOX	NARS, MultiServ/Centrex, one way inward	8			\$ 15.78	\$ 13.13					\$ 25.08	\$ 19.88								
M9QOX	NARS, MultiServ/Centrex, one way inward	9			\$ 16.25	\$ 12.17					\$ 25.08	\$ 17.90								
M9QOX	NARS, MultiServ/Centrex, one way inward	10			\$ 16.60	\$ 12.43														
M9QOX	NARS, MultiServ/Centrex, one way inward	11			\$ 16.96	\$ 12.70														
M9QOX	NARS, MultiServ/Centrex, one way inward	12			\$ 17.26	\$ 12.92	\$ 39.24	\$ 22.72												
MNCCF	ISDN individual business service										\$ 12.25	\$ 8.74			\$ 12.25	\$ 9.08			\$ 12.25	\$ 9.26
MNCCR	ISDN individual business service										\$ 2.50	\$ 1.78							\$ 2.50	\$ 1.89
MWW	Message waiting indication		\$ 0.75	\$ 0.38	\$ 0.75	\$ 0.44	\$ 0.75	\$ 0.37	\$ 0.75	\$ 0.38	\$ 0.75	\$ 0.39	\$ 0.75	\$ 0.38	\$ 0.75	\$ 0.37	\$ 0.75	\$ 0.42	\$ 0.75	\$ 0.38
MWWAV	Message waiting indication audio/visual		\$ 0.75	\$ 0.31	\$ 0.75	\$ 0.37	\$ 0.75	\$ 0.31	\$ 0.75	\$ 0.32	\$ 0.75	\$ 0.33	\$ 0.75	\$ 0.32	\$ 0.75	\$ 0.31	\$ 0.75	\$ 0.35	\$ 0.75	\$ 0.32
NCE	Preferred Call Forwarding - per line		\$ 5.00	\$ 2.09	\$ 5.00	\$ 2.50	\$ 4.95	\$ 2.05	\$ 4.95	\$ 2.09	\$ 4.50	\$ 1.96	\$ 4.95	\$ 2.09	\$ 4.50	\$ 1.85	\$ 6.00	\$ 2.81	\$ 3.50	\$ 1.47
NDT	PBX SVC: DID TRUNK TERMINATION		\$ 20.00	\$ 13.39	\$ 21.80	\$ 16.32			\$ 26.00	\$ 17.57	\$ 27.17	\$ 19.39	\$ 26.00	\$ 19.71	\$ 25.85	\$ 15.98	\$ 30.00	\$ 20.45	\$ 20.00	\$ 13.44
NSD	Caller ID - Basic per line		\$ 10.00	\$ 4.19	\$ 10.00	\$ 4.99	\$ 7.00	\$ 2.89	\$ 9.05	\$ 3.82	\$ 11.00	\$ 4.80	\$ 9.00	\$ 3.79	\$ 7.50	\$ 3.09	\$ 20.00	\$ 9.37	\$ 10.00	\$ 4.20
NSK	Call Selector - per line		\$ 4.50	\$ 1.88	\$ 4.50	\$ 2.25	\$ 4.50	\$ 1.86	\$ 4.95	\$ 2.09	\$ 4.50	\$ 1.96	\$ 4.50	\$ 1.90	\$ 4.50	\$ 1.85	\$ 6.00	\$ 2.81	\$ 3.50	\$ 1.47
NSQ	Repeat Dialing - per line		\$ 4.50	\$ 1.88	\$ 5.00	\$ 2.50	\$ 5.00	\$ 2.07	\$ 4.95	\$ 2.09	\$ 5.00	\$ 2.18	\$ 4.50	\$ 1.90	\$ 4.50	\$ 1.85	\$ 6.00	\$ 2.81	\$ 5.50	\$ 2.31
NSS	Call Return - Per Line		\$ 5.20	\$ 2.18	\$ 6.00	\$ 2.99	\$ 6.00	\$ 2.48	\$ 5.20	\$ 2.20	\$ 5.00	\$ 2.18	\$ 5.75	\$ 2.42	\$ 4.95	\$ 2.04	\$ 6.00	\$ 2.81	\$ 5.50	\$ 2.31
NST	Call Tracing - per line		\$ 5.00	\$ 2.09	\$ 5.00	\$ 2.50	\$ 4.95	\$ 2.05	\$ 5.50	\$ 2.32	\$ 5.00	\$ 2.18	\$ 4.95	\$ 2.09	\$ 5.00	\$ 2.06	\$ 6.00	\$ 2.81	\$ 5.50	\$ 2.31
NSY	Call Block - per line		\$ 4.50	\$ 1.88	\$ 4.50	\$ 2.25	\$ 4.50	\$ 1.86	\$ 4.95	\$ 2.09	\$ 4.50	\$ 1.96	\$ 4.50	\$ 1.90	\$ 4.50	\$ 1.85	\$ 6.00	\$ 2.81	\$ 5.50	\$ 2.31
NWE1X	NAR, ESSX, Inward only	1													\$ 24.42	\$ 15.09				
	COMMUNITY CALLER PLUS SERVICE,																			
NWE1X	NAR, ESSX, Inward only COMMUNITY CALLER PLUS SERVICE,	2													\$ 25.04	\$ 15.47				
NWE1X	NAR, ESSX, Inward only	3													\$ 25.68	\$ 15.87				
	COMMUNITY CALLER PLUS SERVICE,																			
NWE1X	NAR, ESSX, Inward only COMMUNITY CALLER PLUS SERVICE,	4													\$ 26.23	\$ 16.21				
NWE1X	NAR, ESSX, Inward only	5													\$ 26.89	\$ 16.62				
NWE1X	NAR, ESSX, Inward only	6													\$ 27.64	\$ 17.08				
INVVEIX	COMMUNITY CALLER PLUS SERVICE,	0													\$ 27.04	\$ 17.08				
NWE1X	NAR, ESSX, Inward only	7													\$ 28.29	\$ 17.48				
NWE1X	NAR, ESSX, Inward only	8													\$ 28.97	\$ 17.90				
	COMMUNITY CALLER PLUS SERVICE,																			
NWE1X	NAR, ESSX, Inward only COMMUNITY CALLER PLUS SERVICE,	9													\$ 29.38	\$ 18.16				
NWE1X	NAR, ESSX, Inward only	10													\$ 29.38	\$ 18.16				
	COMMUNITY CALLER PLUS SERVICE, NAR, ESSX, Combination	1																		
NWECX	COMMUNITY CALLER PLUS SERVICE,	'													\$ 24.42	\$ 15.09				
NWECX	NAR, ESSX, Combination	2													\$ 25.04	\$ 15.47				
NWECX	COMMUNITY CALLER PLUS SERVICE, NAR, ESSX, Combination	3													\$ 25.68	\$ 15.87				
INVECX	COMMUNITY CALLER PLUS SERVICE,	-													φ 25.00	φ 15.67				
NWECX	NAR, ESSX, Combination	4													\$ 26.23	\$ 16.21				
NWECX	COMMUNITY CALLER PLUS SERVICE, NAR, ESSX, Combination	5													\$ 26.89	\$ 16.62				
	COMMUNITY CALLER PLUS SERVICE,																			
NWECX	NAR, ESSX, Combination COMMUNITY CALLER PLUS SERVICE,	6													\$ 27.64	\$ 17.08				
NWECX	NAR, ESSX, Combination	7													\$ 28.29	\$ 17.48				
	COMMUNITY CALLER PLUS SERVICE,	_																		
NWECX	NAR, ESSX, Combination COMMUNITY CALLER PLUS SERVICE,	8													\$ 28.97	\$ 17.90				
NWECX	NAR, ESSX, Combination	9													\$ 29.38	\$ 18.16				
NN4501:	COMMUNITY CALLER PLUS SERVICE,	10													0 000-	<b>6</b> 10.1-				
NWECX	NAR, ESSX, Combination	10									l				\$ 29.38	\$ 18.16				

USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
Code	COMMUNITY CALLER PLUS SERVICE,	Nate Group	Nate	Nate	Nate	Nate	Nate	Nate	Nate	Nate	Nate	Nate	Nate	IVIO-VOT Kate	Nate	Nate	Nate	ivate	Nate	Nate
NWEOX	NAR, ESSX, Outdial	1													\$ 24.42	\$ 15.09				
NWEOX	COMMUNITY CALLER PLUS SERVICE, NAR, ESSX, Outdial	2													\$ 25.04	\$ 15.47				
	COMMUNITY CALLER PLUS SERVICE,																			
NWEOX	NAR, ESSX, Outdial COMMUNITY CALLER PLUS SERVICE,	3													\$ 25.68	\$ 15.87				
NWEOX	NAR, ESSX, Outdial	4													\$ 26.23	\$ 16.21				
NWEOX	COMMUNITY CALLER PLUS SERVICE, NAR, ESSX, Outdial	5													\$ 26.89	\$ 16.62				
	COMMUNITY CALLER PLUS SERVICE,																			
NWEOX	NAR, ESSX, Outdial COMMUNITY CALLER PLUS SERVICE,	6													\$ 27.64	\$ 17.08				
NWEOX	NAR, ESSX, Outdial	7													\$ 28.29	\$ 17.48				
NWEOX	COMMUNITY CALLER PLUS SERVICE, NAR, ESSX, Outdial	8													\$ 28.97	\$ 17.90				
	COMMUNITY CALLER PLUS SERVICE, NAR, ESSX, Outdial	9																		
NWEOX	COMMUNITY CALLER PLUS SERVICE,	9													\$ 29.38	\$ 18.16				
NWEOX	NAR, ESSX, Outdial	10													\$ 29.38	\$ 18.16				
NWO1X	COMMUNITY CALLER PLUS SERVICE, NAR, Inward only	1													\$ 24.42	\$ 15.09				
NWO1X	COMMUNITY CALLER PLUS SERVICE, NAR, Inward only	2													\$ 25.04	\$ 15.47				
NWOTX	COMMUNITY CALLER PLUS SERVICE,														\$ 25.04	\$ 15.47				
NWO1X	NAR, Inward only COMMUNITY CALLER PLUS SERVICE,	3													\$ 25.68	\$ 15.87				
NWO1X	NAR, Inward only	4													\$ 26.23	\$ 16.21				
NWO1X	NAR, Inward only	5													\$ 26.89	\$ 16.62				
	COMMUNITY CALLER PLUS SERVICE,														ψ 20.09	ÿ 10.02				
NWO1X	NAR, Inward only COMMUNITY CALLER PLUS SERVICE,	6													\$ 27.64	\$ 17.08				
NWO1X	NAR, Inward only	7													\$ 28.29	\$ 17.48				
NWO1X	NAR, Inward only	8													\$ 28.97	\$ 17.90				
	COMMUNITY CALLER PLUS SERVICE,																			
NWO1X	NAR, Inward only COMMUNITY CALLER PLUS SERVICE,	9													\$ 29.38	\$ 18.16				
NWO1X	NAR, Inward only	10													\$ 29.38	\$ 18.16				
NWOCX	NAR, Combination	1													\$ 24.42	\$ 15.09				
	COMMUNITY CALLER PLUS SERVICE,	_																		
NWOCX	NAR, Combination COMMUNITY CALLER PLUS SERVICE,	2													\$ 25.04	\$ 15.47				
NWOCX	NAR, Combination COMMUNITY CALLER PLUS SERVICE,	3													\$ 25.68	\$ 15.87				
NWOCX	NAR, Combination	4													\$ 26.23	\$ 16.21				
NWOCX	COMMUNITY CALLER PLUS SERVICE, NAR, Combination	5													\$ 26.89	\$ 16.62				
	COMMUNITY CALLER PLUS SERVICE,														\$ 20.09					
NWOCX	NAR, Combination COMMUNITY CALLER PLUS SERVICE,	6													\$ 27.64	\$ 17.08				
NWOCX	NAR, Combination	7													\$ 28.29	\$ 17.48				
NWOCX	COMMUNITY CALLER PLUS SERVICE, NAR, Combination	8													\$ 28.97	\$ 17.90				
	COMMUNITY CALLER PLUS SERVICE,	9																		
NWOCX	NAR, Combination COMMUNITY CALLER PLUS SERVICE,	9													\$ 29.38	\$ 18.16				
NWOCX	NAR, Combination COMMUNITY CALLER PLUS SERVICE,	10													\$ 29.38	\$ 18.16				
NWOOX	NAR, Outdial	1													\$ 24.42	\$ 15.09				
NWOOX	COMMUNITY CALLER PLUS SERVICE, NAR, Outdial	2													\$ 25.04	\$ 15.47				
	COMMUNITY CALLER PLUS SERVICE,																			
NWOOX	NAR, Outdial	3													\$ 25.68	\$ 15.87				

						I									=					
USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
NWOOX	COMMUNITY CALLER PLUS SERVICE, NAR, Outdial	4													\$ 26.23	\$ 16.21				
NWOOX	COMMUNITY CALLER PLUS SERVICE, NAR, Outdial	5													\$ 26.89					
	COMMUNITY CALLER PLUS SERVICE,																			
NWOOX	NAR, Outdial COMMUNITY CALLER PLUS SERVICE,	6													\$ 27.64	\$ 17.08				
NWOOX	NAR, Outdial COMMUNITY CALLER PLUS SERVICE,	7													\$ 28.29	\$ 17.48				
NWOOX	NAR, Outdial	8													\$ 28.97	\$ 17.90				
NWOOX	COMMUNITY CALLER PLUS SERVICE, NAR, Outdial	9													\$ 29.38	\$ 18.16				
NWOOX	COMMUNITY CALLER PLUS SERVICE, NAR, Outdial	10													\$ 29.38	\$ 18.16				
	Call Selector, Business, Additional Service, per												_							
NX2	line Call block business and residential service per										\$ 4.50	\$ 1.96	\$ 4.50	\$ 1.90	\$ 3.50	\$ 1.44				
NX5	line Preferred call forwarding, business, additional										\$ 4.50	\$ 1.96	\$ 4.30	\$ 1.81	\$ 3.50	\$ 1.44				
NX6	service, per line										\$ 4.50	\$ 1.96	\$ 5.00	\$ 2.11	\$ 3.50	\$ 1.44				
NX8	Call return business and residential, additional service per line										\$ 5.00	\$ 2.18	\$ 5.30	\$ 2.23	\$ 3.50	\$ 1.44				
NX9	Repeat dial business and residential additional service per line										\$ 5.00					\$ 1.44				
NXECR	Enhanced Caller ID with ACR - per line		\$ 15.99	\$ 6.69	\$ 15.99	\$ 7.98	\$ 15.95	\$ 6.60	\$ 15.95	\$ 6.74	\$ 16.00		\$ 15.95	\$ 6.72	\$ 15.95	\$ 6.57	\$ 25.00	\$ 11.72	\$ 15.99	\$ 6.72
NXMCR	Caller ID - Deluxe with ACR per line		\$ 10.00	\$ 5.02			\$ 7.50	\$ 3.72		\$ 5.07	\$ 11.00			-		\$ 4.94	\$ 20.00	\$ 11.08	\$ 11.00	\$ 5.54
NXMMN	Caller ID - Deluxe without ACR per line for multi- line hunt group arrangement		\$ 10.00	\$ 4.19	\$ 11.00	\$ 5.49	\$ 7.50	\$ 3.10	\$ 10.00	\$ 4.22	\$ 11.00	\$ 4.80	\$ 11.00	\$ 4.63	\$ 10.00	\$ 4.12	\$ 20.00	\$ 9.37	\$ 11.00	\$ 4.62
PBC	Community Caller Plus Service, complex business, without inward call billing	1													\$ 26.81	\$ 16.57				
PBC	Community Caller Plus Service, complex business, without inward call billing	2													\$ 27.50	\$ 17.00				
PBC	Community Caller Plus Service, complex business, without inward call billing	3														\$ 17.44				
	Community Caller Plus Service, complex															•				
PBC	business, without inward call billing Community Caller Plus Service, complex	4													\$ 28.84	\$ 17.82				
PBC	business, without inward call billing Community Caller Plus Service, complex	5													\$ 29.58	\$ 18.28				
PBC	business, without inward call billing	6													\$ 30.42	\$ 18.80				
PBC	Community Caller Plus Service, complex business, without inward call billing	7													\$ 31.15	\$ 19.25				
	Community Caller Plus Service, complex business, without inward call billing	8																		
PBC	Community Caller Plus Service, complex														\$ 31.91	\$ 19.72				
PBC	business, without inward call billing Community Caller Plus Service, complex	9													\$ 32.83	\$ 20.29				
PBC	business, without inward call billing	10													\$ 33.89	\$ 20.94				
PBCCL	COMMUNITY CALLER PLUS Community Caller Plus Service, simple	1													\$ 33.95	\$ 20.98				
PCE	business, without inward call billing	1													\$ 26.81	\$ 16.57				
PCE	Community Caller Plus Service, simple business, without inward call billing	2													\$ 27.50	\$ 17.00				
PCE	Community Caller Plus Service, simple business, without inward call billing	3													\$ 28.22	\$ 17.44				
	Community Caller Plus Service, simple																			
PCE	business, without inward call billing Community Caller Plus Service, simple	4													\$ 28.84					
PCE	business, without inward call billing Community Caller Plus Service, simple	5													\$ 29.58	\$ 18.28				
PCE	business, without inward call billing	6													\$ 30.42	\$ 18.80				
PCE	Community Caller Plus Service, simple business, without inward call billing	7													\$ 31.15	\$ 19.25				
PCE	Community Caller Plus Service, simple business, without inward call billing	8													\$ 31.91	\$ 19.72				
. 01			l				·		1		·		·		اق.ال پ	y 10.12	l			

usoc			AL-Tariff	AL-V&T	FL-Tariff	FL-V&T	GA-Tariff	GA-V&T	KY-Tariff	KY-V&T	LA-Tariff	LA-V&T	MS-Tariff		NC-Tariff	NC-V&T	SC-Tariff	SC-V&T	TN-Tariff	TN-V&T
Code	Community Caller Plus Service, simple	Rate Group	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	MS-V&T Rate	Rate	Rate	Rate	Rate	Rate	Rate
PCE	business, without inward call billing	9													\$ 32.83	\$ 20.29				
	Community Caller Plus Service, simple	40													_	_				
PCE	business, without inward call billing EXPANDED LOCAL SERVICE, THRIFTY	10													\$ 33.89	\$ 20.94				
PCX	CALLER														\$ 15.35	\$ 9.49				
	Thrifty caller service, comples business, w/out																			
PCXCL	inward call billing Community Circle Plan, message rate service														\$ 15.35	\$ 9.49				
РМВСВ	complex business	Asheville													\$ 26.26	\$ 16.23				
DMDOD	Community Circle Plan, message rate service	Charlotte													¢ 07.05	e 40.70				
PMBCB	complex business  Community Circle Plan, message rate service	Chanotte													\$ 27.05	\$ 16.72				
PMBCB	complex business	Davidson													\$ 27.38	\$ 16.92				
PMBCB	Community Circle Plan, message rate service complex business	Greensboro													£ 27.00	\$ 16.69				
PIVIDOD	Community Circle Plan, message rate service	Greensboro													\$ 27.00	\$ 10.09				
PMBCB	complex business	Raleigh													\$ 27.08	\$ 16.74				
	Community Circle Plan, message rate service	Raleigh/Dur																		
РМВСВ	complex business	ham Airport													\$ 33.86	\$ 20.93				
	Community Circle Plan, message rate service														_					
PMBCB	complex business  Community Circle Plan, message rate service	Wilmington Winston-													\$ 25.42	\$ 15.71				
РМВСВ	complex business	Salem													\$ 27.00	\$ 16.69				
D14D0D	Community Circle Plan, message rate service	A ahayilla																		
PMBSB	simple business Community Circle Plan, message rate service	Asheville													\$ 26.26	\$ 16.23				
PMBSB	simple business	Charlotte													\$ 27.05	\$ 16.72				
DMDOD	Community Circle Plan, message rate service	Dovidoon													¢ 07.00	¢ 40.00				
PMBSB	simple business Community Circle Plan, message rate service	Davidson													\$ 27.38	\$ 16.92				
PMBSB	simple business	Greensboro													\$ 27.00	\$ 16.69				
PMBSB	community Circle Plan, message rate service simple business	Raleigh													\$ 27.08	¢ 46.74				
PIVIDOD	Simple business	Raieigii													\$ 27.06	\$ 16.74				
	Community Circle Plan, message rate service	Raleigh/Dur																		
PMBSB	simple business Community Circle Plan, message rate service	ham Airport													\$ 33.86	\$ 20.93				
PMBSB	simple business	Wilmington													\$ 25.42	\$ 15.71				
D14D0D	Community Circle Plan, message rate service	Winston-																		
PMBSB	simple business EXPANDED LOCAL SERVICE SIMPLE	Salem													\$ 27.00	\$ 16.69				
PPB	BUSINESS														\$ 14.80	\$ 9.15				
DDDOI	Thritty Caller Service, simple business, without														£ 4400	6 045				
PPBCL	inward call billing		f 400.00	A 004 00	¢ 400.00	A 000 40	<b>6</b> 400.00	A 007.70	f 400.00	r 004.00	¢ 400.00	¢ 005.44	£ 400.00	<b>6</b> 000.00	\$ 14.80	\$ 9.15	f. 400.00	¢ 000.70	£ 400.00	r 000 40
PR71C	PRIMARY RATE ISDN B-CHANN		\$ 400.00		\$ 400.00			\$ 297.72	\$ 400.00	\$ 304.06			\$ 400.00		\$ 400.00		\$ 400.00	\$ 306.72	\$ 400.00	\$ 302.40
PR71D	Primary Rate ISDN, interface, data only PRIMARY RATE ISDN INWARD		\$ 400.00		\$ 400.00	\$ 299.48	\$ 400.00	\$ 297.72	\$ 400.00	\$ 304.06	\$ 400.00	\$ 285.41	\$ 400.00	\$ 303.30	\$ 400.00	\$ 296.64	\$ 400.00	\$ 306.72	\$ 400.00	\$ 302.40
PR71E	PRIMARY RATE ISDN B-CHANN		\$ 400.00		\$ 400.00	\$ 299.48	\$ 400.00	\$ 297.72	\$ 400.00	\$ 304.06	\$ 400.00	\$ 285.41	\$ 400.00	\$ 303.30	\$ 400.00	\$ 296.64	\$ 400.00	\$ 306.72	\$ 400.00	\$ 302.40
PR71U	PRIMART RATE ISDN B-CHANN		\$ 400.00	\$ 301.32	\$ 400.00	\$ 299.48	\$ 400.00	\$ 297.72	\$ 400.00	\$ 304.06	\$ 400.00	\$ 285.41	\$ 400.00	\$ 303.30	\$ 400.00	\$ 296.64				
	PRIMARY RATE ISDN VOICE D-CHANNEL																			
PR71V	INTERFACE, VOICE/DATA STANDARD		\$ 400.00		\$ 400.00	\$ 299.48		\$ 297.72		\$ 304.06	\$ 400.00	-	\$ 400.00		\$ 400.00	-	\$ 400.00	\$ 306.72	\$ 400.00	\$ 302.40
PR7BD	PRIMARY RATE ISDN B-CHANN		\$ 33.35		\$ 29.00	\$ 21.71	\$ 29.00	\$ 21.58	\$ 41.00	\$ 31.17	\$ 37.65	\$ 26.86	\$ 31.00	\$ 23.51	\$ 37.75	\$ 28.00	\$ 31.00	\$ 23.77	\$ 41.00	\$ 31.00
PR7BE	PRIMARY RATE ISDN B-CHANNEL		\$ 43.00		\$ 38.00	\$ 28.45	\$ 38.00	\$ 28.28	\$ 50.00	\$ 38.01	\$ 47.00	\$ 33.54	\$ 50.00	\$ 37.91	\$ 45.00	\$ 33.37				
PR7BF	PRIMARY RATE ISDN B-CHANNEL		\$ 26.65	\$ 20.08	\$ 24.50	\$ 18.34	\$ 26.65	\$ 19.84	\$ 26.65	\$ 20.26	\$ 26.45	\$ 18.87	\$ 26.65	\$ 20.21	\$ 28.85	\$ 21.40	\$ 31.00	\$ 23.77	\$ 28.85	\$ 21.81
PR7BL	PRIMARY RATE ISDN B-CHANNEL		\$ 59.00	\$ 44.44	\$ 53.00	\$ 39.68	\$ 53.00	\$ 39.45	\$ 67.00	\$ 50.93	\$ 65.00	\$ 46.38	\$ 67.00	\$ 50.80	\$ 60.00	\$ 44.50				
PR7BS	PRIMARY RATE ISDN		\$ 60.00	\$ 45.20					\$ 65.80	\$ 50.02	\$ 66.00	\$ 47.09	\$ 66.00	\$ 50.04						
PR7BU	PRIMARY RATE ISDN		\$ 26.45	\$ 19.92					\$ 26.65	\$ 20.26	\$ 26.45	\$ 18.87	\$ 26.65	\$ 20.21						
PR7BV	PRIMARY RATE ISDN B-CHANNEL		\$ 60.00	\$ 45.20	\$ 48.40	\$ 36.24	\$ 78.85	\$ 58.69	\$ 65.80	\$ 50.02	\$ 66.00	\$ 47.09	\$ 66.00	\$ 50.04	\$ 59.88	\$ 44.41	\$ 70.15	\$ 53.79	\$ 57.10	\$ 43.17
	MAIN STATION LINE TERMINATION, CENTREX/ESSX OE ONLY TERMINATED																			
RXRTX	AS A PBX TRUNK		\$ 27.60	\$ 19.64	\$ 35.22	\$ 27.83			\$ 22.60	\$ 15.27	\$ 17.76	\$ 12.67	\$ 24.65	\$ 15.58	\$ 25.12	\$ 15.52	\$ 51.33	\$ 39.36	\$ 17.25	\$ 11.59
	•										•		•							

USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
		ivate Group	Nate	Nate	Rate	Nate	Rate	ivate	ivate	ivate	Nate	Nate	ivate	WO-VOI Nate			Nate	Nate	Nate	Nate
SM1	EXPANDED LOCAL MEASUSRED SERVICE														\$ 43.69	\$ 27.00			<u> </u>	
	SIMPLIFIED MESSAGE DESK INTERFACE																		,	
SMGP1	(SMDI) PER LINK, ESSA PER LINK, ANALOG/DIGITAL ESSX SERVICE		\$ 210.00	¢ 175.77	\$ 200.00	¢ 166.20			\$ 300.00	\$ 248.31			\$ 300.00	¢ 227.40	\$ 205.00	¢ 126.60	¢ 200 00	¢ 220.04	\$ 250.00	¢ 190.00
SWIGPT	EXPANDED LOCAL SERVICE THRIFTY		\$ 210.00	\$ 175.77	\$ 200.00	\$ 100.30			\$ 300.00	\$ 240.31			\$ 300.00	\$ 221.40	\$ 205.00	\$ 120.09	\$ 300.00	\$ 230.04	\$ 250.00	\$ 109.00
OWNAOV	CALLER SERVICE, SHARING/RESALE,														¢ 00.05	e 40.00			'	
SWMCX	MEAS SRVC REGION SERVICE-BUSINESS TRUNK WITH														\$ 26.35	\$ 16.28				-
T13CX	OUT DISCT COMBINATION																		\$ 24.50	\$ 16.46
TACC1	TENNESSEE AREA CALLING PLAN ECONOMY LINE, BUSINESS LINE																		\$ 24.50	\$ 16.46
	TENNESSEE AREA CALLING PLL+ STD																		ψ 24.00	ψ 10.40
TACC2	BUSINESS LINE TENNESSEE AREA CALLING PLAN TN																		\$ 24.50	\$ 16.46
	AREA CALLING PLAN BOTHWAY, STD,																		'	
TACCS	BUSINESS TRUNK																		\$ 24.50	\$ 16.46
TAKC1	REGIONSERV BUSINESS INDIV																		\$ 25.50	\$ 17.14
TAKCE	DIST BOTHWAY BUSINESS WITH																		\$ 25.50	\$ 17.14
TDD1X	PBX SERVICE, DID INWARD TRUNK	1	\$ 35.76	\$ 29.93	\$ 33.66	\$ 28.00			\$ 32.00	\$ 26.49			\$ 35.78	\$ 27.13			\$ 38.58	\$ 29.58	\$ 47.34	\$ 35.79
TDD1X	PBX SERVICE, DID INWARD TRUNK	2	\$ 36.23			\$ 29.42	\$ 36.16	\$ 25.42	•	\$ 27.23			\$ 36.95				\$ 41.13	\$ 31.54	\$ 53.90	\$ 40.75
TDD1X	PBX SERVICE, DID INWARD TRUNK	3	\$ 36.23	1		\$ 30.97			\$ 32.90	\$ 27.23			\$ 38.12				\$ 43.68	\$ 33.49	\$ 57.31	\$ 43.33
TDD1X	PBX SERVICE, DID INWARD TRUNK	4	\$ 36.23	1		\$ 32.39			\$ 32.90	\$ 27.23			\$ 39.29				\$ 46.23	\$ 35.45		\$ 48.79
TDD1X	PBX SERVICE, DID INWARD TRUNK	5	\$ 36.23	\$ 30.32	\$ 40.55	\$ 33.73	\$ 47.20	\$ 33.18	\$ 32.90	\$ 22.23			\$ 40.46	\$ 30.68			\$ 48.78	\$ 37.40	\$ 52.20	\$ 35.08
TDD1X	PBX SERVICE, DID INWARD TRUNK	6	\$ 36.23	\$ 25.78	\$ 42.33	\$ 33.45											\$ 51.33	\$ 39.36		
TDD1X	PBX SERVICE, DID INWARD TRUNK	7			\$ 43.78	\$ 36.42	\$ 56.00	\$ 34.73									\$ 53.88	\$ 36.72		
TDD1X	PBX SERVICE, DID INWARD TRUNK	8			\$ 45.22	\$ 37.62														
TDD1X	PBX SERVICE, DID INWARD TRUNK	9			\$ 46.58	\$ 34.87														
TDD1X	PBX SERVICE, DID INWARD TRUNK	10			\$ 47.60	\$ 35.64														
TDD1X	PBX SERVICE, DID INWARD TRUNK	11			\$ 48.62	\$ 36.40														
TDD1X	PBX SERVICE, DID INWARD TRUNK	12			\$ 49.47	\$ 37.04	\$ 73.60	\$ 42.61												
TDDCX	PBX SERVICE, DID Combination	1	\$ 71.52	\$ 59.86	\$ 67.32	\$ 56.00			\$ 67.00	\$ 55.46			\$ 71.56	\$ 54.26			\$ 38.58	\$ 29.58	\$ 94.68	\$ 71.57
TDDCX	PBX SERVICE, DID Combination	2	\$ 72.46	\$ 60.65	\$ 70.72	\$ 58.83	\$ 75.82	\$ 53.30	\$ 68.80	\$ 56.95			\$ 73.90	\$ 56.03			\$ 41.13	\$ 31.54	\$ 107.80	\$ 81.50
TDDCX	PBX SERVICE, DID Combination	3	\$ 72.46	\$ 60.65	\$ 74.46	\$ 61.94			\$ 68.80	\$ 56.95			\$ 76.24	\$ 57.81			\$ 43.68	\$ 33.49	\$ 114.63	\$ 86.66
TDDCX	PBX SERVICE, DID Combination	4	\$ 72.46	\$ 60.65	\$ 77.86	\$ 64.77			\$ 68.80	\$ 56.95			\$ 78.58	\$ 59.58			\$ 46.23	\$ 35.45	\$ 136.68	\$ 97.59
TDDCX	PBX SERVICE, DID Combination	5	\$ 72.46	\$ 60.65	\$ 81.10	\$ 67.47	\$ 97.90	\$ 68.82	\$ 68.15	\$ 46.05			\$ 80.92	\$ 61.36			\$ 48.78	\$ 37.40	\$ 104.40	\$ 70.16
TDDCX	PBX SERVICE, DID Combination	6	\$ 72.46	\$ 51.55	\$ 84.66	\$ 66.91											\$ 51.33	\$ 39.36		
TDDCX	PBX SERVICE, DID Combination	7			\$ 87.56	\$ 72.84	\$ 115.50	\$ 71.64									\$ 53.88	\$ 36.72		
TDDCX	PBX SERVICE, DID Combination	8			\$ 90.44	\$ 75.24														
TDDCX	PBX SERVICE, DID Combination	9			\$ 93.16	\$ 69.75														
TDDCX	PBX SERVICE, DID Combination	10			\$ 95.20	\$ 71.28														
TDDCX	PBX SERVICE, DID Combination	11			\$ 97.24	\$ 72.80														
TDDCX	PBX SERVICE, DID Combination PBX SERVICE, COMBINATION BUS FLAT	12			\$ 98.94	\$ 74.08	\$ 150.70	\$ 87.24											<u> </u>	
TFB	RATE COMBINATION TRUNK	1							\$ 35.00	\$ 28.97			\$ 35.78	\$ 27.13					1 '	
TFB	PBX SERVICE, COMBINATION BUS FLAT RATE COMBINATION TRUNK	2					\$ 39.66	\$ 27.88	\$ 35.90	\$ 29.71			\$ 36.95	\$ 28.02						
TFB	PBX SERVICE, COMBINATION BUS FLAT	3							\$ 35.90	\$ 29.71			\$ 38.12	\$ 28.90						
TFB	RATE COMBINATION BUS FLAT	4							\$ 35.90	\$ 29.71			\$ 39.29	\$ 29.79					1	
TFB	PBX SERVICE, COMBINATION BUS FLAT RATE COMBINATION TRUNK	5					\$ 50.70	\$ 35.64		\$ 23.82			\$ 40.46							
TFB	PBX SERVICE, COMBINATION BUS FLAT RATE COMBINATION TRUNK	7					\$ 59.50													

USOC			AL-Tariff	AL-V&T	FL-Tariff	FL-V&T	GA-Tariff	GA-V&T	KY-Tariff	KY-V&T	LA-Tariff	LA-V&T	MS-Tariff		NC-Tariff	NC-V&T	SC-Tariff	SC-V&T	TN-Tariff	TN-V&T
Code	Description	Rate Group	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	MS-V&T Rate	Rate	Rate	Rate	Rate	Rate	Rate
TFB	PBX SERVICE, COMBINATION BUS FLAT RATE COMBINATION TRUNK	12					\$ 77.10	\$ 44.63												
TFC	PBX SERVICE, COMBINATION BUS FLAT RATE COMBINATION TRUNK PBX SERVICE, COMBINATION BUS FLAT	1	\$ 35.76	\$ 29.93	\$ 33.66	\$ 28.00			\$ 35.00	\$ 28.97			\$ 35.78	\$ 27.13			\$ 38.58	\$ 29.58	\$ 47.34	\$ 35.79
TFC	RATE COMBINATION TRUNK	2	\$ 36.23	\$ 30.32	\$ 35.36	\$ 29.42	\$ 39.66	\$ 27.88	\$ 35.90	\$ 29.71			\$ 36.95	\$ 28.02			\$ 41.13	\$ 31.54	\$ 53.90	\$ 40.75
TFC	PBX SERVICE, COMBINATION BUS FLAT RATE COMBINATION TRUNK	3	\$ 36.23	\$ 30.32	\$ 37.23	\$ 30.97			\$ 35.90	\$ 29.71			\$ 38.12	\$ 28.90			\$ 43.68	\$ 33.49	\$ 57.31	\$ 43.33
TFC	PBX SERVICE, COMBINATION BUS FLAT RATE COMBINATION TRUNK	4	\$ 36.23	\$ 30.32	\$ 38.93	\$ 32.39			\$ 35.90	\$ 29.71			\$ 39.29	\$ 29.79			\$ 46.23	\$ 35.45	\$ 68.34	\$ 48.79
TFC	PBX SERVICE, COMBINATION BUS FLAT RATE COMBINATION TRUNK	5	\$ 36.23	\$ 30.32	\$ 40.55	\$ 33.73	\$ 50.70	\$ 35.64	\$ 35.25	\$ 23.82			\$ 40.46	\$ 30.68			\$ 48.78	\$ 37.40	\$ 52.20	\$ 35.08
	PBX SERVICE, COMBINATION BUS FLAT						Ψ 000	ψ 00.01	ψ 00.20	ψ 20.02			ψ 10.10	ψ 00.00					ψ OLILO	Ψ 00.00
TFC	RATE COMBINATION TRUNK PBX SERVICE, COMBINATION BUS FLAT	6	\$ 36.23	\$ 25.78		\$ 33.45												\$ 39.36		
TFC	RATE COMBINATION TRUNK PBX SERVICE, COMBINATION BUS FLAT	7			\$ 43.78	\$ 36.42	\$ 59.50	\$ 36.90									\$ 53.88	\$ 36.72		
TFC	RATE COMBINATION TRUNK	8			\$ 45.22	\$ 37.62														
TFC	PBX SERVICE, COMBINATION BUS FLAT RATE COMBINATION TRUNK	9			\$ 46.58	\$ 34.87														
TFC	PBX SERVICE, COMBINATION BUS FLAT RATE COMBINATION TRUNK	10			\$ 47.60	\$ 35.64														
TFC	PBX SERVICE, COMBINATION BUS FLAT RATE COMBINATION TRUNK	11			\$ 48.62	\$ 36.40														
TFC	PBX SERVICE, COMBINATION BUS FLAT RATE COMBINATION TRUNK	12			\$ 49.47	\$ 37.04	\$ 77.10	\$ 44.63												
TFN	Business, flat rate inward only trunk	1	\$ 35.76	\$ 29.93	\$ 33.66	\$ 28.00			\$ 32.00	\$ 26.49			\$ 35.78	\$ 27.13			\$ 38.58	\$ 29.58	\$ 47.34	\$ 35.79
TFN	Business, flat rate inward only trunk	2	\$ 36.23	\$ 30.32	\$ 35.36	\$ 29.42	\$ 36.16	\$ 25.42	\$ 32.90	\$ 27.23			\$ 36.95	\$ 28.02			\$ 41.13	\$ 31.54	\$ 53.90	\$ 40.75
TFN	Business, flat rate inward only trunk	3	\$ 36.23	\$ 30.32	\$ 37.23	\$ 30.97			\$ 32.90	\$ 27.23			\$ 38.12	\$ 28.90			\$ 43.68	\$ 33.49	\$ 57.31	\$ 43.33
TFN	Business, flat rate inward only trunk	4	\$ 36.23	\$ 30.32	\$ 38.93	\$ 32.39			\$ 32.90	\$ 27.23			\$ 39.29	\$ 29.79			\$ 46.23	\$ 35.45	\$ 68.34	\$ 48.79
TFN	Business, flat rate inward only trunk	5	\$ 36.23	\$ 30.32	\$ 40.55	\$ 33.73	\$ 47.20	\$ 33.18	\$ 32.90	\$ 22.23			\$ 40.46	\$ 30.68			\$ 48.78	\$ 37.40	\$ 52.20	\$ 35.08
TFN	Business, flat rate inward only trunk	6	\$ 36.23	\$ 25.78	\$ 42.33	\$ 33.45											\$ 51.33	\$ 39.36		
TFN	Business, flat rate inward only trunk	7			\$ 43.78	\$ 36.42	\$ 56.00	\$ 34.73									\$ 53.88	\$ 36.72		
TFN	Business, flat rate inward only trunk	8			\$ 45.22	\$ 37.62														
TFN	Business, flat rate inward only trunk	9			\$ 46.58	\$ 34.87														
TFN	Business, flat rate inward only trunk	10			\$ 47.60	\$ 35.64														
TFN	Business, flat rate inward only trunk	11			\$ 48.62	\$ 36.40														
TFN	Business, flat rate inward only trunk	12			\$ 49.47	\$ 37.04	\$ 73.60	\$ 42.61												
TFU	Business, flat rate inward only trunk	1			\$ 33.66	\$ 28.00			\$ 35.00	\$ 28.97			\$ 35.78	\$ 27.13			\$ 38.58	\$ 29.58		
TFU	Business, flat rate inward only trunk	2			\$ 35.36	\$ 29.42	\$ 39.66	\$ 27.88	\$ 35.90	\$ 29.71			\$ 36.95	\$ 28.02			\$ 41.13	\$ 31.54		
TFU	Business, flat rate inward only trunk	3			\$ 37.23	\$ 30.97			\$ 35.90	\$ 29.71			\$ 38.12	\$ 28.90			\$ 43.68	\$ 33.49		
TFU	Business, flat rate inward only trunk	4			\$ 38.93	\$ 32.39			\$ 35.90	\$ 29.71			\$ 39.29	\$ 29.79			\$ 46.23	\$ 35.45		
TFU	Business, flat rate inward only trunk	5			\$ 40.55	\$ 33.73	\$ 50.70	\$ 35.64	\$ 35.25	\$ 23.82			\$ 40.46	\$ 30.68			\$ 48.78	\$ 37.40		
TFU	Business, flat rate inward only trunk	6			\$ 42.33	\$ 33.45											\$ 51.33	\$ 39.36		
TFU	Business, flat rate inward only trunk	7			\$ 43.78	\$ 36.42	\$ 59.50	\$ 36.90									\$ 53.88	\$ 36.72		
TFU	Business, flat rate inward only trunk	8			\$ 45.22	\$ 37.62														
TFU	Business, flat rate inward only trunk	9			\$ 46.58	\$ 34.87														
TFU	Business, flat rate inward only trunk	10			\$ 47.60	\$ 35.64														
TFU	Business, flat rate inward only trunk	11			\$ 48.62	\$ 36.40	:													
TFU	Business, flat rate inward only trunk	12	m 000:		\$ 49.47	\$ 37.04	\$ 77.10	\$ 44.63	e	0 010	A 00.5-	A 22 7		0 07/-					<b>6</b> 40.0-	0 4100
TKG	PBX Service, Measured Rate, Combination	1	\$ 23.24	\$ 19.46					\$ 26.17	\$ 21.66	\$ 28.68	\$ 22.74	\$ 35.78	\$ 27.13					\$ 18.95	\$ 14.33
TKG	PBX Service, Measured Rate, Combination	2	\$ 23.55	\$ 19.71					\$ 28.52	\$ 23.61	\$ 29.36	\$ 23.28	\$ 36.95	\$ 28.02					\$ 21.55	\$ 16.29
TKG	PBX Service, Measured Rate, Combination	3	\$ 23.55	\$ 19.71					\$ 30.52	\$ 25.26	\$ 30.04	\$ 23.82	\$ 38.12	\$ 28.90					\$ 22.95	\$ 17.35
TKG	PBX Service, Measured Rate, Combination	4	\$ 23.55	\$ 19.71					\$ 32.46	\$ 26.87	\$ 30.72	\$ 24.35	\$ 39.29	\$ 29.79					\$ 27.35	\$ 19.53
TKG	PBX Service, Measured Rate, Combination	5	\$ 23.55	\$ 19.71					\$ 38.17	\$ 25.79	\$ 31.40	\$ 24.89	\$ 40.46	\$ 30.68					\$ 27.80	\$ 18.68
TKG	PBX Service, Measured Rate, Combination	6	\$ 23.55	\$ 16.75							\$ 32.08	\$ 25.43								

USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
TKG	PBX Service, Measured Rate, Combination	7									\$ 32.76	\$ 25.97								
TKG	PBX Service, Measured Rate, Combination	8									\$ 32.87	\$ 26.06								
TKG	PBX Service, Measured Rate, Combination	9									\$ 33.00	\$ 23.55								
TKV	PBX Service, Measured Rate, Outward Only	1							\$ 26.17	\$ 21.66			\$ 35.78	\$ 27.13					\$ 18.95	\$ 14.33
TKV	PBX Service, Measured Rate, Outward Only	2							\$ 28.52	\$ 23.61			\$ 36.95	\$ 28.02					\$ 21.55	\$ 16.29
TKV	PBX Service, Measured Rate, Outward Only	3							\$ 30.52	\$ 25.26			\$ 38.12	\$ 28.90					\$ 22.95	\$ 17.35
TKV	PBX Service, Measured Rate, Outward Only	4							\$ 32.46	\$ 26.87			\$ 39.29	\$ 29.79					\$ 27.35	\$ 19.53
TKV	PBX Service, Measured Rate, Outward Only	5							\$ 38.17	\$ 25.79			\$ 40.46	\$ 30.68					\$ 27.80	\$ 18.68
TM2	Message rate, PBX trunk, both way, additional	1							\$ 28.39	\$ 23.50			\$ 21.24	\$ 16.10						
TM2	Message rate, PBX trunk, both way, additional	2							\$ 18.77	\$ 15.54			\$ 22.11	\$ 16.77						
TM2	Message rate, PBX trunk, both way, additional	3							\$ 18.77	\$ 15.54			\$ 22.99	\$ 17.43						
TM2	Message rate, PBX trunk, both way, additional	4							\$ 18.77	\$ 15.54			\$ 23.87	\$ 18.10						
TM2	Message rate, PBX trunk, both way, additional	5							\$ 18.77	\$ 12.68			\$ 24.75	\$ 18.76						
TM3	Message rate, PBX trunk, outdial, additional	1			\$ 7.63	\$ 6.35			\$ 28.39	\$ 23.50			\$ 21.24	-			\$ 36.87	\$ 28.27		
TM3	Message rate, PBX trunk, outdial, additional	2			\$ 8.01	\$ 6.66			\$ 18.77	\$ 15.54			\$ 22.11	\$ 16.77			\$ 39.24	\$ 30.09		
TM3	Message rate, PBX trunk, outdial, additional	3			\$ 8.42	\$ 7.00			\$ 18.77	\$ 15.54			\$ 22.99	-			\$ 41.80	\$ 32.05		
TM3	Message rate, PBX trunk, outdial, additional	4			\$ 8.79	\$ 7.31			\$ 18.77	\$ 15.54			\$ 23.87	\$ 18.10			\$ 44.27	\$ 33.95		
TM3	Message rate, PBX trunk, outdial, additional	5			\$ 9.15	\$ 7.61			\$ 18.77	\$ 12.68			\$ 24.75	\$ 18.76			\$ 46.73	\$ 35.83		
TM3	Message rate, PBX trunk, outdial, additional	6			\$ 9.54	\$ 7.54			ψ .σ	ψ .2.00			¥ 2o	ψ 10.10			\$ 49.20	\$ 37.73		
TM3	Message rate, PBX trunk, outdial, additional	7			\$ 9.86	\$ 8.20											\$ 51.66	\$ 35.21		
TM3	Message rate, PBX trunk, outdial, additional	8			\$ 10.18	\$ 8.47											Ψ 01.00	Ψ 00.21		
TM3	Message rate, PBX trunk, outdial, additional	9			\$ 10.48	\$ 7.85														
TM3	Message rate, PBX trunk, outdial, additional	10			\$ 10.71	\$ 8.02														
TM3	Message rate, PBX trunk, outdial, additional	11			\$ 10.93	\$ 8.18														
TM3	Message rate, PBX trunk, outdial, additional	12			\$ 11.12	\$ 8.33														
TM5	PBX/CENTREX, TOUCH-TONE C, MESSAGE RATE, PBX TRUNK, COMBINATION, ADDTL	1	\$ 15.74	\$ 13.18		\$ 12.69			\$ 28.39	\$ 23.50			\$ 21.24	\$ 16.10			¢ 26.07	f 20.27	\$ 11.45	\$ 8.66
TIVIS	PBX/CENTREX, TOUCH-TONE C,	<u> </u>	φ 15.74	φ 13.10	φ 15.20	φ 12.09			ş 20.39	φ 23.50			\$ 21.24	\$ 10.10			φ 30.07	φ 20.21	\$ 11.45	φ 0.00
TM5	MESSAGE RATE, PBX TRUNK, COMBINATION, ADDTL	2	\$ 16.05	\$ 13.4	\$ 16.01	\$ 13.32			\$ 18.77	\$ 15.54			\$ 22.11	\$ 16.77			\$ 39.24	\$ 30.09	\$ 14.05	\$ 10.62
	PBX/CENTREX, TOUCH-TONE C, MESSAGE RATE, PBX TRUNK,																			
TM5	COMBINATION, ADDTL	3	\$ 16.05	\$ 13.4	\$ 16.84	\$ 14.01			\$ 18.77	\$ 15.54			\$ 22.99	\$ 17.43			\$ 41.80	\$ 32.05	\$ 15.45	\$ 11.68
	PBX/CENTREX, TOUCH-TONE C,																			
TM5	MESSAGE RATE, PBX TRUNK, COMBINATION, ADDTL PBX/CENTREX, TOUCH-TONE C,	4	\$ 16.05	\$ 13.4	\$ 17.59	\$ 14.63			\$ 18.77	\$ 15.54			\$ 23.87	\$ 18.10			\$ 44.27	\$ 33.95	\$ 19.85	\$ 14.17
TM5	MESSAGE RATE, PBX TRUNK, COMBINATION, ADDTL	5	\$ 16.05	\$ 13.4	3 \$ 18.30	\$ 15.22			\$ 18.77	\$ 12.68			\$ 24.75	\$ 18.76			\$ 46.73	\$ 35.83	\$ 20.30	\$ 13.64
	PBX/CENTREX, TOUCH-TONE C,																			
TM5	MESSAGE RATE, PBX TRUNK, COMBINATION, ADDTL PBX/CENTREX, TOUCH-TONE C,	6	\$ 16.05	\$ 11.4	\$ 19.09	\$ 15.09											\$ 49.20	\$ 37.73		
TM5	MESSAGE RATE, PBX TRUNK, COMBINATION, ADDTL	7			\$ 9.86	\$ 8.20											\$ 51.66	\$ 35.21		
TM5	PBX/CENTREX, TOUCH-TONE C, MESSAGE RATE, PBX TRUNK, COMBINATION, ADDTL	8			\$ 10.18	\$ 8.47														
TM5	PBX/CENTREX, TOUCH-TONE C, MESSAGE RATE, PBX TRUNK, COMBINATION, ADDTL	9				\$ 7.85														
TM5	PBX/CENTREX, TOUCH-TONE C, MESSAGE RATE, PBX TRUNK, COMBINATION, ADDTL	10			\$ 10.71															

Attachment 1A Version 1Q01

USOC			AL-Tariff	AL-V&T	FL-Tariff	FL-V&T	GA-Tariff	GA-V&T	KY-Tariff	KY-V&T	LA-Tariff	LA-V&T	MS-Tariff		NC-Tariff	NC-V&T	SC-Tariff	SC-V&T	TN-Tariff	TN-V&T
Code	Description	Rate Group	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	Rate	MS-V&T Rate	Rate	Rate	Rate	Rate	Rate	Rate
TM6	PBX/CENTREX, TOUCH-TONE C, MESSAGE RATE, PBX TRUNK, COMBINATION, ADDTL	11			\$ 10.93	\$ 8.18														
	PBX/CENTREX, TOUCH-TONE C, MESSAGE RATE, PBX TRUNK,																			
TM7	COMBINATION, ADDTL PBX SERVICE, COMBINATION, MSG RATE,	12			\$ 11.12	\$ 8.33														
ТМВ	BOTH WAY FIRST TRUNK PBX SERVICE, COMBINATION, MSG RATE,	1							\$ 33.39	\$ 27.64	\$ 21.51	\$ 17.05	\$ 26.84	\$ 20.35						
ТМВ	BOTH WAY FIRST TRUNK  PBX SERVICE, COMBINATION, MSG RATE,	2							\$ 23.77	\$ 19.67	\$ 22.02	\$ 17.46	\$ 27.71	\$ 21.01						
ТМВ	BOTH WAY FIRST TRUNK	3							\$ 23.77	\$ 19.67	\$ 22.53	\$ 17.86	\$ 28.59	\$ 21.68						
ТМВ	PBX SERVICE, COMBINATION, MSG RATE, BOTH WAY FIRST TRUNK	4							\$ 23.77	\$ 19.67	\$ 23.04	\$ 18.27	\$ 29.47	\$ 22.34						
ТМВ	PBX SERVICE, COMBINATION, MSG RATE, BOTH WAY FIRST TRUNK PBX SERVICE, COMBINATION, MSG RATE,	5							\$ 23.77	\$ 16.06	\$ 23.55	\$ 18.67	\$ 30.35	\$ 23.01						
ТМВ	BOTH WAY FIRST TRUNK	6									\$ 24.06	\$ 19.07								i
ТМВ	PBX SERVICE, COMBINATION, MSG RATE, BOTH WAY FIRST TRUNK	7									\$ 24.57	\$ 19.48								
ТМВ	PBX SERVICE, COMBINATION, MSG RATE, BOTH WAY FIRST TRUNK	8									\$ 24.65	\$ 19.54								
TMB	BOTH WAY FIRST TRUNK	9									\$ 24.75	\$ 17.66								
тмс	PBX SERVICE, COMBINATION, MSG RATE, FIRST TRUNK	1	\$ 23.24	\$ 19.46	\$ 15.26	\$ 12.69			\$ 33.39	\$ 27.64	\$ 21.51	\$ 17.05	\$ 26.84	\$ 20.35			\$ 36.87	\$ 28.27	\$ 27.05	\$ 20.45
тмс	PBX SERVICE, COMBINATION, MSG RATE, FIRST TRUNK	2	\$ 23.55	\$ 19.71	\$ 16.01	\$ 13.32	\$ 26.46	\$ 18.60	\$ 23.77	\$ 19.67	\$ 22.02	\$ 17.46	\$ 27.71	\$ 21.01			\$ 39.24	\$ 30.09	\$ 30.80	\$ 23.28
TMC	PBX SERVICE, COMBINATION, MSG RATE, FIRST TRUNK	3	\$ 23.55	\$ 19.71	\$ 16.84	\$ 14.01			\$ 23.77	\$ 19.67	\$ 22.53	\$ 17.86	\$ 28.59	\$ 21.68			\$ 41.80	\$ 32.05	\$ 32.75	\$ 24.76
тмс	PBX SERVICE, COMBINATION, MSG RATE, FIRST TRUNK	4	\$ 23.55	\$ 19.71	\$ 17.59	\$ 14.63			\$ 23.77	\$ 19.67	\$ 23.04	\$ 18.27	\$ 29.47	\$ 22.34			\$ 44.27	\$ 33.95	\$ 39.05	\$ 27.88
тмс	PBX SERVICE, COMBINATION, MSG RATE, FIRST TRUNK	5	\$ 23.55	\$ 19.71	\$ 18.30	\$ 15.22	\$ 33.47	\$ 23.53	\$ 23.77	\$ 16.06	\$ 23.55	\$ 18.67	\$ 30.35	\$ 23.01			\$ 46.73	\$ 35.83	\$ 39.70	\$ 26.68
тмс	FIRST TRUNK	6	\$ 23.55	\$ 16.75	\$ 19.09	\$ 15.09					\$ 24.06	\$ 19.07					\$ 49.20	\$ 37.73		
TMC	PBX SERVICE, COMBINATION, MSG RATE, FIRST TRUNK	7	•			\$ 16.41	\$ 39.06	\$ 24.23			\$ 24.57	\$ 19.48					\$ 51.66	\$ 35.21		
TMC	PBX SERVICE, COMBINATION, MSG RATE, FIRST TRUNK	8			\$ 20.36	\$ 16.94					\$ 24.65	\$ 19.54								
тмс	PBX SERVICE, COMBINATION, MSG RATE, FIRST TRUNK	9			\$ 20.96	\$ 15.69					\$ 24.75	\$ 17.66								
TMC	PBX SERVICE, COMBINATION, MSG RATE, FIRST TRUNK	10				\$ 16.03					<b>V</b> 20	Ψ 17.00								
TMC	PBX SERVICE, COMBINATION, MSG RATE, FIRST TRUNK	11			\$ 21.86	\$ 16.37														
тмс	PBX SERVICE, COMBINATION, MSG RATE, FIRST TRUNK	12			\$ 22.24	\$ 16.65	\$ 50.24	\$ 29.08												
TMCGE	PBX SERVICE, Georgia Community Calling, Combination	2			ψ 22.24	Ψ 10.03	\$ 26.46	\$ 18.60												
TMCGE	PBX SERVICE, Georgia Community Calling, Combination	5					\$ 20.40	\$ 23.53												
TMCGE	PBX SERVICE, Georgia Community Calling, Combination	7					\$ 39.06	\$ 24.23												
TMCGE	PBX SERVICE, Georgia Community Calling, Combination	12					\$ 50.24	\$ 29.08												
TMD	PBX SERVICE, inward, first	2					\$ 36.16	\$ 25.42												
TMD	PBX SERVICE, inward, first	5					\$ 47.20	\$ 33.18												
TMD	PBX SERVICE, inward, first	7					\$ 56.00	\$ 34.73												
TMD	PBX SERVICE, inward, first PBX SERVICE, Georgia Community Calling,	12					\$ 73.60	\$ 42.61												
TMDGE	Inward  PBX SERVICE, Georgia Community Calling,	2					\$ 36.16	\$ 25.42												
TMDGE	Inward  PBX SERVICE, Georgia Community Calling,	5					\$ 47.20	\$ 33.18												
TMDGE	Inward	7					\$ 56.00	\$ 34.73												i

Attachment 1A Version 1Q01

			=																	
USOC Code	Description	Rate Group	AL-Tariff Rate	AL-V&T Rate	FL-Tariff Rate	FL-V&T Rate	GA-Tariff Rate	GA-V&T Rate	KY-Tariff Rate	KY-V&T Rate	LA-Tariff Rate	LA-V&T Rate	MS-Tariff Rate	MS-V&T Rate	NC-Tariff Rate	NC-V&T Rate	SC-Tariff Rate	SC-V&T Rate	TN-Tariff Rate	TN-V&T Rate
oouc	PBX SERVICE, Georgia Community Calling,		rtute	rtuto	Ruic	ruic	ruic		rtute	Nute	rtuto	Rute	Nutc	mo var kate	rtuto	ruic	Nuic	ruic	rtuto	ruic
TMDGE	Inward	12					\$ 73.60	\$ 42.61												
TMU	Message rate, PBX trunk, outdial, first	1			\$ 15.26	\$ 12.69			\$ 33.39	\$ 27.64	\$ 21.51	\$ 17.05	\$ 26.84	-			\$ 36.87	\$ 28.27		
TMU	Message rate, PBX trunk, outdial, first	2			\$ 16.01	\$ 13.32	\$ 26.46	\$ 18.60		\$ 19.67	\$ 22.02	\$ 17.46	\$ 27.71	-			\$ 39.24	\$ 30.09		
TMU	Message rate, PBX trunk, outdial, first	3			\$ 16.84	\$ 14.01			\$ 23.77	\$ 19.67	\$ 22.53	\$ 17.86	\$ 28.59	*			\$ 41.80	\$ 32.05		
TMU	Message rate, PBX trunk, outdial, first	5			\$ 17.59	\$ 14.63			\$ 23.77	\$ 19.67	\$ 23.04	\$ 18.27	\$ 29.47	\$ 22.34			\$ 44.27	\$ 33.95		
TMU	Message rate, PBX trunk, outdial, first  Message rate, PBX trunk, outdial, first	6			\$ 18.30	\$ 15.22	\$ 33.47	\$ 23.53	\$ 23.77	\$ 16.06	\$ 23.55	\$ 18.67	\$ 30.35	\$ 23.01			\$ 46.73 \$ 49.20	\$ 35.83		
TMU TMU	Message rate, PBX trunk, outdial, first  Message rate, PBX trunk, outdial, first	7			\$ 19.09	\$ 15.09 \$ 16.41	e 0.00	f 100			\$ 24.06 \$ 24.57	\$ 19.07 \$ 19.48					Ψ 10.20	\$ 37.73 \$ 35.21		
TMU	Message rate, PBX trunk, outdial, first	8			\$ 19.73 \$ 20.36	\$ 16.41	\$ 8.00	\$ 4.96			\$ 24.57 \$ 24.65	\$ 19.48 \$ 19.54					\$ 51.66	\$ 35.21		
TMU	Message rate, PBX trunk, outdial, first	9			\$ 20.96	\$ 15.69					\$ 24.05	1								
TMU	Message rate, PBX trunk, outdial, first	10			\$ 21.41	\$ 16.03					ψ 24.73	Ψ 17.00								
TMU	Message rate, PBX trunk, outdial, first	11			\$ 21.86	\$ 16.37														
TMU	Message rate, PBX trunk, outdial, first	12			\$ 22.24	\$ 16.65	\$ 50.24	\$ 29.08												
	PBX SERVICE, Georgia Community Calling,				Ψ ΣΕΙΕ Ι	ψ 10.00														
TMUGE	Out Dial PBX SERVICE, Georgia Community Calling,	2					\$ 26.46	\$ 18.60												
TMUGE	Out Dial  PBX SERVICE, Georgia Community Calling,	5					\$ 33.47	\$ 23.53												
TMUGE	Out Dial	7					\$ 39.06	\$ 24.23												
TMUGE	PBX SERVICE, Georgia Community Calling, Out Dial	12					\$ 50.24	\$ 29.08												
TTT	LONG DISTANCE TRUNKS CONN+				\$ 19.80	\$ 14.82							\$ 34.61	\$ 26.24			\$ 30.87	\$ 21.04		
TTTXB	LONG DISTANCE TRUNKS		\$ 35.79	\$ 23.96	\$ 19.80	\$ 14.82			\$ 35.00	\$ 23.65	\$ 28.68	\$ 20.46					\$ 30.87	\$ 21.04	\$ 27.05	\$ 18.18
T. 1. 0. 1. 1	COMMUNITY CALLING PLUS SERVICE,	,													A 47.00					
TWC1X	PBX Trunk, Inward Only COMMUNITY CALLING PLUS SERVICE,	1													\$ 47.88	\$ 29.59				
TWC1X	PBX Trunk, Inward Only	2													\$ 49.09	\$ 30.34				
TWC1X	PBX Trunk, Inward Only	3													\$ 50.35	\$ 31.12				
TWC1X	COMMUNITY CALLING PLUS SERVICE, PBX Trunk, Inward Only	4													\$ 51.43	\$ 31.78				
TWOAY	COMMUNITY CALLING PLUS SERVICE, PBX Trunk, Inward Only	5													£ 50.70	¢ 00.50				
TWC1X	COMMUNITY CALLING PLUS SERVICE,	5													\$ 52.73	\$ 32.59				
TWC1X	PBX Trunk, Inward Only	6													\$ 53.75	\$ 33.22				
TWC1X	COMMUNITY CALLING PLUS SERVICE, PBX Trunk, Inward Only	7													\$ 53.75	\$ 33.22				
TWC1X	COMMUNITY CALLING PLUS SERVICE, PBX Trunk, Inward Only	8													\$ 53.75	\$ 33.22				
TWCIX	COMMUNITY CALLING PLUS SERVICE,	0													\$ 55.75	\$ 33.22				
TWC1X	PBX Trunk, Inward Only	9													\$ 54.50	\$ 33.68				
TWC1X	COMMUNITY CALLING PLUS SERVICE, PBX Trunk, Inward Only	10													\$ 54.50	\$ 33.68				
TWCCX	PBX Trunk, Combination	1													\$ 47.88	\$ 29.59				
TWCCX	PBX Trunk, Combination	2													\$ 49.09	\$ 30.34				
TWCCX	PBX Trunk, Combination	3													\$ 50.35	\$ 31.12				
THOON	COMMUNITY CALLING PLUS SERVICE,	,													A =1 10					
TWCCX	PBX Trunk, Combination COMMUNITY CALLING PLUS SERVICE,	4													\$ 51.43	\$ 31.78				
TWCCX	PBX Trunk, Combination	5													\$ 52.73	\$ 32.59				
TWCCX	PBX Trunk, Combination COMMUNITY CALLING PLUS SERVICE,	6													\$ 53.75	\$ 33.22				
TWCCX	PBX Trunk, Combination	7													\$ 53.75	\$ 33.22				
TWCCX	PBX Trunk, Combination	8													\$ 53.75	\$ 33.22				
TWCCX	COMMUNITY CALLING PLUS SERVICE, PBX Trunk, Combination	9													\$ 54.50	\$ 33.68				

Attachment 1A Version 1Q01

Part													
MINISTER   CREATE PROPERTY   1		Description	Rate Group						 MS-V&T Rate				 
March   Marc	TWCCY		10							¢ 5450	¢ 22.60		
No.   Common   Comm	TWCCX		10							φ 54.50	φ 33.00		
NO.   Company	TWCND	PBX Trunk, DID Combination	1							\$ 95.76	\$ 59.18		
NORMAN   N			_										
NECKION   OR K. Trance (DIO Commissioner   1	TWCND		2							\$ 98.18	\$ 60.68		
MORROW   PAST No. DIC Combination   S   S   S   S   S   S   S   S   S	TWCND		3							\$ 100.70	\$ 62.23		
WOOD   No. Trach Colombination			1							*	*		
Michael   Mich	TWCND		4							\$ 102.86	\$ 63.57		
WORDON   COMMON PLUS SERVICE,   0   0   0   0   0   0   0   0   0	TWOND		-							C 405 40	e 05.47		
PRINTER   PRINTER   DIG COMPANION	TWCND		3							\$ 105.46	\$ 65.17		
Pick   Pick   Disconnections   7	TWCND		6							\$ 108.40	\$ 66.99		
TYCHO   PRAT TURK DID Consistence   Program													
PRIX TURK, DID COMPARISON   S   S   70.22   S   S   S   S   S   S   S   S   S	TWCND		7							\$ 110.96	\$ 68.57		
NAME   NAME	TWCND	PBX Trunk, DID Combination	8							\$ 113.62	\$ 70.22		
TWOMEN   PORT TOTAL DID COMMUNITY CALLING PLUS SERVICE.	TVOILD		<del>                                     </del>							Ψ 110.02	Ψ 70.22		
Pick Trusk, DID Confinement   10	TWCND		9							\$ 116.84	\$ 72.21		
TYCKN   POX TAX B. DID   1   1   1   1   1   1   1   1   1			4.0										
PACKED   P	TWCND		10							\$ 120.54	\$ 74.49		
COMMUNITY CALLING PLUS SERVICE,   2	TWCNX		1							\$ 47.88	\$ 29.59		
TYCKN   PATTURE DID SERVICE			+ +							Ψσσ	ψ 20.00		
March   Pax Trank   Did	TWCNX		2							\$ 49.09	\$ 30.34		
TOWNS   PRATTING DID	T11/01/1/		2							• ====			
March   Pax Truck   DID	TWCNX		3							\$ 50.35	\$ 31.12		
COMMUNITY CALINO PLUS SERVICE, TWCKX   PAS Trurk, Old   COMMUNITY CALINO PLUS SERVICE, TWCKX   PAS TRURK   COMMUNITY CALINO PLUS SERVICE, TW	TWCNX		4							\$ 51.43	\$ 31.78		
TOWNING   PAST   TAIL NO PLUS SERVICE,   TOWNING   TOW		The state of the s											
TWCKN,   PBX Trurk, DID	TWCNX		5							\$ 52.73	\$ 32.59		
TWCKN   Past Trunk Outdail   Trunk Outdail	TWCNY		6							¢ 53.75	¢ 33.22		
COMMONITY CALLING PLUS SERVICE,	TVVCIVX		<del>                                     </del>							ψ 33.73	ψ 33.22		
TWONN   PRX Trurk   DID	TWCNX		7							\$ 53.75	\$ 33.22		
COMMONITY CALLING PLUS SERVICE, TOWNOWN PS TURK, DUIdou   9   9   9   9   9   9   9   9   9											_		
TWCNX   PRX Trurk, DID   PRX Trurk, DI	TWCNX		8							\$ 53.75	\$ 33.22		
COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   COMMUNITY CALLING PLUS SERVICE, TWCOX   PX Trurk, Outdial   PX T	TWCNX		9							\$ 54.50	\$ 33.68		
COMMUNITY CALLING PLUS SERVICE,   TWCOX   PBX Trurk, Outdial   TUCX   PBX Trurk, Out		COMMUNITY CALLING PLUS SERVICE,											
TWCOX   PBX Trurk, Outdial	TWCNX		10							\$ 54.50	\$ 33.68		
COMMUNITY CALLING PLUS SERVICE, PBX Trurk, Outdial   2   2   3   3   3   3   3   3   3   3	TWCOX		1							¢ 47.88	\$ 20.50		
COMMUNITY CALLING PLUS SERVICE,	TWCOX		· ·							φ 47.00	φ 29.59		
TWCOX   PBX Trunk, Outdial   3	TWCOX	PBX Trunk, Outdial	2							\$ 49.09	\$ 30.34		
COMMUNITY CALLING PLUS SERVICE,   S   51.43   \$ 31.78   S   51.43   S   S   S   S   S   S   S   S   S										_			
TWCOX   PBX Trunk, Outdial   4   8   8   8   8   8   8   8   8   8	TWCOX		3							\$ 50.35	\$ 31.12		
COMMUNITY CALLING PLUS SERVICE,   Foundation   Foundati	TWCOX		4							\$ 51.43	\$ 31.78		
COMMUNITY CALLING PLUS SERVICE,   Factor   Fac		COMMUNITY CALLING PLUS SERVICE,								+ 010	Ç 00		
TWCOX   PBX Trunk, Outdial   6	TWCOX		5							\$ 52.73	\$ 32.59		
COMMUNITY CALLING PLUS SERVICE,   7	TWCOY		6							¢ 50.75	e 22.00		
TWCOX         PBX Trunk, Outdial         7         8         53.75         \$ 33.22         8         8         53.75         \$ 53.75         \$ 33.22         8	TWCOX		0							φ 53./5	э 33.22		
COMMUNITY CALLING PLUS SERVICE,	TWCOX	PBX Trunk, Outdial	7							\$ 53.75	\$ 33.22		
COMMUNITY CALLING PLUS SERVICE, TWCOX PBX Trunk, Outdial 9 \$ 54.50 \$ 33.68													
TWCOX         PBX Trunk, Outdial         9         S 54.50         \$ 33.68           COMMUNITY CALLING PLUS SERVICE,         S 54.50         \$ 33.68         S 33.68	TWCOX		8							\$ 53.75	\$ 33.22		
COMMUNITY CALLING PLUS SERVICE,	TWCOY		9							\$ 54.50	\$ 33.60		
	1 *****		-							ψ 54.50	ψ 33.08		
TWCOX PBX Trunk, Outdial 10 \$ 54.50 \$ 33.68	TWCOX	PBX Trunk, Outdial	10							\$ 54.50	\$ 33.68		

### EXHIBIT 1

26540ff 145189

## **Attachment 4**

**Physical Collocation** 

Version 3Q00: 11/16/00 EXHIBIT 1

# BELLSOUTH PHYSICAL COLLOCATION

#### 1. Scope of Attachment

1.1 Scope of Attachment. The rates, terms, and conditions contained within this Attachment shall only apply when BTI is occupying the Collocation Space as a sole occupant or as a Host within a Premises location pursuant to Section 4. This Attachment is applicable to Premises owned or leased by BellSouth. However, if the Premises occupied by BellSouth is leased by BellSouth from a third party, special considerations and intervals may apply in addition to the terms and conditions of this Attachment.

All the negotiated rates, terms and conditions set forth in this Attachment pertain to collocation and the provisioning of Collocation Space.

- Right to Occupy. BellSouth shall offer to BTI collocation on rates, terms, and conditions that are just, reasonable, non-discriminatory and consistent with the rules of the Federal Communications Commission ("FCC"). Subject to Section 4 of this Attachment, BellSouth allows BTI to occupy that certain area designated by BellSouth within a BellSouth Premises, or on BellSouth property upon which the BellSouth Premises is located, of a size which is specified by BTI and agreed to by BellSouth (hereinafter "Collocation Space"). BellSouth Premises include BellSouth Central Offices and Serving Wire Centers. The necessary rates, terms and conditions for BellSouth locations other than BellSouth Premises shall be negotiated upon request for collocation at such location(s). Neither BellSouth nor any of BellSouth's affiliates may reserve space for future use on more preferential terms than those set forth above.
- 1.2.1 In all states other than Florida, the size specified by BTI may contemplate a request for space sufficient to accommodate BTI's growth within a two-year period.
- 1.2.2 In the state of Florida, the size specified by BTI may contemplate a request for space sufficient to accommodate BTI's growth within an eighteen (18) month period.
- Space Reclamation. In the event of space exhaust within a Central Office Premises, BellSouth may include in its documentation for the Petition for Waiver filing any unutilized space in the Central Office Premises. BTI will be responsible for any justification of unutilized space within its space, if such justification is required by the appropriate state commission.
- Use of Space. BTI shall use the Collocation Space for the purposes of installing, maintaining and operating BTI's equipment (to include testing and monitoring equipment) necessary for interconnection with BellSouth services and facilities, including access to unbundled network elements, for the provision of telecommunications services. Pursuant to Section 5 following, BTI may, at its option, place BTI-owned fiber entrance facilities to the Collocation Space. The Collocation

Version 3Q00: 11/16/00 EXHIBIT 1

- Space may be used for no other purposes except as specifically described herein or authorized in writing by BellSouth.
- 1.5 Rates and Charges. BTI agrees to pay the rates and charges identified in Exhibit A attached hereto. Rates that are interim are subject to true-up pursuant to section 7.11 of this attachment
- 1.6 Due Dates. In all states other than Georgia, if any due date contained in this Attachment falls on a weekend or National holiday, then the due date will be the next business day thereafter.
- 1.7 The parties agree to comply with all applicable federal, state, county, local and administrative laws, rules, ordinances, regulations and codes in the performance of their obligations hereunder.

#### 2. Space Notification

- 2.1 Availability of Space. Upon submission of an Application pursuant to Section 6, BellSouth will permit BTI to physically collocate, pursuant to the terms of this Attachment, at any BellSouth Premises, unless BellSouth is prepared to demonstrate to the Public Service Commission that there is no space available due to space limitations or no space available due to technical infeasibility. BellSouth shall, upon request from BTI, remove obsolete unused equipment from its premises prior to BellSouth's scheduled removal of such equipment, to increase the amount of space available for collocation. There will be no charge to BTI for removal of obsolete unused equipment from BellSouth's premises.
- 2.1.1 Availability Notification. Unless otherwise specified, BellSouth will respond to an application within ten (10) calendar days as to whether space is available or not available within a BellSouth Premises. This interval excludes National Holidays. If the amount of space requested is not available, BellSouth will notify BTI of the amount of space that is available.
- 2.1.2 BellSouth will respond to a Florida Application within fifteen (15) calendar days as to whether space is available or not available within a BellSouth Premises. If the amount of space requested is not available, BellSouth will notify BTI of the amount of space that is available.
- 2.1.3 BellSouth will respond to a Louisiana Application within ten (10) calendar days for space availability for one (1) to ten (10) Applications; fifteen (15) calendar days for eleven (11) to twenty (20) Applications; and for more than twenty (20) Applications, it is increased by five (5) calendar days for every five additional Applications received within five (5) business days. If the amount of space requested is not available, BellSouth will notify BTI of the amount of space that is available.
- 2.1.4 BellSouth will respond to a Mississippi Application within ten (10) business days as to whether space is available or not available within a BellSouth Premises. If the

- amount of space requested is not available, BellSouth will notify BTI of the amount of space that is available.
- <u>2.2</u> Reporting. Upon request from BTI, BellSouth will provide a written report ("Space Availability Report") specifying the amount of Collocation Space available at the Premises requested, the number of collocators present at the Premises, any modifications in the use of the space since the last report on the Premises requested and the measures BellSouth is taking to make additional space available for collocation arrangements.
- 2.2.1 The request from BTI for a Space Availability Report must be written and must include the Premises and Common Language Location Identification ("CLLI") code of the Premises. Such information regarding Premises and CLLI code is located in the National Exchange Carriers Association (NECA) Tariff FCC No. 4.
- 2.2.2 BellSouth will respond to a request for a Space Availability Report for a particular Premises within ten (10) calendar days of receipt of such request. BellSouth will make best efforts to respond in ten (10) calendar days to such a request when the request includes from two (2) to five (5) Premises within the same state. The response time for requests of more than five (5) Premises shall be negotiated between the Parties. If BellSouth cannot meet the ten calendar day response time, BellSouth shall notify BTI and inform BTI of the time frame under which it can respond.
- 2.2.2.1 In Mississippi, BellSouth will respond to a request for a Space Availability Report for a particular Premises within ten (10) business days of receipt of such request. BellSouth will make best efforts to respond in ten (10) business days to such a request when the request includes from two (2) to five (5) Premises within the same state. The response time for requests of more than five (5) Premises shall be negotiated between the Parties. If BellSouth cannot meet the ten business day response time, BellSouth shall notify BTI and inform BTI of the time frame under which it can respond.
- 2.3 Denial of Application. After notifying BTI that BellSouth has no available space in the requested Premises ("Denial of Application"), BellSouth will allow BTI, upon request, to tour the entire Premises within ten (10) calendar days of such Denial of Application. In order to schedule said tour within ten (10) calendar days, the request for a tour of the Premises must be received by BellSouth within five (5) calendar days of the Denial of Application.
- 2.3.1 Denial of Application. In Mississippi, after notifying BTI that BellSouth has no available space in the requested Premises ("Denial of Application"), BellSouth will allow BTI, upon request, to tour the entire Premises within ten (10) calendar days of such Denial of Application. In order to schedule said tour within ten (10) calendar days, the request for a tour of the Premises must be received by BellSouth within five (5) calander days of the Denial of Application.

- 2.4 Filing of Petition for Waiver. Upon Denial of Application BellSouth will timely file a petition with the Commission pursuant to 47 U.S.C. § 251(c)(6). BellSouth shall provide to the Commission any information requested by that Commission. Such information shall include which space, if any, BellSouth or any of BellSouth's affiliates have reserved for future use and a detailed description of the specific future uses for which the space has been reserved. Subject to an appropriate nondisclosure agreement or provision, BellSouth shall permit BTI to inspect any floor plans or diagrams that BellSouth provides to the Commission.
- 2.5 Waiting List. Unless otherwise specified, on a first-come, first-served basis governed by the date of receipt of an Application or Letter of Intent, BellSouth will maintain a waiting list of requesting carriers who have either received a Denial of Application or, where it is publicly known that the Premises is out of space, have submitted a Letter of Intent to collocate. BellSouth will notify the telecommunications carriers on the waiting list when space becomes available according to how much space becomes available and the position of telecommunications carrier on said waiting list. BTI must submit an updated, complete, and correct Application to BellSouth within 30 calendar days (in Mississippi, 30 business days) of such notification or notify BellSouth in writing within that time that BTI wants to maintain its place on the waiting list either without accepting such space or accepting an amount of space less than its original request. If BTI does not submit such an Application or notify BellSouth in writing as described above, BellSouth will offer such space to the next CLEC on the waiting list and remove BTI from the waiting list. Upon request, BellSouth will advise BTI as to its position on the list.
- 2.5.1 In Florida, on a first-come, first-served basis governed by the date of receipt of an Application or Letter of Intent, BellSouth will maintain a waiting list of requesting carriers who have either received a Denial of Application or, where it is publicly known that the Premises is out of space, have submitted a Letter of Intent to collocate. Sixty (60) days prior to space becoming available, if known, BellSouth will notify the Florida PSC and the telecommunications carriers on the waiting list by mail when space becomes available according to the position of telecommunications carrier on said waiting list. If not known sixty (60) days in advance, BellSouth shall notify the Florida PSC and the telecommunications carriers on the waiting list within two days of the determination that space is available.
- 2.6 Public Notification. BellSouth will maintain on its Interconnection Services website a notification document that will indicate all Central Offices that are without available space. BellSouth shall update such document within ten (10) calendar days (in Mississippi, 10 business days) of the Denial of Application due to Space Exhaust. BellSouth will also post a document on its Interconnection Services website that contains a general notice where space has become available in a Central Office previously on the space exhaust list. BellSouth shall allocate said available space pursuant to the waiting list referenced in Section 2.5.
- 2.7 Regulatory Agency Procedures. Notwithstanding the foregoing, should any state or federal regulatory agency impose procedures or intervals applicable to BTI that are

different from procedures or intervals set forth in this section, whether now in effect or that become effective after execution of this Agreement, those procedures or intervals shall supersede the requirements set forth herein for that jurisdiction for all applications submitted for the first time after the effective date thereof.

#### 3. Collocation Options

- 3.1 Cageless. BellSouth shall allow BTI to collocate BTI's equipment and facilities without requiring the construction of a cage or similar structure. BellSouth shall allow BTI to have direct access to its equipment and facilities. BellSouth shall make cageless collocation available in single bay increments pursuant to Section 7. Except where BTI's equipment requires special technical considerations (e.g., special cable racking, isolated ground plane), BellSouth shall assign cageless Collocation Space in conventional equipment rack lineups where feasible. For equipment requiring special technical considerations, BTI must provide the equipment layout, including spatial dimensions for such equipment pursuant to generic requirements contained in BellCore (Telcordia) GR-63-Core and shall be responsible for constructing all special technical requirements associated with such equipment pursuant to Section 6 following.
- 3.2 Cages. BellSouth shall construct enclosures in compliance with BTI's collocation request. At BTI's request, BellSouth shall permit BTI to subcontract the construction of physical collocation arrangements with a contractor certified by BellSouth ("BellSouth Certified Contractor"), provided however, that BellSouth shall not unreasonably withhold approval of contractors.
- 3.3 When BTI subcontracts the construction, BTI must arrange with a BellSouth Certified Contractor to construct a collocation arrangement enclosure in accordance with BellSouth's guidelines and specifications prior to starting equipment installation and at BTI's sole expense. BellSouth will provide guidelines and specifications upon request. Where local building codes require enclosure specifications more stringent than BellSouth's standard enclosure specification, BTI and BTI's BellSouth Certified Contractor must comply with the more stringent local building code requirements. BTI's BellSouth Certified Contractor shall be responsible for filing and receiving any and all necessary permits and/or licenses for such construction. BellSouth shall cooperate with BTI and provide, at BTI's expense, the documentation, including architectural drawings, necessary for BTI to obtain the zoning, permits and/or other licenses. BellSouth shall pass on to BTI the costs of providing the documentation. The BellSouth Certified Contractor shall bill BTI directly for all work performed for BTI pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the BellSouth Certified Contractor. BTI must provide the local BellSouth building contact with two Access Keys used to enter the locked enclosure. Except in case of emergency, BellSouth will not access BTI's locked enclosure prior to notifying BTI.

- 3.3.1 BellSouth may elect to review BTI's plans and specifications prior to allowing construction to start to ensure compliance with BellSouth's guidelines and specifications. BellSouth shall complete its review within fifteen (15) calendar days. BTI shall be able to design caged enclosures in amounts as small as sufficient to house and maintain a single rack or bay of equipment. If BellSouth reviews BTI's plans and specifications prior to construction, then BellSouth will have the right to inspect the enclosure after construction to make sure it is constructed according to the submitted plans and specifications. If BellSouth elects not to review BTI's plans and specifications prior to construction, BTI will be entitled to request BellSouth to review; and in the event BTI does not request a BellSouth review, BellSouth shall have the right to inspect the enclosure after construction to make sure it is constructed according to BellSouth's guidelines and specifications. BellSouth may require BTI to remove or correct within seven (7) calendar days at BTI's expense any structure that does not meet these plans and specifications or, where applicable, BellSouth guidelines and specifications.
- 3.4 Shared (Subleased) Caged Collocation. BTI may allow other telecommunications carriers to share BTI's caged collocation arrangement pursuant to terms and conditions agreed to by BTI ("Host") and other telecommunications carriers ("Guests") and pursuant to this section, except where the BellSouth Premises is located within a leased space and BellSouth is prohibited by said lease from offering such an option. BTI shall notify BellSouth in writing upon execution of any agreement between the Host and its Guest within ten (10) calendar days (in Mississippi, ten (10) business days) of its execution and prior to any Firm Order. Further, such notice shall include the name of the Guest(s) and the term of the agreement, and shall contain a certification by BTI that said agreement imposes upon the Guest(s) the same terms and conditions for Collocation Space as set forth in this Attachment between BellSouth and BTI.
- 3.4.1 BTI, as the host CLEC shall be the sole interface and responsible Party to BellSouth for the assessment and billing of rates and charges contained within this Attachment; and for the purposes of ensuring that the safety and security requirements of this Attachment are fully complied with by the Guest, its employees and agents. BellSouth shall prorate the costs of the collocation space based on the number of collocators and the space used by each. In all states other than Florida, and in addition to the foregoing, BTI shall be the responsible party to BellSouth for the purpose of submitting Applications for initial and additional equipment placement of Guest. In the event the Host and Guest jointly submit an initial Application, only one Application Fee will be assessed. A separate initial Guest application shall require the assessment of a Subsequent Application Fee, as set forth in Exhibit A, if this Application is not the initial Application made for the arrangement. Notwithstanding the foregoing, Guest may arrange directly with BellSouth for the provision of the interconnecting facilities between BellSouth and Guest and for the provision of the services and access to unbundled network elements.
- 3.4.2 BTI shall indemnify and hold harmless BellSouth from any and all claims, actions, causes of action, of whatever kind or nature arising out of the presence of BTI's

Guests in the Collocation Space except to the extent caused by BellSouth's sole negligence, gross negligence, or willful misconduct.

- 3.5 Adjacent Collocation. BellSouth will permit adjacent collocation arrangements ("Adjacent Arrangement") on the Premises' property where physical collocation space within the Premises is legitimately exhausted, subject to technical feasibility, where the Adjacent Arrangement does not interfere with access to existing or planned structures or facilities on the Premises property and where permitted by zoning and other applicable state and local regulations. The Adjacent Arrangement shall be constructed or procured by BTI and in conformance with BellSouth's design and construction specifications. Further, BTI shall construct, procure, maintain and operate said Adjacent Arrangement(s) pursuant to all of the terms and conditions set forth in this Attachment. Rates shall be negotiated at the time of the request for the Adjacent Arrangement.
- 3.5.1 Should BTI elect such option, BTI must arrange with a BellSouth Certified Contractor to construct an Adjacent Arrangement structure in accordance with BellSouth's guidelines and specifications. BellSouth will provide guidelines and specifications upon request. Where local building codes require enclosure specifications more stringent than BellSouth's standard specification, BTI and BTI's BellSouth Certified Contractor must comply with the more stringent local building code requirements. BTI's BellSouth Certified Contractor shall be responsible for filing and receiving any and all necessary zoning, permits and/or licenses for such construction. BTI's BellSouth Certified Contractor shall bill BTI directly for all work performed for BTI pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the BellSouth Certified Contractor. BTI must provide the local BellSouth building contact with two cards, keys or other access device used to enter the locked enclosure. Except in cases of emergency, BellSouth shall not access BTI's locked enclosure prior to notifying BTI.
- 3.5.2 BTI must submit its plans and specifications to BellSouth with its Firm Order. BellSouth may elect to review BTI's plans and specifications prior to construction of an Adjacent Arrangement(s) to ensure compliance with BellSouth's guidelines and specifications. BellSouth shall complete its review within fifteen (15) calendar days. If BellSouth reviews BTI's plans and specifications prior to construction, then BellSouth will have the right to inspect the Adjacent Arrangement after construction to make sure it is constructed according to the submitted plans and specifications. If BellSouth elects not to review BTI's plans and specifications prior to construction, BTI will be entitled to request BellSouth to review; and in the event BTI does not request a BellSouth review, BellSouth shall have the right to inspect the Adjacent Arrangement after construction to make sure it is constructed according to BellSouth's guidelines and specifications. BellSouth may require BTI to remove or correct within seven (7) calendar days at BTI's expense any structure that does not meet these plans and specifications or, where applicable, BellSouth's guidelines and specifications

- 3.5.3 BTI shall provide a concrete pad, the structure housing the arrangement, heating/ventilation/air conditioning ("HVAC"), lighting, and all facilities that connect the structure (i.e. racking, conduits, etc.) to the BellSouth point of demarcation. At BTI's option, and where the local authority having jurisdiction permits, BellSouth shall provide an AC power source and access to physical collocation services and facilities subject to the same nondiscriminatory requirements as applicable to any other physical collocation arrangement. BTI's BellSouth Certified Contractor shall be responsible for filing and receiving any and all necessary zoning, permits and/or licenses for such arrangement.
- 3.5.4 BellSouth shall allow Shared (Subleased) Caged Collocation within an Adjacent Arrangement pursuant to the terms and conditions set forth in Section 3.4 preceding.
- 3.6 BTI may seek a particular physical collocation arrangement, and is entitled to a presumption that such arrangement is technically feasible, if any local exchange company has deployed such collocation arrangement in any incumbent local exchange carrier premises.

#### 4. Occupancy

- 4.1 <u>Commencement Date</u>. The "Commencement Date" shall be the day BTI's equipment becomes operational as described in Article 4.2, following.
- 4.2 Occupancy. BellSouth will notify BTI in writing that the Collocation Space is ready for occupancy. BTI must notify BellSouth in writing that collocation equipment installation is complete and is operational with BellSouth's network. BellSouth may, at its option, not accept orders for interconnected service until receipt of such notice. For purposes of this paragraph, BTI's telecommunications equipment will be deemed operational when cross-connected to BellSouth's network for the purpose of service provision.
- 4.3 <u>Termination of Occupancy</u>. In addition to any other provisions addressing Termination of Occupancy in this Attachment, Termination of Occupancy may occur in the following circumstances:
- 4.3.1 BTI may terminate occupancy in a particular Collocation Space by submitting a Subsequent Application requesting termination of occupancy.
- 4.3.2 Upon termination of such occupancy, BTI at its expense shall remove its equipment and other property from the Collocation Space. BTI shall have thirty (30) calendar days from the termination date to complete such removal, including the removal of all equipment and facilities of BTI's Guests, unless BTI's guest has assumed responsibility for the collocation space housing the guest equipment and executed the documentation required by BellSouth prior to such removal date. BTI shall continue payment of monthly fees to BellSouth until such date as BTI has fully vacated the Collocation Space. Should BTI or BTI's Guest fail to vacate the Collocation Space within thirty (30) calendar days from the termination date, BellSouth shall have the right to remove the equipment and other property of BTI or BTI's Guest at BTI's

expense and with no liability for damage or injury to BTI or BTI's Guest's property unless caused by the gross negligence or intentional misconduct of BellSouth. Upon termination of BTI's right to occupy Collocation Space, BTI shall surrender such Collocation Space to BellSouth in the same condition as when first occupied by BTI except for ordinary wear and tear, unless otherwise agreed to by the Parties. BTI shall be responsible for the cost of removing any enclosure, together with all support structures (e.g., racking, conduits), at the termination of occupancy and restoring the grounds to their original condition.

#### 5. Use of Collocation Space

- 5.1 Equipment Type. BellSouth permits the collocation of any type of equipment necessary for interconnection to BellSouth's network or for access to unbundled network elements in the provision of telecommunications services. Such equipment includes, but is not limited to transmission equipment including, but not limited to, optical terminating equipment and multiplexers, and digital subscriber line access multiplexers, routers, asyncronous transfer mode multiplexers, and remote switching modules. Nothing in this section requires BellSouth to permit collocation of equipment used solely to provide enhanced services.
- Such equipment must at a minimum meet the following BellCore (Telcordia)
  Network Equipment Building Systems (NEBS) General Equipment Requirements:
  Criteria Level 1 requirements as outlined in the BellCore (Telcordia) Special Report SR-3580, Issue 1; equipment design spatial requirements per GR-63-CORE, Section 2; thermal heat dissipation per GR-063-CORE, Section 4, Criteria 77-79; acoustic noise per GR-063-CORE, Section 4, Criterion 128, and National Electric Code standards. Except where otherwise required by a Commission, BellSouth shall comply with the applicable FCC rules relating to denial of collocation based on BTI's failure to comply with this section.
- BTI shall not request more DS0, DS1, DS3 and optical terminations for a collocation arrangement than the total port or termination capacity of the transmission equipment physically installed in the arrangement. The total capacity of the transmission equipment collocated in the arrangement will include equipment contained in the application in question as well as equipment already placed in the arrangement. Collocated cross-connect devices are not considered transmission equipment. If full network termination capacity of the transmission equipment being installed is not requested in the application, additional network terminations for the installed equipment will require the submission of another application. In the event that BTI submits an application for terminations that exceed the total capacity of the collocated equipment, BTI will be informed of the discrepancy and will be required to submit a revision to the application.
- 5.1.3 BTI shall not use the Collocation Space for marketing purposes nor shall it place any identifying signs or markings in the area surrounding the Collocation Space or on the grounds of the Premises.

- 5.1.4 BTI shall place a plaque or other identification affixed to BTI's equipment necessary to identify BTI's equipment, including a list of emergency contacts with telephone numbers.
- 5.2 Entrance Facilities. BTI may elect to place BTI-owned or BTI-leased fiber entrance facilities into the Collocation Space. BellSouth will designate the point of interconnection in close proximity to the Premises building housing the Collocation Space, such as an entrance manhole or a cable vault which are physically accessible by both Parties. BTI will provide and place fiber cable at the point of entrance of sufficient length to be pulled through conduit and into the splice location. BTI will provide and install a sufficient length of fire retardant riser cable, to which the entrance cable will be spliced, which will extend from the splice location to BTI's equipment in the Collocation Space. In the event BTI utilizes a non-metallic, risertype entrance facility, a splice will not be required. BTI must contact BellSouth for instructions prior to placing the entrance facility cable in the manhole. BTI is responsible for maintenance of the entrance facilities. At BTI's option BellSouth will accommodate where technically feasible a microwave entrance facility pursuant to separately negotiated terms and conditions. In the case of adjacent collocation, unless BellSouth determines that limited space is available for the entrance facilities, copper facilities may be used between the adjacent collocation arrangement and the central office termination point.
- Dual Entrance. BellSouth will provide at least two interconnection points at each Premises where there are at least two such interconnection points available and where capacity exists. Upon receipt of a request for physical collocation under this Attachment, BellSouth shall provide BTI with information regarding BellSouth's capacity to accommodate dual entrance facilities. If conduit in the serving manhole(s) is available and is not reserved for another purpose for utilization within 12 months of the receipt of an application for collocation, BellSouth will make the requested conduit space available for installing a second entrance facility to BTI's arrangement. The location of the serving manhole(s) will be determined at the sole discretion of BellSouth. Where dual entrance is not available due to lack of capacity, BellSouth will so state in the Application Response.
- 5.2.2 <u>Shared Use</u>. BTI may utilize spare capacity on an existing interconnector entrance facility for the purpose of providing an entrance facility to another BTI collocation arrangement within the same BellSouth Premises. BTI must arrange with BellSouth for BellSouth to splice the utilized entrance facility capacity to BTI-provided riser cable.
- 5.2.3 Splicing in the Entrance Manhole. Although not generally permitted, should BTI request a splice to occur in the entrance manhole(s), BellSouth, at its sole discretion, may grant such a request, provided that BellSouth will not unreasonably withhold approval of requests to make such a splice. When the request for a splice is granted to BTI by BellSouth, BTI shall ensure its employees or agents entering and/or performing work in the entrance manhole(s) are trained and comply with BellSouth procedures which shall be no less than industry standards and OSHA requirements

regarding access to manholes and that BellSouth personnel are notified and present for all entrances and work performed in the entrance manhole(s). Manhole covers shall be properly closed and secured at the conclusion of entry and/or work. Advance notification to BellSouth shall occur at a minimum of 48 hours prior to desired entry for normal work activities and at a minimum of 2 hours prior to desired entry in an out of service condition.

- 5.3 Demarcation Point. BellSouth will designate the point(s) of demarcation between BTI's equipment and/or network and BellSouth's network. Each Party will be responsible for maintenance and operation of all equipment/facilities on its side of the demarcation point. For 2-wire and 4-wire connections to BellSouth's network, the demarcation point shall be a common block on the BellSouth designated conventional distributing frame. BTI shall be responsible for providing, and a supplier certified by BellSouth ("BTI's BellSouth Certified Supplier") shall be responsible for installing and properly labeling/stenciling, the common block, and necessary cabling pursuant to Section 6.5. For all other terminations BellSouth shall designate a demarcation point on a per arrangement basis. Where it designates the demarcation point on a per arrangement basis, BellSouth will make best efforts to designate the demarcation point closest to BTI's collocated equipment. BTI or its agent must perform all required maintenance to equipment/facilities on its side of the demarcation point, pursuant to Section 5.4, following, and may self-provision cross-connects that may be required within the Collocation Space to activate service requests. At BTI's option and expense, a Point of Termination ("POT") bay or frame may be placed in the Collocation Space. BTI must make arrangements with a BellSouth Certified Supplier for such placement.
- BTI's Equipment and Facilities. BTI, or if required by this Attachment, BTI's BellSouth Certified Supplier, is solely responsible for the design, engineering, installation, testing, provisioning, performance, monitoring, maintenance and repair of the equipment and facilities used by BTI which must be performed in compliance with all applicable BellSouth policies and guidelines. Such equipment and facilities may include but are not limited to cable(s); equipment; and point of termination connections. BTI and its selected BellSouth Certified Supplier must follow and comply with all BellSouth requirements outlined in BellSouth's TR 73503, TR 73519, TR 73572, and TR 73564, copies of which BellSouth will supply upon request.
- 5.5 BellSouth's Access to Collocation Space. From time to time BellSouth may require access to the Collocation Space. BellSouth retains the right to access such space for the purpose of making BellSouth equipment and building modifications (e.g., running, altering or removing racking, ducts, electrical wiring, HVAC, and cables). BellSouth will give notice to BTI at least three (3) calendar days before access to the Collocation Space is required. BTI may elect to be present whenever BellSouth performs work in the Collocation Space. The Parties agree that BTI will not bear any of the expense associated with this work.

- 5.6 Access. Pursuant to Section 11, BTI shall have access to the Collocation Space twenty-four (24) hours a day, seven (7) days a week. BTI agrees to provide the name and social security number or date of birth or driver's license number of each employee, contractor, or agents of BTI or BTI's Guests provided with access keys or devices ("Access Keys") prior to the issuance of said Access Keys. Key acknowledgement forms must be signed by BTI and returned to BellSouth Access Management within 15 calendar days of BTI's receipt. Failure to return properly acknowledged forms will result in the holding of subsequent requests until acknowledgements are current. Access Keys shall not be duplicated under any circumstances. BTI agrees to be responsible for all Access Keys and for the return of all said Access Keys in the possession of BTI employees, contractors, Guests, or agents after termination of the employment relationship, contractual obligation with BTI or upon the termination of this Attachment or the termination of occupancy of an individual collocation arrangement.
- 5.6.1 <u>Lost or Stolen Access Keys</u>. BTI shall notify BellSouth in writing within 24 hours of becoming aware in the case of lost or stolen Access Keys. Should it become necessary for BellSouth to re-key buildings or deactivate a card as a result of a lost Access Key(s) or for failure to return an Access Key(s), BTI shall pay for all reasonable costs associated with the re-keying or deactivating the card.
- 5.7 Interference or Impairment. Notwithstanding any other provisions of this Attachment, BTI shall not use any product or service provided under this Agreement, any other service related thereto or used in combination therewith, or place or use any equipment or facilities in the Collocation Space in any manner that 1) significantly degrades, interferes with or impairs service provided by BellSouth or by any other entity or any person's use of its telecommunications service; 2) endangers or damages the equipment, facilities or other property of BellSouth or of any other entity or person; 3) compromises the privacy of any communications; or 4) creates an unreasonable risk of injury or death to any individual or to the public. If BellSouth reasonably determines that any equipment or facilities of BTI violates the provisions of this paragraph, BellSouth shall give written notice to BTI, which notice shall direct BTI to cure the violation within forty-eight (48) hours of BTI's actual receipt of written notice or, at a minimum, to commence curative measures within 24 hours and to exercise reasonable diligence to complete such measures as soon as possible thereafter. After receipt of the notice, the Parties agree to consult immediately and, if necessary, to inspect the arrangement.
- 5.7.1 Except in the case of the deployment of an advanced service which significantly degrades the performance of other advanced services or traditional voice band services, if BTI fails to take curative action within 48 hours or if the violation is of a character which poses an immediate and substantial threat of damage to property, injury or death to any person, or any other significant degradation, interference or impairment of BellSouth's or another entity's service, then and only in that event BellSouth may take such action as it deems appropriate to correct the violation, including without limitation the interruption of electrical power to BTI's equipment. BellSouth will endeavor, but is not required, to provide notice to BTI prior to taking

- such action and shall have no liability to BTI for any damages arising from such action, except to the extent that such action by BellSouth constitutes willful misconduct or gross negligence.
- 5.7.2 For purposes of this Section 5.7, the term significantly degrade shall mean an action that noticeably impairs a service from a user's perspective. In the case of the deployment of an advanced service which significantly degrades the performance of other advanced services or traditional voice band services and BTI fails to take curative action within 48 hours then BellSouth will establish before the relevant Commission that the technology deployment is causing the significant degradation. Any claims of network harm presented to BTI or, if subsequently necessary, the relevant Commission, must be supported with specific and verifiable information. Where BellSouth demonstrates that a deployed technology is significantly degrading the performance of other advanced services or traditional voice band services, BTI shall discontinue deployment of that technology and migrate its customers to technologies that will not significantly degrade the performance of other such services. Where the only degraded service itself is a known disturber, and the newly deployed technology satisfies at least one of the criteria for a presumption that is acceptable for deployment under section 47 C.F.R. 51.230, the degraded service shall not prevail against the newly-deployed technology.
- <u>5.8</u> Personalty and its Removal. Facilities and equipment placed by BTI in the Collocation Space shall not become a part of the Collocation Space, even if nailed, screwed or otherwise fastened to the Collocation Space, but shall retain their status as personal property and may be removed by BTI at any time. Any damage caused to the Collocation Space by BTI's employees, agents or representatives during the removal of such property shall be promptly repaired by BTI at its expense.
- Alterations. In no case shall BTI or any person acting on behalf of BTI make any rearrangement, modification, improvement, addition, repair, or other alteration which could affect in any way space, power, HVAC, and/or safety considerations to the Collocation Space or the BellSouth Premises without the written consent of BellSouth, which consent shall not be unreasonably withheld. The cost of any such specialized alterations shall be paid by BTI. Any material rearrangement, modification, improvement, addition, repair, or other alteration shall require a Subsequent Application and Subsequent Application Fee, pursuant to sub-section 6.2.2.
- 5.10 Janitorial Service. BTI shall be responsible for the general upkeep of the Collocation Space. BTI shall arrange directly with a BellSouth Certified Contractor for janitorial services applicable to Caged Collocation Space. BellSouth shall provide a list of such contractors on a site-specific basis upon request.

#### 6. Ordering and Preparation of Collocation Space

- Should any state or federal regulatory agency impose procedures or intervals applicable to BTI that are different from procedures or intervals set forth in this section, whether now in effect or that become effective after execution of this Agreement, those procedures or intervals shall supersede the requirements set forth herein for that jurisdiction for all applications submitted for the first time after the effective date thereof.
- 6.2 <u>Application for Space</u>. BTI shall submit an application document when BTI or BTI's Guest(s), as defined in Section 3.4, desires to request or modify the use of the Collocation Space.
- 6.2.1 <u>Initial Application</u>. For BTI or BTI's Guest(s) initial equipment placement, BTI shall submit to BellSouth a Physical Expanded Interconnection Application Document ("Application"). The Application is Bona Fide when it is complete and accurate, meaning that all required fields on the application are completed with the appropriate type of information. The Bona Fide Application shall contain a detailed description and schematic drawing of the equipment to be placed in BTI's Collocation Space(s) and an estimate of the amount of square footage required. BellSouth will act responsibly in it's determination of the completeness and accuracy of the application document as described in this section.
- 6.2.2 Subsequent Application. In the event BTI or BTI's Guest(s) desires to modify the use of the Collocation Space ("Augmentation"), BTI shall complete an Application detailing all information regarding the modification to the Collocation Space ("Subsequent Application"). The minimum Subsequent Application Fee shall be considered a partial payment of the applicable Subsequent Application Fee which shall be calculated as set forth below. BellSouth shall determine what modifications, if any, to the Premises are required to accommodate the change requested by BTI in the Application. Such necessary modifications to the Premises may include but are not limited to, floor loading changes, changes necessary to meet HVAC requirements, changes to power plant requirements, equipment additions, etc. The fee paid by BTI for its request to modify the use of the Collocation Space shall be dependent upon the level of assessment needed for the modification requested. Where the Subsequent Application does not require assessment for provisioning or construction work by BellSouth, no Subsequent Application Fee will be required. The fee for an Application where the modification requested has limited effect (e.g., does not require assessment related to capital expenditure by BellSouth) shall be the Subsequent Application Fee as set forth in Exhibit A. If the modification requires capital expenditure assessment, a full Application Fee shall apply. In the event such modifications require the assessment of a full Application Fee as set forth in Exhibit A, the outstanding balance shall be due by BTI within 30 calendar days following BTI's receipt of a bill or invoice from BellSouth. The Subsequent Application is Bona Fide when it is complete and accurate, meaning that all required fields on the Application are completed with the appropriate type of information.

Version 3Q00: 11/16/00 EXHIBIT 1

- Application Response. In Alabama, Kentucky, North Carolina, and Tennessee, when space has been determined to be available, BellSouth will provide a written response ("Application Response") within twenty-three (23) business days of the receipt of a Bona Fide Application, which will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7.
- 6.3.1 Application Response. Except as otherwise provided, for all States that have ordered provisioning intervals but not application response intervals, the following will apply. In addition to the notice of space availability pursuant to Section 2.1, BellSouth will respond as to whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. When space has been determined to be available, BellSouth will provide a written response ("Application Response") within thirty (30) calendar days of receipt of a Bona Fide Application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7. When multiple applications are submitted in a state within a fifteen (15) calendar day window, BellSouth will respond to the Bona Fide Applications as soon as possible, but no later than the following: within thirty (30) calendar days for Bona Fide Applications 1-5; within thirty-six (36) calendar days for Bona Fide Applications 6-10; within forty-two (42) calendar days for Bona Fide Applications 11-15. Response intervals for multiple Bona Fide Applications submitted within the same timeframe for the same state in excess of 15 must be negotiated. All negotiations shall consider the total volume from all requests from telecommunications companies for collocation.
- Application Response (Florida). Within fifteen (15) calendar days of receipt of a Bona Fide Application, BellSouth will respond as to whether space is available or not available within a particular Premises. Additionally, when space has been determined to be available or when a lesser amount of space than that requested is available, then with respect to the space available, BellSouth will provide a written response ("Application Response") including sufficient information to enable BTI to place a Firm Order. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7. When BTI submits ten (10) or more Applications within ten (10) calendar days, the initial fifteen (15) day response period will increase by ten (10) days for every additional ten (10) Applications or fraction thereof.
- Application Response (Georgia) In addition to the notice of space availability pursuant to Section 2.1, BellSouth will respond as to whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. When space has been determined to be available for caged or cageless arrangements, BellSouth will provide a written response ("Application Response") within thirty (30) calendar days of receipt of a Bona Fide Application. The Application Response will include, at a minimum, the configuration of the

space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7.

- Application Response (Louisiana). In addition to the notice of space availability pursuant to Section 2.1, BellSouth will respond as to whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. BellSouth will respond as to whether space is available or not available within a particular Premises in accordance with Section 2. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7. BellSouth will respond with a full Application Response within 30 calendar days for one to ten Applications; 35 calendar days for eleven to twenty Applications; and for requests of more than twenty Application it is increased by five calendar days for every five Applications received within five business days.
- 6.3.5 Application Response (Mississippi). In addition to the notice of space availability pursuant to Section 2. BellSouth will respond as to whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. When space has been determined to be available, BellSouth will provide a written response ("Application Response") within thirty (30) business days of receipt of a Bona Fide Application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7. When multiple applications are submitted in a state within a fifteen (15) business day window, BellSouth will respond to the Bona Fide Applications as soon as possible, but no later than the following: within thirty (30) business days for Bona Fide Applications 1-5; within thirty-six (36) business days for Bona Fide Applications 6-10; within forty-two (42) business days for Bona Fide Applications 11-15. Response intervals for multiple Bona Fide Applications submitted within the same timeframe for the same state in excess of 15 must be negotiated. All negotiations shall consider the total volume from all requests from telecommunications companies for collocation.
- Application Modifications. In Alabama, Kentucky, North Carolina, and Tennessee, if a modification or revision is made to any information in the Bona Fide Application prior to Bona Fide Firm Order, with the exception of modifications to Customer Information, Contact Information or Billing Contact Information, either at the request of BTI or necessitated by technical considerations, said Application shall be considered a new Application and shall be handled as a new Application for purposes of the provisioning interval and BellSouth may charge BTI an application fee. Where the Application Modification does not require assessment for provisioning or construction work by BellSouth, no application fee will be required. The fee for an Application Modification where the modification requested has limited effect (e.g., requires limited assessment and no capital expenditure by BellSouth) shall be the Subsequent Application Fee as set forth in Exhibit A. Major changes such as requesting additional space or adding equipment may require BTI to submit the Application with an Application Fee.

- Application Modifications. For all States that have ordered provisioning intervals but not application response intervals, and except as otherwise specified, the following will apply: If a modification or revision is made to any information in the Bona Fide Application for Physical Collocation or the Bona Fide Application for Adjacent Collocation, with the exception of modifications to Customer Information, Contact Information or Billing Contact Information, either at the request of BTI or necessitated by technical considerations, BellSouth will respond to the Bona Fide Application within thirty (30) calendar days after BellSouth receives such revised application or at such other date as the Parties agree. If, at any time, BellSouth needs to reevaluate BTI's Bona Fide Application as a result of changes requested by BTI to BTI's original application, then BellSouth will charge BTI a Subsequent Application Fee. Major changes such as requesting additional space or adding additional equipment may require BTI to resubmit the Application with an Application Fee.
- Application Modifications (Florida). If a modification or revision is made to any information in the Bona Fide Application for Physical Collocation or the Bona Fide Application for Adjacent Collocation, with the exception of modifications to Customer Information, Contact Information or Billing Contact Information, either at the request of BTI or necessitated by technical considerations, BellSouth will respond to the Bona Fide Application within fifteen (15) calendar days after BellSouth receives such revised Application or at such other date as the Parties agree. If, at any time, BellSouth needs to reevaluate BTI's Bona Fide Application as a result of changes requested by BTI to BTI's original Application, then BellSouth will charge BTI a Subsequent Application Fee. Major changes such as requesting additional space or adding additional equipment may require BTI to resubmit the Application with an Application Fee.
- Application Modifications (Georgia). If a modification or revision is made to any information in the Bona Fide Application for Physical Collocation or the Bona Fide Application for Adjacent Collocation, with the exception of modifications to Customer Information, Contact Information or Billing Contact Information, either at the request of BTI or necessitated by technical considerations, BellSouth will respond to the Bona Fide Application within thirty (30) calendar days after BellSouth receives such revised Application or at such other date as the Parties agree. If, at any time, BellSouth needs to reevaluate BTI's Bona Fide Application as a result of changes requested by BTI to BTI's original Application, then BellSouth will charge BTI a Subsequent Application Fee. Major changes such as requesting additional space or adding additional equipment may require BTI to resubmit the Application with an Application Fee.
- 6.4.4 <u>Application Modifications (Mississippi)</u>. If a modification or revision is made to any information in the Bona Fide Application for Physical Collocation or the Bona Fide Application for Adjacent Collocation, with the exception of modifications to Customer Information, Contact Information or Billing Contact Information, either at the request of BTI or necessitated by technical considerations, BellSouth will respond to the Bona Fide Application within thirty (30) business days after BellSouth receives such revised Application or at such other date as the Parties agree. If, at any time,

BellSouth needs to reevaluate BTI's Bona Fide Application as a result of changes requested by BTI to BTI's original Application, then BellSouth will charge BTI a Subsequent Application Fee. Major changes such as requesting additional space or adding additional equipment may require BTI to resubmit the Application with an Application Fee.

- 6.4.5 Application Modifications (Louisiana). If a modification or revision is made to any information in the Bona Fide Application for Physical Collocation or the Bona Fide Application for Adjacent Collocation, with the exception of modifications to Customer Information, Contact Information or Billing Contact Information, either at the request of BTI or necessitated by technical considerations, BellSouth will respond to the Bona Fide Application within thirty (30) calendar days after BellSouth receives such revised application or at such other date as the Parties agree. BellSouth will respond to such modifications or revisions within 30 calendar days for one to ten revised Applications; 35 calendar days for eleven to twenty revised Applications; and for requests of more than twenty revised Applications it is increased by five calendar days for every five revised Applications received within five business days. If, at any time, BellSouth needs to reevaluate BTI's Bona Fide Application as a result of changes requested by BTI to BTI's original application, then BellSouth will charge BTI a Subsequent Application Fee. Major changes such as requesting additional space or adding additional equipment may require BTI to resubmit the application with an Application Fee.
- 8.5 Bona Fide Firm Order. In Alabama, Kentucky, North Carolina, and Tennessee, BTI shall indicate its intent to proceed with equipment installation in a Company Premises by submitting a Physical Expanded Interconnection Firm Order document ("Firm Order") to the Company. A Firm Order shall be considered Bona Fide when BTI has completed the Application/Inquiry process described in Section 6.2, preceding, and has submitted the Firm Order document indicating acceptance of the Application Response provided by the Company. The Bona Fide Firm Order must be received by the Company no later than five (5) business days after the Company's Application Response to BTI's Bona Fide Application.
- Bona Fide Firm Order. Except as otherwise provided, in all States that have ordered provisioning intervals but not addressed Firm Order intervals, the following shall apply\_BTI shall indicate its intent to proceed with equipment installation in a BellSouth Premises by submitting a Physical Expanded Interconnection Firm Order document ("Firm Order") to BellSouth. A Firm Order shall be considered Bona Fide when BTI has completed the Application/Inquiry process described in Section 6.2, preceding and has submitted the Firm Order document indicating acceptance of the Application Response provided by BellSouth. The Bona Fide Firm Order must be received by BellSouth no later than thirty (30) calendar days after BellSouth's Application Response to BTI's Bona Fide Application or the Application will expire.
- 6.5.2 <u>Bona Fide Firm Order (Mississippi).</u> BTI shall indicate its intent to proceed with equipment installation in a BellSouth Premises by submitting a Physical Expanded Interconnection Firm Order document ("Firm Order") to BellSouth. A Firm Order

shall be considered Bona Fide when BTI has completed the Application/Inquiry process described in Section 6.2, preceding and has submitted the Firm Order document indicating acceptance of the Application Response provided by BellSouth. The Bona Fide Firm Order must be received by BellSouth no later than thirty (30) business days after BellSouth's Application Response to BTI's Bona Fide Application or the Application will expire.

- 6.5.3 BellSouth will establish a firm order date based upon the date BellSouth is in receipt of a Bona Fide Firm Order. BellSouth will acknowledge the receipt of BTI's Bona Fide Firm Order within seven (7) calendar days of receipt indicating that the Bona Fide Firm Order has been received. A BellSouth response to a Bona Fide Firm Order will include a Firm Order Confirmation containing the firm order date. No revisions will be made to a Bona Fide Firm Order.
- 6.5.4 BellSouth will permit one accompanied site visit to BTI's designated collocation arrangement location after receipt of the Bona Fide Firm Order without charge to BTI.
- 6.5.5 BTI must submit to BellSouth the completed Access Control Request Form for all employees or agents requiring access to the BellSouth Premises a minimum of 30 calendar days prior to the date BTI desires access to the Collocation Space. BTI may submit such a request at any time subsequent to BellSouth's receipt of the Bona Fide Firm Order. In the event BTI desires access to the Collocation Space after submitting such a request but prior to access being approved, BellSouth shall permit BTI to access the Collocation Space, accompanied by a security escort at BTI's expense. BTI must request escorted access at least three (3) business days prior to the date such access is desired.

6.6

6.6.1 Construction and Provisioning Interval. In Kentucky, Tennessee and North Carolina, the Company will complete construction for collocation arrangements within seventysix (76) business days from receipt of an Application or as agreed to by the Parties. Under extraordinary conditions, the Company will complete construction for collocation arrangements within ninety-one (91) business days. Examples of extraordinary conditions include, but are not limited to, extended license or permitting intervals; major Company equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. In the event BTI submits a forecast as described in the following section three (3) months or more prior to the application date, the above intervals shall apply. In the event BTI submits such a forecast between two (2) months and three (3) months prior to the application date, the above intervals may be extended by one (1) additional month. In the event BTI submits such a forecast less than two (2) months prior to the application date, the above intervals may be extended by sixty (60) calendar days. The Company will attempt to meet standard intervals for unforecasted requests and

Version 3Q00: 11/16/00 EXHIBIT 1

any interval adjustments will be discussed with BTI at the time the application is received. Raw space, which is space lacking the necessary infrastructure to provide collocation space including but not limited to HVAC, Power, etc.), conversion time frames fall outside the normal intervals and are negotiated on an individual case basis. Additionally, installations to existing collocation arrangements for line sharing or line splitting, which include adding cable, adding cable and splitter, and adding a splitter, will be forty five (45) business days from receipt of an Application.

- 6.6.1.2 To be considered a timely and accurate forecast, BTI must submit to the Company the CLEC Forecast Form, as set forth in Exhibit C attached hereto, containing the following information: Central Office/Serving Wire Center CLLI, number of Caged square feet and/or Cageless bays, number of DS0, DS1, DS3 frame terminations, number of fused amps and planned application date.
- In Alabama, BellSouth will complete construction for cageless collocation arrangements under ordinary conditions as soon as possible and within a maximum of sixty (60) calendar days from receipt of a Bona Fide Firm Order and ninety (90) calendar days for extraordinary conditions or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Extraordinary conditions are defined to include but are not limited to major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Commission.
- 6.6.2.1 In Alabama, for caged collocation arrangements, the Company will complete construction within seventy-six (76) business days from receipt of an Application or as agreed to by the Parties. Under extraordinary conditions, the Company will complete construction for collocation arrangements within ninety-one (91) business days. Examples of extraordinary conditions include, but are not limited to, extended license or permitting intervals; major Company equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. In the event BTI submits a forecast as described in the following section three (3) months or more prior to the application date, the above intervals shall apply. In the event BTI submits such a forecast between two (2) months and three (3) months prior to the application date, the above intervals may be extended by one (1) additional month. In the event BTI submits such a forecast less than two (2) months prior to the application date, the above intervals may be extended by sixty (60) calendar days. The Company will attempt to meet standard intervals for unforecasted requests and any interval adjustments will be discussed with BTI at the time the application is received. Raw space, which is space lacking the necessary infrastructure to provide collocation space including but not limited to HVAC, Power,

- etc.), conversion time frames fall outside the normal intervals and are negotiated on an individual case basis. Additionally, installations to existing collocation arrangements for line sharing or line splitting, which include adding cable, adding cable and splitter, and adding a splitter, will be forty five (45) business days from receipt of an Application.
- 6.6.2.2 To be considered a timely and accurate forecast, BTI must submit to the Company the CLEC Forecast Form, as set forth in Exhibit C attached hereto, containing the following information: Central Office/Serving Wire Center CLLI, number of Caged square feet, number of DS0, DS1, DS3 frame terminations, number of fused amps and planned application date.
- 6.6.3 Construction and Provisioning Interval (Florida). BellSouth will complete construction for collocation arrangements as soon as possible and within a maximum of 90 calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. For changes to collocation space after initial space completion, BellSouth will complete construction for collocation arrangements as soon as possible and within a maximum of 45 calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. If BellSouth does not believe that construction will be completed within the relevant time frame and BellSouth and BTI cannot agree upon a completion date, within 45 calendar days of receipt of the Bona Fide Firm Order for an initial request, and within 30 calendar days for Augmentations, BellSouth may seek an extension from the Florida PSC.
- 6.6.4 Construction and Provisioning Interval (Georgia). BellSouth will use best efforts to complete construction for caged collocation arrangements under ordinary conditions as soon as possible and within a maximum of 90 calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. BellSouth will use best efforts to complete construction for cageless collocation arrangements under ordinary conditions as soon as possible and within a maximum of 60 calendar days from receipt of a Bona Fide Firm Order and 90 calendar days for extraordinary conditions or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Extraordinary conditions are defined to include but are not limited to major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. BellSouth may elect to renegotiate an alternative provisioning interval with BTI or seek a waiver from this interval from the Commission.
- 6.6.5 Construction and Provisioning Interval (Louisiana). BellSouth will complete construction for collocation arrangements under ordinary conditions as soon as possible and within a maximum of 120 calendar days from receipt of a Bona Fide Firm Order for an initial request, and within 60 calendar days for an Augmentation, or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC,

cabling and the power plant(s). BellSouth will complete construction of all other Collocation Space ("extraordinary conditions") within 180 calendar days of the receipt of a Bona Fide Firm Order. Examples of extraordinary conditions include but are not limited to, extended license or permitting intervals; major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. BellSouth may elect to renegotiate an alternative provisioning interval with BTI or seek a waiver from this interval from the Commission.

- 6.6.6 Construction and Provisioning Interval (Mississippi). Excluding the time interval required to secure the appropriate government licenses and permits, BellSouth will complete construction for collocation arrangements under ordinary conditions as soon as possible and within a maximum of 120 calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Excluding the time interval required to secure the appropriate government licenses and permits, BellSouth will complete construction of all other Collocation Space ("extraordinary conditions") within 180 calendar days of the receipt of a Bona Fide Firm Order. Examples of extraordinary conditions include but are not limited to, extended license or permitting intervals; major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. BellSouth may elect to renegotiate an alternative provisioning interval with BTI or seek a waiver from this interval from the Commission.
- 6.6.7 <u>Construction and Provisioning Interval (South Carolina)</u>. BellSouth will complete the construction and provisioning activities for cageless and caged collocation arrangements as soon as possible, but no later than 90 calendar days from receipt of a bona fide firm order. BellSouth may elect to renegotiate an alternative provisioning interval with BTI or seek a waiver from this interval from the Commission.
- Joint Planning Meeting. Unless otherwise agreed to by the Parties, a joint planning meeting or other method of joint planning between BellSouth and BTI will commence within a maximum of 15 business days from BellSouth's receipt of a Bona Fide Firm Order and the payment of agreed upon fees. At such meeting, the Parties will agree to the preliminary design of the Collocation Space and the equipment configuration requirements as reflected in the Application and affirmed in the Bona Fide Firm Order. The Collocation Space Completion time period will be provided to BTI during the joint planning meeting or as soon as possible thereafter. If "extraordinary conditions" will delay completion beyond the seventy-six (76) business day construction and provisioning interval provided in Section 6.6.1, BellSouth will so advise BTI at or before the joint planning meeting. If "extraordinary conditions" that will delay completion beyond the seventy-six (76)

business day construction and provisioning interval provided in Section 6.6.1 are not reasonably determined by the time of the joint planning meeting, BellSouth will provide BTI notice as soon as reasonably possible upon discovering a condition that would negatively impact the interval. provided that such "extraordinary conditions" will, consistent with Section 6.6.1, not delay completion beyond ninety-one (91) business days. BellSouth will complete all design work following the joint planning meeting.

- 6.8 <u>Permits</u>. Each Party or its agents will diligently pursue filing for the permits required for the scope of work to be performed by that Party or its agents within ten (10) calendar days of the completion of finalized construction designs and specifications.
- 6.9 <u>Acceptance Walk Through</u>. BTI and BellSouth will complete an acceptance walk through of each Collocation Space requested from BellSouth by BTI. BellSouth will correct any deviations to BTI's original or jointly amended requirements within five (5) business days after the walk through, unless the Parties jointly agree upon a different time frame.
- 6.10 Use of BellSouth Certified Supplier. BTI shall select a supplier which has been approved as a BellSouth Certified Supplier to perform all engineering and installation work. BTI and BTI's BellSouth Certified Supplier must follow and comply with all BellSouth requirements outlined in BellSouth's TR 73503, TR 73519, TR 73572, and TR 73564. In some cases, BTI must select separate BellSouth Certified Suppliers for transmission equipment, switching equipment and power equipment. BellSouth shall provide BTI with a list of BellSouth Certified Suppliers upon request. The BellSouth Certified Supplier(s) shall be responsible for installing BTI's equipment and components, extending power cabling to the BellSouth power distribution frame, performing operational tests after installation is complete, and notifying BellSouth's equipment engineers and BTI upon successful completion of installation, etc. The BellSouth Certified Supplier shall bill BTI directly for all work performed for BTI pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the BellSouth Certified Supplier. BellSouth shall consider certifying BTI or any supplier proposed by BTI. All work performed by or for BTI shall conform to generally accepted industry guidelines and standards.
- Alarm and Monitoring. BellSouth shall place environmental alarms in the Premises for the protection of BellSouth equipment and facilities. BTI shall be responsible for placement, monitoring and removal of environmental and equipment alarms used to service BTI's Collocation Space. Upon request, BellSouth will provide BTI with applicable tariffed service(s) to facilitate remote monitoring of collocated equipment by BTI. Both Parties shall use best efforts to notify the other of any verified environmental hazard known to that Party.

- 6.12 <u>Basic Telephone Service</u>. Upon request of BTI, BellSouth will provide basic telephone service to the Collocation Space under the rates, terms and conditions of the current tariff offering for the service requested.
- 6.13 Virtual to Physical Collocation Relocation. In the event physical Collocation Space was previously denied at a location due to technical reasons or space limitations, and that physical Collocation Space has subsequently become available, BTI may relocate its virtual collocation arrangements to physical collocation arrangements and pay the appropriate non-recurring fees for physical collocation and for the rearrangement or reconfiguration of services terminated in the virtual collocation arrangement, as outlined in the appropriate BellSouth tariffs. In the event that BellSouth knows when additional space for physical collocation may become available at the location requested by BTI, such information will be provided to BTI in BellSouth's written denial of physical collocation. To the extent that (i) physical Collocation Space becomes available to BTI within 180 calendar days of BellSouth's written denial of BTI's request for physical collocation, (ii) BellSouth had knowledge that the space was going to become available, and (iii) BTI was not informed in the written denial that physical Collocation Space would become available within such 180 calendar days, then BTI may transition its virtual collocation arrangement to a physical collocation arrangement and will receive a credit for any nonrecurring charges previously paid for such virtual collocation. BTI must arrange with a BellSouth Certified Supplier for the relocation of equipment from its virtual Collocation Space to its physical Collocation Space and will bear the cost of such relocation.
- 6.14 <u>Cancellation</u>. If, at anytime prior to space acceptance, BTI cancels its order for the Collocation Space(s), BTI will reimburse BellSouth in the following manner: BellSouth will ascertain how far preparation work has progressed. BTI will be billed the applicable non recurring rate for any and all work processes for which work has begun.
- 6.15 <u>Licenses.</u> BTI, at its own expense, will be solely responsible for obtaining from governmental authorities, and any other appropriate agency, entity, or person, all rights, privileges, and licenses necessary or required to operate as a provider of telecommunications services to the public or to occupy the Collocation Space.
- 6.16 <u>Environmental Compliance.</u> The Parties agree to utilize and adhere to the Environmental Hazard Guidelines identified as Exhibit B attached hereto.

#### 7. Rates and Charges

- 7.1 BellSouth shall assess an Application Fee via a service order which shall be issued at the time BellSouth responds that space is available pursuant to section 2.1. Payment of said Application Fee will be due as dictated by BTI's current billing cycle and is non-refundable.
- 7.2 <u>Space Preparation.</u> Space preparation fees consist of a nonrecurring charge for Firm Order Processing and monthly recurring charges for Central Office Modifications,

assessed per arrangement, per square foot, and Common Systems Modifications, assessed per arrangement, per square foot for cageless and per cage for caged collocation. BTI shall remit payment of the nonrecurring Firm Order Processing Fee coincident with submission of a Bona Fide Firm Order. The recurring charges for space preparation apply beginning on the date on which BellSouth releases the Collocation Space for occupancy or on the date BTI first occupies the Collocation Space, whichever is sooner. The charges recover the costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, building and support systems. In the event BTI opts for cageless space, the space preparation fees will be assessed based on the total floor space dedicated to BTI as prescribed in Section 7.7.

- Space Preparation Fee in Florida. Space preparation fees include a nonrecurring charge for Firm Order Processing and monthly recurring charges for Central Office Modifications, assessed per arrangement, per square foot, and Common Systems Modifications, assessed per arrangement, per square foot for cageless and per cage for caged collocation. BTI shall remit payment of the nonrecurring Firm Order Processing Fee coincident with submission of a Bona Fide Firm Order. The recurring charges for space preparation apply beginning on the date on which BellSouth releases the Collocation Space for occupancy or on the date BTI first occupies the Collocation Space, whichever is sooner. The charges recover the costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, building and support systems. In the event BTI opts for cageless space, space preparation fees will be assessed based on the total floor space dedicated to BTI as prescribed in Section 7.7
- 5.4 Space Preparation Fee in Georgia. In Georgia, the Space Preparation Fee is a one-time fee, assessed per arrangement, per location. It recovers a portion of costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, power, building and support systems. This is a set fee of \$100 per square foot as established by the Georgia Public Service Commission Order in Docket No. 7061-U. In the event BTI opts for non-enclosed space, the space preparation fee will be assessed based on the total floor space dedicated to BTI as prescribed in Section 7.7.
- 7.5 Space Preparation Fee in North Carolina. In North Carolina, space preparation fees consist of monthly recurring charges for Central Office Modifications, assessed per arrangement, per square foot; Common Systems Modifications, assessed per arrangement, per square foot for cageless and per cage for caged collocation; and Power, assessed per the nominal –48V DC ampere requirements specified by BTI on the Bona Fide Application. The space preparation charges apply beginning on the date on which BellSouth releases the Collocation Space for occupancy or on the date BTI first occupies the Collocation Space, whichever is sooner. The charges recover the costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, building and support systems. In the event BTI opts for cageless space, the space

preparation fees will be assessed based on the total floor space dedicated to BTI as described in Section 7.7.

- 7.6 <u>Cable Installation</u>. Cable Installation Fee(s) are assessed per entrance cable placed.
- 7.7 Floor Space. The Floor Space Charge includes reasonable charges for lighting, HVAC, and other allocated expenses associated with maintenance of the Premises but does not recover any power-related costs incurred by BellSouth. When the Collocation Space is enclosed, BTI shall pay floor space charges based upon the number of square feet so enclosed. When the Collocation Space is not enclosed, BTI shall pay floor space charges based upon the following floor space calculation: [(depth of the equipment lineup in which the rack is placed) + (0.5 x maintenance aisle depth) + (0.5 x wiring aisle depth)] X (width of rack and spacers). For purposes of this calculation, the depth of the equipment lineup shall consider the footprint of equipment racks plus any equipment overhang. BellSouth will assign unenclosed Collocation Space in conventional equipment rack lineups where feasible. In the event BTI's collocated equipment requires special cable racking, isolated grounding or other treatment which prevents placement within conventional equipment rack lineups, BTI shall be required to request an amount of floor space sufficient to accommodate the total equipment arrangement. Floor space charges are due beginning with the date on which BellSouth releases the Collocation Space for occupancy or on the date BTI first occupies the Collocation Space, whichever is sooner.
- 7.8 <u>Power</u>. BellSouth shall make available –48 Volt (-48V) DC power for BTI's Collocation Space at a BellSouth Power Board or BellSouth Battery Distribution Fuse Bay ("BDFB") at BTI's option within the Premises.
- 7.8.1 Recurring charges for -48V DC power will be assessed per ampere per month based upon the BellSouth Certified Supplier engineered and installed power feed fused ampere capacity. Rates include redundant feeder fuse positions (A&B) and common cable rack to BTI's equipment or space enclosure. When obtaining power from a BDFB, fuses and power cables (A&B) must be engineered (sized), and installed by BTI's BellSouth Certified Supplier. When obtaining power from a BellSouth power board, power cables (A&B) must be engineered (sized), and installed by BTI's BellSouth Certified power Supplier. BTI is responsible for contracting with a BellSouth Certified Supplier for power distribution feeder cable runs from a BellSouth BDFB or power board to BTI's equipment. Determination of the BellSouth BDFB or BellSouth power board as the power source will be made at BellSouth's sole, but reasonable, discretion. The BellSouth Certified Supplier contracted by BTI must provide BellSouth a copy of the engineering power specification prior to the day on which BTI's equipment becomes operational ("Commencement Date"). BellSouth will provide the common power feeder cable support structure between the BellSouth BDFB or power board and BTI's arrangement area. BTI shall contract with a BellSouth Certified Supplier who will be responsible for the following: dedicated power cable support structure within BTI's arrangement; power cable feeds; terminations of cable. Any terminations at a

BellSouth power board must be performed by a BellSouth Certified power Supplier. BTI shall comply with all applicable National Electric Code (NEC), BellSouth TR73503, Telcordia (BellCore) and ANSI Standards regarding power cabling.

- 7.8.2 If BellSouth has not previously invested in power plant capacity for collocation at a specific site, BTI has the option to add its own dedicated power plant; provided, however, that such work shall be performed by a BellSouth Certified Supplier who shall comply with BellSouth's guidelines and specifications. Where the addition of BTI's dedicated power plant results in construction of a new power plant room, upon termination of BTI's right to occupy collocation space at such site, BTI shall have the right to remove its equipment from the power plant room, but shall otherwise leave the room intact.
- 7.8.3 If BTI elects to install its own DC Power Plant, BellSouth shall provide AC power to feed BTI's DC Power Plant. Charges for AC power will be assessed per breaker ampere per month. Rates include the provision of commercial and standby AC power. When obtaining power from a BellSouth service panel, protection devices and power cables must be engineered (sized), and installed by BTI's BellSouth Certified Supplier except that BellSouth shall engineer and install protection devices and power cables for Adjacent Collocation. BTI's BellSouth Certified Supplier must also provide a copy of the engineering power specification prior to the Commencement Date. Charges for AC power shall be assessed pursuant to the rates specified in Exhibit A. AC power voltage and phase ratings shall be determined on a per location basis. At BTI's option, BTI may arrange for AC power in an Adjacent Collocation arrangement from a retail provider of electrical power.
- 7.9 Security Escort. A security escort will be required whenever BTI or its approved agent desires access to the entrance manhole or must have access to the Premises after the one accompanied site visit allowed pursuant to Section 6.6.2 prior to completing BellSouth's Security Training requirements and/or prior to Space Acceptance. Rates for a security escort are assessed according to the schedule appended hereto as Exhibit A beginning with the scheduled escort time. BellSouth will wait for one-half (1/2) hour after the scheduled time for such an escort and BTI shall pay for such half-hour charges in the event BTI fails to show up.
- 7.10 Cable Record charges. These charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.
- 7.11 Rate "True-Up". The Parties agree that the prices reflected as interim herein shall be "trued-up" (up or down) based on final prices either determined by further agreement or by an effective order, in a proceeding involving BellSouth before the regulatory authority for the state in which the services are being performed or any other body having jurisdiction over this Agreement (hereinafter "Commission"). Under the "true-up" process, the interim price for each service shall be multiplied by the volume of that service purchased to arrive at the total interim amount paid for that service ("Total Interim Price"). The final price for that service shall be multiplied by the

volume purchased to arrive at the total final amount due ("Total Final Price"). The Total Interim Price shall be compared with the Total Final Price. If the Total Final Price is more than the Total Interim Price, BTI shall pay the difference to BellSouth. If the Total Final Price is less than the Total Interim Price, BellSouth shall pay the difference to BTI. Each Party shall keep its own records upon which a "true-up" can be based and any final payment from one Party to the other shall be in an amount agreed upon by the Parties based on such records. In the event of any disagreement as between the records or the Parties regarding the amount of such "true-up," the Parties agree that the Commission shall be called upon to resolve such differences.

7.12 Other. If no rate is identified in the contract, the rate for the specific service or function will be negotiated by the Parties upon request by either Party. Payment of all other charges under this Attachment shall be due as dictated by BTI's current billing cycle. BTI will pay a late payment charge as specified in the current State Tariff.

#### 8. Insurance

- 8.1 BTI shall, at its sole cost and expense, procure, maintain, and keep in force insurance as specified in this Section 8 and underwritten by insurance companies licensed to do business in the states applicable under this Attachment and having a Best's Insurance Rating of A-.
- 8.2 BTI shall maintain the following specific coverage:
- 8.2.1 Commercial General Liability coverage in the amount of ten million dollars (\$10,000,000.00) or a combination of Commercial General Liability and Excess/Umbrella coverage totaling not less than ten million dollars (\$10,000,000.00). BellSouth shall be named as an Additional Insured on the Commercial General Liability policy as specified herein.
- 8.2.2 Statutory Workers Compensation coverage and Employers Liability coverage in the amount of one hundred thousand dollars (\$100,000.00) each accident, one hundred thousand dollars (\$100,000.00) each employee by disease, and five hundred thousand dollars (\$500,000.00) policy limit by disease.
- 8.2.3 All Risk Property coverage on a full replacement cost basis insuring all of BTI's real and personal property situated on or within BellSouth's Central Office location(s).
- 8.2.4 BTI may elect to purchase business interruption and contingent business interruption insurance, having been advised that BellSouth assumes no liability for loss of profit or revenues should an interruption of service occur.
- 8.3 The limits set forth in Section 8.2 above may be increased by BellSouth from time to time during the term of this Attachment upon thirty (30) days notice to BTI to at least such minimum limits as shall then be customary with respect to comparable occupancy of BellSouth structures.

Version 3Q00: 11/16/00 EXHIBIT 1

- 8.4 All policies purchased by BTI shall be deemed to be primary and not contributing to or in excess of any similar coverage purchased by BellSouth. All insurance must be in effect on or before the date equipment is delivered to BellSouth's Premises and shall remain in effect for the term of this Attachment or until all BTI's property has been removed from BellSouth's Premises, whichever period is longer. If BTI fails to maintain required coverage, BellSouth may pay the premiums thereon and seek reimbursement of same from BTI.
- 8.5 BTI shall submit certificates of insurance reflecting the coverage required pursuant to this Section a minimum of ten (10) business days prior to the commencement of any work in the Collocation Space. Failure to meet this interval may result in construction and equipment installation delays. BTI shall arrange for BellSouth to receive thirty (30) business days' advance notice of cancellation from BTI's insurance company. BTI shall forward a certificate of insurance and notice of cancellation/non-renewal to BellSouth at the following address:

BellSouth Telecommunications, Inc. Attn.: Risk Management Coordinator 17H53 BellSouth Center 675 W. Peachtree Street Atlanta, Georgia 30375

- 8.6 BTI must conform to recommendations made by BellSouth's fire insurance company to the extent BellSouth has agreed to, or shall hereafter agree to, such recommendations.
- 8.7 <u>Self-Insurance</u>. If BTI's net worth exceeds five hundred million dollars (\$500,000,000), BTI may elect to request self-insurance status in lieu of obtaining any of the insurance required in Sections 8.2.1 and 8.2.2. BTI shall provide audited financial statements to BellSouth thirty (30) days prior to the commencement of any work in the Collocation Space. BellSouth shall then review such audited financial statements and respond in writing to BTI in the event that self-insurance status is not granted to BTI. If BellSouth approves BTI for self-insurance, BTI shall annually furnish to BellSouth, and keep current, evidence of such net worth that is attested to by one of BTI's corporate officers. The ability to self-insure shall continue so long as the BTI meets all of the requirements of this Section. If the BTI subsequently no longer satisfies this Section, BTI is required to purchase insurance as indicated by Sections 8.2.1 and 8.2.2.
- 8.8 The net worth requirements set forth in Section 8.7 may be increased by BellSouth from time to time during the term of this Attachment upon thirty (30) days' notice to BTI to at least such minimum limits as shall then be customary with respect to comparable occupancy of BellSouth structures.
- 8.9 Failure to comply with the provisions of this Section will be deemed a material breach of this Attachment.

#### 9. Mechanics Liens

9.1 If any mechanics lien or other liens shall be filed against property of either Party (BellSouth or BTI), or any improvement thereon by reason of or arising out of any labor or materials furnished or alleged to have been furnished or to be furnished to or for the other Party or by reason of any changes, or additions to said property made at the request or under the direction of the other Party, the other Party directing or requesting those changes shall, within thirty (30) business days after receipt of written notice from the Party against whose property said lien has been filed, either pay such lien or cause the same to be bonded off the affected property in the manner provided by law. The Party causing said lien to be placed against the property of the other shall also defend, at its sole cost and expense, on behalf of the other, any action, suit or proceeding which may be brought for the enforcement of such liens and shall pay any damage and discharge any judgment entered thereon.

#### 10. Inspections

10.1 BellSouth may conduct an inspection of BTI's equipment and facilities in the Collocation Space(s) prior to the activation of facilities between BTI's equipment and equipment of BellSouth. BellSouth may conduct an inspection if BTI adds equipment and may otherwise conduct routine inspections at reasonable intervals mutually agreed upon by the Parties. BellSouth shall provide BTI with a minimum of forty-eight (48) hours or two (2) business days, whichever is greater, advance notice of all such inspections. All costs of such inspection shall be borne by BellSouth.

### 11. Security and Safety Requirements

- 11.1 The security and safety requirements set forth in this section are as stringent as the security requirements BellSouth maintains at its own premises either for their own employees or for authorized contractors. Only BellSouth employees, BellSouth Certified Contractors and authorized employees, authorized Guests, pursuant to Section 3.4, preceding, or authorized agents of BTI will be permitted in the BellSouth Premises. BTI shall provide its employees and agents with picture identification which must be worn and visible at all times while in the Collocation Space or other areas in or around the Premises. The photo Identification card shall bear, at a minimum, the employee's name and photo, and the BTI name. BellSouth reserves the right to remove from its premises any employee of BTI not possessing identification issued by BTI or who have violated any of BellSouth's policies as outlined in the CLEC Security Training documents. BTI shall hold BellSouth harmless for any damages resulting from such removal of its personnel from BellSouth premises. BTI shall be solely responsible for ensuring that any Guest of BTI is in compliance with all subsections of this Section 11.
- 11.1.1 BTI will be required, at its own expense, to conduct a statewide investigation of criminal history records for each BTI employee being considered for work on the BellSouth Premises, for the states/counties where the BTI employee has worked and

lived for the past five years. Where state law does not permit statewide collection or reporting, an investigation of the applicable counties is acceptable. BTI shall not be required to perform this investigation if an affiliated company of BTI has performed an investigation of the BTI employee seeking access, if such investigation meets the criteria set forth above. This requirement will not apply if BTI has performed a preemployment statewide investigation of criminal history records, or where state law does not permit an investigation of the applicable counties for the BTI employee seeking access, for the states/counties where the BTI employee has worked and lived for the past five years.

- 11.1.2 BTI will be required to administer to their personnel assigned to the BellSouth Premises security training either provided by BellSouth, or meeting criteria defined by BellSouth.
- 11.1.3 BTI shall not assign to the BellSouth Premises any personnel with records of felony criminal convictions. BTI shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions, except for misdemeanor traffic violations, without advising BellSouth of the nature and gravity of the offense(s). BellSouth reserves the right to refuse building access to any BTI personnel who have been identified to have misdemeanor criminal convictions. Notwithstanding the foregoing, in the event that BTI chooses not to advise BellSouth of the nature and gravity of any misdemeanor conviction, BTI may, in the alternative, certify to BellSouth that it shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions (other than misdemeanor traffic violations).
- 11.1.4 BTI shall not knowingly assign to the BellSouth Premises any individual who was a former employee of BellSouth and whose employment with BellSouth was terminated for a criminal offense whether or not BellSouth sought prosecution of the individual for the criminal offense.
- 11.1.5 BTI shall not knowingly assign to the BellSouth Premises any individual who was a former contractor of BellSouth and whose access to a BellSouth Premises was revoked due to commission of a criminal offense whether or not BellSouth sought prosecution of the individual for the criminal offense.
- 11.1.6 For each BTI employee requiring access to a BellSouth Premises pursuant to this Attachment, BTI shall furnish BellSouth, prior to an employee gaining such access, a certification that the aforementioned background check and security training were completed. The certification will contain a statement that no felony convictions were found and certifying that the security training was completed by the employee. If the employee's criminal history includes misdemeanor convictions, BTI will disclose the nature of the convictions to BellSouth at that time. In the alternative, BTI may certify to BellSouth that it shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions other than misdemeanor traffic violations.
- 11.1.7 At BellSouth's request, BTI shall promptly remove from BellSouth's Premises any employee of BTI BellSouth does not wish to grant access to its premises 1) pursuant

to any investigation conducted by BellSouth or 2) prior to the initiation of an investigation in the event that an employee of BTI is found interfering with the property or personnel of BellSouth or another CLEC, provided that an investigation shall promptly be commenced by BellSouth.

- 11.2 Notification to BellSouth. BellSouth reserves the right to interview BTI's employees, agents, or contractors in the event of wrongdoing in or around BellSouth's property or involving BellSouth's or another CLEC's property or personnel, provided that BellSouth shall provide reasonable notice to BTI's Security contact of such interview. BTI and its contractors shall reasonably cooperate with BellSouth's investigation into allegations of wrongdoing or criminal conduct committed by, witnessed by, or involving BTI's employees, agents, or contractors. Additionally, BellSouth reserves the right to bill BTI for all reasonable costs associated with investigations involving its employees, agents, or contractors if it is established and mutually agreed in good faith that BTI's employees, agents, or contractors are responsible for the alleged act. BellSouth shall bill BTI for BellSouth property which is stolen or damaged where an investigation determines the culpability of BTI's employees, agents, or contractors and where BTI agrees, in good faith, with the results of such investigation. BTI shall notify BellSouth in writing immediately in the event that the CLEC discovers one of its employees already working on the BellSouth premises is a possible security risk. Upon request of the other Party, the Party who is the employer shall discipline consistent with its employment practices, up to and including removal from BellSouth Premises, any employee found to have violated the security and safety requirements of this section. BTI shall hold BellSouth harmless for any damages resulting from such removal of its personnel from BellSouth premises.
- 11.3 <u>Use of Supplies</u>. Unauthorized use of telecommunications equipment or supplies by either Party, whether or not used routinely to provide telephone service (e.g. plug-in cards,) will be strictly prohibited and handled appropriately. Costs associated with such unauthorized use may be charged to the offending Party, as may be all associated investigative costs.
- 11.4 <u>Use of Official Lines</u>. Except for non-toll calls necessary in the performance of their work, neither Party shall use the telephones of the other Party on the BellSouth Premises. Charges for unauthorized telephone calls may be charged to the offending Party, as may be all associated investigative costs.
- 11.5 <u>Accountability</u>. Full compliance with the Security requirements of this section shall in no way limit the accountability of either Party to the other for the improper actions of its employees.

#### 12. Destruction of Collocation Space

12.1 In the event a Collocation Space is wholly or partially damaged by fire, windstorm, tornado, flood or by similar causes to such an extent as to be rendered wholly unsuitable for BTI's permitted use hereunder, then either Party may elect within ten

(10) business days after such damage, to terminate occupancy of the damaged Collocation Space, and if either Party shall so elect, by giving the other written notice of termination, both Parties shall stand released of and from further liability under the terms hereof. If the Collocation Space shall suffer only minor damage and shall not be rendered wholly unsuitable for BTI's permitted use, or is damaged and the option to terminate is not exercised by either Party, BellSouth covenants and agrees to proceed promptly without expense to BTI, except for improvements not the property of BellSouth, to repair the damage. BellSouth shall have a reasonable time within which to rebuild or make any repairs, and such rebuilding and repairing shall be subject to delays caused by storms, shortages of labor and materials, government regulations, strikes, walkouts, and causes beyond the control of BellSouth, which causes shall not be construed as limiting factors, but as exemplary only. BTI may, at its own expense, accelerate the rebuild of its collocated space and equipment provided however that a BellSouth Certified Contractor is used and the necessary space preparation has been completed. Rebuild of equipment must be performed by a BellSouth Certified Vendor. If BTI's acceleration of the project increases the cost of the project, then those additional charges will be incurred by BTI. Where allowed and where practical, BTI may erect a temporary facility while BellSouth rebuilds or makes repairs. In all cases where the Collocation Space shall be rebuilt or repaired, BTI shall be entitled to an equitable abatement of rent and other charges, depending upon the unsuitability of the Collocation Space for BTI's permitted use, until such Collocation Space is fully repaired and restored and BTI's equipment installed therein (but in no event later than thirty (30) business days after the Collocation Space is fully repaired and restored). Where BTI has placed an Adjacent Arrangement pursuant to Section 3.5, BTI shall have the sole responsibility to repair or replace said Adjacent Arrangement provided herein. Pursuant to this section, BellSouth will restore the associated services to the Adjacent Arrangement.

#### 13. Eminent Domain

13.1 If the whole of a Collocation Space or Adjacent Arrangement shall be taken by any public authority under the power of eminent domain, then this Attachment shall terminate with respect to such Collocation Space or Adjacent Arrangement as of the day possession shall be taken by such public authority and rent and other charges for the Collocation Space or Adjacent Arrangement shall be paid up to that day with proportionate refund by BellSouth of such rent and charges as may have been paid in advance for a period subsequent to the date of the taking. If any part of the Collocation Space or Adjacent Arrangement shall be taken under eminent domain, BellSouth and BTI shall each have the right to terminate this Attachment with respect to such Collocation Space or Adjacent Arrangement and declare the same null and void, by written notice of such intention to the other Party within ten (10) business days after such taking.

### 14. Nonexclusivity

14.1 BTI understands that this Attachment is not exclusive and that BellSouth may enter into similar agreements with other Parties. Assignment of space pursuant to all such agreements shall be determined by space availability and made on a first come, first served basis.

## EXHIBIT A: BELLSOUTH/BTI RATES – ALABAMA PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and are subject to true-up

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3,760.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,134.00 Minimum
PE1SJ PE1SK PE1SL	Space Preparation Fees Firm Order Processing* Central Office Modifications* Common Systems Modifications –	Per sq. ft. Per sq. ft.	\$2.24 \$3.01	\$1,211.00
PE1SM	Cageless* Common Systems Modifications – Caged*	Per cage	\$102.16	
	Space Enclosure (100 sq. ft. minimum)			
PE1BW PE1CW	Welded Wire-mesh Welded Wire-mesh	Per first 100 sq. ft. Per add'1 50 sq. ft.	\$178.65 \$17.52	NA NA
PE1PJ	Floor Space	Per sq. ft.	\$3.68	NA
PE1BD	Cable Installation	Per cable	NA	\$1,751.00
PE1PM	Cable Support Structure	Per entrance cable	\$19.67	NA
PE1PL PE1FB PE1FD PE1FE PE1FG	Power -48V DC Power* 120V AC Power single phase* 240V AC Power single phase* 120V AC Power three phase* 277 AC Power three phase*	Per amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp	\$9.00 \$5.63 \$11.26 \$16.89 \$38.99	NA - - - -
PE1P2 PE1P4 PE1P1 PE1P3 PE1F2 PE1F4	Cross Connects 2-wire 4-wire DS-1 DS-3 2-fiber 4-fiber	Per cross connect	\$.031 \$.062 \$1.28 \$16.27 \$3.23 \$5.73	First/Add'1 \$33.68/\$31.79 \$33.63/\$31.67 \$52.93/\$39.87 \$51.99/\$38.59 \$52.00/\$38.60 \$64.54/\$51.14

	Al	LABAMA (continued)		
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
	Co-Carrier Cross-Connect  Prior to the effective date of this agreement for existing arrangements. (Note 3)			
PE1ES Fiber	Fiber Arrangement	Cable Support Structure, per linear	\$0.06	NA
PE1DS Copper	Copper or Coaxial	foot (existing)	\$0.03	NA
		Cable Support Structure (new)	NA	ICB
PE1AX	Security Access System Security System*	Per central office	\$52.27	
PE1A1	New Access Card Activation*	Per card	\$.059	\$55.57
PE1AA	Administrative change, existing card*	Per card		\$15.58
PE1AR	Replace lost or stolen card*	Per card		\$45.56
PE1AK	Initial Key	Per key	NA	\$26.19
PE1AL	Replace lost or stolen key	Per key	NA	\$26.19
PE1SR	Space Availability Report*	Per premises requested		\$2,150.00
	POT Bay Arrangements Prior to 6/1/99	Per cross connect		
PE1PE	2-Wire Cross-Connect		\$0.08	NA
PE1PF	4-Wire Cross-Connect		\$0.17	NA
PE1PG	DS1 Cross-Connect		\$0.69	NA
PE1PH	DS3 Cross-Connect		\$4.74	NA
PE1B2 PE1B4	2-Fiber Cross-Connect 4-Fiber Cross-Connect		\$32.02 \$40.48	NA NA
	Cable Records <sup>1</sup>			Note 2
				Initial/Subsequent
PE1CR	Cable Records	Per request	NA	\$1708/\$1166
PE1CD	VG/DS0 Cable	Per cable record	NA	\$923.51/\$923.51
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.02/\$18.02
PE1C1	DS1	Per T1TIE	NA	\$8.44/\$8.44
PE1C3	DS3	Per T3TIE	NA	\$29.53/\$29.53
PE1CB	Fiber Cable	Per cable record	NA	\$278.95/\$278.95

	ALABAMA (continued)					
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)		
	Security Escort	Per half hr/add'l half				
		hr				
PE1BT	Basic Time		NA	\$33.85/\$21.45		
PE1OT	Overtime		NA	\$44.09/\$27.71		
PE1PT	Premium Time		NA	\$54.33/\$33.96		

#### Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3 **Co-Carrier Cross-Connect.**. For arrangements in place prior to the effective date of this agreement, the recurring charges as stated in this Exhibit A shall apply.

## EXHIBIT A: BELLSOUTH/BTI RATES – FLORIDA PHYSICAL COLLOCATION

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per request		\$3,791.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,160.00
	Space Preparation Fees			
PE1SJ	Firm Order Processing		<b>42.7</b> 0	\$1,211.00
PE1SK	Central Office Modifications	Per sq. ft.	\$2.58	
PE1SL	Common Systems Modifications –	Per sq. ft.	\$2.96	
DE1CM	Cageless	D	φ100 cc	
PE1SM	Common Systems Modifications –	Per cage	\$100.66	
	Caged			
	Space Enclosure (100 sq. ft.			
	minimum)			
PE1BW	Wire Cage	Per first 100 sq. ft.	\$205.93	NA
PE1CW	Wire Cage	Per add'l 50 sq. ft.	\$20.20	NA
TLICW	Whe Cage	Ter add 150 sq. it.	Ψ20.20	1111
PE1PJ	Floor Space	Per sq. ft.	\$6.57	NA
	T. W.	1 1		<u></u>
PE1BD	Cable Installation	Per cable		\$1,826.00
PE1PM	Cable Support Structure		\$21.66	NA
	Power			
PE1PL	-48V DC Power	Per amp	\$8.86	NA
PE1FB	120V AC Power single phase	Per breaker amp	\$5.62	-
PE1FD	240V AC Power single phase	Per breaker amp	\$11.26	-
PE1FE	120V AC Power three phase	Per breaker amp	\$16.88	-
PE1FG	277 AC Power three phase	Per breaker amp	\$38.98	-
				TP - / A 1 144
	Cross Connects	D .	ф o <del>д</del> 4	First/Add'l
	2-wire	Per cross connect	\$.074	\$34.53/\$32.51
	4-wire	Per cross connect	\$.148	\$34.54/\$32.53
	DS1	Per cross connect	\$1.29	\$54.15/\$40.94
	DS3 2-fiber	Per cross connect	\$17.48	\$53.28/\$39.65
		Per cross connect	\$2.96 \$5.66	\$53.28/\$39.66
	4-fiber	Per cross connect	\$5.66	\$66.08/\$52.47

	FLORIDA (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
	Co-Carrier Cross-Connect  Prior to the effective date of this agreement for existing arrangements. (Note 4)		(233)		
PE1ES Fiber	Fiber Cable Support Structure, existing	Per linear foot	\$0.06	NA	
PE1DS Copper	Copper or Coaxial Cable Support Structure, existing	Per linear foot	\$0.03	NA	
TBD	Cable Support Structure Construction, new	Per new construction	NA	ICB	
PE1AX	Security Access System Security System	Per premises	\$89.48		
PE1A1	New Access Card Activation	Per card	\$.06	\$56.03	
PE1AA	Administrative change, existing card	Per card		\$15.71	
PE1AR	Replace lost or stolen card	Per card		\$45.93	
PE1AK PE1AL	Initial Key Replace lost or stolen key	Per key Per key	NA NA	\$26.41 \$26.41	
PE1SR	Space Availability Report	Per premises requested		\$2,168.00	
	POT Bay (Note 1)		NA	NA	
	Cable Records <sup>2</sup>			Note 3	
				initial/subsequent	
PE1CR	Cable Records	Per request	NA	\$1709/\$1166	
PE1CD	VG/DS0 Cable	Per cable record	NA	\$923.86/\$923.86	
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.03/\$18.03	
PE1C1	DS1	Per T1TIE	NA	\$8.44/\$8.44	
PE1C3	DS3	Per T3TIE	NA	\$29.54/\$29.54	
PE1CB	Fiber Cable	Per cable record	NA	\$279.05/\$279.05	
	Security Escort	Per ¼ hour			
PE1BQ	Basic Time		NA	\$10.89	
PE1OQ	Overtime		NA	\$13.64	
PE1PQ	Premium Time		NA	\$16.40	

#### Florida (Continued)

#### **Note(s):**

N/A refers to rate elements which do not have a negotiated rate.

- (1) **POT Bays**: BellSouth's Florida specific rates were established in the Florida Public Service Commission Docket No. 960833. The Commission did not set permanent rates for <u>POT Bays</u>, given the assumption by the Parties to the Proceeding that they will always provide their own POT Bays. It will be necessary for BTI to provide its own POT Bays per BellSouth specifications and provide the necessary information from which BellSouth can inventory.
- (2) Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.
- (3) The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.
- (4) **Co-Carrier Cross-Connect.**. For arrangements in place prior to the effective date of this agreement, the recurring charges as stated in this Exhibit A shall apply.

# EXHIBIT A: BELLSOUTH/BTI RATES – GEORGIA PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and subject to true-up

USOC	Rate Element Description	Unit	Recurring Rate	Non-Recurring
	P		(RC)	Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3,755.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,130.00
TETETT	Subsequent rippineuron rec	Ter request	1171	Minimum
PE1BB	Space Preparation Fee (Note 2)	Per sq. ft.	NA	\$100.00
TEIDD	Space Freparation Fee (140te 2)	Ter sq. it.	IVA	Ψ100.00
	Space Enclosure (100 sq. ft.			
PE1BW	minimum) Welded Wire-mesh	Per first 100 sq. ft.	\$187.36	NA
PE1CW	Welded Wire-mesh	Per add'l 50 sq. ft.	\$18.38	NA NA
	Floor Space		7 3 3 3 3	
PE1PJ	Zone A	Per sq. ft.	\$4.47	NA
PE1PK	Zone B	Per sq. ft.	\$4.47	NA
PE1BD	Cable Installation	Per cable	NA	\$1,693.00
PE1PM	Cable Support Structure	Per entrance cable	\$19.26	NA
	Power			
PE1PL	-48V DC Power	Per amp	\$5.00	NA
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.52	-
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.05	-
PE1FE	120V AC Power three phase*	Per breaker amp	\$16.58	-
PE1FG	277 AC Power three phase*	Per breaker amp	\$38.27	-
	Cross Connects	Per cross connect		First/Add'1
PE1P2	2-wire		\$0.030	\$33.76/\$31.86
PE1P4	4-wire		\$0.061	\$33.77/\$31.80
PE1P1	DS-1		\$1.13	\$53.05/\$39.99
PE1P3	DS-3		\$14.43	\$52.14/\$38.71
PE1F2	2-fiber		\$2.86	\$52.14/\$38.72
PE1F4	4-fiber		\$5.08	\$64.74/\$51.31

GEORGIA (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
	Co-Carrier Cross-Connect  Prior to the effective date of this agreement for existing arrangements. (Note 3)			( 7
PE1ES Fiber	Fiber Cable Support Structure, existing	Per linear foot	\$0.06	NA
PE1DS Copper	Copper or Coaxial Cable Support Structure, existing	Per linear foot	\$0.03	NA
TBD	Cable Support Structure Construction, new	Per new construction	NA	ICB
PE1AX	Security Access System Security System*	Per premises	\$40.00	
PE1A1	New Access Card Activation*	Per card	\$.058	\$55.51
PE1AA	Administrative change, existing card*	Per card		\$15.56
PE1AR	Replace lost or stolen card*	Per card		\$45.50
PE1AK	Initial Key	Per key	NA	\$26.16
PE1AL	Replace lost or stolen key	Per key	NA	\$26.16
PE1SR	Space Availability Report*	Per premises requested		\$2,148.00
	POT Bay Arrangements	Dor aross connect		
	Prior to 6/1/99	Per cross-connect		
PE1PE	2-Wire Cross-Connect		\$0.40	NA
PE1PF	4-Wire Cross-Connect		\$1.20	NA
PE1PG	DS1 Cross-Connect		\$1.20	NA
PE1PH	DS3 Cross-Connect		\$8.00	NA
PE1B2	2 Fiber Cross-Connect		\$38.79	NA
PE1B4	4 Fiber Cross-Connect		\$52.31	NA
	Cable Records <sup>1</sup>			Note 2
				Initial/subsequent
PE1CR	Cable Records	Per request	NA	\$1706/\$1164
PE1CD	VG/DS0 Cable	Per cable record	NA	\$922.38/\$922.38
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.00/\$18.00
PE1C1	DS1	Per T1TIE	NA	\$8.43/\$8.43
PE1C3	DS3	Per T3TIE	NA	\$29.49/\$29.49
PE1CB	Fiber Cable	Per cable record	NA	\$278.61/\$278.61

### Georgia (Continued)

	Security Escort	Per half hr./Add'l		
		half hr.		
PE1BT	Basic Time		NA	\$33.81/\$21.42
PE1OT	Overtime		NA	\$44.03/\$27.67
PE1PT	Premium Time		NA	\$54.26/\$33.92

N/A refers to rate elements which do not have a negotiated rate.

- Note (1) Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.
- Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.
- Note 3 **Co-Carrier Cross-Connect.**. For arrangements in place prior to the effective date of this agreement, the recurring charges as stated in this Exhibit A shall apply.

## EXHIBIT A: BELLSOUTH/BTI RATES – KENTUCKY PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate	Non-Recurring
DE1D A	A 1' 4' E	D 4	(RC)	Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3,761.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,135.00
FEICA	Subsequent Application Fee	rei request	INA	Minimum
				IVIIIIIIIIIIII
	Space Preparation Fees			
PE1SJ	Firm Order Processing*			\$1,202.00
PE1SK	Central Office Modifications*	Per sq. ft.	\$2.38	. ,
PE1SL	Common Systems Modifications –	Per sq. ft.	\$3.30	
	Cageless*	1	·	
PE1SM	Common Systems Modifications –	Per cage	\$112.11	
	Caged*			
	Space Enclosure (100 sq. ft.			
DE1DIV	minimum)	D C . 100 C	Φ100.0 <b>7</b>	NTA
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$189.85	NA
PE1CW	Welded Wire-mesh	Per add'l 50 sq. ft.	\$18.62	NA
PE1PJ	Floor Space	Per sq. ft.	\$8.20	NA
PE1BD	Cable Installation	Per cable	NA	\$1,755.00
FEIDD	Cable Histaliation	r ei cable	IVA	\$1,733.00
PE1PM	Cable Support Structure	Per entrance cable	\$20.14	NA
DE1DI	Power	D	Φ0.77	NTA
PE1PL	-48V DC Power*	Per amp	\$8.77	NA
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.58	-
PE1FD	240V AC Power three phase*	Per breaker amp	\$11.16	-
PE1FE	120V AC Power three phase*	Per breaker amp	\$16.74	-
PE1FG	277 AC Power three phase*	Per breaker amp	\$38.65	<u>-</u>
	Cross Connects	Per cross connect		First/Add'1
PE1P2	2-wire	1 of cross connect	\$0.037	\$33.67/\$31.78
PE1P4	4-wire		\$0.075	\$33.66/\$31.70
PE1P1	DS-1		\$1.51	\$52.97/\$39.90
PE1P3	DS-3		\$19.15	\$52.04/\$38.62
PE1F2	2-fiber		\$3.80	\$52.04/\$38.63
PE1F4	4-fiber		\$6.75	\$64.59/\$51.18

		TUCKY (continued)		
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
	Co-Carrier Cross-Connect  Prior to the effective date of this agreement for existing arrangements. (Note 3)			
PE1ES Fiber	Fiber Cable Support Structure, existing	Per linear foot	\$0.06	NA
PE1DS Copper	Copper or Coaxial Cable Support Structure, existing	Per linear foot	\$0.03	NA
TBD	Cable Support Structure Construction, new	Per new construction	NA	ICB
PE1AX	Security Access System Security System*	Per premises	\$78.11	
PE1A1	New Access Card Activation	Per card	\$.059	\$55.59
PE1AA	Administrative change, existing card	Per card		\$15.59
PE1AR	Replace lost or stolen card	Per card		\$45.58
PE1AK PE1AL	Initial Key Replace lost or stolen key	Per key Per key	NA NA	\$26.20 \$26.20
PE1SR	Space Availability Report	Per premises requested		\$2,151
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect		
PE1PE	2-Wire Cross-Connect		\$0.06	NA
PE1PF	4-Wire Cross-Connect		\$0.15	NA
PE1PG	DS1 Cross-Connect		\$0.58	NA
PE1PH	DS3 Cross-Connect		\$4.51	NA
PE1B2	2 Fiber Cross-Connect		\$38.79	NA
PE1B4	4 Fiber Cross-Connect		\$52.31	NA
	Security Escort	Per half hr./Add'l half hr.		
PE1BT	Basic Time		NA	\$33.86/\$21.46
PE1OT	Overtime		NA	\$44.10/\$27.72
PE1PT	Premium Time		NA	\$54.35/\$33.97

#### **Kentucky (Continued)**

	Cable Records <sup>1</sup>			Note 2
				Initial/subsequent
				1
PE1CR	Cable Records	Per request	NA	\$1709/1166
PE1CD	VG/DS0 Cable	Per cable record	NA	\$923.83/\$923.83
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.03/\$18.03
PE1C1	DS1	Per T1TIE	NA	\$8.44/\$8.44
PE1C3	DS3	Per T3TIE	NA	\$29.54/\$29.54
PE1CB	Fiber Cable	Per cable record	NA	\$279.05/\$279.05

#### Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3 **Co-Carrier Cross-Connect.**. For arrangements in place prior to the effective date of this agreement, the recurring charges as stated in this Exhibit A shall apply.

# EXHIBIT A: BELLSOUTH/BTI RATES – LOUISIANA PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3756.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3131.00 Minimum
PE1SJ PE1SK PE1SL	Space Preparation Fees Firm Order Processing* Central Office Modifications* Common Systems Modifications –	Per sq. ft. Per sq. ft.	\$2.60 \$3.15	\$1,200.00
PE1SM	Cageless* Common Systems Modifications – Caged*	Per cage	\$105.87	
	Space Enclosure (100 sq. ft.			
	minimum)			
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$207.06	NA
PE1CW	Welded Wire-mesh	Per add'1 50 sq. ft.	\$20.31	NA
PE1PJ	Floor Space	Per sq. ft.	\$5.94	NA
PE1BD	Cable Installation	Per cable	NA	\$1,753.00
PE1PM	Cable Support Structure	Per entrance cable	\$21.16	NA
DE1DI	Power -48V DC Power*	December	¢0.20	NT A
PE1PL PE1FB	120V AC Power single phase*	Per amp Per breaker amp	\$9.20 \$5.66	NA
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.34	-
PE1FE	120V AC Power three phase*	Per breaker amp	\$17.00	_
PE1FG	277 AC Power three phase*	Per breaker amp	\$39.26	
	Cross Connects	Per cross connect		First/Add'1
PE1P2	2-wire		\$0.036	\$33.61/\$31.76
PE1P4	4-wire		\$0.073	\$33.53/\$31.58
PE1P1	DS-1		\$1.20	\$52.80/\$39.76
PE1P3	DS-3		\$15.26	\$51.86/\$38.49

	LOUISIANA (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
	Cross Connects (continued)	Per cross connect		First/Add'1	
PE1F2	2-fiber		\$3.03	\$51.86/\$38.49	
PE1F4	4-fiber		\$5.38	\$64.36/\$50.99	
	Co-Carrier Cross-Connect				
	Prior to the effective date of this				
	agreement for existing				
	arrangements. (Note 3)				
PE1ES Fiber	Fiber Cable Support Structure, existing	Per linear foot	\$0.06	NA	
	Copper or Coaxial Cable Support				
PE1DS	Structure, existing	Per linear foot	\$0.03	NA	
Copper	Calla Carrant Street Carration				
TBD	Cable Support Structure Construction, new	Per new construction	NA	ICB	
	new				
PE1AX	Security Access System Security	Per premises	\$60.60		
FLIAA	System*	rei premises	\$00.00		
PE1A1	New Access Card Activation*	Per card	\$.060	\$55.51	
PE1AA	Administrative change, existing	Per card	φ.σσσ	\$15.57	
	card*			,	
PE1AR	Replace lost or stolen card	Per card		\$45.51	
PE1AK	Initial Key	Per key	NA	\$26.16	
PE1AL	Replace lost or stolen key	Per key	NA	\$26.16	
PE1SR	Space Availability Report*	Per premises		\$2,148	
LISK	Space Wandomity Report	requested		Ψ2,140	
	DOT D	<b>D</b>			
	POT Bay Arrangements  Prior to 6/1/99	Per cross-connect			
PE1PE	2-Wire Cross-Connect		\$0.0776	NA	
PE1PF	4-Wire Cross-Connect		\$0.0776	NA NA	
PE1PG	DS1 Cross-Connect		\$0.1332	NA NA	
PE1PH	DS3 Cross-Connect		\$4.75	NA NA	
PE1B2	2 Fiber Cross-Connect		\$47.44	NA NA	
PE1B4	4 Fiber Cross-Connect		\$63.97	NA NA	
TEIDT	+ 1 lber Cross Connect		ψ03.71	141	
	Cable Records <sup>1</sup>			Note 2	
			1		
				Initial/subsequent	
PE1CR	Cable Records	Per request	NA	Initial/subsequent \$1706/\$1165	
PE1CR PE1CD	Cable Records VG/DS0 Cable	Per request Per cable record	NA NA	•	
				\$1706/\$1165	
PE1CD	VG/DS0 Cable	Per cable record	NA	\$1706/\$1165 \$922.51/\$922.51	
PE1CD PE1CO	VG/DS0 Cable VG/DS0 Cable	Per cable record Per each 100 pair	NA NA	\$1706/\$1165 \$922.51/\$922.51 \$18.00/\$18.00	

	LOUISIANA (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
	Security Escort	Per half hr./Add'l half hr.			
PE1BT	Basic Time		NA	\$33.97/\$21.53	
PE1OT	Overtime		NA	\$44.25/\$27.81	
PE1PT	Premium Time		NA	\$54.53/\$34.09	

#### Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3 **Co-Carrier Cross-Connect.**. For arrangements in place prior to the effective date of this agreement, the recurring charges as stated in this Exhibit A shall apply.

# EXHIBIT A: BELLSOUTH/BTI RATES – MISSISSIPPI PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and are subject to true-up.

Rates marked with an asterisk (*) are interim and are subject to true-up.					
USOC	Rate Element Description	Unit	Recurring Rate	Non-Recurring	
			(RC)	Rate (NRC)	
PE1BA	Application Fee	Per request	NA	\$3,755.00	
PE1CA	Subsequent Application Fee	Per request	NA	\$3,130.00	
				Minimum	
DE161	Space Preparation Fees			44.000.00	
PE1SJ	Firm Order Processing*			\$1,200.00	
PE1SK	Central Office Modifications*	Per sq. ft.	\$2.61		
PE1SL	Common Systems Modifications –	Per sq. ft.	\$2.88		
DE1CM	Cageless*	D	¢07.05		
PE1SM	Common Systems Modifications –	Per cage	\$97.85		
	Caged*				
	Space Enclosure(100 sq. ft.				
	minimum)				
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$208.30	NA	
PE1CW	Welded Wire-mesh	Per add'1 50 sq. ft.	\$20.43	NA	
			7-0110		
PE1PJ	Floor Space	Per sq. ft.	\$6.53		
PE1BD	Cable Installation	Per cable	NA	\$1,871.00	
PE1PM	Cable Commont Standards	Per entrance cable	\$19.90	NA	
PEIPM	Cable Support Structure	Per entrance cable	\$19.90	NA	
	Power				
PE1PL	-48V DC Power*	Per amp	\$8.96	NA	
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.61	-	
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.23	-	
PE1FE	120V AC Power three phase*	Per breaker amp	\$16.84	-	
PE1FG	277 AC Power three phase*	Per breaker amp	\$38.89		
	Cross Connects	Per cross connect		First/Add'1	
PE1P2	2-wire		\$.038	\$33.65/\$31.77	
PE1P4	4-wire		\$.076	\$33.46/\$31.52	

	MISSISSIPPI (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
	Cross Connects (continued)	Per cross connect	` '	First/Add'1	
PE1P1	DS-1		\$1.30	\$52.73/\$39.70	
PE1P3	DS-3		\$16.55	\$51.78/\$38.43	
PE1F2	2-fiber		\$3.28	\$51.78/\$38.43	
PE1F4	4-fiber		\$5.83	\$64.27/\$50.91	
	Co-Carrier Cross-Connect  Prior to the effective date of this agreement for existing arrangements. (Note 3)				
PE1ES Fiber	Fiber Arrangement Cable Support Structure, existing	Per linear foot	\$0.06	NA	
PE1DS Copper	Copper or Arrangement, existing	Per linear foot	\$0.03	NA	
TBD	Cable Support Structure Construction, new	Per new construction	NA	ICB	
PE1AX	Security Access System Security	Per premises	\$85.54		
PE1A1	System* New Access Card Activation*	Per card	\$.061	\$55.50	
PE1AA	Administrative change, existing card*	Per card	φ.001	\$15.56	
PE1AR	Replace lost or stolen card	Per card		\$45.50	
PE1AK	Initial Key	Per key	NA	\$26.16	
PE1AL	Replace lost or stolen key	Per key	NA	\$26.16	
PE1SR	Space Availability Report*	Per premises requested		\$2,147.00	
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect			
PE1PE	2-Wire Cross-Connect		\$0.1195	NA	
PE1PF	4-Wire Cross-Connect		\$0.2389	NA	
PE1PG	DS1 Cross-Connect		\$0.9862	NA	
PE1PH	DS3 Cross-Connect		\$5.81	NA	
PE1B2	2 Fiber Cross-Connect		\$38.79	NA	
PE1B4	4 Fiber Cross-Connect		\$52.31	NA	

Mississippi (Continued)

	Cable Records <sup>1</sup>			Note 2
				Initial/subsequent
PE1CR	Cable Records	Per request	NA	\$1706/1164
PE1CD	VG/DS0 Cable	Per cable record	NA	\$922.28/\$922.28
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.00/\$18.00
PE1C1	DS1	Per T1TIE	NA	\$8.42/\$8.42
PE1C3	DS3	Per T3TIE	NA	\$29.49/\$29.49
PE1CB	Fiber Cable	Per cable record	NA	\$278.58/\$278.58

MISSISSIPPI (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
	Security Escort	Per half hr./Add'l		
		half hr.		
PE1BT	Basic Time		NA	\$33.80/\$21.42
PE1OT	Overtime		NA	\$44.03/\$27.67
PE1PT	Premium Time		NA	\$54.26/\$33.92

#### Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3 **Co-Carrier Cross-Connect.**. For arrangements in place prior to the effective date of this agreement, the recurring charges as stated in this Exhibit A shall apply.

## EXHIBIT A: BELLSOUTH/BTI RATES – NORTH CAROLINA PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate	Non-Recurring
DE1D A	A 1' To w	D .	(RC)	Rate (NRC)
PE1BA	Application Fee*	Per request	NA	\$3,850.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,119.00
ILICA	Subsequent Application rec	1 ci request	IVA	Minimum
				William
	Space Preparation Fees			
	Central Office Modification*	Per sq. ft.	\$1.57	
	Common Systems Modification –	Per sq. ft.	\$3.26	
	Cageless*	1		
	Common Systems Modification –	Per cage	\$110.79	
	Caged*			
	Power*	Per nominal –48v	\$5.76	
		DC Amp		
	G F 1 (100 C			
	Space Enclosure (100 sq. ft.			
DE 1 DIVI	minimum)	D 6 . 100 6	φ10 <b>2.7</b> ς	27.4
PE1BW	Welded Wire-mesh*	Per first 100 sq. ft.	\$102.76	NA
PE1CW	Welded Wire-mesh*	Per add'1 50 sq. ft.	\$10.44	NA
PE1PJ	Floor Space*	Per sq. ft.	\$3.45	NA
PE1BD	Cable Installation*	Per cable	NA	\$2,305.00
PE1PM	Cable Support Structure*	Per entrance cable	\$21.33	NA
	D			
PE1PL	Power -48V DC Power*	Danama	\$6.65	NA
		Per amp	· ·	NA
PE1FB PE1FD	120V AC Power single phase*	Per breaker amp	\$5.50	-
PE1FD PE1FE	240V AC Power single phase* 120V AC Power three phase*	Per breaker amp Per breaker amp	\$11.01 \$16.51	-
PE1FE PE1FG	277 AC Power three phase*	_	\$38.12	-
FEIFU	211 AC Power tillee phase.	Per breaker amp	\$30.12	
	Cross Connects (Note 1)	Per cross connect		First/Add'1
PE1P2	2-wire*	1 tr tropp connect	\$0.32	\$41.78/\$39.23
PE1P4	4-wire*		\$0.64	\$41.91/\$39.25
PE1P1	DS-1*		\$2.34	\$71.02/\$51.08
PE1P3	DS-3*		\$42.84	\$69.84/\$49.43
PE1F2	2-fiber		\$2.94	\$51.97/\$38.59
PE1F4	4-fiber		\$5.62	\$64.53/\$51.15

USOC	Rate Element Description	CAROLINA (continu	Recurring Rate	Non-Recurring
CBOC	Nate Element Description	Cint	(RC)	Rate (NRC)
	Co-Carrier Cross-Connect  Prior to the effective date of this agreement for existing arrangements. (Note 3)			
PE1ES Fiber	Fiber Arrangement Cable Support Structure, existing	Per linear foot	\$0.06	NA
PE1DS Copper	Copper or Coaxial Arrangement, existing	Per linear foot	\$0.03	NA
TBD	Cable Support Structure Construction, new	Per new construction	NA	ICB
PE1AX	Security Access System Security System*	Per premises	\$41.03	
PE1A1	New Access Card Activation*	Per card	\$.062	\$55.30
PE1AA	Administrative change, existing card*	Per card		\$15.51
PE1AR	Replace lost or stolen card	Per card		\$45.34
PE1AK PE1AL	Initial Key Replace lost or stolen key	Per key Per key	NA NA	\$26.18 \$26.18
PE1SR	Space Availability Report*	Per premises requested		\$2,140.00
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect		
PE1PE	2-Wire Cross-Connect		\$0.10	NA
PE1PF	4-Wire Cross-Connect		\$0.19	NA
PE1PG	DS1 Cross-Connect		\$0.79	NA
PE1PH	DS3 Cross-Connect		\$4.85	NA
PE1B2	2 Fiber Cross-Connect		\$45.30	NA
PE1B4	4 Fiber Cross-Connect		\$61.09	NA
	Security Escort	Per half hr./Add'l half hr.		
PE1BT	Basic Time		NA	\$42.92/\$25.56
PE1OT	Overtime		NA	\$54.51/\$32.44
PE1PT	Premium Time		NA	\$66.10/\$39.32

#### **North Carolina (Continued)**

	Cable Records <sup>1</sup>			Note 2
				Initial/subsequent
DE1CD	Cohla Dagarda	Dan magnage	NIA	\$1707/\$11 <i>65</i>
PE1CR	Cable Records	Per request	NA	\$1707/\$1165
PE1CD	VG/DS0 Cable	Per cable record	NA	\$923.08/\$923.08
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.02/\$18.02
PE1C1	DS1	Per T1TIE	NA	\$8.43/\$8.43
PE1C3	DS3	Per T3TIE	NA	\$29.51/\$29.51
PE1CB	Fiber Cable	Per cable record	NA	\$278.82/\$278.82

### EXHIBIT A: BELLSOUTH/BTI RATES – NORTH CAROLINA PHYSICAL COLLOCATION (continued)

### Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3 **Co-Carrier Cross-Connect.**. For arrangements in place prior to the effective date of this agreement, the recurring charges as stated in this Exhibit A shall apply.

## EXHIBIT A: BELLSOUTH/BTI RATES – SOUTH CAROLINA PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per request	NA NA	\$3768.00
TEIDII	1 ipplication 1 cc	1 of request	1111	Ψ3700.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,141.00
	Tr	1 1 1 1 1 1 1		Minimum
	Space Preparation Fees			
PE1SJ	Firm Order Processing*			\$1,204.00
PE1SK	Central Office Modifications*	Per sq. ft.	\$2.75	
PE1SL	Common Systems Modifications –	Per sq. ft.	\$3.24	
	Cageless*			
PE1SM	Common Systems Modifications –	Per cage	\$110.17	
	Caged*			
	Space Enclosure (100 sq. ft.			
	minimum)			
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$219.19	NA
PE1CW	Welded Wire-mesh	Per add'l 50 sq. ft.	\$21.50	NA
DE1DI	Elean Conse	D	¢2.05	NTA
PE1PJ	Floor Space	Per sq. ft.	\$3.95	NA
PE1BD	Cable Installation	Per cable	NA	\$1,621.00
TEIDD		1 of cubic	1171	Ψ1,021.00
PE1PM	Cable Support Structure	Per entrance cable	\$21.33	NA
	Power			
PE1PL	-48V DC Power*	Per amp	\$9.19	NA
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.67	-
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.36	-
PE1FE	120V AC Power three phase*	Per breaker amp	\$17.03	-
PE1FG	277 AC Power three phase*	Per breaker amp	\$39.33	-
	Cross Connects	Per cross connect		First/Add'l
PE1P2	2-wire		\$.034	\$33.75/\$31.86
PE1P4	4-wire		\$.068	\$33.71/\$31.75
PE1P1	DS-1		\$1.12	\$53.05/\$39.96
PE1P3	DS-3		\$14.21	\$52.11/\$38.68
PE1F2	2-fiber		\$2.82	\$52.11/\$38.69
PE1F4	4-fiber		\$5.01	\$64.69/\$51.26

TICOC		CAROLINA (continu	,	Non De
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
	Co-Carrier Cross-Connect  Prior to the effective date of this agreement for existing arrangements. (Note 3)			
PE1ES Fiber	Fiber Arrangement Cable Support Structure, existing	Per linear foot	\$0.06	NA
PE1DS Copper	Copper or Arrangement, existing	Per linear foot	\$0.03	NA
TBD	Cable Support Structure Construction, new	Per new construction	NA	ICB
PE1AX	Security Access System Security System*	Per premises	\$74.12	
PE1A1	New Access Card Activation*	Per card	\$.060	\$55.70
PE1AA	Administrative change, existing card*	Per card		\$15.62
PE1AR	Replace lost or stolen card	Per card		\$45.66
PE1AK	Initial Key	Per key	NA	\$26.25
PE1AL	Replace lost or stolen key	Per key	NA	\$26.25
PE1SR	Space Availability Report*	Per premises requested		\$2,155.00
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect		
PE1PE	2-Wire Cross-Connect		\$0.1091	NA
PE1PF	4-Wire Cross-Connect		\$0.2181	NA
PE1PG	DS1 Cross-Connect		\$0.9004	NA
PE1PH	DS3 Cross-Connect		\$5.64	NA
PE1B2	2 Fiber Cross-Connect		\$37.36	NA
PE1B4	4 Fiber Cross-Connect		\$50.38	NA
	Security Escort	Per half hr./Add'l half hr.		
PE1BT	Basic Time		NA	\$33.92/\$21.50
PE1OT	Overtime		NA	\$44.19/\$27.77
PE1PT	Premium Time		NA	\$54.45/\$34.04

#### **South Carolina (Continued)**

	Cable Records <sup>1</sup>			Note 2
				Initial/subsequent
PE1CR	Cable Records	Per request	NA	\$1712/\$1168
PE1CD	VG/DS0 Cable	Per cable record	NA	\$925.57/\$925.57
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.06/\$18.06
PE1C1	DS1	Per T1TIE	NA	\$8.45/\$8.45
PE1C3	DS3	Per T3TIE	NA	\$29.59/\$29.59
PE1CB	Fiber Cable	Per cable record	NA	\$279.57/\$279.57

### EXHIBIT A: BELLSOUTH/BTI RATES – SOUTH CAROLINA PHYSICAL COLLOCATION (continued)

### Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3 **Co-Carrier Cross-Connect.**. For arrangements in place prior to the effective date of this agreement, the recurring charges as stated in this Exhibit A shall apply.

## EXHIBIT A: BELLSOUTH/BTI RATES – TENNESSEE PHYSICAL COLLOCATION

\* Rates are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3,767.00
		1		. ,
PE1CA	Subsequent Application Fee	Per request	NA	\$3,140.00
		•		Minimum
	Space Preparation Fees			
PE1SJ	Firm Order Processing*			\$1,204.00
PE1SK	Central Office Modifications*	Per sq. ft.	\$2.74	
PE1SL	Common Systems Modifications –	Per sq. ft.	\$2.95	
	Cageless*			
PE1SM	Common Systems Modifications –	Per cage	\$100.14	
	Caged*			
	Space Enclosure (100 sq. ft.			
	minimum)			
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$218.53	NA
PE1CW	Welded Wire-mesh	Per add'1 50 sq. ft.	\$21.44	NA
DEADY	771 0	7 0	A = 77	
PE1PJ	Floor Space	Per sq. ft.	\$6.75	NA
DE1DD	C 11 T . II d	D 11	27.4	Φ1. <b>757.</b> 00
PE1BD	Cable Installation	Per cable	NA	\$1,757.00
DE1DM	Cohla Congrant Structure	Dan autuanaa aalala	\$10.00	NT A
PE1PM	Cable Support Structure	Per entrance cable	\$19.80	NA
	Power			
PE1PL	-48V DC Power*	Per amp	\$8.87	NA
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.60	IVA
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.22	_
PE1FE	120V AC Power three phase*	Per breaker amp	\$16.82	_
PE1FG	277 AC Power three phase*	Per breaker amp	\$38.84	_
TEITO	277 Tie Tower three phase	Ter breaker amp	Ψ50.04	
	Cross Connects	Per cross connect		First/Add'1
PE1P2	2-wire	2 ST CLOSS COMMEN	\$0.033	\$33.82/\$31.92
PE1P4	4-wire		\$0.066	\$33.94/\$31.95
PE1P1	DS-1		\$1.51	\$53.27/\$40.16
PE1P3	DS-3		\$19.26	\$52.37/\$38.89
PE1F2	2-fiber		\$3.82	\$52.37/\$38.89
PE1F4	4-fiber		\$6.79	\$65.03/\$51.55

TIGOG		NESSEE (continued)	<u> </u>	N. D.
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
	Co-Carrier Cross-Connect  Prior to the effective date of this agreement for existing arrangements. (Note 3)			
PE1ES Fiber	Fiber Cable Support Structure, existing	Per linear foot	\$0.06	NA
PE1DS Copper	Copper or Coaxial Cable Support Structure, existing	Per linear foot	\$0.03	NA
TBD	Cable Support Structure Construction, new	Per new construction	NA	ICB
PE1AX	Security Access System Security System	Per premises	\$55.99	
PE1A1	New Access Card Activation	Per card	\$.059	\$55.67
PE1AA	Administrative change, existing card	Per card		\$15.61
PE1AR	Replace lost or stolen card	Per card		\$45.64
PE1AK PE1AL	Initial Key Replace lost or stolen key	Per key Per key	NA NA	\$26.24 \$26.24
PE1SR	Space Availability Report*	Per premises requested		\$2,154.00
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect		
PE1PE	2-Wire Cross-Connect		\$0.40	NA
PE1PF	4-Wire Cross-Connect		\$1.20	NA
PE1PG	DS1 Cross-Connect		\$1.20	NA
PE1PH	DS3 Cross-Connect		\$8.00	NA
PE1B2 PE1B4	2 Fiber Cross-Connect 4 Fiber Cross-Connect		\$38.79 \$52.31	NA NA
	Security Escort	Per half hr./Add'l half hr.		
PE1BT	Basic Time		NA	\$33.91/\$21.49
PE1OT	Overtime		NA	\$44.17/\$27.76
PE1PT	Premium Time		NA	\$54.42/\$34.02

#### **Tennessee (Continued)**

				N. 4. 2
	Cable Records <sup>1</sup>			Note 2
				Initial/subsequent
PE1CR	Cable Records	Per request	NA	\$1711/\$1168
PE1CD	VG/DS0 Cable	Per cable record	NA	\$925.06/\$925.06
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.05/\$18.05
PE1C1	DS1	Per T1TIE	NA	\$8.45/\$8.45
PE1C3	DS3	Per T3TIE	NA	\$29.57/\$29.57
PE1CB	Fiber Cable	Per cable record	NA	\$279.42/\$279.42

#### Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3 **Co-Carrier Cross-Connect.**. For arrangements in place prior to the effective date of this agreement, the recurring charges as stated in this Exhibit A shall apply.

#### **EXHIBIT B**

Page 1 of 4

### ENVIRONMENTAL AND SAFETY PRINCIPLES

The following principles provide basic guidance on environmental and safety issues when applying for and establishing Physical Collocation arrangements.

#### 1. GENERAL PRINCIPLES

- 2.1 Compliance with Applicable Law. BellSouth and BTI agree to comply with applicable federal, state, and local environmental and safety laws and regulations including U.S. Environmental Protection Agency (USEPA) regulations issued under the Clean Air Act (CAA), Clean Water Act (CWA), Resource Conservation and Recovery Act (RCRA), Comprehensive Environmental Response, Compensation and Liability Act (CERCLA), Superfund Amendments and Reauthorization Act (SARA), the Toxic Substances Control Act (TSCA), and OSHA regulations issued under the Occupational Safety and Health Act of 1970, as amended and NFPA and National Electrical Codes (NEC) and the NESC ("Applicable Laws"). Each Party shall notify the other if compliance inspections are conducted by regulatory agencies and/or citations are issued that relate to any aspect of this Attachment.
- Notice. BellSouth and BTI shall provide notice to the other, including Material Safety Data Sheets (MSDSs), of known and recognized physical hazards or Hazardous Chemicals existing on site or brought on site. Each Party is required to provide specific notice for known potential Imminent Danger conditions. BTI should contact 1-800-743-6737 for BellSouth MSDS sheets.
- 1.3 Practices/Procedures. BellSouth may make available additional environmental control procedures for BTI to follow when working at a BellSouth Premises (See Section 2, below). These practices/procedures will represent the regular work practices required to be followed by the employees and contractors of BellSouth for environmental protection. BTI will require its contractors, agents and others accessing the BellSouth Premises to comply with these practices. Section 2 lists the Environmental categories where BST practices should be followed by BTI when operating in the BellSouth Premises.
- 1.4 Environmental and Safety Inspections. BellSouth reserves the right to inspect the BTI space with proper notification. BellSouth reserves the right to stop any BTI work operation that imposes Imminent Danger to the environment, employees or other persons in the area or Facility.
- 1.5 Hazardous Materials Brought On Site. Any hazardous materials brought into, used, stored or abandoned at the BellSouth Premises by BTI are owned by BTI. BTI will indemnify BellSouth for claims, lawsuits or damages to persons or property caused by these materials. Without prior written BellSouth approval, no substantial new

safety or environmental hazards can be created by BTI or different hazardous materials used by BTI at BellSouth Facility. BTI must demonstrate adequate emergency response capabilities for its materials used or remaining at the BellSouth Facility.

- 1.6 Spills and Releases. When contamination is discovered at a BellSouth Premises, the Party discovering the condition must notify BellSouth. All Spills or Releases of regulated materials will immediately be reported by BTI to BellSouth.
- 1.7 Coordinated Environmental Plans and Permits. BellSouth and BTI will coordinate plans, permits or information required to be submitted to government agencies, such as emergency response plans, spill prevention control and countermeasures (SPCC) plans and community reporting. If fees are associated with filing, BellSouth and BTI will develop a cost sharing procedure. If BellSouth's permit or EPA identification number must be used, BTI must comply with all of BellSouth's permit conditions and environmental processes, including environmental "best management practices (BMP)" (see Section 2, below) and/or selection of BST disposition vendors and disposal sites.
- 1.8 Environmental and Safety Indemnification. BellSouth and BTI shall indemnify, defend and hold harmless the other Party from and against any claims (including, without limitation, third-party claims for personal injury or death or real or personal property damage), judgments, damages, (including direct and indirect damages, and punitive damages), penalties, fines, forfeitures, costs, liabilities, interest and losses arising in connection with the violation or alleged violation of any Applicable Law or contractual obligation or the presence or alleged presence of contamination arising out of the acts or omissions of the indemnifying Party, its agents, contractors, or employees concerning its operations at the Facility.

#### 2. CATEGORIES FOR CONSIDERATION OF ENVIRONMENTAL ISSUES

When performing functions that fall under the following Environmental categories on BellSouth's Premises, BTI agrees to comply with the applicable sections of the current issue of BellSouth's Environmental and Safety Methods and Procedures (M&Ps), incorporated herein by this reference. BTI further agrees to cooperate with BellSouth to ensure that BTI's employees, agents, and/or subcontractors are knowledgeable of and satisfy those provisions of BellSouth's Environmental M&Ps which apply to the specific Environmental function being performed by BTI, its employees, agents and/or subcontractors.

The most current version of reference documentation must be requested from BellSouth.

ENVIRONMENTAL CATEGORIES	ENVIRONMENTAL ISSUES	ADDRESSED BY THE FOLLOWING DOCUMENTATION
Disposal of hazardous material or other regulated material (e.g., batteries, fluorescent tubes, solvents & cleaning	Compliance with all applicable local, state, & federal laws and regulations	<ul><li>Std T&amp;C 450</li><li>Fact Sheet Series 17000</li></ul>
materials)	Pollution liability insurance	• Std T&C 660-3
	EVET approval of contractor	<ul> <li>Approved Environmental Vendor List (Contact E/S Management)</li> </ul>
Emergency response	Hazmat/waste release/spill firesafety emergency	<ul> <li>Fact Sheet Series 1700</li> <li>Building Emergency Operations Plan (EOP) (specific to and located on Premises)</li> </ul>
Contract labor/outsourcing for services with environmental implications to be performed on BellSouth Premises	Compliance with all applicable local, state, & federal laws and regulations	• Std T&C 450
(e.g., disposition of hazardous	Performance of services in	• Std T&C 450-B
material/waste; maintenance of storage tanks)	accordance with BST's environmental M&Ps	• (Contact E/S for copy of appropriate E/S M&Ps.)
	Insurance	• Std T&C 660
Transportation of hazardous material	Compliance with all applicable local, state, & federal laws and regulations	<ul><li>Std T&amp;C 450</li><li>Fact Sheet Series 17000</li></ul>
	Pollution liability insurance	• Std T&C 660-3
	EVET approval of contractor	Approved Environmental Vendor List (Contact E/S Management)
Maintenance/operations work which may produce a waste	Compliance with all application local, state, & federal laws and regulations	• Std T&C 450
Other maintenance work	Protection of BST employees and equipment	<ul> <li>29CFR 1910.147 (OSHA Standard)</li> <li>29CFR 1910 Subpart O (OSHA Standard)</li> </ul>
Janitorial services	All waste removal and disposal must conform to all applicable	P&SM Manager -     Procurement

	federal, state and local regulations	Procurement
	All Hazardous Material and Waste  Asbestos notification and protection of employees and equipment	<ul> <li>Fact Sheet Series 17000</li> <li>GU-BTEN-001BT, Chapter 3</li> <li>BSP 010-170-001BS (Hazcom)</li> </ul>
Manhole cleaning	Compliance with all applicable local, state, & federal laws and regulations	<ul> <li>Std T&amp;C 450</li> <li>Fact Sheet 14050</li> <li>BSP 620-145-011PR Issue A, August 1996</li> </ul>
	Pollution liability insurance	• Std T&C 660-3
	EVET approval of contractor	Approved Environmental Vendor List (Contact E/S Management)
Removing or disturbing building materials that may contain asbestos	Asbestos work practices	• GU-BTEN-001BT, Chapter 3

#### 3. **DEFINITIONS**

<u>Generator</u>. Under RCRA, the person whose act produces a Hazardous Waste, as defined in 40 CFR 261, or whose act first causes a Hazardous Waste to become subject to regulation. The Generator is legally responsible for the proper management and disposal of Hazardous Wastes in accordance with regulations.

<u>Hazardous Chemical</u>. As defined in the U.S. Occupational Safety and Health (OSHA) hazard communication standard (29 CFR 1910.1200), any chemical which is a health hazard or physical hazard.

Hazardous Waste. As defined in section 1004 of RCRA.

<u>Imminent Danger</u>. Any conditions or practices at a facility which are such that a danger exists which could reasonably be expected to cause immediate death or serious harm to people or immediate significant damage to the environment or natural resources.

Spill or Release. As defined in Section 101 of CERCLA.

#### 4. ACRONYMS

<u>E/S</u> – Environmental/Safety

EVET - Environmental Vendor Evaluation Team

Version 3Q00: 11/16/00 EXHIBIT 1

 $\underline{DEC/LDEC} \text{ - Department Environmental Coordinator/Local Department Environmental Coordinator}$ 

<u>GU-BTEN-001BT</u> - BellSouth Environmental Methods and Procedures

NESC - National Electrical Safety Codes

P&SM - Property & Services Management

Std. T&C - Standard Terms & Conditions

#### Three Month CLEC Forecast

CLEC NAME	DATE
-----------	------

STATE	Central Office/City	CAGED Sq. Ft.	CAGELE	SS # Bays	FRAME TERMINA TIONS	CLEC Provided BDFB Amps Load	BST Provided BDFB Amps Load		Entrance Facilities # sheaths & # fibers	Proposed Application Date	NOTES
			Standard Bays*	Non- Standard Bays**							
North Carolina	CHRLNCLP - Charlotte		5		36/DS3's 84/DS1`'s 1200/DS0 's					6/1/01	Example
Tennessee	KNVLTNBE - Knoxville	100			36/DS3's 84/DS1`'s 1200/DS0 's					6/1/01	Example
*Standard ba	ays are defined as racks, l collocated equipment bay	pays or cat	pinets, inclu	ding equipm	ent and cab	ole, with meas	urements equa	al to or less th	an the followin	g: Width - 26"	, Depth - 25". The standard
-	ast for non-standard cagelo				nt describin	g the quantity	and width and				

Version 3Q00: 11/16/00 EXHIBIT 1

#### **EXHIBIT 2**

### **Attachment 9**

**Performance Measurements** 

#### **Performance Measurements**

Upon a particular Commission's issuance of an Order pertaining to Performance Measurements in a proceeding expressly applicable to all CLECs generally, BellSouth shall implement in that state such Performance Measurements as of the date specified by the Commission.

# Amendment to the Interconnection Agreement between Business Telecom, Inc. and BellSouth Telecommunications, Inc. Dated February 21, 2000

Pursuant to this Amendment ("Amendment"), Business Telecom, Inc. ("BTI") and BellSouth Telecommunications, Inc. ("BellSouth"), hereinafter referred to collectively as the "Parties", hereby agree to amend the Interconnection Agreement between BellSouth and BTI dated February 21, 2000 ("Interconnection Agreement").

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, BTI and BellSouth hereby covenant and agree as follows:

- 1. The following paragraph will be added to Attachment 2 as section 5.6.3.7:
  - 5.6.3.7 2-wire voice grade coin port, shared interoffice transport, and 2-wire analog loop. This Unbundled Coin port/loop combination is to be utilized with a payphone set capable of performing coin phone functions such as call rating, coin return, and timing within the set itself.
- 2. Rates for the Unbundled Coin Port/Loop Combinations are attached hereto as Exhibit 1.
- 3. All other provisions of the Interconnection Agreement dated February 21, 2000 shall remain unchanged and in full force and effect until the expiration date.
- 4. Either or both of the Parties is authorized to submit this Amendment to the appropriate regulatory agencies for approval subject to section 252(e) of the Federal Telecommunications Act of 1996.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

BellSouth Telecommunications, Inc.	Business Telecom, Inc.
Ву:	By:
Name: Greg Follensbee	Name:
Title: <u>Senior Director</u>	Title:
Date:	Date:

#### **UNBUNDLED COIN PORT/LOOP COMBINATION RATE MATRIX**

1. UNE COIN PORT/LOOP COMBINATION:	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
1.1 RECURRING CHARGES (RCs), By Month				(Note 7)						
1.1.1 UNE COIN PORTS (Note 1) (Note 2)										
2-WIRE COIN 2-WAY WITHOUT OPERATOR SCREENING AND	UEPRF	\$2.53	NA	NA	\$2.91	\$2.74	\$2.47	NA	NA	NA
WITHOUT BLOCKING (AL, KY, LA, MS)										
2-WIRE COIN 2-WAY WITHOUT OPERATOR SCREENING AND	UEPND	NA	NA	NA	NA	NA	NA	\$2.62	NA	NA
WITHOUT BLOCKING (NC)										
2-WIRE COIN 2-WAY WITHOUT OPERATOR SCREENING AND	UEPSD	NA	NA	NA	NA	NA	NA	NA	\$4.04	NA
WITHOUT BLOCKING (SC)										
2-WIRE COIN 2-WAY WITHOUT OPERATOR SCREENING AND	UEPTB	NA	NA	NA	NA	NA	NA	NA	NA	\$4.72
WITHOUT BLOCKING (TN)										
2-WIRE COIN 2-WAY WITHOUT OPERATOR SCREENING AND	UEPMC	NA	NA	NA	NA	NA	\$2.47	NA	NA	NA
WITHOUT BLOCKING; WITH DIALING PARITY (Note 3) (MS)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING (AL, KY)	UEPRE	\$2.53	NA	NA	\$2.91	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING (GA)	UEPGC	NA	NA	\$1.89	NA	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING (NC)	UEPNC	NA	NA	NA	NA	NA	NA	\$2.62	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPRA	\$2.53	NA	NA	\$2.91	\$2.74	\$2.47	NA	\$4.04	NA
BLOCKING: 011, 900/976, 1+DDD (AL, KY, LA, MS, SC)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEP2F	NA	\$1.48	NA	NA	NA	NA	NA	NA	NA
BLOCKING: 011, 900/976, 1+DDD (FL)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEP2G	NA	NA	\$1.89	NA	NA	NA	NA	NA	NA
BLOCKING: 011, 900/976, 1+DDD (GA)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPMA	NA	NA	NA	NA	NA	\$2.47	NA	NA	NA
BLOCKING: 011, 900/976, 1+DDD; WITH DIALING PARITY (MS)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPSA	NA	NA	NA	NA	NA	NA	NA	\$4.04	NA
BLOCKING: 011, 900/976, 1+DDD (SC)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPRP	NA	NA	NA	NA	NA	NA	\$2.62	NA	\$4.72
BLOCKING: 011, 900/976, 1+DDD (NC, TN)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPRB	\$2.53	NA	NA	NA	\$2.74	\$2.47	NA	NA	NA
BLOCKING (AL, LA, MS)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPFA	NA	\$1.48	NA	NA	NA	NA	NA	NA	NA
BLOCKING (FL)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPGA	NA	NA	\$1.89	NA	NA	NA	NA	NA	NA
BLOCKING (GA)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPKA	NA	NA	NA	\$2.91	NA	NA	NA	NA	NA
BLOCKING (KY)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPSH	NA	NA	NA	NA	NA	NA	NA	\$4.04	NA
BLOCKING (SC)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPNB	NA	NA	NA	NA	NA	NA	\$2.62	NA	NA
BLOCKING (NC)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPTA	NA	NA	NA	NA	NA	NA	NA	NA	\$4.72
BLOCKING (TN)										1

2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPMB	NA	NA	NA	NA	NA	\$2.47	NA	NA	NA
BLOCKING; WITH DIALING PARITY (MS)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPSC	NA	NA	NA	NA	NA	NA	NA	\$4.04	NA
BLOCKING; WITH DIALING PARITY (SC)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 900/976	UEPGB	NA	NA	\$1.89	NA	NA	NA	NA	NA	NA
BLOCKING (GA)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCD	\$2.53	NA	NA	\$2.91	\$2.74	\$2.47	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (AL, KY, LA, MS)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCG	NA	\$1.48	NA	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (FL)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCH	NA	NA	\$1.89	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (GA)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCC	NA	NA	NA	NA	NA	NA	NA	\$4.04	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (SC)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCA	NA	NA	NA	NA	NA	NA	\$2.62	NA	\$4.72
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (NC, TN)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCE	NA	NA	NA	NA	NA	NA	NA	\$4.04	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL; ENHANCED										
CALLING OPT 3YV (SC)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCF	NA	NA	NA	NA	NA	NA	NA	\$4.04	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL; ENHANCED										
CALLING OPT AP7 (SC)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCJ	NA	NA	NA	NA	NA	\$2.47	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL; WITH DIALING										
PARITY (MS)										
2-WIRE COIN OUTWARD WITHOUT BLOCKING AND WITHOUT	UEPRN	NA	NA	NA	\$2.91	\$2.74	\$2.47	NA	NA	NA
OPERATOR SCREENING (KY, LA, MS)										
2-WIRE COIN OUTWARD WITHOUT BLOCKING AND WITHOUT	UEPSG	NA	NA	NA	NA	NA	NA	NA	\$4.04	NA
OPERATOR SCREENING (SC)										
2-WIRE COIN OUTWARD WITHOUT BLOCKING AND WITHOUT	UEPME	NA	NA	NA	NA	NA	\$2.47	NA	NA	NA
OPERATOR SCREENING; WITH DIALING PARITY (MS)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPRJ	NA	NA	\$1.89	\$2.91	NA	\$2.47	NA	NA	NA
BLOCKING (GA, KY, MS)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPLA	NA	NA	NA	NA	\$2.74	NA	NA	NA	NA
BLOCKING (LA)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPSF	NA	NA	NA	NA	NA	NA	NA	\$4.04	NA
BLOCKING (SC)	===	*				L				
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPRK	\$2.53	\$1.48	NA	NA	NA	NA	NA	NA	NA
BLOCKING (AL, FL)	LIEDNIE	A1.A	N.1.A	N. A.	110	110	N. A.	<b>#0.00</b>	N. A.	<b>.</b>
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPNE	NA	NA	NA	NA	NA	NA	\$2.62	NA	NA
BLOCKING (NC)	LIEDTO	N.1.4	N 1 A	N1 A	N1.4	N1.4	N 1 A	N 1 A	N 1 A	Φ4.70
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPTC	NA	NA	NA	NA	NA	NA	NA	NA	\$4.72
BLOCKING (TN)	LIEBOE	N.1.4	M4 40	N1 A	N1.4	N1.4	N 1 A	N 1 A	N 1 A	NI A
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPOF	NA	\$1.48	NA	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+ (FL)			<u> </u>	<u> </u>	<u> </u>	<u> </u>	L	ļ	<u> </u>	

2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPMD	NA	NA	NA	NA	NA	\$2.47	NA	NA	NA
BLOCKING: WITH DIALING PARITY (MS)	DEPIND	INA	INA	INA	INA	INA	\$2.47	INA	INA	INA
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPRH	\$2.53	NA	NA	\$2.91	\$2.74	\$2.47	NA	NA	NA
BLOCKING: 011, 900/976, 1+DDD (AL, KY, LA, MS)	UEPKH	φ2.55	INA	INA	φ2.91	Φ2.74	Φ2.47	INA	INA	INA
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPSJ	NA	NA	NA	NA	NA	NA	NA	\$4.04	NA
	UEFSJ	INA	INA	INA	INA	INA	INA	INA	\$4.04	INA
BLOCKING: 011, 900/976, 1+DDD (SC) 2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCN	\$2.53	NA	NA	\$2.91	\$2.74	\$2.47	NA	NA	NA
	UEPCIN	φ∠.53	INA	INA	φ2.91	\$2.74	\$∠.47	INA	INA	INA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (AL, KY, LA, MS)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCQ	NA	\$1.48	\$1.89	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (FL, GA)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCM	\$2.53	NA	NA	NA	NA	NA	NA	\$4.04	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (SC)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCL	NA	NA	NA	NA	NA	NA	\$2.62	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (NC)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPOT	NA	NA	NA	NA	NA	NA	NA	NA	\$4.72
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (TN)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCP	NA	NA	NA	NA	NA	NA	NA	\$4.04	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL; WITH ENHANCED										
CALLING OPT 3YW (SC)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCS	NA	NA	NA	NA	NA	\$2.47	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL; WITH DAILING										
PARITY (MS)										
1.1.2 UNE 2-WIRE VOICE GRADE LOOPS (RCs)										
RC – 2-WIRE VOICE GRADE LOOP – STATEWIDE	UEPLX	NA	NA	NA	NA	NA	NA	\$14.18	NA	NA
RC – 2-WIRE VOICE GRADE LOOP – ZONE 1 (Note 6)	UEPLX	\$14.35	\$14.90	\$10.80	\$14.79	\$14.05	\$14.59	NA	\$17.02	\$12.48
RC – 2-WIRE VOICE GRADE LOOP – ZONE 2 (Note 6)	UEPLX	\$23.31	\$18.51	\$12.47	\$27.68	\$24.14	\$19.33	NA	\$25.66	\$14.42
RC – 2-WIRE VOICE GRADE LOOP – ZONE 3 (Note 6)	UEPLX	\$42.24	\$24.25	\$19.83	\$47.78	\$49.30	\$27.63	NA	\$33.99	\$21.77
RC – 2-WIRE VOICE GRADE LOOP – ZONE 4 (Note 6)	UEPLX	NA	NA	NA	NA	NA	\$36.47	NA	NA	NA
1.1.3 UNE COIN PORT/LOOP COMBINATION (RCs)										
RC – 2-WIRE VOICE GRADE LOOP WITH APPLICABLE UNE COIN	Note 4	NA	NA	NA	NA	NA	NA	\$16.80	NA	NA
PORT – STATEWIDE										
RC – 2-WIRE VOICE GRADE LOOP WITH APPLICABLE UNE COIN	Note 4	\$16.88	\$16.38	\$12.69	\$17.70	\$16.79	\$17.06	NA	\$21.06	\$17.20
PORT – ZONE 1										
RC – 2-WIRE VOICE GRADE LOOP WITH APPLICABLE UNE COIN	Note 4	\$25.84	\$19.99	\$14.36	\$30.59	\$26.88	\$21.80	NA	\$29.70	\$19.14
PORT – ZONE 2										
RC – 2-WIRE VOICE GRADE LOOP WITH APPLICABLE UNE COIN	Note 4	\$44.77	\$25.73	\$21.72	\$50.69	\$52.04	\$30.10	NA	\$38.03	\$26.49
PORT – ZONE 3										
RC – 2-WIRE VOICE GRADE LOOP WITH APPLICABLE UNE COIN	Note 4	NA	NA	NA	NA	NA	\$38.94	NA	NA	NA
PORT – ZONE 4										
	İ									
1.1.4 ADDITIONAL UNE COIN PORT/LOOP (RCs)										
UNE COIN PORT/LOOP COMBO USAGE (FLAT RATE)	URECU	\$1.56	\$1.86	\$3.59	\$2.57	\$1.81	\$4.62	\$3.70	\$4.05	\$3.45

1.2 NONRECURRING CHARGES (NRC)										
1.2.1 CURRENTLY COMBINED UNE COIN PORT/LOOP COMBO										
(NRCs) (Note 8)										
CONVERSION CHARGE – SWITCH AS IS – 1ST	USAC2	\$2.80	\$0.20	\$2.01	\$10.00	\$3.80	\$5.20	\$2.77	\$1.59	\$1.03
CONVERSION CHARGE – SWITCH AS IS – ADD'L	USAC2	\$0.41	\$0.20	\$0.31	\$10.00	\$0.29	\$0.41	\$0.40	\$0.40	\$0.29
CONVERSION CHARGE – SWITCH WITH CHANGE – 1ST	USACC	\$2.80	\$0.20	\$2.01	\$10.00	\$3.80	\$5.20	\$2.77	\$1.59	\$1.03
CONVERSION CHARGE – SWITCH WITH CHANGE – ADD'L	USACC	\$0.41	\$0.20	\$0.31	\$10.00	\$0.29	\$0.41	\$0.40	\$0.40	\$0.29
1.2.1.1 ADDITIONAL NRCs (Note 9)										
SUBSEQUENT ACTIVITY (Note 5)	USAS2	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
1.2.1.2 OSS NRCs (Note 9)										
OSS LSR CHARGE, ELECTRONIC, PER LSR RECEIVED FROM THE CLEC BY ONE OF THE OSS INTERACTIVE INTERFACES	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
INCREMENTAL MANUAL SERVICE ORDER CHARGE – 1ST	SOMAN	\$40.71	\$21.56	\$33.67	\$19.99	\$31.92	\$43.52	\$40.18	\$43.19	\$30.89
INCREMENTAL MANUAL SERVICE ORDER CHARGE – ADD'L	SOMAN	\$9.58	\$21.56	\$7.88	\$19.99	\$7.32	\$9.99	\$9.45	\$9.91	\$7.03
SUBSEQUENT DATABASE UPDATE - ELECTRONIC		\$1.44	NA	NA	NA	\$2.11	\$2.87	\$1.42	\$0.71	\$0.76
SUBSEQUENT DATABASE UPDATE – MANUAL SERVICE ORDER		\$8.25	NA	NA	NA	\$5.12	\$6.88	\$10.27	\$8.91	\$7.97
ELECTRONIC SERVICE ORDER DISCONNECT	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
INCREMENTAL MANUAL SERVICE ORDER DISCONNECT	SOMAN	\$20.00	\$3.84	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00
1.2.2 NOT CURRENTLY COMBINED (NRCs) (APPLICABLE IN GA AS SPECIFIED IN Note 8)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING (GA) – 1ST	UEPGC	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING (GA) - ADD'L	UEPGC	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEP2G	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
BLOCKING: 011, 900/976, 1+DDD (GA) - 1ST				,						
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND BLOCKING: 011, 900/976, 1+DDD (GA) – ADD'L	UEP2G	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPGA	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
BLOCKING (GA) – 1ST										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011 BLOCKING (GA) – ADD'L	UEPGA	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 900/976	UEPGB	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
BLOCKING (GA) – 1ST										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 900/976	UEPGB	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
BLOCKING (GA) – ADD'L 2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCH	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (GA) – 1ST	OLI OII	107	10/	Ψ22.14	1471	14/1	14/1	1473	177	14/1
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCH	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (GA) – ADD'L										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPRJ	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
BLOCKING (GA, KY, MS) – 1ST										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPRJ	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
BLOCKING (GA, KY, MS) – ADD'L		L				l			l	

CAMPE CONTRACTOR OF THE CONTRACTOR CONTENTS AND				1 000 11						
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCQ	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (GA) – 1ST	LIEDOO			045.05					N. A	110
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCQ	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (GA) – ADD'L										
1.2.2.1 DISCONNECT NRCs (APPLICABLE IN GA AS SPECIFIED II	u									
Note 8)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING (GA) – 1ST	UEPGC	NA	NA	\$8.45	NA	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING (GA) – ADD'L	UEPGC	NA	NA	\$3.91	NA	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEP2G	NA	NA	\$8.45	NA	NA	NA	NA	NA	NA
BLOCKING: 011, 900/976, 1+DDD (GA) – 1ST	OLI 20	INA	INA	ψ0.45	INA	INA	INA	INA	INA	INA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEP2G	NA	NA	\$3.91	NA	NA	NA	NA	NA	NA
BLOCKING: 011, 900/976, 1+DDD (GA) – ADD'L	02.20			ψο.σ.						
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPGA	NA	NA	\$8.45	NA	NA	NA	NA	NA	NA
BLOCKING (GA) – 1ST										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPGA	NA	NA	\$3.91	NA	NA	NA	NA	NA	NA
BLOCKING (GA) – ADD'L										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 900/976	UEPGB	NA	NA	\$8.45	NA	NA	NA	NA	NA	NA
BLOCKING (GA) – 1ST										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 900/976	UEPGB	NA	NA	\$3.91	NA	NA	NA	NA	NA	NA
BLOCKING (GA) – ADD'L										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCH	NA	NA	\$8.45	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (GA) – 1ST	l									
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCH	NA	NA	\$3.91	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (GA) – ADD'L	LIEDD I		110	<b>#0.45</b>	114	110	111	N 1 A	N1.0	110
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPRJ	NA	NA	\$8.45	NA	NA	NA	NA	NA	NA
BLOCKING (GA, KY, MS) – 1ST	LIEDDI	NIA.	NIA	<b>#0.04</b>	A I A	NI A	NIA.	NIA	NI A	NIA
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011 BLOCKING (GA, KY, MS) – ADD'L	UEPRJ	NA	NA	\$3.91	NA	NA	NA	NA	NA	NA
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCQ	NA	NA	\$8.45	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (GA) – 1ST	ULFCQ	INA	INA	φ0.45	INA	INA	INA	INA	INA	INA
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCQ	NA	NA	\$3.91	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (GA) – ADD'L	OLI OQ	14/-1	INA	ψ5.51	INA	14/3	IVA	11/3	INA	IVA
BEGGRANG. 300/070, 11888, 0111, 1118 EGGRE (GR) TRBBE										
1.2.2.2 ADDITIONAL NRCs (APPLICABLE IN GA AS SPECIFIED IN										
Note 8)										
SUBSEQUENT ACTIVITY (Note 5)	USAS2	NA	NA	\$10.00	NA	NA	NA	NA	NA	NA
`										
1.2.2.3 OSS NRCs (APPLICABLE IN GA AS SPECIFIED IN Note 8)										
(Note 9)										
OSS LSR CHARGE, ELECTRONIC, PER LSR RECEIVED FROM THE	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
CLEC BY ONE OF THE OSS INTERACTIVE INTERFACES										
INCREMENTAL MANUAL SERVICE ORDER CHARGE – 1ST	SOMAN	NA	NA	\$37.06	NA	NA	NA	NA	NA	NA
INCREMENTAL MANUAL SERVICE ORDER CHARGE – ADD'L	SOMAN	NA	NA	\$8.19	NA	NA	NA	NA	NA	NA
ELECTRONIC SERVICE ORDER DISCONNECT	SOMEC	NA	NA	NA	NA	NA	NA	NA	NA	NA
INCREMENTAL MANUAL SERVICE ORDER DISCONNECT	SOMAN	NA	NA	\$11.17	NA	NA	NA	NA	NA	NA

2. UNE COIN PORT/LOOP COMBINATION:	USOC	AL	FL Orlando, Ft. Lauderdale, Miami	<b>GA</b> Atlanta	КҮ	<b>LA</b> New Orleans	MS	NC Greensboro- Winston Salem- Highpoint / Charlotte- Gastonia- Rock Hill	SC	TN Nashville
2.1 RECURRING CHARGES (RCs), By Month										
2.1.1 UNE COIN PORTS (Note 10) (Note 11) (Note 12)										
2-WIRE COIN 2-WAY WITHOUT OPERATOR SCREENING AND WITHOUT BLOCKING (AL, KY, LA, MS)	UEPRF	\$21.00	NA	NA	\$14.00	\$17.50	\$24.15	NA	NA	NA
2-WIRE COIN 2-WAY WITHOUT OPERATOR SCREENING AND WITHOUT BLOCKING (NC)	UEPND	NA	NA	NA	NA	NA	NA	\$17.50	NA	NA
2-WIRE COIN 2-WAY WITHOUT OPERATOR SCREENING AND WITHOUT BLOCKING (SC)	UEPSD	NA	NA	NA	NA	NA	NA	NA	\$14.00	NA
2-WIRE COIN 2-WAY WITHOUT OPERATOR SCREENING AND WITHOUT BLOCKING (TN)	UEPTB	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-WIRE COIN 2-WAY WITHOUT OPERATOR SCREENING AND WITHOUT BLOCKING; WITH DIALING PARITY (MS)	UEPMC	NA	NA	NA	NA	NA	\$24.15	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING (AL, KY)	UEPRE	\$21.00	NA	NA	\$14.00	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING (GA)	UEPGC	NA	NA	\$17.15	NA	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING (NC)	UEPNC	NA	NA	NA	NA	NA	NA	\$17.50	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND BLOCKING: 011, 900/976, 1+DDD (AL, KY, LA, MS, SC)	UEPRA	\$21.00	NA	NA	\$14.00	\$17.50	\$24.15	NA	\$14.00	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND BLOCKING: 011, 900/976, 1+DDD (FL)	UEP2F	NA	\$14.00	NA	NA	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND BLOCKING: 011, 900/976, 1+DDD (GA)	UEP2G	NA	NA	\$17.15	NA	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND BLOCKING: 011, 900/976, 1+DDD; WITH DIALING PARITY (MS)	UEPMA	NA	NA	NA	NA	NA	\$24.15	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND BLOCKING: 011, 900/976, 1+DDD (SC)	UEPSA	NA	NA	NA	NA	NA	NA	NA	\$14.00	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND BLOCKING: 011, 900/976, 1+DDD (NC, TN)	UEPRP	NA	NA	NA	NA	NA	NA	\$17.50	NA	\$14.00
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011 BLOCKING (AL, LA, MS)	UEPRB	\$21.00	NA	NA	NA	\$17.50	\$24.15	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011 BLOCKING (FL)	UEPFA	NA	\$14.00	NA	NA	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011 BLOCKING (GA)	UEPGA	NA	NA	\$17.15	NA	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011 BLOCKING (KY)	UEPKA	NA	NA	NA	\$14.00	NA	NA	NA	NA	NA
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011 BLOCKING (SC)	UEPSH	NA	NA	NA	NA	NA	NA	NA	\$14.00	NA

2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPNB	NA	NA	NA	NA	NA	NA	\$17.50	NA	NA
BLOCKING (NC)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPTA	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
BLOCKING (TN)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPMB	NA	NA	NA	NA	NA	\$24.15	NA	NA	NA
BLOCKING; WITH DIALING PARITY (MS)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 011	UEPSC	NA	NA	NA	NA	NA	NA	NA	\$14.00	NA
BLOCKING; WITH DIALING PARITY (SC)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND 900/976	UEPGB	NA	NA	\$17.15	NA	NA	NA	NA	NA	NA
BLOCKING (GA)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCD	\$21.00	NA	NA	\$14.00	\$17.50	\$24.15	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (AL, KY, LA, MS)	021 02	Ψ21.00	1473	1473	Ψ14.00	Ψ17.00	Ψ24.10	1473	1 17	147
			<b>**</b> * * * * * * * * * * * * * * * * * *							
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCG	NA	\$14.00	NA	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (FL)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCH	NA	NA	\$17.15	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (GA)				·						
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCC	NA	NA	NA	NA	NA	NA	NA	\$14.00	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (SC)									,	
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCA	NA	NA	NA	NA	NA	NA	\$17.50	NA	\$14.00
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (NC, TN)	02.07.							ψσ		Ψσσ
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCE	NA	NA	NA	NA	NA	NA	NA	\$14.00	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL; ENHANCED	OLI OL	1473	1473	1473	1473	1471	1471	1473	Ψ14.00	147
CALLING OPT 3YV (SC)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCF	NA	NA	NA	NA	NA	NA	NA	\$14.00	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL; ENHANCED	OLFCI	INA	INA	INA	INA	INA	INA	INA	\$14.00	INA
CALLING OPT AP7 (SC)										
2-WIRE COIN 2-WAY WITH OPERATOR SCREENING AND	UEPCJ	NA	NA	NA	NA	NIA	<b>CO445</b>	NA	NA	NIA
	UEPCJ	NA NA	NA	INA	INA	NA	\$24.15	NA	NA NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL; WITH DIALING										
PARITY (MS)					<b>*</b>	<b>A</b> 1 = = 0	<b>***</b>			
2-WIRE COIN OUTWARD WITHOUT BLOCKING AND WITHOUT	UEPRN	NA	NA	NA	\$14.00	\$17.50	\$24.15	NA	NA	NA
OPERATOR SCREENING (KY, LA, MS)										
2-WIRE COIN OUTWARD WITHOUT BLOCKING AND WITHOUT	UEPSG	NA	NA	NA	NA	NA	NA	NA	\$14.00	NA
OPERATOR SCREENING (SC)										
2-WIRE COIN OUTWARD WITHOUT BLOCKING AND WITHOUT	UEPME	NA	NA	NA	NA	NA	\$24.15	NA	NA	NA
OPERATOR SCREENING; WITH DIALING PARITY (MS)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPRJ	NA	NA	\$17.15	\$14.00	NA	\$24.15	NA	NA	NA
BLOCKING (GA, KY, MS)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPLA	NA	NA	NA	NA	\$17.50	NA	NA	NA	NA
BLOCKING (LA)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPSF	NA	NA	NA	NA	NA	NA	NA	\$14.00	NA
BLOCKING (SC)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPRK	\$21.00	\$14.00	NA	NA	NA	NA	NA	NA	NA
BLOCKING (AL, FL)								",	""	""
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPNE	NA	NA	NA	NA	NA	NA	\$17.50	NA	NA
BLOCKING (NC)		'*'	''''	1471				ψ11.00	''''	
DECOMING (NO)	<u> </u>	<u> </u>		1	1		L			1

2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPTC	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
BLOCKING (TN)										,
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPOF	NA	\$14.00	NA	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+ (FL)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND 011	UEPMD	NA	NA	NA	NA	NA	\$24.15	NA	NA	NA
BLOCKING; WITH DIALING PARITY (MS)							·			
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPRH	\$21.00	NA	NA	\$14.00	\$17.50	\$24.15	NA	NA	NA
BLOCKING: 011, 900/976, 1+DDD (AL, KY, LA, MS)					·	·	·			
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPSJ	NA	NA	NA	NA	NA	NA	NA	\$14.00	NA
BLOCKING: 011, 900/976, 1+DDD (SC)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCN	\$21.00	NA	NA	\$14.00	NA	\$24.15	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (AL, KY, LA, MS)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCQ	NA	\$14.00	\$17.15	NA	NA	NA	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (FL, GA)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCM	NA	NA	NA	NA	NA	NA	NA	\$14.00	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (SC)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCL	NA	NA	NA	NA	NA	NA	\$17.50	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (NC)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPOT	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL (TN)										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCP	NA	NA	NA	NA	NA	NA	NA	\$14.00	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL; WITH ENHANCED										
2-WIRE COIN OUTWARD WITH OPERATOR SCREENING AND	UEPCS	NA	NA	NA	NA	NA	\$24.15	NA	NA	NA
BLOCKING: 900/976, 1+DDD, 011+, AND LOCAL; WITH DAILING										
2.1.2 UNE 2-WIRE VOICE GRADE LOOPS (RCs)										
RC – 2-WIRE VOICE GRADE LOOP – STATEWIDE	UEPLX	NA	NA	NA	NA	NA	NA	\$14.18	NA	NA
RC – 2-WIRE VOICE GRADE LOOP – ZONE 1 (Note 6)	UEPLX	\$14.35	\$14.90	\$10.80	\$14.79	\$14.05	\$14.59	NA	\$17.02	\$12.48
RC – 2-WIRE VOICE GRADE LOOP – ZONE 2 (Note 6)	UEPLX	\$23.31	\$18.51	\$12.47	\$27.68	\$24.14	\$19.33	NA	\$25.66	\$14.42
RC – 2-WIRE VOICE GRADE LOOP – ZONE 3 (Note 6)	UEPLX	\$42.24	\$24.25	\$19.83	\$47.78	\$49.30	\$27.63	NA	\$33.99	\$21.77
RC – 2-WIRE VOICE GRADE LOOP – ZONE 4 (Note 6)	UEPLX	NA	NA	NA	NA	NA	\$36.47	NA	NA	NA
2.1.3 UNE COIN PORT/LOOP COMBINATION (RCs)										
RC – 2-WIRE VOICE GRADE LOOP WITH APPLICABLE UNE COIN	Note 4	NA	NA	NA	NA	NA	NA	\$31.68	NA	NA
PORT – STATEWIDE										
RC – 2-WIRE VOICE GRADE LOOP WITH APPLICABLE UNE COIN	Note 4	\$35.35	\$28.90	\$27.95	\$28.79	\$31.55	\$38.74	NA	\$31.02	\$26.48
PORT – ZONE 1										
RC – 2-WIRE VOICE GRADE LOOP WITH APPLICABLE UNE COIN	Note 4	\$44.31	\$32.51	\$29.62	\$41.68	\$41.64	\$43.48	NA	\$39.66	\$28.42
PORT – ZONE 2										
RC – 2-WIRE VOICE GRADE LOOP WITH APPLICABLE UNE COIN	Note 4	\$63.24	\$38.25	\$36.98	\$61.78	\$66.80	\$51.78	NA	\$47.99	\$35.77
PORT – ZONE 3										
RC – 2-WIRE VOICE GRADE LOOP WITH APPLICABLE UNE COIN	Note 4	NA	NA	NA	NA	NA	\$60.62	NA	NA	NA
PORT – ZONE 4			<u> </u>	<u> </u>				<u> </u>		
2.1.4 ADDITIONAL UNE COIN PORT/LOOP (RCs)										

2.2 NONRECURRING CHARGES (NRC)										
2.2.1 CURRENTLY COMBINED UNE COIN PORT/LOOP COMBO										
(NRCs) (Note 9)										
CONVERSION CHARGE - SWITCH AS IS - 1ST	USAC2	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50
CONVERSION CHARGE - SWITCH AS IS - ADD'L	USAC2	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50
CONVERSION CHARGE - SWITCH WITH CHANGE - 1ST	USACC	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50
CONVERSION CHARGE – SWITCH WITH CHANGE – ADD'L	USACC	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50
2.2.2 NOT CURRENTLY COMBINED UNE COIN PORT/LOOP										
COMBO (NRCs)										
INSTALLATION CHARGE – 1ST	UEPRF	\$90.00	NA	NA	\$90.00	\$90.00	\$90.00	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPRF	\$50.00	NA	NA	\$50.00	\$50.00	\$50.00	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPND	NA	NA	NA	NA	NA	NA	\$90.00	NA	NA
INSTALLATION CHARGE – ADD'L	UEPND	NA	NA	NA	NA	NA	NA	\$50.00	NA	NA
INSTALLATION CHARGE – 1ST	UEPSD	NA	NA	NA	NA	NA	NA	NA	\$90.00	NA
INSTALLATION CHARGE – ADD'L	UEPSD	NA	NA	NA	NA	NA	NA	NA	\$50.00	NA
INSTALLATION CHARGE – 1ST	UEPTB	NA	NA	NA	NA	NA	NA	NA	NA	\$90.00
INSTALLATION CHARGE – ADD'L	UEPTB	NA	NA	NA	NA	NA	NA	NA	NA	\$50.00
INSTALLATION CHARGE – 1ST	UEPMC	NA	NA	NA	NA	NA	\$90.00	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPMC	NA	NA	NA	NA	NA	\$50.00	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPRE	\$90.00	NA	NA	\$90.00	NA	NA	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPRE	\$50.00	NA	NA	\$50.00	NA	NA	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPGC	NA	NA	\$90.00	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPGC	NA	NA	\$50.00	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPNC	NA	NA	NA	NA	NA	NA	\$90.00	NA	NA
INSTALLATION CHARGE – ADD'L	UEPNC	NA	NA	NA	NA	NA	NA	\$50.00	NA	NA
INSTALLATION CHARGE – 1ST	UEPRA	\$90.00	NA	NA	\$90.00	\$90.00	\$90.00	NA	\$90.00	NA
INSTALLATION CHARGE – ADD'L	UEPRA	\$50.00	NA	NA	\$50.00	\$50.00	\$50.00	NA	\$50.00	NA
INSTALLATION CHARGE – 1ST	UEP2F	NA	\$90.00	NA	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEP2F	NA	\$50.00	NA	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEP2G	NA	NA	\$90.00	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEP2G	NA	NA	\$50.00	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPMA	NA	NA	NA	NA	NA	\$90.00	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPMA	NA	NA	NA	NA	NA	\$50.00	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPSA	NA	NA	NA	NA	NA	NA	NA	\$90.00	NA
INSTALLATION CHARGE – ADD'L	UEPSA	NA	NA	NA	NA	NA	NA	NA	\$50.00	NA
INSTALLATION CHARGE – 1ST	UEPRP	NA	NA	NA	NA	NA	NA	\$90.00	NA	\$90.00
INSTALLATION CHARGE – ADD'L	UEPRP	NA	NA	NA	NA	NA	NA	\$50.00	NA	\$50.00
INSTALLATION CHARGE – 1ST	UEPRB	\$90.00	NA	NA	NA	\$90.00	\$90.00	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPRB	\$50.00	NA	NA	NA	\$50.00	\$50.00	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPFA	NA	\$90.00	NA	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPFA	NA	\$50.00	NA	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPGA	NA	NA	\$90.00	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPGA	NA	NA	\$50.00	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPKA	NA	NA	NA	\$90.00	NA	NA	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPKA	NA	NA	NA	\$50.00	NA	NA	NA	NA	NA

INSTALLATION CHARGE – 1ST	UEPSH	NA	NA	NA	NA	NA	NA	NA	\$90.00	NA
INSTALLATION CHARGE – ADD'L	UEPSH	NA	NA	NA	NA	NA	NA	NA	\$50.00	NA
INSTALLATION CHARGE – 1ST	UEPNB	NA	NA	NA	NA	NA	NA	\$90.00	NA	NA
INSTALLATION CHARGE – ADD'L	UEPNB	NA	NA	NA	NA	NA	NA	\$50.00	NA	NA
INSTALLATION CHARGE – 1ST	UEPTA	NA	NA	NA	NA	NA	NA	NA	NA	\$90.00
INSTALLATION CHARGE – ADD'L	UEPTA	NA	NA	NA	NA	NA	NA	NA	NA	\$50.00
INSTALLATION CHARGE – 1ST	UEPMB	NA	NA	NA	NA	NA	\$90.00	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPMB	NA	NA	NA	NA	NA	\$50.00	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPSC	NA	NA	NA	NA	NA	NA	NA	\$90.00	NA
INSTALLATION CHARGE – ADD'L	UEPSC	NA	NA	NA	NA	NA	NA	NA	\$50.00	NA
INSTALLATION CHARGE – 1ST	UEPGB	NA	NA	\$90.00	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPGB	NA	NA	\$50.00	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPCD	\$90.00	NA	NA	\$90.00	\$90.00	\$90.00	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPCD	\$50.00	NA	NA	\$50.00	\$50.00	\$50.00	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPCG	NA	\$90.00	NA	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPCG	NA	\$50.00	NA	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPCH	NA	NA	\$90.00	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPCH	NA	NA	\$50.00	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPCC	NA	NA	NA	NA	NA	NA	NA	\$90.00	NA
INSTALLATION CHARGE – ADD'L	UEPCC	NA	NA	NA	NA	NA	NA	NA	\$50.00	NA
INSTALLATION CHARGE – 1ST	UEPCA	NA	NA	NA	NA	NA	NA	\$90.00	NA	\$90.00
INSTALLATION CHARGE – ADD'L	UEPCA	NA	NA	NA	NA	NA	NA	\$50.00	NA	\$50.00
INSTALLATION CHARGE – 1ST	UEPCE	NA	NA	NA	NA	NA	NA	NA	\$90.00	NA
INSTALLATION CHARGE – ADD'L	UEPCE	NA	NA	NA	NA	NA	NA	NA	\$50.00	NA
INSTALLATION CHARGE – 1ST	UEPCF	NA	NA	NA	NA	NA	NA	NA	\$90.00	NA
INSTALLATION CHARGE – ADD'L	UEPCF	NA	NA	NA	NA	NA	NA	NA	\$50.00	NA
INSTALLATION CHARGE – 1ST	UEPCJ	NA	NA	NA	NA	NA	\$90.00	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPCJ	NA	NA	NA	NA	NA	\$50.00	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPRN	NA	NA	NA	\$90.00	\$90.00	\$90.00	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPRN	NA	NA	NA	\$50.00	\$50.00	\$50.00	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPSG	NA	NA	NA	NA	NA	NA	NA	\$90.00	NA
INSTALLATION CHARGE – ADD'L	UEPSG	NA	NA	NA	NA	NA	NA	NA	\$50.00	NA
INSTALLATION CHARGE – 1ST	UEPME	NA	NA	NA	NA	NA	\$90.00	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPME	NA	NA	NA	NA	NA	\$50.00	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPRJ	NA	NA	\$90.00	\$90.00	NA	\$90.00	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPRJ	NA	NA	\$50.00	\$50.00	NA	\$50.00	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPLA	NA	NA	NA	NA	\$90.00	NA	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPLA	NA	NA	NA	NA	\$50.00	NA	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPSF	NA	NA	NA	NA	NA	NA	NA	\$90.00	NA
INSTALLATION CHARGE – ADD'L	UEPSF	NA	NA	NA	NA	NA	NA	NA	\$50.00	NA
INSTALLATION CHARGE – 1ST	UEPRK	\$90.00	\$90.00	NA	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPRK	\$50.00	\$50.00	NA	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPNE	NA	NA	NA	NA	NA	NA	\$90.00	NA	NA
INSTALLATION CHARGE – ADD'L	UEPNE	NA	NA	NA	NA	NA	NA	\$50.00	NA	NA
INSTALLATION CHARGE – 1ST	UEPTC	NA	NA	NA	NA	NA	NA	NA	NA	\$90.00
INSTALLATION CHARGE – ADD'L	UEPTC	NA	NA	NA	NA	NA	NA	NA	NA	\$50.00

INICTALLATION CHARGE ACT	LIEDOE	1 A I A	<b>\$00.00</b>	1 114		N.1.A		A.1.A	A.I.A.	N.1.A
INSTALLATION CHARGE – 1ST	UEPOF	NA	\$90.00	NA	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPOF	NA	\$50.00	NA	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPMD	NA	NA	NA	NA	NA	\$90.00	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPMD	NA	NA	NA	NA	NA	\$50.00	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPRH	\$90.00	NA	NA	\$90.00	\$90.00	\$90.00	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPRH	\$50.00	NA	NA	\$50.00	\$50.00	\$50.00	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPSJ	NA	NA	NA	NA	NA	NA	NA	\$90.00	NA
INSTALLATION CHARGE – ADD'L	UEPSJ	NA	NA	NA	NA	NA	NA	NA	\$50.00	NA
INSTALLATION CHARGE – 1ST	UEPCN	\$90.00	NA	NA	\$90.00	\$90.00	\$90.00	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPCN	\$50.00	NA	NA	\$50.00	\$50.00	\$50.00	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPCQ	NA	\$90.00	\$90.00	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPCQ	NA	\$50.00	\$50.00	NA	NA	NA	NA	NA	NA
INSTALLATION CHARGE – 1ST	UEPCM	NA	NA	NA	NA	NA	NA	NA	\$90.00	NA
INSTALLATION CHARGE – ADD'L	UEPCM	NA	NA	NA	NA	NA	NA	NA	\$50.00	NA
INSTALLATION CHARGE – 1ST	UEPCL	NA	NA	NA	NA	NA	NA	\$90.00	NA	NA
INSTALLATION CHARGE – ADD'L	UEPCL	NA	NA	NA	NA	NA	NA	\$50.00	NA	NA
INSTALLATION CHARGE – 1ST	UEPOT	NA	NA	NA	NA	NA	NA	NA	NA	\$90.00
INSTALLATION CHARGE – ADD'L	UEPOT	NA	NA	NA	NA	NA	NA	NA	NA	\$50.00
INSTALLATION CHARGE – 1ST	UEPCP	NA	NA	NA	NA	NA	NA	NA	\$90.00	NA
INSTALLATION CHARGE – ADD'L	UEPCP	NA	NA	NA	NA	NA	NA	NA	\$50.00	NA
INSTALLATION CHARGE – 1ST	UEPCS	NA	NA	NA	NA	NA	\$90.00	NA	NA	NA
INSTALLATION CHARGE – ADD'L	UEPCS	NA	NA	NA	NA	NA	\$50.00	NA	NA	NA
							·			
2.2.3 ADDITIONAL NRCs (Note 9)										
SUBSEQUENT ACTIVITY (Note 5)	USAS2	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
,										
2.2.4 OSS NRCs (Note 9)										
OSS LSR CHARGE, ELECTRONIC, PER LSR RECEIVED FROM THE	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
CLEC BY ONE OF THE OSS INTERACTIVE INTERFACES										
INCREMENTAL MANUAL SERVICE ORDER CHARGE – 1ST	SOMAN	\$40.71	\$21.56	\$33.67	\$19.99	\$31.92	\$43.52	\$40.18	\$43.19	\$30.89
INCREMENTAL MANUAL SERVICE ORDER CHARGE – ADD'L	SOMAN	\$9.58	\$21.56	\$7.88	\$19.99	\$7.32	\$9.99	\$9.45	\$9.91	\$7.03
SUBSEQUENT DATABASE UPDATE - ELECTRONIC		\$1.44	NA	NA	NA	\$2.11	\$2.87	\$1.42	\$0.71	\$0.76
SUBSEQUENT DATABASE UPDATE – MANUAL SERVICE ORDER		\$8.25	NA	NA	NA	\$5.12	\$6.88	\$10.27	\$8.91	\$7.97
ELECTRONIC SERVICE ORDER DISCONNECT	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
INCREMENTAL MANUAL SERVICE ORDER DISCONNECT	SOMAN	\$20.00	\$3.84	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00
3. ADDITIONAL USOCs										
(APPLICABLE IN ALL STATES) (Note 9)										
UNBUNDLED CONTACT NAME	UNECN	NA	NA	NA	NA	NA	NA	NA	NA	NA
l .										

#### NOTES:

- 1. The UNE Coin Port rate within this section includes all available features in GA, KY and TN.
- 2. Rates within this section apply for "Currently Combined" (and "Not Currently Combined" as noted in GA) UNE Coin Port/Loop Combinations ordered by the CLEC within all MSAs of th states listed to serve end users with less than four (4) DS0 equivalent lines; and for all MSAs of the states listed, except Orlando, Ft. Lauderdale, Miami, Atlanta, Greensboro-Winston Salem-Highpoint / Charlotte-Gastonia-Rock Hill, Nashville and New Orleans, to serve end users with four (4) or more DS0 equivalent line:
- 3. As clarification, the term Dialing Parity refers to the BellSouth local calling plans that BellSouth must provide to comply with local dialing parity. The CLEC may request such a local calling scope with its UNE port/loop combination by selecting the appropriate USOC.

Attachment 2 Exhibit C Rates

- 4. There is not a unique UNE Coin Port/Loop Combination USOC. The service is ordered by submitting the UNE Coin Port USOC.
- 5. Subsequent Activity is charged for service order activity after the line has been established as a UNE Combination. This charge is in addition to other applicable NRCs, such as OSS charges and installation charges.
- 6. This Amendment contains Deaveraged Loop Rates by Zone. Until approximately December 31, 2000 or until such time that BellSouth billing systems have been developed to handle the new zone rate structure, BellSouth will bill at the Zone 1 Deaveraged Loop rate level only. After December 31, 2000 or such time that the billing systems have been developed to handle the new zone rate structure, BellSouth will begin billing pursuant to CLEC-1's interconnection agreement.
- 7. In Georgia the UNE Coin Port Recurring Rates within this section will apply for "Currently Combined" as well as "Not Currently Combined" Port/Loop Combinations except for the Atlanta MSA if the CLEC is serving an end user with four or more DS0 equivalent lines.
- 8. The nonrecurring charges in this category apply to "Not Currently Combined" UNE Coin Port/Loop Combinations in GA only, except for the Atlanta MSA if the CLEC is serving an end user with four or more DS0 equivalent lines
- 9. The USOCs listed within this category will appear on the CLEC's bill with the charges listed but are not required on the Local Service Request (LSR) when ordering.
- 10. Rates within this section apply (1) if the CLEC is ordering a UNE Coin Port/loop Combination that is "Currently Combined" or "Not Currently Combined" to serve an end user with four or more DS0 equivalent lines in the Orlando, Ft. Lauderdale, Miami, Atlanta, Greensboro-Winston Salem-Highpoint / Charlotte-Gastonia-Rock Hill, Nashville or New Orleans MSAs or (2) if the CLEC is ordering a UNE Coin Port/loop Combination that is "Not Currently Combined" to serve an end user in any MSA in AL, FL, KY, LA, MS, NC, SC or TN.
- 11. The UNE Coin Port rate within this section includes all available features.
- 12. BellSouth currently is developing the billing capability to bill unbundled recurring port rates for this section 2.1.1. In the interim, BellSouth shall bill the recurring unbundled port rates listed in 1.1.1 of this Exhibit 1 and shall true-up the billing differences once the billing capability is developed.

## AMENDMENT TO INTERCONNECTION AGREEMENT BETWEEN BELLSOUTH TELECOMMUNICATIONS, INC. AND

BUSINESS TELECOM, INC. DATED FEBRUARY 21, 2000

This Agreement (the "Amendment") is made and entered into between BellSouth Telecommunications, Inc. ("BellSouth") a Georgia corporation, and Business Telecom, Inc. d/b/a BTI, a North Carolina corporation.

WHEREAS, The Parties desire to amend that certain Interconnection Agreement between BellSouth and BTI dated February 21, 2000 (the "Interconnection Agreement") in order to incorporate rates established by the Tennessee Regulatory Authority ("TRA") in Docket Number 97-01262, on December 19, 2000, as amended by BellSouth's corrected submissions of January 31, 2001 and February 12, 2001;

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, BellSouth and BTI hereby covenant and agree as follows:

- 1. Those permanent rates established by the TRA in Docket No. 97-01262 for certain Unbundled Network Elements and Local Interconnection in Tennessee are as set forth in Exhibit 1-TN attached hereto and incorporated herein by this reference.
- 2. To the extent that any product or service set forth in Exhibit 2-TN corresponds to a product or service set forth in the Interconnection Agreement, all rate elements and rates associated with such product or service in the Interconnection Agreement are hereby deleted and replaced with the corresponding rates and rate elements in Exhibit 1-TN.
- 3. Any rate element and rate for products or services in the Interconnection Agreement that is not expressly replaced by the rates and rate elements set forth in Exhibit 1-TN as described in paragraph 2 above shall remain in full force and effect in accordance with the terms of the Interconnection Agreement.
- 4. To the extent BTI and BellSouth have not previously negotiated terms and conditions corresponding to any rate element set forth in Exhibit 1-TN, then any Order for such element shall be provisioned in accordance to the terms and conditions set forth in the Competitive Local Exchange Carrier Tariff for the State of Tennessee, incorporated herein by this reference.

- 5. The Parties agree that all of the other provisions of the Interconnection Agreement, dated February 21, 2000, shall remain in full force and effect.
- 6. The Parties further agree that either or both of the Parties is authorized to submit this Amendment to the Tennessee Regulatory Authority or other regulatory body having jurisdiction over the subject matter of this Amendment, for approval subject to Section 252(e) of the federal Telecommunications Act of 1996.

This Amendment is made effective upon the date that it is signed by both Parties.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the data indicated below.

BellSouth Telecommunications, Inc.	Business Telecom, Inc.
Signature on File	Signature on File
By: <u>Gregory R. Follensbee</u>	By: Anthony M. Copeland
Title: Senior Director	Title:Exec.VP/General Counsel_
Date:8/13/01	Date:August 7, 2001

								RATES					oss	RATES	
			Not in TRA												Incremental Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETWORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97- 01262	BCS	USOC				Nonre	curring	Svc Order Submitted Elec	Svc Order Submitted	Incremental Charge - Manual Svc Order vs.		Manual Svc Order vs.
							Nonr	ecurring	Disco	onnect	per LSR	Manually per LSR		Svc Order vs. Electronic-Add'l	Electronic-Disc 1st
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN
															<u> </u>
ODL	UF/EDOUF/CMDS/CNAM-Resale														<b></b>
															<b></b> '
	OPTIONAL DAILY USAGE FILE (ODUF)														
		OSS OLEC Daily Usage File													1
	ODUF: Recording, per message	Recording per message			N/A	\$0.0000044									<b></b> '
															1
		OSS OLEC Daily Usage File													1
	ODUF: Message Processing, per message	Message Distribution per message			N/A	\$0.0027366									1
		OSS OLEC Daily Usage File													ı
	ODUF: Message Processing, per Magnetic Tape														1
	provisioned	tape provisioned			N/A	\$52.75									
		OSS OLEC Daily Usage File Data													1
	ODUF: Data Transmission (CONNECT:DIRECT)	Transmission (Connect: Direct), per													1
	per message	message			N/A	\$0.0000339									<b>I</b>
															<b></b>
															1

Incremental Charge -Manual Svc Order vs. Electronic-Disc Add'l SOMAN

									RATES					oss	RATES		
																Incremental Charge -	Incremental Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS	Not in TRA Docket 97-	Zone	BCS	usoc						Svc Order	Svc Order	Incremental	Incremental	Manual Svc	Manual Svc
		STATED IN DOCKET 97-01262	01262							Nonre	ecurring	Submitted Elec	Submitted Manually per	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Order vs. Electronic-Disc	Order vs. Electronic-Disc
									curring		onnect	per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'l
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Th	ne "Zone" shown in the section for stand-alone loop	os or loops as part of a combination ref	e														
to	Geographically Deaveraged UNE Zones. To view		ne														i
	Designations by Central Office, ref http://www.interconnection.bellsouth.com/becon																i
		T						İ									
LINDIIN	NDLED EXCHANGE ACCESS LOOP																
ONDO	IDEED EXCITAINGE AGGESS EGGI																
2-	WIRE ANALOG VOICE GRADE LOOP																
	2-Wire Analog Voice Grade Loop - Service																
	Level 1- Zone 1	Service Level 1 - Zone 1		1	UEANL	UEAL2	\$13.19	\$31.99	\$20.02	\$10.65	\$1.41			\$20.35	\$10.54	\$13.32	l
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2		-	2	UEANL	UEAL2	£47.00	624.00	<b>#</b> 20.00	£40.05	C4 44			#20.2F	£40.54	£40.00	İ
	2-Wire Analog Voice Grade Loop - Service	Service Level 1 - Zone 2			UEAINL	UEALZ	\$17.23	\$31.99	\$20.02	\$10.65	\$1.41			\$20.35	\$10.54	\$13.32	
	Level 1- Zone 3	Service Level 1 - Zone 3		3	UEANL	UEAL2	\$22.53	\$31.99	\$20.02	\$10.65	\$1.41			\$20.35	\$10.54	\$13.32	İ
	Manual Order Coordination for UVL-SL1s	2-WAVGL - SL1 - Manual Order					*==:	******	*	*******	*****			4=0.00		******	
	(per loop)*	Coordination			UEANL	UEAMC		\$36.52	\$36.52	\$9.18	\$9.18						
		2-WAVGL - SL1 - Order															İ
	Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR) *	Conversion Time			UEANL	OCOSL		\$34.29	\$34.29								İ
	2-Wire Analog Voice Grade Loop - Service	Conversion Time			ULANL	OCOSL		\$34.29	\$34.29								ĺ
	Level 2 w/Loop or Ground Start Signaling -	2-Wire Analog Voice Grade Loop															İ
	Zone 1	Service Level 2 - Zone 1		1	UEA	UEAL2	\$16.56	\$75.06	\$48.20	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	i
	2-Wire Analog Voice Grade Loop - Service																1
	Level 2 w/Loop or Ground Start Signaling -	2-Wire Analog Voice Grade Loop	-	_						***	00.			***	010 = 1		İ
	Zone 2  2-Wire Analog Voice Grade Loop - Service	Service Level 2 - Zone 2		2	UEA	UEAL2	\$21.63	\$75.06	\$48.20	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	
	Level 2 w/Loop or Ground Start Signaling -	2-Wire Analog Voice Grade Loop															İ
	Zone 3	Service Level 2 - Zone 3		3	UEA	UEAL2	\$28.28	\$75.06	\$48.20	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	İ
		2-WAVGL - SL2 - Order															
	Order Coordination for Specified Conversion																İ
	Time (per LSR)	Conversion Time			UEA	OCOSL		\$34.29	\$34.29								
	2-Wire Analog Voice Grade Loop - Service	2-Wire Analog Voice Grade Loop															İ
	Level 2 w/Reverse Battery Signaling - Zone			1	UEA	UEAR2	\$16.56	\$75.06	\$48.20	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	İ
	, , ,																
	2-Wire Analog Voice Grade Loop - Service		-														İ
	Level 2 w/Reverse Battery Signaling - Zone	Service Level 2 - Zone 2		2	UEA	UEAR2	\$21.63	\$75.06	\$48.20	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	
	2-Wire Analog Voice Grade Loop - Service	2-Wire Analog Voice Grade Loop															İ
	Level 2 w/Reverse Battery Signaling - Zone			3	UEA	UEAR2	\$28.28	\$75.06	\$48.20	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	İ
		2-WAVGL - SL2 - Order															
	Order Coordination for Specified Conversion																İ
	Time (per LSR)	Conversion Time			UEA	OCOSL		\$34.29	\$34.29								
	WIRE ANALOG VOICE GRADE LOOP																
4-	WINE ANALOG VOICE GRADE LOUP	4-Wire Analog Voice Grade Loop										1					
	4-Wire Analog Voice Grade Loop - Zone 1	Zone 1		1	UEA	UEAL4	\$24.70	\$122.76	\$85.57	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	li
		4-Wire Analog Voice Grade Loop															
	4-Wire Analog Voice Grade Loop - Zone 2	Zone 2		2	UEA	UEAL4	\$32.25	\$122.76	\$85.57	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	l
	4 Wire Applea Voice Crede Lees 7 2	4-Wire Analog Voice Grade Loop		3	UEA	UEAL4	£40.47	£400.70	<b>605.53</b>	Ф <b>7</b> С ОБ	¢20.40			\$20.25	640.54	£40.00	İ
	4-Wire Analog Voice Grade Loop - Zone 3	Zone 3		3	UEA	UEAL4	\$42.17	\$122.76	\$85.57	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
	Order Coordination for Specified Conversion	4-Wire AVGL - Order Coordination	n														İ
	Time (per LSR)	for Specified Conversion Time	<u>L</u>		UEA	OCOSL		\$34.29	\$34.29	<u></u>			<u></u>	<u></u>			<u>.                                    </u>
2-	WIRE ISDN DIGITAL GRADE LOOP																<u> </u>
		2-Wire ISDN Digital Grade Loop -															
	2-Wire ISDN Digital Grade Loop - Zone 1	Zone 1		1	UDN	U1L2X	\$22.00	\$142.76	\$88.88	\$76.35	\$39.16	L		\$20.35	\$10.54	\$13.32	

	T		1					RATES			1		0881	RATES		
							T	RAILS	I 1			T	033	MILS	Incremental	Incremental
	UNDUNDLED METEODY ELEMENT 40	Not in TRA									1				Charge -	Charge -
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97-	Zone	BCS	USOC				Nonre	curring	Svc Order Submitted	Svc Order Submitted	Incremental Charge - Manual	Incremental Charge - Manual	Manual Svc Order vs.	Manual Svc Order vs.
		01262					1				Elec	Manually per	Svc Order vs.	Svc Order vs.	Electronic-Disc	Electronic-Dis
								curring		onnect	per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'l
	0.100		$\vdash$			Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Wire ISDN Digital Grade Loop - Zone 2	2-Wire ISDN Digital Grade Loop - Zone 2		2	UDN	U1L2X	\$29.02	\$142.76	\$88.88	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
2-Wife ISBN Digital Grade Loop - Zorie Z	2-Wire ISDN Digital Grade Loop -			ODIN	UILZX	\$29.02	\$142.76	Φ00.00	\$76.33	\$39.10			\$20.33	\$10.54	\$13.32	<b>_</b>
2-Wire ISDN Digital Grade Loop - Zone 3	Zone 3		3	UDN	U1L2X	\$37.95	\$142.76	\$88.88	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
	2-Wire ISDN Digital Grade Loop -		Ě			ψ07.00	ψ <u></u>	<b>\$60.00</b>	ψ. σ.σσ	ψουο			Ψ20.00	ψ.σ.σ.	ψ.σ.σ <u>2</u>	
Order Coordination For Specified Conversio																
Time (per LSR)	Conversion Time			UDN	OCOSL		\$34.29	\$34.29								
2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBE	R LINE (ADSL) COMPATIBLE LO	OP														
2-Wire ADSL Compatible Loop ncluding																
Man Svc Inquiry and Facility Reservation -	2-Wire ADSL Compatible Loop -															
	Zone 1		1	UAL	UAL2X	\$13.82	\$270.01	\$234.63	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
2-Wire ADSL Compatible Loop ncluding																
	2-Wire ADSL Compatible Loop -							_								
	Zone 2		2	UAL	UAL2X	\$18.05	\$270.01	\$234.63	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
2-Wire ADSL Compatible Loop ncluding	2 Wire ADSI Compatible !															
Man Svc Inquiry and Facility Reservation - Zone 3	2-Wire ADSL Compatible Loop - Zone 3		3	UAL	UAL2X	\$23.60	\$270.01	\$234.63	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
Zone 3	2-Wire ADSL Digital Grade Loop -		3	UAL	UALZA	<b>⊅∠ა.</b> σ∪	φ∠10.01	φ234.03	φ14.54	φ39.14			φ <b>∠</b> 0.35	φ10.5 <del>4</del>	\$13.32	
Order Coordination for Specified Conversion																
Time	Conversion Time			UAL	OCOSL		\$34.29	\$34.29								
								*								
2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER	R LINE (HDSL) COMPATIBLE LOO	F	t t													
2-Wire HDSL Compatible Loopincluding																
manual service inquiry and facility	2-Wire High Bit Rate DSL															
reservation - Zone 1	Compatible Loop - Zone 1		1	UHL	UHL2X	\$10.83	\$270.01	\$234.63	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
2-Wire HDSL Compatible Loop <u>including</u>																
	2-Wire High Bit Rate DSL							_								
reservation - Zone 2	Compatible Loop - Zone 2		2	UHL	UHL2X	\$14.15	\$270.01	\$234.63	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
2-Wire HDSL Compatible Loopincluding manual service inquiry and facility	2-Wire High Bit Rate DSL															
reservation - Zone 3	Compatible Loop - Zone 3		3	UHL	UHL2X	\$18.50	\$270.01	\$234.63	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
	2-Wire HDSL Loop - Order		٦	OTTL	OTILZX	\$10.50	\$270.01	\$234.03	\$74.54	φ55.14			\$20.55	\$10.54	ψ13.32	
Order Coordination for Specified Conversion																
Time	Conversion Time			UHL	OCOSL		\$34.29	\$34.29								
								-								
4-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER	R LINE (HDSL) COMPATIBLE LOO	P	ΙI								1					
4-Wire HDSL Compatible Loopincluding	- (		t				l				1					
manual service inquiry and facility	4-Wire HDSL Compatible Loop -															
	Zone 1		1	UHL	UHL4X	\$13.93	\$279.60	\$244.22	\$74.54	\$39.14	<u> </u>	L	\$20.35	\$10.54	\$13.32	
4-Wire HDSL Compatible Loop <u>including</u>						<u></u>								-		
manual service inquiry and facility	4-Wire HDSL Compatible Loop -		$\begin{bmatrix} & & 1 \end{bmatrix}$		l											
	Zone 2		2	UHL	UHL4X	\$18.20	\$279.60	\$244.22	\$74.54	\$39.14	1		\$20.35	\$10.54	\$13.32	
4-Wire HDSL Compatible Loopincluding	4 Wire HDSI Commercials I am															
manual service inquiry and facility reservation - Zone 3	4-Wire HDSL Compatible Loop - Zone 3		3	UHL	UHL4X	\$23.80	\$279.60	\$244.22	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
16361 VALIOIT - ZOTIE 3	4-Wire HDSL Loop Order		3	UIIL	OI IL4A	φ23.00	\$219.00	<b>Φ</b> 244.22	₹14.54	φ39.14			Φ∠0.33	φ10.5 <del>4</del>	\$13.32	-
Order Coordination for Specified Conversion																
Time	Conversion Time			UHL	OCOSL		\$34.29	\$34.29								
4-WIRE DS1 DIGITAL LOOP																
							l				1					
4-Wire DS1 Digital Loop - Zone 1	4-Wire DS1 Digital Loop - Zone 1		$\lfloor \ \rfloor$	USL	USLXX	\$57.73	\$313.08	\$219.72	\$96.86	\$40.45			\$18.98	\$8.43	\$11.95	\$0.00
T WITE DOT DIGITAL LOOP - ZOTIE T	- Trie Do i Digital Loop - Zolle I		<del>                                     </del>	UUL	OOLAA	φυ1.13	φυ 10.06	φ∠19.1∠	φ30.00	φ40.40	1	1	φ10.90	φυ.43	φ11. <del>3</del> 3	φυ.υυ
	4 Wiss DC4 Disital Lass 7sss 2		2	USL	USLXX	\$75.40	\$313.08	\$219.72	\$96.86	\$40.45			\$18.98	\$8.43	\$11.95	<b>60.00</b>
4 Wire DC1 Digital Leas Zone 2															DII.95	\$0.00
4-Wire DS1 Digital Loop - Zone 2	4-Wire DS1 Digital Loop - Zone 2			001	OOLAA	Ψ13.40	ψ313.00	Ψ213.72	ψ30.00	ψ+0.+3	1		Ψ10.00	ψ0.40	***************************************	· ·

									RATES			l		OSS	RATES		
			Not in TRA													Incremental Charge -	Incremental Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97- 01262	Zone	BCS	usoc				Nonre	curring	Svc Order Submitted Elec	Svc Order Submitted	Incremental Charge - Manual	Incremental Charge - Manual	Manual Svc Order vs.	Manual Svc Order vs.
									curring		nnect	per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'l	Electronic-Disc 1st	Electronic-Disc Add'l
$\vdash$		4-Wire DS1 Loop - Order					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Order Coordination for Specified Conversion																1
	Time	Conversion Time			USL	OCOSL		\$34.59	\$34.59								1
	1-WIRE 56 OR 64 KBPS DIGITAL GRADE LOO																
	4-Wire 56 Kbps Digital Grade Loop - Zone 1	4-Wire 56 or 64 Kbps Digital Grade		1	UDL	UDL56	\$31.10	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	1
	4-Wire 50 Rbps Digital Grade Loop - Zone 1	4-Wire 56 or 64 Kbps Digital Grade		<u> </u>	ODL	ODLOG	φ31.10	φ207.01	\$141.30	φ90.70	ψ <del>44</del> .10			φ20.33	\$10.54	\$13.32	
	4-Wire 56 Kbps Digital Grade Loop - Zone 2			2	UDL	UDL56	\$40.61	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
	4 Wiss 50 Khan Dinital Conda Lana 7ana 2	4-Wire 56 or 64 Kbps Digital Grade		3	UDL	UDL56	050.44	#007.04	<b>044400</b>	000.70	04440			#00.0F	040.54	640.00	1
	4-Wire 56 Kbps Digital Grade Loop - Zone 3	4-Wire 56/64 kbps Dig. GL - Order		3	UDL	UDLS6	\$53.11	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
	Order Coordination for Specified Conversion																1
	Time	Conversion Time			UDL	OCOSL		\$34.29	\$34.29								
	4-Wire 64 Kbps Digital Grade Loop - Zone 1	4-Wire 56 or 64 Kbps Digital Grade		1	UDL	UDL64	\$31.10	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	1
+	- Title of Rops Digital Glade Loop - Zolle 1	4-Wire 56 or 64 Kbps Digital Grade			UDL	UDL04	φυ1.10	φ∠01.01	φ141.30	φσ0.70	φ++.10			φ20.33	φ10.34	φ13.32	
	4-Wire 64 Kbps Digital Grade Loop - Zone 2			2	UDL	UDL64	\$40.61	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
	4-Wire 64 Kbps Digital Grade Loop - Zone 3	4-Wire 56 or 64 Kbps Digital Grade		3	UDL	UDL64	\$53.11	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	1
	4-Wife 64 Kbps Digital Grade Loop - Zoffe 3	4-Wire 56/64 kbps Dig. GL - Order		3	ODL	UDL64	\$53.11	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
	Order Coordination for Specified Conversion																1
	Time	Conversion Time			UDL	OCOSL		\$34.29	\$34.29								
CLID	LOOPS																
306-																	
	Sub-Loop Distribution																
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-Up		*		UEANL	USBSA		\$517.25	\$517.25					\$20.35	\$10.54	\$13.32	
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up		*		UEANL	USBSB		\$42.68	\$42.68					\$20.35	\$10.54	\$13.32	
	Sub-Loop Distribution Per 2-Wire Analog																
	Voice Grade Loop - Statewide Sub-Loop Distribution - Order Coordination	Loop Distribution - per 2-WAVGL		SW	UEANL	USBN2	\$10.02	\$148.84	\$112.34	\$73.14	\$36.65			\$20.35	\$10.54	\$13.32	$\vdash$
	per sub-loop pair		*		UEANL	USBMC		\$36.52									1
		Sub-Loop Distribution - Order															
	Sub-Loop Distribution - Order Coordination	Coordination for Specified			UEANL	OCOSL		©24.00									1
	for Specified Time Conversion, per LSR	Conversion Time			UEAINL	OCOSL		\$34.29									
<u> </u>	Sub-Loop Feeder							<b>-</b>									
	DS0 Set-up per Cross Box location - CLEC				UEA, UDN,U												1
	Distribution Facility set-up for access to Feeder)		*			USBFW		\$517.25	\$517.25					\$20.35	\$10.54	\$13.32	1
	7 coder y				OL,ODL	OOD! !!		ψ517.25	ψ317.23					Ψ20.33	\$10.5 <del>4</del>	ψ10.02	
					UEA,												1
	DS0 Set-up per Cross Box location - per 25 pair pasnel set-up <i>for access to Feeder</i> )		*		UDN,U CL,UDL	USBFX		\$42.68	\$42.68					\$20.35	\$10.54	\$13.32	1
$\vdash$	pair paorier set-up (or access to reeder)				JL,UDL	30Di X		φ42.00	φ42.00					φ20.33	φ10.5 <del>4</del>	φ13.32	
	Sub-Loop Feeder- Per 2-Wire Analog Voice																1
$\vdash$	Grade Ground-Start Loop - Statewide Sub-Loop Feeder- Per 2-Wire Analog Voice	Loop Feeder per 2-WAVGL		sw	UEA	USBFA	\$12.05	\$122.24	\$85.05	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	$\vdash$
	Grade Loop-Start Loop - Statewide	Loop Feeder per 2-WAVGL		sw	UEA	USBFB	\$12.05	\$122.24	\$85.05	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
	0.1.1																
	Sub-Loop Feeder - Per 2-Wire Analog Voice Grade Reverse Battery Loop - Statewide	Loop Feeder per 2-WAVGL		sw	UEA	USBFC	\$12.05	\$122.24	\$85.05	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	1
	5.335 Stored Ballory Loop Statewide	Sub-Loop Feeder - Order		5,,,	02,1	305.0	Ψ12.00	Ψ122.24	ψ00.00	ψ10.55	ψυσ.10			Ψ20.00	Ψ10.04	ψ10.02	
	Sub-Loop Feeder - Order Coordination for Specified Conversion Time, per LSR	Coordination for Specified Conversion Time			UEA	OCOSL		\$34.29									
$\Box$	epecined conversion rime, per LON				OLA	JUUGE		Ψ04.23	l	ı .		I	I				

						_	RATES					OSS	RATES		
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS D	ot in TRA ocket 97- Zon 01262	e BCS	usoc				Nonre	curring	Svc Order Submitted	Svc Order Submitted	Incremental Charge - Manual	Incremental Charge - Manual	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.
						Nome	curring	Dies	onnect	Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'l	Electronic-Disc 1st	Electronic-Disc Add'l
					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
					Nec	11130	Auu i	11131	Auu	COMILO	COMPLE	COMPLIA	JOHNA	JOHAN	COMPLE
						+									
Network Interface Device (NID)															
Network Interface Device Cross Connect	NID per 2-Wire Loop			UNDC2	\$1.15	\$0.74				\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
Network Interface Device Cross Connect	NID per 4-Wire Loop		UENTW	UNDC4	\$1.27	\$0.74				\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
UNBUNDLED LOOP CONCENTRATION															
	Loop Channelization system - DLC									1					
Loop Channelization System	(Inside CO)		ULC	ULCCS	\$307.07	\$307.34	\$74.37	\$4.18		\$3.50		\$20.35	\$10.54	\$13.32	
	CO Channel Interface - 2-Wire				400	***************************************		*****		70.00		4-0.00	4.0.0	*******	
CO Channel Interface - 2-Wire Voice Gra			ULC	ULCC2	\$1.20	\$9.57	\$9.52	\$8.66	\$8.60	\$3.50		\$20.35	\$10.54	\$13.32	
Unbundled Loop Concentration - Channe					<b>\$1.20</b>	ψ0.01	Ψ0.02	ψ0.00	ψ0.00	ψο.σσ		Ψ20.00	<b>\$</b> 10.01	ψ.σ.σ <u>2</u>	
Interface-2 Wire Voice-Loop Start or Grou	ind CO Channel Interface - 2-Wire														
Start	Voice Grade		UEA	ULCC2	\$1.20	\$9.57	\$9.52	\$8.66	\$8.60	\$3.50		\$20.35	\$10.54	\$13.32	
	1000000				<b>\$1.20</b>	ψ0.01	Ψ0.02	ψ0.00	ψ0.00	ψο.σσ		4_0.00	Ţ	ψ.σ.σ <u>2</u>	
UNDUNDLED SUD LOOD CONSENTS ATION (OU	TSIDE CO.			1		1		1			1				
UNBUNDLED SUB-LOOP CONCENTRATION (OU	Loop Concentration -					1		1							
	Channelization System (Outside														
Loop Concentration Channelization Sys			TBD	UCT8A	¢220.20	CC4 00	<b>#000 40</b>	\$207.00	<b>CCO 04</b>	<b>60.50</b>		<b>600.05</b>	640.54	640.00	
Loop Concentration - Channelization Sys	Loop Concentration - Remote		IBD	UCT8A	\$328.28	\$651.09	\$283.42	\$207.92	\$50.94	\$3.50		\$20.35	\$10.54	\$13.32	
Remote Channel Interface - 2-Wire Voice															
			TBD	ULCC2	00.00	00.40	<b>60.40</b>	04.74	<b>#4.70</b>	00.50		<b>600.05</b>	640.54	040.00	
Grade Loop	(Outside CO)		IBD	ULCC2	\$0.88	\$9.43	\$9.40	\$4.71	\$4.70	\$3.50		\$20.35	\$10.54	\$13.32	
Loop Concentration - Remote terminal	Loop Concentration - Remote				105										
Cabinet (Outside CO)	Terminal Cabinet (Outside CO)				ICB					\$3.50					
01	Loop Concentration - Remote														
Channel Interface - 2 Wire Voice-Loop St									4			***			
or Ground Start	(Outside CO)		TBD	ULCC2	\$0.88	\$9.43	\$9.40	\$4.71	\$4.70	\$3.50		\$20.35	\$10.54	\$13.32	
UNBUNDLED LOCAL EXCHANGE SWITCHING(P	ORTS)														
Exchange Ports															
Exolutinge Forts						+		<b>†</b>							
Exchange Ports - 2-Wire Analog Line Por	t- Evchange Ports - 2-Wire Analog														
Res.	Line Port (Res./Bus.)		HEDSE	UEPRL	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
						φ9.93	φ5.15	\$5.00	Ψ2.92	1		\$20.55	\$10.54	\$13.32	\$1.40
Note: Although the Port rate includes all avail		feature(s) v	will need t	be order	ed.)			1							
Exchange Ports - 2-Wire Analog Line Por		I	1	1		1 .		1 .	_			l .			_
with Caller ID - Res.	Line Port (Res./Bus.)		UEPSR	UEPRC	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Exchange Ports - 2-Wire Analog Line Por		I	1	1		1 .		1 .	_			l .			_
outgoing only - Res.	Line Port (Res./Bus.)		UEPSR	UEPRO	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Exchange Ports - 2-Wire VG unbundled T		I													
extended local dialing parity Port with Cal		1				1 .	_		_					_	
ID - Res.	Line Port (Res./Bus.)		UEPSR	UEPAQ	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Exchange Ports - 2-Wire VG unbundlec	_	1													
Tennessee Area Calling port with Caller II		I	1	1		1 .		1 .	_			l .			_
Res (F2R)	Line Port (Res./Bus.)		UEPSR	UEPAK	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Exchange Ports - 2-Wire VG unbundlec		1													
Tennessee Area Calling port with Caller II		I						I							
Res (TACER)	Line Port (Res./Bus.)		UEPSR	UEPAL	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Exchange Ports - 2-Wire VG unbundlec		]						1							
Tennessee Area Calling port with Caller II		1													
Res (TACSR)	Line Port (Res./Bus.)	L	UEPSR	UEPAM	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92	<u></u>		\$20.35	\$10.54	\$13.32	\$1.40
Exchange Ports - 2-Wire VG unbundlec															
Tennessee Area Calling port with Caller II		1						I							
Res (1MF2X)	Line Port (Res./Bus.)		1	UEPAN	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40

									RATES					oss	RATES		
			Not in TRA													Incremental Charge -	Incremental Charge -
UNBUNDLED NE	TWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97- 01262	Zone	BCS	usoc				Nonre	ecurring	Svc Order Submitted	Svc Order Submitted	Incremental Charge - Manual	Incremental Charge - Manual	Manual Svc Order vs.	Manual Svc Order vs.
			01262			-		Nonre	curring		onnect	Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs.	Electronic-Disc	Electronic-Disc
						-	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Exchange Ports - 2-W																	
Tennessee Area Calli Res (2MR)	ing port with Caller ID	- Exchange Ports - 2-Wire Analog Line Port (Res./Bus.)		- I.,	IEDED	UEPAO	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Res (ZIVIK)		Life Fort (Res./Bus.)		U	IEPOR	UEFAU	\$1.89	\$9.93	\$9.19	\$3.00	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Exchange Ports - 2-W	/ire VG unbundled res	s, Exchange Ports - 2-Wire Analog															
low usage line port wi	th Caller ID (LUM)	Line Port (Res./Bus.)			IEPSR	UEPAP	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Subsequent Activity			*	U	IEPSR	USASC		\$10.00	\$10.00								
All Available Vertical	Footures	[Exchange Ports includes all Applicable Features.]		- 1	IEDOD	UEPVF	\$0.00	\$0.00	\$0.00								
Exchange Ports - 2-W		Exchange Ports - 2-Wire Analog		-	ILFOR	OLF VI	φυ.υυ	φυ.υυ	φυ.υυ								
without Caller ID - Bu	s	Line Port (Res./Bus.)		U	JEPSB	UEPBL	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Exchange Ports - 2-W																	
- Bus.	ort with Caller+E484	ID Exchange Ports - 2-Wire Analog Line Port (Res./Bus.)		- 1	IEPSB	UEPBC	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Exchange Ports - 2-W	/ire Analog Line Por	Exchange Ports - 2-Wire Analog		Ť	LI OD	OLI DO	ψ1.03	ψ3.33	ψ3.13	ψ3.00	Ψ2.32			Ψ20.33	\$10.54	ψ13.32	Ψ1.40
outgoing only - Bus.		Line Port (Res./Bus.)		U	JEPSB	UEPBO	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Exchange Ports - 2-W																	
ID - Bus.	panty Port with Calle	Exchange Ports - 2-Wire Analog Line Port (Res./Bus.)		U	JEPSB	UEPAV	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Exhange Ports - 2-Wi	ire VG unbundlec	Exchange Ports - 2-Wire Analog		Ť		02.711	ψ1.00	ψ0.00	ψ0.10	ψ0.00	Ψ2.02			Ψ20.00	ψ10.04	Ψ10.02	Ψ1.40
incoming only port wit		Line Port (Res./Bus.)		U	JEPSB	UEPB1	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Exchange Ports - 2-W Bus 2-Way Area Calli		Exchange Ports - 2-Wire Analog															
Option - Bus (TACC1)		Line Port (Res./Bus.)		U	IEPSB	UEPAC	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Exchange Ports - 2-W	/ire VG unbundled TN						ψ1.00	ψ0.00	φοιτο	ψ0.00	<b>\$2.02</b>			Ψ20.00	<b>\$10.01</b>	ψ10.0 <u>2</u>	<b>V</b>
Bus 2-Way Area Calli		Exchange Ports - 2-Wire Analog		- I													
Option - Bus (TACC2 Exchange Ports-2-Wi		Line Port (Res./Bus.)		U	JEPSB	UEPAD	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Bus 2-Way Collierville		Exchange Ports - 2-Wire Analog															
Calling Port-Bus (B2F	-)	Line Port (Res./Bus.)		U	JEPSB	UEPAE	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Subsequent Activity			*	U	JEPSB	USASC		\$10.00	\$10.00								
All Available Vertical	F4	[Exchange Ports includes all Applicable Features.]		- I.,	IEDOD	UEPVF	\$0.00	00.00	<b>#</b> 0.00								
Exchange Ports - 4-W		Exchange Ports - 4-Wire Analog		- 10	IEPOD	UEPVF	\$0.00	\$0.00	\$0.00								
Grade Port		Voice Grade Port				UEP4A	\$8.27	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Subsequent Activity			*			USASC		\$10.00	\$10.00								
		[Exchange Ports includes all															
All Available Vertical	Features	Applicable Features.]				UEPVF	\$0.00	\$0.00	\$0.00								
Exchange Ports - 2-W	/ire DID Port	Exchange Ports - 2-Wire DID Port		- 1	JEPEX	UEPP2	\$8.97	\$47.75	\$47.01	\$9.21	\$8.47			\$20.35	\$10.54	\$13.32	\$1.40
Exchange Forts - 2-Vi	THE DID FOR	Exchange Forts - 2-wife bib Fort		Ť	LILX	OLITZ	ψ0.31	ψ41.13	Ψ47.01	ψ3.21	ψ0.47			Ψ20.33	\$10.54	ψ13.32	Ψ1.40
Subsequent Activity			*	U	JEPEX	USASC		\$10.00	\$10.00								
All Available Variant	Footures	[Exchange Ports includes all		<b>.</b>	IEDEV	UEPVF	<b>60.00</b>	<b>\$0.00</b>	<b>\$0.00</b>								
All Available Vertical Exchange Ports - DDI		Applicable Features.]	<del>                                     </del>	U	JEPEX	DEPVF	\$0.00	\$0.00	\$0.00			-					
Port with DID capabili		Exchange Ports - 4-Wire DID Port		U	IEPDD	UEPDD	\$35.74	\$75.93	\$38.15	\$8.77	\$8.04			\$20.35	\$10.54	\$13.32	\$1.40
Subsequent Activity			*	U	IEPDD	USACP		\$10.00	\$10.00								
Exchange Ports - 2-W	/ire ISDN Port (See	Exchange Ports - 2-Wire ISDN															
Notes below.)		Port				U1PMA	\$16.26	\$30.23	\$29.49	\$4.10	\$4.10			\$41.43	\$42.17	\$9.80	\$9.80
Subsequent Activity			*	U	EPDD	USASC		\$10.00	\$10.00								1
All Available Vertical	Footures	[Exchange Ports includes all Applicable Features.]			IEPDD	UEPVF	\$0.00	\$0.00	\$0.00								
NOTE: Transmission/us				- 10	LPUU	UEFVF	\$0.00	\$0.00	\$0.00								
with POTS circuit switched	ed usage will also app	ly															
to circuit switched voice data transmission by B-C																	
2-wire ISD																	
2		1							1		l				1		

								RATES					oss	RATES		
					ľ										Incremental Charge -	Incremental Charge -
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS	Not in TRA Docket 97-	Zone	BCS	usoc						Svc Order	Svc Order	Incremental	Incremental	Manual Svc	Manual Svc
ONDONDEED NETWORK ELEMENT	STATED IN DOCKET 97-01262	01262	Zone	БОО	0000				Nonre	curring	Submitted Elec	Submitted	Charge - Manual	Charge - Manual	Order vs. Electronic-Disc	Order vs.
							Nonre	curring	Disc	onnect	per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'l	1st	Electronic-Dis Add'l
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
NOTE: Access to B Channel or D Channel																
acket capabilities will be available only throug																
R/New Business Request Process. Rates fo																
e packet capabilities will be determined via th	<b>e</b>															
Bona Fide Request/New Business Request																
Exchange Ports - 2-Wire ISDN Port Channel Profiles		*			U1UMA	\$0.00	\$0.00	\$0.00								
Charlier Fromes	Exchange Ports - 4-Wire ISDN				UTUNA	\$0.00	\$0.00	\$0.00								
Exchange Ports - 4-Wire ISDN DS1 Port	DS1 Port			UEPEX	UEPEX	\$75.04	\$148.66	\$147.18	\$38.46	\$36.98			\$40.69	\$42.17	\$9.07	\$10.54
2-Wire VG Unbundled 2-Way PBX Trunk	Exchange Ports - 2-Wire Analog					Ţ	***********	*********	***************************************	400.00	1		¥	<b>*</b>	40.0.	
Res	Line Port (PBX)			UEPSE	UEPRD	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire VG Line Side Unbundled 2-Way PB)																
Trunk - Bus	Line Port (PBX)			UEPSP	UEPPC	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire VG Line Side Unbundled Outward	Exchange Ports - 2-Wire Analog															
PBX Trunk - Bus	Line Port (PBX)			UEPSP	UEPPO	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus	Exchange Ports - 2-Wire Analog Line Port (PBX)			UEPSP	UEPP1	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Analog Long Distance Terminal PBX				ULFSF	ULFFI	\$1.79	\$9.93	ф9.19	φ3.00	\$2.92	1		\$20.33	\$10.54	\$13.32	\$1.40
Trunk - Bus	Line Port (PBX)			UEPSP	UEPLD	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Analog TN 2-Way Calling Plan PBX	Exchange Ports - 2-Wire Analog			02. 0.	02.20	ψ1.73	ψ5.55	ψο.10	ψ0.00	Ψ2.52			Ψ20.00	ψ10.04	ψ10.02	Ψ1το
Trunk - Bus	Line Port (PBX)			UEPSP	UEPT2	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire TN Outward Calling Plan PBX Trunk	Exchange Ports - 2-Wire Analog					•			•							
Bus	Line Port (PBX)			UEPSP	UEPTO	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Voice Unbundled PBX LD Terminal	Exchange Ports - 2-Wire Analog															
Ports	Line Port (PBX)			UEPSP	UEPLD	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Voice Unbundled 2-Way PBX	Exchange Ports - 2-Wire Analog				l l				_	_			_	_	_	
Tennessee Calling Port	Line Port (PBX)			UEPSP	UEPT2	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Voice Unbundled 1-Way Outgoing PBX Tennessee Calling Port	Exchange Ports - 2-Wire Analog Line Port (PBX)			UEPSP	UEPTO	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Vice Unbundled 2-Way PBX Usage	Exchange Ports - 2-Wire Analog			UEFSF	UEPIU	\$1.79	\$9.93	\$9.19	\$3.00	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Port	Line Port (PBX)			UEPSP	UEPXA	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Voice Unbundled PBX Toll Termina	Exchange Ports - 2-Wire Analog			OLI OI	OLI 701	Ψ1.73	ψ3.33	ψ3.13	ψ3.00	Ψ2.32			Ψ20.33	ψ10.5 <del>4</del>	ψ13.32	Ψ1.40
Hotel Ports	Line Port (PBX)			UEPSP	UEPXB	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Voice Unbundled PBX LD DDD	Exchange Ports - 2-Wire Analog								*	, ,			, , , , , ,			1
Terminals Port	Line Port (PBX)			UEPSP	UEPXC	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Voice Unbundled PBX LD Terminal	Exchange Ports - 2-Wire Analog															
Switchboard Port	Line Port (PBX)			UEPSP	UEPXD	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Voice Unbundled PBX LD Terminal	Exchange Ports - 2-Wire Analog			LIEDOD	UEPXE	0		00.40	•••					212 = 1		
Switchboard IDD Capable Port 2-Wire Voice Unbundled 2-Way PBX	Line Port (PBX)			UEPSP	UEPXE	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Hotel/Hospital Economy Administrative	Exchange Ports - 2-Wire Analog															
Calling Port	Line Port (PBX)			UEPSP	UEPXL	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Voice Unbundled 2-Way PBX	Exchange Ports - 2-Wire Analog			02. 0.	OL: AL	ψ1.73	ψ5.55	ψ5.15	ψ0.00	Ψ2.02	1		Ψ20.00	ψ10.04	ψ10.02	Ψ110
Hotel/Hospital Economy Room Calling Port				UEPSP	UEPXM	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Voice Unbundled 1-Way Outgoing																
PBX Hotel/Hospital Economy Admin Calling																
Port TN Calling Port	Line Port (PBX)			UEPSP	UEPXN	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Voice Unbundled 1-Way Outgoing	Full and Burton C.W. And a															
PBX Hotel/Hospital Discount Room Calling Port	Exchange Ports - 2-Wire Analog Line Port (PBX)			UEPSP	UEPXO	£4.70	<b>60.00</b>	<b>CO 40</b>	<b>#2.00</b>	<b>#0.00</b>			<b>#20.25</b>	640.54	£40.00	ma 40
2-Wire Voice Unbundled 1-Way Outgoing	Exchange Ports - 2-Wire Analog			UEPSP	UEPXU	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92	1		\$20.35	\$10.54	\$13.32	\$1.40
PBX Measured Port	Line Port (PBX)			UEPSP	UEPXS	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Voice Unbundled PBX Collierville an				021 01	321 7.0	ψ1.10	ψυ.υυ	ψυ.1υ	ψ5.00	Ψ2.32			Ψ20.55	ψ10.54	ψ10.02	Ψ1τ0
Memphis Calling Port	Line Port (PBX)			UEPSP	UEPXU	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Voice Unbundled 2-Way PBX	Exchange Ports - 2-Wire Analog							*								1
Tennessee RegionServ Calling Port	Line Port (PBX)			UEPSP	UEPXV	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Subsequent Activity		*		UEPSP	USASC		\$10.00	\$10.00								
	[Exchange Ports includes all															
All Available Vertical Features	Applicable Features.]	1			UEPVF	\$0.00	\$0.00	\$0.00				I	I			1

1									RATES					oss	RATES		
																Incremental Charge -	Incremental Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262	Zone	BCS	usoc			<u>I</u>	Nonr	ecurring	Svc Order Submitted	Svc Order Submitted	Incremental Charge - Manual	Incremental Charge - Manual	Manual Svc Order vs.	Manual Svc Order vs.
								Nonre	curring	Disc	connect	Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'l	Electronic-Disc 1st	Electronic-Dis Add'l
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Exchange Ports - Coin Port	Exchange Ports - Coin Port					\$2.11	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
NBUN	DLED LOCAL SWITCHING, PORT USAGE																
En	d Office Switching (Port Usage)																
	End Office Switching Function, Per MOU	End Office Switching Function End Office Interoffice Trunk Port -					\$0.0008041										
	End Office Trunk Port - Shared, Per MOU	Shared, per MOU															
+	End Office Frank Fore Officed, For Moo	Onarea, per Mee															+
Та	ndem Switching (Port Usage) (Local or Acc	ess Tandem)															+
	Tandem Switching Function Per MOU	Tandem Switching Function					\$0.0009778										+
	g a said	Tandem Interoffice Trunk Port -					***************************************										
	Tandem Trunk Port - Shared, Per MOU	Shared, per MOU															
NBUN	DLED TRANSPORT																
CC	MMON TRANSPORT (Shared)																
	Common Transport - Per Mile, Per MOU	Common Transport - per mile, per MOU					\$0.0000064										
-+	Common Transport - Facilities Termination	Common Transport - Facilities					\$0.0000064										+
	Per MOU	Termination per MOU					\$0.0003871										
IN	TEROFFICE CHANNEL - DEDICATED TRANS	SPORT - VOICE GRADE															
	Interoffice Channel - Dedicated Transport - 2																
	Wire Voice Grade - Per Mile per month Interoffice Channel - Dedicated Transport- 2	Voice Grade - per mile per MOU			U1TVX	1L5XX	\$0.0174										
	Wire Voice Grade - Facility Termination per																
	month	Termination			U1TVX	U1TV2	\$18.58	\$55.39	\$17.37	\$27.96	\$3.51			\$20.35	\$21.09	\$9.80	\$10.54
	Interoffice Channel - Dedicated Transpor t-							•		•				-		*	
	Wire Voice Grade Rev Bat Per Mile per	Interoffice Transport - Dedicated -															
	month Interoffice Channel-Dedicated Transport- 2	Voice Grade - per mile per MOU Interoffice Transport - Dedicated -			U1TVX	1L5XX	\$0.0174										
	Wire Voice Grade Rev BatFacility	2-Wire Voice Grade - Facility															
	Termination per month	Termination			U1TVX	U1TR2	\$18.58	\$55.39	\$17.37	\$27.96	\$3.51			\$20.35	\$21.09	\$9.80	\$10.54
IN	TEROFFICE CHANNEL - DEDICATED TRANS																
	Interoffice Channel - Dedicated Transport	Interoffice Transport - Dedicated -				41.500											
-	56 kbps - per mile per month	DSO - 56/64 kbps - per mile Interoffice Transport - Dedicated -		H	U1TDX	1L5XX	\$0.0174										
	Interoffice Channel - Dedicated Transport -	DSO - 56/64 kbps - Facility		l l													
	56 kbps - Facility Termination per month	Termination		l l	U1TDX	U1TD5	\$17.98	\$55.39	\$17.37	\$27.96	\$3.51			\$20.35	\$21.09	\$9.80	\$10.54
	Interoffice Channel - Dedicated Transport	Interoffice Transport - Dedicated -														*****	
$\perp$	64 kbps - per mile per month	DSO - 56/64 kbps - per mile		Ш	U1TDX	1L5XX	\$0.0174										
	Interoffice Channel - Dedicated Transport -	Interoffice Transport - Dedicated - DSO - 56/64 kbps - Facility															
	64 kbps - Facility Termination per month	Termination			U1TDX	U1TD6	\$17.98	\$55.39	\$17.37	\$27.96	\$3.51			\$20.35	\$21.09	\$9.80	\$10.54
+	- In the state of				35/		<b>\$</b> 00	ψ00.00	ψ	Ψ27.00	ψ3.01	1	1	7-3.00	7=1.00	ψ5.00	\$.0.04
IN	TEROFFICE CHANNEL - DEDICATED TRANS	SPORT - DS1															
	Interoffice Channel - Dedicated Channel	Interoffice Transport - Dedicated -															+
	DS1 - Per Mile per month	DS1 - per mile		L_l	U1TD1	1L5XX	\$0.3562					<u> </u>		<u></u>			1
		Interoffice Transport - Dedicated -				==								000	***		
+	DS1 - Facility Termination per month	DS1 - Facility Termination		H	U1TD1	U1TF1	\$77.86	\$112.40	\$76.27	\$19.55	\$14.99			\$20.35	\$21.09	\$9.80	\$10.54
	CAL CHANNEL - DEDICATED TRANSPORT			H													
			1						l .		1				1		1

									RATES					oss	RATES		
						1										Incremental Charge -	Incremental Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97-	Zone	BCS	usoc						Svc Order	Svc Order	Incremental	Incremental	Manual Svc	Manual Svc
		STATED IN DOCKET 97-01262	01262			ł				Nonre	ecurring	Submitted Elec	Submitted Manually per	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Order vs. Electronic-Disc	Order vs. Electronic-Disc
									curring		onnect	per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'l
		Local Channel - Dedicated - 2		_			Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Local Channel - Dedicated - 2-Wire Voice	Wire Voice Grade [shown here															
	Grade per month	deaveraged]		U	ILDVX	ULDV2	\$17.18	\$199.33	\$24.16	\$54.81	\$4.80			\$20.35	\$10.54	\$13.30	\$0.00
	•	<u> </u>							•	, , , , , , , , , , , , , , , , , , , ,							
		Local Channel - Dedicated - 2															
	Local Channel - Dedicated - 2-Wire Voice	Wire Voice Grade [shown here															
	Grade Rev. Bat. Per month	deaveraged] Local Channel - Dedicated - 4		U	ILDVX	ULDR2	\$17.18	\$199.33	\$24.16	\$54.81	\$4.80			\$20.35	\$10.54	\$13.30	\$0.00
	Local Channel - Dedicated - 4-Wire Voice	Wire Voice Grade [shown here															
	Grade per month	deaveraged]		U	ILDVX	ULDV4	\$18.18	\$201.53	\$24.83	\$55.52	\$5.51			\$20.35	\$10.54	\$13.30	\$0.00
	·																
		Local Channel - Dedicated DS1															
	Local Channel - Dedicated - DS1 per month	[shown here deaveraged]		U	JLDD1	ULDF1	\$36.24	\$277.35	\$233.26	\$33.18	\$22.30			\$45.68	\$1.76	\$21.75	\$1.76
																	<u> </u>
MUL	Channelization - DS1 to DS0 Channe	Channelization Channel Co.															<u> </u>
	System	Channelization - Channel System DS1 to DSO		- 1	JXTD1	MQ1	\$80.77	\$141.87	\$77.11	\$14.51	\$13.46			\$20.35	\$9.80	\$11.49	\$1.18
	OCU-DP COCI (data) - DS1 to DS0 Channe			U.	וטואנ	IVIQI	φου.//	\$141.0 <i>1</i>	φ//.II	\$14.51	\$13.40			\$20.33	\$9.60	\$11.49	Φ1.10
	System - per month (2.4-64kbs)	DSO - OCU - DP Card			UDL	1D1DD	\$1.82	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
	2-wire ISDN COCI (BRITE) - DS1 to DS0	Interface Unit - Interface DS1 to															
	Channel Systsem - per month	DSO - Brite Card		ι	UDN	UC1CA	\$3.10	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
	Voice Grade COCI - DS1 to DS0 Channel System - per month	Interface Unit - Interface DS1 to DSO - Voice Grade Card		Ι.		1D1VG	f. 04	ФС 07	£4.00					#20.2F	<b>©0.00</b>	C44 40	C4 40
	System - per month	Channelization - Channel System			UEA	IDIVG	\$.91	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
	DS3 to DS1 Channel System per month	DS3 to DS1		U.	JXTD3	MQ3	\$222.98	\$308.03	\$108.47	\$44.47	\$42.62			\$20.35	\$9.80	\$11.49	\$1.18
		Channelization - Channel System					•	,		·				*	*		
	STS1 to DS1 Channel System per month	DS3 to DS1		U	JXTS1	MQ3	\$222.98	\$308.03	\$108.47	\$44.47	\$42.62			\$20.35	\$9.80	\$11.49	\$1.18
	DS3 Interface Unit (DS1 COCI) used with	Interface Unit - Interface DS3 to DS1		Ι.		LICADA	017.50	00.07	04.00					#00.0F	<b>60.00</b>	044.40	04.40
	Loop per month DS3 Interface Unit (DS1 COCI) used with	Interface Unit - Interface DS3 to			USL	UC1D1	\$17.58	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
	Local Channel per month	DS1		U	ILDD1	UC1D1	\$17.58	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
	DS3 Interface Unit (DS1 COCI) used with	Interface Unit - Interface DS3 to					•	,	,					*	*		
	Interoffice Channel per month	DS1		U	J1TD1	UC1D1	\$17.58	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
DARI	( FIBER																<del>                                     </del>
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local	Dark Fiber, per 4 fiber strands, pe	r														
	Channel	route mile or fraction thereof	1	l l	UDF	1L5DC	\$53.23										
		Dark Fiber, per 4 fiber strands, pe	r				+										
	NRC Dark Fiber - Local Channel	route mile or fraction thereof		ı	UDF	UDFC4		\$1,219.22	\$169.75	\$453.22	\$339.34						
	Dark Fiber, Four Fiber Strands, Per Route	Dork Eibor, por 4 fiber street	_ T		Ţ	T											
	Mile or Fraction Thereof per month - Interoffice Channel	Dark Fiber, per 4 fiber strands, per route mile or fraction thereof	1	Ι,	UDF	1L5DF	\$53.23										
	interonice Chainei	Dark Fiber, per 4 fiber strands, pe	r	-+	ODI	I LUDI	<b>Φ</b> υυ.∠υ										
	NRC Dark Fiber - Interoffice Channel	route mile or fraction thereof		ı	UDF	UDF14		\$1,219.22	\$169.75	\$453.22	\$339.34						
	Dark Fiber, Four Fiber Strands, Per Route																
	Mile or Fraction Thereof per month - Local	Dark Fiber, per 4 fiber strands, pe	r	Ι.		41.501	<b>AFO OC</b>										
	Loop	route mile or fraction thereof  Dark Fiber, per 4 fiber strands, pe			UDF	1L5DL	\$53.23			<b></b>		1					<del></del>
	NRC Dark Fiber - Local Loop	route mile or fraction thereof	1	_ [ ,	UDF	UDFL4		\$1,219.22	\$169.75	\$453.22	\$339.34						
	zami nor zoda zoop					20.21		ψ1,210.22	¥100.70	ψ-100.22	ψ000.04						
8XX	ACCESS TEN DIGIT SCREENING																
		800 Access Ten digit screening															
	8XX Access Ten Digit Screening, Per Call	(800 ATDS) per call					\$0.0005192										

Re Re Re Re Re Re Re Re Re Re Re Re Re R	UNBUNDLED NETWORK ELEMENT  XX Access Ten Digit Screening, esservation Charge Per 8XX Number esserved  XX Access Ten Digit Screening, Per 8XX OLE Stablished W/O POTS Translations  XX Access Ten Digit Screening, Per 8XX OLE Stablished With POTS Translations  XX Access Ten Digit Screening, Stablished With POTS Translations  XX Access Ten Digit Screening, Multiple terLATA CXR Routing Per CXR Requester 8XX No.  XX Access Ten Digit Screening, Change harge Per Request  XX Access Ten Digit Screening, Change harge Per Request  XX Access Ten Digit Screening, Call andling and Destination Features	B00 Access Ten digit screening Reservation Charge per 800 Number Reserved 800 Access Ten digit screening Pestations 800 Access Ten digit screening Pestations 800 # established w/o POTS Translations 800 Access Ten digit screening Pestations 800 Access Ten digit screening Pestations 800 Access Ten digit screening Customized Area of Service per 800 ATDS, Multiple InterLATA	t in TRA cket 97- Zon 11262	BCS	N8R1X  N8FTX  N8FCX  N8FMX	Rec	\$5.21 \$11.47	\$0.76 \$1.46		onnect Add'I  \$0.7602	Svc Order Submitted Elec per LSR SOMEC	Svc Order Submitted Manually per LSR SOMAN	Incremental Charge - Manual Svc Order vs. Electronic-1st SOMAN \$20.35	Incremental Charge - Manual Svc Order vs. Electronic-Add'l SOMAN	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st SOMAN	Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'I SOMAN
Re Re Re Re Re Re Re Re Re Re Re Re Re R	eservation Charge Per 8XX Number eserved  (X Access Ten Digit Screening, Per 8XX o. Established W/O POTS Translations  (X Access Ten Digit Screening, Per 8XX o. Established With POTS Translations  (X Access Ten Digit Screening, Per 8XX o. Established With POTS Translations  (X Access Ten Digit Screening, ustomized Area of Service Per 8XX umber  (X Access Ten Digit Screening, Multiple terLATA CXR Routing Per CXR Requester 8XX No.  (X Access Ten Digit Screening, Change harge Per Request  (X Access Ten Digit Screening, Call andling and Destination Features	800 Access Ten digit screening Reservation Charge per 800 Number Reserved 800 Access Ten digit screening pe 800 # established w/o POTS Translations 800 Access Ten digit screening pe 800 # established with POTS Translations 800 Access Ten digit screening Customized Area of Service per 800 Number 800 ATDS, Multiple InterLATA CXR Routing per CXR Requested per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination			N8FTX N8FCX	Rec	\$5.21 \$11.47	\$0.76 \$1.46	First \$7.34	Add'I \$0.7602	per LSR	LSR	SOMAN \$20.35	Electronic-Add'l	1st SOMAN	Add'l
Re Re Re Re Re Re Re Re Re Re Re Re Re R	eservation Charge Per 8XX Number eserved  (X Access Ten Digit Screening, Per 8XX o. Established W/O POTS Translations  (X Access Ten Digit Screening, Per 8XX o. Established With POTS Translations  (X Access Ten Digit Screening, Per 8XX o. Established With POTS Translations  (X Access Ten Digit Screening, ustomized Area of Service Per 8XX umber  (X Access Ten Digit Screening, Multiple terLATA CXR Routing Per CXR Requester 8XX No.  (X Access Ten Digit Screening, Change harge Per Request  (X Access Ten Digit Screening, Call andling and Destination Features	Reservation Charge per 800 Number Reserved 800 Access Ten digit screening pe 800 # established w/o POTS Translations 800 Access Ten digit screening pe 800 # established with POTS Translations 800 Access Ten digit screening Customized Area of Service per 800 Number 800 ATDS, Multiple InterLATA CXR Routing per CXR Requested per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination			N8FTX N8FCX	Rec	\$5.21 \$11.47	\$0.76 \$1.46	First \$7.34	Add'I \$0.7602			\$20.35		SOMAN	
Re Re Re Re Re Re Re Re Re Re Re Re Re R	eservation Charge Per 8XX Number eserved  (X Access Ten Digit Screening, Per 8XX o. Established W/O POTS Translations  (X Access Ten Digit Screening, Per 8XX o. Established With POTS Translations  (X Access Ten Digit Screening, Per 8XX o. Established With POTS Translations  (X Access Ten Digit Screening, ustomized Area of Service Per 8XX umber  (X Access Ten Digit Screening, Multiple terLATA CXR Routing Per CXR Requester 8XX No.  (X Access Ten Digit Screening, Change harge Per Request  (X Access Ten Digit Screening, Call andling and Destination Features	Reservation Charge per 800 Number Reserved 800 Access Ten digit screening pe 800 # established w/o POTS Translations 800 Access Ten digit screening pe 800 # established with POTS Translations 800 Access Ten digit screening Customized Area of Service per 800 Number 800 ATDS, Multiple InterLATA CXR Routing per CXR Requested per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination			N8FTX N8FCX		\$11.47	\$1.46							\$13.28	
Re  8XX No  8XX No  8XX Cu  8XX Inte Pee  8XX Ch  8XX LINE INFOR  LID LID LID LID LID SIGNALING CCC	eserved  XX Access Ten Digit Screening, Per 8XX b. Established W/O POTS Translations  XX Access Ten Digit Screening, Per 8XX b. Established With POTS Translations  XX Access Ten Digit Screening,  ustomized Area of Service Per 8XX  umber  XX Access Ten Digit Screening, Multiple  terLATA CXR Routing Per CXR Requeste  er 8XX No.  XX Access Ten Digit Screening, Change  harge Per Request  XX Access Ten Digit Screening, Call  andling and Destination Features	Number Reserved 800 Access Ten digit screening pe 800 # established w/o POTS Translations 800 Access Ten digit screening pe 800 # established with POTS Translations 800 Access Ten digit screening Customized Area of Service per 800 Number 800 ATDS, Multiple InterLATA CXR Routing per CXR Requested per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination			N8FTX N8FCX		\$11.47	\$1.46							\$13.28	
8XX No 8XX No 8XX Cu Nu 8XX Inte Peie 8XX Ch 8XX HA LINE INFOR	XX Access Ten Digit Screening, Per 8XX b. Established W/O POTS Translations  XX Access Ten Digit Screening, Per 8XX b. Established With POTS Translations b. Established With POTS Translations b. Established With POTS Translations b. Established With POTS Translations b. Established With POTS Translations b. Established With POTS Translations b. Established With PoTS Trans	800 Access Ten digit screening pe 800 # established w/o POTS Translations 800 Access Ten digit screening pe 800 # established with POTS Translations 800 Access Ten digit screening Customized Area of Service per 800 Number 800 ATDS, Multiple InterLATA CXR Routing per CXR Requested per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination			N8FTX N8FCX		\$11.47	\$1.46							\$13.28	
No  8X: No 8X: Cu Nu 8X: Inte Pei 8X: Ch LID LID LID LID SIGNALING CCC	o. Established W/O POTS Translations  XX Access Ten Digit Screening, Per 8XX b. Established With POTS Translations  XX Access Ten Digit Screening, ustomized Area of Service Per 8XX umber  XX Access Ten Digit Screening, Multiple terLATA CXR Routing Per CXR Requester er 8XX No.  XX Access Ten Digit Screening, Change harge Per Request  XX Access Ten Digit Screening, Change harge Per Request  XX Access Ten Digit Screening, Call andling and Destination Features	800 # established w/o POTS Translations 800 Access Ten digit screening pe 800 # established with POTS Translations 800 Access Ten digit screening Customized Area of Service per 800 Number 800 ATDS, Multiple InterLATA CXR Routing per CXR Requested per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination			N8FCX			·					\$20.35		\$13.28	
No  8X: No 8X: Cu Nu 8X: Inte Pei 8X: Ch LID LID LID LID SIGNALING CCC	o. Established W/O POTS Translations  XX Access Ten Digit Screening, Per 8XX b. Established With POTS Translations  XX Access Ten Digit Screening, ustomized Area of Service Per 8XX umber  XX Access Ten Digit Screening, Multiple terLATA CXR Routing Per CXR Requester er 8XX No.  XX Access Ten Digit Screening, Change harge Per Request  XX Access Ten Digit Screening, Change harge Per Request  XX Access Ten Digit Screening, Call andling and Destination Features	Translations 800 Access Ten digit screening pe 800 # established with POTS Translations 800 Access Ten digit screening Customized Area of Service per 800 Number 800 ATDS, Multiple InterLATA CXR Routing per CXR Requested per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination			N8FCX			·					\$20.35		\$13.28	
BXX No BX Cu Nu BX Inte Pei BX Ch BX LINE INFOR LID LID CO SIGNALING CCC	XX Access Ten Digit Screening, Per 8XX b. Established With POTS Translations XX Access Ten Digit Screening, ustomized Area of Service Per 8XX umber XX Access Ten Digit Screening, Multiple terLATA CXR Routing Per CXR Requeste ber 8XX No. XX Access Ten Digit Screening, Change harge Per Request XX Access Ten Digit Screening, Call andling and Destination Features	800 # established with POTS Translations 800 Access Ten digit screening Customized Area of Service per 800 Number 800 ATDS, Multiple InterLATA CXR Routing per CXR Requested per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination			N8FCX			·					720.00		¥10.20	<del>                                     </del>
No 8XX Cu Nu 8XX Inte Pei 8XX Ch  8XX LINE INFOR LIC LIC Or ( SIGNALING	o. Established With POTS Translations (XX Access Ten Digit Screening, ustomized Area of Service Per 8XX umber (XX Access Ten Digit Screening, Multiple terLATA CXR Routing Per CXR Requester ar 8XX No. (XX Access Ten Digit Screening, Change harge Per Request (XX Access Ten Digit Screening, Call andling and Destination Features	Translations 800 Access Ten digit screening Customized Area of Service per 800 Number 800 ATDS, Multiple InterLATA CXR Routing per CXR Requested per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination			N8FCX		\$11.47	\$1.46	\$7.34							
BX. Cu Nu 8X. Inte Pee 8X. Ch  8X. LINE INFOR LID LID LID SIGNALING CCC	XX Access Ten Digit Screening, ustomized Area of Service Per 8XX umber XX Access Ten Digit Screening, Multiple terLATA CXR Routing Per CXR Requester 8XX No. XX Access Ten Digit Screening, Change harge Per Request XX Access Ten Digit Screening, Change harge Per Request XX Access Ten Digit Screening, Call andling and Destination Features	800 Access Ten digit screening Customized Area of Service per 800 Number 800 ATDS, Multiple InterLATA CXR Routing per CXR Requested per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination			N8FCX		\$11.47	\$1.46	\$7.34							, ,
Cu Nu 8X. Inte Pei 8X. Ch 2 Ch 2 LID LID Or 0	ustomized Area of Service Per 8XX umber  KX Access Ten Digit Screening, Multiple terLATA CXR Routing Per CXR Requester er 8XX No. KX Access Ten Digit Screening, Change harge Per Request  KX Access Ten Digit Screening, Call andling and Destination Features	Customized Area of Service per 800 Number 800 ATDS, Multiple InterLATA CXR Routing per CXR Requested per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination							Ψ1.0.	\$0.7602			\$20.35		\$13.28	
Nu 8XX Inte	umber  (XX Access Ten Digit Screening, Multiple terLATA CXR Routing Per CXR Requester re 8XX No.  (XX Access Ten Digit Screening, Change harge Per Request  (XX Access Ten Digit Screening, Call andling and Destination Features	800 Number 800 ATDS, Multiple InterLATA CXR Routing per CXR Requested per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination														,
BX. Inte Pere  BX. Ch  BX. Ha  LINE INFOR  LIC LIC OF (  SIGNALING	XX Access Ten Digit Screening, Multiple terLATA CXR Routing Per CXR Requester are 8XX No.  XX Access Ten Digit Screening, Change harge Per Request  XX Access Ten Digit Screening, Call andling and Destination Features	800 ATDS, Multiple InterLATA CXR Routing per CXR Requested per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination					\$4.47	\$2.24								,
Inte Pei 8XX Ch 8XX Ha LINE INFOR LIC LIC Cor (SIGNALING CCC	terLATA CXR Routing Per CXR Requester or 8XX No. XX Access Ten Digit Screening, Change harge Per Request XX Access Ten Digit Screening, Call andling and Destination Features	CXR Routing per CXR Requested per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination			NIOEMY		\$4.47	\$2.24								
Pei 8X. Ch. 8X. Ha LINE INFOR LIC LIC SIGNALING CCC	er 8XX No.  XX Access Ten Digit Screening, Change harge Per Request  XX Access Ten Digit Screening, Call andling and Destination Features	per 800 # 800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination			NIOEMAV											,
LINE INFOR LIC LIC OT (  SIGNALING	harge Per Request  (X Access Ten Digit Screening, Call andling and Destination Features	Change Charge per Request 800 Access Ten digit screening, Call Handling and Destination			INOLIVIV		\$5.23	\$3.00								,
LINE INFOR	XX Access Ten Digit Screening, Call andling and Destination Features	800 Access Ten digit screening, Call Handling and Destination						*****								
LINE INFOR	andling and Destination Features	Call Handling and Destination			N8FAX		\$5.97	\$0.76					\$20.35			,
LINE INFOR	andling and Destination Features															
LINE INFOR		Features					4									,
LID LID Or ( SIGNALING	RMATION DATA BASE ACCESS (LIDB)				N8FDX		\$4.47									ļ'
LID LID Or ( SIGNALING	RMATION DATA BASE ACCESS (LIDB)															ļ!
LIE LIE OF (		LIDB Common Transport per														ļ!
LIE LIE OF (	DB Common Transport Per Query	Query			OQT	\$0.0000354										,
SIGNALING CC	DB Validation Per Query	LIDB Validation per Query		-	OQU	\$0.0000334			1		1					
SIGNALING CC	DB Originating Point Code Establishmen	LIDB Originating Point Code		1	UQU	\$0.0117403										
SIGNALING CC	Change	Establishment or Change					\$49.03						\$20.35			,
cc	Change	Establishment of Ghange		1			ψ49.03						Ψ20.33			
cc	C(CCS7)			1												
	5 (CC37)	CCS7 Signaling Termination per														
	CS7 Signaling Termination, Per STP Port	STP Port				\$138.41										,
CC	oor olgitaling rollimation, roll of the	CCS7 Signaling Usage per TCAP				ψ100.41										
	CS7 Signaling Usage, Per TCAP Message					\$0.0000916										,
1 1		CCS7 Signaling Connection per 56														
	CS7 Signaling Connection, Per link (A link					\$17.84	\$130.84						\$20.35			
	CS7 Signaling Connection, Per link (B link					4										,
(als	Iso known as D link)	kbps faciliity CCS7 Signaling Usage, per call				\$17.84	\$130.84						\$20.35			
cc	CS7 Signaling Usage, Per ISUP Message	setup message				\$0.0000373										,
	graining coago, i or root mossage	CCS7 Signaling Usage Surrogate,		1	1	\$0.0000073										
	CS7 Signaling Usage Surrogate, per link	per 56 kbps facility per LATA per		1												
per	er LATA	month		1	<u> </u>	\$352.30			<u> </u>		<u></u>		\$20.35			
SELECTIVE	ROUTING		T													
		Selective Routing (Interim Solution														
	elective Deutine Deutlichen Line Ci	Line Class Codes) per Unique Line														
	elective Routing Per Unique Line Class ode Per Request Per Switch	Class Code per Request per Switch			USRCR		¢170.00						\$20.25			
	oue Fei Request Per Switch	SWIICH		-	USKCK		\$179.60						\$20.35			
AINL BELL	COUTU AIN OMO ACCECO CERVICE			-												
AIN - BELLS	SOUTH AIN SMS ACCESS SERVICE	AIN SMS Access Service - Service		1												
11A		Establishment, per State, Initial		1												
	N SMS Access Service - Service	Setup		1	CAMSE		\$135.56									
	N SMS Access Service - Service stablishment, Per State, Initial Setup				1		\$100.00									
	N SMS Access Service - Service stablishment, Per State, Initial Setup															
	stablishment, Per State, Initial Setup	AINLONG Assess Committee Days		1	CAMDP		\$41.75				l	I				
	stablishment, Per State, Initial Setup  N SMS Access Service - Port Connection			1	CAIVIDP	<b> </b>	\$41./5									
ISE	stablishment, Per State, Initial Setup	Connection - Dial/Shared Access														

									RATES					oss	RATES		
														1		Incremental	Incremental
		UNBUNDLED NETEORK ELEMENT AS	Not in TRA	l_								Svc Order	Svc Order	Incremental	Incremental	Charge - Manual Svc	Charge - Manual Svc
	UNBUNDLED NETWORK ELEMENT	STATED IN DOCKET 97-01262	Docket 97- 01262	Zone	BCS	USOC				Nonre	curring	Submitted	Submitted	Charge - Manual	Charge - Manual	Order vs.	Order vs.
			0.202						currina	8	onnect	Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs.	Electronic-Disc 1st	Electronic-Dis
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	Electronic-Add'I SOMAN	SOMAN	SOMAN
		AIN SMS Access Service - User		H			Nec	11131	Auu	11130	Auu i	COMILO	COMPAR	COMPAR	JOHNAIN	JOHIAN	OOMAN
	AIN SMS Access Service - User Identification																
	Codes - Per User ID Code	Code				CAMAU		\$96.63									
		AIN SMS Access Service -						<b>\$00.00</b>									1
	AIN SMS Access Service - Security Card,	Security Card, per User ID Code,															
	Per User ID Code, Initial or Replacement	Initial or Replacement				CAMRC		\$113.67									
	AIN SMS Access Service - Storage, Per Uni	t AIN SMS Access Service -															
	(100 Kilobytes)	Storage, per unit (100 kilobytes)					\$0.0024										
	AIN SMS Access Service - Session, Per	AIN SMS Access Service -															
	Minute	Session, per minute					\$0.0820123										
		AIN SMS Access Service -															
	AIN SMS Access Service - Company	company performed session, per															
	Performed Session, Per Minute	minute					\$2.27										
AIN - BE	LLSOUTH AIN TOOLKIT SERVICE																
	AIN Toolkit Service - Service Establishmen	AIN TS - Service Establishment															
	Charge, Per State, Initial Setup	Charge per State, Initial Setup				BAPSC		\$132.04									
	AIN Toolkit Service - Training Session, Pe	AIN TS - Training Session, per															
	Customer	Customer				BAPVX		\$7,915									
	AIN Toolkit Service - Trigger Access Charge																
	Per Trigger, Per DN, Term. Attempt	per Trigger per DN, Term Attempt				BAPTT		\$31.21									
		AIN TS - Trigger Access Charge,															
	AIN Toolkit Service - Trigger Access Charge					DADTD											
	Per Trigger, Per DN, Off-Hook Delay	Delay				BAPTD		\$31.21									
	AINI Tabilit Candas Trinnas Assaul Channe	AIN TS - Trigger Access Charge,															
	AIN Toolkit Service - Trigger Access Charge	Immediate				ВАРТМ											
	Per Trigger, Per DN, Off-Hook Immediate	Immediate		-		BAPTM		\$31.21				1					ļ
	AIN Toolkit Service - Trigger Access Charge	AINITS Trigger Assess Charge															
	Per Trigger, Per DN, 10-Digit PODP	per Trigger per DN, 10-Digit PODF				варто		\$85.24									
	AIN Toolkit Service - Trigger Access Charge					BAFTO		\$65.24				1					
	Per Trigger, Per DN, CDP	per Trigger per DN, CDP				BAPTC		\$85.24									
	r er ringger, r er biv, ebr	per migger per bry, ebr				D/ 11 10		ψ03.24		1							-
	AIN Toolkit Service - Trigger Access Charge	AIN TS - Trigger Access Charge.															
	Per Trigger, Per DN, Feature Code	per Trigger per DN, Feature Code				BAPTF		\$85.24									
	AIN Toolkit Service - Query Charge, Per	Fee mager per and reasons are						<b>\$50.2</b> .									
	Query	AIN TS - Query Charge, Per Query	,				\$.0211882										
	AIN Toolkit Service - Type 1 Node Charge	AIN TS - Type 1 Node Charge, per		t			***************************************										
	Per AIN Toolkit Subscription, Per Node, Per	AIN Toolkit subscription, per Node															
	Query	per query					\$.0054774										
		AIN TS - SCP Storage charge, per															
	AIN Toolkit Service - SCP Storage Charge,	SMS access account, per 100															
	Per SMS Access Account, Per 100 Kilobytes						\$1.50										
	AIN Toolkit Service - Monthly report - Per Al			1 ]													
	Toolkit Service Subscription	TS Subscription				BAPMS	\$17.43	\$33.52		1							
	AIN Toolkit Service - Special Study - Per All									I							
	Toolkit Service Subscription	TS Subscription				BAPLS	\$0.1321116	\$36.23		1							
	AIN Toolkit Service - Call Event Report - Pe					DARSS		005									
	AIN Toolkit Service Subscription	AIN TS Subscription				BAPDS	\$17.35	\$33.52				1					
	AINI Tarillit Camina Call Francis	AIN TO CHI Franci Constitution															
	AIN Toolkit Service - Call Event Special	AIN TS - Call Event Special Study	1			DADEO	00.0544405	#00.0C		I							
	Study - Per AIN Toolkit Service Subscription	per AIN 1S Subscription		$\sqcup$		BAPES	\$0.0511435	\$36.23		<b> </b>							
										ļ							
ODUF/EI	DOUF/ADUF/CMDS		<u> </u>	L I									<u> </u>				
OP.	TIONAL DAILY USAGE FILE (ODUF)			ΙT		1											
		OSS OLEC Daily Usage File		t		1				1		1					<b>†</b>
	1	Recording per message	1	ī I		1	\$0.0000044				1	1	1	I	Ì	l	1

OD Tap OD (CC UNBUN Cost Ba requirer provic Ports.	pe provisioned  OUF: Data Transmission	OSS OLEC Daily Usage File Message Distribution per magnetit tape provisioned OSS OLEC Daily Usage File Data Transmission (Connect: Direct), per message			BCS	usoc	Rec \$0.0027366	Nonro First	ecurring Add'l		ecurring onnect Add'I	Svc Order Submitted Elec per LSR SOMEC	Svc Order Submitted Manually per LSR SOMAN	Incremental Charge - Manual Svc Order vs. Electronic-1st SOMAN	Incremental Charge - Manual Svc Order vs. Electronic-Add'I SOMAN	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st SOMAN	Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'I SOMAN
OD Tap OD (CC UNBUN Cost Ba requirer provic Ports.	DUF: Message Processing, per message DUF: Message Processing, per Magnetic pe provisioned DUF: Data Transmission DNNECT:DIRECT), per message  NDLED PORT/LOOP COMBINATIONS - C ased Rates are applied where BellSouth is d by FCC and/or State Commission rule to de Unbundled Local Switching or Switch BellSouth is required to offer unbundled	OSS OLEC Daily Usage File Message Distribution per message OSS OLEC Daily Usage File Message Distribution per magnetic tape provisioned OSS OLEC Daily Usage File Data Transmission (Connect: Direct), per message	Docket 97- 01262		BCS	usoc				Disc	onnect	Submitted Elec per LSR	Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic-1st	Charge - Manual Svc Order vs. Electronic-Add'l	Manual Svc Order vs. Electronic-Disc 1st	Manual Svc Order vs. Electronic-Dis Add'l
OD Tap OD (CC UNBUN Cost Ba requirer provic Ports.	DUF: Message Processing, per Magnetic pe provisioned DUF: Data Transmission DNNECT:DIRECT), per message  NDLED PORT/LOOP COMBINATIONS - Cased Rates are applied where BellSouth is d by FCC and/or State Commission rule to de Unbundled Local Switching or Switch BellSouth is required to offer unbundled	OSS OLEC Daily Usage File Message Distribution per message OSS OLEC Daily Usage File Message Distribution per magnetic tape provisioned OSS OLEC Daily Usage File Data Transmission (Connect: Direct), per message	8							Disc	onnect	Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'l	Electronic-Disc 1st	Electronic-Dis
OD Tap OD (CC UNBUN Cost Ba requirer provic Ports.	DUF: Message Processing, per Magnetic pe provisioned DUF: Data Transmission DNNECT:DIRECT), per message  NDLED PORT/LOOP COMBINATIONS - Cased Rates are applied where BellSouth is d by FCC and/or State Commission rule to de Unbundled Local Switching or Switch BellSouth is required to offer unbundled	Message Distribution per message OSS OLEC Daily Usage File Message Distribution per magnetic tape provisioned OSS OLEC Daily Usage File Data Transmission (Connect: Direct), per message															
OD Tap OD (CC UNBUN Cost Ba requirer provic Ports.	DUF: Message Processing, per Magnetic pe provisioned DUF: Data Transmission DNNECT:DIRECT), per message  NDLED PORT/LOOP COMBINATIONS - Cased Rates are applied where BellSouth is d by FCC and/or State Commission rule to de Unbundled Local Switching or Switch BellSouth is required to offer unbundled	Message Distribution per message OSS OLEC Daily Usage File Message Distribution per magnetic tape provisioned OSS OLEC Daily Usage File Data Transmission (Connect: Direct), per message															
OD Tap OD (CC UNBUN Cost Ba requirer provic Ports.	DUF: Message Processing, per Magnetic pe provisioned DUF: Data Transmission DNNECT:DIRECT), per message  NDLED PORT/LOOP COMBINATIONS - Cased Rates are applied where BellSouth is d by FCC and/or State Commission rule to de Unbundled Local Switching or Switch BellSouth is required to offer unbundled	Message Distribution per message OSS OLEC Daily Usage File Message Distribution per magnetic tape provisioned OSS OLEC Daily Usage File Data Transmission (Connect: Direct), per message					\$0.0027366										
OD Tap OD (CC UNBUN Cost Ba requirer provic Ports.	DUF: Message Processing, per Magnetic pe provisioned DUF: Data Transmission DNNECT:DIRECT), per message  NDLED PORT/LOOP COMBINATIONS - Cased Rates are applied where BellSouth is d by FCC and/or State Commission rule to de Unbundled Local Switching or Switch BellSouth is required to offer unbundled	OSS OLEC Daily Usage File Message Distribution per magnetit tape provisioned OSS OLEC Daily Usage File Data Transmission (Connect: Direct), per message					\$0.0027366										
UNBUN Cost Barequired provide Ports.	pe provisioned  PUF: Data Transmission  DNNECT:DIRECT), per message  NDLED PORT/LOOP COMBINATIONS - Coased Rates are applied where BellSouth is to by FCC and/or State Commission rule to de Unbundled Local Switching or Switch BellSouth is required to offer unbundled	Message Distribution per magnetic tape provisioned OSS OLEC Daily Usage File Data Transmission (Connect: Direct), per message	c														1
UNBUN Cost Barequired provide Ports.	pe provisioned  PUF: Data Transmission  DNNECT:DIRECT), per message  NDLED PORT/LOOP COMBINATIONS - Coased Rates are applied where BellSouth is to by FCC and/or State Commission rule to de Unbundled Local Switching or Switch BellSouth is required to offer unbundled	tape provisioned OSS OLEC Daily Usage File Data Transmission (Connect: Direct), per message															I
UNBUN Cost Ba required provid Ports.	DUF: Data Transmission DNNECT:DIRECT), per message  NDLED PORT/LOOP COMBINATIONS - Cased Rates are applied where BellSouth is d by FCC and/or State Commission rule to de Unbundled Local Switching or Switch BellSouth is required to offer unbundled	OSS OLEC Daily Usage File Data Transmission (Connect: Direct), per message					\$52.75										I
UNBUN Cost Ba required provid Ports.	DNNECT:DIRECT), per message  NDLED PORT/LOOP COMBINATIONS - C ased Rates are applied where BellSouth it d by FCC and/or State Commission rule it de Unbundled Local Switching or Switch BellSouth is required to offer unbundled	per message					*										
UNBUN Cost Ba required provid Ports.	NDLED PORT/LOOP COMBINATIONS - C ased Rates are applied where BellSouth is d by FCC and/or State Commission rule to de Unbundled Local Switching or Switch BellSouth is required to offer unbundled																I
Cost Ba required provide Ports.	ased Rates are applied where BellSouth is d by FCC and/or State Commission rule to de Unbundled Local Switching or Switch BellSouth is required to offer unbundled	COST BASED RATES					\$0.0000339										1
Cost Ba required provid Ports. por	ased Rates are applied where BellSouth is d by FCC and/or State Commission rule to de Unbundled Local Switching or Switch BellSouth is required to offer unbundled	COST BASED RATES															1
required provide Ports. por	d by FCC and/or State Commission rule to de Unbundled Local Switching or Switch BellSouth is required to offer unbundled																t
provide Ports.	de Unbundled Local Switching or Switch BellSouth is required to offer unbundled																I
Ports.	BellSouth is required to offer unbundled																
	rt/loop combinations that are Currently																I
Combin																	I
	ned in Tennessee unless the end user ha																I
	more DS0 equivalent lines and is in Zone e of the Top 8 MSAs in BellSouth's region																I
1 of one	e of the Top 8 MSAs in BellSouth's region																
In th	he absence of ordered rates by a State																I
	ssion, the recurring rates for combination																
	loop network elements will the sum of the																l
	ing rates for the UNEs which make up the																1
combin	ations, and the nonrecurring rates shall b																
4	as set forth in this section.																<del>                                     </del>
	cal features shall apply to the Unbundled																I
	t/Loop Combination - Cost Based Rate																
	n in the same manner as they are applied Stand-Alone Unbundled Port section of thi																
to the S	Rate Exhibit.																I
End (	Office and Tandem Switching Usage and																
	nmon Transport Usage rates in the Port																l
	tion of this rate exhibit shall apply to all																l
	binations of loop/port network elements																l
	ot for UNE Coin Port/Loop Combinations h have a flat rate usage charge (USOC:																
WHICH	URECU).																1
	•																
	Currently Combined Combos in TN, the curring charges shall be those identified in																
	curring charges shall be those identified it inrecurring Charges - Currently Combined																
110 110	sections.																
	2 2 2 2 2 2																
2-WIRE	VOICE GRADE LOOP WITH 2-WIRE LIN	NE PORT (RES)															
$\perp$																	<del> </del>
UNE Po	ort/Loop Combination Rates	2-Wire Voice Grade Loop with 2		$\vdash$		1						1		<b> </b>			
2-V	Vire VG Loop/Port Combo - Zone 1	Wire Line Port - Zone 1		1			\$14.18										1
++-		2-Wire Voice Grade Loop with 2-		H			ψ. n. 10					l e		1			
2-V		Wire Line Port - Zone 2		2			\$18.01										1
	·	2-Wire Voice Grade Loop with 2															
2-V	Vire VG Loop/Port Combo - Zone 3	Wire Line Port - Zone 3		3			\$23.02										<b></b>
LINE:	an Datas																<b>+</b>
UNE LC	Dop Rates Vire Voice Grade Loop (SL1) - Zone		**	1	UEPRX	UEPLX	\$12.29										
2-V	Vire Voice Grade Loop (SL1) - Zone 2		**	2	UEPRX	UEPLX	\$16.12			1			1				
2-V	Vire Voice Grade Loop (SL1) - Zone :		**	3	UEPRX	UEPLX	\$21.13										
$\perp \!\!\! \perp \!\!\! \perp$	Voice Grade Line Port Rates (Res)																<u> </u>

								RATES					oss	RATES		
															Incremental Charge -	Incremen Charge
	UNBUNDLED NETWORK ELEMENT UNBUNDLED NETEORK ELEMENT		A 7- Zone	BCS	USOC						Svc Order	Svc Order	Incremental	Incremental	Manual Svc	Manual S
	STATED IN DOCKET 97-0126	01262							Nonre	ecurring	Submitted Elec	Submitted Manually per	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Order vs. Electronic-Disc	Order v
							Nonre	ecurring	Disc	onnect	per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'l
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAI
	Vire voice unbundled port - residence	**		UEPRX	UEPRL	\$1.89										
	Vire voice unbundled port with Caller ID															
res		**		UEPRX	UEPRC	\$1.89										
	Vire voice unbundled port outgoing only	**		LIEDDY	LIEDDO											
res	Vire voice Grade unbundled Tennessee		-	UEPRX	UEPRO	\$1.89										
	ended local dialing parity port with Caller															
	- res	**		UEPRX	UEPAQ	\$1.89										
2-V	Vire voice unbundled Tennessee Area					• • • • • • • • • • • • • • • • • • • •										
	lling port with Caller ID - res (F2R)	**		UEPRX	UEPAK	\$1.89										
	Vire voice unbundled Tennessee Area															
	lling port with Caller ID - res (TACER)	**		UEPRX	UEPAL	\$1.89										
	Vire voice unbundled Tennessee Area Iling port with Caller ID - res (TACSR)	**		UEPRX	UEPAM	\$1.89										
	Vire voice unbundled Tennessee Area		-	ULFIX	OLFAIVI	\$1.09	-				+					
	Iling port with Caller ID - res (1MF2X)	**		UEPRX	UEPAN	\$1.89										
	Vire voice unbundled Tennessee Area					ψσσ										
	lling port with Caller ID - res (2MR)	**		UEPRX	UEPAO	\$1.89										
	Vire voice unbundles res, low usage line															
por	t with Caller ID (LUM)	**		UEPRX	UEPAP	\$1.89										
			-													
FEATU	RES [Exchange Ports includes al						1				1					
ΔΙΙ	Available Vertical Features Applicable Features.	**		UEPRX	UEPVF	\$0.00	\$0.00	\$0.00								
ZIII.	Available vertical realures Applicable realures.]		-	OLI IXX	OLF VI	\$0.00	φ0.00	\$0.00								
LOCAL	NUMBER PORTABILITY		-													
	cal Number Portability (1 per port)	**	-	LIEPRX	LNPCX	\$0.35										
-	sar rumber r ertability (1 per perty		-	02.100	2.1.1	ψ0.00										
NONRE	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED		1				1				1					
2-V	Vire Voice Grade Loop / Line Port   2-Wire Voice Grade Loop/Li	ne														
Cor	mbination - Conversion - Switch-as-is Port Combo - Switch-as-is			UEPRX	USAC2		\$1.03	\$0.29					\$30.89	\$7.03		
	Vire Voice Grade Loop / Line Port															
	mbination - Conversion - Switch with	*						_						_		
	ange Vire Voice Grade Loop / Line Port 2-Wire Voice Grade Loop/Li			UEPRX	USACC		\$1.03	\$0.29					\$30.89	\$7.03		
	mbination - Conversion - Subsequent Port Combo - Subsequent	ie														
	tabase Update Database Update						\$0.76						\$7.97			
							ψο σ						<b>\$1.01</b>			
ADDITI	ONAL NRCs		1													1
2-V	Vire Voice Grade Loop/Line Port			1			1				1	1				1
Cor	mbination - Subsequent Activity	**		UEPRX	USAS2		\$10.00	\$10.00								
2-WIRE	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															
UNE Po	ort/Loop Combination Rates															1
	2-Wire Voice Grade Loop wi	ın 2				64440										
2-V	Vire VG Loop/Port Combo - Zone 1 Wire Line Port - Zone 1  2-Wire Voice Grade Loop wi	th 2	+1	-	ļ —	\$14.18				-	1		-			
2-W	Vire VG Loop/Port Combo - Zone 2 Wire Line Port - Zone 2	2	2			\$18.01										
	2-Wire Voice Grade Loop wi	th 2		1		ψ.σ.σι					1		l .			
2-V	Vire VG Loop/Port Combo - Zone 3 Wire Line Port - Zone 3		3			\$23.02										
	pop Rates															
0.14	Vire Voice Grade Loop (SL1) - Zone 1	**	1	UEPBX		\$12.29										
					1					1		i —				1
	Vire Voice Grade Loop (SL1) - Zone 2	**	3	UEPBX UEPBX	UEPLX	\$16.12										

								RATES					oss	RATES		
															Incremental	Incremental
	UNBUNDLED NETWORK ELEMENT UNBUNDLED NETEORK ELEMENT AS	Not in TRA Docket 97-	Zone	BCS	usoc					1	Svc Order	Svc Order	Incremental	Incremental	Charge - Manual Svc	Charge - Manual Svc
	STATED IN DOCKET 97-01262	01262	Zone	ВСЗ	0300				Nonre	ecurring	Submitted	Submitted	Charge - Manual	Charge - Manual	Order vs.	Order vs.
							Nonre	curring	Disc	connect	Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'l	Electronic-Disc 1st	Electronic-Dis Add'l
						Rec	First	Add'l	First	Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2	Wire Voice Grade Line Port (Bus)															
	2-Wire voice unbundled port without Calle															
	ID - bus	**		UEPBX	UEPBL	\$1.89										
	2-Wire voice unbundled port with Caller -	**														
	E484 ID - bus  2-Wire voice unbundled port outgoing only	**		UEPBX	UEPBC	\$1.89										
	bus	**		UEPBX	UEPBO	\$1.89										
	2-Wire voice Grade unbundled Tennessee			OLI DX	OLI DO	\$1.09										
	extended local dialing parity port with Caller															
	ID - bus	**		UEPBX	UEPAV	\$1.89										
	2-Wire voice unbundled incoming only por															
	with Caller ID - Bus	**		UEPBX	UPEB1	\$1.89										
	2-Wire voice unbundled Tennessee Bus 2															
	Way Area Calling Port Economy Option (TACC1)	**		UEPBX	UEPAC	£4.00										
	2-Wire voice unbundled Tennessee Bus 2		-	OEPBX	UEPAC	\$1.89					-		-			
	Way Area Calling Port Standard Option															
	(TACC2)	**		UEPBX	UEPAD	\$1.89										
	2-Wire voice unbundled Tennessee Bus 2															
	Way Collierville and Memphis Local Calling															
	Port (B2F)	**		UEPBX	UEPAE	\$1.89										
L	OCAL NUMBER PORTABILITY															
	Local Number Portability (1 per port)	**		UEPBX	LNPCX	\$0.35										
F	EATURES															
	[Exchange Ports includes all All Available Vertical Features Applicable Features.]	**		LIEDDY	UEPVF	\$0.00	<b>CO.00</b>	\$0.00								
	All Available vertical Features Applicable Features.]			UEPBA	UEPVF	\$0.00	\$0.00	\$0.00								
-	ONDEOLIDDING OLLADOEG (NIDO.) OLIDDENTI V COMPINED										+					
N	ONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED    2-Wire Voice Grade Loop / Line Port   2-Wire Voice Grade Loop/Line															
	Combination - Conversion - Switch-as-is  Port Combo - Switch-as-is			UEPBX	USAC2		\$1.03	\$0.29					\$30.89	\$7.03		
	2-Wire Voice Grade Loop / Line Port			02. 5/	00/102		ψ1.00	ψ0.20					ψου.σσ	ψ1.00		
	Combination - Conversion - Switch with															
	change	*		UEPBX	USACC		\$1.03	\$0.29					\$30.89	\$7.03		
	2-Wire Voice Grade Loop / Line Port 2-Wire Voice Grade Loop/Line															
	Combination - Conversion - Subsequent Database Update  Port Combo - Subsequent Database Update						\$0.76						\$7.97			
-	Database Opuate Database Opuate						\$0.76						\$7.97			
Α.	DDITIONAL NRCs										-		-			
^	2-Wire Voice Grade Loop/Line Port						<b>I</b>				1		1			
	Combination - Subsequent Activity	**		UEPBX	USAS2		\$10.00	\$10.00								
2	WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)															
1																
U	NE Port/Loop Combination Rates															
	2-Wire Voice Grade Loop with 2															
_	2-Wire VG Loop/Port Combo - Zone 1 Wire Line Port - Zone 1		1			\$14.18										
	2-Wire VG Loop/Port Combo - Zone 2  Wire VG Loop/Port Combo - Zone 2  2-Wire Voice Grade Loop with 2  Wire Line Port - Zone 2		اما			£40.04										
+	2-Wire VG Loop/Port Combo - Zone 2 Wire Line Port - Zone 2 2-Wire Voice Grade Loop with 2	1	2		1	\$18.01	<del>                                     </del>			1	1	-				
	2-Wire VG Loop/Port Combo - Zone 3 Wire Line Port - Zone 3		3			\$23.02										
+			H			ΨΕ0.0Ε							t			
,,	NE Loop Rates						1				1	1				<del>                                     </del>
U	2-Wire Voice Grade Loop (SL 1) - Zone 1	**	1	UEPRG	UEPLX	\$12.39				1	1	<b> </b>	1			-
+	2-Wire Voice Grade Loop (SL 1) - Zone 1	**		UEPRG		\$16.22					1		1			
-		**					<u> </u>				1		1			
- 1	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPRG	UEPLX	\$21.23						I				

INDICATE PROPERTY   CONTINUE OF THE PORT									RATES					OSSI	RATES		
MEANS DET NOTE AS EASTER   MARKED NET NOTE AND ADMINISTRATION   MARKED NET NOTE AND																	Incremental Charge -
Second   S	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97-	Zone	BCS	usoc				Manag						Manual Svc	Manual Svc Order vs.
2-Wire Visio Grade Line Pyri Rates (RES - PRX)		OTATED IN BOOKET 37-01202	01262			1		1			•	Elec	Manually per	Svc Order vs.	Svc Order vs.	Electronic-Disc	Electronic-Di
2							D										Add'I SOMAN
2-Wire VG Unbronded Contention 2-Way   PRX Trust Print Prior   Wire United States   Wire Un							Rec	FIISt	Add I	First	Add I	SOMEC	SUWAN	SOWAN	SOWAN	SUMAN	SUMAN
2-Wine VG Unborned Combination (2-Wine VG Unborned V	2-Wire Voice Grade Line Port Rates (RES - PE	3X)															-
PPEX Trunce Port - Res		 															-
Local Number Partability (1 per port)			**		UEPRG	UEPRD	\$1.79										
Local Number Partability (1 per port)																	
Section   Sect	LOCAL NUMBER PORTABILITY																
All Available Vertical Features	Local Number Portability (1 per port)		**		UEPRG	LNPCP	\$3.15										
All Available Ventral Features   Exchange Ports Include all Applicable Features																	
All Available Vertical Features	FEATURES	(Cush as as Danta in aluda a all															
2Wire Voice Grade Loop Line Port   Voice Grade Loop Line   Voice Grade Line Loop Line   Voice Grade Loop Line   Voice Grade Loop Line   Voice Grade Loop Line   Voice Grade Loop Line   Voice Grade Loop Line   Voice Grade Loop Line   Voice Grade Loop Line   Voice Grade Loop Line   Voice Line Loop Line   Voice Grade Loop Line   Voice Grade Loop Line   Voice Line Line Line Line Line Line Line Lin	All Available Vertical Features		**		UEPRG	UEPVF	\$0.00	\$0.00	\$0.00								
Combination (PBX) - Conversion - Switch-Agi-Wire Viole Grade Loop Line   Pint Combo - Switch-Agi-Wire Viole Grade Loop Line Port Combo - Switch-Agi-Wire Viole Grade Loop Line Port Combo - Switch-Agi-Wire Viole Grade Loop Line Port Combo - Switch-Agi-Wire Viole Grade Loop Line Port Combo - Switch-Agi-Wire Viole Grade Loop Line Port Combo - Switch-Agi-Wire Viole Grade Loop Line Port Combo - Switch-Agi-Wire Viole Grade Loop Line Port Combo - Switch-Agi-Wire Viole Grade Loop Line Port Combo - Switch-Agi-Wire Viole Grade Loop Line Port Combo - Switch-Agi-Wire Viole Grade Loop Line Port Combo - Switch-Agi-Wire Viole Grade Loop Line Port (BUS - PBX)   Wire Viole Grade Loop Wire Line Port - Zone 1   Wire Line Port - Zone 2   Wire Line Port - Zone 2   Wire Line Port - Zone 2   Wire Line Port - Zone 2   Wire Line Port - Zone 2   Wire Line Port - Zone 2   Wire Line Port - Zone 2   Wire Line Port - Zone 3   Wire Line Port - Zone 2   Wire Line Port - Zone 3   Wire Line Port - Zone 2   Wire Line Port - Zone 3   Wire Line Port - Zone 2   Wire Line Port - Zone 3   Wire Line Port - Zone 2   Wire Line Port - Zone 3   Wire Line Port -		ENTLY COMBINED															
Section   Port Combo - Switch-ae-is   UEPRG USAC2   \$1.03   \$0.29   \$30.89   \$7.03		s2-Wire Voice Grade Loop/Line															
Combination (PBX) - Conversion - Switch with Change   2-Wire Voice Grade Loop/ Line Port Combio - Subsequent Database Update   Database	Is				UEPRG	USAC2		\$1.03	\$0.29	<u> </u>				\$30.89	\$7.03		<u></u>
With Change   2-Wire Voice Grade Loop / Line Port   Combination - Conversion - Subsequent   Database Update Update Up																	
2-Wire Voice Grade Loop / Line Port Combons - Subsequent			*		LIEPRG	USACC		\$1.03	\$0.29					\$30.80	\$7.03		
Database Update		2-Wire Voice Grade Loop/Line			OLI KO	00/100		Ψ1.03	Ψ0.23					ψ50.05	Ψ1.03		
ADDITIONAL NRCS   2-Wire Voice Grade Lopy Line Port																	
2-Wire Voice Grade Loop Line Port (BUS - PBX)   UEPRG USAS2   S10.00   S10.00   S30.89   S7.03	Database Update	Database Update						\$0.76						\$7.97			
2-Wire Voice Grade Loop Line Port   Sassa	ADDITIONAL NDCo																
2-Wire Voice Grade Loop (St. 1) - Zone 2 2-Wire Voice Grade Loop (St. 1) - Zone 3 3																	
UNE Port/Loop Combination Rates   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop (St. 1) - Zone 2   2-Wire Voice Grade Loop (St. 1) - Zone 2   3-Z-Wire Voice Grade Loop (St. 1) - Zone 2   3-Z-Wire Voice Grade Loop (St. 1) - Zone 2   3-Z-Wire Voice Grade Loop (St. 1) - Zone 3   3-Z-Wire Voice Unbundled Off Unward PBX Trush Port - Bus   UEPPX UE	Combination (PBX) - Subsequent Activity		**		UEPRG	USAS2		\$10.00	\$10.00					\$30.89	\$7.03		
UNE Port/Loop Combination Rates   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop with 2   2-Wire Voice Grade Loop (St. 1) - Zone 2   2-Wire Voice Grade Loop (St. 1) - Zone 2   3-Z-Wire Voice Grade Loop (St. 1) - Zone 2   3-Z-Wire Voice Grade Loop (St. 1) - Zone 2   3-Z-Wire Voice Grade Loop (St. 1) - Zone 3   3-Z-Wire Voice Unbundled Off Unward PBX Trush Port - Bus   UEPPX UE																	
2-Wire VG Loop/Port Combo - Zone 1   Wire Line Port - Zone 1   1   \$14.18	2-WIRE VOICE GRADE LOOP WITH 2-WIRE L	INE PORT (BUS - PBX)															
2-Wire VG Loop/Port Combo - Zone 1   Wire Line Port - Zone 1   1   \$14.18	LINE Port/Loop Combination Pates																-
2-Wire VG Loop/Port Combo - Zone 1	ONE FOR LOOP COmbination Rates	2-Wire Voice Grade Loop with 2															
2-Wire VG Loop/Port Combo - Zone 2	2-Wire VG Loop/Port Combo - Zone 1			1			\$14.18										
2-Wire Voice Grade Loop with 2   Wire Line Port - Zone 3   Wire Line Port - Zone 3   3   \$23.02																	
2-Wire VG Loop/Port Combo - Zone 3   Wire Line Port - Zone 3   3   \$23.02	2-Wire VG Loop/Port Combo - Zone 2			2			\$18.01										
UNE Loop Rates   2-Wire Voice Grade Loop (St. 1) - Zone   ** 1 UEPPX UEPLX \$12.39	2-Wire VG Loop/Port Combo - Zone 3			3			\$23.02										
2-Wire Voice Grade Loop (SL 1) - Zone   **	·						*										
2-Wire Voice Grade Loop (St. 1) - Zone 2																	
2-Wire Voice Grade Loop (SL 1) - Zone 3																	
2-Wire Voice Grade Line Port Rates (BUS - PBX)    Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus																	-
Line Side Unbundled Combination 2-Way	2-wire voice Grade Loop (SL 1) - Zone 3			3	UEPPX	UEPLX	\$21.23										-
Line Side Unbundled Combination 2-Way	2-Wire Voice Grade Line Port Rates (BUS - PE	J BX)										1					
Line Side Unbundled Outward PBX Trunk	Line Side Unbundled Combination 2-Way																
Port - Bus			**		UEPPX	UEPPC	\$1.79										
Line Side Unbundled Incoming PBX Trunk			**		∏EPD∀	HEPP∩	\$1.70										
Port - Bus					JEITA		φ1.10										
Ports	Port - Bus		**		UEPPX	UEPP1	\$1.79										
2-Wire Voice Unbundled 2-Way Combination			**		LIEBBY	LIED! D	04 ===										
PBX Tennessee Calling Port         **         UEPPX         UEPT2         \$1.79           2-Wire Voice Unbundled 1-Way Outgoinç         **         UEPPX         UEPTO         \$1.79           PBX Tennessee Calling Port         **         UEPPX         UEPTO         \$1.79		OI.			UEPPX	UEPLD	\$1.79	-		-							1
2-Wire Voice Unbundled 1-Way Outgoinç PBX Tennessee Calling Port ** UEPPX UEPTO \$1.79			**		UEPPX	UEPT2	\$1.79										
I BY TOTILICOCCO COMMING FOR	2-Wire Voice Unbundled 1-Way Outgoing																
O Missa Vision High we fleet we fleet to a first to the f			**		UEPPX	UEPTO	\$1.79					1					1
2-Wire Voice Unbundled 2-Way Combination PBX Usage Port  ** UEPPX UEPX \$1.79		OI .	**		I IEDD^	HEDVA	¢1 70										

							RATES					oss	RATES		
				ĺ										Incremental Charge -	Incrementa Charge -
UNBUNDLED NETWORK ELEMENT UNBUNDLED NETEORK ELEMENT AS	Not in TRA Docket 97- Z	one	BCS	USOC						Svc Order	Svc Order	Incremental	Incremental	Manual Svc	Manual Svo
STATED IN DOCKET 97-01262	01262		200			4		Nonre	ecurring	Submitted Elec	Submitted Manually per	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Order vs. Electronic-Disc	Order vs. Electronic-D
						Nonre	curring	Disc	onnect	per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'l
					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Wire Voice Unbundled PBX Toll Termina															
Hotel Ports	**	ı	UEPPX	UEPXB	\$1.79										
2-Wire Voice Unbundled PBX LD DDD Terminals Port	**	Ι.	UEPPX	UEPXC	64.70										
2-Wire Voice Unbundled PBX LD Terminal		- 1	UEPPA	UEFAC	\$1.79										
Switchboard Port	**	ı	UEPPX	UEPXD	\$1.79										
2-Wire Voice Unbundled PBX LD Terminal					<b>\$0</b>										
Switchboard IDD Capable Port	**	ι	UEPPX	UEPXE	\$1.79										
2-Wire Voice Unbundled 2-Way PBX															
Hotel/Hospital Economy Administrative															
Calling Port	**	Ų	UEPPX	UEPXL	\$1.79										
2-Wire Voice Unbundled 2-Way PBX	**	Ι.	LIEDDY	UEPXM	64.70										
Hotel/Hospital Economy Room Calling Port  2-Wire Voice Unbundled 1-Way Outgoing			UEPPX	DEPAIN	\$1.79	-									
PBX Hotel/Hospital Economy Admin Calling															
Port TN Calling Port	**	l	UEPPX	UEPXN	\$1.79										
2-Wire Voice Unbundled 1-Way Outgoinç					*****										
PBX Hotel/Hospital Discount Room Calling															
Port	**	ı	UEPPX	UEPXO	\$1.79										
2-Wire Voice Unbundled 1-Way Outgoing	**														
PBX Measured Port	**	Ų	UEPPX	UEPXS	\$1.79										
2-Wire Voice Unbundled PBX Collierville and Memphis Calling Port	**	Ι.	UEPPX	UEPXU	\$1.79										
2-Wire Voice Unbundled 2-Way PBX			ULFFA	ULFXU	\$1.79					-					
Tennessee RegionServ Callling Port	**	l l	LIEPPX	UEPXV	\$1.79										
					<b>\$</b> •										
LOCAL NUMBER PORTABILITY															
Local Number Portability (1 per port)	**	٠,	LIEDDY	LNPCP	\$3.15										
Local Number Fortability (1 per port)		- 1	OLITA	LIVI OI	φ3.13										
FEATURES	_					1				+					
[Exchange Ports includes all															
All Available Vertical Features Applicable Features.]	**	ι	UEPPX	UEPVF	\$0.00	\$0.00	\$0.00								
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
2-Wire Voice Grade Loop/ Line Port															
Combination (PBX) - Conversion - Switch-As 2-Wire Voice Grade Loop/Line															
Is Port Combo - Switch-as-is		ı	UEPPX	USAC2		\$1.03	\$0.29					\$30.89	\$7.03		
2-Wire Voice Grade Loop/ Line Port															
Combination (PBX) - Conversion - Switch with Change	*	Ι.	LIEDDY	USACC		24.00	•					***	•		
2-Wire Voice Grade Loop / Line Port 2-Wire Voice Grade Loop/Line	-		UEPPX	USACC		\$1.03	\$0.29					\$30.89	\$7.03		
Combination - Conversion - Subsequent Port Combo - Subsequent															
Database Update Database Update						\$0.76						\$7.97			
ADDITIONAL NRCs										1					
2-Wire Voice Grade Loop/ Line Port															
Combination (PBX) - Subsequent Activity	**	ι	UEPPX	USAS2		\$10.00	\$10.00					\$30.89	\$7.03		
2-WIRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT															
UNE Port/Loop Combination Rates															
2-Wire VG Loop/2-Wire DID Trunk Port 2-Wire Voice Grade Loop with 2		, [			040.00										
Combo - Zone 1 Wire DID Trunk Port - Zone 1  2-Wire VG Loop/2-Wire DID Trunk Port 2-Wire Voice Grade Loop with 2		1			\$18.38										
Combo - Zone 2		2			\$19.87										
2-Wire VG Loop/2-Wire DID Trunk Port 2-Wire Voice Grade Loop with 2		-			\$19.87										
Combo - Zone 3 Wire DID Trunk Port - Zone 3		3			\$25.52										
11.00	1	- 1			Ψ <b>-</b> 0.0 <b>-</b>					1	1				

				1 1					RATES					OSS	RATES		
														1		Incremental	Incremental
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS	Not in TRA Docket 97-	1_	BCS	USOC						Svc Order	Svc Order	Incremental	Incremental	Charge - Manual Svc	Charge - Manual Svc
	UNBUNDLED NETWORK ELEMENT	STATED IN DOCKET 97-01262	Docket 97- 01262	Zone	BCS	USOC				Nonre	ecurring	Submitted	Submitted	Charge - Manual	Charge - Manual	Order vs.	Order vs.
			*****					Nonre	curring	Diec	onnect	Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'l	Electronic-Disc 1st	Electronic-Dis Add'l
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
NC	NRECURRING CHARGES - CURRENTLY	COMBINED						1				1					
+		2-Wire Voice Grade Loop with 2															
	2-Wire Voice Grade Loop / 2-Wire DID	Wire DID Trunk Port Combo -															
	Trunk Port Combination - Switch-as-is	Switch-as-is			UEPPX	USAC1		\$8.76	\$5.75					\$41.43	\$9.80		
	2-Wire Voice Grade Loop / 2-Wire DID																
	Trunk Port Combination Conversion with		*		LIEDDY	USA1C									••••		
+	Changes		-		UEPPX	USATC		\$8.76	\$5.75			\$3.50		\$41.43	\$9.80	\$20.00	\$20.00
+-	lankana Nimekan/Tumb Casus Fatabliana																
	lephone Number/Trunk Group Establisme	nt															
- Cn	DID Trunk Termination (One Per Port)		**		UEPPX	NDT	60.00	<b>#</b> 0.00	<b>#</b> 0.00			00.50		044.40	00.00	600.00	600.00
+	DID Numbers, Establish Trunk Group and				UEPPX	NDT	\$0.00	\$0.00	\$0.00			\$3.50		\$41.43	\$9.80	\$20.00	\$20.00
	Provide First Group of 20 DID Numbers		**		UEPPX	NDZ	\$0.00	\$0.00	\$0.00			\$3.50		\$41.43	\$9.80	\$20.00	\$20.00
+	Additional DID Numbers for each Group of						ψ0.00	ψ0.00	ψ0.00			ψ0.00		Ųo	\$0.00	<b>Q2</b> 0.00	Ψ <u></u> 20.00
	20 DID Numbers		**		UEPPX	ND4	\$0.00	\$0.00	\$0.00			\$3.50		\$41.43	\$9.80	\$20.00	\$20.00
	DID Numbers, Non- consecutive DID						-										
	Numbers , Per Number		**		UEPPX	ND5	\$0.00	\$0.00	\$0.00			\$3.50		\$41.43	\$9.80	\$20.00	\$20.00
$\perp$																	
LO	CAL NUMBER PORTABILITY																
	Local Number Portability (1 per port)		**		UEPPX	LNPCP	\$3.15	\$0.00	\$0.00								
2-V	VIRE ISDN DIGITAL GRADE LOOP WITH 2	-WIRE ISDN DIGITAL LINE SIDE PO	RT														
UN	IE Port/Loop Combination Rates							1									
		2-Wire ISDN Digital Grade Loop															
	2W ISDN Digital Grade Loop/2W ISDN	with 2-Wire ISDN Digital Line Side	1	١. ١													
+	Digital Line Side Port - Zone 1	Port - Zone 1 2-Wire ISDN Digital Grade Loop		1			\$32.27					1		\$19.99	\$19.99	\$20.00	\$20.00
	2W ISDN Digital Grade Loop/2W ISDN	with 2-Wire ISDN Digital Line Side															
	Digital Line Side Port - Zone 2	Port - Zone 2		2			\$34.78							\$19.99	\$19.99	\$20.00	\$20.00
+	3	2-Wire ISDN Digital Grade Loop					*******							¥10100	7.0.00	<b>V</b>	<b>V</b>
	2W ISDN Digital Grade Loop/2W ISDN	with 2-Wire ISDN Digital Line Side															
	Digital Line Side Port - Zone 3	Port - Zone 3		3			\$44.32							\$19.99	\$19.99	\$20.00	\$20.00
NC	NRECURRING CHARGES - CURRENTLY																
	2-Wire ISDN Digital Grade Loop / 2-Wire	2-Wire ISDN Digital Grade Loop 2															
	ISDN Line Side Port Combination -	Wire ISDN Line Side Port Combo -	1						_					_	_	_	_
-	Conversion	Switch-as-is			UEPPB	USACB		\$117.23	\$117.23			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
												1					
AD	DITIONAL NRCs							-									
	2-Wire ISDN Digital Grade Loop / 2-Wire	2-Wire ISDN Digital Grade Loop 2-															
	ISDN Line Side Port Combination-Non	Wire ISDN Line Side Port Combo -															
	Feature Subs Activity	Non Feature Subsequent Activity			UEPPB	USASB		\$212.88				\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
LO	CAL NUMBER PORTABILITY																
	Local Number Portability (1 per port)		**		UEPPR	LNPCX	\$0.35	\$0.00	\$0.00								
B-1	CHANNEL USER PROFILE ACCESS:							1				1					
	CVS/CSD (DMS/5ESS)		**		UEPPB	U1UCA	\$0.00	\$0.00	\$0.00								
$\top$	CVS (EWSD)		**		UEPPB	U1UCB	\$0.00	\$0.00	\$0.00								
$\top$	CSD		**		UEPPB	U1UCC	\$0.00	\$0.00	\$0.00								
$\top$								1				1					
	OLIANDEL AREA RILIGUER RROEL E	1	<b>†</b>					1				1		1			
B-0	CHANNEL AREA PLUS USER PROFILE																
	CHANNEL AREA PLUS USER PROFILE CESS: (AL,KY,LA,MS SC,MS, & TN)																

								RATES					oss	RATES		
															Incremental	Incremental
	UNBUNDLED NETWORK ELEMENT UNBUNDLED NETEORK ELEMENT AS	Not in TRA Docket 97-	Zone	BCS	usoc						Svc Order	Svc Order	Incremental	Incremental	Charge - Manual Svc	Charge - Manual Svo
	STATED IN DOCKET 97-01262	01262	Zone	ВСЗ	0300				Nonre	ecurring	Submitted Elec	Submitted	Charge - Manual	Charge - Manual	Order vs.	Order vs.
							Nonre	curring	Disc	onnect	per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'l	Electronic-Disc 1st	Electronic-Dis Add'l
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	CVS (EWSD)	**		UEPPB	U1UCE	\$0.00	\$0.00	\$0.00								
	CSD	**		UEPPB	U1UCF	\$0.00	\$0.00	\$0.00								
U	SER TERMINAL SERVICE PROFILE (EWSD	**		UEPPB	U1UMA	\$0.00	\$0.00	\$0.00								
V	ERTICAL FEATURES	**														
	One per Channel B User Profile	**		OEPPB	UEPVF	\$0.00	\$0.00	\$0.00								
٠.	WIRE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT															
4-	WIRE DST DIGITAL LOOP WITH 4-WIRE ISDN DST DIGITAL TRUNK PORT															
	NE Port/Loop Combination Rates										-					
-	4-Wire DS1 Digital Loop with 4															+
	4W DS1 Digital Loop/4W ISDN DS1 Digital Wire ISDN DS1 Digital Trunk Por	t -														
	Trunk Port - Zone 1 Zone 1		1			\$132.58							\$19.99	\$19.99	\$20.00	\$20.00
	4-Wire DS1 Digital Loop with 4															
	4W DS1 Digital Loop/4W ISDN DS1 Digital Wire ISDN DS1 Digital Trunk Por Trunk Port - Zone 2	11	2			\$150.25					Ī		\$19.99	\$19.99	\$20.00	\$20.00
-	4-Wire DS1 Digital Loop with 4					φ100.20	1				1		φ।ਰ.ੲੲ	φ13.33	φ∠0.00	φ20.00
	4W DS1 Digital Loop/4W ISDN DS1 Digital Wire ISDN DS1 Digital Trunk Por	t-														
	Trunk Port - Zone 3 Zone 3		3			\$173.44							\$19.99	\$19.99	\$20.00	\$20.00
N	ONRECURRING CHARGES - CURRENTLY COMBINED															
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 4-Wire DS1 Digital Loop with 4															
	Digital Trunk Port Combination - Conversion Wire ISDN DS1 Digital Trunk Por Switch-as-is Combo - Switch-as-is	τ		LIEDDD	USACP		\$328.53	\$328.53					\$19.99	\$19.99	\$20.00	\$20.00
$\dashv$	OWIGH-43-13			OLITI	OOACI		φ320.33	φ320.33					\$19.99	\$15.55	φ20.00	\$20.00
_	DDITIONAL NRCs															-
А	DDITIONAL NRCS															
	4-Wire DS1 Digital Loop with 4-															
	4-Wire DS1 Dig Loop / 4-Wire ISDN DS1 Wire ISDN DS1 Digital Trunk Por	t														
	Dig Trunk Port Combination-Sub Channel Combo - Subsequent Channel															
	Activation-Per Channel Activation - Per Channel			UEPPP	USASP		\$28.39						\$19.99	\$19.99	\$20.00	\$20.00
	4-Wire DS1 Digital Loop with 4-															
	4-Wire DS1 Dig Loop / 4-Wire ISDN DS1 DigWire ISDN DS1 Digital Trunk Por Trunk Port Combination-Sub Inward/2-Way Combo - Subsequent Inward/2-	τ														
	Telephone Numbers way Telephone Numbers			UEPPP	PR7TG		\$0.94						\$19.99	\$19.99	\$20.00	\$20.00
+	4-Wire DS1 Digital Loop with 4						₩J.0¬				1		<b>\$.3.55</b>	ψ.σ.σσ	ψ <u>-</u> 3.00	<b>\$20.00</b>
	4-Wire DS1 Dig Loop / 4-Wire ISDN DS1 Dig Wire ISDN DS1 Digital Trunk Por	t									Ī					
	Trunk Port Combination-Sub Outward  Combo - Subsequent Outward			LIEBSE	DD====		000						0.6			
+	Telephone Numbers Telephone Numbers 4-Wire DS1 Digital Loop with 4			UEPPP	PR7TP		\$22.36				1		\$19.99	\$19.99	\$20.00	\$20.00
	4-Wire DS1 Dig Loop / 4-Wire ISDN DS1 DigWire ISDN DS1 Digital Ecop with 4	t									Ī					
	Trunk Port Combination-Subsequent Inward Combo - Subsequent Inward															
	Telephone Numbers Telephone Numbers			UEPPP	PR7ZT		\$44.71						\$19.99	\$19.99	\$20.00	\$20.00
T	4-Wire DS1 Digital Loop with 4															
	4-Wire DS1 Dig Loop / 4-Wire ISDN DS1 DigWire ISDN DS1 Digital Trunk Por Trunk Port Combination-Subsequent Service Combo - Subsequent Service	τ									Ī					
	Order Per Order Order			UEPPP	USASP		\$189.76						\$19.99	\$19.99	\$20.00	\$20.00
+	5.55. 5. 5.441				30, 101		\$100.70						Ψ10.00	Ψ10.00	Ψ20.00	Ψ20.00
14	OCAL NUMBER PORTABILITY			1	1		1				1	1				1
	Local Number Portability (1 per port)	**		UEPPP	LNPCN	\$1.75										
	Crasmy (1 por port)			22	2.1. 011	ψ1.70										
IN	ITERFACE (Provsioning Only)															
	Voice/Data	**		UEPPP	PR71V	\$0.00					1	1				
	Digital Data	**		UEPPP		\$0.00										

			1					RATES			1		OSS	RATES		
								RAILO					000	LATES	Incremental	Incremental
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS	Not in TRA	_	BCS	USOC						Svc Order	Svc Order	Incremental	Incremental	Charge - Manual Svc	Charge - Manual Svc
UNBUNDLED NETWORK ELEMENT	STATED IN DOCKET 97-01262	Docket 97- 01262	Zone	BCS	USOC				Nonrec	curring	Submitted	Submitted	Charge - Manual	Charge - Manual	Order vs.	Order vs.
							Nonre	curring	Disco	nnect	Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'l	Electronic-Disc 1st	Electronic-Disc Add'I
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Inward Data		**		UEPPP	PR71E	\$0.00										
CALL TYPES																
Inward		**		UEPPP	PR7C1	\$0.00										
Outward		**		UEPPP	PR7C0	\$0.00										
Two-way		**		UEPPP	PR7CC	\$0.00										
4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS	S TRUNK PORT															
UNE Port/Loop Combination Rates													1	1		
4W DS1 Digital Loop/4W DDITS Trunk Port 4																
	Wire DID Trunk Port - Zone 1		1			\$93.28										
4W DS1 Digital Loop/4W DDITS Trunk Port 4													1	, J		
	Wire DID Trunk Port - Zone 2		2			\$110.95										
4W DS1 Digital Loop/4W DDITS Trunk Port 4						_							1	, J		
Zone 3 V	Wire DID Trunk Port - Zone 3		3			\$134.14										
NONRECURRING CHARGES - CURRENTLY COM	MBINED 4-Wire DS1 Digital Loop/4-Wire															
	i-wire DST Digital Loop/4-wire DID Trunk Port Combo - Switch-as												1	, J		
Trunk Port Combination - Switch-as-is	on Trunk Port Combo - Switch-as	1		HEDDO	USAC4		\$312.91	\$312.91			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
4-Wire DS1 Digital Loop / 4-Wire DDITS	5		-	UEPDC	U3AC4		\$312.91	\$312.91			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Trunk Port Combination - Conversion with													1	, J		
DS1 Changes		*		LIEPDC	USAWA		\$312.91	\$312.91			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
	4-Wire DS1 Digital Loop/4-Wire			OLI DO	00/11//1		ψ312.91	ψ312.31			ψ3.30		Ψ13.33	Ψ19.99	Ψ20.00	Ψ20.00
	OID Trunk Port Combo - Switch-as												1	, J		
Change - Trunk is	s			UEPDC	USAWB		\$312.91	\$312.91			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
															-	
ADDITIONAL NRCs																
	4-Wire DS1 Digital Loop 4-Wire												1			
4-Wire DS1 Dig Loop / 4-Wire DDITS Trunk D													1	, J		
	Subsequent Service Order Per												1	, J		
	Order			UEPDC	USAS4		\$94.88	\$94.88			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
	4-Wire DS1 Digital Loop/4-Wire												1	, J		
4-Wire DS1 Dig Loop/4-Wire DDITS Trk Port D Combination-NRC-Sub Channel Activation-													1	, J		
	Per Channel			HEDDO	UDTTA		\$108.67				\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
	4-Wire DS1 Digital Loop/4-Wire			OLFDC	JULIA		φ100.07				φ3.50		क्।इ.इइ	क्।अ.अअ	<b>⊅∠∪.∪∪</b>	φ∠U.UU
4-Wire DS1 Dig Loop/4-Wire DDITS Trk Port													1	, J		
Combo-Sub Channel Activation-Per Chanl-1S													1	, J		
	Per Channel			UEPDC	UDTTB		\$108.67				\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
4	4-Wire DS1 Digital Loop/4-Wire						1				1		7.2.22		<del></del>	<del></del>
4-Wire DS1 Dig Loop/4-Wire DDITS Trk Port D											1		1	, ,		
	Subsequent Channel Activation												1	, ,		
	Per Channel			UEPDC	UDTTC		\$108.67				\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
	4-Wire DS1 Digital Loop/4-Wire						1				1		ı ¬	, 7		
4-Wire DS1 Dig Loop/4-Wire DDITS Trk PortD											I		1	, J		
	Subsequent Channel Activation			LIEBBO	LIDTTE		0400 05				00.70		040.00	046.55	000.00	000.00
	Per Channel 4-Wire DS1 Digital Loop/4-Wire		H	UEPDC	UDTTD		\$108.67				\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
4-Wire DS1 Dig Loop/4-Wire DDITS Trk Port											1		1	, ,		
	Subsequent Channel Activation												1	, ,		
	Per Channel			UEPDC	UDTTE		\$108.67				\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
		0		25, 50	2011L		ψ100.07	1			ψ3.50	1	Ψ10.00	ψ10.33	Ψ20.00	Ψ20.00
4-VVICE US LUIGURALLOOD / 4-VVICE LILLIS IA													•	' h		
	4-Wire DS1 Digital Loop 4-Wire DID Trunk Port Combo -												1			

Married St. Organization   Company									RATES					OSS	RATES		
Second Control   Seco			Net in TDA														
A	UNBUNDLED NETWORK ELEMENT		Docket 97-	Zone	BCS	usoc				Nonro	ourring					Manual Svc	
### AVER DRI TIDIDIA LOUGE / AVERS DRITTS Trust Port Combination - Subsequent DRITT Trust Port DRITT Trust Port Combination - Subsequent DRITT Trust Port Co			01262									Elec	Manually per	Svc Order vs.	Svc Order vs.	Electronic-Disc	Electronic-Disc
A-Vive DST Digital Looy 1 - Vive DST Digital Loop 1 - Vive DST Digital Loop 1 - Vive DST Digital Loop 1 - Vive DST Digital Loop 1 - Vive DST Digital Loop 1 - Vive DST Digital Loop 1 - Vive DST Digital Loop 4 - Vive DST Digit							Rec										
Suncia Confession	4 Wire DS4 Digital Loop / 4 Wire DDITS																
Service Order For Order   Order   UEPOC   SAS4   S94.88   S93.00   S19.99   S19.99   S19.90   S20.00																	
Digital Trusk Prior Combonation - Subsequent Telephone Numbers   Subsequent Telephone Numbe					UEPDC	USAS4		\$94.88	\$94.88			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Digital Trusk Prior Combonation - Subsequent Telephone Numbers   Subsequent Telephone Numbe	4 W/ PO4 Pin Lang /4 W/ IODN PO4	4 M// DO4 D/-// 4 M//															
Telephone Number   Subsequent Pelephone Numbers   See   Se																	
Superframe Format - Conversion on New Install   Superframe Format - Conversion   VEPDC   COSEF   \$0.00   \$0.					UEPDC			\$88.68				\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Install																	
Common			**		LIEDDO	00005	•		***								
or New Install   Frame   Superframe   Supe			**		UEPDC	CCOSF	\$0.00	\$0.00	\$0.00								
Second Spiritrame Format			**		UEPDC	CC0EF	\$0.00	\$0.00	\$0.00								
Second Spiritrame Format																	
Alternate Mark Inversion			**		UEPDC	CCOSF	\$0.00	\$590.00	\$590.00			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Alternate Mark Inversion			**		UEPDC	CCOEF	\$0.00	\$590.00	\$590.00			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Superframe Format							***	,	,			,			,		
Estended SuperFrame Format	Alternate Mark Inversion																
Extended SuperFrame Format	Superframe Format		**		LIEPDC	MCOSE		\$0.00	\$0.00								
Telephone Number for 2-Way Trunk Group   **   UEPOC   UDTGX   \$0.00   \$3.50   \$19.99   \$19.99   \$20.00   \$20.	Cupername i ormat				OLI DO	WOOOI		Ψ0.00	Ψ0.00								
Charges	Extended SuperFrame Format		**		UEPDC	MCOPO		\$0.00	\$0.00								
Telephone Number for 2-Way Trunk Group   " UEPDC UDTGX \$0.00   \$3.50   \$19.99   \$19.99   \$20.00   \$20.00																	
Telephone Number for 1-Way Outward			**		LIEDDC	LIDTGY	\$0.00					\$2.50		¢10.00	\$10.00	\$20.00	\$20.00
Telephone Number for 1-Way Inward Trunk   Corp.   Co					OLI DO	ODIGA	φ0.00					φ3.50		φ19.99	φ13.33	\$20.00	φ20.00
Group Without DID			**		UEPDC	UDTGY	\$0.00					\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
DID Numbers   Establish Trunk Group and   Provide First Group of 20 DID Numbers   Provide First Provide			**		LIEDDC	LIDTGZ	\$0.00					\$2.50		¢10.00	\$10.00	\$20.00	\$20.00
Provide First Group of 20 DID Numbers					OLFDC	UDIGZ	\$0.00					\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
20 DID Numbers			**		UEPDC	NDZ	\$0.00					\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Did Numbers, Non-consecutive DID   Numbers, Non-consecutive DID   Numbers, Per Number   Non-consecutive DID   Numbers, Per Number   Non-consecutive DID   Numbers, Per Number   Non-consecutive DID   Numbers, Per Number   Non-consecutive DID   Numbers, Per Number   Non-consecutive DID   Numbers, Per Number   Non-consecutive DID   Numbers, Per Number   Non-consecutive DID   Numbers, Per Number   Non-consecutive DID   Numbers, Per Number   Non-consecutive DID   Numbers, Per Number   Non-consecutive DID   Numbers, Per Number   Numbers, Per Number   Numbers, Per Number   Numbers, Per Number   Numbers, Per Number   Numbers, Per Number   Numbers, Per Number   Numbers, Per Number   Numbers, Per Number   Numbers, Per Number   Numbers, Per Number   Numbers, Per Number   Numbers, Per Number   Numbers, Per Number   Numbers, Per Numbers,			**		LIEDDO		•							***	0.000		***
Numbers   Per Number					UEPDC	ND4	\$0.00					\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
FX/FCO for 4-Wire DSI Digital Loop with 4-Wire DDITS Trunk Port   Fixed rate 0-1   Fixed Port			**		UEPDC	ND5	\$0.00					\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Interoffice Channel Mileage - Fixed rate 0-{   miles (Facilities Termination)   ** UEPDC   LINO1   \$75.83   \$3.50   \$19.99   \$19.99   \$20.00   \$20.00     Interoffice Channel Mileage - Additional rate per mile - 0-8 miles (Facilities Termination)   ** UEPDC   LINOA   \$0.3525   \$3.50   \$19.99   \$19.99   \$20.00   \$20.00     Interoffice Channel Mileage - Fixed rate 9-2! miles (Facilities Termination)   ** UEPDC   LINO2   \$0.0000       Interoffice Channel Mileage - Additional rate per mile - 9-25 miles (Facilities Termination)   ** UEPDC   LINOB   \$0.3525   \$3.50   \$19.99   \$19.99   \$20.00   \$20.00     Interoffice Channel Mileage - Fixed rate 25- miles (Facilities Termination)   ** UEPDC   LINOB   \$0.3525   \$3.50   \$19.99   \$19.99   \$20.00   \$20.00     Interoffice Channel Mileage - Additional rate per mile - 25+ miles (Facilities Termination)   ** UEPDC   LINOS   \$0.0000   \$0.000   \$0	Dedicated DS1 (Interoffice Channel Mileage) -							-				1					<del>                                     </del>
Interoffice Channel Mileage - Fixed rate 0-{   miles (Facilities Termination)																	
miles (Facilities Termination)			-									1					
Interoffice Channel Mileage - Additional rati			**		UEPDC	1LNO1	\$75.83					\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Interoffice Channel Mileage - Fixed rate 9-25	Interoffice Channel Mileage - Additional rate																
miles (Facilities Termination)			**		UEPDC	1LNOA	\$0.3525					\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Interoffice Channel Mileage - Additional rati			**		UEPDC	1LNO2	\$0,0000										
Interoffice Channel Mileage - Fixed rate 25+	Interoffice Channel Mileage - Additional rate						ψ0.0000										
miles (Facilities Termination)			**		UEPDC	1LNOB	\$0.3525					\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Interoffice Channel Mileage - Additional rate			**		LIEPDO	1I NO3	\$0,000										
Local Number Portability, per DS0 Activated							ψυ.υυυ										
Central Office Termininating Point ** UEPDC CTG \$0.00												\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Central Office Tennininating Form																	
NHANCED EXTENDED LINK (EELs)	Central Office Termininating Point		**		UEPDC	CTG	\$0.00										
NRIANGEU EATENDEU LINA (EELS)	TANLANCED EXTENDED LINE (EEL -)		-														
	ENHANGED EXTENDED LINK (EELS)		-									1					

							RATES					oss	RATES		
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS DO	t in TRA ocket 97- Zo 01262	ne BC	s uso	:				ecurring	Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Increment Charge - Manual So Order vs Electronic-E
					Rec	Nonre	ecurring Add'l	Disc First	onnect Add'l	per LSR SOMEC	LSR SOMAN	Electronic-1st SOMAN	Electronic-Add'l SOMAN	1st SOMAN	Add'I SOMAN
NOTE: New (not currently combined) loop/transport (EEL) combinations are only available in density Zone 1 of the Nashville MSA in TN	4				Rec	riist	Addi	First	Addi	SOMEC	SUMAN	SOMAN	SOMAN	SOMAN	SOMAN
For Currently Combined EELs, network element recurring and Switch As Is Charges apply. For new EELs, network element recurring and nonrecurring apply (no Switch As Is Charge).															
2-WIRE VOICE GRADE EXTENDED LOOP WITH	H DEDICATED DS1 INTEROFFICE TR	ANSPOR	T (EEL)												
First 2-Wire Analog Voice Grade Loop - SL2/DS1 Interofficed Transport Combination Zone 1	Voice Grade with DS1 excluding	1	UNC	VX UEA	2 \$16.56	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	\$10.54
First 2-Wire Analog Voice Grade Loop - SL2/DS1 Interofficed Transport Combination Zone 2	Voice Grade with DS1 excluding	2	2 UNC	VX UEA	2 \$21.63	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	\$10.54
First 2-Wire Analog Voice Grade Loop - SL2/DS1 Interofficed Transport Combination Zone 3	Loop With DS1 Dedicated Interoffice Transport (First 2-Wire Voice Grade with DS1 excluding	3	3 UNC	VX UEA	2 \$28.28	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	\$10.54
Interoffice Transport - Dedicated - DS1 combination - Per Mile per month	Interoffice Transport Dedicated DS1 - Per Mile		UNC	1X 1L5X	X \$0.3562										
Interoffice Transport - Dedicated - DS1 Facility Termination in combination- Per Month			UNC	1X U1TI	1 \$77.8600	\$171.24	\$113.12	\$70.70	\$30.90			\$20.35	\$21.09	\$9.80	\$10.5
DS1 Channelization System in combination Per Month			UNC	1X MQ	\$80.7700	\$214.52	\$49.95	\$75.98	\$13.60						
Voice Grade COCI - DS1 to DS0 Interface in combination - per month  Each Additional 2-Wire Analog VG Loop-	Loop With DS1 Dedicated		UNC	VX 1D1\	G \$0.9100										
	Interoffice Transport (Additional 2- Wire Voice Grade Loop in same Loop With DS1 Dedicated	1	UNC	VX UEA	2 \$16.56	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$0.00	\$0.00	\$0.0
SL2 in the same DS1 Interoffice Transport Combination-Zone 2	Interoffice Transport (Additional 2- Wire Voice Grade Loop in same	2	2 UNC	VX UEA	2 \$21.63	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	\$10.5
Each Additional 2-Wire Analog VG Loop- SL2 in the same DS1 Interoffice Transport Combination-Zone 3	Interoffice Transport (Additional 2- Wire Voice Grade Loop in same	3	3 UNC	VX UEA	2 \$28.28	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$0.00	\$0.00	\$0.0
Each Additional VG Loop COCI - DS1 to DS0 Interface in combination - per month	Lance and cond Characteristics		UNC	VX 1D1\	G \$0.9100										
Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	Loop or Local Channel and Interoffice Combination - Switch-as is		UNC	1X UNC	:C	\$52.73	\$24.62	\$9.12	\$9.12						
4-WIRE VOICE GRADE EXTENDED LOOP WITH	H DEDICATED DS1 INTEROFFICE TO	ANSPOR	T (EEL)		+	1									-
	4-Wire Voice Grade Extended Loop With DS1 Dedicated Interoffice Transport (First 4-Wire 56 or 64 kbps Digital Grade Loop with DS1 excluding mileage)	- HOFOR	(122)												
	Zone 1	1	UNC	VX UEA	.4 \$24.70	\$108.76	\$35.47	\$72.04	\$10.86			\$20.35	\$21.09	\$9.80	\$10.

								RATES					oss	RATES		
													1	l I	Incremental	Incremental
	UNBUNDLED NETEORK ELEMENT AS	Not in TRA					1		<b>}</b>		Svc Order	Svc Order	Incremental	Incremental	Charge - Manual Svc	Charge - Manual Svc
UNBUNDLED NETWORK ELEMENT	STATED IN DOCKET 97-01262	Docket 97- 01262	Zone	BCS	USOC				Nonre	curring	Submitted	Submitted	Charge - Manual	Charge - Manual	Order vs.	Order vs.
		01202					1			-	Elec	Manually per	Svc Order vs.	Svc Order vs.	Electronic-Disc	Electronic-Disc
						Rec	First	curring Add'l	First	onnect Add'l	per LSR SOMEC	LSR SOMAN	Electronic-1st SOMAN	Electronic-Add'l SOMAN	1st SOMAN	Add'I SOMAN
	4-Wire Voice Grade Extended					Rec	Filst	Auu i	Filst	Addi	SOMEC	JOWAN	SOWAN	JOWAN	JOWAN	SOWAN
	Loop With DS1 Dedicated															
	Interoffice Transport (First 4-Wire															
	56 or 64 kbps Digital Grade Loop															
First 4-Wire Analog Voice Grade Loop/DS1	with DS1 excluding mileage) -															
Interoffice Transport Combination - Zone 2			2	UNCVX	UEAL4	\$32.25	\$108.76	\$35.47	\$72.04	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
	4-Wire Voice Grade Extended															
	Loop With DS1 Dedicated															
	Interoffice Transport (First 4-Wire 56 or 64 kbps Digital Grade Loop															
First 4-Wire Analog Voice Grade Loop/DS1	with DS1 excluding mileage) -															
Interoffice Transport Combination - Zone 3			3	UNCVX	UEAL4	\$42.17	\$108.76	\$35.47	\$72.04	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
Interoffice Transport - Dedicated - DS1	Interoffice Transport - Dedicated -		3	UNCVX	ULAL4	Ψ42.17	\$100.70	\$55.47	\$72.04	\$10.00			\$20.33	\$21.09	\$5.00	\$10.54
combination - Per Mile Per Month	DS1 - Per Mile			UNC1X	1L5XX	\$0.3562										
Interoffice Transport - Dedicated - DS1	2 22			J. 10 .X		\$0.000 <u>L</u>										
Facility Termination in combination- Per																
Month				UNC1X	U1TF1	\$77.8600	\$171.24	\$113.12	\$70.70	\$30.90			\$20.35	\$21.09	\$9.80	\$10.54
DS1 Channelization System in combination																
Per Month				UNC1X	MQ1	\$80.7700	\$214.52	\$49.95	\$75.98	\$13.60						
Voice Grade COCI - DS1 to DS0 Interface in	וו															
combination - per month				UNCVX	1D1VG	\$0.9100										
	4-Wire Voice Grade Extended															
Additional 4-Wire Analog Voice Grade Loop	Loop With DS1 Dedicated															
in same DS1 Interoffice Transport	Wire 56 or 64 kbps in same DS1															
Combination - Zone 1	excluding mileage) -Zone 1		1	UNCVX	UEAL4	\$24.70	\$108.76	\$35.47	\$72.04	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
Combination - Zone 1	4-Wire Voice Grade Extended			UNCVX	ULAL4	\$24.70	\$100.76	\$33.47	\$72.04	\$10.00			\$20.35	\$21.09	φ9.00	\$10.54
	Loop With DS1 Dedicated															
Additional 4-Wire Analog Voice Grade Loop																
in same DS1 Interoffice Transport	Wire 56 or 64 kbps in same DS1															
Combination - Zone 2	excluding mileage) -Zone 2		2	UNCVX	UEAL4	\$32.25	\$108.76	\$35.47	\$72.04	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
	4-Wire Voice Grade Extended									ì					,	
	Loop With DS1 Dedicated															
Additional 4-Wire Analog Voice Grade Loop																
in same DS1 Interoffice Transport	Wire 56 or 64 kbps in same DS1		_			_		_		_				_	_	_
Combination - Zone 3	excluding mileage) -Zone 3		3	UNCVX	UEAL4	\$42.17	\$108.76	\$35.47	\$72.04	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
Voice Grade COCI - DS1 to DS0 Interface in	1			LINIOVA	4041/0	<b>CO 0400</b>										
combination - per month	Non-Recurring Cost for Extended			UNCVX	1D1VG	\$0.9100	-									
	Loop or Local Channel and															
Nonrecurring Currently Combined Network	Interoffice combination Switch-as-															
Elements Switch -As-Is Charge	is			UNC1X	UNCCC		\$52.73	\$24.62	\$9.12	\$9.12						
							T									
4-WIRE 56 KBPS EXTENDED DIGITAL LOOP \	WITH DEDICATED DS1 INTEROFE	ICF TRAN	SPO	RT (FFI \			l l						l .			
CO NOI O EXTENDED DIGITAL ECON	4-Wire 56 or 64 kbps Extended		J. J.	(===)												
	Digital Loop With Dedicated DS1															
	Interoffice Transport (First 4-Wire															
	56 or 64 kbps Digital Grade Loop															
First 4-Wire 56Kbps Digital Grade Loop/DS																
Interoffice Transport Combination - Zone 1	Zone 1		1	UNCDX	UDL56	\$31.10	\$108.76	\$35.47	\$72.94	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
	4-Wire 56 or 64 kbps Extended															
	Digital Loop With Dedicated DS1															
	Interoffice Transport (First 4-Wire 56 or 64 kbps Digital Grade Loop						I		1							
First 4-Wire 56Kbps Digital Grade Loop/DS																
Interoffice Transport Combination - Zone 2			2	LINCDY	UDL56	\$40.76	\$108.76	\$35.47	\$72.94	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
interenice transport Combination - Zone z	2010 2	L		OINODY	ODESO	φ+∪./0	φ100.70	φυυ.41	φ12.94	φ10.00	1	<u> </u>	φ <b>∠</b> U.33	φ <u>2</u> 1.09	ψ5.0U	φ10.04

									RATES					oss	RATES		
																Incremental Charge -	Incremental Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS	Not in TRA Docket 97-	Zone	BCS	usoc						Svc Order	Svc Order	Incremental	Incremental	Manual Svc	Manual Svc
		STATED IN DOCKET 97-01262	01262							Nonre	ecurring	Submitted Elec	Submitted Manually per	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Order vs. Electronic-Disc	Order vs. Electronic-Disc
									curring		onnect	per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'l
		4-Wire 56 or 64 kbps Extended					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Digital Loop With Dedicated DS1															
		Interoffice Transport (First 4-Wire															
		56 or 64 kbps Digital Grade Loop															
	First 4-Wire 56Kbps Digital Grade Loop/DS			_			_								_		
	Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1	Zone 3 Interoffice Transport - Dedicated -		3	UNCDX	UDL56	\$53.11	\$108.76	\$35.47	\$72.94	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
	combination - Per Mile Per Month	DS1 - Per Mile			UNC1X	1L5XX	\$0.3562										
	Interoffice Transport - Dedicated - DS1	20 6			ONOTA	120701	ψ0.0002										
	Facility Termination in combination- Per																
	Month				UNC1X	U1TF1	\$77.8600	\$171.24	\$113.12	\$70.70	\$30.90			\$20.35	\$21.09	\$9.80	\$10.54
	DS1 Channelization System in combination Per Month				LINGAY	NO4	<b>*</b> 00 <b>77</b> 00	004450	640.05	A75.00	040.00						
	OCU-DP COCI (data) - DS1 to DS0 Channe	21			UNC1X	MQ1	\$80.7700	\$214.52	\$49.95	\$75.98	\$13.60						
	System - per month (2.4-64kbs)				UNCDX	1D1DD	\$1.82	\$6.07	\$4.66								
		4-Wire 56 or 64 kbps Extended					¥ · · · · ·	1	Ţ								
		Digital Loop With Dedicated DS1															
	Additional A Wiss FCKhas Digital Conds	Interoffice Transport (Additional 4-															
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport	Wire 56 or 64 kbps Digital Grade Loop in same DS1 excluding															
	Combination - Zone 1	mileage) -Zone 1		1	UNCDX	UDL56	\$31.10	\$108.76	\$35.47	\$72.94	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
	Combination Zone :	4-Wire 56 or 64 kbps Extended			ONODA	02200	ψ01.10	ψ100.70	ψ00.41	Ψ12.04	ψ10.00			Ψ20.00	Ψ21.00	ψ0.00	ψ10.04
		Digital Loop With Dedicated DS1															
	A LUC LA MC FOIGH Pi-ti-l O I-	Interoffice Transport (Additional 4-															
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport	Wire 56 or 64 kbps Digital Grade Loop in same DS1 excluding															
	Combination - Zone 2	mileage) -Zone 2		2	LINICDY	UDL56	\$40.76	\$108.76	\$35.47	\$72.94	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
	Combination - Zone Z	4-Wire 56 or 64 kbps Extended			UNCDA	ODLOG	φ40.76	\$100.70	\$33.41	\$12.54	ψ10.00			\$20.33	φ21.09	φ3.00	\$10.54
		Digital Loop With Dedicated DS1															
	_	Interoffice Transport (Additional 4-															
	Additional 4-Wire 56Kbps Digital Grade	Wire 56 or 64 kbps Digital Grade															
	Loopin same DS1 Interoffice Transport Combination - Zone 3	Loop in same DS1 excluding mileage) -Zone 3		3	UNCDX	UDL56	\$53.11	\$108.76	\$35.47	\$72.94	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
	OCU-DP COCI (data) - DS1 to DS0 Channe			-	UNCDX	ODLSO	φοσ.11	\$100.70	\$33.4 <i>1</i>	\$72.94	\$10.00			\$20.33	\$21.09	φ9.00	\$10.54
	System - per month (2.4-64kbs)				UNCDX	1D1DD	\$1.82	\$6.07	\$4.66								
		Nonrecurring Cost for Extended						*	•								
		Loop or Local Channel and															
	Nonrecurring Currently Combined Network	Interoffice Combination Switch-as-			11041/	LINICCO		650.70	004.00	00.40	<b>#0.40</b>						
	Elements Switch -As-Is Charge	IS			UNCTX	UNCCC		\$52.73	\$24.62	\$9.12	\$9.12						
4 14	VIRE 64 KBPS EXTENDED DIGITAL LOOP	WITH DEDICATED DO4 INTEROFF	CE TRAN	EPO!	) 								-	<b> </b>			<del>                                     </del>
4-V	VINE 04 RBF3 EXTENDED DIGITAL LOOP	4-Wire 56 or 64 kbps Extended	ICE IKAN	3701	(EEL)							-		1			<del>                                     </del>
		Digital Loop With Dedicated DS1															
		Interoffice Transport (First 4-Wire															
		56 or 64 kbps Digital Grade Loop															
	First 4-Wire 64Kbps Digital Grade Loop/DS					LIDLOA	****	<b>*</b> * * * * * * * * * * * * * * * * * *						***			
	Interoffice Transport Combination - Zone 1	Zone 1 4-Wire 56 or 64 kbps Extended		1	UNCDX	UDL64	\$31.10	\$108.76	\$35.47	\$72.94	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
		Digital Loop With Dedicated DS1															
		Interoffice Transport (First 4-Wire															
		56 or 64 kbps Digital Grade Loop															
	First 4-Wire 64Kbps Digital Grade Loop/DS				l	l		1									1
	Interoffice Transport Combination - Zone 2	Zone 2 4-Wire 56 or 64 kbps Extended		2	UNCDX	UDL64	\$40.76	\$108.76	\$35.47	\$72.94	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
		Digital Loop With Dedicated DS1															
		Interoffice Transport (First 4-Wire															
		56 or 64 kbps Digital Grade Loop															
	First 4-Wire 64Kbps Digital Grade Loop/DS																
	Interoffice Transport Combination - Zone 3	Zone 3		3	UNCDX	UDL64	\$53.11	\$108.76	\$35.47	\$72.94	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54

								RATES					oss	RATES		
UNBUNDLED NETWORK ELEMENT			Zone	BCS	usoc				Nonre	curring	Svc Order Submitted	Svc Order Submitted	Incremental Charge - Manual	Incremental Charge - Manual	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.
		01262					Nonre	currina		onnect	Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'l	Electronic-Disc	Electronic-Disc
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Interoffice Transport - Dedicated - DS1	Interoffice Transport - Dedicated -				41.500	******										
combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1	DS1 - Per Mile	-		UNC1X	1L5XX	\$0.3562										
Facility Termination in combination- Per																
Month				UNC1X	U1TF1	\$77.8600	\$171.24	\$113.12	\$70.70	\$30.90			\$20.35	\$21.09	\$9.80	\$10.54
DS1 Channelization System in combination Per Month				UNC1X	MQ1	\$80.7700	\$214.52	\$49.95	\$75.98	\$13.60						
OCU-DP COCI (data) - DS1 to DS0 Channel	1			ONOTA	IVIQI	ψ00.7700	Ψ214.02	Ψ-3.33	Ψ10.00	ψ13.00						
System - per month (2.4-64kbs)	4 Wiss 50 64 librar Friday			UNCDX	1D1DD	\$1.82	\$6.07	\$4.66								
Additional 4-Wire546Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 1	4-Wire 56 or 64 kbps Extended Digital Loop With Dedicated DS1 Interoffice Transport (Additional 4- Wire 56 or 64 kbps Digital Grade Loop in same DS1 excluding mileage) -Zone 1		1	UNCDX	UDL64	\$31.10	\$108.76	\$35.47	\$72.94	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
Additional 4-Wire546Kbps Digital Grade Loopin same DS1 Interoffice Transport	4-Wire 56 or 64 kbps Extended Digital Loop With Dedicated DS1 Interoffice Transport (Additional 4- Wire 56 or 64 kbps Digital Grade Loop in same DS1 excluding		0	LINODY	LIDI 64	040.70	0400.70	<b>005.47</b>	ф <b>т</b> о о 4	<b>#</b> 40.00			#00.0F	<b>#</b> 04.00	<b>*</b>	040.54
Combination - Zone 2	mileage) -Zone 2 4-Wire 56 or 64 kbps Extended		2	UNCDX	UDL64	\$40.76	\$108.76	\$35.47	\$72.94	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
Additional 4-Wire546Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 3	Digital Loop With Dedicated DS1 Interoffice Transport (Additional 4- Wire 56 or 64 kbps Digital Grade Loop in same DS1 excluding mileage) -Zone 3 Nonrecurring Cost for Extended		3	UNCDX	UDL64	\$53.11	\$108.76	\$35.47	\$72.94	\$10.86			\$20.35	\$21.09	\$9.80	\$10.54
Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	Loop or Local Channel and Interoffice Combination Switch-as- is			UNC1X	UNCCC		\$52.73	\$24.62	\$9.12	\$9.12						
2-WIRE VOICE GRADE DEDICATED EXTENDE	ED LOCAL CHANNEL WITH DEDICA	ATED DS1	INT	FROFFI	CF TRAN	SPORT (FFL)										
First 2-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 1	Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 excluding mileage) - Zone 1				ULDV2	\$17.18			\$72.94	440.00				\$21.09		\$10.54
First 2-Wire Local Channel with/DS1	Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 excluding					•	\$108.76	\$35.47		\$10.88			\$20.35		\$9.80	•
Interoffice Transport Combination - Zone 2  First 2-Wire Local Channel with/DS1	mileage) - Zone 2 Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 excluding		2	UNCVX	ULDV2	\$22.44	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	\$10.54
Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1	mileage) - Zone 3 Interoffice Transport - Dedicated -		3	UNCVX	ULDV2	\$29.34	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	\$10.54
combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 Facility Termination in combination- Per Month	DS1 - Per Mile			UNC1X		\$0.3562 \$77.8600	¢474.04	\$113.12	\$70.70	\$30.90			\$20.35	\$21.09	\$9.80	\$10.54
DS1 Channelization System in combination Per Month				UNC1X	MQ1	\$80.7700	\$171.24 \$214.52	\$113.12	\$70.70	\$30.90			Φ∠∪.30	φ∠1.U9	⊅ä.δU	φ1U.54
Voice Grade COCI - DS1 to DS0 Interface in combination - per month	n			UNCVX		\$0.9100	Ψ217.02	ψ-τυ.συ	ψ10.00	ψ10.00						

									RATES					oss	RATES		
			Not in TRA													Incremental Charge -	Incremental Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97-	Zone	BCS	usoc				Nonre	curring	Svc Order Submitted	Svc Order Submitted	Incremental Charge - Manual	Incremental Charge - Manual	Manual Svc Order vs.	Manual Svc Order vs.
			01262						currina		onnect	Elec per LSR	Manually per LSR	Svc Order vs.	Svc Order vs. Electronic-Add'l	Electronic-Disc	Electronic-Disc
							Rec	First	Add'l	First	Add'I	SOMEC	SOMAN	Electronic-1st SOMAN	SOMAN	SOMAN	SOMAN
	Additional 2-Wire Voice Grade Channel in Same DSI Interoffice Transport Combination	Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (Additional 2-Wire Voice) Grade Channel in same DS1															
	Zone 1	excluding mileage) -Zone 1 Extended 2-Wire Voice Grade		1	UNCVX	ULDV2	\$17.18	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	\$10.54
	Additional 2-Wire Voice Grade Channel in Same DSI Interoffice Transport Combination Zone 2	Dedicated Local Channel with Dedicated DS1 Interoffice Transport (Additional 2-Wire Voice		2	LINCVX	ULDV2	\$22.44	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	<b>\$</b> 10.54
	Additional 2-Wire Voice Grade Channel in Same DSI Interoffice Transport Combination Zone 3	Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (Additional 2-Wire Voice			UNCVX		\$29.34	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	\$10.54
	Voice Grade COCI - DS1 to DS0 Interface in				ONCVX	OLDVZ	Ψ23.34	ψ100.70	Ψ55.47	Ψ12.3 <del>4</del>	ψ10.00			Ψ20.55	Ψ21.03	ψ3.00	ψ10.5 <del>4</del>
	combination - per month	Nonrecurring Cost for Extended			UNCVX	1D1VG	\$0.9100										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	Loop or Local Channel and Interoffice combination Switch-as- is			UNC1X	UNCCC		\$52.73	\$24.62	\$9.12	\$9.12						
4-1	WIRE VOICE GRADE DEDICATED EXTENDE		ATED DS	1 INT	EROFFIC	CE TRAN	SPORT (EEL)										
	First 4-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 1	Extended 4-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 4-Wire Local Channel with DS1 excluding mileage) - Zone 1		1			\$18.18	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	\$10.54
	First 4-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 2	Extended 4-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 4-Wire Local Channel with DS1 excluding mileace) - Zone 2		2			\$23.75	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	\$10.54
	First 4-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 3	Extended 4-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 4-Wire Local Channel with DS1 excluding mileage) - Zone 3		3			\$31.05	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	\$10.54
	Interoffice Transport - Dedicated - DS1	Interoffice Transport - Dedicated -				1											
	combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1	DS1 - Per Mile			UNC1X	U1TF1	\$0.3562										
	Facility Termination in combination- Per Month				UNC1X	U1TF1	\$77.8600	\$171.24	\$113.12	\$70.70	\$30.90			\$20.35	\$21.09	\$9.80	\$10.54
	DS1 Channelization System in combination Per Month				UNC1X		\$80.7700	\$214.52	\$49.95	\$75.98	\$13.60			Ψ20.33	Ψ21.03	ψσ.00	ψ10.54
	Voice Grade COCI - DS1 to DS0 Interface in	n e						<del>+=:2</del>	Ţ.3.00	Ţ. 3.00	+ : 3:00	1					
	combination - per month				UNCVX	1D1VG	\$0.9100										

									RATES					oss	RATES		
			N													Incremental Charge -	Incremental Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97-	Zone	BCS	USOC			•			Svc Order	Svc Order Submitted	Incremental	Incremental	Manual Svc	Manual Svc
		STATED IN DOCKET 97-01262	01262							Nonre	ecurring	Submitted Elec	Manually per	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Order vs. Electronic-Disc	Order vs.
									curring		onnect	per LSR	LSR	Electronic-1st	Electronic-Add'l	1st SOMAN	Add'l
+							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Extended 4-Wire Voice Grade															
		Dedicated Local Channel with															
	Additional 4-Wire Voice Grade Channel in	Dedicated DS1 Interoffice Transport (Additional 4-Wire Voice															
	Same DSI Interoffice Transport Combination		1														
	Zone 1	excluding mileage) -Zone 1		1			\$18.18	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	\$10.54
		Extended 4-Wire Voice Grade Dedicated Local Channel with															
		Dedicated DS1 Interoffice															
	Additional 4-Wire Voice Grade Channel in	Transport (Additional 4-Wire Voice															
	Same DSI Interoffice Transport Combination																
	Zone 2	excluding mileage) -Zone 2 Extended 4-Wire Voice Grade		2			\$23.75	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	\$10.54
		Dedicated Local Channel with															
		Dedicated DS1 Interoffice															
	Additional 4-Wire Voice Grade Channel in	Transport (Additional 4-Wire Voice															
	Same DSI Interoffice Transport Combination			_			***				***			***	201.00		210 = 1
	Zone 3 Voice Grade COCI - DS1 to DS0 Interface in	excluding mileage) -Zone 3		3			\$31.05	\$108.76	\$35.47	\$72.94	\$10.88			\$20.35	\$21.09	\$9.80	\$10.54
	combination - per month				UNCVX	1D1VG	\$0.9100										
		Nonrecurring Cost for Extended															
	Name a surviva a Compania Compilia and National	Loop or Local Channel and Interoffice Combination Switch-as-															
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	is service Combination Switch-as-			LINC1X	UNCCC		\$52.73	\$24.62	\$9.12	\$9.12						
-	Elements Owten 7to 15 Charge	10			ONOTA	011000		Ψ02.70	ΨΣ-1.02	ψ5.12	ψ0.12						
4-V	VIRE DS1 DIGITAL EXTENDED LOOP WITH	DEDICATED DS1 INTEROFFICE	TRANSPO	RT (I	EEL)												
		Extended 4-Wire DS1 Digital Loop															
		With Dedicated DS1 Interoffice															
	4-Wire DS1 Digital Loop in Combination with	Transport (First 4-Wire DS1 Digita															
	DS1 Interoffice Transport - Zone 1	Zone 1		1	UNC1X	USLXX	\$57.73	\$228.40	\$161.74	\$79.87	\$24.88			\$20.35	\$21.09	\$9.60	\$10.54
	,	Extended 4-Wire DS1 Digital Loop															
		With Dedicated DS1 Interoffice															
	4-Wire DS1 Digital Loop in Combination with	Transport (First 4-Wire DS1 Digita															
	DS1 Interoffice Transport - Zone 2	Zone 2		2	UNC1X	USLXX	\$75.40	\$228.40	\$161.74	\$79.87	\$24.88			\$20.35	\$21.09	\$9.60	\$10.54
		Extended 4-Wire DS1 Digital Loop															
		With Dedicated DS1 Interoffice Transport (First 4-Wire DS1 Digita															
	4-Wire DS1 Digital Loop in Combination with																
	DS1 Interoffice Transport - Zone 3	Zone 3		3	UNC1X	USLXX	\$98.56	\$228.40	\$161.74	\$79.87	\$24.88			\$20.35	\$21.09	\$9.60	\$10.54
	Interoffice Transport - Dedicated - DS1	Interoffice Transport - Dedicated -															
	combination - Per Mile Per Month	DS1 - Per Mile			UNC1X	1L5XX	\$0.3562										
	Interoffice Transport - Dedicated - DS1												Ī				
	combination - Facility Termination Per Mon		<u> </u>		UNC1X	U1TF1	\$77.8300	\$171.24	\$113.12	\$70.07	\$30.90	<u>L</u>	<u></u>	\$20.35	\$21.05	\$9.60	\$10.54
		Nonrecurring Cost for Extended															
	Nonrecurring Currently Combined Network	Loop or Local Channel and Interoffice Combination Switch-as-															
	Elements Switch -As-Is Charge	is			UNC1X	UNCCC		\$52.73	\$24.62	\$9.12	\$9.12		Ī				
$\top$																	
Cu	rrently Combined Network Transport Elem	nents (Non-Switched Combinations	Resultin	g fro	m a Con	version)											
Lo	cal Channel - Dedicated - 2-Wire VG																
	Per Month - Zone 1		**	1	UNCVX	ULDV2	\$17.18										
1	Per Month - Zone 2				UNCVX	ULDV2	\$22.44										
-																	1
	Per Month - Zone 3			3	UNCVX	ULDV2	\$29.34										+

								RATES			OSS RATES Incremental Inc					
											Incremental Charge -					
	UNBUNDLED NETWORK ELEMENT  UNBUNDLED NETEORK ELEMENT AS	Not in TRA Docket 97-	Zone	BCS	USOC						Svc Order	Svc Order	Incremental	Incremental	Charge - Manual Svc	Charge - Manual Svc
	STATED IN DOCKET 97-01262	01262	Zone	ВСЗ	USUC				Nonre	ecurring	Submitted	Submitted	Charge - Manual	Charge - Manual	al Order vs.	Order vs.
							Nonre	curring	Disc	onnect	Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'l	Electronic-Disc 1st	Electronic-Dis Add'l
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
F	Per Month - Zone 1			UNCVX	ULDR2	\$17.18										
F	Per Month - Zone 2			UNCVX	ULDV2	\$22.44										
F	Per Month - Zone 3	**	3	UNCVX	ULDV2	\$29.34										
Loca	al Channel - Dedicated - 4-Wire VG															
1	Monthly Recurring per month	**		UNCVX	ULDV4	\$20.56										
Loca	al Channel - Dedicated - DS1															
Г	DS1 Monthly Recurring per month	**		UNC1X	ULDF1	\$20.56	\$201.53	\$24.83	\$55.52	\$5.51			\$20.35	\$10.54	\$13.30	
Inter	office Channel - Dedicated - 2-wire VG															
	nteroffice Channel - Dedicated 2-wire VG ·															
	per mile per month	**		UNCVX	1L5XX	\$0.0174										
	nteroffice Channel - Dedicated 2-wire VG	**			LIATIVO	010 =5										
+	Facility Termination per month	**		UNCVX	U1TV2	\$18.58										
Inter	office Channel - Dedicated - 2-wire VG Rev Battery nteroffice Channel - Dedicated 2-wire VG -															
	per mile per month	**	l	UNCVX	1L5XX	\$0.0174										
	nteroffice Channel - Dedicated 2-wire VG -			ONOVA	120701	Ψ0.017-7										
	Facility Termination per month	**		UNCVX	U1TR2	\$18.58										
Inte	roffice Channel - Dedicated - 56kbps															
	nteroffice Channel - Dedicated- 56 kbps ·															
	per mile per month	**		UNCDX	1L5XX	\$0.0174										
	nteroffice Channel - Dedicated - 56 kbps	**		LINODY	LIATOR	0.7										
	Facility Termination per month	**		UNCDX	U1TD5	\$17.98										
							-						-			
	roffice Channel - Dedicated - 64kbps nteroffice Channel - Dedicated - 64 kbps ·															
	per mile per month	**		UNCDX	1L5XX	\$0.0174										
	nteroffice Channel - Dedicated - 64 kbps			0.1027	120701	ψο.σ										
F	Facility Termination per month	**		UNCDX	U1TD6	\$17.98										
Inte	roffice Channel - Dedicated - DS1															
	nteroffice Channel - Dedicated - DS1 - per															
	mile per month	**		UNC1X	1L5XX	\$0.3562										
	Interoffice Channel - Dedicated - DS1 · Facility Termination per month	**		UNC1X	U1TF1	\$77.86										
++	acting remination per month		1	ONCIX	UIIFI	Φ11.00	1				1		1			
Local	al Loop - Dedicated - 2-Wire VG		<del>                                     </del>				1		1		1	1	1			
	Loop - Dedicated - 2-Wire VG -		<del>                                     </del>				1		1		1	1	1			
	Monthly Recurring per month - Zone 1	**	1	UNCVX	UEAL2	\$16.56										
L	Local Loop - Dedicated - 2-Wire VG -		Ī													
M	Monthly Recurring per month - Zone 2	**	2	UNCVX	UEAL2	\$21.63										
	Local Loop - Dedicated - 2-Wire VG -	**	١.			***										
<u> </u>	Monthly Recurring per month - Zone 3 Local Loop - Dedicated - 2-Wire VG -	**	3	UNCVX	UEAL2	\$28.28										
	Monthly Recurring per mile per month	**		UNCVX	1L5ND	\$0.00										
	normany recogning por mine por monar		<b>l</b>	5110VX	ILUIAD	ψυ.υυ			<b>†</b>		1	<b> </b>	1			
Loca	al Loop - Dedicated - 2-Wire VG - Rev Bat.		<del>                                     </del>				1		1		1		1			
L	Local Loop - Dedicated - 2-Wire VG - Rev		<b>1</b>								1	1				
	Bat - Monthly Recurring per month - Zone 1	**	1	UNCVX	UEAR2	\$16.56										
	Local Loop - Dedicated - 2-Wire VG - Rev		l . ¯													
	Bat - Monthly Recurring per month - Zone 2 Local Loop - Dedicated - 2-Wire VG - Rev	**	2	UNCVX	UEAR2	\$21.63										
1.0		1		1	1		i		1	1	1	1	I	1	l	1

			I		1			RATES					oss	RATES		
								101120							Incremental	Incremental
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS	Not in TRA	7	BCS	usoc						Svc Order	Svc Order	Incremental	Incremental	Charge - Manual Svc	Charge - Manual Svc
UNBUNDLED NETWORK ELEMENT	STATED IN DOCKET 97-01262	Docket 97- 01262	Zone	BCS	USOC				Nonre	curring	Submitted	Submitted	Charge - Manual	Charge - Manual	Order vs.	Order vs.
							Nonr	ecurring	Disco	nnect	Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'l	Electronic-Disc 1st	Electronic-Disc Add'l
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Local Loop - Dedicated - 4-Wire VG																
Local Loop - Dedicated - 4-Wire VG -		**			l											
Monthly Recurring per month - Zone 1 Local Loop - Dedicated - 4-Wire VG -			1	UNCVX	UEAL4	\$24.70										
Monthly Recurring per month - Zone 2		**	2	UNCVX	UEAL4	\$32.25										
Local Loop - Dedicated - 4-Wire VG -			É	ONOVA	OLAL	Ψ32.23										
Monthly Recurring per month - Zone 3		**	3	UNCVX	UEAL4	\$42.17										
Local Loop - Dedicated - 2-Wire ISDN Digital   Local Loop - Dedicated - 2-Wire ISDN -																
Local Loop - Dedicated - 2-Wire ISDN -																
Monthly Recurring per month - Zone 1		**	1	UNCVX	U1L2X	\$22.00										
Local Loop - Dedicated - 2-Wire ISDN - Monthly Recurring per month - Zone 2		**	2	UNCVX	U1L2X	\$29.02										
Local Loop - Dedicated - 2-Wire ISDN -				DINCVX	JILZA	φ29.UZ					1					
Monthly Recurring per month - Zone 3		**	3	UNCVX	U1L2X	\$37.95										
				1	1		1				1					
Local Loop - Dedicated - 4-Wire 56 kbps			l		1		1				1	1	1			
Local Loop - Dedicated - 4-Wire 56 kbps																
Digital - Monthly Recurring per month - Zone		**														
1   Local Loop - Dedicated - 4-Wire 56 kbps		**	1	UNCVX	UDL56	\$31.10										
Digital - Monthly Recurring per month - Zone																
2		**	2	UNCVX	UDL56	\$40.61										
Local Loop - Dedicated - 4-Wire 56 kbps			Ħ	O. TO TA	02200	ψ.σ.σ.										
Digital - Monthly Recurring per month - Zone																
3		**	3	UNCVX	UDL56	\$53.11										
Local Loop - Dedicated - 4-Wire 64 kbps																
Local Loop - Dedicated - 4-Wire 64 kbps Digital - Monthly Recurring per month - Zone																
1		**	1	UNCVX	UDL64	\$31.10										
Local Loop - Dedicated - 4-Wire 64 kbps			H	ONCVA	ODLOT	ψ51.10										
Digital - Monthly Recurring per month - Zone																
2		**	2	UNCVX	UDL64	\$40.61										
Local Loop - Dedicated - 4-Wire 64 kbps																
Digital - Monthly Recurring per month - Zone		**	2	UNCVX	UDL64	\$53.11										
3			3	UNCVA	UDL64	φ33.11										
Local Loop - Dedicated - DS1			1								1					
Local Loop - Dedicated - DS1 - Monthly			1	1	1						1	1	1			
Recurring per month - Zone 1		**	1	UNC1X	USLXX	\$57.73										
Local Loop - Dedicated - DS1 - Monthly						**					1					
Recurring per month - Zone 2		**	2	UNC1X	USLXX	\$75.40										
Local Loop - Dedicated - DS1 - Monthly		**	_													
Recurring per month - Zone 3		**	3	UNC1X	USLXX	\$98.59					1		1			
					-											
DS1 Channelization		**		11010111	1401	005										
DS1 to DS0 Channel System per month		**	<b>!</b>	UNC1X	MQ1	\$80.77										
OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)		**		UNCDX	1D1DD	\$1.82										
2-wire ISDN COCI (BRITE) - DS1 to DS0			l -	SINCEA	טטוטו	۷۱.0∠	<b> </b>				1		1			
Channel Systsem - per month		**		UNCNX	UC1CA	\$3.10										
Voice Grade COCI - DS1 to DS0 Channel				1			1				1					
System - per month		**		UNCVX	1D1VG	\$0.91										
MULTIPLEXERS			L		<u></u>											
DS3 to DS1 Channel System per month		**		UNC3X	MQ3	\$222.98										

## Unbundled Network Elements TENNESSEE

							RATES					oss	RATES		
														Incremental	Incremental
		Not in TRA												Charge -	Charge -
	UNBUNDLED NETWORK ELEMENT UNBUNDLED NETEORK ELEMENT AS	Docket 97- Z	Zone BCS	usoc						Svc Order	Svc Order	Incremental	Incremental	Manual Svc	Manual Svc
	STATED IN DOCKET 97-01262	01262						Nonre	curring	Submitted		Charge - Manual			Order vs.
										Elec	Manually per			Electronic-Disc	
						Nonre	curring	Disc	onnect	per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'l
					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	STS1 to DS1 Channel System per month	**	UNCSX	MQ3	\$222.98										
	DS3 Interface Unit (DS1 COCI) used with														
	Loop per month	**	UNC1X	UC1D1	\$17.58										
	Rates for these elements, products or services were not ordered by the TRA in Docket No. 97-01262	2. However, s	uch elements,	products or s	services are neces	ssary for the pr	ovision of other	elements, prod	ucts or services	for which perm	anent rates wer	re ordered.			
**	* Rates for these elements, products or services are related to a combination(s) ordered by the TRA i	in Docket No.	** Rates for these elements, products or services are related to a combination(s) ordered by the TRA in Docket No. 97-01262.												

## LOCAL INTERCONNECTION Tennessee

UNBUNDLED NETWORK ELEMENT								RATES					00.	SRATES		
UNBUNDLED NETWORK ELEMENT															Incremental	Incrementa
	DOCKET 97-01262	Not in TRA Docket 97- 01262	Zone	BCS	usoc				Noni	ecurring	Svc Order Sub-mitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs. Electronic-Disc	Charge - Manual Svo Order vs. Electronic-Di
						Rec	Nonre First	curring Add'l	Dis First	connect Add'l	per LSR SOMEC	LSR SOMAN	Electronic-1st SOMAN	Electronic-Add'l SOMAN	1st SOMAN	Add'I SOMAN
						Rec	FIRST	Addi	FIRST	Addi	SOMEC	SOMAN	SUMAN	SOMAN	SOMAN	SOMAN
LOCAL INTERCONNECTION (CALL TRANSPORT AN	ND TERMINATION)															
END OFFICE SWITCHING End Office Switching Function, Per MOL E	End office switcing functior					\$0.0008041										
TANDEM SWITCHING Tandem Switching Function Per MOL T	andem switching function					\$0.0009778										
LOCAL INTERCONNECTION (TRANSPORT)																
COMMON TRANSPORT (Shared)																
	Common transport - per mile, per MOL Common Transport - Facilities					\$0.0000064										
MOU T	ermination per MOU					\$0.0003871										
INTEROFFICE CHANNEL - DEDICATED TRANSP Interoffice Channel - Dedicated Transport - 2 Ir Wire Voice Grade - Per Mile per month Interoffice Channel - Dedicated Transport - 2			c	DHL, OH1MS	1L5NF	\$0.0174										
Wire Voice Grade - Facility Termination per Ir	nteroffice Transport - Dedicated - 2-Wire															
month V	oice Grade - Facility Termination		c	OHL, OH1MS	1L5NF	\$18.58	\$83.35									
INTEROFFICE CHANNEL - DEDICATED TRANSP																
Interoffice Channel - Dedicated Transport - 565	nteroffice Transport - Dedicated DSO -															
kbps - per mile per month D	Dedicated - DSO - per mile nteroffice Transport - Dedicated DSO -		c	OHL, OH1MS	1L5NK	\$0.0174										
Interoffice Channel - Dedicated Transport - 565	66/64 kbps - Interoffice Transport -															
i ir	Dedicated - DSO - Facility Termination interoffice Transport - Dedicated DSO -		C	OHL, OH1MS	1L5NK	\$17.98	\$83.35									
	Dedicated - DSO - per mile  nteroffice Transport - Dedicated DSO -		c	OHL, OH1MS	1L5NK	\$0.0174										
Interoffice Channel - Dedicated Transport - 645																
kbps - Facility Termination per month D	Dedicated - DSO - Facility Termination		d	OHL, OH1MS	1L5NK	\$17.98	\$83.35									
INTEROFFICE CHANNEL - DEDICATED TRANSPI																
	per mile		d	OH1, OH1MS	1L5NL	\$0.3562										
Interoffice Channel - Dedicated Tranport - DS Ir	nteroffice Transport - Dedicated - DS1 -															
- Facility Termination per month F	Facility Termination		ď	OH1, OH1MS	1L5NL	\$77.86	\$131.95									
LOCAL CHANNEL - DEDICATED TRANSPORT  Local Channel - Deldicated - 2-Wire Voice L	ocal Channel - Dedicated - 2-Wire Voice															
	Grade [shown here deaveraged]			OHL	TEFV2	\$17.18	\$254.14									
	ocal Channel - Dedicated - 4-Wire Voice															
	Grade [shown here deaveraged] Local Channel - Dedicated - Dedicated			OHL	TEFV4	\$18.18	\$257.05									
Local Channel - Dedicated - DS1 per month D				OH1	TEFHG	\$36.24	\$310.53									
MULTIPLEXERS																
	Channelization - Channel System DS1 to				CATNI	<b>₾</b> 00.77	<b>6400.04</b>									
Channelization - DS1 to DS0 Channel SystemD	DSO Channelization - Channel System DS3 to		ď	OH1, OH1MS	SATN1	\$80.77	\$186.34									
DS3 to DS1 Channel System per month D	DS1  nterface Unit - Interface DS3 to DS1		c	OH3. OH3MS OH1, OH1MS	SATNS SATCO	\$222.98 \$17.58	\$314.48 \$6.07						\$20.35	\$9.80	\$11.49	\$1.18

										RATES			OSS RATES					
		UNBUNDLED NETWORK ELEMENT		Not In TRA													Incremental Charge -	
			UNBUNDLED NETWORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97- 01262	Zone	BCS	USOC				Nonrecurring		Svc Order Submitted	Svc Order Submitted		Incremental Charge - Manual	Manual Svc Order vs.	
									Nonrecurring		Disconnect		Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st		Electronic-Disc 1st	
								Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	
ODU	F/ED	OUF/ADUF/CMDS																
	OPT	IONAL DAILY USAGE FILE (ODUF)																
		ODUF: Recording, per message	OSS OLEC Daily Usage File Recording per message				N/A	\$0.0000044										
		ODUF: Message Processing, per message	OSS OLEC Daily Usage File Message Distribution per message				N/A	\$0.0027366										
		ODUF: Message Processing, per Magnetic Tape provisioned	OSS OLEC Daily Usage File Message Distribution per magnetic tape provisioned				N/A	\$52.75										
		ODUF: Data Transmission (CONNECT:DIRECT), per message	OSS OLEC Daily Usage File Data Transmission (Connect: Direct), per message				N/A	\$0.0000339										

Incremental Charge -Manual Svc Order vs. Electronic-Disc Add'l SOMAN

```
2-WIRE ANALOG VOICE GRADE LOOP
```

4-WIRE ANALOG VOICE GRADE LOOP

2-WIRE ISDN DIGITAL GRADE LOOP

2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP

2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOF

4-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP

4-WIRE DS1 DIGITAL LOOP

4-WIRE 56 OR 64 KBPS DIGITAL GRADE LOOP

SUB-LOOP DISTRIBUTION

**SUB-LOOP FEEDER** 

**NETWORK INTERFACE DEVICE (NID)** 

UNBUNDLED LOOP CONCENTRATION

**UNBUNDLED SUB-LOOP CONCENTRATION (OUTSIDE CO)** 

UNBUNDLED LOCAL EXCHANGE SWITCHING(PORTS)

UNBUNDLED LOCAL SWITCHING, PORT USAGE

**COMMON TRANSPORT (SHARED)** 

INTEROFFICE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE

INTEROFFICE CHANNEL - DEDICATED TRANSPORT- 56/64 KBPS

**INTEROFFICE CHANNEL - DEDICATED TRANSPORT - DS1** 

LOCAL CHANNEL - DEDICATED TRANSPORT

**MULTIPLEXERS** 

**DARK FIBER** 

**8XX ACCESS TEN DIGIT SCREENING** 

LINE INFORMATION DATA BASE ACCESS (LIDB)

SIGNALING (CCS7)

SELECTIVE ROUTING

**AIN - BELLSOUTH AIN SMS ACCESS SERVICE** 

AIN - BELLSOUTH AIN TOOLKIT SERVICE

**OPTIONAL DAILY USAGE FILE (ODUF)** 

**UNBUNDLED PORT/LOOP COMBINATIONS - COST BASED RATES** 

2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)

2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)

2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)

2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)

2-WIRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT

2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT

4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT

4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT

2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

4-WIRE 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

4-WIRE 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

2-WIRE VOICE GRADE DEDICATED EXTENDED LOCAL CHANNEL WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

4-WIRE VOICE GRADE DEDICATED EXTENDED LOCAL CHANNEL WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

## Amendment to the Interconnection Agreement between Business Telecom, Inc. and BellSouth Telecommunications, Inc. Dated February 21, 2000

Pursuant to this Amendment ("Amendment"), Business Telecom, Inc. ("BTI"), collectively "BTI", a North Carolina corporation, as it is certificated in each state and identified in Appendix A hereto, and BellSouth Telecommunications, Inc. ("BellSouth"), hereinafter referred to collectively as the "Parties", hereby agree to amend the Interconnection Agreement between BellSouth and BTI dated February 21, 2000 ("Interconnection Agreement").

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, BTI and BellSouth hereby covenant and agree as follows:

1. The Parties agree to amend the October 20, 2000 Amendment to the Interconnection Agreement to revise the USOC for the non-recurring charge (NRC) for the 2-Wire Voice Grade Loop with 2-Wire DID Port – 1<sup>st</sup> and Add'l from USACC to USAC1 as reflected below:

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
Combination Rates										
NRC – 2–Wire Voice	USAC1	TBD	\$14.62	\$166.08	TBD	TBD	TBD	\$13.26	TBD	\$8.76
Grade Loop with 2-Wire										
DID Port – 1st										
NRC – 2-Wire Vice	USAC1	TBD	\$3.73	\$140.01	TBD	TBD	TBD	\$8.39	TBD	\$5.75
Grade Loop with 2-Wire										
DID Port – Add'l										

- 2. All other provisions of the Interconnection Agreement dated February 21, 2000 shall remain unchanged and in full force and effect until the expiration date.
- Either or both of the Parties is authorized to submit this Amendment to the appropriate regulatory agencies for approval subject to section 252(e) of the Federal Telecommunications Act of 1996.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

BellSouth Telecommunications, Inc.	Business Telecom, Inc.
By: Signature on File	By: Signature on File
Name: <u>Greg Follensbee</u>	Name: Anthony M. Copeland
Title: <u>Senior Director</u>	Title: <u>Executive Vice President/Gen Counsel</u>
Date: <u>03/18/02</u>	Date: <u>March 14, 2002</u>

## SCHEDULE OF BUSINESS TELECOM, INC.'S ("BTI") CERTIFICATED NAMES

Business Telecom, Inc. ("BTI, Inc.) [AL]

Business Telecom, Inc. d/b/a BTI [FL]

Business Telecom, Inc. ("BTI") [GA, KY, LA, MS, NC, SC]

Business Telecom, Inc. ("BTI") [resale - TN]

Business Telecom, Inc. ("BTI Telecommunications, Inc. 'BTI') [facilities - TN]